Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act).

YES □

The number of shares outstanding of the registrant's common shares, par value \$0.125 per share, was 297,733,061 as of April 20, 2018.

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

▼ OHADTEDLY DEDODT DUDGHANT TO SECTION 12 OD 15(4) OF THE

		HANGE ACT OF 1934	I) OF THE	
	, , ,	od ended March 31, 2018 OR		
		OK		
		NT TO SECTION 13 OR 15(c HANGE ACT OF 1934 d from to	I) OF THE	
	Commission Fi	e Number: 1-8944		
	→ C	LIFFS		
	_	D-CLIFFS INC. nt as Specified in Its Charter)		
C	Dhio		34-1464672	
,	er Jurisdiction of or Organization)		(I.R.S. Employer Identification No.)	
200 Public Squa	re, Cleveland, Ohio		44114-2315	
(Address of Princip	al Executive Offices)		(Zip Code)	
	Registrant's Telephone Number,	ncluding Area Code: (216) 69	4-5700	
	registrant (1) has filed all reports required shorter period that the registrant was red			
	YES ⊠	NO □		
	registrant has submitted electronically an Rule 405 of Regulation S-T (§232.405 o post such files).			
3	YES ⊠	NO □		
	e registrant is a large accelerated filer, definitions of "large accelerated filer," "ac			
Large accelerated filer	\boxtimes		Accelerated filer	
Non-accelerated filer	☐ (Do not check if a smaller reportin	g company)	Smaller reporting company	
			Emerging growth company	
	cate by check mark if the registrant has s provided pursuant to Section 13(a) of th		ded transition period for complying	with any new or

NO ⊠

TABLE OF CONTENTS

		Page Number
DEFINITION	s	1
PART I - FIN	IANCIAL INFORMATION	
Item 1.	Financial Statements	
	Statements of Unaudited Condensed Consolidated Financial Position as of March 31, 2018 and December 31, 2017	<u>2</u>
	Statements of Unaudited Condensed Consolidated Operations for the Three Months Ended March 31, 2018 and 2017	<u>4</u>
	Statements of Unaudited Condensed Consolidated Comprehensive Loss for the Three Months Ended March 31, 2018 and 2017	<u>5</u>
	Statements of Unaudited Condensed Consolidated Cash Flows for the Three Months Ended March 31, 2018 and 2017	<u>6</u>
	Notes to Unaudited Condensed Consolidated Financial Statements	<u>7</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>45</u>
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	<u>61</u>
Item 4.	Controls and Procedures	<u>62</u>
PART II - 01	THER INFORMATION	
Item 1.	Legal Proceedings	<u>63</u>
Item 1A.	Risk Factors	<u>63</u>
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	<u>63</u>
Item 4.	Mine Safety Disclosures	<u>64</u>
Item 5.	Other Information	<u>64</u>
Item 6.	Exhibits	<u>65</u>
Signatures		<u>66</u>

DEFINITIONS

The following abbreviations or acronyms are used in the text. References in this report to the "Company," "we," "us," "our" and "Cliffs" are to Cleveland-Cliffs Inc. and subsidiaries, collectively. References to "A\$" or "AUD" refer to Australian currency, "C\$" or "CAD" to Canadian currency and "\$" to United States currency.

Abbreviation or acronym	Term
A&R 2015 Equity Plan	Amended and Restated Cliffs Natural Resources Inc. 2015 Equity and Incentive Compensation Plan
ABL Facility	Amended and Restated Syndicated Facility Agreement by and among Bank of America, N.A., as Administrative Agent and Australian Security Trustee, the Lenders that are parties hereto, as the Lenders, Cleveland-Cliffs Inc., as Parent and a Borrower, and the Subsidiaries of Parent party hereto, as Borrowers dated as of March 30, 2015, and Amended and Restated as of February 28, 2018
Adjusted EBITDA	EBITDA excluding certain items such as impairment of inventory and long-lived assets, severance and retention costs, impacts of discontinued operations, foreign currency exchange remeasurement, and extinguishment of debt
ArcelorMittal	ArcelorMittal (as the parent company of ArcelorMittal Mines Canada, ArcelorMittal USA and ArcelorMittal Dofasco, as well as, many other subsidiaries)
ALJ	Administrative Law Judge
AMT	Alternative Minimum Tax
ASC	Accounting Standards Codification
ASU	Accounting Standards Update
Bloom Lake Group	Bloom Lake General Partner Limited and certain of its affiliates, including Cliffs Quebec Iron Mining ULC
Canadian Entities	Bloom Lake Group, Wabush Group and certain other wholly-owned Canadian subsidiaries
CCAA	Companies' Creditors Arrangement Act (Canada)
Dodd-Frank Act	Dodd-Frank Wall Street Reform and Consumer Protection Act
DR-grade	Direct Reduction-grade
EBITDA	Earnings before interest, taxes, depreciation and amortization
Empire	Empire Iron Mining Partnership
Exchange Act	Securities Exchange Act of 1934, as amended
FASB	Financial Accounting Standards Board
Fe	Iron
FERC	Federal Energy Regulatory Commission
FMSH Act	U.S. Federal Mine Safety and Health Act 1977, as amended
GAAP	Accounting principles generally accepted in the United States
HBI	Hot briquetted iron
Hibbing	Hibbing Taconite Company, an unincorporated joint venture
Koolyanobbing	Collective term for the operating deposits at Koolyanobbing, Mount Jackson and Windarling
Long ton	2,240 pounds
LTVSMC	LTV Steel Mining Company
Metric ton	2,205 pounds
MISO	
MMBtu	Midcontinent Independent System Operator, Inc. Million British Thermal Units
MSHA	
	U.S. Mine Safety and Health Administration
Monitor	FTI Consulting Canada Inc.
Net ton	2,000 pounds
Northshore	Northshore Mining Company
OPEB	Other postretirement employment benefits
Platts 62% Price	Platts IODEX 62% Fe Fines Spot Price
SEC	U.S. Securities and Exchange Commission
SG&A	Selling, general and administrative
Securities Act	Securities Act of 1933, as amended
SSR	System Support Resource
Tilden	Tilden Mining Company L.C.
Topic 606	ASC Topic 606, Revenue from Contracts with Customers
TSR	Total Shareholder Return
United Taconite	United Taconite LLC
U.S.	United States of America
U.S. Steel	U.S Steel Corporation and all subsidiaries
Wabush Group	Wabush Iron Co. Limited and Wabush Resources Inc., and certain of its affiliates, including Wabush Mines (an unincorporated joint venture of Wabush Iron Co. Limited and Wabush Resources Inc.), Arnaud Railway Company and Wabush Lake Railway Company

PART I

Item 1. Financial Statements

Statements of Unaudited Condensed Consolidated Financial Position

Cleveland-Cliffs Inc. and Subsidiaries

	(In Millions)		
	March 31, 2018		ecember 31, 2017
ASSETS			
CURRENT ASSETS			
Cash and cash equivalents	\$ 786.6	\$	1,007.7
Accounts receivable, net	47.2		140.6
Inventories	324.4		183.4
Supplies and other inventories	81.7		93.9
Derivative assets	93.6		39.4
Loans to and accounts receivable from the Canadian Entities	50.4		51.6
Other current assets	28.5		28.0
TOTAL CURRENT ASSETS	1,412.4		1,544.6
PROPERTY, PLANT AND EQUIPMENT, NET	1,047.3		1,051.0
OTHER ASSETS			
Deposits for property, plant and equipment	74.1		17.8
Income tax receivable	219.9		235.3
Other non-current assets	109.2		104.7
TOTAL OTHER ASSETS	 403.2		357.8
TOTAL ASSETS	\$ 2,862.9	\$	2,953.4

(continued)

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements .

Statements of Unaudited Condensed Consolidated Financial Position

Cleveland-Cliffs Inc. and Subsidiaries - (Continued)

	(I	(In Millions)		
	March 31, 2018	December 31, 2017		
LIABILITIES				
CURRENT LIABILITIES				
Accounts payable	\$ 99	9.5 \$ 127.7		
Accrued expenses	94	4.4 107.1		
Accrued interest	28	8.2 31.4		
Contingent claims	54	4.3 55.6		
Partnership distribution payable	44	4.2 44.2		
Other current liabilities	104	4.3 86.2		
TOTAL CURRENT LIABILITIES	424	4.9 452.2		
PENSION AND POSTEMPLOYMENT BENEFIT LIABILITIES	25 1	1.4 257.7		
ENVIRONMENTAL AND MINE CLOSURE OBLIGATIONS	181	1.2 196.5		
LONG-TERM DEBT	2,308	8.2 2,304.2		
OTHER LIABILITIES	182	2.0 186.9		
TOTAL LIABILITIES	3,347	7.7 3,397.5		
COMMITMENTS AND CONTINGENCIES (REFER TO NOTE 19)				
EQUITY				
CLIFFS SHAREHOLDERS' DEFICIT				
Preferred Stock - no par value				
Class A - 3,000,000 shares authorized				
Class B - 4,000,000 shares authorized				
Common Shares - par value \$0.125 per share				
Authorized - 600,000,000 shares (2017 - 600,000,000 shares);				
Issued - 301,886,794 shares (2017 - 301,886,794 shares);				
Outstanding - 297,733,061 shares (2017 - 297,400,968 shares)	37	7.7 37.7		
Capital in excess of par value of shares	3,918	8.0 3,933.9		
Retained deficit	(4,257	7.6) (4,207.3		
Cost of 4,153,733 common shares in treasury (2017 - 4,485,826 shares)	(151	1.8) (169.6		
Accumulated other comprehensive loss	(31	1.3) (39.0		
TOTAL CLIFFS SHAREHOLDERS' DEFICIT	(485	5.0) (444.3		
NONCONTROLLING INTEREST		0.2		
TOTAL DEFICIT	(484	4.8) (444.1		
TOTAL LIABILITIES AND DEFICIT	\$ 2,862	2.9 \$ 2,953.4		

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements .

Statements of Unaudited Condensed Consolidated Operations

Cleveland-Cliffs Inc. and Subsidiaries

	(In Mi	(In Millions, Except Per Share Amounts		
	Three Months Ended March 31,			ded
		2018		2017
REVENUES FROM PRODUCT SALES AND SERVICES				
Product	\$	220.7	\$	412.8
Freight and venture partners' cost reimbursements		18.3		48.8
		239.0		461.6
COST OF GOODS SOLD AND OPERATING EXPENSES		(242.6)		(365.3)
SALES MARGIN		(3.6)		96.3
OTHER OPERATING INCOME (EXPENSE)				
Selling, general and administrative expenses		(27.7)		(27.7)
Miscellaneous – net		(8.7)		11.5
		(36.4)	,	(16.2)
OPERATING INCOME (LOSS)		(40.0)		80.1
OTHER INCOME (EXPENSE)				
Interest expense, net		(33.5)		(42.8)
Loss on extinguishment of debt		_		(71.9)
Other non-operating income		4.4		2.5
		(29.1)		(112.2)
LOSS FROM CONTINUING OPERATIONS BEFORE INCOME TAXES		(69.1)		(32.1)
INCOME TAX BENEFIT (EXPENSE)		(15.7)		1.8
LOSS FROM CONTINUING OPERATIONS		(84.8)		(30.3)
INCOME FROM DISCONTINUED OPERATIONS, NET OF TAX		0.5		0.5
NET LOSS		(84.3)		(29.8)
LOSS ATTRIBUTABLE TO NONCONTROLLING INTEREST		_		1.7
NET LOSS ATTRIBUTABLE TO CLIFFS SHAREHOLDERS	\$	(84.3)	\$	(28.1)
LOSS PER COMMON SHARE ATTRIBUTABLE TO CLIFFS SHAREHOLDERS – BASIC				
Continuing operations	\$	(0.29)	\$	(0.11)
Discontinued operations	•	` _ ´		
	\$	(0.29)	\$	(0.11)
LOSS PER COMMON SHARE ATTRIBUTABLE TO CLIFFS SHAREHOLDERS – DILUTED	<u> </u>			,
Continuing operations	\$	(0.29)	\$	(0.11)
Discontinued operations	Ψ	(0.29)	Ψ	(0.11)
Discontinued operations	\$	(0.29)	\$	(0.11)
AVEDACE NUMBER OF CHARES (IN THOUSANDS)	Ψ	(0.29)	Ψ	(0.11)
AVERAGE NUMBER OF SHARES (IN THOUSANDS)		207.202		005.404
Basic		297,266		265,164
Diluted		297,266		265,164

The accompanying notes are an integral part of theseunaudited condensed consolidated financial statements.

Statements of Unaudited Condensed Consolidated Comprehensive Loss

Cleveland-Cliffs Inc. and Subsidiaries

	(In Millions) Three Months Ended March 31,			
	<u> </u>	2018		2017
NET LOSS ATTRIBUTABLE TO CLIFFS SHAREHOLDERS	\$	(84.3)	\$	(28.1)
OTHER COMPREHENSIVE INCOME (LOSS)				
Changes in pension and other post-retirement benefits, net of tax		6.7		4.7
Unrealized net gain (loss) on foreign currency translation		0.7		(12.7)
Unrealized net gain on derivative financial instruments, net of tax		0.3		_
OTHER COMPREHENSIVE INCOME (LOSS)		7.7		(8.0)
OTHER COMPREHENSIVE LOSS ATTRIBUTABLE TO THE NONCONTROLLING INTEREST				5.0
TOTAL COMPREHENSIVE LOSS ATTRIBUTABLE TO CLIFFS SHAREHOLDERS	\$	(76.6)	\$	(31.1)

 $The\ accompanying\ notes\ are\ an\ integral\ part\ of\ these\quad unaudited\ condensed\ consolidated\ financial\ statements\ .$

Statements of Unaudited Condensed Consolidated Cash Flows

Cleveland-Cliffs Inc. and Subsidiaries

	(In Millions)			
		Three Months Ended March 31,		
		2018		2017
OPERATING ACTIVITIES				
Net loss	\$	(84.3)	\$	(29.8)
Adjustments to reconcile net loss to net cash used by operating activities:				
Depreciation, depletion and amortization		23.9		23.2
Loss on extinguishment/restructuring of debt		_		71.9
Gain on derivatives		(40.8)		(17.7)
Other		25.9		0.8
Changes in operating assets and liabilities:				
Receivables and other assets		196.3		86.5
Inventories		(193.0)		(70.0)
Payables, accrued expenses and other liabilities		(70.9)		(90.0)
Net cash used by operating activities		(142.9)		(25.1)
INVESTING ACTIVITIES				
Purchase of property, plant and equipment		(12.4)		(25.9)
Deposits for property, plant and equipment		(59.0)		(2.0)
Other investing activities		_		0.5
Net cash used by investing activities		(71.4)	'	(27.4)
FINANCING ACTIVITIES				
Proceeds from issuance of debt		_		500.0
Debt issuance costs		(1.5)		(8.5)
Net proceeds from issuance of common shares		_		661.3
Repurchase of debt		_		(1,115.5)
Distributions of partnership equity		_		(8.7)
Other financing activities		(5.5)		(5.6)
Net cash provided (used) by financing activities		(7.0)	'	23.0
EFFECT OF EXCHANGE RATE CHANGES ON CASH		0.2		1.4
DECREASE IN CASH AND CASH EQUIVALENTS		(221.1)		(28.1)
CASH AND CASH EQUIVALENTS AT BEGINNING OF PERIOD		1,007.7		323.4
CASH AND CASH EQUIVALENTS AT END OF PERIOD	\$	786.6	\$	295.3

 $The\ accompanying\ notes\ are\ an\ integral\ part\ of\ these\quad unaudited\ condensed\ consolidated\ financial\ statements\ .$

Cleveland-Cliffs Inc. and Subsidiaries

Notes to Unaudited Condensed Consolidated Financial Statements

NOTE 1 - BASIS OF PRESENTATION AND SIGNIFICANT ACCOUNTING POLICIES

The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with SEC rules and regulations and, in the opinion of management, include all adjustments (consisting of normal recurring adjustments) necessary to present fairly the financial position, results of operations, comprehensive income (loss) and cash flows for the periods presented. The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. Management bases its estimates on various assumptions and historical experience, which are believed to be reasonable; however, due to the inherent nature of estimates, actual results may differ significantly due to changed conditions or assumptions. The results of operations for the three months ended March 31, 2018 are not necessarily indicative of results to be expected for the year ending December 31, 2018 or any other future period. These unaudited condensed consolidated financial statements should be read in conjunction with the financial statements and notes included in our Annual Report on Form 10-K for the year ended December 31, 2017.

On January 25, 2018, we announced that we would accelerate the projected time frame for the planned closure of our Asia Pacific Iron Ore mining operations in Australia. On April 6, 2018, we committed to a course of action expected to lead to the permanent closure of the Asia Pacific Iron Ore mining operations and expect our final Asia Pacific Iron Ore shipment to occur by June 30, 2018. Factors considered in this decision include increasingly discounted prices for lower-iron-content ore, the quality of the remaining iron ore reserves and the lack of a legitimate offer from a qualified buyer. As a result, we recorded various adjustments to *Inventories*, *Property*, *Plant and Equipment*, *Environmental and mine closure obligations* and *Supplies and other inventories* consistent with our current mine plan. Refer to NOTE 5 - INVENTORIES, NOTE 6 - PROPERTY, PLANT AND EQUIPMENT and NOTE 13 - ENVIRONMENTAL AND MINE CLOSURE OBLIGATIONS for further information.

We report our results from continuing operations in two reportable segments: U.S. Iron Ore and Asia Pacific Iron Ore.

Basis of Consolidation

The unaudited condensed consolidated financial statements include our accounts and the accounts of our wholly-owned subsidiaries, including the following operations as of March 31, 2018:

Name	Location	Status of Operations
Northshore	Minnesota	Active
United Taconite	Minnesota	Active
Tilden	Michigan	Active
Empire	Michigan	Indefinitely Idled
Koolyanobbing ¹	Western Australia	Active

¹ On April 6, 2018, we committed to a course of action expected to lead to the permanent closure of the Asia Pacific Iron Ore mining operations and expect our final Asia Pacific Iron Ore shipment to occur by June 30, 2018.

Intercompany transactions and balances are eliminated upon consolidation.

Equity Method Investments

Our 23% ownership interest in Hibbing is recorded as an equity method investment. As of March 31, 2018 and December 31, 2017, our investment in Hibbing was \$7.3 million and \$11.0 million, respectively, classified as *Other liabilities* in the Statements of Unaudited Condensed Consolidated Financial Position.

Foreign Currency

Our financial statements are prepared with the U.S. dollar as the reporting currency. The functional currency of our Australian subsidiaries is the Australian dollar. The functional currency of all other international subsidiaries is the U.S. dollar. The financial statements of our Australian subsidiaries are translated into U.S. dollars using the exchange rate at each balance sheet date for assets and liabilities and a weighted average exchange rate for each period for revenues, expenses, gains and losses. Translation adjustments are recorded as *Accumulated other comprehensive*

loss. Income taxes generally are not provided for foreign currency translation adjustments. To the extent that monetary assets and liabilities, including short-term intercompany loans, are recorded in a currency other than the functional currency, these amounts are remeasured each reporting period, with the resulting gain or loss being recorded in the Statements of Unaudited Condensed Consolidated Operations . Transaction gains and losses resulting from remeasurement of short-term intercompany loans are included in Miscellaneous – net in the Statements of Unaudited Condensed Consolidated Operations .

The following represents the transaction gains and losses resulting from remeasurement:

	(In Millions) Three Months Ended March 31,				
		2018		2017	
Short-term intercompany loans	\$	(0.2)	\$	15.1	
Cash and cash equivalents		0.1		(1.2)	
Other		(0.2)		(0.3)	
Net impact of transaction gains (losses) resulting from remeasurement	\$	(0.3)	\$	13.6	

Significant Accounting Policies

A detailed description of our significant accounting policies can be found in the audited financial statements for the fiscal year ended December 31, 2017 included in our Annual Report on Form 10-K filed with the SEC. There have been no material changes in our significant accounting policies and estimates from those disclosed therein other than those related to the adoption of Topic 606. Refer to NOTE 2 - NEW ACCOUNTING STANDARDS for further information.

NOTE 2 - NEW ACCOUNTING STANDARDS

Adoption of New Accounting Standards

ASC Topic 606, Revenue from Contracts with Customers (Topic 606). On January 1, 2018, we adopted Topic 606 and applied it to all contracts that were not completed using the modified retrospective method. We recognized the cumulative effect of initially applying Topic 606 as an adjustment to the opening balance of Retained deficit of \$34.0 million. The comparative period information has not been restated and continues to be reported under the accounting standards in effect for those periods. We do not expect that the adoption of Topic 606 will have a material impact to our annual net income on an ongoing basis.

Under Topic 606, revenue will generally be recognized upon delivery for our U.S. Iron Ore customers, which is earlier than under the previous guidance. As an example, for certain iron ore shipments where revenue was previously recognized upon title transfer when payment was received, we will now recognize revenue when control transfers, which is generally upon delivery. While we continue to retain title until we receive payment, we determined upon review of our customer contracts that the preponderance of control indicators pass to our customers' favor when we deliver our products; thus, we generally concluded control transfers at that point. As a result of the adoption of Topic 606 and vessel deliveries not occurring during the winter months because of the closure of the Soo Locks and the Welland Canal, our revenues and net income will be relatively lower than historical levels during the first quarter of each year and relatively higher than historical levels during the remaining three quarters in future years. However, the total amount of revenue recognized during the year should remain substantially the same as under previous accounting standards, assuming revenue rates and volumes are consistent between years.

The adoption of Topic 606 will not change the pattern or timing of revenue recognition for Asia Pacific Iron Ore, as control transfers when vessels are loaded, which is the same time title and the risk of loss transfers to our customers.

The cumulative effect of the changes made to our consolidated January 1, 2018 balance sheet for the adoption of Topic 606 were as follows:

	(\$ in Millions)					
		alance at nber 31, 2017	•	ments due to		ce at January 1, 2018
ASSETS					<u> </u>	
CURRENT ASSETS						
Cash and cash equivalents	\$	1,007.7	\$	_	\$	1,007.7
Accounts receivable, net		140.6		76.6		217.2
Inventories		183.4		(51.4)		132.0
Supplies and other inventories		93.9		_		93.9
Derivative assets		39.4		11.6		51.0
Loans to and accounts receivable from the Canadian Entities		51.6		_		51.6
Other current assets		28.0		_		28.0
TOTAL CURRENT ASSETS		1,544.6		36.8		1,581.4
PROPERTY, PLANT AND EQUIPMENT, NET		1,051.0		_		1,051.0
OTHER ASSETS						
Deposits for property, plant and equipment		17.8		_		17.8
Income tax receivable		235.3		_		235.3
Other non-current assets		104.7		_		104.7
TOTAL OTHER ASSETS		357.8		_		357.8
TOTAL ASSETS	\$	2,953.4	\$	36.8	\$	2,990.2
LIABILITIES CURRENT LIABILITIES						
Accounts payable	\$	127.7	\$	1.4	\$	129.1
Accrued expenses	Ψ	107.1	Ψ	- 1.4	Ψ	107.1
Accrued interest		31.4				31.4
Contingent claims		55.6		_		55.6
Partnership distribution payable		44.2				44.2
Other current liabilities		86.2		1.4		87.6
TOTAL CURRENT LIABILITIES		452.2		2.8		455.0
PENSION AND POSTEMPLOYMENT BENEFIT LIABILITIES		257.7		2.0		257.7
ENVIRONMENTAL AND MINE CLOSURE OBLIGATIONS		196.5				196.5
LONG-TERM DEBT		2,304.2		_		2,304.2
OTHER LIABILITIES		186.9				186.9
TOTAL LIABILITIES		3,397.5		2.8		3,400.3
EQUITY		3,391.5		2.0		3,400.3
CLIFFS SHAREHOLDERS' DEFICIT		(444.3)		34.0		(410.3)
NONCONTROLLING INTEREST		0.2		J4.0		0.2
TOTAL DEFICIT		(444.1)		34.0		(410.1)
	•		¢		•	2,990.2
TOTAL LIABILITIES AND DEFICIT	\$	2,953.4	\$	36.8	\$	2,990.2

The impact of adoption on our Statements of Unaudited Condensed Consolidated Operations and Statements of Unaudited Condensed Consolidated Financial Position is as follows:

	(\$ in Millions) Three Months Ended March 31, 2018						
					8		
	As Papartad		Balances without Adoption of Topic As Reported 606		Eff	Effect of Change	
REVENUES FROM PRODUCT SALES AND SERVICES		Reported				lect of Change	
Product	\$	220.7	\$	279.1	\$	(58.4)	
Freight and venture partners' cost reimbursements	Ψ	18.3	Ψ	22.4	Ψ	(4.1)	
Troight and voltare partitions soot formbal sometime		239.0		301.5	_	(62.5)	
COST OF GOODS SOLD AND OPERATING EXPENSES		(242.6)		(286.2)		43.6	
SALES MARGIN		(3.6)		15.3		(18.9)	
OTHER OPERATING EXPENSE		(0.0)		10.0		(10.0)	
Selling, general and administrative expenses		(27.7)		(27.7)		_	
Miscellaneous – net		(8.7)		(8.7)		_	
		(36.4)		(36.4)		_	
OPERATING LOSS		(40.0)		(21.1)		(18.9)	
OTHER INCOME (EXPENSE)		, ,		,		,	
Interest expense, net		(33.5)		(33.5)		_	
Other non-operating income		4.4		4.4		_	
		(29.1)		(29.1)	-	_	
LOSS FROM CONTINUING OPERATIONS BEFORE INCOME TAXES		(69.1)		(50.2)		(18.9)	
INCOME TAX EXPENSE		(15.7)		(15.7)		_	
LOSS FROM CONTINUING OPERATIONS		(84.8)		(65.9)		(18.9)	
INCOME FROM DISCONTINUED OPERATIONS, NET OF TAX		0.5		0.5		_	
NET LOSS		(84.3)		(65.4)		(18.9)	
LOSS ATTRIBUTABLE TO NONCONTROLLING INTEREST		_		_		_	
NET LOSS ATTRIBUTABLE TO CLIFFS SHAREHOLDERS	\$	(84.3)	\$	(65.4)	\$	(18.9)	
LOSS PER COMMON SHARE ATTRIBUTABLE TO CLIFFS SHAREHOLDERS – BASIC							
Continuing operations	\$	(0.29)	\$	(0.23)	\$	(0.06)	
Discontinued operations		_				_	
	\$	(0.29)	\$	(0.23)	\$	(0.06)	
LOSS PER COMMON SHARE ATTRIBUTABLE TO CLIFFS SHAREHOLDERS – DILUTED							
Continuing operations	\$	(0.29)	\$	(0.23)	\$	(0.06)	
Discontinued operations		_		_		_	
	\$	(0.29)	\$	(0.23)	\$	(0.06)	
AVERAGE NUMBER OF SHARES (IN THOUSANDS)							
Basic		297,266		297,266			
Diluted		297,266		297,266			

	(\$ in Millions)								
			March 31, 2018						
			Balances without Adoption of Topic						
	As	Reported	606	Effec	t of Change				
ASSETS									
CURRENT ASSETS									
Cash and cash equivalents	\$	786.6	\$ 786.6	\$	_				
Accounts receivable, net		47.2	24.9		22.3				
Inventories		324.4	332.0		(7.6)				
Supplies and other inventories		81.7	81.7		_				
Derivative assets		93.6	91.3		2.3				
Loans to and accounts receivable from the Canadian Entities		50.4	50.4		_				
Other current assets		28.5	28.5		_				
TOTAL CURRENT ASSETS		1,412.4	1,395.4		17.0				
PROPERTY, PLANT AND EQUIPMENT, NET		1,047.3	1,047.3		_				
OTHER ASSETS									
Deposits for property, plant and equipment		74.1	74.1		_				
Income tax receivable		219.9	219.9		_				
Other non-current assets		109.2	109.2		_				
TOTAL OTHER ASSETS		403.2	403.2						
TOTAL ASSETS		2,862.9	2,845.9		17.0				
LIABILITIES									
CURRENT LIABILITIES									
Accounts payable	\$	99.5	\$ 99.2	\$	0.3				
Accrued expenses	·	94.4	94.4	•	_				
Accrued interest		28.2	28.2		_				
Contingent claims		54.3	54.3		_				
Partnership distribution payable		44.2	44.2		_				
Other current liabilities		104.3	104.0		0.3				
TOTAL CURRENT LIABILITIES		424.9	424.3	•	0.6				
PENSION AND POSTEMPLOYMENT BENEFIT LIABILITIES		251.4	251.4		_				
ENVIRONMENTAL AND MINE CLOSURE OBLIGATIONS		181.2	181.2		_				
LONG-TERM DEBT		2,308.2	2,308.2		_				
OTHER LIABILITIES		182.0	182.0		_				
TOTAL LIABILITIES		3,347.7	3,347.1	_	0.6				
EQUITY		.,.	-7-						
CLIFFS SHAREHOLDERS' DEFICIT		(485.0)	(501.4)		16.4				
NONCONTROLLING INTEREST		0.2	0.2		_				
TOTAL DEFICIT		(484.8)	(501.2)		16.4				
TOTAL LIABILITIES AND DEFICIT	\$	2,862.9	\$ 2,845.9	\$	17.0				
	_ _	_,	-,- 1010	<u> </u>					

The adoption of Topic 606 did not have an impact on net cash flows in our Statements of Unaudited Condensed Consolidated Cash Flows .

ASU 2017-07, Retirement Benefits - Improving the Presentation of Net Periodic Pension Cost and Net Periodic Postretirement Benefit Cost . On January 1, 2018, we adopted the amendments to ASC 715 regarding the presentation of net periodic pension and postretirement benefit costs. We retrospectively adopted the presentation of service cost

separate from the other components of net periodic costs. The interest cost, expected return on assets, amortization of prior service costs, net remeasurement, and other costs have been reclassified from Cost of goods sold and operating expenses, Selling, general and administrative expenses and Miscellaneous – net to Other non-operating income. We elected to apply the practical expedient, which allows us to reclassify amounts disclosed previously in our Pension and other postretirement benefits footnote as the basis for applying retrospective presentation for comparative periods. On a prospective basis, only service costs will be included in amounts capitalized in inventory or property, plant, and equipment.

The effect of the retrospective presentation change related to the net periodic cost of our defined benefit pension and other postretirement employee benefits plans on our Statements of Unaudited Condensed Consolidated Operations was as follows:

	(\$ in Millions) Three Months Ended March 31, 2017							
		As Revised	Pre	eviously Reported	ed Effect of Change			
Cost of goods sold and operating expenses	\$	(365.3)	\$	(365.9)	\$	0.6		
Selling, general and administrative expenses	\$	(27.7)	\$	(25.7)	\$	(2.0)		
Miscellaneous - net	\$	11.5	\$	11.9	\$	(0.4)		
Operating income	\$	80.1	\$	81.9	\$	(1.8)		
Other non-operating income	\$	2.5	\$	0.7	\$	1.8		
Net Loss	\$	(29.8)	\$	(29.8)	\$	_		

Recent Accounting Pronouncements

Issued and Not Effective

In February 2016, the FASB issued ASU No. 2016-02, Leases (Topic 842). The new standard requires lessees to recognize a right-of-use asset and a lease liability on the balance sheet for all leases except for short-term leases. For lessees, leases will continue to be classified as either operating or finance leases in the Statements of Unaudited Condensed Consolidated Operations. We plan to adopt the standard on its effective date of January 1, 2019. The new standard may be adopted using either the modified retrospective approach, which requires application of the new guidance at the beginning of the earliest comparative period presented or the optional alternative approach, which requires application of the new guidance at the beginning of the standards effective date. We are currently finalizing our implementation plan, compiling an inventory of existing leases and evaluating the effect the updated standard will have on our consolidated financial statements and related disclosures.

NOTE 3 - SEGMENT REPORTING

Our continuing operations are organized and managed according to geographic location: U.S. Iron Ore and Asia Pacific Iron Ore. Our U.S. Iron Ore segment is a major supplier of iron ore pellets to the North American steel industry from our mines and pellet plants located in Michigan and Minnesota. The Asia Pacific Iron Ore segment is located in Western Australia and provides iron ore to the seaborne market for Asian steel producers. There were no intersegment revenues in the first quarter of 2018 or 2017.

We evaluate segment performance based on sales margin, defined as revenues less cost of goods sold and operating expenses identifiable to each segment. Additionally, we evaluate performance on a segment basis, as well as a consolidated basis, based on EBITDA and Adjusted EBITDA. These measures allow management and investors to focus on our ability to service our debt as well as illustrate how the business and each operating segment are performing. Additionally, EBITDA and Adjusted EBITDA assist management and investors in their analysis and forecasting as these measures approximate the cash flows associated with operational earnings.

The following tables present a summary of our reportable segments including a reconciliation of segment sales margin to Loss from Continuing Operations Before Income Taxes and a reconciliation of Net Loss to EBITDA and Adjusted EBITDA:

	(In Millions)								
		Three Months Ended March 31,							
		2018			2017				
Revenues from product sales and services:			,						
U.S. Iron Ore	\$	180.0	75%	\$	286.2	62%			
Asia Pacific Iron Ore		59.0	25%		175.4	38%			
Total revenues from product sales and services	\$	239.0	100%	\$	461.6	100%			
Sales margin:									
U.S. Iron Ore	\$	61.5		\$	49.0				
Asia Pacific Iron Ore		(65.1)			47.3				
Sales margin		(3.6)			96.3				
Other operating expense		(36.4)			(16.2)				
Other expense		(29.1)			(112.2)				
Loss from continuing operations before income taxes	\$	(69.1)		\$	(32.1)				

		(In Millions)				
	Thr	e Months E March 31,				
	2018		2017			
Net Loss	\$	84.3) \$	(29.8)			
Less:						
Interest expense, net		33.5)	(42.8)			
Income tax benefit (expense)		15.7)	1.8			
Depreciation, depletion and amortization		23.9)	(23.2)			
EBITDA	\$	11.2) \$	34.4			
Less:						
Inventory impairments	\$	18.9) \$	_			
Impairment of long-lived assets		(2.6)	_			
Severance and retention costs		(1.5)	_			
Impact of discontinued operations		0.5	0.5			
Foreign exchange remeasurement		(0.3)	13.6			
Loss on extinguishment of debt			(71.9)			
Adjusted EBITDA	\$	11.6 \$	92.2			
EBITDA						
U.S. Iron Ore	•	72.5 \$	57.9			
Asia Pacific Iron Ore		63.7)	51.4			
Other		20.0)	(74.9)			
Total EBITDA	\$	11.2) \$	34.4			
Adjusted EBITDA:						
U.S. Iron Ore	· ·	77.1 \$	64.1			
Asia Pacific Iron Ore		39.6)	53.8			
Other		25.9)	(25.7)			
Total Adjusted EBITDA	\$	11.6 \$	92.2			

		(In Million					
		Three Months Ended March 31,					
	201	18		2017			
Depreciation, depletion and amortization:							
U.S. Iron Ore	\$	15.8	\$	16.4			
Asia Pacific Iron Ore		6.7		4.7			
Other		1.4		2.1			
Total depreciation, depletion and amortization	\$	23.9	\$	23.2			
Capital additions ¹ :							
U.S. Iron Ore	\$	18.7	\$	27.1			
Asia Pacific Iron Ore		_		0.2			
Other ²		60.2					
Total capital additions	\$	78.9	\$	27.3			

¹ Includes cash paid for capital additions of \$71.4 million, including deposits of \$59.0 million, and an increase in non-cash accruals of \$7.5 million for the three months ended March 31, 2018 compared to cash paid for capital additions of \$27.9 million, including deposits of \$2.0 million, and a decrease in non-cash accruals of \$0.6 million for the three months ended March 31, 2017.

A summary of assets by segment is as follows:

		(In Millions)					
	M	arch 31, 2018	De	cember 31, 2017			
Assets:							
U.S. Iron Ore	\$	1,646.8	\$	1,500.6			
Asia Pacific Iron Ore		78.4		138.8			
Total segment assets		1,725.2		1,639.4			
Corporate and Other		1,137.7		1,314.0			
Total assets	\$	2,862.9	\$	2,953.4			

NOTE 4 - REVENUE

Revenue is recognized generally when iron ore is delivered to our customers. Revenue is measured at the point control transfers and represents the amount of consideration we expect to receive in exchange for transferring goods. We offer standard payment terms to our customers, generally requiring settlement within 30 days.

We enter into supply contracts of varying lengths to provide customers iron ore to use in their blast furnaces. Blast furnaces run continuously with a constant feed of iron ore and once shut down, cannot easily be restarted. As a result, we ship iron ore in large quantities for storage and use by customers at a later date. Customers do not simultaneously receive and consume the benefits of the iron ore. Based on our assessment of the factors that indicate the pattern of satisfaction, we transfer control of the iron ore at a point in time upon shipment or delivery of the product. The customer is able to direct the use of, and obtain substantially all of the benefits from, the product at the time the product is delivered.

We disaggregate Revenues from product sales and services based on geographical location. We sell a single product, iron ore, in the North American and Asian markets. Refer to NOTE 3 - SEGMENT REPORTING for further information on disaggregated revenue.

Certain of our U.S. Iron Ore and Asia Pacific Iron Ore customer supply agreements specify a provisional price, which is used for initial billing and cash collection. Revenue recorded in accordance with Topic 606 is calculated using the expected revenue rate at the point when control transfers. The final settlement includes market inputs for a specified period of time, which may vary by customer, but typically include one or more of the following: Platts 62% Price, pellet premiums, Platts international indexed freight rates and changes in specified Producer Price Indices, including industrial

² Includes capital additions related to our HBI project.

commodities, energy and steel. Changes in the expected revenue rate from the date control transfers through final settlement of contract terms is recorded in accordance with ASC Topic 815. Refer to NOTE 15 - DERIVATIVE INSTRUMENTS for further information on how our estimated expected and final revenue rates are determined.

A supply agreement with one U.S. Iron Ore customer provides for supplemental revenue or refunds based on the average annual daily market price for hot-rolled coil steel at the time the iron ore is consumed in the customer's blast furnaces. As control transfers prior to consumption, the supplemental revenue is recorded in accordance with ASC Topic 815. Refer to NOTE 15 - DERIVATIVE INSTRUMENTS for further information on supplemental revenue or refunds.

Included within *Revenues from product sales and services* is derivative revenue related to ASC Topic 815 of \$43.8 million and \$1.3 million, for three months ended March 31, 2018 at our U.S. Iron Ore and Asia Pacific Iron Ore segments, respectively.

Practical expedients and exemptions

We have elected to treat all shipping and handling costs as fulfillment costs as a significant portion of these costs are incurred prior to control transfer.

We have various long-term sales contracts with minimum purchase and supply requirement provisions that extend beyond the current reporting period. The portion of our transaction price for these contracts that is allocated entirely to wholly unsatisfied performance obligations is based on market prices that have not yet been determined and therefore is variable in nature. As such, we have not disclosed the value of unsatisfied performance obligations pursuant to the practical expedient.

Deferred Revenue

The table below summarizes our deferred revenue balances:

	Deferred Revenue (Current) ¹	De	ferred Revenue (Long-Term)
Opening balance as of January 1, 2018	\$ 23.8	\$	51.4
Closing balance as of March 31, 2018	31.0		51.4
Increase	\$ 7.2	\$	_

¹ The opening balance includes a \$1.4 million adjustment from the December 31, 2017 balance due to the adoption of Topic 606.

The terms of one of our U.S. Iron Ore pellet supply agreements required supplemental payments to be paid by the customer during the period 2009 through 2012, with the option to defer a portion of the 2009 monthly amount in exchange for interest payments until the deferred amount was repaid in 2013. Installment amounts received under this arrangement in excess of sales were classified as *Other current liabilities* and *Other liabilities* in the Statements of Unaudited Condensed Consolidated Financial Position upon receipt of payment. Revenue is recognized over the life of the supply agreement, which extends until 2022, in equal annual installments. As of March 31, 2018 and December 31, 2017, installment amounts received in excess of sales totaled \$64.2 million related to this agreement. As of March 31, 2018 and December 31, 2017, deferred revenue of \$12.8 million was recorded in *Other current liabilities* and \$51.4 million was recorded as long-term in *Other liabilities* in the Statements of Unaudited Condensed Consolidated Financial Position, related to this agreement.

Due to the payment terms and the timing of cash receipts near a period end, cash receipts can exceed shipments for certain customers. Revenue recognized on these transactions totaling \$18.2 million and \$9.6 million was deferred and included in *Other current liabilities* in the Statements of Unaudited Condensed Consolidated Financial Position as of March 31, 2018 and December 31, 2017, respectively.

NOTE 5 - INVENTORIES

The following table presents the detail of our *Inventories* in the Statements of Unaudited Condensed Consolidated Financial Position:

					(In Mill	ions)						
	March 31, 2018						December 31, 2017					
Segment	inished Goods	Work-in Process	Tota	I Inventory	Finished Goods Work-in Process			Total Inventory				
U.S. Iron Ore	\$ 267.2	\$	36.0	\$	303.2	\$	127.1	\$	11.3	\$	138.4	
Asia Pacific Iron Ore	20.2		1.0		21.2		33.3		11.7		45.0	
Total	\$ 287.4	\$	37.0	\$	324.4	\$	160.4	\$	23.0	\$	183.4	

We recorded lower of cost or net realizable value inventory charges of \$13.0 million and \$9.1 million related to finished goods inventory and work-in process inventory, respectively, at Asia Pacific Iron Ore in *Cost of goods sold and operating expenses* in the Statements of Unaudited Condensed Consolidated Operations for the three months ended March 31, 2018. The charges were a result of the decline in our expected realized revenue rates for future sales of these tons. There were no lower of cost or net realizable value inventory adjustments recorded for the three months ended March 31, 2017.

We recorded an impairment charge of \$1.4 million and \$13.2 million related to finished goods inventory and work-in process inventory, respectively, at Asia Pacific Iron Ore in *Cost of goods sold and operating expenses* in the Statements of Consolidated Operations for the three months ended March 31, 2018. Inventory not expected to be sold prior to the closure of operations was impaired. There were no inventory impairment adjustments recorded for the three months ended March 31, 2017.

NOTE 6 - PROPERTY, PLANT AND EQUIPMENT

The following table indicates the value of each of the major classes of our consolidated depreciable assets:

	(In Millions)				
	March 31, 2018			December 31, 2017	
Land rights and mineral rights	\$	549.6	\$	549.6	
Office and information technology		66.3		66.3	
Buildings		85.5		86.8	
Mining equipment		594.0		594.4	
Processing equipment		619.8		617.0	
Electric power facilities		57.0		57.0	
Land improvements		23.6		23.7	
Asset retirement obligation		16.9		19.2	
Other		30.3		30.3	
Construction in-progress		48.1		35.1	
		2,091.1		2,079.4	
Allowance for depreciation and depletion		(1,043.8)		(1,028.4)	
	\$	1,047.3	\$	1,051.0	

We recorded depreciation and depletion expense of \$21.3 million and \$22.6 million in the Statements of Unaudited Condensed Consolidated Operations for the three months ended March 31, 2018 and March 31, 2017, respectively.

As of March 31, 2018, based on the anticipated closure of the Asia Pacific Iron Ore operations we determined that we would not recover the value of certain long-lived assets at our Asia Pacific Iron Ore operations. As a result, we recorded an impairment of \$2.6 million in *Miscellaneous – net* in the Statements of Unaudited Condensed Consolidated Operations.

NOTE 7 - DEBT AND CREDIT FACILITIES

The following represents a summary of our long-term debt:

(In M	:111	anc'	١

	March	31, 2018	3				
Debt Instrument	Annual Effective Interest Rate		Principal nount	Issuance Costs	Unamortized Discounts	T	otal Debt
Secured Notes							
\$400 Million 4.875% 2024 Senior Notes	5.00%	\$	400.0	\$ (6.7)	\$ (2.5)	\$	390.8
Unsecured Notes							
\$400 Million 5.90% 2020 Senior Notes	5.98%		88.9	(0.2)	(0.1)		88.6
\$500 Million 4.80% 2020 Senior Notes	4.83%		122.4	(0.2)	(0.1)		122.1
\$700 Million 4.875% 2021 Senior Notes	4.89%		138.4	(0.3)	(0.1)		138.0
\$316.25 Million 1.50% 2025 Convertible Senior Notes	6.26%		316.3	(6.3)	(83.2)		226.8
\$1.075 Billion 5.75% 2025 Senior Notes	6.01%		1,075.0	(11.1)	(16.0)		1,047.9
\$800 Million 6.25% 2040 Senior Notes	6.34%		298.4	(2.3)	(3.4)		292.7
ABL Facility	N/A		450.0	N/A	N/A		_
Fair Value Adjustment to Interest Rate Hedge							1.3
Long-term debt						\$	2,308.2

(In Millions)

	Decembe	er 31,	2017						
Debt Instrument	Annual Effective Total Principal Debt Issuance Debt Instrument Interest Rate Amount Costs			Unamortized Discounts		-	Total Debt		
Secured Notes									
\$400 Million 4.875% 2024 Senior Notes	5.00%	\$	400.0	\$	(7.1)	\$	(2.6)	\$	390.3
Unsecured Notes									
\$400 Million 5.90% 2020 Senior Notes	5.98%		88.9		(0.2)		(0.1)		88.6
\$500 Million 4.80% 2020 Senior Notes	4.83%		122.4		(0.3)		(0.1)		122.0
\$700 Million 4.875% 2021 Senior Notes	4.89%		138.4		(0.3)		(0.1)		138.0
\$316.25 Million 1.50% 2025 Convertible Senior Notes	6.26%		316.3		(6.6)		(85.6)		224.1
\$1.075 Billion 5.75% 2025 Senior Notes	6.01%		1,075.0		(11.3)		(16.5)		1,047.2
\$800 Million 6.25% 2040 Senior Notes	6.34%		298.4		(2.4)		(3.4)		292.6
ABL Facility	N/A		550.0		N/A		N/A		_
Fair Value Adjustment to Interest Rate Hedge									1.4
Lona-term debt								\$	2,304.2

\$1.075 Billion 5.75% 2025 Senior Notes

On February 27, 2017, we entered into an indenture among the Company, the guarantors party thereto and U.S. Bank National Association, as trustee, relating to the issuance of \$500 million aggregate principal amount of 5.75% 2025 Senior Notes. On August 7, 2017, we issued an additional \$575 million aggregate principal amount of our 5.75% 2025 Senior Notes. The second tranche was issued at 97.0% of face value. The 5.75% 2025 Senior Notes were issued in private transactions exempt from the registration requirements of the Securities Act. Pursuant to the registration rights agreement executed as part of these issuances, we filed on February 14, 2018 a registration statement with the

SEC with respect to a registered offer to exchange the 5.75% 2025 Senior Notes for publicly registered notes, with all significant terms and conditions remaining the same.

Debt Maturities

The following represents a summary of our maturities of debt instruments based on the principal amounts outstanding at March 31, 2018:

	(In	Millions)
	Matur	ities of Debt
2018	\$	_
2019		_
2020		211.3
2021		138.4
2022		_
2023		_
2024 and thereafter		2,089.7
Total maturities of debt	\$	2,439.4

ABL Facility

On February 28, 2018, we entered into an amended and restated senior secured asset-based revolving credit facility with various financial institutions. The ABL Facility amends and restates our prior \$550.0 million Syndicated Facility Agreement, dated as of March 30, 2015. The ABL Facility will mature upon the earlier of February 28, 2023 or 60 days prior to the maturity of certain other material debt, and provides for up to \$450.0 million in borrowings, comprised of (i) a \$400.0 million U.S. tranche, including a \$248.8 million sublimit for the issuance of letters of credit and a \$100.0 million sublimit for U.S. swingline loans, and (ii) a \$50.0 million Australian tranche, including a \$24.4 million sublimit for the issuance of letters of credit and a \$20.0 million sublimit for Australian swingline loans. Availability under both the U.S. tranche and Australian tranche of the ABL Facility is limited to an eligible U.S. borrowing base and Australian borrowing base, as applicable, determined by applying customary advance rates to eligible accounts receivable, inventory and certain mobile equipment.

The ABL Facility and certain bank products and hedge obligations are guaranteed by us and certain of our existing wholly-owned U.S. and Australian subsidiaries and are required to be guaranteed by certain of our future U.S. and Australian subsidiaries; provided, however, that the obligations of any U.S. entity will not be guaranteed by any Australian entity. Amounts outstanding under the ABL Facility are secured by (i) a first-priority security interest in the accounts receivable and other rights to payment, inventory, as-extracted collateral, certain investment property, deposit accounts, securities accounts, certain general intangibles and commercial tort claims, certain mobile equipment, commodities accounts, deposit accounts, securities accounts and other related assets of ours, the other borrowers and the guarantors, and proceeds and products of each of the foregoing (collectively, the "ABL Collateral"); provided, however, that the ABL Collateral owned by a borrower or guarantor that is organized under the laws of Australia (the "Australian Loan Parties") shall only secure the Australian tranche and obligations of the borrowers and guarantors organized under the laws of Australia, (ii) a second-priority security interest in substantially all of our assets and the assets of the other borrowers and the guarantors (other than the Australian Loan Parties) other than the ABL Collateral (collectively, the "Notes Collateral" and, together with the ABL Collateral, the "Collateral") and (iii) solely in the case of the obligations of the Australian Loan Parties under the ABL Facility, a featherweight floating security interest over substantially all assets of the Australian Loan Parties other than ABL Collateral, in each case, subject to certain customary exceptions.

Borrowings under the ABL Facility bear interest, at our option, at a base rate, an Australian base rate or, if certain conditions are met, a LIBOR rate, in each case plus an applicable margin. The base rate is equal to the greatest of the federal funds rate plus ½ of 1%, the LIBOR rate based on a one-month interest period plus 1% and the floating rate announced by Bank of America Merrill Lynch as its "prime rate" and 1%. The Australian base rate is equal to the LIBOR rate as of 11:00 a.m. on the first business day of each month for a one-month period. The LIBOR rate is a per annum fixed rate equal to LIBOR with respect to the applicable interest period and amount of LIBOR rate loan requested.

The ABL Facility contains customary representations and warranties and affirmative and negative covenants including, among others, covenants regarding the maintenance of certain financial ratios if certain conditions are

triggered, covenants relating to financial reporting, covenants relating to the payment of dividends on, or purchase or redemption of, our capital stock, covenants relating to the incurrence or prepayment of certain debt, covenants relating to the incurrence of liens or encumbrances, covenants relating to compliance with laws, covenants relating to transactions with affiliates, covenants relating to mergers and sales of all or substantially all of our assets and limitations on changes in the nature of our business

The ABL Facility provides for customary events of default, including, among other things, the event of nonpayment of principal, interest, fees, or other amounts, a representation or warranty proving to have been materially incorrect when made, failure to perform or observe certain covenants within a specified period of time, a cross-default to certain material indebtedness, the bankruptcy or insolvency of the Company and certain of its subsidiaries, monetary judgment defaults of a specified amount, invalidity of any loan documentation, a change of control of the Company, and ERISA defaults resulting in liability of a specified amount. If an event of a default exists (beyond any applicable grace or cure period, if any), the administrative agent may and, at the direction of the requisite number of lenders, shall declare all amounts owing under the ABL Facility immediately due and payable, terminate such lenders' commitments to make loans under the ABL Facility. For certain events of default related to insolvency and receivership, the commitments of the lenders will be automatically terminated and all outstanding loans and other amounts will become immediately due and payable.

As of March 31, 2018 and December 31, 2017, we were in compliance with the ABL Facility liquidity requirements and, therefore, the springing financial covenant requiring a minimum fixed charge coverage ratio of 1.0 to 1.0 was not applicable.

As of March 31, 2018 and December 31, 2017, no loans were drawn under the ABL Facility and we had total availability of \$314.1 million and \$273.2 million, respectively, as a result of borrowing base limitations. As of March 31, 2018 and December 31, 2017, the principal amount of letter of credit obligations totaled \$46.6 million and \$46.5 million, respectively, to support business obligations primarily related to workers compensation and environmental obligations, thereby further reducing available borrowing capacity on our ABL Facility to \$267.5 million and \$226.7 million, respectively.

NOTE 8 - FAIR VALUE MEASUREMENTS

The following represents the assets and liabilities of the Company measured at fair value:

		(In Millions)											
		March 31, 2018											
Description	Quoted Prices in Active Markets for Identical Assets/Liabilities (Level 1)			Significant Other Observable Inputs (Level 2)		nt Unobservable Inputs Level 3)	Total						
Assets:													
Cash equivalents	\$	36.0	\$	490.6	\$	_	\$	526.6					
Derivative assets		_		_		93.6		93.6					
Total	\$	36.0	\$	490.6	\$	93.6	\$	620.2					
Liabilities:													
Derivative liabilities	\$	_	\$	0.2	\$	4.2	\$	4.4					
Total	\$	_	\$	0.2	\$	4.2	\$	4.4					

	(In Millions)										
	December 31, 2017										
Description	Markets Asset	Quoted Prices in Active Markets for Identical Assets/Liabilities (Level 1)		Significant Other Observable Inputs (Level 2)		ificant Unobservable Inputs (Level 3)		Total			
Assets:											
Cash equivalents	\$	66.3	\$	550.6	\$	_	\$	616.9			
Derivative assets		_		_		39.4		39.4			
Total	\$	66.3	\$	550.6	\$	39.4	\$	656.3			
Liabilities:											
Derivative liabilities	\$	_	\$	0.3	\$	2.4	\$	2.7			
Total	\$	_	\$	0.3	\$	2.4	\$	2.7			

Financial assets classified in Level 1 include money market funds and treasury bonds. The valuation of these instruments is based upon unadjusted quoted prices for identical assets in active markets.

The valuation of financial assets and liabilities classified in Level 2 is determined using a market approach based upon quoted prices for similar assets and liabilities in active markets or other inputs that are observable. Level 2 assets include commercial paper and certificates of deposit. Level 2 liabilities include commodity hedge contracts.

The Level 3 assets and liabilities include derivative assets that consist of freestanding derivative instruments related to certain supply agreements with one of our U.S. Iron Ore customers and derivative assets and liabilities related to certain provisional pricing arrangements with our U.S. Iron Ore and Asia Pacific Iron Ore customers.

The supply agreement included in our Level 3 assets includes provisions for supplemental revenue or refunds based on the average annual daily market price for hot-rolled coil steel at the time the iron ore product is consumed in the customer's blast furnaces. We account for these provisions as derivative instruments at the time of sale and adjust the corresponding asset or liability to fair value as an adjustment to *Product revenues* each reporting period until the product is consumed and the amounts are settled. The fair value of the instruments are determined using a market approach based on the estimate of the average annual daily market price for hot-rolled coil steel. This estimate takes into consideration current market conditions and nonperformance risk. We had assets of \$91.2 million and \$37.9 million at March 31, 2018 and December 31, 2017, respectively, related to the supply agreement.

The provisional pricing arrangements included in our Level 3 assets/liabilities specify provisional price calculations, where the pricing mechanisms generally are based on market pricing, with the final revenue rate to be based on market inputs at a specified point in time in the future, per the terms of the supply agreements. The difference between the estimated final revenue rate at the date of sale and the estimated final revenue rate at the measurement date is characterized as a derivative and is required to be accounted for separately once the revenue has been recognized. The derivative instrument is adjusted to fair value through *Product revenues* each reporting period based upon current market data and forward-looking estimates provided by management until the final revenue rate is determined. We had assets of \$2.4 million and \$1.5 million at March 31, 2018 and December 31, 2017, respectively, related to provisional pricing arrangements. In addition, we had liabilities of \$4.2 million and \$2.4 million related to provisional pricing arrangements at March 31, 2018 and December 31, 2017, respectively.

The following table illustrates information about quantitative inputs and assumptions for the assets and liabilities categorized in Level 3 of the fair value hierarchy:

Qualitative/Quantitative Information About Level 3 Fair Value Measurements

	(In Millions) Fair Value at March 31 2018		Balance Sheet Location	Valuation Technique	Unobservable Input	Range or Point Estimate (Weighted Average)
Customer supply agreements	\$	91.2	Derivative assets	Market Approach	Management's Estimate of Market Hot-Rolled Coil Steel per net ton	\$752
Provisional pricing arrangements	\$	2.4	Derivative assets	Market Approach	Management's Estimate of Platts 62% Price per dry metric ton	\$63 - \$71 (\$66)
Provisional pricing arrangements	\$	4.2	Other Current Liabilities	Market Approach	Management's Estimate of Platts 62% Price per dry metric ton	\$63 - \$71 (\$66)

The significant unobservable input used in the fair value measurement of our customer supply agreement is an estimate determined by management including the forward-looking estimate for the average annual daily market price for hot-rolled coil steel.

The significant unobservable inputs used in the fair value measurement of our provisional pricing arrangements are management's estimates of Platts 62% Price based upon current market data and index pricing, of which includes forward-looking estimates determined by management.

We recognize any transfers between levels as of the beginning of the reporting period, including both transfers into and out of levels. There were no transfers between Level 1 and Level 2 and no transfers into or out of Level 3 of the fair value hierarchy during the three months ended March 31, 2018 and 2017. The following tables represent a reconciliation of the changes in fair value of financial instruments measured at fair value on a recurring basis using significant unobservable inputs (Level 3):

	(In Mi	llions)				
	Level 3 Assets					
	Three Months Ended March 31,					
	2018		2017			
Beginning balance ¹	\$ 51.0	\$	31.6			
Total gains (losses)						
Included in earnings	49.1		42.1			
Settlements	(6.5)		(14.3)			
Ending balance - March 31	\$ 93.6	\$	59.4			
Total gains for the period included in earnings attributable to the change in unrealized gains on assets still held at the reporting date	\$ 44.5	\$	33.2			

¹ Beginning balance as of January 1, 2018 includes an \$11.6 million adjustment for adoption of Topic 606.

	 (In Mi	llions)				
	Level 3 Liabilities					
	Three Months Ended March 31,					
	2018		2017			
Beginning balance	\$ (2.4)	\$	(0.5)			
Total gains (losses)						
Included in earnings	(4.0)		(8.6)			
Settlements	2.2		_			
Ending balance - March 31	\$ (4.2)	\$	(9.1)			
Total losses for the period included in earnings attributable to the change in unrealized losses on liabilities still held at the reporting date	\$ (4.2)	\$	(9.1)			

The carrying amount of certain financial instruments (e.g., Accounts receivable, net, Accounts payable and Accrued expenses) approximates fair value and, therefore, has been excluded from the table below. A summary of the carrying amount and fair value of other financial instruments were as follows:

					(In Mi	llion	s)		
		March 31, 2018				December 31, 2			2017
	Classification	Carrying Value		Fair Value		Carrying Value		F	air Value
Long-term debt:									
Secured Notes									
\$400 Million 4.875% 2024 Senior Notes	Level 1	\$	390.8	\$	389.0	\$	390.3	\$	398.0
Unsecured Notes									
\$400 Million 5.90% 2020 Senior Notes	Level 1		88.6		89.1		88.6		88.0
\$500 Million 4.80% 2020 Senior Notes	Level 1		122.1		120.3		122.0		118.8
\$700 Million 4.875% 2021 Senior Notes	Level 1		138.0		135.4		138.0		130.8
\$316.25 Million 1.50% 2025 Convertible Senior Notes	Level 1		226.8		340.0		224.1		352.9
\$1.075 Billion 5.75% 2025 Senior Notes	Level 1		1,047.9		1,026.6		1,047.2		1,029.3
\$800 Million 6.25% 2040 Senior Notes	Level 1		292.7		251.1		292.6		227.1
ABL Facility	Level 2		_		_		_		_
Fair value adjustment to interest rate hedge	Level 2		1.3		1.3		1.4		1.4
Total long-term debt		\$	2,308.2	\$	2,352.8	\$	2,304.2	\$	2,346.3

The fair value of long-term debt was determined using quoted market prices based upon current borrowing rates.

Items Measured at Fair Value on a Non-Recurring Basis

The following tables present information about the financial assets and liabilities that were measured on a fair value basis. The tables also indicate the fair value hierarchy of the valuation techniques used to determine such fair value.

					(In I	Millions)				
					March	า 31, 2018				
Description	Ma Identi Li	ed Prices in Active rkets for cal Assets/ abilities .evel 1)		nificant Other ervable Inputs (Level 2)		Significant nobservable Inputs (Level 3)		Total		tal Year-to- Date Loss
Assets:										
Loans to and accounts receivables from the Canadian Entities	\$	_	\$	_	\$	50.4	\$	50.4	\$	(1.2)
Long-lived assets - Asia Pacific Iron Ore	\$	_	\$	_	\$	_	\$	_	\$	(2.6)
		(In Millions) December 31, 2017								
Description	Market / L	ed Prices in Active s for Identical Assets/ iabilities Level 1)	Significant Other			Significant Unobservable Inputs (Level 3)		Total	Total Year-to-Date Gains	
Assets:										
Loans to and accounts receivables from the Canadian Entities	\$	_	\$	_	\$	51.6	\$	51.6	\$	3.0
Liabilities:										
Guarantees	\$	_	\$	_	\$	_	\$	_	\$	31.4

To assess the fair value and recoverability of the accounts receivable from the Canadian Entities, we estimated the fair value of the underlying net assets of the Canadian Entities available for distribution to their creditors in relation to the estimated creditor claims and the priority of those claims. These underlying amounts are denominated primarily in Canadian dollars and are remeasured on a quarterly basis.

We determined the fair value and recoverability of our Canadian investments by comparing the estimated fair value of the remaining underlying assets of the Canadian Entities to remaining estimated liabilities. We recorded the Canadian denominated guarantees at book value, which best approximated fair value.

Our estimates involve significant judgment and are based on currently available information, an assessment of the validity of certain claims and estimated payments made by the Canadian Entities. Our ultimate recovery is subject to the final liquidation value of the Canadian Entities.

During the three months ended March 31, 2018, we recorded an impairment of \$2.6 million for our Asia Pacific Iron Ore reporting segment. Based on the anticipated closure of the Asia Pacific Iron Ore operations, we stated the value of these assets within *Property, plant and equipment* at their estimated fair value.

NOTE 9 - PENSIONS AND OTHER POSTRETIREMENT BENEFITS

We offer defined benefit pension plans, defined contribution pension plans and OPEB plans, primarily consisting of retiree healthcare benefits, to most employees in the U.S. as part of a total compensation and benefits program. We do not have employee retirement benefit obligations at our Asia Pacific Iron Ore operations. The defined benefit pension

plans largely are noncontributory and benefits generally are based on a minimum formula or employees' years of service and average earnings for a defined period prior to retirement.

On January 1, 2018, we adopted the amendments to ASC 715 regarding the presentation of net periodic pension and postretirement benefit costs. We retrospectively adopted the presentation of service cost separate from the other components of net periodic costs. Service costs are classified within Cost of goods sold and operating expenses, Selling, general and administrative expenses and Miscellaneous – net while the interest cost, expected return on assets, amortization of prior service costs, net remeasurement, and other costs are classified within Other non-operating income in our Statements of Unaudited Condensed Consolidated Operations.

The following are the components of defined benefit pension and OPEB costs and credits:

Defined Benefit Pension Costs

	(In Millions) Three Months Ended March 31,				
		2018		2017	
Service cost	\$	4.7	\$	4.8	
Interest cost		7.6		7.5	
Expected return on plan assets		(15.0)		(13.5)	
Amortization:					
Prior service costs		0.5		0.6	
Net actuarial loss		5.3		5.3	
Net periodic benefit cost	\$	3.1	\$	4.7	

Other Postretirement Benefits Credits

		(In Millions) Three Months Ended March 31,				
		2018		2017		
Service cost	\$	0.5	\$	0.5		
Interest cost		2.1		2.1		
Expected return on plan assets		(4.6)		(4.4)		
Amortization:						
Prior service credits		(8.0)		(0.7)		
Net actuarial loss		1.2		1.2		
Net periodic benefit credit	\$	(1.6)	\$	(1.3)		

Based on funding requirements, we made pension contributions of \$2.3 million for the three months ended March 31, 2018, compared to no pension contributions for the three months ended March 31, 2017. OPEB contributions are typically made on an annual basis in the first quarter of each year, but due to plan funding requirements being met, no OPEB contributions were required or made for the three months ended March 31, 2018 and March 31, 2017.

NOTE 10 - STOCK COMPENSATION PLANS

Employees' Plans

On February 21, 2018, the Compensation and Organization Committee of the Board of Directors approved grants under the A&R 2015 Equity Plan to certain officers and employees for the 2018 to 2020 performance period. Shares granted under the awards consisted of 0.7 million restricted stock units and 0.7 million performance shares.

Restricted stock units granted during 2018 are subject to continued employment, are retention based and are payable in common shares or cash at a time determined by the Compensation Committee at its discretion. The outstanding restricted stock units that were granted in 2018 cliff vest on December 31, 2020.

Performance shares are subject to continued employment, and each performance share, if earned, entitles the holder to be paid out in common shares or cash in certain circumstances. Performance is measured on the basis of relative TSR for the period of January 1, 2018 to December 31, 2020 and measured against the constituents of the S&P Metals and Mining ETF Index at the beginning of the relevant performance period. The final payouts for the outstanding performance period grants will vary from zero to 200% of the original grant depending on whether and to what extent the Company achieves certain objectives and performance goals as established by the Compensation Committee.

Determination of Fair Value

The fair value of each performance share grant is estimated on the date of grant using a Monte Carlo simulation to forecast relative TSR performance. A correlation matrix of historic and projected stock prices was developed for both the Company and our predetermined peer group of mining and metals companies. The fair value assumes that performance goals will be achieved.

The expected term of the grant represents the time from the grant date to the end of the service period. We estimate the volatility of our common shares and that of the peer group of mining and metals companies using daily price intervals for all companies. The risk-free interest rate is the rate at the grant date on zero-coupon government bonds with a term commensurate with the remaining life of the performance period.

The following assumptions were utilized to estimate the fair value for the 2018 performance share grant:

		Average						Fair Value (Percent of	
Grant Date	ant Date ket Price	Expected Term (Years)	Expected Volatility	Risk-Free Interest Rate	Dividend Yield	Fa	ir Value	Grant Date Market Price)	
February 21, 2018	\$ 7.53	2.86	86.8%	2.42%	—%	\$	11.93	158.43%	

NOTE 11 - INCOME TAXES

Our 2018 estimated annual effective tax rate before discrete items is approximately 0.1%. The annual effective tax rate differs from the U.S. statutory rate of 21% primarily due to the deductions for percentage depletion in excess of cost depletion related to U.S. operations and the reversal of valuation allowance from operations in the current year. The 2017 estimated annual effective tax rate before discrete items at March 31, 2017 was 5.4%.

For the three months ended March 31, 2018 and 2017, we recorded discrete items that resulted in an income tax expense of \$15.7 million and a benefit of \$0.1 million, respectively. The current year items relate primarily to a \$14.5 million reduction of the refundable AMT credit recorded in *Income tax receivable* in our Statements of Unaudited Condensed Consolidated Financial Position based on the sequestration guidance issued by the Internal Revenue Service during the period ended March 31, 2018. This \$14.5 million current year expense is a reduction of an asset and will not result in a cash tax outlay.

NOTE 12 - LEASE OBLIGATIONS

We lease certain mining, production and other equipment under operating and capital leases. The capital leases are for varying lengths, generally at market interest rates and contain purchase and/or renewal options at the end of the terms. Some capital lease payments could be accelerated upon cancellation of certain contracts at Asia Pacific Iron Ore. Our operating lease expense was \$1.6 million for the three months ended March 31, 2018, compared with \$1.7 million for the comparable period in 2017.

Future minimum payments under capital leases and non-cancellable operating leases as of March 31, 2018 are as follows:

		(In Millions)			
	Capital	Leases	Operati	ng Leases	
2018 (April 1 - December 31)	\$	14.7	\$	3.3	
2019		12.0		1.9	
2020		11.0		1.8	
2021		10.3		1.8	
2022		2.1		1.8	
2023 and thereafter		_		7.5	
Total minimum lease payments	\$	50.1	\$	18.1	
Amounts representing interest		7.6			
Present value of net minimum lease payments ¹	\$	42.5			

¹ The total is comprised of \$14.6 million and \$27.9 million classified as *Other current liabilities*, respectively, in the Statements of Unaudited Condensed Consolidated Financial Position as of March 31, 2018.

NOTE 13 - ENVIRONMENTAL AND MINE CLOSURE OBLIGATIONS

We had environmental and mine closure liabilities of \$202.5 million and \$200.1 million at March 31, 2018 and December 31, 2017, respectively. The following is a summary of the obligations:

		(In Millions)			
	Marc 20	h 31,)18	Dec	ember 31, 2017	
Environmental	\$	3.1	\$	2.9	
Mine closure					
U.S. Iron Ore ¹		170.7		168.4	
Asia Pacific Iron Ore		28.7		28.8	
Total mine closure		199.4		197.2	
Total environmental and mine closure obligations		202.5		200.1	
Less current portion		21.3		3.6	
Long-term environmental and mine closure obligations	\$	181.2	\$	196.5	

¹U.S. Iron Ore includes our active operating mines, our indefinitely idled Empire mine and a closed mine formerly operating as LTVSMC.

As of March 31, 2018, we reclassified \$17.7 million of our mine closure liability from long-term *Environmental and mine closure obligations* to *Other current liabilities* based on our plan to begin reclamation activities at Asia Pacific Iron Ore later this year.

Mine Closure

The accrued mine closure obligation for our active mining operations provides for contractual and legal obligations associated with the eventual closure of the mining operations. The accretion of the liability and amortization of the related asset is recognized over the estimated mine lives for each location.

The following represents a roll forward of our mine closure obligation liability for the three months ended March 31, 2018 and for the year ended December 31, 2017:

	(In Millions)			
	rch 31, 2018		ember 31, 2017	
Mine closure obligation at beginning of period	\$ 197.2	\$	204.0	
Accretion expense	2.7		14.9	
Remediation payments	(0.1)		(5.6)	
Exchange rate changes	(0.5)		1.5	
Revision in estimated cash flows	0.1		(17.6)	
Mine closure obligation at end of period	\$ 199.4	\$	197.2	

For the year ended December 31, 2017, the revision in estimated cash flows relates primarily to updates to our estimates resulting from our three-year in-depth review of our mine closure obligations for each of our U.S. mines. The primary driver of the decrease in estimated cash flows was the Empire mine, as the mine closure obligation was reduced \$26.2 million as a result of the refinement of the cash flows required for reclamation, remediation and structural removal. Prior estimates were based on RS Means (a common costing methodology used in the construction and demolition industry) costing data while the current estimate was compiled using a more detailed cost build-up approach. The overall decrease in estimated cash flows for our U.S. Iron Ore mines was offset partially by an increase in costs of \$10.1 million relating to the refinement of expected costs to be incurred at the end of life of mine at our Asia Pacific Iron Ore operations.

NOTE 14 - GOODWILL AND OTHER INTANGIBLE ASSETS

Goodwill

The carrying amount of goodwill as of March 31, 2018 and December 31, 2017 was \$2.0 million and related to our U.S. Iron Ore operating segment.

Other Intangible Assets

The following table is a summary of definite-lived intangible assets:

						(In Mi	llions)			
				Marc	h 31, 2018				De	cember 31, 2017	
	Classification	Ca	Gross arrying mount		umulated ortization	Net Carrying Amount		Gross Carrying Amount		Accumulated Amortization	Net Carrying Amount
Permits	Other non-current assets	\$	78.8	\$	(28.9)	\$ 49.9	\$	78.8	\$	(26.5)	\$ 52.3

Amortization expense relating to other intangible assets was \$2.6 million and \$0.6 million for the three months ended March 31, 2018 and 2017, respectively, and is recognized in *Cost of goods sold and operating expenses* in the Statements of Unaudited Condensed Consolidated Operations. Amortization expense of other intangible assets is expected to continue to be immaterial going forward.

NOTE 15 - DERIVATIVE INSTRUMENTS

The following table presents the fair value of our derivative instruments and the classification of each in the Statements of Unaudited Condensed Consolidated Financial Position:

						(In Mi	llions)								
			Derivativ	e Assets			Derivative Liabilities								
	March	31, 20	31, 2018 Decembe			, 2017	March	31, 2	.018	December 31, 2017					
Derivative Instrument	Balance Sheet Location		Fair Value	Balance Sheet Location		Fair Value	Balance Sheet Location		Fair Value	Balance Sheet Location		Fair Value			
Derivatives designated as hedging instruments under ASC 815:															
Commodity Contracts		\$	_		\$	_	Other current liabilities	\$	0.2	Other current liabilities	\$	0.3			
Derivatives not designated as hedging instruments under ASC 815:															
Customer supply agreements	Derivative assets	\$	91.2	Derivative assets	\$	37.9		\$	_		\$	_			
Provisional pricing arrangements	Derivative assets		2.4	Derivative assets		1.5	Other current liabilities		4.2	Other current liabilities		2.4			
Total derivatives not designated as hedging instruments under ASC 815		\$	93.6		\$	39.4		\$	4.2		\$	2.4			
Total derivatives		\$	93.6		\$	39.4		\$	4.4		\$	2.7			

Cash Flow Hedges

Commodity Contracts

As of March 31, 2018, we had outstanding natural gas hedge contracts for a notional amount of 3.5 million MMBtu in the form of forward contracts with varying maturity dates ranging from April 2018 to February 2019. As of December 31, 2017, we had outstanding natural gas hedge contracts for a notional amount of 3.5 million MMBtu in the form of forward contracts with varying maturity dates ranging from January 2018 to November 2018. Changes in fair value of highly effective hedges are recorded as a component of *Accumulated other comprehensive loss* in the Statements of Unaudited Condensed Consolidated Financial Position.

During the three months ended March 31, 2018, we recorded an unrealized gain of \$0.4 million in *Other comprehensive income (loss)* for changes in the fair value of these instruments and \$0.1 million has been reclassified from *Accumulated other comprehensive loss* into earnings. We had no commodity contracts designated as hedge instruments for the three months ended March 31, 2017.

Derivatives Not Designated as Hedging Instruments

Customer Supply Agreements

Most of our U.S. Iron Ore long-term supply agreements are comprised of a base price with annual price adjustment factors. The base price is the primary component of the purchase price for each contract. The indexed price adjustment factors are integral to the iron ore supply contracts and vary based on the agreement, but typically include adjustments based upon changes in the Platts 62% Price, along with pellet premiums, published Platts international indexed freight rates and changes in specified Producer Price Indices, including those for industrial commodities, fuel and steel. The pricing adjustments generally operate in the same manner, with each factor typically comprising a portion of the price adjustment, although the weighting of each factor varies based upon the specific terms of each agreement. In most cases, these adjustment factors have not been finalized at the time our product is sold. In these cases, we historically have estimated the adjustment factors at each reporting period based upon the best third-party information available. The estimates are then adjusted to actual when the information has been finalized. The price adjustment factors have been evaluated to determine if they contain embedded derivatives. The price adjustment factors share the same economic

characteristics and risks as the host contract and are integral to the host contract as inflation adjustments; accordingly, they have not been separately valued as derivative instruments.

A supply agreement with one U.S. Iron Ore customer provides for supplemental revenue or refunds to the customer based on the average annual daily steel market price for hot-rolled coil steel at the time the iron ore product is consumed in the customer's blast furnace. The supplemental pricing is characterized as a freestanding derivative and is required to be accounted for separately once the product is delivered. The derivative instrument, which is finalized based on a future price, is adjusted to fair value as a revenue adjustment each reporting period until the pellets are consumed and the amounts are settled.

We recognized net derivative revenue of \$41.9 million and \$17.8 million in *Product revenues* in the Statements of Unaudited Condensed Consolidated Operations for the three months ended March 31, 2018 and 2017, respectively, related to the supplemental payments. *Derivative assets*, representing the fair value of the supplemental revenue, were \$91.2 million and \$37.9 million as of March 31, 2018 and December 31, 2017 in the Statements of Unaudited Condensed Consolidated Financial Position, respectively.

Provisional Pricing Arrangements

Certain of our U.S. Iron Ore and Asia Pacific Iron Ore customer supply agreements specify provisional price calculations, where the pricing mechanisms generally are based on market pricing, with the final revenue rate based on certain market inputs at a specified period in time in the future, per the terms of the supply agreements. Market inputs are tied to indexed price adjustment factors that are integral to the iron ore supply contracts and vary based on the agreement. The pricing mechanisms typically include adjustments based upon changes in the Platts 62% Price, along with pellet premiums, published Platts international indexed freight rates and changes in specified Producer Price Indices, including those for industrial commodities, fuel and steel. The pricing adjustments generally operate in the same manner, with each factor typically comprising a portion of the price adjustment, although the weighting of each factor varies based upon the specific terms of each agreement.

Revenue is recognized generally when iron ore is delivered to our customers. Revenue is measured at the point control transfers and represents the amount of consideration we expect to receive in exchange for transferring goods. Changes in the expected revenue rate from the date control transfers through final settlement of contract terms is recorded in accordance with ASC Topic 815 and is characterized as a freestanding derivative and accounted for separately. Subsequently, the derivative instruments for both U.S. Iron Ore and Asia Pacific Iron Ore are adjusted to fair value through *Product revenues* each reporting period based upon current market data and forward-looking estimates provided by management until the final revenue rate is determined.

At March 31, 2018, we recorded \$2.4 million as *Derivative assets* and \$4.2 million as derivative liabilities classified as *Other current liabilities* related to our estimate of the final revenue rate with our U.S. Iron Ore and Asia Pacific Iron Ore customers in the Statements of Unaudited Condensed Consolidated Financial Position. At December 31, 2017, we recorded \$1.5 million as *Derivative assets* and \$2.4 million as derivative liabilities classified as *Other current liabilities* related to our estimate of the final revenue rate with our U.S. Iron Ore and Asia Pacific Iron Ore customers in the Statements of Unaudited Condensed Consolidated Financial Position. These amounts represent the difference between the amount we expect to receive when revenue is initially measured at the point control transfers and our subsequent estimate of the final revenue rate based on the price calculations established in the supply agreements. We recognized net increases of \$3.2 million and \$15.7 million in *Product revenues* in the Statements of Unaudited Condensed Consolidated Operations for the three months ended March 31, 2018 and 2017, respectively, related to these arrangements.

The following summarizes the effect of our derivatives that are not designated as hedging instruments in the Statements of Unaudited Condensed Consolidated Operations:

(In Millions)

Derivatives Not Designated as Hedging Instruments	Location of Income (Loss) Recognized on Derivative	Amount of Income (Loss) Recognized on Derivatives						
		 Three Mo Mar	nths Ei ch 31,	nded				
		 2018		2017				
Customer Supply Agreements	Product revenues	\$ 41.9	\$	17.8				
Provisional Pricing Arrangements	Product revenues	3.2		15.7				
Commodity Contracts	Cost of goods sold and operating expenses	_		(1.3)				
Total		\$ 45.1	\$	32.2				

Refer to NOTE 8 - FAIR VALUE MEASUREMENTS for additional information.

NOTE 16 - SHAREHOLDERS' DEFICIT

The following table reflects the changes in shareholders' deficit attributable to both us and the noncontrolling interests, primarily related to Tilden and Empire. We own 100% of both mines as of March 31, 2018 and 85% and 79% of each mine, respectively, as of March 31, 2017:

				(In Millions)	
	_	Cliffs Shareholders' Equity (Deficit)	ı	Noncontrolling Interest	Total Equity (Deficit)
December 31, 2017	\$	(444.3)	\$	0.2	\$ (444.1)
Adoption of accounting standard (Note 2)		34.0		_	34.0
Comprehensive loss					
Net loss		(84.3)		_	(84.3)
Other comprehensive income		7.7		_	7.7
Total comprehensive loss		(76.6)		_	(76.6)
Stock and other incentive plans		1.9		_	1.9
March 31, 2018	\$	(485.0)	\$	0.2	\$ (484.8)

	(In Millions)					
		Cliffs Shareholders' Equity (Deficit)		Noncontrolling Interest		Total Equity (Deficit)
December 31, 2016	\$	(1,464.3)	\$	133.8	\$	(1,330.5)
Comprehensive loss						
Net loss		(28.1)		(1.7)		(29.8)
Other comprehensive loss		(3.0)		(5.0)		(8.0)
Total comprehensive loss		(31.1)		(6.7)		(37.8)
Issuance of common shares		661.3		_		661.3
Stock and other incentive plans		4.0				4.0
March 31, 2017	\$	(830.1)	\$	127.1	\$	(703.0)

The following table reflects the changes in Accumulated other comprehensive loss related to Cliffs shareholders' deficit:

		(In I	Millions)	
	Changes in Pension and Other Post- Retirement Benefits, net of tax	Unrealized Net Gain on Foreign Currency Translation	Net Unrealized Gain (Loss) on Derivative Financial Instruments, net of tax	Accumulated Other Comprehensive Loss
December 31, 2017	\$ (263.9)	\$ 225.4	\$ (0.5)	\$ (39.0)
Other comprehensive income before reclassifications	0.5	0.7	0.4	1.6
Net loss (gain) reclassified from accumulated other comprehensive loss	6.2	_	(0.1)	6.1
March 31, 2018	\$ (257.2)	\$ 226.1	\$ (0.2)	\$ (31.3)

	(In Millions)						
	and (Retiren	es in Pension Other Post- nent Benefits, let of tax	(Loss	zed Net Gain) on Foreign cy Translation		accumulated Other omprehensive Loss	
December 31, 2016	\$	(260.6)	\$	239.3	\$	(21.3)	
Other comprehensive income (loss) before reclassifications		3.3		(12.7)		(9.4)	
Net loss reclassified from accumulated other comprehensive loss		6.4		_		6.4	
March 31, 2017	\$	(250.9)	\$	226.6	\$	(24.3)	

The following table reflects the details about Accumulated other comprehensive loss components related to Cliffs shareholders' deficit:

		(In Mi	llions)							
D. () () () ()	Amo	unt of (Gain)/Lo	ss Rec	lassified into	Aff. (11: 14 : 41 O) (
Details about Accumulated Other Comprehensive Loss Components		Three Mor Marc	ths End	led	Affected Line Item in the Statement of Unaudited Condensed Consolidated Operations					
		2018		2017						
Amortization of pension and OPEB liability:										
Prior service credits	\$	(0.3)	\$	(0.1)	Other non-operating income					
Net actuarial loss		6.5		6.5	Other non-operating income					
	\$	6.2	\$	6.4	Net of taxes					
Unrealized loss on derivative financial instruments:										
Commodity contracts	\$	(0.1)	\$		Cost of goods sold and operating expenses					
	\$	(0.1)	\$		Net of taxes					
Total reclassifications for the period, net of tax	\$	6.1	\$	6.4						

NOTE 17 - RELATED PARTIES

One of our four operating U.S. iron ore mines is a co-owned joint venture with companies that are integrated steel producers or their subsidiaries. We are the manager of such co-owned mine and rely on our joint venture partners to make their required capital contributions and to pay for their share of the iron ore pellets that we produce. Our joint venture partners are also our customers. The following is a summary of the mine ownership of the co-owned iron ore mine at March 31, 2018:

Mine	Cleveland-Cliffs Inc.	ArcelorMittal	U.S. Steel
Hibbing	23.0%	62.3 %	14.7 %

Product revenues from related parties were as follows:

	(In M	illions)		
	Three Mo Mar	nths E ch 31,	nded		
	 2018 2017				
Product revenues from related parties	\$ 62.1	\$	118.5		
Total product revenues	\$ 220.7	\$	412.8		
Related party product revenue as a percent of total product revenue	28.1%		28.7 %		

The following table presents the classification of related party assets and liabilities in the Statements of Unaudited Condensed Consolidated Financial Position:

	(In Millions)								
	Balance Sheet Location	Marc	h 31, 2018	Decem	ber 31, 2017				
Amounts due from related parties	Accounts receivable, net	\$	7.9	\$	68.1				
Customer supply agreements and provisional pricing agreements	Derivative assets		91.3		37.9				
Amounts due to related parties	Accounts payable		(1.2)		_				
Amounts due to related parties	Partnership distribution payable		(44.2)		(44.2)				
Amounts due to related parties	Other current liabilities		(0.4)		(12.3)				
Amounts due to related parties	Other liabilities		(42.0)		(41.4)				
Net amounts due from related parties		\$	11.4	\$	8.1				

During 2017, our ownership interest in Empire increased to 100% as we reached an agreement to distribute the noncontrolling interest net assets of \$132.7 million to ArcelorMittal, in exchange for its interest in Empire. The net assets were agreed to be distributed in three installments of \$44.2 million each, the first of which was paid upon the execution of the agreement and the remaining distributions are due in August 2018 and August 2019. The remaining two outstanding installments are reflected in *Partnership distribution payable* and *Other liabilities* in the Statements of Unaudited Condensed Consolidated Financial Position as of March 31, 2018.

A supply agreement with one U.S. Iron Ore customer provides for supplemental revenue or refunds to the customer based on the average annual daily market price for hot-rolled coil steel at the time the product is consumed in the customer's blast furnace. The supplemental pricing is characterized as a freestanding derivative. Refer to NOTE 15 - DERIVATIVE INSTRUMENTS for further information.

NOTE 18 - EARNINGS PER SHARE

The following table summarizes the computation of basic and diluted earnings (loss) per share:

	(In Millions, Ex	(In Millions, Except Per Share Amounts)					
	Three	Three Months Ended March 31,					
	2018		2017				
Loss from Continuing Operations	\$ (8	4.8) \$	(30.3)				
Loss from Continuing Operations Attributable to Noncontrolling Interest		_	1.7				
Net Loss from Continuing Operations Attributable to Cliffs Shareholders	\$ (8	4.8) \$	(28.6)				
Income from Discontinued Operations, net of tax		0.5	0.5				
Net Loss Attributable to Cliffs Shareholders	\$ (8	4.3) \$	(28.1)				
Weighted Average Number of Shares:							
Basic	29	7.3	265.2				
Employee Stock Plans		_	_				
Diluted	29	7.3	265.2				
Loss per Common Share Attributable to Cliffs Common Shareholders - Basic:							
Continuing operations	\$ (0	.29) \$	(0.11)				
Discontinued operations		_	_				
	\$ (0	.29) \$	(0.11)				
Loss per Common Share Attributable to Cliffs Common Shareholders - Diluted:							
Continuing operations	\$ (0	.29) \$	(0.11)				
Discontinued operations			_				
	\$ (0	.29) \$	(0.11)				

The diluted earnings per share calculation excludes 3.8 million and 4.6 million shares for the three months ended March 31, 2018 and 2017, respectively, related to equity plan awards that would have been anti-dilutive.

NOTE 19 - COMMITMENTS AND CONTINGENCIES

Contingencies

We are currently the subject of, or party to, various claims and legal proceedings incidental to our operations. If management believes that a loss arising from these matters is probable and can reasonably be estimated, we record the amount of the loss or the minimum estimated liability when the loss is estimated using a range, and no point within the range is more probable than another. As additional information becomes available, any potential liability related to these matters is assessed and the estimates are revised, if necessary. Based on currently available information, management believes that the ultimate outcome of these matters, individually and in the aggregate, will not have a material effect on our financial position, results of operations or cash flows. However, these claims and legal proceedings are subject to inherent uncertainties and unfavorable rulings could occur. An unfavorable ruling could include monetary damages, additional funding requirements or an injunction. If an unfavorable ruling were to occur, there exists the possibility of a material impact on the financial position and results of operations for the period in which the ruling occurs or future periods. However, we do not believe that any pending claims or legal proceedings will result in a material liability in relation to our consolidated financial statements.

Currently, we have recorded a liability in the Statements of Unaudited Condensed Consolidated Financial Position related to the following legal matters:

Michigan Electricity Matters. On February 19, 2015, in connection with various proceedings before FERC with respect to certain cost allocations for continued operation of the Presque Isle Power Plant in Marquette, Michigan, FERC issued an order directing MISO to submit a revised methodology for allocating SSR costs that identified the load serving entities that require the operation of SSR units at the power plant for reliability purposes. On September 17, 2015, FERC

issued an order conditionally approving MISO's revised allocation methodology. On September 22, 2016, FERC denied requests for rehearing of the February 19 order, rejecting arguments that FERC did not have the authority to order refunds in a cost allocation case and to impose retroactive surcharges to effectuate such refunds. FERC, however, suspended any refunds and surcharges pending its review of a July 25, 2016 ALJ initial decision on the appropriate amount of SSR compensation. This suspension was ultimately lifted after FERC's Order on Initial Decision of October 19, 2017, affirming in part and reversing in part certain aspects of the ALJ's decision and FERC's order on February 28, 2018, directing that refunds and surcharges be effectuated over a ten-month period beginning on the date of the order. Our current estimate of the potential liability to the Empire and Tilden mines is \$13.0 million in the aggregate, based on a schedule of anticipated surcharges (including interest) for the Escanaba, White Pine and Presque Isle SSRs from Empire and Tilden's electricity supplier. Separate from these SSR compensation issues, Tilden and Empire, along with various Michigan-aligned parties, had filed petitions for review regarding allocation and non-cost SSR issues with the U.S. Court of Appeals for the D.C. Circuit. Oral arguments on those issues were completed on April 6, 2018. We will continue to vigorously challenge the imposition of any retroactive SSR costs before the U.S. Court of Appeals for the D.C. Circuit. As of March 31, 2018, \$13.0 million is included in our Statements of Unaudited Condensed Consolidated Financial Position as part of Accrued expenses.

CCAA Proceedings

Effective January 27, 2015, following the commencement of CCAA proceedings for the Bloom Lake Group, we deconsolidated the Bloom Lake Group and certain other wholly-owned subsidiaries comprising substantially all of our Canadian operations. Additionally, on May 20, 2015, the Wabush Group commenced CCAA proceedings which resulted in the deconsolidation of the remaining Wabush Group entities that were not previously deconsolidated. As a result of this action, the CCAA protection granted to the Bloom Lake Group was extended to include the Wabush Group to facilitate the reorganization of each of their businesses and operations.

Prior to the deconsolidations, certain of our wholly-owned subsidiaries made loans to the Canadian Entities for the purpose of funding their operations and had accounts receivable generated in the ordinary course of business. The loans, corresponding interest and the accounts receivable were considered intercompany transactions and eliminated in our consolidated financial statements. Since the deconsolidations, the loans, associated interest and accounts receivable are considered related party transactions and have been recognized in our consolidated financial statements at their estimated fair value of \$50.4 million and \$51.6 million classified as *Loans to and accounts receivable from the Canadian Entities* in the Statements of Unaudited Condensed Consolidated Financial Position as of March 31, 2018 and December 31, 2017, respectively.

As of March 31, 2018, CCAA proceedings are ongoing and the majority of the assets of each of the Bloom Lake Group and the Wabush Group have been liquidated. The Monitor appointed by the court in the CCAA proceedings for the Bloom Lake Group and the Wabush Group has conducted a claims process pursuant to which creditors have filed claims against the Bloom Lake Group and the Wabush Group. The Monitor is reviewing all claims filed as part of this claims process. Currently, there is uncertainty as to the amount of the distribution that will be made to the creditors of the Bloom Lake Group and the Wabush Group, including, if any, to us, and whether we could be held liable for claims that may be asserted by or on behalf of the Bloom Lake Group or the Wabush Group or by their respective representatives against non-debtor affiliates of the Bloom Lake Group and the Wabush Group.

The net proceeds of sale of the assets of the Bloom Lake Group and the Wabush Group are currently being held by the Monitor. Certain of these funds will be utilized to fund the accrued and ongoing costs of the CCAA proceedings and the remaining funds will be available for distribution to the creditors of the Bloom Lake Group and the Wabush Group.

During 2017, we became aware that it was probable the Monitor will assert a preference claim against us and/or certain of our affiliates. We have an estimated liability of \$54.3 million, which includes the value of our related-party claims against the Bloom Lake Group and the Wabush Group, classified as Contingent claims in the Statements of Unaudited Condensed Consolidated Financial Position as of March 31, 2018.

During March 2018, we entered into a restructuring term sheet with the Bloom Lake Group and the Wabush Group, which documents the proposed terms of a plan of compromise or arrangement in the CCAA proceedings (the "Proposed Plan") to be sponsored by us as negotiated between us and the Monitor. This Proposed Plan requires both creditor and court approval.

Under the terms of this Proposed Plan, we and certain of our wholly owned subsidiaries have agreed to forego the benefit of any distributions or payments we may be entitled to receive as creditors of the Bloom Lake Group and the Wabush Group and to also make a C\$5.0 million cash contribution to the Bloom Lake Group and the Wabush Group for distribution to other creditors. It is important to note that the Proposed Plan, as currently drafted, will not resolve certain

employee claims which have been raised outside of the CCAA proceedings against us and certain of our affiliates and which will be addressed separately.

If this Proposed Plan is approved by a majority of the creditors of both the Bloom Lake Group and the Wabush Group, and is also approved by the court in the CCAA proceedings, and is implemented in accordance with its terms, then the Proposed Plan will resolve all of our claims against the Bloom Lake Group and the Wabush Group and all claims by the Bloom Lake Group, the Wabush Group and their respective creditors against us, except as noted above. The net financial impact of the Proposed Plan is materially consistent with amounts previously recorded in our financial statements.

However, if the creditors and the court do not approve the Proposed Plan, it is reasonably possible that future changes to our estimates and the ultimate amount paid on any claims could be material to our results of discontinued operations in future periods. We are not able to reasonably estimate such impact because there would be significant factual and legal issues to be resolved if the Proposed Plan is not approved. We will vigorously defend any claims if the Proposed Plan is not approved and implemented in accordance with its terms.

The motion to authorize the Bloom Lake Group and the Wabush Group to file the Proposed Plan and to schedule meetings of the creditors of the Bloom Lake Group and the Wabush Group to consider and vote to approve or reject the Proposed Plan was approved by the court on April 20, 2018.

NOTE 20 - SUBSEQUENT EVENTS

We have evaluated subsequent events through the date of financial issuance.

NOTE 21 - SUPPLEMENTARY GUARANTOR INFORMATION

The accompanying unaudited condensed consolidating financial information has been prepared and presented pursuant to SEC Regulation S-X, Rule 3-10, "Financial Statements of Guarantors and Issuers of Guaranteed Securities Registered or Being Registered." Certain of our subsidiaries (the "Guarantors") have guaranteed the obligations under the \$1.075 billion 5.75% 2025 Senior Notes issued by Cleveland-Cliffs Inc. See NOTE 7 - DEBT AND CREDIT FACILITIES for further information.

The following presents the unaudited condensed consolidating financial information for: (i) the Parent Company and the Issuer of the guaranteed obligations (Cleveland-Cliffs Inc.); (ii) the Guarantor subsidiaries, on a combined basis; (ivi) consolidating eliminations; and (v) Cleveland-Cliffs Inc. and subsidiaries on a consolidated basis. Each Guarantor subsidiary is 100% owned by the Parent Company as of March 31, 2018 and December 31, 2017. The unaudited condensed consolidating financial information is presented as if the Guarantor structure at March 31, 2018 existed for all periods presented. As a result, the Guarantor subsidiaries within the unaudited condensed consolidating financial information as of March 31, 2018 and December 31, 2017 and for the three months ended March 31, 2018 and 2017 include results of subsidiaries that were previously less than wholly-owned and were historically non-guarantors until 100% ownership was obtained.

Each of the Guarantor subsidiaries fully and unconditionally guarantee, on a joint and several basis, the obligations of Cleveland-Cliffs Inc. under the \$1.075 billion 5.75% 2025 Senior Notes. The guarantee of a Guarantor subsidiary will be automatically and unconditionally released and discharged, and such Guarantor subsidiary's obligations under the guarantee and the related indenture governing the \$1.075 billion 5.75% 2025 Senior Notes (the "Indenture") will be automatically and unconditionally released and discharged, upon:

- (a) any sale, exchange, transfer or disposition of such Guarantor subsidiary (by merger, consolidation, or the sale of) or the capital stock of such Guarantor subsidiary after which the applicable Guarantor subsidiary is no longer a subsidiary of the Company or the sale of all or substantially all of such Guarantor subsidiary's assets (other than by lease);
- (b) upon designation of any Guarantor subsidiary as an "excluded subsidiary" (as defined in the Indenture); and
- (c) upon defeasance or satisfaction and discharge of the Indenture.

Each entity in the unaudited consolidating financial information follows the same accounting policies as described in the consolidated financial statements. The accompanying unaudited condensed consolidating financial information has been presented on the equity method of accounting for all periods presented. Under this method, investments in subsidiaries are recorded at cost and adjusted for the subsidiaries' cumulative results of operations, capital contributions and distributions, and other changes in equity. Elimination entries include consolidating and eliminating entries for investments in subsidiaries, and intra-entity activity and balances.

Unaudited Condensed Consolidating Statement of Financial Position

As of March 31, 2018

/ln	мн	lion	٠١
ciri	IVIII	HOH	51

	Clev	eland-Cliffs Inc.		Guarantor Subsidiaries	Non-Guarantor Subsidiaries		Eliminations		Consolidated
ASSETS									
CURRENT ASSETS									
Cash and cash equivalents	\$	753.6	\$	1.0	\$	32.0	\$	_	\$ 786.6
Accounts receivable, net		5.7		27.7		15.9		(2.1)	47.2
Inventories		_		303.2		21.2		_	324.4
Supplies and other inventories		_		81.4		0.3		_	81.7
Derivative assets		_		93.6		_		_	93.6
Loans to and accounts receivable from the Canadian Entities		43.5		6.9		_		_	50.4
Other current assets		15.5		7.9		5.1		_	28.5
TOTAL CURRENT ASSETS		818.3		521.7		74.5		(2.1)	1,412.4
PROPERTY, PLANT AND EQUIPMENT, NET		16.2		956.9		74.2		_	1,047.3
OTHER ASSETS									
Deposits for property, plant and equipment		_		1.9		72.2		_	74.1
Income tax receivable		219.9		_		_		_	219.9
Investment in subsidiaries		1,185.7		27.4		_		(1,213.1)	_
Long-term intercompany notes		_		_		242.0		(242.0)	_
Other non-current assets		8.9		97.9		2.4		_	109.2
TOTAL OTHER ASSETS		1,414.5	_	127.2		316.6		(1,455.1)	403.2
TOTAL ASSETS	\$	2,249.0	\$	1,605.8	\$	465.3	\$	(1,457.2)	\$ 2,862.9
LIABILITIES					-				
CURRENT LIABILITIES									
Accounts payable	\$	4.6	\$	71.8	\$	25.2	\$	(2.1)	\$ 99.5
Accrued expenses		11.1		60.0		23.3			94.4
Accrued interest		28.2		_		_		_	28.2
Contingent claims		54.3		_		_		_	54.3
Partnership distribution payable		_		44.2		_		_	44.2
Other current liabilities		1.8		63.3		39.2		_	104.3
TOTAL CURRENT LIABILITIES		100.0		239.3		87.7		(2.1)	424.9
PENSION AND POSTEMPLOYMENT BENEFIT LIABILITIES		66.1		429.6		(244.3)		_	251.4
ENVIRONMENTAL AND MINE CLOSURE OBLIGATIONS		_		143.1		38.1		_	181.2
LONG-TERM DEBT		2,308.2		_		_		_	2,308.2
LONG-TERM INTERCOMPANY NOTES		242.0		_		_		(242.0)	_
OTHER LIABILITIES		17.5		142.9		21.6		_	182.0
TOTAL LIABILITIES		2,733.8		954.9		(96.9)		(244.1)	3,347.7
EQUITY									
TOTAL CLIFFS SHAREHOLDERS' DEFICIT		(484.8)		650.9		562.0		(1,213.1)	(485.0)
NONCONTROLLING INTEREST		_		_		0.2		_	0.2
TOTAL DEFICIT		(484.8)		650.9		562.2		(1,213.1)	(484.8)
TOTAL LIABILITIES AND DEFICIT	\$	2,249.0	\$	1,605.8	\$	465.3	\$	(1,457.2)	\$ 2,862.9
					_				

Unaudited Condensed Consolidating Statement of Financial Position

As of December 31, 2017

	Clev	eland-Cliffs Inc.	Guarantor Subsidiaries		Non-Guarantor Subsidiaries		Eliminations		Consolidated
ASSETS									
CURRENT ASSETS									
Cash and cash equivalents	\$	948.9	\$	2.1	\$	56.7	\$	_	\$ 1,007.7
Accounts receivable, net		4.5		102.9		33.9		(0.7)	140.6
Inventories		_		138.4		45.0		_	183.4
Supplies and other inventories		_		88.8		5.1		_	93.9
Derivative assets		_		37.9		1.5		_	39.4
Loans to and accounts receivable from the Canadian Entities		44.7		6.9		_		_	51.6
Other current assets		16.4		7.5		4.1		_	28.0
TOTAL CURRENT ASSETS		1,014.5		384.5		146.3		(0.7)	1,544.6
PROPERTY, PLANT AND EQUIPMENT, NET		17.5		959.0		74.5		_	1,051.0
OTHER ASSETS									
Deposits for property, plant and equipment		_		1.3		16.5		_	17.8
Income tax receivable		235.3		_		_		_	235.3
Investment in subsidiaries		1,024.3		29.9		_		(1,054.2)	_
Long-term intercompany notes		_		_		242.0		(242.0)	_
Other non-current assets		7.8		91.7		5.2		_	104.7
TOTAL OTHER ASSETS		1,267.4		122.9		263.7		(1,296.2)	357.8
TOTAL ASSETS	\$	2,299.4	\$	1,466.4	\$	484.5	\$	(1,296.9)	\$ 2,953.4
LIABILITIES					_				
CURRENT LIABILITIES									
Accounts payable	\$	7.1	\$	89.7	\$	31.6	\$	(0.7)	\$ 127.7
Accrued expenses		19.0		59.9		28.2			107.1
Accrued interest		31.4		_		_		_	31.4
Contingent claims		55.6		_		_		_	55.6
Partnership distribution payable		_		44.2		_		_	44.2
Other current liabilities		2.1		63.5		20.6		_	86.2
TOTAL CURRENT LIABILITIES		115.2		257.3	_	80.4		(0.7)	452.2
PENSION AND POSTEMPLOYMENT BENEFIT LIABILITIES		66.4		430.6		(239.3)		_	257.7
ENVIRONMENTAL AND MINE CLOSURE OBLIGATIONS		_		140.6		55.9		_	196.5
LONG-TERM DEBT		2,304.2		_		_		_	2,304.2
LONG-TERM INTERCOMPANY NOTES		242.0		_		_		(242.0)	_
OTHER LIABILITIES		15.7		147.2		24.0		_	186.9
TOTAL LIABILITIES		2,743.5		975.7	_	(79.0)		(242.7)	3,397.5
EQUITY									
TOTAL CLIFFS SHAREHOLDERS' DEFICIT		(444.1)		490.7		563.3		(1,054.2)	(444.3)
NONCONTROLLING INTEREST		_		_		0.2		_	0.2
TOTAL DEFICIT		(444.1)		490.7	_	563.5		(1,054.2)	(444.1)
TOTAL LIABILITIES AND DEFICIT	\$	2,299.4	\$	1,466.4	\$	484.5	\$	(1,296.9)	\$ 2,953.4

Unaudited Condensed Consolidating Statement of Operations and Comprehensive Income (Loss) For the Three Months Ended March 31, 2018

	Cleveland-Cliffs Guarantor Inc. Subsidiaries		Non-Guarantor Subsidiaries		Eliminations		Cor	solidated	
REVENUES FROM PRODUCT SALES AND SERVICES									
Product	\$ _	\$	169.2	\$	51.5	\$	_	\$	220.7
Freight and venture partners' cost									
reimbursements	 		10.8		7.5		_		18.3
	_		180.0		59.0		_		239.0
COST OF GOODS SOLD AND OPERATING EXPENSES	 		(118.5)		(124.1)		_		(242.6)
SALES MARGIN	_		61.5		(65.1)		_		(3.6)
OTHER OPERATING EXPENSE									
Selling, general and administrative expenses	(20.1)		(4.3)		(3.3)		_		(27.7)
Miscellaneous – net	(0.2)		(5.3)		(3.2)		_		(8.7)
	(20.3)		(9.6)		(6.5)		_		(36.4)
OPERATING INCOME (LOSS)	(20.3)		51.9		(71.6)		_		(40.0)
OTHER INCOME (EXPENSE)									
Interest expense, net	(31.9)		(0.8)		(8.0)		_		(33.5)
Other non-operating income (expense)	(0.9)		0.5		4.8		_		4.4
	(32.8)		(0.3)		4.0		_		(29.1)
INCOME (LOSS) FROM CONTINUING OPERATIONS BEFORE INCOME TAXES	(53.1)		51.6		(67.6)		_		(69.1)
INCOME TAX EXPENSE	(15.6)		(0.1)		_		_		(15.7)
EQUITY IN INCOME (LOSS) OF SUBSIDIARIES	(15.7)		4.5		_		11.2		_
INCOME (LOSS) FROM CONTINUING OPERATIONS	(84.4)		56.0		(67.6)		11.2		(84.8)
INCOME FROM DISCONTINUED OPERATIONS, NET OF TAX	0.1		0.2		0.2		_		0.5
NET INCOME (LOSS) ATTRIBUTABLE TO CLIFFS SHAREHOLDERS	\$ (84.3)	\$	56.2	\$	(67.4)	\$	11.2	\$	(84.3)
OTHER COMPREHENSIVE INCOME	7.7		5.9		0.8		(6.7)		7.7
TOTAL COMPREHENSIVE INCOME (LOSS) ATTRIBUTABLE TO CLIFFS SHAREHOLDERS	\$ (76.6)	\$	62.1	\$	(66.6)	\$	4.5	\$	(76.6)

Unaudited Condensed Consolidating Statement of Operations and Comprehensive Income (Loss) For the Three Months Ended March 31, 2017

	Cleveland-Cliffs	 Guarantor Subsidiaries	١	Non-Guarantor Subsidiaries	E	liminations	Consolidated
REVENUES FROM PRODUCT SALES AND SERVICES							
Product	\$ —	\$ 247.3	\$	165.5	\$		\$ 412.8
Freight and venture partners' cost reimbursements		38.9		9.9			 48.8
	_	286.2		175.4		_	461.6
COST OF GOODS SOLD AND OPERATING EXPENSES		(237.2)		(128.1)			 (365.3)
SALES MARGIN	_	49.0		47.3		_	96.3
OTHER OPERATING INCOME (EXPENSE)							
Selling, general and administrative expenses	(19.5)	(4.4)		(3.8)		_	(27.7)
Miscellaneous – net	(0.1)	(5.5)		17.1			 11.5
	(19.6)	(9.9)		13.3			(16.2)
OPERATING INCOME (LOSS)	(19.6)	39.1		60.6		_	 80.1
OTHER INCOME (EXPENSE)							
Interest expense, net	(41.6)	_		(1.2)		_	(42.8)
Loss on extinguishment of debt	(71.9)	_		_		_	(71.9)
Other non-operating income (expense)	(1.0)	(8.0)		4.3			2.5
	(114.5)	(8.0)		3.1		_	(112.2)
INCOME (LOSS) FROM CONTINUING OPERATIONS BEFORE INCOME TAXES	(134.1)	38.3		63.7		_	(32.1)
INCOME TAX BENEFIT (EXPENSE)	5.2	(0.8)		(2.6)		_	1.8
EQUITY IN INCOME OF SUBSIDIARIES	100.4	3.2		_		(103.6)	_
INCOME (LOSS) FROM CONTINUING OPERATIONS	(28.5)	40.7		61.1		(103.6)	(30.3)
INCOME (LOSS) FROM DISCONTINUED OPERATIONS, net of tax	0.4	0.2		(0.1)		_	0.5
NET INCOME (LOSS)	(28.1)	40.9		61.0		(103.6)	 (29.8)
LOSS ATTRIBUTABLE TO NONCONTROLLING INTEREST	_	1.7		_		_	1.7
NET INCOME (LOSS) ATTRIBUTABLE TO CLIFFS SHAREHOLDERS	\$ (28.1)	\$ 42.6	\$	61.0	\$	(103.6)	\$ (28.1)
OTHER COMPREHENSIVE INCOME (LOSS)	(3.0)	10.8		(17.8)		7.0	(3.0)
TOTAL COMPREHENSIVE INCOME (LOSS) ATTRIBUTABLE TO CLIFFS SHAREHOLDERS	\$ (31.1)	\$ 53.4	\$	43.2	\$	(96.6)	\$ (31.1)

Unaudited Condensed Consolidating Statement of Cash Flows For the Three Months Ended March 31, 2018

		and-Cliffs Inc.	Guarantor Subsidiaries	 n-Guarantor ubsidiaries	Eliminations	(Consolidated
Net cash used by operating activities	\$	(54.7)	\$ (62.8)	\$ (25.4)	\$ _	\$	(142.9)
INVESTING ACTIVITIES							
Purchase of property, plant and equipment		_	(8.1)	(4.3)	_		(12.4)
Deposits for property, plant and equipment		_	(0.8)	(58.2)	_		(59.0)
Intercompany investing		(137.7)	 (4.8)	 	 142.5		_
Net cash used by investing activities	<u> </u>	(137.7)	(13.7)	(62.5)	 142.5		(71.4)
FINANCING ACTIVITIES							
Debt issuance costs		(1.5)	_	_	_		(1.5)
Intercompany financing		_	75.9	66.6	(142.5)		_
Other financing activities		(1.4)	(0.5)	(3.6)			(5.5)
Net cash provided (used) by financing activities		(2.9)	 75.4	 63.0	(142.5)		(7.0)
EFFECT OF EXCHANGE RATE CHANGES ON CASH				0.2			0.2
DECREASE IN CASH AND CASH EQUIVALENTS		(195.3)	 (1.1)	 (24.7)			(221.1)
CASH AND CASH EQUIVALENTS AT BEGINNING OF PERIOD		948.9	2.1	56.7	_		1,007.7
CASH AND CASH EQUIVALENTS AT END OF PERIOD	\$	753.6	\$ 1.0	\$ 32.0	\$ _	\$	786.6

Unaudited Condensed Consolidating Statement of Cash Flows For the Three Months Ended March 31, 2017

(In Millions)

	Cleve	eland-Cliffs Inc.	S Guarantor Subsidiaries		Non-Guarantor Subsidiaries		Eliminations		Consolidated
Net cash provided (used) by operating activities	\$	(99.5)	\$	(19.7)	\$	94.1	\$	\$	(25.1)
INVESTING ACTIVITIES									
Purchase of property, plant and equipment		(8.0)		(24.9)		(0.2)	_		(25.9)
Deposits for property, plant and equipment		_		(2.0)		_	_		(2.0)
Intercompany investing		(56.5)		(0.5)		(45.0)	102.0		_
Other investing activities		_		0.5		_	_		0.5
Net cash used by investing activities		(57.3)		(26.9)		(45.2)	102.0		(27.4)
FINANCING ACTIVITIES									
Net proceeds from issuance of common shares		661.3		_		_	_		661.3
Proceeds from issuance of debt		500.0		_		_	_		500.0
Debt issuance costs		(8.5)		_		_	_		(8.5)
Repurchase of debt		(1,115.5)		_		_	_		(1,115.5)
Distributions of partnership equity		_		(8.7)		_	_		(8.7)
Intercompany financing		45.1		55.8		1.1	(102.0)		_
Other financing activities		(0.5)		(0.7)		(4.4)	_		(5.6)
Net cash provided (used) by financing activities	,	81.9		46.4		(3.3)	(102.0)		23.0
EFFECT OF EXCHANGE RATE CHANGES ON CASH		_		_		1.4	_		1.4
INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS		(74.9)		(0.2)		47.0	_		(28.1)
CASH AND CASH EQUIVALENTS AT BEGINNING OF PERIOD		283.4		2.5		37.5			323.4
CASH AND CASH EQUIVALENTS AT END OF PERIOD	\$	208.5	\$	2.3	\$	84.5	\$	\$	295.3

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Management's Discussion and Analysis of Financial Condition and Results of Operations ("MD&A") is designed to provide a reader of our financial statements with a narrative from the perspective of management on our financial condition, results of operations, liquidity and other factors that may affect our future results. We believe it is important to read our MD&A in conjunction with our Annual Report on Form 10-K for the year ended December 31, 2017 as well as other publicly available information.

Overview

Founded in 1847, Cleveland-Cliffs Inc. is the largest and oldest independent iron ore mining company in the United States. We are a major supplier of iron ore pellets to the North American steel industry from our mines and pellet plants located in Michigan and Minnesota. Additionally, we operate an iron ore mining complex in Western Australia. By 2020, we expect to be the sole producer of HBI in the Great Lakes region with the development of our first production plant in Toledo, Ohio. Driven by the core values of safety, social, environmental and capital stewardship, our employees endeavor to provide all stakeholders with operating and financial transparency.

The key driver of our business is demand for steelmaking raw materials from U.S. steelmakers. During the first two months of 2018, the U.S. produced approximately 13 million metric tons of crude steel, or about 5% of total global crude steel production, which is in line with the same period in 2017. U.S. total steel capacity utilization was approximately 75% in the first two months of 2018, consistent with the same period in 2017. Additionally, in the first two months of 2018, China produced approximately 137 million metric tons of crude steel, or about 49% of total global crude steel production. These figures represent an approximate 6% increase in Chinese crude steel production when compared to the same period in 2017. Through the first two months of 2018, global crude steel production increased about 4% compared to the same period in 2017.

The Platts 62% Price decreased 13% to an average price of \$74 per metric ton for the three months ended March 31, 2018, compared to the same period in 2017. Volatility in the iron ore price impacts our realized revenue rates at each of our segments to varying extents, but our revenue realizations are not fully correlated. Pricing mechanisms in our U.S. Iron Ore contracts reference this metric, but our prices are somewhat protected from the volatility given that it is just one of many of the inputs used in contract pricing formulas. Asia Pacific Iron Ore revenue rates do not see a full correlation to the Platts 62% Price due to the discounts on the lower iron content of the ore sold there.

We recognize the volatility of iron ore supply-demand dynamics and that changes in behaviors of the major iron ore producers and/or Chinese steelmakers could either lift or put pressure on iron ore prices in the near term. We also continue to observe vastly improved demand for higher grade iron ore products, typically those of benchmark grade (62% iron content) and above, as Chinese mills put more emphasis on the productive and environmentally friendly nature of these ores. Assuming the margins at Chinese mills remain strong and the government continues to crack down on pollution, we believe that the mills will favor benchmark quality ore, placing additional pricing pressure on lower quality ore.

This flight to quality has also manifested itself in increased pellet premiums during the first three months of 2018. The Atlantic Basin pellet premium, another important pricing factor in our U.S. Iron Ore contracts, averaged \$58 per metric ton for the first three months of 2018, a 28% increase compared to the same period in 2017. We believe this market will remain tight during 2018, especially with recent labor disputes at a major pelletizing operation in Eastern Canada. We believe this scarcity will support these multi-year high premiums for pellet products.

The price for domestic hot-rolled coil steel, which is an important attribute in the calculation of supplemental revenue in one of our customer's supply agreement, averaged \$757 per net ton for the first three months of 2018, 20% higher than the same period last year. The price of steel was impacted positively in the first quarter by healthy U.S. manufacturing activity, limited imports, and inflation on major steel input costs. Furthermore, during the quarter the U.S. Department of Commerce recommended restrictions on imported steel and aluminum under Section 232 of the Trade Expansion Act of 1962, as amended, on the basis of national security. In response to this recommendation, the President ultimately instituted a 25% tariff on steel imports into the United States, with exemptions for certain allies. Because the United States is the largest importer of steel in the world, we believe these tariffs should not only alleviate some national security concerns, but also keep the prices for domestic hot-rolled coil steel elevated above historical averages for as long as the tariffs remain in place. As such, we remain positive on our outlook for the domestic steel market.

For the three months ended March 31, 2018 and 2017, our consolidated revenues were \$239.0 million and \$461.6 million, respectively, with net loss from continuing operations per diluted share of \$0.29 and \$0.11, respectively. Despite having significantly lower sales volume at our U.S. Iron Ore segment, sales margin increased for the segment during the quarter when compared to the same period in 2017. However, total sales margin decreased by \$99.9 million in the three months ended March 31, 2018 when compared to the same period in 2017, due to our Asia Pacific Iron Ore segment which realized lower revenue rates, lower sales volumes and incurred certain charges as a result of our current mine plan, which anticipates the closure of the Asia Pacific Iron Ore mining operations by June 30, 2018.

First Quarter 2018 Recent Developments

On April 6, 2018, we committed to a course of action expected to lead to the permanent closure of the Asia Pacific Iron Ore mining operations and expect our final Asia Pacific Iron Ore shipment to occur by June 30, 2018. Factors considered in this decision include increasingly discounted prices for lower-iron-content ore, the quality of the remaining iron ore reserves at Asia Pacific Iron Ore and the lack of a commercially reasonable offer from a qualified buyer. Upon the completion of shipping activity, expected in the second quarter of 2018, this business will be treated as a discontinued operation and no longer included in continuing business results.

We estimate total costs that will be incurred in connection with the closure of the Asia Pacific Iron Ore mining operations of approximately \$140 to \$170 million. This amount does not include previously accrued asset retirement obligations, release of cumulative translation adjustments or any proceeds we expect to receive from asset sales, such as rail cars, mobile equipment and the ore handling facility. The expected costs of implementing this closure primarily consist of potential contract termination costs in the range of \$60 to \$70 million; employee severance obligations, demobilization and other closure-related costs of \$30 to \$40 million; and non-cash asset impairments and write-offs of \$50 to \$60 million. We expect that the majority of these charges will be recorded in the first half of 2018. Of the total charges expected to be incurred, we anticipate future cash expenditures of approximately \$120 to \$140 million, including certain capital lease liabilities previously recorded. This range of cash expenditures also excludes any proceeds we may receive from asset sales and other mitigation strategies, to the extent successful.

During the first quarter of 2018, we amended and restated our senior secured ABL Facility, extending its maturity to the earlier of February 28, 2023 or 60 days prior to the maturity of certain other material debt, and introducing several

improvements from the previous facility, which was put into place in March 2015. The changes were introduced in alignment with our vastly improved financial condition since the initial facility was adopted, while continuing to provide us with the financial flexibility needed to operate our business and execute our strategic initiatives. In the amended and restated ABL Facility, the overall size of the credit facility was reduced from \$550 million to \$450 million and borrowing costs and unused commitment fees were also reduced.

During March 2018, we entered into a restructuring term sheet with the Bloom Lake Group and the Wabush Group, which documents the proposed terms of a plan of compromise or arrangement in the CCAA proceedings (the "Proposed Plan") to be sponsored by us as negotiated between us and the Monitor. This Proposed Plan requires both creditor and court approval.

Under the terms of this Proposed Plan, we and certain of our wholly owned subsidiaries have agreed to forego the benefit of any distributions or payments we may be entitled to receive as creditors of the Bloom Lake Group and the Wabush Group and to also make a C\$5.0 million cash contribution to the Bloom Lake Group and the Wabush Group for distribution to other creditors. It is important to note that the Proposed Plan, as currently drafted, will not resolve certain employee claims which have been raised outside of the CCAA proceedings against us and certain of our affiliates and which will be addressed separately.

If this Proposed Plan is approved by a majority of the creditors of both the Bloom Lake Group and the Wabush Group, and is also approved by the court in the CCAA proceedings, and is implemented in accordance with its terms, then the Proposed Plan will resolve all of our claims against the Bloom Lake Group and the Wabush Group and all claims by the Bloom Lake Group, the Wabush Group and their respective creditors against us, except as noted above. The net financial impact of the Proposed Plan is materially consistent with amounts previously recorded in our financial statements.

However, if the creditors and the court do not approve the Proposed Plan, it is reasonably possible that future changes to our estimates and the ultimate amount paid on any claims could be material to our results of discontinued operations in future periods. We are not able to reasonably estimate such impact because there would be significant factual and legal issues to be resolved if the Proposed Plan is not approved. We will vigorously defend any claims if the Proposed Plan is not approved and implemented in accordance with its terms.

The motion to authorize the Bloom Lake Group and the Wabush Group to file the Proposed Plan and to schedule meetings of the creditors of the Bloom Lake Group and the Wabush Group to consider and vote to approve or reject the Proposed Plan was approved by the court on April 20, 2018.

Business Segments

Our company's primary continuing operations are organized and managed according to geographic location: U.S. Iron Ore and Asia Pacific Iron Ore.

Results of Operations - Consolidated

2018 Compared to 2017

The following is a summary of our consolidated results of operations:

			(1	n Millions)						
		Three Months Ended March 31,								
		2018		2017		Variance Favorable/ (Unfavorable)				
Revenues from product sales and services	\$	239.0	\$	461.6	\$	(222.6)				
Cost of goods sold and operating expenses		(242.6)		(365.3)		122.7				
Sales margin	\$	(3.6)	\$	96.3	\$	(99.9)				
Sales margin %		(1.5)%		20.9%		(22.4)%				

Revenues from Product Sales and Services

The decrease in *Revenues from product sales and services* of \$192.1 million or 46.5%, excluding the decrease in freight and reimbursements of \$30.5 million, for the three months ended March 31, 2018 from the comparable period in 2017 was driven by a decrease in sales volume of 1.5 million long tons totaling \$119.7 million from our U.S. Iron Ore operations and a decrease in sales volume of 1.4 million metric tons totaling \$76.5 million from our Asia Pacific Iron Ore operations. In addition, our realized revenue rate decreased 42.8% or \$40.9 million from our Asia Pacific Iron Ore operations. These decreases were offset partially by an increase in our realized revenue rate of 32.4% or \$41.5 million from our U.S. Iron Ore operations.

Cost of Goods Sold and Operating Expenses

The decrease in *Cost of goods sold and operating expenses* of \$92.2 million or 29.1%, excluding the decrease in freight and reimbursements of \$30.5 million, for the three months ended March 31, 2018 from the comparable period in 2017 was primarily due to lower sales volume across all operations that resulted in decreased costs of \$154.2 million. This decrease was offset partially by inventory lower of cost or net realizable value and impairment charges of \$22.1 million and \$18.9 million, respectively, from our Asia Pacific Iron Ore operations.

Refer to "Results of Operations – Segment Information" for additional information regarding the specific factors that impacted our revenue and operating results during the period.

Other Operating Income (Expense)

The following is a summary of Other operating income (expense):

	(In Millions)					
		2018		2017		Variance Favorable/ (Unfavorable)
Selling, general and administrative expenses	\$	(27.7)	\$	(27.7)	\$	_
Miscellaneous – net		(8.7)		11.5		(20.2)
	\$	(36.4)	\$	(16.2)	\$	(20.2)

The following is a summary of *Miscellaneous – net*:

	(In Millions)							
			Three Mon Marc					
	2018		201	17		Variance Favorable/ (Unfavorable)		
Foreign exchange remeasurement	\$	(0.3)	\$	13.6	\$	(13.9)		
Empire idle costs		(4.9)		(6.8)		1.9		
Impairment		(2.6)		_		(2.6)		
Other		(0.9)		4.7		(5.6)		
	\$	(8.7)	\$	11.5	\$	(20.2)		

Miscellaneous – net decreased by \$20.2 million for the three months ended March 31, 2018, from the comparable period in 2017. For the three months ended March 31, 2018, there was an incrementally unfavorable impact of \$13.9 million due to a change in foreign exchange remeasurement of intercompany loans.

Other Income (Expense)

The following is a summary of Other expense:

				(In Millions)				
	Three Months Ended March 31,							
		2018		2017		Variance Favorable/ (Unfavorable)		
Interest expense, net	\$	(33.5)	\$	(42.8)	\$	9.3		
Loss on extinguishment of debt		_		(71.9)		71.9		
Other non-operating income		4.4		2.5		1.9		
	\$	(29.1)	\$	(112.2)	\$	83.1		

Interest expense, net for the three months ended March 31, 2018 had a favorable variance of \$9.3 million versus the comparable prior-year period, predominantly as a result of the debt restructuring activities that occurred throughout 2017. These debt restructurings resulted in a net reduction of our effective interest rate and extended our debt maturities.

The loss on extinguishment of debt of \$71.9 million for the three months ended March 31, 2017 primarily related to the repurchase of certain of our senior notes and the redemption in full of all of our outstanding 1.5 lien notes and second lien notes.

Income Taxes

Our effective tax rate is impacted by permanent items, such as depletion and the relative mix of income we earn in various foreign jurisdictions with tax rates that differ from the U.S. statutory rate. It also is affected by discrete items that may occur in any given period but are not consistent from period to period. The following represents a summary of our tax provision and corresponding effective rates:

		(In	Millions)	
			onths Ended arch 31,	
	 2018		2017	Variance
Income tax benefit (expense)	\$ (15.7)	\$	1.8	\$ (17.5)
Effective tax rate	(22.7)%		5.6 %	(28.3)%

Our tax provision for the three months ended March 31, 2018 was an expense of \$15.7 million and a negative 22.7% effective tax rate compared with a benefit of \$1.8 million and a positive 5.6% effective tax rate for the comparable prior-year period. The difference in the effective rate and income tax expense from the comparable prior-year period is primarily related to discrete items recorded in each period.

For the three months ended March 31, 2018 and 2017, we recorded discrete items that resulted in an income tax expense of \$15.7 million and a benefit of \$0.1 million, respectively. The current year items relate primarily to a \$14.5 million reduction of the refundable AMT credit recorded in *Income tax receivable* in our Statements of Unaudited Condensed Consolidated Financial Position based on the sequestration guidance issued by the Internal Revenue Service during the period ended March 31, 2018. This \$14.5 million current year expense is a reduction of an asset and will not result in a cash tax outlay.

Our 2018 estimated annual effective tax rate before discrete items is 0.1%. This estimated annual effective tax rate differs from the U.S. statutory rate of 21% primarily due to the deductions for percentage depletion in excess of cost depletion related to U.S. operations and the reversal of valuation allowance from operations in the current year.

Results of Operations - Segment Information

We evaluate segment performance based on sales margin, defined as revenues less cost of goods sold and operating expenses identifiable to each segment. Additionally, we evaluate performance on a segment basis, as well as a consolidated basis, based on EBITDA and Adjusted EBITDA. These measures allow management and investors to focus on our ability to service our debt as well as illustrate how the business and each operating segment are performing.

Additionally, EBITDA and Adjusted EBITDA assist management and investors in their analysis and forecasting as these measures approximate the cash flows associated with operational earnings.

EBITDA and Adjusted EBITDA

	(In Millions)						
	Three Mor Marc	ed					
	 2018		2017				
Net Loss	\$ (84.3)	\$	(29.8)				
Less:							
Interest expense, net	(33.5)		(42.8)				
Income tax benefit (expense)	(15.7)		1.8				
Depreciation, depletion and amortization	(23.9)		(23.2)				
EBITDA	\$ (11.2)	\$	34.4				
Less:	 						
Inventory impairments	\$ (18.9)	\$	_				
Impairment of long-lived assets	(2.6)		_				
Severance and retention costs	(1.5)		_				
Impact of discontinued operations	0.5		0.5				
Foreign exchange remeasurement	(0.3)		13.6				
Loss on extinguishment of debt	_		(71.9)				
Adjusted EBITDA	\$ 11.6	\$	92.2				
EBITDA:							
U.S. Iron Ore	\$ 72.5	\$	57.9				
Asia Pacific Iron Ore	(63.7)		51.4				
Other	(20.0)		(74.9)				
Total EBITDA	\$ (11.2)	\$	34.4				
Adjusted EBITDA:							
U.S. Iron Ore	\$ 77.1	\$	64.1				
Asia Pacific Iron Ore	(39.6)		53.8				
Other	 (25.9)		(25.7)				
Total Adjusted EBITDA	\$ 11.6	\$	92.2				

EBITDA decreased \$45.6 million for the three months ended March 31, 2018 on a consolidated basis from the comparable periods in 2017. The unfavorable variance in EBITDA for the three months ended March 31, 2018 was driven primarily by a decrease in sales margin of \$112.4 million from our Asia Pacific Iron Ore operations compared to the prior-year period, offset partially by the negative impact of \$71.9 million from debt extinguishment activities during the prior-year period and an increase in sales margin of \$12.5 million from our U.S. Iron Ore operations compared to the prior-year period.

Adjusted EBITDA decreased \$80.6 million for the three months ended March 31, 2018 from the comparable period in 2017. The decrease primarily was attributable to the lower sales margin of \$112.4 million from our Asia Pacific Iron Ore operations for the three months ended March 31, 2018 compared to the prior-year period, offset partially by Asia Pacific Iron Ore impairments of \$21.5 million which are excluded from Adjusted EBITDA. Refer to further detail below for additional information regarding the specific factors that impacted each reportable segment's sales margin during the three months ended March 31, 2018 and 2017.

2018 Compared to 2017

U.S. Iron Ore

The following is a summary of U.S. Iron Ore results:

					(In Mi	lions))								
	 Changes due to:														
	 Three Months Ended March 31,			Re	evenue			Fre	ight and						
	 2018 2		2017		2017		2017		and cost rate		es volume		urse-ment	Tota	al change
Revenues from product sales and services	\$ 180.0	\$	286.2	\$	41.5	\$	(119.7)	\$	(28.0)	\$	(106.2)				
Cost of goods sold and operating expenses	(118.5)		(237.2)		(5.1)		95.8		28.0		118.7				
Sales margin	\$ 61.5	\$	49.0	\$	36.4	\$	(23.9)	\$	_	\$	12.5				

	(in Millions)										
		Three Mo Mar	nths Ei ch 31,	nded							
Per Ton Information		2018	2017		Difference		Percent change				
Realized product revenue rate ¹	\$	105.03	\$	79.35	\$	25.68	32.4 %				
Cash cost of goods sold and operating expense rate ^{1,2}		57.05		58.37		(1.32)	(2.3)%				
Depreciation, depletion & amortization		9.81		5.26		4.55	86.5 %				
Total cost of goods sold and operating expenses rate		66.86		63.63		3.23	5.1 %				
Sales margin	\$	38.17	\$	15.72	\$	22.45	142.8 %				
Sales tons ³ (In thousands)		1,611		3,118							
Production tons ³ (In thousands)											
Total		5,890		5,814							
Cliffs' share of total		4,500		4,277							

¹ Excludes revenues and expenses related to domestic freight, which are offsetting and have no impact on sales margin. Revenues and expenses also exclude venture partner cost reimbursements, where applicable.

Sales margin for U.S. Iron Ore was \$61.5 million for the three months ended March 31, 2018, compared with \$49.0 million for the three months ended March 31, 2017. Sales margin per long ton increased 142.8% to \$38.17 in the first three months of 2018 compared to the first three months of 2017.

Revenue decreased by \$78.2 million during the three months ended March 31, 2018, compared to the prior-year period, excluding the freight and reimbursements decrease of \$28.0 million, driven by:

- · Lower sales volume of 1.5 million long tons, which resulted in decreased revenues of \$120 million predominantly due to:
 - Lower carry-over tonnage from the prior-year nomination of 0.7 million long tons during the three months ended March 31, 2018 compared to the prior year period; and
 - The adoption of Topic 606 resulted in a reduction of 0.6 million long tons during the three months ended March 31, 2018 compared to the prioryear period as discussed further in NOTE 2 - NEW ACCOUNTING STANDARDS.
- This decrease was offset partially by an increase in the average year-to-date realized product revenue rate of \$26 per long ton or 32.4% during the three months ended March 31, 2018, compared to the same period in the previous year, which resulted in an increase of \$42 million. This is predominantly due to:

² Cash cost of goods sold and operating expense rate is a non-GAAP financial measure. Refer to "Non-GAAP Reconciliation" for reconciliation in dollars back to our consolidated financial statements.

³ Tons are long tons.

- An increase related to supplemental revenue associated with prior-period sales tons that were not consumed during 2017, and therefore are valued based on the 2018 estimated average annual daily market price for hot-rolled coil steel as final pricing is determined in the year the iron ore is consumed in the customer's blast furnaces. The increase to the 2018 estimated average annual daily market price for hot-rolled coil steel positively affected the realized revenue rate by \$17 per long ton or \$28 million.
- Platts 62% related pricing for carry-over tonnage and for tons priced on a lag basis was more favorable during the three months ended March 31, 2018 than the prior-year period, which positively affected the realized revenue rate by \$5 per long ton or \$8 million, despite the decrease in Platts 62% Price;
- Changes in customer and contract mix, which positively affected the realized revenue rate by \$5 per long ton or \$8 million;
- Higher pellet premiums, which positively affected the realized revenue rate by \$3 per long ton or \$4
- These increases were offset partially by higher index freight rates, a component in some of our contract pricing formulas, which negatively affected the realized revenue rate by \$2 per long ton or \$3 million.

Cost of goods sold and operating expenses decreased \$90.7 million during the three months ended March 31, 2018, excluding the freight and reimbursements decrease of \$28.0 million, compared to the same period in 2017. This was predominantly as a result of a decrease in sales volume as discussed above, which decreased costs of \$96 million period-over-period.

Production

Our share of production in the U.S. Iron Ore segment increased by 5.2% in the first three months of 2018 when compared to the same period in 2017. The increase in production volume primarily is attributable to incremental tonnage of 0.2 million long tons as a result of the increase in ownership of the Tilden mine to 100% in the third quarter of 2017.

Asia Pacific Iron Ore

The following is a summary of Asia Pacific Iron Ore results:

		(In Millions)														
		Change due to:														
	Three Months Ended March 31,				Revenue					Fr	eight and					
		2018 2017		2018 2017		2017		d cost rate	Sal	es volume	Exc	hange rate	reimburse-ment		Total change	
Revenues from product sales and services	\$	59.0	\$	175.4	\$	(40.9)	\$	(76.5)	\$	3.5	\$	(2.5)	\$	(116.4)		
Cost of goods sold and operating expenses		(124.1)		(128.1)		(53.2)		58.4		(3.7)		2.5		4.0		
Sales margin	\$	(65.1)	\$	47.3	\$	(94.1)	\$	(18.1)	\$	(0.2)	\$	_	\$	(112.4)		

Per Ton Information Realized product revenue rate¹ Cash cost of goods sold and operating expense rate¹.² Depreciation, depletion & amortization Total cost of goods sold and operating expenses rate Sales margin	(in Millions)											
		Three Mor Marc	ths End	ded								
Per Ton Information		2018		2017	[Difference	Percent change					
Realized product revenue rate ¹	\$	31.10	\$	54.35	\$	(23.25)	(42.8)%					
Cash cost of goods sold and operating expense rate ^{1,2}		66.36		37.27		29.09	78.1 %					
Depreciation, depletion & amortization		4.05		1.54		2.51	163.0 %					
Total cost of goods sold and operating expenses rate		70.41		38.81		31.60	81.4 %					
Sales margin	\$	(39.31)	\$	15.54	\$	(54.85)	(353.0)%					
Sales tons ³ (In thousands)		1,656		3,043								
Production tons ³ (In thousands)		1,637		2,671								

¹ The information above excludes revenues and expenses related to freight, which are offsetting and have no impact on sales margin.

Sales margin for Asia Pacific Iron Ore was negative \$65.1 million or negative \$39.31 per metric ton, for the three months ended March 31, 2018, compared with sales margin of \$47.3 million or \$15.54 per metric ton, for the three months ended March 31, 2017.

Revenue decreased \$113.9 million in the three months ended March 31, 2018, compared to the prior-year period, excluding the freight and reimbursements decrease of \$2.5 million, predominantly due to:

- Decreased sales volume of 1.4 million metric tons, or 45.6%, to 1.7 million metric tons during the three months ended March 31, 2018 compared to the prior-year period. The decrease in tons sold was driven by lower production, as discussed below, which resulted in decreased revenue of \$77 million during the three months ended March 31, 2018, compared to the prior-year period.
- A decrease in the average year-to-date realized product revenue rate of \$23 per metric ton or 42.8% during the three months ended March 31, 2018, compared to the same period in the previous year, which resulted in a decrease of \$37 million, including the impact of foreign exchange. This decrease is predominantly a result of:
 - A decrease in revenue rate of \$13 per metric ton or \$21 million due to price and quality adjustments to meet market competition for lower grade ore and to compensate for varying quality ore and a reduction in iron content; and
 - A decrease in the Platts 62% Price, which negatively affected the realized revenue rate by \$11 per metric ton or \$18 million.

² Cash cost of goods sold and operating expense rate is a non-GAAP financial measure. Refer to "Non-GAAP Reconciliation" for reconciliation in dollars back to our consolidated financial statements.

³ Tons are metric tons.

Cost of goods sold and operating expenses decreased \$1.5 million during the three months ended March 31, 2018 compared to the same period in 2017, excluding the freight and reimbursements decrease of \$2.5 million, predominantly as a result of:

- A decrease in sales volume of 1.4 million metric tons, which decreased costs by \$58 million.
- This decrease was offset partially by an increase in costs of \$53 million or \$32 per metric ton, driven by the following changes associated with our commitment to a course of action expected to lead to the permanent closure of the Asia Pacific Iron Ore mining operations:
 - A lower of cost or net realizable value adjustment of \$22 million or \$13 per metric ton, to reduce work-in process and finished goods inventory to the future expected realized revenue rate;
 - An inventory impairment charge of \$15 million or \$9 per metric ton, for inventory not expected to be sold prior to the closure of operations; and
 - An increase to the supply inventory reserve of \$4 million or \$3 per metric ton.
 - Increased depreciation, depletion and amortization expense driven by the shortened remaining mine life

Production

Production at our Asia Pacific Iron Ore mining complex decreased by 38.7% or 1.0 million metric tons during the first three months of 2018 compared to the same period in 2017. On April 6, 2018, we committed to a course of action expected to lead to the permanent closure of the Asia Pacific Iron Ore mining operations and expect our final Asia Pacific Iron Ore shipment to occur by June 30, 2018. The increasingly discounted prices for lower iron content ore and the quality of the remaining iron ore reserves put significant pressure on production volumes and were significant factors in reaching the decision to close these operations.

Liquidity, Cash Flows and Capital Resources

Our primary sources of liquidity are *Cash and cash equivalents* and cash generated from our operating and financing activities. Our capital allocation decision-making process is focused on maintaining the strength of our balance sheet and creating financial flexibility to manage through the inherent cyclical demand for our products and volatility in commodity prices. We are focused on the preservation of liquidity in our business through maximizing the cash generation of our operations as well as actively managing operating costs, aligning capital investments with our strategic priorities and the requirements of our business plan, including regulatory and permission-to-operate related projects, and managing SG&A expenses.

During the first three months of 2018, we took action consistent with our capital allocation priorities and our stated objective of improving the strength of our balance sheet, improving our financial flexibility and executing on opportunities that will allow us to increase our long-term profitability. We have remained focused on protecting our core U.S. Iron Ore business based on our plan to allocate capital to both sustaining our existing operations and our two major expansion projects: the HBI plant in Toledo, Ohio and the upgrades at the Northshore plant to enable it to produce significantly increased levels of DR-grade pellets. We held a ground breaking event on April 5, 2018 to celebrate the start of construction of our first HBI production plant, as we have made our first major deposit for that project in the first quarter of 2018. Additionally, we have employed a strategy to mitigate our costs as we begin to permanently close the Asia Pacific Iron Ore operations.

Based on our outlook for the next 12 months, which is subject to continued changing demand from steelmakers that utilize our products and volatility in iron ore and domestic steel prices, we expect to generate cash from operations sufficient to meet the needs of our existing operations, including closure costs associated with our Asia Pacific Iron Ore operations and to service our debt obligations.

We estimate total cash outflows of approximately \$120 to \$140 million will be incurred in connection with the closure of the Asia Pacific Iron Ore mining operations. This amount does not include asset retirement obligation spend or any cash inflows we may receive from asset sales, such as rail cars, mobile equipment and the ore handling facility. These future expected cash outflows primarily consist of potential contract termination costs in the range of \$60 to \$70 million; employee severance obligations, demobilization and other closure-related costs of \$30 to \$40 million; and payments associated with capital lease liabilities that could be accelerated upon cancellation of certain contracts of \$30 to \$40 million. We expect that the majority of these cash outflows will be incurred in the next eighteen months.

Refer to "Outlook" for additional guidance regarding expected future results, including projections on sales volume and production.

The following discussion summarizes the significant activities impacting our cash flows during the three months ended March 31, 2018 and 2017 as well as expected impacts to our future cash flows over the next 12 months. Refer to the Statements of Unaudited Condensed Consolidated Cash Flows for additional information.

Operating Activities

Net cash used by operating activities was \$142.9 million and \$25.1 million for the three months ended March 31, 2018 and 2017, respectively. The increase in cash used by operating activities in the first three months of 2018 was primarily due to the unfavorable operating results previously discussed in our Asia Pacific Iron Ore operating segment.

Our U.S. cash and cash equivalents balance at March 31, 2018 was \$755.7 million, or approximately 96.1% of our consolidated total cash and cash equivalents balance of \$786.6 million.

Investing Activities

Net cash used by investing activities was \$71.4 million for the three months ended March 31, 2018, compared with \$27.4 million for the comparable period in 2017. We spent approximately \$12 million and \$28 million on expenditures related to sustaining capital during the three months ended March 31, 2018 and 2017, respectively. Sustaining capital spend includes infrastructure, mobile equipment, environment, safety, fixed equipment, product quality and health. Additionally, during the first three months of 2018, we had cash outflows, including deposits of approximately \$58 million on development of the HBI production plant in Toledo, Ohio.

We anticipate total cash used for capital expenditures, excluding amounts attributable to construction-related contingencies and capitalized interest, during the next twelve months to be approximately \$420 million. Included within this estimate is approximately \$100 million for sustaining capital, \$250 million related to development of the HBI production plant in Toledo, Ohio and \$70 million for upgrades at the Northshore plant to enable it to produce significantly increased levels of DR-grade pellets that could be sold commercially or used as feedstock for the HBI production plant. In total, we expect to spend approximately \$700 million on the HBI production plant and \$90 million on the Northshore upgrades, exclusive of construction-related contingencies and capitalized interest, through 2020.

Financing Activities

Net cash used by financing activities in the first three months of 2018 was \$7.0 million, compared to net cash provided by financing activities of \$23.0 million for the comparable period in 2017. Uses of cash for financing activities during the first three months of 2018 primarily related to the repayment of lease liabilities.

Net cash provided by financing activities for the first three months of 2017 included an equity offering, generating net proceeds of \$661.3 million. Additionally, we issued \$500.0 million aggregate principal amount of 5.75% Senior Notes due 2025, which provided further net proceeds of \$491.5 million. The proceeds from these offerings were used to redeem in full all of our outstanding 8.00% 1.5 lien notes due 2020 and 7.75% second lien notes due 2020 and to purchase certain other outstanding senior notes through tender offers. The total debt redeemed and purchased through tender offers during the first quarter of 2017 was \$1,115.5 million.

Capital Resources

The following represents a summary of key liquidity measures:

	(In Millions)							
	 March 31, 2018		December 31, 2017					
Cash and cash equivalents	\$ 786.6	1,007.7						
Available borrowing base on ABL Facility ¹	\$ 314.1	\$	273.2					
ABL Facility loans drawn	_		_					
Letter of credit obligations and other commitments	(46.6)		(46.5)					
Borrowing capacity available	\$ 267.5	\$	226.7					

¹ The ABL Facility had a maximum borrowing base of \$450 million and \$550 million at March 31, 2018 and December 31, 2017, respectively, determined by applying customary advance rates to eligible accounts receivable, inventory and certain mobile equipment.

Our primary sources of funding are cash on hand, which totaled \$786.6 million as of March 31, 2018, cash generated by our business and availability under the ABL Facility. The combination of cash and availability under the ABL Facility gives us \$1,054.1 million in liquidity entering the second quarter of 2018, which is expected to be adequate to fund operations, letter of credit obligations, sustaining and expansion capital expenditures and other cash commitments for at least the next 12 months.

As of March 31, 2018, we were in compliance with the ABL Facility liquidity requirements and, therefore, the springing financial covenant requiring a minimum Fixed Charge Coverage Ratio of 1.0 to 1.0 was not applicable.

Off-Balance Sheet Arrangements

In the normal course of business, we are a party to certain arrangements that are not reflected on our Statements of Unaudited Condensed Consolidated Financial Position. These arrangements include minimum "take or pay" purchase commitments, such as minimum electric power demand charges, minimum coal, diesel and natural gas purchase commitments, minimum railroad transportation commitments and minimum port facility usage commitments; financial instruments with off-balance sheet risk, such as bank letters of credit and bank guarantees; and operating leases, which relate primarily to equipment and office space.

Market Risks

We are subject to a variety of risks, including those caused by changes in commodity prices, foreign currency exchange rates and interest rates. We have established policies and procedures to manage such risks; however, certain risks are beyond our control.

Pricing Risks

Commodity Price Risk

Our consolidated revenues include the sale of iron ore pellets, iron ore lump and iron ore fines. Our financial results can vary significantly as a result of fluctuations in the market prices of iron ore and hot-rolled coil steel. World market prices for these commodities have fluctuated historically and are affected by numerous factors beyond our control. The world market price that is most commonly utilized in our iron ore sales contracts is the Platts 62% Price, which can fluctuate widely due to numerous factors, such as global economic growth or contraction, change in demand for steel or changes in availability of supply. The other important metric in our price realizations in the U.S. is the price for hot-rolled coil steel, which can fluctuate due to similar factors.

Customer Supply Agreements

A supply agreement with one U.S. Iron Ore customer provides for supplemental revenue or refunds based on the average annual daily market price for hot-rolled coil steel at the time the iron ore product is consumed in the customer's blast furnaces. At March 31, 2018, we had derivative assets of \$91.2 million, representing the fair value of the pricing factors, based upon the amount of unconsumed long tons and an estimated average hot-rolled coil steel price related to the period in which the iron ore is expected to be consumed in the customer's blast furnaces, subject to final pricing at a future date. We estimate that a \$75 positive or negative change in the average daily market price for hot-rolled coil steel realized from the March 31, 2018 estimated price recorded would cause the fair value of the derivative instrument to increase or decrease by approximately \$18 million, respectively, thereby impacting our consolidated revenues by the same amount. We have not entered into any hedging programs to mitigate the risk of adverse price fluctuations.

Provisional Pricing Arrangements

Certain of our U.S. Iron Ore and Asia Pacific Iron Ore customer supply agreements specify provisional price calculations, where the pricing mechanisms generally are based on market pricing, with the final revenue rate to be based on market inputs at a specified point in time in the future, per the terms of the supply agreements. At March 31, 2018, we had derivative assets and liabilities of \$2.4 million and \$4.2 million, respectively, reflected as part of our U.S. Iron Ore and Asia Pacific Iron Ore segment revenue, representing the fair value of the provisional price calculations. We estimate that a positive or negative \$10 change in the Platts 62% Price from the March 31, 2018 estimated price recorded would cause the fair value of the derivative instrument to increase or decrease by approximately \$2 million and \$5 million, for our U.S. Iron Ore and Asia Pacific Iron Ore segments, respectively.

We have not entered into any hedging programs to mitigate the risk of adverse price fluctuations; however, most of our Asia Pacific Iron Ore supply agreements are short-term in nature and therefore do not expose us to long-term risk.

Volatile Energy and Fuel Costs

The volatile cost of energy is an important factor affecting the production costs at our iron ore operations. Our consolidated U.S. Iron Ore operations consumed 4.4 million MMBtu's of natural gas at an average delivered price of \$3.92 per MMBtu, excluding the natural gas hedge impact, or \$3.91 per MMBtu net of the natural gas hedge impact during the first three months of 2018. Additionally, our consolidated U.S. Iron Ore operations consumed 5.7 million gallons of diesel fuel at an average delivered price of \$2.19 per gallon during the first three months of 2018. Consumption of diesel fuel by our Asia Pacific operations was 1.9 million gallons at an average delivered price of \$2.07 per gallon for the same period.

In the ordinary course of business, there may also be increases in prices relative to electricity costs at our U.S. mine sites. Specifically, our Tilden mine in Michigan has entered into large curtailable special contracts with Wisconsin Electric Power Company. Charges under those special contracts are subject to a power supply cost recovery mechanism that is based on variations in the utility's actual fuel and purchase power expenses.

Our strategy to address volatile natural gas and diesel rates includes improving efficiency in energy usage, identifying alternative providers and utilizing the lowest cost alternative fuels. A full-year hedging program was implemented during the fourth quarter of 2017 in order to manage the price risk of natural gas at our U.S. Iron Ore mines. We will continue to monitor relevant energy markets for risk mitigation opportunities and may make additional forward purchases or employ other hedging instruments in the future as warranted and deemed appropriate by management. In the near term, a 10% change in the remaining nine months from our current average year-to-date natural gas and diesel fuel prices would result in a change of approximately \$9.0 million in our annual fuel and energy cost based on expected consumption for 2018.

Valuation of Other Long-Lived Assets

Long-lived assets are reviewed for impairment upon the occurrence of events or changes in circumstances that would indicate that the carrying value of the assets may not be recoverable. Such indicators may include, among others: a significant decline in expected future cash flows; a sustained, significant decline in market pricing; a significant adverse change in legal or environmental factors or in the business climate; changes in estimates of our recoverable reserves; unanticipated competition; and slower growth or production rates. Any adverse change in these factors could have a significant impact on the recoverability of our long-lived assets and could have a material impact on our consolidated statements of operations and statement of financial position.

A comparison of each asset group's carrying value to the estimated undiscounted future cash flows expected to result from the use of the assets, including cost of disposition, is used to determine if an asset is recoverable. Projected

future cash flows reflect management's best estimates of economic and market conditions over the projected period, including growth rates in revenues and costs, estimates of future expected changes in operating margins and capital expenditures. If the carrying value of the asset group is higher than its undiscounted future cash flows, the asset group is measured at fair value and the difference is recorded as a reduction to the long-lived assets. We estimate fair value using a market approach, an income approach or a cost approach. As of March 31, 2018, there were no indicators present indicative of potential impairment or the inability to recover the value of our long-lived assets at our U.S. Iron Ore Operations. As of March 31, 2018, based on the anticipated closure of the Asia Pacific Iron Ore operations we determined that we would not recover the value of certain long-lived assets at our Asia Pacific Iron Ore operations. As a result, we recorded an impairment of \$2.6 million in *Miscellaneous – net* in the Statements of Unaudited Condensed Consolidated Operations.

Foreign Currency Exchange Rate Risk

We are subject to changes in foreign currency exchange rates as a result of our operations in Australia, which could impact our financial condition. With respect to Australia, foreign exchange risk arises from our exposure to fluctuations in foreign currency exchange rates because our reporting currency is the U.S. dollar but the functional currency of our Asia Pacific operations is the Australian dollar. Our Asia Pacific operations receive funds in U.S. currency for their iron ore sales and incur costs in Australian currency. We estimate that if the average Australian dollar to U.S. dollar exchange rate during the remaining nine months of 2018 is \$0.05 higher or lower than the March 31, 2018 rate, this would cause our cash cost of goods sold and operating expense as well as additional costs related to the exit of our mining operations to increase or decrease by approximately \$10 million, respectively, for our Asia Pacific Iron Ore segment.

We have not entered into any hedging programs to mitigate the risk of adverse currency fluctuations. We have suspended entering into new foreign exchange rate contracts as we have indefinitely deferred the program. In the future, we may enter into additional hedging instruments as needed in order to hedge our exposure to changes in foreign currency exchange rates.

Interest Rate Risk

Interest payable on our senior notes is at fixed rates. Interest payable under our ABL Facility is at a variable rate based upon the base rate plus the base rate margin depending on the excess availability. As of March 31, 2018, we had no amounts drawn on the ABL Facility.

During 2017, we issued the 5.75% Senior Notes in private transactions exempt from the registration requirements of the Securities Act. Pursuant to the registration rights agreement executed as part of the issuances, we agreed to file a registration statement with the SEC with respect to a registered offer to exchange the 5.75% Senior Notes for publicly registered notes within 365 days of the issue date and consummate the exchange offer within 120 days after the first anniversary of the issue date. We filed the registration statement for the exchange offer in February 2018 and commenced the exchange offer on March 29, 2018. If we fail to consummate the exchange offer within the required deadline, the interest rate on the 5.75% Senior Notes will be increased by 0.25% per annum during the 90-day period immediately following such default, and such rate shall increase by 0.25% per annum at the end of each subsequent 90-day period until all defaults have been cured, up to a maximum additional interest rate of 1.00% per annum.

Supply Concentration Risks

Many of our mines are dependent on one source each of electric power and natural gas. A significant interruption or change in service or rates from our energy suppliers could impact materially our production costs, margins and profitability.

Outlook

	2018 Outlook Summary
Per Long Ton Information	U.S. Iron Ore
Revenues from product sales and services ¹	\$102 - \$107
Cost of goods sold and operating expense rate	\$68 - \$73
Less:	
Freight expense rate ²	\$7
Depreciation, depletion & amortization rate	\$3
Cash cost of goods sold and operating expense rate	\$58 - \$63
Sales volume (million long tons)	20.5
Production volume (million long tons)	20.0

¹ This expectation is based on the assumption that iron ore prices, steel prices, and pellet premiums will average for the remainder of 2018 their respective year-to-date averages.

U.S. Iron Ore Outlook (Long Tons)

Based on the assumption that iron ore prices, steel prices, and pellet premiums will average for the remainder of 2018 their respective year-to-date averages, we would realize U.S. Iron Ore revenue rates in the range of \$102 to \$107 per long ton. This represents an increase from the prior calculation based on the substantial increase in hot-rolled coil steel prices, partially offset by lower iron ore prices.

As a result of strong market demand for pellets in the Great Lakes, we have increased our full-year sales volume expectation by 500,000 long tons to 20.5 million long tons. Our production volume expectation of 20 million tons was maintained.

Our full-year 2018 U.S. Iron Ore cash cost of goods sold and operating expense expectation was maintained at \$58 - \$63 per long ton.

SG&A Expenses and Other Expectations

Our full-year 2018 SG&A expense expectation of \$115 million was maintained. We also note that of the \$115 million expectation, approximately \$20 million is considered non-cash.

Our full-year 2018 interest expense is expected to be approximately \$130 million. Full-year 2018 depreciation, depletion and amortization associated with U.S. Iron Ore and Corporate/Other is expected to be approximately \$80 million.

Capital Expenditures

We provided the following updates to our 2018 capital expenditures budget:

- The Toledo HBI Project spend expectation was reduced by \$25 million to \$225 million due to further development and refined timing of the project spending plan;
- · The sustaining capital expectation of \$85 million was maintained; and
- The Northshore Mine upgrade spend expectation of \$50 million was maintained.

² Freight has an offsetting amount in revenue and has no impact on sales margin.

Forward-Looking Statements

This report contains statements that constitute "forward-looking statements" within the meaning of the federal securities laws. As a general matter, forward-looking statements relate to anticipated trends and expectations rather than historical matters. Forward-looking statements are subject to uncertainties and factors relating to Cliffs' operations and business environment that are difficult to predict and may be beyond our control. Such uncertainties and factors may cause actual results to differ materially from those expressed or implied by the forward-looking statements. These statements speak only as of the date of this report, and we undertake no ongoing obligation, other than that imposed by law, to update these statements. Uncertainties and risk factors that could affect Cliffs' future performance and cause results to differ from the forward-looking statements in this report include, but are not limited to:

- uncertainty and weaknesses in global economic conditions, including downward pressure on prices caused by oversupply or imported products, the
 impact of any reduced barriers to trade, the outcomes of recently filed and forthcoming trade cases, reduced market demand and any change to the
 economic growth rate in China;
- continued volatility of iron ore and steel prices and other trends, including the supply approach of the major iron ore producers, affecting our financial condition, results of operations or future prospects, specifically the impact of price-adjustment factors on our sales contracts;
- our level of indebtedness could limit cash flow available to fund working capital, capital expenditures, acquisitions and other general corporate purposes or ongoing needs of our business;
- · availability of capital and our ability to maintain adequate liquidity;
- our ability to successfully conclude the CCAA process and plan to close our Asia Pacific operations in a manner that minimizes cash outflows and associated liabilities, including our ability to successfully mitigate the closure costs of our Asia Pacific operations;
- the impact of our customers reducing their steel production due to increased market share of steel produced using other methods or lighter-weight steel alternatives:
- uncertainty relating to restructurings in the steel industry and/or affecting the steel industry;
- the outcome of any contractual disputes with our customers, joint venture partners or significant energy, material or service providers or any other litigation or arbitration;
- · the ability of our customers and joint venture partners to meet their obligations to us on a timely basis or at all;
- problems or uncertainties with productivity, tons mined, transportation, mine-closure obligations, environmental liabilities, employee-benefit costs and other risks of the mining industry;
- our ability to reach agreement with our customers regarding any modifications to sales contract provisions, renewals or new arrangements;
- our actual levels of capital spending;
- · our ability to successfully diversify our product mix and add new customers beyond our traditional blast furnace clientele;
- our actual economic iron ore reserves or reductions in current mineral estimates, including whether any mineralized material qualifies as a reserve;
- · our ability to cost-effectively achieve planned production rates or levels, including at our HBI production plant;
- our ability to successfully identify and consummate any strategic investments or development projects, including our HBI production plant;
- · changes in sales volume or mix;
- events or circumstances that could impair or adversely impact the viability of a mine and the carrying value of associated assets, as well as any resulting impairment charges;

- our ability to maintain appropriate relations with unions and employees;
- impacts of existing and increasing governmental regulation and related costs and liabilities, including failure to receive or maintain required
 operating and environmental permits, approvals, modifications or other authorization of, or from, any governmental or regulatory entity and costs
 related to implementing improvements to ensure compliance with regulatory changes;
- uncertainties associated with natural disasters, weather conditions, unanticipated geological conditions, supply or price of energy, equipment failures and other unexpected events;
- adverse changes in currency values, currency exchange rates, interest rates and tax laws;
- · risks related to international operations; and
- · the potential existence of significant deficiencies or material weakness in our internal control over financial reporting.

For additional factors affecting the business of Cliffs, refer to Part II - Item 1A. Risk Factors. You are urged to carefully consider these risk factors.

Non-GAAP Reconciliation

We present cash cost of goods sold and operating expense rate per long/metric ton, which is a non-GAAP financial measure that management uses in evaluating operating performance. We believe our presentation of non-GAAP cash cost of goods sold and operating expenses is useful to investors because it excludes depreciation, depletion and amortization, which are non-cash, and freight and joint venture partners' cost reimbursements, which have no impact on sales margin, thus providing a more accurate view of the cash outflows related to the sale of iron ore. The presentation of this measure is not intended to be considered in isolation from, as a substitute for, or as superior to, the financial information prepared and presented in accordance with GAAP. The presentation of this measure may be different from non-GAAP financial measures used by other companies. Below is a reconciliation in dollars of this non-GAAP financial measure to our consolidated financial statements.

						(In Mil	lions)							
		Three	Month	s Ended Mar	rch 31	,	Three Months Ended March 31,							
				2018			2017							
	U.S.	Iron Ore	Pacific Iron Ore		Total	U.S	. Iron Ore	Asia Pacific Iron on Ore Ore			Total			
Cost of goods sold and operating expenses	\$	118.5	\$	124.1	\$	242.6	\$	237.2	\$	128.1	\$	365.3		
Less:														
Freight and reimbursements		10.8		7.5		18.3		38.8		10.0		48.8		
Depreciation, depletion & amortization		15.8		6.7		22.5		16.4		4.7		21.1		
Cash cost of goods sold and operating expenses	\$	91.9	\$	109.9	\$	201.8	\$	182.0	\$	113.4	\$	295.4		

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Information regarding our Market Risk is presented under the caption *Market Risks*, which is included in our Annual Report on Form 10-K for the year ended December 31, 2017, and in the Management's Discussion and Analysis section of this report.

Item 4. Controls and Procedures

We maintain disclosure controls and procedures that are designed to ensure that information required to be disclosed in our Exchange Act reports is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including the Chief Executive Officer and the Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosure based solely on the definition of "disclosure controls and procedures" in Rule 13a-15(e) promulgated under the Exchange Act. In designing and evaluating the disclosure controls and procedures, management recognized that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives, and management necessarily was required to apply its judgment in evaluating the cost-benefit relationship of possible controls and procedures.

As of the end of the period covered by this report, we carried out an evaluation under the supervision and with the participation of our management, including the Chief Executive Officer and the Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures. Based on the foregoing, the Chief Executive Officer and the Chief Financial Officer, concluded that our disclosure controls and procedures were effective.

There have been no changes in our internal control over financial reporting or in other factors that occurred during our last fiscal quarter that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II

Item 1. Legal Proceedings

CCAA Proceedings. Refer to NOTE 19 - COMMITMENTS AND CONTINGENCIES of the notes to our condensed consolidated financial statements included in Item 1 of Part 1 of this report for a description of the CCAA Proceedings underway with respect to the Bloom Lake Group and the Wabush Group. Such description is incorporated by reference into this Item 1.

Mesabi Metallics Adversary Proceeding. On September 7, 2017, Mesabi Metallics Company LLC (f/k/a Essar Steel Minnesota LLC) ("Mesabi Metallics") filed a complaint against Cleveland-Cliffs Inc. in the Essar Steel Minnesota LLC and ESML Holdings Inc. bankruptcy proceeding that is pending in the United States Bankruptcy Court, District of Delaware. Mesabi Metallics alleges tortious interference with its contractual rights and business relations involving certain vendors, suppliers and contractors, violations of federal and Minnesota antitrust laws through monopolization, attempted monopolization and restraint of trade, violation of the automatic stay, and civil conspiracy with unnamed Doe defendants. Mesabi Metallics amended its complaint to add additional defendants, including, among others, our subsidiary, Cleveland-Cliffs Minnesota Land Development Company LLC ("Cliffs Minnesota Land"), and to add additional claims, including avoidance and recovery of unauthorized post-petition transfers, claims disallowance, civil contempt and declaratory relief. Mesabi Metallics seeks, among other things, unspecified damages and injunctive relief. Cliffs and Cliffs Minnesota Land filed counterclaims against Mesabi Metallics, Chippewa Capital Partners ("Chippewa"), and Thomas M. Clarke ("Clarke"), for tortious interference and civil conspiracy, as well as additional claims against Chippewa and Clarke for aiding and abetting tortious interference and against Clarke for libel, for which we seek, among other things, damages and injunctive relief. The parties have filed various dispositive motions on certain of the claims. We believe the claims asserted against us are unmeritorious and intend to continue to vigorously defend the lawsuit.

Item 1A. Risk Factors

Our Annual Report on Form 10-K for the year ended December 31, 2017, includes a detailed discussion of our risk factors.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

The following table presents information with respect to repurchases by the Company of our common shares during the periods indicated.

ISSUER PURCHASES OF EQUITY SECURITIES

Period	Total Number of Shares (or Units) Purchased ¹	A	verage Price Paid per Share (or Unit)	Total Number of Shares (or Units) Purchased as Part of Publicly Announced Plans or Programs	Appro Shares	eximum Number (or ximate Dollar Value) of (or Units) that May Yet chased Under the Plans or Programs
January 1 - 31, 2018	1,248	\$	7.21	_	\$	_
February 1 - 28, 2018	171,592	\$	7.52	_	\$	_
March 1 - 31, 2018		\$	_	_	\$	_
	172,840	\$	7.52	_	\$	_

¹ These shares were delivered to us to satisfy tax withholding obligations due upon the vesting or payment of stock awards.

Item 4. Mine Safety Disclosures

We are committed to protecting the occupational health and well-being of each of our employees. Safety is one of our core values and we strive to ensure that safe production is the first priority for all employees. Our internal objective is to achieve zero injuries and incidents across the Company by focusing on proactively identifying needed prevention activities, establishing standards and evaluating performance to mitigate any potential loss to people, equipment, production and the environment. We have implemented intensive employee training that is geared toward maintaining a high level of awareness and knowledge of safety and health issues in the work environment through the development and coordination of requisite information, skills and attitudes. We believe that through these policies we have developed an effective safety management system.

Under the Dodd-Frank Act, each operator of a coal or other mine is required to include certain mine safety results within its periodic reports filed with the SEC. As required by the reporting requirements included in §1503(a) of the Dodd-Frank Act and Item 104 of Regulation S-K, the required mine safety results regarding certain mining safety and health matters for each of our mine locations that are covered under the scope of the Dodd-Frank Act are included in Exhibit 95 of Item 6. Exhibits of this Quarterly Report on Form 10-Q.

Item 5. Other Information

None.

Item 6. Exhibits

All documents referenced below have been filed pursuant to the Securities Exchange Act of 1934 by Cleveland-Cliffs Inc., file number 1-09844, unless otherwise indicated.

Exhibit Number	Exhibit
10.1	Amended and Restated Syndicated Facility Agreement, by and among Bank of America, N.A., as Administrative Agent and Australian Security Trustee, the Lenders that are Parties hereto, as the Lenders, Cleveland-Cliffs Inc., as Parent and a Borrower, and the Subsidiaries of Parent Party hereto, as Borrowers, dated as of February 28, 2018 (filed herewith)
10.2	* Form of Cleveland-Cliffs Inc. Amended and Restated 2015 Equity and Incentive Compensation Plan Restricted Stock Unit Award Memorandum (Vesting December 31, 2020) and Restricted Stock Unit Award Agreement (filed herewith)
10.3	* Form of Cleveland-Cliffs Inc. Amended and Restated 2015 Equity and Incentive Compensation Plan Performance Share Award Memorandum and Performance Share Award Agreement (filed herewith)
10.4	* Form of Cleveland-Cliffs Inc. Amended and Restated 2015 Equity and Incentive Compensation Plan Cash Incentive Award Memorandum (TSR) (Vesting December 31, 2020) and Cash Incentive Award Agreement (TSR) (filed herewith)
<u>31.1</u>	Certification Pursuant to 15 U.S.C. Section 7241, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, signed and dated by Lourenco Goncalves as of April 24, 2018 (filed herewith)
31.2	Certification Pursuant to 15 U.S.C. Section 7241, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002, signed and dated by Timothy K. Flanagan as of April 24, 2018 (filed herewith)
<u>32.1</u>	Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, signed and dated by Lourenco Goncalves, Chairman, President and Chief Executive Officer of Cleveland-Cliffs Inc., as of April 24, 2018 (filed herewith)
<u>32.2</u>	Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, signed and dated by Timothy K. Flanagan, Executive Vice President, Chief Financial Officer of Cleveland-Cliffs Inc., as of April 24, 2018 (filed herewith)
<u>95</u>	Mine Safety Disclosures (filed herewith)
101.INS	XBRL Instance Document
101.SCH	XBRL Taxonomy Extension Schema Document
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document
101.LAB	XBRL Taxonomy Extension Label Linkbase Document
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document

^{*} Indicates management contract or other compensatory arrangement.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

CLEVELAND-CLIFFS INC.

By: /s/ R. Christopher Cebula

Name: R. Christopher Cebula

Title: Vice President, Corporate Controller & Chief Accounting Officer

Date: 4/24/2018

AMENDED AND RESTATED SYNDICATED FACILITY AGREEMENT by and among

BANK OF AMERICA, N.A.,

as Agent and Australian Security Trustee,

THE LENDERS THAT ARE PARTIES HERETO,

as the Lenders,

CLEVELAND-CLIFFS INC.,

as Parent and a Borrower, and

THE SUBSIDIARIES OF PARENT PARTY HERETO,

as Borrowers

MERRILL LYNCH, PIERCE, FENNER & SMITH INCORPORATED,

PNC CAPITAL MARKETS LLC,

DEUTSCHE BANK SECURITIES INC.,

citizens bank, n.a.,

and

REGIONS BUSINESS CAPITAL, A DIVISION OF REGIONS BANK,

as Joint Lead Arrangers

and

Joint Book Runners

Dated as of March 30, 2015

and

Amended and Restated as of February 28, 2018

TABLE OF CONTENTS

			Page
1	DEFINI	TIONS AND CONSTRUCTION	<u>1</u>
	1.1	Definitions	<u>1</u>
	1.2	Accounting Terms	<u>1</u>
	1.3	Code; Australian PPSA	<u>2</u>
	1.4	Construction	<u>2</u>
	1.5	Time References	<u>4</u>
	1.6	Schedules and Exhibits	<u>4</u>
	1.7	Dollar Equivalent	<u>4</u>
	1.8	LIBOR Rate	<u>5</u>
	1.9	Borrowers	<u>5</u>
	1.10	Australian Terms	<u>5</u>
2	REVOL'	VING LOANS AND TERMS OF PAYMENT	<u>5</u> <u>5</u>
	2.1	Revolving Loans	<u>5</u>
	2.2	Reallocation of the Australian Revolver Commitments	<u>7</u>
	2.3	Borrowing Procedures and Settlements	<u>7</u>
	2.4	Payments; Reductions of Commitments; Prepayments	<u>16</u>
	2.5	Promise to Pay; Promissory Notes	
	2.6	Interest Rates and Letter of Credit Fee; Rates; Payments, and Calculations	<u>24</u>
	2.7	Crediting Payments	<u>25</u>
	2.8	Designated Accounts	<u>26</u>
	2.9	Maintenance of Loan Accounts; Statements of Obligations	<u></u>
	2.10	Fees	
	2.11	U.S. Letters of Credit	<u>27</u>
	2.12	Australian Letters of Credit	<u>32</u>
	2.13	LIBOR Option	<u>38</u>
	2.14	Capital Requirements	<u>40</u>
	2.15	Joint and Several Liability	<u>41</u>
	2.16	Incremental Borrowings	<u>44</u>
3	CONDI	TIONS; TERM OF AGREEMENT	<u>45</u>
	3.1	Conditions Precedent to the Effectiveness of this Agreement	<u>45</u>
	3.2	Consequences of Closing Date	<u>45</u>
	3.3	Conditions Precedent to all Extensions of Credit	<u>45</u>
	3.4	Maturity	<u>46</u>
	3.5	Effect of Maturity	<u>46</u>
	3.6	Early Termination by Borrowers	<u>46</u>
	3.7	Conditions Subsequent	<u>47</u>
4	REPRE	SENTATIONS AND WARRANTIES	<u>47</u>
	4.1	Due Organization and Qualification; Subsidiaries	<u>47</u>
	4.2	Due Authorization; No Conflict	<u>47</u>
	4.3	Governmental Consents	<u>48</u>
	4.4	Binding Obligations; Perfected Liens	<u>48</u>
	4.5	Title to Assets; No Encumbrances	<u>48</u>
	4.6	Litigation	<u>48</u>
	4.7	Compliance with Laws	<u>48</u>
	4.8	No Material Adverse Effect	<u>49</u>

TABLE OF CONTENTS

(continued)

			Page
	4.9	Solvency	<u>49</u>
	4.10	Employee Benefits	<u>49</u>
	4.11	Environmental Condition	<u>49</u>
	4.12	Complete Disclosure	<u>50</u>
	4.13	Sanctions, PATRIOT Act, and FCPA	<u>50</u>
	4.14	[Reserved.]	<u>50</u>
	4.15	Payment of Taxes	<u>50</u>
	4.16	Margin Stock	<u>51</u>
	4.17	Governmental Regulation	<u>51</u>
	4.18	OFAC	<u>51</u>
	4.19	Employee and Labor Matters	<u>51</u>
	4.20	[Reserved.]	<u>51</u>
	4.21	[Reserved.]	<u>51</u>
	4.22	Eligible Accounts	<u>51</u>
	4.23	Eligible Inventory and Eligible Equipment	<u>52</u>
	4.24	[Reserved.]	<u>52</u>
	4.25	Inventory and Equipment Records	<u>52</u>
	4.26	Tax Consolidation	<u>52</u>
	4.27	Commercial Benefit	<u>52</u>
	4.28	No Immunity	<u>52</u>
	4.29	[Reserved.]	<u>52</u>
	4.30	Material Contracts	<u>52</u>
	4.31	EEA Financial Institutions	<u>52</u>
5	AFFIRM.	ATIVE COVENANTS	<u>52</u>
	5.1	Financial Statements, Reports, Certificates	<u>53</u>
	5.2	Reporting	<u>53</u>
	5.3	Existence	<u>53</u>
	5.4	Maintenance of Properties	<u>53</u>
	5.5	Taxes	<u>53</u>
	5.6	Insurance	<u>53</u>
	5.7	Inspection	<u>54</u>
	5.8	Compliance with Laws	<u>54</u>
	5.9	Environmental	<u>54</u>
	5.10	[Reserved]	<u>55</u>
	5.11	Formation of Subsidiaries	<u>55</u>
	5.12	Further Assurances	<u>55</u>
	5.13	Lender Meetings	<u>56</u>
	5.14	[Reserved]	
	5.15	Compliance with ERISA and the IRC	<u>56</u>
	5.16	Cash Management	<u>56</u>
6		VE COVENANTS	<u>58</u>
	6.1	Indebtedness	<u></u>
	6.2	Liens	
	6.3	Restrictions on Fundamental Changes	
	6.4	Disposal of Assets	
		•	<u></u>

TABLE OF CONTENTS

(continued)

			Page
	6.5	Nature of Business	<u>58</u>
	6.6	Prepayments and Amendments	<u>58</u>
	6.7	Restricted Payments	<u>59</u>
	6.8	Accounting Methods	<u>60</u>
	6.9	Investments	<u>60</u>
	6.10	Transactions with Affiliates	<u>60</u>
	6.11	Use of Proceeds	<u>61</u>
7	FINANC	CIAL COVENANT	<u>61</u>
8	EVENT:	S OF DEFAULT	<u>61</u>
	8.1	Payments	<u>61</u>
	8.2	Covenants	<u>62</u>
	8.3	Judgments	<u>62</u>
	8.4	Voluntary Bankruptcy, etc.	<u>62</u>
	8.5	Involuntary Bankruptcy, etc.	<u>62</u>
	8.6	Default Under Other Agreements	<u>62</u>
	8.7	Representations, etc.	<u>62</u>
	8.8	Security Documents	<u>63</u>
	8.9	Loan Documents	<u>63</u>
	8.10	Change in Control	<u>63</u>
	8.11	ERISA and Australian Pension Events	<u>63</u>
	8.12	Expropriation, etc.	<u>63</u>
9	RIGHTS	S AND REMEDIES	<u>63</u>
	9.1	Rights and Remedies	<u>63</u>
	9.2	Remedies Cumulative	<u>64</u>
10	WAIVER	RS; INDEMNIFICATION	<u>64</u>
	10.1	Demand; Protest; etc.	<u>64</u>
	10.2	The Lender Group's Liability for Collateral	<u>64</u>
	10.3	Indemnification	<u>65</u>
11	NOTICE	ES .	<u>65</u>
12	CHOICE	E OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION	<u>66</u>
13	ASSIGN	IMENTS AND PARTICIPATIONS; SUCCESSORS	<u>67</u>
	13.1	Assignments and Participations	<u>67</u>
	13.2	Successors	<u>70</u>
14	AMEND	MENTS; WAIVERS	<u>70</u>
	14.1	Amendments and Waivers	<u></u>
	14.2	Replacement of Certain Lenders	<u></u>
	14.3	No Waivers; Cumulative Remedies	<u>73</u>
15	AGENT	AND AUSTRALIAN SECURITY TRUSTEE	<u>73</u>
	15.1	Appointment; Authority and Duties of Agent	<u>73</u>
	15.2	Liability of Agent and Australian Security Trustee	<u></u>
	15.3	Reliance by Agent	
	15.4	Notice of Default or Event of Default	<u>75</u>
	15.5	Due Diligence and Non-Reliance	<u>75</u>
	15.6	Indemnification	<u>75</u>
	15.7	Individual Capacities	<u></u> <u>76</u>
	-	·	

TABLE OF CONTENTS

(continued)

			Page	
	15.8	Successor Agent	<u>76</u>	
	15.9	Agreements Regarding Collateral and Borrower Materials	<u>76</u>	
	15.10	Ratable Sharing	77	
	15.11	Remittance of Payments and Collections	<u>77</u>	
	15.12	Titles	<u>78</u>	
	15.13	Bank of Product Providers	78 78	
	15.14	Australian Security Trustee	<u>78</u>	
	15.15	No Third Party Beneficiaries	<u>79</u>	
16	COLLEC	<u>79</u>		
17	WITHHOLDING TAXES			
	17.1	Payments	<u>80</u> 80	
	17.2	Exemptions	<u>80</u>	
	17.3	Reductions	<u>81</u>	
	17.4	Refunds	<u>82</u>	
18	GENERA	<u>82</u>		
	18.1	Effectiveness	<u>82</u> 82	
	18.2	Section Headings	<u>82</u>	
	18.3	Interpretation	<u>83</u>	
	18.4	Severability of Provisions	<u>83</u>	
	18.5	Bank Product Providers	<u>83</u>	
	18.6	Debtor-Creditor Relationship	<u>83</u>	
	18.7	Counterparts; Electronic Execution	<u>83</u>	
	18.8	Revival and Reinstatement of Obligations; Certain Waivers	<u>84</u>	
	18.9	Confidentiality	<u>84</u> <u>84</u>	
	18.10	Survival	<u>85</u>	
	18.11	Patriot Act	<u>85</u>	
	18.12	Integration	<u>86</u>	
	18.13	Parent as Agent for Borrowers	<u>86</u>	
	18.14	Public Offer Representations	<u>87</u>	
	18.15	Acknowledgement and Consent of Bail-In of EEA Financial Institutions	<u>88</u>	
	18.16	Lender Representation	<u>89</u>	
19				

EXHIBITS AND SCHEDULES

Exhibit A-1 Form of Assignment and Acceptance
Exhibit B-1 Form of Borrowing Base Certificate

Exhibit B-2 Form of Australian Commitment Reallocation Notice

Exhibit B-3 Form of Bank Product Provider Agreement

Exhibit C-1 Form of Compliance Certificate

Exhibit L-1 Form of LIBOR Notice

Exhibit P-1 Form of U.S. Perfection Certificate
Exhibit P-2 Form of Australian Perfection Certificate

Schedule 1.1 Definitions
Schedule A-1 Agent's Account
Schedule A-2 Authorized Persons
Schedule A-3 Competitors

Schedule C-1
Schedule D-1
U.S. Designated Account
Schedule D-2
Australian Designated Account
Schedule E-1
Existing Hedge Obligations
Schedule E-2
Existing Letters of Credit

Schedule E-3 Excluded Subsidiary Indebtedness

Schedule I-1 Immaterial Subsidiaries Schedule P-1 Permitted Investments Schedule P-2 Permitted Liens Schedule P-3 **Permitted Dispositions** Schedule P-4 Permitted Indebtedness Schedule 3.1 Conditions Precedent Schedule 3.7 Conditions Subsequent Schedule 4.1(b) Capitalization of Borrowers

Schedule 4.1(c) Capitalization of Borrowers' Subsidiaries Schedule 4.1(d) Subscriptions, Options, Warrants, Calls

Schedule 4.11 Environmental Matters

Schedule 5.1 Financial Statements, Reports, Certificates

Schedule 5.16(a) ADI Accounts
Schedule 5.16(b) Deposit Accounts
Schedule 5.2 Collateral Reporting

AMENDED AND RESTATED SYNDICATED FACILITY AGREEMENT

THIS AMENDED AND RESTATED SYNDICATED FACILITY AGREEMENT, is dated as of March 30, 2015 and amended and restated as of February 28, 2018 by and among the lenders identified on the signature pages hereof (each of such lenders, together with its successors and permitted assigns, is referred to hereinafter as a "Lender", as that term is hereinafter further defined), BANK OF AMERICA, N.A., as administrative agent for each member of the Lender Group and the Bank Product Providers (in such capacity, together with its successors and assigns in such capacity, "Agent"), BANK OF AMERICA, N.A., as Australian security trustee (in such capacity, together with its successors and assigns in such capacity, "Australian Security Trustee"), CLEVELAND-CLIFFS INC., an Ohio corporation ("Parent"), and the Subsidiaries of Parent identified on the signature pages hereof (such Subsidiaries, together with Parent, are referred to hereinafter each individually as a "Borrower", and individually and collectively as the "Borrowers").

WHEREAS, the Borrowers, Agent and Australian Security Trustee are party to that certain Syndicated Facility Agreement, dated as of March 30, 2015 (as amended, supplemented or otherwise modified prior to the date hereof, including pursuant to that certain First Amendment to Syndicated Facility Agreement dated as of June 17, 2016, the "Existing Syndicated Facility Agreement");

WHEREAS, the parties hereto desire to amend and restate the Existing Syndicated Facility Agreement as provided in this Agreement (as defined in Schedule 1.1 hereto), subject to the terms and conditions set forth in Sections 3.1 and 3.2;

NOW, THEREFORE, the Existing Syndicated Facility Agreement is amended and restated in its entirety as follows:

- 1 DEFINITIONS AND CONSTRUCTION.
 - 1.1 <u>Definitions</u>. Capitalized terms used in this Agreement shall have the meanings specified therefor on Schedule 1.1.
- 1.2 Accounting Terms. All accounting terms not specifically defined herein shall be construed in accordance with GAAP; provided, that if Parent notifies Agent that Borrowers request an amendment to any provision hereof to eliminate the effect of any Accounting Change occurring after the Closing Date or in the application thereof on the operation of such provision (or if Agent notifies Borrowers that the Required Lenders request an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such Accounting Change or in the application thereof, then Agent and Borrowers agree that they will negotiate in good faith amendments to the provisions of this Agreement that are directly affected by such Accounting Change with the intent of having the respective positions of the Lenders and Borrowers after such Accounting Change conform as nearly as possible to their respective positions as of the date of this Agreement and, until any such amendments have been agreed upon and agreed to by the Required Lenders, the provisions in this Agreement shall be calculated as if no such Accounting Change had occurred; provided further that the parties hereto agree that the adoption of ASC 606 by the Borrowers and their Subsidiaries prior to the date hereof shall not constitute an Accounting Change. When used herein, the term "financial statements" shall include the notes and schedules thereto. Whenever the term "Parent" is used in respect of a financial covenant or a related definition, it shall be understood to mean Parent and its Subsidiaries on a consolidated basis, unless the context clearly requires otherwise. For purposes of calculating the U.S. Borrowing Base with respect to the U.S. Iron Ore Business, such calculation shall be on a "first-in, first-out" basis. Notwithstanding anything to the contrary contained herein, (a) all financial statements delivered hereunder shall be prepared, and all financial covenants contained herein shall be calculated, without giving effect to any election under the Statement of Financial Accounting Standards No. 159 (or any similar accounting principle) permitting a Person to value its financial liabilities or Indebtedness at the fair value thereof. (b) the term "unqualified opinion" as used herein to refer to opinions or reports provided by accountants shall mean an opinion or report that does not include any explanation, supplemental comment, or other comment concerning the ability of the applicable Person to continue as a going concern or concerning the scope of the audit (other than a "going concern" or like gualification or exception resulting solely from (i) maturity of

any Indebtedness (including the Revolver Commitments) occurring within one year from the time such opinion is delivered, and/or (ii) the projected or potential breach of any of the financial covenants set forth in this Agreement or any agreement governing any Indebtedness during the one-year period following the date such opinion is delivered), and (c) for purposes of any calculations hereunder, including calculating Fixed Charge, and the determination of Indebtedness hereunder, any lease (or similar arrangement) that would constitute an "operating lease" under GAAP as in effect on the Closing Date (or would have constituted an "operating lease" had such lease or similar arrangement been in effect on the Closing Date) shall constitute an "operating lease" hereunder and the obligations thereunder shall not constitute Capitalized Lease Obligations. For purposes of determining satisfaction of the Payment Conditions set forth in this Agreement or the financial covenant set forth in Section 7 of this Agreement, such determination shall be calculated on a *pro forma* basis (including *pro forma* adjustments arising out of events which are directly attributable to any Permitted Acquisition, Permitted Disposition or Permitted Investment that are factually supportable, and are expected to have a continuing impact, in each case determined on a basis consistent with Article 11 of Regulation S-X promulgated under the Securities Act and as interpreted by the staff of the SEC or in such other manner acceptable to Agent).

1.3 Code; Australian PPSA. Any terms used in this Agreement or the other Loan Documents (except as otherwise specifically provided herein or therein) that are defined in (a) the Code shall be construed and defined as set forth in the Code; provided, that to the extent that the Code is used to define any term herein and such term is defined differently in different Articles of the Code, the definition of such term contained in Article 9 of the Code shall govern, and (b) the Australian PPSA shall be construed and defined as set forth in the Australian PPSA. Notwithstanding the foregoing, and where the context so requires, (i) any term defined in this Agreement or the Loan Documents by reference to the "Code", "UCC" or the "Uniform Commercial Code" shall also have any extended, alternative or analogous meaning given to such term in applicable Australian PPS Law, in all cases for the extension, preservation or betterment of the security and rights of the Collateral, (ii) all references in this Agreement to a financing statement, continuation statement, amendment or termination statement shall be deemed to refer also to the analogous documents used under applicable Australian PPS Law, (iii) all references to the United States of America, or to any subdivision, department, agency or instrumentality thereof, and (iv) all references to federal or state securities laws of the United States shall be deemed to refer also to analogous federal and state securities laws in Australia.

1.4 Construction.

- (a) Unless the context of this Agreement or any other Loan Document clearly requires otherwise, references to the plural include the singular, references to the singular include the plural, the terms "includes" and "including" are not limiting, and the term "or" has, except where otherwise indicated, the inclusive meaning represented by the phrase "and/or." The words "hereof," "herein," "hereby," "hereunder," and similar terms in this Agreement or any other Loan Document refer to this Agreement or such other Loan Document, as the case may be, as a whole and not to any particular provision of this Agreement or such other Loan Document, as the case may be. Section, subsection, clause, schedule, and exhibit references herein are to this Agreement unless otherwise specified. Any reference in this Agreement or in any other Loan Document to any agreement, instrument, or document shall include all alterations, amendments, changes, extensions, modifications, renewals, replacements, thereto and thereof, as applicable (subject to any restrictions on such alterations, amendments, changes, extensions, modifications, renewals, replacements, substitutions, joinders, and supplements and supplements set forth herein). The words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties. Any reference herein to any Person shall be construed to include such Person's successors and assigns. Any requirement of a writing contained herein or in any other Loan Document shall be satisfied by the transmission of a Record.
- (b) Any reference herein or in any other Loan Document to the satisfaction, repayment, or payment in full of the Obligations shall mean (i) the payment or repayment in full in immediately available funds of (A) the principal amount of, and interest accrued and unpaid with respect to, all outstanding Loans, together with the payment of any premium applicable to the repayment of the Loans, (B) all Lender Group Expenses that have accrued and are unpaid for which an invoice has been provided to Parent, and (C) all fees or charges that have accrued hereunder or under any other Loan Document (including the Letter of Credit Fees and the Unused Line Fee) and are unpaid, (ii) in the case of contingent reimbursement obligations with respect to Letters of Credit, providing Letter of Credit

Collateralization, (iii) in the case of obligations with respect to Bank Products (other than Hedge Obligations), providing Bank Product Collateralization, (iv) the receipt by Agent of cash collateral in order to secure any other contingent Obligations for which a claim or demand for payment has been made on or prior to such time or in respect of matters or circumstances known to Agent or a Lender at such time that are reasonably expected to result in any loss, cost, damage, or expense (including attorneys' fees and legal expenses), such cash collateral to be in such amount as Agent reasonably determines is appropriate to secure such contingent Obligations, (v) the payment or repayment in full in immediately available funds of all other outstanding Obligations (including the payment of any termination amount then applicable (or which would or could become applicable as a result of the repayment of the other Obligations) under Hedge Agreements provided by Hedge Providers) other than (A) unasserted contingent indemnification Obligations, (B) any Bank Product Obligations (other than Hedge Obligations) that, at such time, are allowed by the applicable Hedge Provider to remain outstanding without being required to be repaid, and (V) any Hedge Obligations that, at such time, are allowed by the applicable Hedge Provider to remain outstanding without being required to be repaid, and (Vi) the termination of all of the Revolver Commitments of the Lenders.

- (c) Any reference herein or in any other Loan Document to the satisfaction, repayment, or payment in full of the U.S. Obligations shall mean (i) the payment or repayment in full in immediately available funds of (A) the principal amount of, and interest accrued and unpaid with respect to, all outstanding U.S. Revolving Loans, together with the payment of any premium applicable to the repayment of the U.S. Revolving Loans, (B) all Lender Group Expenses of Agent, the U.S. Issuing Banks, and the U.S. Revolving Lenders that have accrued and are unpaid and for which an invoice has been provided to Parent in respect of the U.S. Obligations, and (C) all fees or charges that have accrued hereunder or under any other Loan Document with respect to the U.S. Obligations (including the U.S. Letter of Credit Fee and the Unused Line Fee) and are unpaid, (ii) in the case of contingent reimbursement obligations with respect to U.S. Letters of Credit, providing Letter of Credit Collateralization, (iii) in the case of obligations with respect to U.S. Bank Products (other than U.S. Hedge Obligations), providing Bank Product Collateralization, (iv) the receipt by Agent of cash collateral in order to secure any other contingent U.S. Obligations for which a claim or demand for payment has been made on or prior to such time or in respect of matters or circumstances known to Agent or a U.S. Revolving Lender at such time that are reasonably expected to result in any loss, cost, damage, or expense (including attorneys' fees and legal expenses), such cash collateral to be in such amount as Agent reasonably determines is appropriate to secure such contingent U.S. Obligations, (v) the payment or repayment in full in immediately available funds of all other outstanding U.S. Obligations (including the payment of any termination amount then applicable (or which would or could become applicable as a result of the repayment of the other U.S. Obligations) under Hedge Agreements provided by U.S. Hedge Providers) other than (A) unasserted contingent indemnification U.S. Obligations, (B) any U.S. Bank Product Obligations (other than U.S. Hedge Obligations) that, at such time, are allowed by the applicable U.S. Bank Product Provider to remain outstanding without being required to be repaid or cash collateralized, and (C) any U.S. Hedge Obligations that, at such time, are allowed by the applicable U.S. Hedge Provider to remain outstanding without being required to be repaid, and (vi) the termination of all of the U.S. Revolver Commitments of the U.S. Revolving Lenders.
- (d) Any reference herein or in any other Loan Document to the satisfaction, repayment, or payment in full of the Australian Obligations shall mean (i) the payment or repayment in full in immediately available funds of (A) the principal amount of, and interest accrued and unpaid with respect to, all outstanding Australian Revolving Loans, together with the payment of any premium applicable to the repayment of the Australian Revolving Loans, (B) all Lender Group Expenses of Agent, the Australian Security Trustee, the Australian Issuing Banks, and the Australian Revolving Lenders that have accrued and are unpaid and for which an invoice has been provided to Parent in respect of the Australian Obligations, and (C) all fees or charges that have accrued hereunder or under any other Loan Document with respect to the Australian Obligations (including the Australian Letter of Credit Fee) and are unpaid, (ii) in the case of contingent reimbursement obligations with respect to Australian Letters of Credit, providing Letter of Credit Collateralization, (iii) in the case of obligations with respect to Australian Bank Products (other than Australian Hedge Obligations), providing Bank Product Collateralization, (iv) the receipt by Agent of cash collateral in order to secure any other contingent Australian Obligations for which a claim or demand for payment has been made on or prior to such time or in respect of matters or circumstances known to Agent or an Australian Revolving Lender at such time that are reasonably expected to result in any loss, cost, damage, or expense (including attorneys' fees and legal expenses), such cash collateral to be in such amount as Agent reasonably determines is appropriate to secure such contingent Australian Obligations, (v) the payment or repayment in full in immediately available funds of all other outstanding Australian Obligations (including the payment of any termination amount then applicable (or which would or could become applicable as a result of the repayment of the other Australian Obligations) under Hedge Agreements provided by Australian Hedge Providers) other than (A) unasserted contingent indemnification Australian Obligations, (B) any Australian Bank Product Obligations (other than Australian Hedge Obligations) that, at such time, are allowed by the applicable Australian Bank Product Provider to remain outstanding without being required to be repaid or cash

collateralized, and (C) any Australian Hedge Obligations that, at such time, are allowed by the applicable Australian Hedge Provider to remain outstanding without being required to be repaid, and (vi) the termination of all of the Australian Revolver Commitments of the Australian Revolving Lenders.

- (e) Any reference herein or in any other Loan Document to any law, statute, rule or regulation shall include all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such law.
- (f) Any reference in this Agreement to Liens stated to be in favor of Agent shall be construed so as to include a reference to Liens granted in favor of Australian Security Trustee.
- 1.5 <u>Time References</u>. Unless the context of this Agreement or any other Loan Document clearly requires otherwise, all references to time of day refer to (i) Chicago time for U.S. Revolving Loans and (ii) Sydney time for Australian Revolving Loans. For purposes of the computation of a period of time from a specified date to a later specified date, the word "from" means "from and including" and the words "to" and "until" each means "to and including"; provided that, with respect to a computation of fees or interest payable to Agent or any Lender, such period shall include the first day, but not the last day so long as payment thereof is received prior to the time specified in Section 2.6 hereof, but in any event shall consist of at least one full day.
 - 1.6 Schedules and Exhibits. All of the schedules and exhibits attached to this Agreement shall be deemed incorporated herein by reference.

1.7 **Dollar Equivalent.**

- (a) Agent shall determine the Dollar Equivalent of any Letter of Credit not denominated in Dollars (i) on the date of issuance thereof and the date of any amendment thereto that increases the face amount thereof, in each case using the Spot Rate for the Agreed Currency other than Dollars in relation to Dollars in effect on date immediately prior to the applicable issuance or amendment date, and (ii) at such other times as may be determined by either Agent or an Issuing Bank in its Permitted Discretion, in each case using the Spot Rate for the Agreed Currency other than Dollars in relation to Dollars in effect on the date of determination. Each amount determined pursuant to this <u>Section 1.7(a)</u>, shall be the Dollar Equivalent of the applicable Letter of Credit until the next calculation thereof pursuant to the preceding sentences of this <u>Section 1.7(a)</u>, absent manifest error. Upon the request of the Borrowers, Agent shall notify Borrowers and the applicable Lenders of each calculation of the Dollar Equivalent of each Letter of Credit denominated in an Agreed Currency other than Dollars.
- (b) Wherever in this Agreement in connection any Letter of Credit, an amount, such as a required minimum, sublimit, maximum, or multiple amount, is expressed in Dollars, but such Letter of Credit is denominated in an Agreed Currency other than Dollars, such amount shall be the Agreed Currency Equivalent of such Dollar amount (rounded to the nearest unit of such Agreed Currency that is equivalent to one Dollar, with 0.5 of a unit being rounded upward).
- (c) Principal, interest, reimbursement obligations, cash collateral for reimbursement obligations, fees, and all other amounts payable to Agent or Lenders under this Agreement and the other Loan Documents shall be payable (except as otherwise specifically provided herein) in Dollars.
- (d) If at any time following one or more fluctuations in the exchange rate of any Agreed Currency other than Dollars against the Dollar, all or any part of the Obligations exceeds more than 102% of any other limit set forth herein for such Obligations, the Borrowers of such Obligations shall within 1 Business Day of written notice of same from Agent immediately make the necessary payments or repayments to reduce such Obligations, or Cash Collateralize such Obligations, to an amount necessary to eliminate such excess over 100% of such limit.

- 1.8 <u>LIBOR Rate</u>. Agent does not warrant, nor accept responsibility, nor shall Agent have any liability with respect to the administration, submission, or any other matter related to the rates in the definition of LIBOR Rate or with respect to any comparable or successor rate thereto (including as may be provided pursuant to <u>Section 2.13(d)(iv)</u>).
- 1.9 <u>Borrowers</u>. Except as otherwise set forth herein, any reference herein to the "applicable Borrowers" or "Borrowers" making payment or receiving extensions of credit in respect of Obligations shall be deemed to refer to the U.S. Borrowers making payment of, or receiving extensions of credit that constitute, U.S. Obligations and the Australian Borrowers making payment of, or receiving extensions of credit that shall constitute, Australian Obligations.
- 1.10 <u>Australian Terms</u>. Without prejudice to the generality of any provision of this Agreement, in this Agreement where it relates to an Australian Loan Party or any of its Subsidiaries incorporated under the laws of Australia or any state or territory thereof, a reference in this Agreement to:
 - (a) "Affiliate" has the meaning given to it in section 50AA of the Corporations Act;
 - (b) "Authorized Person" means the company secretary or any director of that Australian Loan Party or Subsidiary;
 - (c) "Controller" has the meaning given to it in Section 9 of the Corporations Act;
 - (d) "Fair Labor Standards Act" or "Worker Adjustment and Retraining Notification Act" means the Fair Work Act 2009 (Cth) of Australia;
 - (e) "Lien" also includes any 'security interest' as defined in sections 12(1) and 12(2) of the Australian PPSA;
 - (f) "Solvent" means that it is not 'insolvent' within the meaning of section 95A(2) of the Corporations Act; and
 - (g) a "Subsidiary" means a subsidiary within the meaning given in Part 1.2 Division 6 of the Corporations Act.

2 REVOLVING LOANS AND TERMS OF PAYMENT.

2.1 Revolving Loans.

- (a) Subject to the terms and conditions of this Agreement, and during the term of this Agreement, each U.S. Revolving Lender agrees (severally, not jointly or jointly and severally) to make revolving loans ("<u>U.S. Revolving Loans</u>") in Dollars to U.S. Borrowers in a principal amount not to exceed the lesser of:
 - (i) such Lender's U.S. Revolver Commitment at such time, or
- (ii) such Lender's Pro Rata Share of an amount equal to (x) the U.S. Line Cap at such time less (y) the sum of (A) the U.S. Letter of Credit Usage at such time, (B) the principal amount of U.S. Swing Loans outstanding at such time and (C) the principal amount of U.S. Revolving Loans outstanding at such time.

Anything to the contrary in this <u>Section 2.1(a)</u> notwithstanding, Agent shall have the right (but not the obligation), in the exercise of its Permitted Discretion, to establish and increase or decrease U.S. Receivable Reserves, U.S. Inventory/Equipment Reserves, U.S. Bank Product Reserves, U.S. Dilution Reserves and other Reserves against the U.S. Borrowing Base or the U.S. Maximum Revolver Amount, including based on review of the Borrowers' employment

and labor contracts; provided that no reserve shall be established, increased or decreased except upon not less than three Business Days' prior written notice from the Agent to the Parent, during which period employees, agents or other representatives of the Agent shall use commercially reasonable efforts to be available during regular business hours to discuss any such proposed establishment or modification of such reserve with the Parent and, without limiting the right of the Agent to establish or modify reserves in its Permitted Discretion, the Parent may take such action as may be required so that the circumstances, conditions, events or contingencies that are the basis for such reserve or modification thereof no longer exist, in a manner and to the extent reasonably satisfactory to the Agent in its Permitted Discretion; provided that no such prior written notice shall be required for any modifications to any reserves during any Cash Dominion Trigger Period, during a Financial Covenant Period, during any period of weekly collateral reporting as provided in Section 5.2, or after the occurrence and during the continuance of any Event of Default, in each case resulting by virtue of mathematical calculations of the amount of the reserves in accordance with the methodology of calculation previously utilized. The amount of any U.S. Receivable Reserve, U.S. Inventory/Equipment Reserve, U.S. Bank Product Reserve, or other Reserve established by Agent shall have a reasonable relationship to the event, condition, other circumstance, or fact that is the basis for such reserve and shall not be duplicative of any other reserve established and currently maintained. Upon establishment or increase in reserves, Agent agrees to make itself available to discuss the reserve or increase, and Borrowers may take such action as may be required so that the event, condition, circumstance, or fact that is the basis for such reserve or increase no longer exists, in a manner and to the extent reasonably satisfactory to Agent in the exercise of its Permitted Discretion. In no event shall such notice and opportunity limit the right of Agent to establish or change such U.S. Receivable Reserve, U.S. Inventory/Equipment Reserve, U.S. Bank Product Reserve, or other Reserves, unless Agent shall have determined, in its Permitted Discretion, that the event, condition, other circumstance, or fact that was the basis for such U.S. Receivable Reserve, U.S. Inventory/Equipment Reserve, U.S. Bank Product Reserve, or other Reserves or such change no longer exists or has otherwise been adequately addressed by Borrowers.

- (b) Subject to the terms and conditions of this Agreement, and during the term of this Agreement, each Australian Revolving Lender agrees (severally, not jointly or jointly and severally) to make revolving loans ("Australian Revolving Loans") and together with the U.S. Revolving Loans, "Revolving Loans" in Dollars to Australian Borrowers in a principal amount not to exceed the lesser of:
 - (i) such Lender's Australian Revolver Commitment at such time, or
- (ii) such Lender's Pro Rata Share of an amount equal to (x) the Australian Line Cap at such time less (y) the sum of (A) the Australian Letter of Credit Usage at such time, (B) the principal amount of Australian Swing Loans outstanding at such time and (C) the principal amount of Australian Revolving Loans outstanding at such time.

Anything to the contrary in this Section 2.1(b) notwithstanding, Agent shall have the right (but not the obligation), in the exercise of its Permitted Discretion, to establish and increase or decrease Australian Receivable Reserves, Australian Inventory/Equipment Reserves, Australian Bank Product Reserves, Australian Priority Payables Reserves, Australian Dilution Reserves and other Reserves against the Australian Borrowing Base or the Australian Maximum Revolver Amount, including based on review of the Australian Borrowers' employment and labor contracts; provided that no reserve shall be established, increased or decreased except upon not less than three Business Days' prior written notice from the Agent to the Parent, during which period employees, agents or other representatives of the Agent shall use commercially reasonable efforts to be available during regular business hours to discuss any such proposed establishment or modification of such reserve with the Parent and, without limiting the right of the Agent to establish or modify reserves in its Permitted Discretion, the Parent may take such action as may be required so that the circumstances, conditions, events or contingencies that are the basis for such reserve or modification thereof no longer exist, in a manner and to the extent reasonably satisfactory to the Agent in its Permitted Discretion; provided that no such prior written notice shall be required for any modifications to any reserves during a Financial Covenant Period, during any period of weekly collateral reporting as provided in Section 5.2, or after the occurrence and during the continuance of any Event of Default, in each case resulting by virtue of mathematical calculations of the amount of the reserves in accordance with the methodology of calculation previously utilized. The amount of any Australian Receivable Reserve, Australian Inventory/Equipment Reserve, Australian Bank Product Reserve, Australian Priority Payables Reserves, Australian Dilution Reserve, or other Reserve established by Agent shall have a reasonable relationship to the event, condition, other circumstance, or fact that is the basis for such reserve as determined by Agent in its Permitted Discretion and shall not be duplicative of any other reserve established and currently maintained. Upon establishment or increase in reserves, Agent agrees to make itself available to discuss the reserve or increase, and Borrowers may take such action as may be required so that the event, condition, circumstance, or fact that is the basis for such reserve or increase no longer exists, in a manner and to the extent reasonably satisfactory to Agent in

the exercise of its Permitted Discretion. In no event shall such notice and opportunity limit the right of Agent to establish or change such Australian Receivable Reserve, Australian Inventory/Equipment Reserve, Australian Bank Product Reserve, Australian Priority Payables Reserves, Australian Dilution Reserve or other Reserves, unless Agent shall have determined, in its Permitted Discretion, that the event, condition, other circumstance, or fact that was the basis for such Receivable Reserve, Inventory Reserve, Bank Product Reserve, or other Reserves or such change no longer exists or has otherwise been adequately addressed by Borrowers.

- (c) The outstanding principal amount of the U.S. Revolving Loans, together with interest accrued and unpaid thereon, shall constitute U.S. Obligations and shall be due and payable on the Maturity Date or, if earlier, on the date on which they are declared due and payable pursuant to the terms of this Agreement.
- (d) The outstanding principal amount of the Australian Revolving Loans, together with interest accrued and unpaid thereon, shall constitute Australian Obligations and shall be due and payable on the Maturity Date or, if earlier, on the date on which they are declared due and payable pursuant to the terms of this Agreement.
- (e) Amounts borrowed pursuant to this <u>Section 2.1</u> may be repaid and, subject to the terms and conditions of this Agreement, reborrowed at any time during the term of this Agreement.
- 2.2 <u>Reallocation of the Australian Revolver Commitments</u>. For so long as any Revolver Commitments are in effect and/or any Obligations are outstanding, all or a portion of the Australian Revolver Commitments may be permanently reallocated as U.S. Revolver Commitments in accordance with this <u>Section 2.2</u>.
- (a) So long as no Default or Event of Default shall have occurred and be continuing, from time to time but in no event more than (A) two (2) times per twelve-month period, upon at least five (5) Business Days (or such shorter period as Agent may agree) prior written notice to the Agent, the Borrowers may request that all or a portion of the Australian Revolver Commitments be permanently reallocated (any such reallocation, an "Australian Commitment By providing a written notice to the Agent requesting such Australian Commitment Reallocation (each such notice herein an "Australian Commitment Reallocation Notice"); provided that at no time shall (i) the Revolver Commitment of any Lender be decreased by virtue of any reallocation under this Section 2.2(a) or (ii) any Lender be required to agree to reallocate any portion of its Australian Revolver Commitment as a U.S. Revolving Commitment.

(b) [reserved]

- (c) In connection with each Australian Commitment Reallocation Notice, the Agent shall calculate and provide notice to each Lender of its Pro Rata Share of the U.S. Revolver Commitments and the Australian Revolver Commitments. On each date on which a reallocation of the Revolver Commitments pursuant to an Australian Commitment Reallocation Notice which results in a change in the any Lender's Pro Rata Share of the U.S. Revolver Commitments and Australian Revolver Commitments becomes effective, if any Loans or Letters of Credit are outstanding as of such date, then the outstanding principal balance of the Loans (and participations in Letters of Credit and Swing Loans) shall be reallocated in accordance with such new Pro Rata Share of the U.S. Revolver Commitments and Australian Revolver Commitments and each Lender whose Pro Rata Share has increased shall, by wire transfer of immediately available funds, purchase a pro rata portion of the outstanding Loans (and participation interests in Letters of Credit and Swing Loans) of each Lender whose Pro Rata Share of the Australian Revolver Commitments has decreased (and such Lenders hereby agree to sell and to take all such further action to effectuate such sale) such that each Lender shall hold its new Pro Rata Share of the outstanding Loans (and participation interests) after giving effect to such purchases and sales.
- (d) Each Australian Commitment Reallocation Notice delivered by any Borrower to the Agent shall be substantially in the form of <u>Exhibit B-2</u> or such other form as the Agent may agree.

2.3 **Borrowing Procedures and Settlements.**

(a) **Procedure for Borrowings**. Each Borrowing shall be made by a written request by an Authorized Person delivered to Agent and received by Agent no later than (i) 12:00 noon on the Business Day that is

the requested Funding Date in the case of a request for a Swing Loan or a U.S. Revolving Loan that is a Base Rate Loan, (ii) 11:00 a.m. on the Business Day that is one Business Day prior to the requested Funding Date in the case of a request for an Australian Base Rate Loan, and (iii) 12:00 noon on the Business Day that is 3 Business Days prior to the requested Funding Date in the case of all other LIBOR Rate Loans, specifying (A) the amount of such Borrowing, (B) whether such Borrowing will be a U.S. Revolving Loan or an Australian Revolving Loan, (C) the requested Funding Date (which shall be a Business Day), (D) whether such Borrowing is to be a Borrowing of Base Rate Loans, a Borrowing of LIBOR Rate Loans or a Borrowing of Australian Base Rate Loans, and (E) in the case of a Borrowing of LIBOR Rate Loans, the initial Interest Period to be applicable thereto, which shall be a period contemplated by the definition of the term "Interest Period"; provided that Agent may, in its sole discretion, elect to accept as timely requests that are received later than the times specified above on the applicable Business Day; provided, further that (i) the U.S. Borrowers shall only be permitted to borrow Base Rate Loans and LIBOR Rate Loans and LIBOR Rate Loans. In lieu of delivering the above-described written request, any Authorized Person may give Agent electronic notice of such request by the required time. In such circumstances, Borrowers agree that any such electronic notice will be confirmed in writing within 24 hours of the giving of such electronic notice, but the failure to provide such written confirmation shall not affect the validity of the request.

(b) Reserved.

(c) Making of U.S. Swing Loans. In the case of a request for a U.S. Revolving Loan and so long as either (i) the aggregate amount of U.S. Swing Loans made since the last Settlement Date, *minus* all payments or other amounts applied to U.S. Swing Loans since the last Settlement Date, *plus* the amount of the requested U.S. Swing Loan does not exceed \$100,000,000, or (ii) U.S. Swing Lender, in its sole discretion, agrees to make a U.S. Swing Loan notwithstanding the foregoing limitation, U.S. Swing Lender shall make a U.S. Revolving Loan (any such U.S. Revolving Loan made by U.S. Swing Lender pursuant to this Section 2.3(c) being referred to as "U.S. Swing Loan" and all such U.S. Revolving Loans being referred to as "U.S. Swing Loans") available to U.S. Borrowers on the Funding Date applicable thereto by transferring immediately available funds in the amount of such requested Borrowing to the U.S. Designated Account. Each U.S. Swing Loan shall be deemed to be a U.S. Revolving Loan hereunder and shall be subject to all the terms and conditions (including Section 3) applicable to other U.S. Revolving Loans, except that all payments (including interest) on any U.S. Swing Loan shall be payable to U.S. Swing Lender solely for its own account. Subject to the provisions of Section 2.3(f)(iii), U.S. Swing Lender shall not make and shall not be obligated to make any U.S. Swing Lender shall not make actual knowledge that (i) one or more of the applicable conditions precedent set forth in Section 3 will not be satisfied on the requested Funding Date for the applicable Borrowing, or (ii) the requested Borrowing would exceed the U.S. Excess Availability on such Funding Date. U.S. Swing Lender shall not otherwise be required to determine whether the applicable conditions precedent set forth in Section 3 have been satisfied on the Funding Date applicable thereto prior to making any U.S. Swing Loan. The U.S. Swing Loans shall be secured by Liens granted under the Loan Documents (other than the Australian Security Documents), co

(d) Making of Australian Swing Loans. In the case of a request for an Australian Revolving Loan and so long as either (i) the aggregate amount of Australian Swing Loans made since the last Settlement Date, *minus* all payments or other amounts applied to Australian Swing Loans since the last Settlement Date, *plus* the amount of the requested Australian Swing Loan does not exceed \$20,000,000, or (ii) Australian Swing Lender, in its sole discretion, agrees to make an Australian Swing Loan notwithstanding the foregoing limitation, Australian Swing Lender shall make an Australian Revolving Loan made by Australian Swing Lender pursuant to this Section 2.3(d) being referred to as an "Australian Swing Loan" and all such Australian Revolving Loans being referred to as "Australian Swing Loans"; and U.S. Swing Loans and Australian Swing Loan may also each be referred to herein as "Swing Loans") available to Australian Borrowers on the Funding Date applicable thereto by transferring immediately available funds in the amount of such requested Borrowing to the Australian Designated Account. Each Australian Swing Loan shall be deemed to be an Australian Revolving Loan hereunder and shall be subject to all the terms and conditions (including Section 3) applicable to other Australian Revolving Loans, except that all payments (including interest) on any Australian Swing Loan shall be payable to Australian Swing Lender solely for its own account. Subject to the provisions of Section 2.3(f)(iv). Australian Swing Lender shall not make and shall not be obligated to make any Australian Swing Loan if Australian Swing Lender has actual knowledge that (A) one or more of the applicable conditions precedent set forth in Section 3 will not be satisfied on the requested Funding Date or the applicable Borrowing, or (B) the requested Borrowing would exceed the Australian Excess Availability on such Funding Date. Australian Swing Lender shall not otherwise be required to determine whether the applicable conditions precedent set forth in Sect

Australian Swing Loan. The Australian Swing Loans shall be secured by Liens granted under the Loan Documents, constitute Australian Revolving Loans and Australian Obligations, and bear interest at the rate applicable from time to time to Australian Revolving Loans that are Base Rate Loans.

(e) Making of Revolving Loans.

- (i) In the event that a Swing Lender refuses to or is not obligated to make a Swing Loan, then after receipt of a request for a Borrowing pursuant to Section 2.3(a), Agent shall notify the applicable Lenders by telecopy, telephone, email, or other electronic form of transmission, of the requested Borrowing; such notification to be sent on the Business Day that is one Business Day prior to the requested Funding Date (or, in the case of a request for a Loan delivered on the requested Funding Date, no later than 1:00 p.m. on the requested Funding Date). If Agent has notified the Lenders of a requested Borrowing on the Business Day that is one Business Day prior to the Funding Date (or, in the case of a request for a Loan delivered on the requested Funding Date), then each Lender shall make the amount of such Lender's Pro Rata Share of the requested Borrowing available to Agent in immediately available funds, to the Agent's Account, not later than 3:00 p.m., in the case of any Loan requested to be funded on the Funding Date, or 10:00 a.m., in the case of any other request, in each case, on the Business Day that is the requested Funding Date. After Agent's receipt of the proceeds of such Revolving Loans from the Lenders, Agent shall make the proceeds thereof available to applicable Borrowers on the applicable Funding Date by transferring immediately available funds equal to such proceeds received by Agent to the U.S. Designated Account (with respect to U.S. Revolving Loans) and the Australian Designated Account (with respect to Australian Revolving Loans); provided, that, subject to the provisions of Sections 2.3(f)(iii) and 2.3(f)(iv), no Lender shall have an obligation to make any Revolving Loan, if (1) one (1) or more of the applicable conditions precedent set forth in Section 3 are not satisfied on the requested Funding Date for the applicable Borrowing unless such condition has been waived, or (2) the requested Borrowing would exceed the U.S. Excess Availability or the Australian Excess Availability on such Fun
- (ii) Unless Agent receives notice from a Lender prior to 9:30 a.m. on the Business Day that is the requested Funding Date relative to a requested Borrowing as to which Agent has notified the Lenders of a requested Borrowing that such Lender will not make available as and when required hereunder to Agent for the account of the applicable Borrowers the amount of that Lender's Pro Rata Share of the Borrowing, Agent may assume that each Lender has made or will make such amount available to Agent in immediately available funds on the Funding Date and Agent may (but shall not be so required), in reliance upon such assumption, make available to the applicable Borrowers a corresponding amount. If, on the requested Funding Date, any Lender shall not have remitted the full amount that it is required to make available to Agent in immediately available funds and if Agent has made available to the applicable Borrowers such amount on the requested Funding Date, then such Lender shall make the amount of such Lender's Pro Rata Share of the requested Borrowing available to Agent in immediately available funds, to the Agent's Account, no later than 10:00 a.m. on the Business Day that is the first Business Day after the requested Funding Date (in which case, the interest accrued on such Lender's portion of such Borrowing for the Funding Date shall be for Agent's separate account). If any Lender shall not remit the full amount that it is required to make available to Agent in immediately available funds as and when required hereby and if Agent has made available to Borrowers such amount, then that Lender shall be obligated to immediately remit such amount to Agent, together with interest at the Defaulting Lender Rate for each day until the date on which such amount is so remitted. A notice submitted by Agent to any Lender with respect to amounts owing under this Section 2.3(e)(ii) shall be conclusive, absent manifest error. If the amount that a Lender is required to remit is made available to Agent, then such payment to Agent shall constitute such Lender's Revolving Loan for all purposes of this Agreement. If such amount is not made available to Agent on the Business Day following the Funding Date, Agent will notify Borrowers of such failure to fund and, within one Business Day after receipt of such notice, the applicable Borrowers shall pay such amount to Agent for Agent's account, together with interest thereon for each day elapsed since the date of such Borrowing, at a rate per annum equal to the interest rate applicable at the time to the Revolving Loans composing such Borrowing.

(f) Protective Advances and Optional Overadvances.

(i) Any contrary provision of this Agreement or any other Loan Document notwithstanding, but subject to Section 2.3(f)(vii), at any time after the occurrence and during the continuance of a Default or an Event of Default, or that any of the other applicable conditions precedent set forth in Section 3 are not satisfied, Agent hereby is authorized by Borrowers and the Lenders, from time to time, in Agent's Permitted Discretion, to make U.S. Revolving Loans that are Base Rate Loans to, or for the benefit of, U.S. Borrowers, on behalf of the U.S. Revolving Lenders, that Agent, in its Permitted Discretion, deems necessary or desirable (1) to preserve or protect the Collateral, or any portion thereof, or (2) to enhance the collectability or likelihood of repayment of the U.S. Obligations, so long as such U.S. Revolving Loans do not cause U.S. Revolver Usage to exceed the U.S. Maximum Revolver

Amount (the U.S. Revolving Loans described in this <u>Section 2.3(f)(i)</u> shall be referred to as "<u>U.S. Protective Advances</u>"). U.S. Revolving Lenders shall participate on a pro rata basis in U.S. Protective Advances outstanding from time to time. Required Lenders may at any time revoke Agent's authority to make further Protective Advances by written notice to Agent. Agent shall use reasonable efforts to notify Parent of the existence of any U.S. Protective Advance on or about the date when made (it being understood that the failure to provide such notification to Parent shall have no effect on such U.S. Protective Advance).

- (ii) Any contrary provision of this Agreement or any other Loan Document notwithstanding, but subject to Section 2.3(f)(vii), if there is a U.S. Overadvance at any time, the excess amount shall be payable by U.S. Borrowers within one Business Day after receipt of demand from Agent, but all such U.S. Revolving Loans shall nevertheless constitute U.S. Obligations of the U.S. Borrowers secured by the applicable Collateral of the U.S. Loan Parties and entitled to all benefits of the Loan Documents (other than Australian Collateral Documents). Agent may require U.S. Revolving Lenders to honor requests for U.S. Overadvance Loans and to forbear from requiring U.S. Borrowers to cure a U.S. Overadvance, (a) when no other Event of Default is known to the Agent, as long as (i) the U.S. Overadvance does not continue for more than 30 consecutive days (and no U.S. Overadvance may exist for at least five (5) consecutive days thereafter before further U.S. Overadvance Loans are required), and (ii) the U.S. Overadvance is not known by Agent to exceed 5.0% of the U.S. Maximum Revolver Amount and (b) regardless of whether an Event of Default exists, if Agent discovers a U.S. Overadvance not previously known by it to exist, as long as from the date of such discovery the U.S. Overadvance is not increased and does not continue for more than 30 consecutive days. In no event shall U.S. Overadvance Loans be required that would cause (x) U.S. Revolver Usage (including for this purpose, the aggregate principal amount of U.S. Overadvance Loans) to exceed the aggregate principal amount of U.S. Overadvance Loans to exceed the aggregate principal amount of U.S. Overadvance Loans to exceed the aggregate principal amount of U.S. Overadvance Loans) to exceed the aggregate principal amount of U.S. Overadvance Loans to exceed the aggregate principal amount of U.S. Overadvance Loans to exceed the Event of Default caused thereby. Agent shall use reasonable efforts to notify Parent of the existence of any U.S. Overadvance on or about the date when m
- (iii) Each U.S. Protective Advance and each U.S. Overadvance (each, a "<u>U.S. Extraordinary Advance</u>") shall be deemed to be a U.S. Revolving Loan hereunder, except that no U.S. Extraordinary Advance shall be eligible to be a LIBOR Rate Loan and, prior to Settlement therefor, all payments on the U.S. Extraordinary Advances shall be payable to Agent solely for its own account. The U.S. Extraordinary Advances shall be repayable within one Business Day of demand thereof, secured by the Liens granted under the Loan Documents (other than the Australian Security Documents), constitute U.S. Obligations hereunder, and bear interest at the rate applicable from time to time to U.S. Revolving Loans that are Base Rate Loans.
- (iv) Any contrary provision of this Agreement or any other Loan Document notwithstanding, but subject to Section 2.3(f)(vii), at any time after the occurrence and during the continuance of a Default or an Event of Default, or that any of the other applicable conditions precedent set forth in Section 3 are not satisfied, Agent hereby is authorized by Borrowers and the Lenders, from time to time, in Agent's Permitted Discretion, to make Australian Revolving Loans that are Base Rate Loans to, or for the benefit of, Australian Borrowers, on behalf of the Australian Revolving Lenders, that Agent, in its Permitted Discretion, deems necessary or desirable (1) to preserve or protect the Collateral, or any portion thereof, or (2) to enhance the collectability or the likelihood of repayment of the Australian Obligations, so long as such Australian Revolving Loans do not cause Australian Revolver Usage to exceed the Australian Maximum Revolver Amount (the Australian Revolving Loans described in this Section 2.3(f)(iv) shall be referred to as "Australian Protective Advances"; and U.S. Protective Advances and Australian Protective Advances may also each be referred to herein as "Protective Advances"). Australian Revolving Lenders shall participate on a pro rata basis in Australian Protective Advances outstanding from time to time. Required Lenders may at any time revoke Agent's authority to make further Protective Advances by written notice to Agent. Agent shall use reasonable efforts to notify Parent of the existence of any Australian Protective Advance on or about the date when made (it being understood that the failure to provide such notification to Parent shall not invalidate such Australian Protective Advance).
- (v) Any contrary provision of this Agreement or any other Loan Document notwithstanding, but subject to Section 2.3(f)(vii), if there is an Australian Overadvance at any time, the excess amount shall be payable by Australian Borrowers within one Business Day after receipt of demand from Agent, but all such Australian Revolving Loans shall nevertheless constitute Australian Obligations secured by the applicable Collateral and entitled to all benefits of the Loan Documents. Agent may require Australian Revolving Lenders to honor requests for Australian Overadvance Loans and to forbear from requiring Australian Borrowers to cure an. Australian Overadvance, (a) when no other Event of Default is known to the Agent, as long as (i) the Australian Overadvance

does not continue for more than 30 consecutive days (and no Australian Overadvance may exist for at least five (5) consecutive days thereafter before further Australian Overadvance Loans are required), and (ii) the Australian Overadvance is not known by Agent to exceed 5.0% of the Australian Maximum Revolver Amount and (b) regardless of whether an Event of Default exists, if Agent discovers an Australian Overadvance not previously known by it to exist, as long as from the date of such discovery the Australian Overadvance is not increased and does not continue for more than 30 consecutive days. In no event shall Australian Overadvance Loans be required that would cause (x) Australian Revolver Usage (including, for this purpose, the aggregate principal amount of Australian Revolver Usage (including, for this purpose, the aggregate principal amount of Australian Overadvance Loans) to exceed its Australian Revolver Commitments or (y) any Australian Revolving Lenders' Pro Rata Share of Australian Revolver Usage (including, for this purpose, the aggregate principal amount of Australian Overadvance Loans) to exceed its Australian Revolver Commitment. Any funding of an Australian Overadvance Loan or sufferance of an. Australian Overadvance shall not constitute a waiver by Agent or Lenders of the Event of Default caused thereby. Agent shall use reasonable efforts to notify Parent of the existence of any Australian Overadvance)

- (vi) Each Australian Protective Advance and each Australian Overadvance (each, an "<u>Australian Extraordinary Advances</u>"; and U.S. Extraordinary Advances and Australian Extraordinary Advances may also each be referred to herein as "<u>Extraordinary Advances</u>") shall be deemed to be an Australian Revolving Loan hereunder, except that no Australian Extraordinary Advance shall be eligible to be a LIBOR Rate Loan and, prior to Settlement therefor, all payments on the Australian Extraordinary Advances shall be payable to Agent solely for its own account. The Australian Extraordinary Advances shall be repayable within one Business Day of demand thereof, secured by the Liens granted under the Loan Documents, constitute Australian Obligations hereunder, and bear interest at the rate applicable from time to time to Australian Revolving Loans that are Base Rate Loans.
- (vii) Notwithstanding anything contained in this Agreement or any other Loan Document to the contrary: (A) no U.S. Extraordinary Advance may be made by Agent if such U.S. Extraordinary Advance would cause the aggregate principal amount of U.S. Extraordinary Advances outstanding to exceed an amount equal to 10% of the U.S. Maximum Revolver Amount and (B) no Australian Extraordinary Advance may be made by Agent if such Australian Extraordinary Advance would cause the aggregate principal amount of Australian Extraordinary Advances outstanding to exceed an amount equal to 10% of the Australian Maximum Revolver Amount.
- (viii) The provisions of this <u>Section 2.3(f)</u> are for the exclusive benefit of Agent, Swing Lender, and the Lenders and are not intended to benefit Borrowers (or any other Loan Party) in any way.

(g) Settlement.

- (i) <u>U.S. Obligations</u>. It is agreed that each U.S. Revolving Lender's funded portion of the U.S. Revolving Loans is intended by the U.S. Revolving Lenders to equal, at all times, such Lender's Pro Rata Share of the outstanding U.S. Revolving Loans. Such agreement notwithstanding, Agent, U.S. Swing Lender, and the other U.S. Revolving Lenders agree (which agreement shall not be for the benefit of Borrowers) that in order to facilitate the administration of this Agreement and the other Loan Documents, Settlement among the U.S. Revolving Lenders as to the U.S. Revolving Loans, the U.S. Swing Loans, and the U.S. Extraordinary Advances shall take place on a periodic basis in accordance with the following provisions:
- (A) Agent shall request settlement ("Settlement") with the U.S. Revolving Lenders on a weekly basis, or on a more frequent basis if so determined by Agent in its sole discretion (1) on behalf of U.S. Swing Lender, with respect to the outstanding U.S. Swing Loans, (2) for itself, with respect to the outstanding U.S. Extraordinary Advances, and (3) with respect to U.S. Borrowers' or any of their Subsidiaries' payments or other amounts received, as to each by notifying the U.S. Revolving Lenders by telecopy, telephone, email or other electronic form of transmission, of such requested Settlement, no later than 2:00 p.m. on the Business Day immediately prior to the date of such requested Settlement (the date of such requested Settlement being the "Settlement Date"). Such notice of a Settlement Date shall include a summary statement of the amount of outstanding U.S. Revolving Loans, U.S. Swing Loans, and U.S. Extraordinary Advances for the period since the prior Settlement Date. Subject to the terms and conditions contained herein (including Section 2.3(i)): (y) if the amount of the U.S. Revolving Loans (including U.S. Swing Loans, and U.S. Extraordinary Advances) made by a U.S. Revolving Lender that is not a Defaulting Lender exceeds such U.S. Revolving Lender's Pro Rata Share of the U.S. Revolving Loans (including U.S. Swing Loans, and U.S. Extraordinary Advances) as of a Settlement Date, then Agent shall, by no later than 12:00 p.m. on the Settlement Date, transfer in immediately available funds to a Deposit Account of such U.S. Revolving Lender (as such U.S. Revolving Lender may designate), an amount such that each such U.S. Revolving Lender shall, upon receipt of such

amount, have as of the Settlement Date, its Pro Rata Share of the U.S. Revolving Loans (including U.S. Swing Loans, and U.S. Extraordinary Advances), and (z) if the amount of the U.S. Revolving Loans (including U.S. Swing Loans, and U.S. Extraordinary Advances) made by a U.S. Revolving Lender is less than such Lender's Pro Rata Share of the U.S. Revolving Loans (including U.S. Swing Loans, and U.S. Extraordinary Advances) as of a Settlement Date, such U.S. Revolving Lender shall no later than 12:00 p.m. on the Settlement Date transfer in immediately available funds to Agent's Account, an amount such that each such U.S. Revolving Lender shall, upon transfer of such amount, have as of the Settlement Date, its Pro Rata Share of the U.S. Revolving Loans (including U.S. Swing Loans and U.S. Extraordinary Advances). Such amounts made available to Agent under clause (z) of the immediately preceding sentence shall be applied against the amounts of the applicable U.S. Swing Loans or U.S. Extraordinary Advances representing U.S. Swing Loans or U.S. Extraordinary Advances representing U.S. Swing Lender's Pro Rata Share thereof, shall constitute U.S. Revolving Loans of such Lenders. If any such amount is not made available to Agent by any U.S. Revolving Lender on the Settlement Date applicable thereto to the extent required by the terms hereof, Agent shall be entitled to recover for its account such amount on demand from such U.S. Revolving Lender together with interest thereon at the Defaulting Lender Rate.

- (B) In determining whether a U.S. Revolving Lender's balance of the U.S. Revolving Loans, U.S. Swing Loans, and U.S. Extraordinary Advances is less than, equal to, or greater than such U.S. Revolving Lender's Pro Rata Share of the U.S. Revolving Loans, U.S. Swing Loans, and U.S. Extraordinary Advances as of a Settlement Date, Agent shall, as part of the relevant Settlement, apply to such balance the portion of payments actually received in good funds by Agent with respect to principal, interest, fees payable by U.S. Borrowers and allocable to the U.S. Revolving Lenders hereunder, and proceeds of Collateral securing the U.S. Obligations.
- (C) Between Settlement Dates, Agent, to the extent U.S. Extraordinary Advances or U.S. Swing Loans are outstanding, may pay over to Agent or U.S. Swing Lender, as applicable, any payments or other amounts received by Agent, that in accordance with the terms of this Agreement would be applied to the reduction of the U.S. Revolving Loans, for application to the U.S. Extraordinary Advances or U.S. Swing Loans. Between Settlement Dates, Agent, to the extent no U.S. Extraordinary Advances or U.S. Swing Loans are outstanding, may pay over to U.S. Swing Lender any payments or other amounts received by Agent, that in accordance with the terms of this Agreement would be applied to the reduction of the U.S. Revolving Loans, for application to U.S. Swing Lender's Pro Rata Share of the U.S. Revolving Loans. If, as of any Settlement Date, payments or other amounts of Parent and its Subsidiaries received since the then immediately preceding Settlement Date have been applied to U.S. Swing Lender's Pro Rata Share of the U.S. Revolving Loans other than to U.S. Swing Loans, as provided for in the previous sentence, U.S. Swing Lender shall pay to Agent for the accounts of the U.S. Revolving Lenders, and Agent shall pay to the U.S. Revolving Lenders (other than a Defaulting Lender if Agent has implemented the provisions of Section 2.3(i)), to be applied to the outstanding U.S. Revolving Loans of such U.S. Revolving Lender shall, upon receipt of such amount, have, as of such Settlement Date, its Pro Rata Share of the U.S. Revolving Loans. During the period between Settlement Dates, U.S. Swing Lender with respect to U.S. Swing Loans, Agent with respect to U.S. Extraordinary Advances, and each U.S. Revolving Lender with respect to the U.S. Revolving Loans other than U.S. Swing Loans and U.S. Extraordinary Advances, shall be entitled to interest at the applicable rate or rates payable under this Agreement on the daily amount of funds employed by U.S. Swing Lender, Agent, or the U.S. Revolving Lenders, as applicable.
- (ii) <u>Australian Obligations</u>. It is agreed that each Australian Revolving Lender's funded portion of the Australian Revolving Loans is intended by the U.S. Revolving Lenders to equal, at all times, such Lender's Pro Rata Share of the outstanding Australian Revolving Loans. Such agreement notwithstanding, Agent, Australian Swing Lender, and the other Australian Revolving Lenders agree (which agreement shall not be for the benefit of Borrowers) that in order to facilitate the administration of this Agreement and the other Loan Documents, Settlement among the Lenders as to the Australian Revolving Loans, the Australian Swing Loans, and the Australian Extraordinary Advances shall take place on a periodic basis in accordance with the following provisions:
- (A) Agent shall request Settlement with the Australian Revolving Lenders on a weekly basis, or on a more frequent basis if so determined by Agent in its sole discretion (1) on behalf of Australian Swing Lender, with respect to the outstanding Australian Swing Loans, (2) for itself, with respect to the outstanding Australian Extraordinary Advances, and (3) with respect to Australian Borrowers or any of their respective Subsidiaries' payments or other amounts received, as to each by notifying the Australian Revolving Lenders by telecopy, telephone, email or other electronic form of transmission, of such requested Settlement, no later than 2:00 p.m. on the Business Day immediately prior to the Settlement Date. Such notice of a Settlement Date shall include a summary statement of the amount of outstanding Australian Revolving Loans, Australian Swing Loans, and Australian Extraordinary Advances for the period since the prior Settlement Date. Subject to the terms and conditions contained herein (including

Section 2.3(i)): (y) if the amount of the Australian Revolving Loans (including Australian Swing Loans, and Australian Extraordinary Advances) made by an Australian Revolving Lender that is not a Defaulting Lender exceeds such Australian Revolving Lender's Pro Rata Share of the Australian Revolving Loans (including Australian Swing Loans, and Australian Extraordinary Advances) as of a Settlement Date, then Agent shall, by no later than 12:00 p.m. on the Settlement Date, transfer in immediately available funds to a Deposit Account of such Australian Revolving Lender (as such Australian Revolving Lender may designate), an amount such that each such Australian Revolving Lender shall, upon receipt of such amount, have as of the Settlement Date, its Pro Rata Share of the Australian Revolving Loans (including Australian Swing Loans, and Australian Extraordinary Advances), and (z) if the amount of the Australian Revolving Loans (including Australian Swing Loans, and Australian Extraordinary Advances) made by an Australian Lender is less than such Lender's Pro Rata Share of the Australian Revolving Loans (including Australian Swing Loans, and Australian Extraordinary Advances) as of a Settlement Date, such Australian Revolving Lender shall no later than 12:00 p.m. on the Settlement Date transfer in immediately available funds to Agent's Account, an amount such that each such Australian Revolving Lender shall, upon transfer of such amount, have as of the Settlement Date, its Pro Rata Share of the Australian Revolving Loans (including Australian Swing Loans and Australian Extraordinary Advances). Such amounts made available to Agent under clause (z) of the immediately preceding sentence shall be applied against the amounts of the applicable Australian Swing Loans or Australian Extraordinary Advances and, together with the portion of such Australian Swing Loans or Australian Extraordinary Advances representing Australian Swing Lender's Pro Rata Share thereof, shall constitute Australian Revolving Loans of such Lenders. If any such amount is not made available to Agent by any Australian Revolving Lender on the Settlement Date applicable thereto to the extent required by the terms hereof, Agent shall be entitled to recover for its account such amount on demand from such Australian Revolving Lender together with interest thereon at the Defaulting Lender Rate.

- (B) In determining whether an Australian Revolving Lender's balance of the Australian Revolving Loans, Australian Swing Loans, and Australian Extraordinary Advances is less than, equal to, or greater than such Australian Revolving Lender's Pro Rata Share of the Australian Revolving Loans, Australian Swing Loans, and Australian Extraordinary Advances as of a Settlement Date, Agent shall, as part of the relevant Settlement, apply to such balance the portion of payments actually received in good funds by Agent with respect to principal, interest, fees payable by Australian Borrowers and allocable to the Australian Revolving Lenders hereunder, and proceeds of Collateral securing the Australian Obligations.
- (C) Between Settlement Dates, Agent, to the extent Australian Extraordinary Advances or Australian Swing Loans are outstanding, may pay over to Agent or Australian Swing Lender, as applicable, any payments or other amounts received by Agent, that in accordance with the terms of this Agreement would be applied to the reduction of the Australian Revolving Loans, for application to the Australian Extraordinary Advances or Australian Swing Loans. Between Settlement Dates, Agent, to the extent no Australian Extraordinary Advances or Australian Swing Loans are outstanding, may pay over to Australian Swing Lender any payments or other amounts received by Agent, that in accordance with the terms of this Agreement would be applied to the reduction of the Australian Revolving Loans, for application to Australian Swing Lender's Pro Rata Share of the Australian Revolving Loans. If, as of any Settlement Date, payments or other amounts of Parent and its Subsidiaries received since the then immediately preceding Settlement Date have been applied to Australian Swing Lender's Pro Rata Share of the Australian Revolving Loans other than to Australian Swing Loans, as provided for in the previous sentence, Australian Swing Lender shall pay to Agent for the accounts of the Australian Revolving Lenders, and Agent shall pay to the Australian Revolving Lenders (other than a Defaulting Lender if Agent has implemented the provisions of Section 2.3(i)), to be applied to the outstanding Australian Revolving Loans of such Australian Revolving Lenders, an amount such that each such Australian Revolving Lender shall, upon receipt of such amount, have, as of such Settlement Date, its Pro Rata Share of the Australian Revolving Loans. During the period between Settlement Dates, Australian Swing Lender with respect to Australian Revolving Loans, Agent with respect to Australian Revolving Loans other than Australian Swing Loans and Australian Extraordinary Advances, shall be entitled to interest at the applicable rate or rates payable under this A
- (iii) Anything in this <u>Section 2.3(g)</u> to the contrary notwithstanding, in the event that a Lender is a Defaulting Lender, Agent shall be entitled to refrain from remitting settlement amounts to the Defaulting Lender and, instead, shall be entitled to elect to implement the provisions set forth in <u>Section 2.3(i)</u>.
- (h) **Notation.** Agent, as a non-fiduciary agent for Borrowers, shall maintain a register (including the Register required pursuant to <u>Section 13.1(h)</u>) showing the principal amount of the Revolving Loans owing to each Lender, including the Swing Loans owing to each Swing Lender, and Extraordinary Advances owing to Agent, and the

interests therein of each Lender, from time to time and such register shall, absent manifest error, conclusively be presumed to be correct and accurate.

(i) Defaulting Lenders.

(i) Generally. Notwithstanding the provisions of Section 2.4(b)(iv) and Section 2.4(b)(v), Agent shall not be obligated to transfer to a Defaulting Lender any payments made by Borrowers (or any of them) to Agent for the Defaulting Lender's benefit or any proceeds of Collateral that would otherwise be remitted hereunder to the Defaulting Lender, and, in the absence of such transfer to the Defaulting Lender, Agent shall transfer any such payments (A) first, to Swing Lenders to the extent of any Swing Loans that were made by any Swing Lender and that were required to be, but were not, paid by the Defaulting Lender, (B) second, to Issuing Banks, to the extent of the portion of a Letter of Credit Disbursement that was required to be, but was not, paid by the Defaulting Lender, (C) third, to each Non-Defaulting Lender ratably in accordance with their applicable Revolver Commitments (but, in each case, only to the extent that such Defaulting Lender's portion of a Revolving Loan (or other funding obligation) was funded by such other Non-Defaulting Lender), (D) to a suspense account maintained by Agent, the proceeds of which shall be retained by Agent and may be made available to be re-advanced to or for the benefit of the applicable Borrowers (upon the request of Borrowers and subject to the conditions set forth in Section 3.3) as if such Defaulting Lender had made its portion of applicable Revolving Loans (or other funding obligations) hereunder, and (E) from and after the date on which all other Obligations have been paid in full, to such Defaulting Lender in accordance with tier (Y) of Section 2.4(b)(iv) or tier (M) of Section 2.4(b)(v) (as applicable). Subject to the foregoing, Agent may hold and, in its discretion, re-lend to the applicable Borrowers for the account of such Defaulting Lender the amount of all such payments received and retained by Agent for the account of such Defaulting Lender. Solely for the purposes of voting or consenting to matters with respect to the Loan Documents (including the calculation of Pro Rata Share in connection therewith) and for the purpose of calculating the fee payable under Section 2.10(b), such Defaulting Lender shall be deemed not to be a "Lender" and such Lender's Revolver Commitments shall be deemed to be zero; provided, that the foregoing shall not apply to any of the matters governed by Section 14.1(a)(i) through (iv). The provisions of this Section 2.3(i) shall remain effective with respect to such Defaulting Lender until the earlier of (y) the date on which all of the Non-Defaulting Lenders, Agent, Issuing Banks, and Borrowers shall have waived, in writing, the application of this Section 2.3(i) to such Defaulting Lender, or (z) the date on which such Defaulting Lender makes payment of all amounts that it was obligated to fund hereunder, pays to Agent all amounts owing by Defaulting Lender in respect of the amounts that it was obligated to fund hereunder, and, if requested by Agent, provides adequate assurance of its ability to perform its future obligations hereunder (on which earlier date, so long as no Event of Default has occurred and is continuing, any remaining cash collateral held by Agent pursuant to Section 2.3(i)(iii) or 2.3(i)(iiii) shall be released to the applicable Borrowers). The operation of this Section 2.3(i) shall not be construed to increase or otherwise affect the Revolver Commitment of any Lender, to relieve or excuse the performance by such Defaulting Lender or any other Lender of its duties and obligations hereunder, or to relieve or excuse the performance by any Borrower of its duties and obligations hereunder to Agent, Issuing Banks, or to the Lenders other than such Defaulting Lender. Any failure by a Defaulting Lender to fund amounts that it was obligated to fund hereunder shall constitute a material breach by such Defaulting Lender of this Agreement and shall entitle Borrowers, at their option, upon written notice to Agent, to arrange for a substitute Lender to assume the Revolver Commitment of such Defaulting Lender, such substitute Lender to be reasonably acceptable to Agent. In connection with the arrangement of such a substitute Lender, the Defaulting Lender shall have no right to refuse to be replaced hereunder, and agrees to execute and deliver a completed form of Assignment and Acceptance in favor of the substitute Lender (and agrees that it shall be deemed to have executed and delivered such document if it fails to do so) subject only to being paid its share of the outstanding Obligations (other than Bank Product Obligations, but including (1) all interest, fees, and other amounts that may be due and payable in respect thereof, and (2) an assumption of its Pro Rata Share of its participation in the applicable Letters of Credit); provided, that, subject to Section 18.15, any such assumption of the Revolver Commitment of such Defaulting Lender shall not be deemed to constitute a waiver of any of the Lender Group's or Borrowers' rights or remedies against any such Defaulting Lender arising out of or in relation to such failure to fund. In the event of a direct conflict between the priority provisions of this Section 2.3(i) and any other provision contained in this Agreement or any other Loan Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, the terms and provisions of this Section 2.3(i) shall control and govern.

(ii) <u>U.S. Revolving Lenders as Defaulting Lenders</u>. If any U.S. Swing Loan or U.S. Letter of Credit is outstanding at the time that a U.S. Revolving Lender becomes a Defaulting Lender then:

(A) such Defaulting Lender's U.S. Swing Loan Exposure and U.S. Letter of Credit Exposure shall be reallocated among the Non-Defaulting Lenders in accordance with their respective Pro Rata Shares

but only to the extent (x) the sum of all Non-Defaulting Lenders' U.S. Revolving Loan Exposures *plus* such Defaulting Lender's U.S. Swing Loan Exposure and U.S. Letter of Credit Exposure does not exceed the total of all Non-Defaulting Lenders' U.S. Revolver Commitments, (y) no Non-Defaulting Lenders' Pro Rata Share of U.S. Revolver Usage exceeds its U.S. Revolver Commitment, and (z) the conditions set forth in <u>Section 3.3</u> are satisfied at such time;

- (B) if the reallocation described in clause (A) above cannot, or can only partially, be effected, U.S. Borrowers shall within one Business Day following notice by the Agent (x) first, prepay such Defaulting Lender's U.S. Swing Loan Exposure (after giving effect to any partial reallocation pursuant to clause (A) above) and (y) second, cash collateralize such Defaulting Lender's U.S. Letter of Credit Exposure (after giving effect to any partial reallocation pursuant to clause (A) above), pursuant to a cash collateral agreement to be entered into in form and substance reasonably satisfactory to the Agent, for so long as such U.S. Letter of Credit Exposure is outstanding; provided, that U.S. Borrowers shall not be obligated to cash collateralize any Defaulting Lender's U.S. Letter of Credit Exposure if such Defaulting Lender is also a U.S. Issuing Bank;
- (C) if U.S. Borrowers cash collateralize any portion of such Defaulting Lender's U.S. Letter of Credit Exposure pursuant to this <u>Section 2.3(i)(ii)</u>, U.S. Borrowers shall not be required to pay any U.S. Letter of Credit Fees to Agent for the account of such Defaulting Lender pursuant to <u>Section 2.6(b)</u> with respect to such cash collateralized portion of such Defaulting Lender's U.S. Letter of Credit Exposure during the period such U.S. Letter of Credit Exposure is cash collateralized;
- (D) to the extent the U.S. Letter of Credit Exposure of the Non-Defaulting Lenders is reallocated pursuant to this Section 2.3(i)(ii), then the U.S. Letter of Credit Fees payable to the Non-Defaulting Lenders pursuant to Section 2.6(b) shall be adjusted in accordance with such Non-Defaulting Lenders' U.S. Letter of Credit Exposure;
- (E) to the extent any Defaulting Lender's U.S. Letter of Credit Exposure is neither cash collateralized nor reallocated pursuant to this <u>Section 2.3(i)(ii)</u>, then, without prejudice to any rights or remedies of any U.S. Issuing Bank or any U.S. Revolving Lender hereunder, all U.S. Letter of Credit Fees that would have otherwise been payable to such Defaulting Lender under <u>Section 2.6(b)</u> with respect to such portion of such U.S. Letter of Credit Exposure shall instead be payable to the applicable U.S. Issuing Bank until such portion of such Defaulting Lender's U.S. Letter of Credit Exposure is cash collateralized or reallocated:
- (F) so long as any U.S. Revolving Lender is a Defaulting Lender, the U.S. Swing Lender shall not be required to make any U.S. Swing Loan and no U.S. Issuing Bank shall be required to issue, amend, or increase any U.S. Letter of Credit, in each case, to the extent (x) the Defaulting Lender's Pro Rata Share of such U.S. Swing Loans or U.S. Letter of Credit cannot be reallocated pursuant to this Section 2.3(i)(ii) or (y) the U.S. Swing Lender or U.S. Issuing Banks, as applicable, have not otherwise entered into arrangements reasonably satisfactory to the U.S. Swing Lender or U.S. Issuing Banks, as applicable, and Borrowers to eliminate the U.S. Swing Lender's or U.S. Issuing Banks' risk with respect to the Defaulting Lender's participation in U.S. Swing Loans or U.S. Letters of Credit; and
- (G) Agent may release any cash collateral provided by U.S. Borrowers pursuant to this <u>Section 2.3(i)(ii)</u> to any U.S. Issuing Bank and such U.S. Issuing Bank may apply any such cash collateral to the payment of such Defaulting Lender's Pro Rata Share of any Letter of Credit Disbursement that is not reimbursed by U.S. Borrowers pursuant to Section 2.11(d).
- (iii) <u>Australian Revolving Lenders as Defaulting Lenders</u>. If any Australian Swing Loan or Australian Letter of Credit is outstanding at the time that an Australian Revolving Lender becomes a Defaulting Lender then:
- (A) such Defaulting Lender's Australian Swing Loan Exposure and Australian Letter of Credit Exposure shall be reallocated among the Non-Defaulting Lenders in accordance with their respective Pro Rata Shares but only to the extent (x) the sum of all Non-Defaulting Lenders' Australian Revolving Loan Exposures plus such Defaulting Lender's Australian Swing Loan Exposure and Australian Letter of Credit Exposure does not exceed the total of all Non-Defaulting Lenders' Australian Revolver Commitments, (y) no Non-Defaulting Lenders' Pro Rata Share of Australian Revolver Usage exceeds its Australian Revolver Commitment, and (z) the conditions set forth in Section 3.3 are satisfied at such time;

(B) if the reallocation described in clause (A) above cannot, or can only partially, be effected, Australian Borrowers sha
within one Business Day following notice by the Agent (x) first, prepay such Defaulting Lender's Australian Swing Loan Exposure (after giving effect to ar
partial reallocation pursuant to clause (A) above) and (y) second, cash collateralize such Defaulting Lender's Australian Letter of Credit Exposure (after givin
effect to any partial reallocation pursuant to clause (A) above), pursuant to a cash collateral agreement to be entered into in form and substance reasonab
satisfactory to the Agent, for so long as such Australian Letter of Credit Exposure is outstanding; provided, that Australian Borrowers shall not be obligated to
cash collateralize any Defaulting Lender's Australian Letter of Credit Exposure if such Defaulting Lender is also an Australian Issuing Bank;

- (C) if Australian Borrowers cash collateralize any portion of such Defaulting Lender's Australian Letter of Credit Exposure pursuant to this <u>Section 2.3(i)(iii)</u>, Australian Borrowers shall not be required to pay any Australian Letter of Credit Fees to Agent for the account of such Defaulting Lender pursuant to <u>Section 2.6(b)</u> with respect to such cash collateralized portion of such Defaulting Lender's Australian Letter of Credit Exposure during the period such Australian Letter of Credit Exposure is cash collateralized;
- (D) to the extent the Australian Letter of Credit Exposure of the Non-Defaulting Lenders is reallocated pursuant to this Section 2.3(i)(iii), then the Australian Letter of Credit Fees payable to the Non-Defaulting Lenders pursuant to Section 2.6(b) shall be adjusted in accordance with such Non-Defaulting Lenders' U.S. Letter of Credit Exposure;
- (E) to the extent any Defaulting Lender's Australian Letter of Credit Exposure is neither cash collateralized nor reallocated pursuant to this <u>Section 2.3(i)(iiii)</u>, then, without prejudice to any rights or remedies of any Australian Issuing Bank or any Australian Revolving Lender hereunder, all Australian Letter of Credit Fees that would have otherwise been payable to such Defaulting Lender under <u>Section 2.6(b)</u> with respect to such portion of such Australian Letter of Credit Exposure shall instead be payable to the applicable Australian Issuing Bank until such portion of such Defaulting Lender's Australian Letter of Credit Exposure is cash collateralized or reallocated;
- (F) so long as any Australian Revolving Lender is a Defaulting Lender, the Australian Swing Lender shall not be required to make any Australian Swing Loan and no Australian Issuing Bank shall be required to issue, amend, or increase any Australian Letter of Credit, in each case, to the extent (x) the Defaulting Lender's Pro Rata Share of such Australian Swing Loans or Australian Letter of Credit cannot be reallocated pursuant to this Section 2.3(i)(iii) or (y) the Australian Swing Lender or Australian Issuing Banks, as applicable, have not otherwise entered into arrangements reasonably satisfactory to the Australian Swing Lender or Australian Issuing Banks, as applicable, and Australian Borrowers to eliminate the Australian Swing Lender's or Australian Issuing Banks' risk with respect to the Defaulting Lender's participation in Australian Swing Loans or Australian Letters of Credit; and
- (G) Agent may release any cash collateral provided by Australian Borrowers pursuant to this Section 2.3(i)(iii) to any Australian Issuing Bank and such Australian Issuing Bank may apply any such cash collateral to the payment of such Defaulting Lender's Pro Rata Share of any Letter of Credit Disbursement that is not reimbursed by Australian Borrowers pursuant to Section 2.12(d).
- (j) Independent Obligations. All Revolving Loans (other than Swing Loans and Extraordinary Advances) shall be made by the Lenders contemporaneously and in accordance with their Pro Rata Shares. The obligations of the Lenders under this Agreement to make Loans and to fund participations in Letters of Credit, Swing Loans and Extraordinary Advances are several (and not joint and several). It is understood that (i) no Lender shall be responsible for any failure by any other Lender to perform its obligation to make any Revolving Loan (or other extension of credit) hereunder, nor shall any Revolver Commitment of any Lender be increased or decreased as a result of any failure by any other Lender to perform its obligations hereunder, and (ii) no failure by any Lender to perform its obligations hereunder shall excuse any other Lender from its obligations hereunder.

2.4 Payments; Reductions of Commitments; Prepayments.

(a) Payments by Borrowers.

(i) Except as otherwise expressly provided herein, all payments by Borrowers (or any of them) shall be made to the Agent's Account designated for the currency of the applicable payment and shall be made in immediately available funds, no later than 1:30 p.m. on the date specified herein. Any payment received by

Agent later than 1:30 p.m. shall be deemed to have been received (unless Agent, in its sole discretion, elects to credit it on the date received) on the following Business Day and any applicable interest or fee shall continue to accrue until such following Business Day.

(ii) Unless Agent receives notice from Borrowers (or any of them) prior to the date on which any payment is due to the Lenders (or any of them) that such Borrowers will not make such payment in full as and when required, Agent may assume that such Borrowers have made (or will make) such payment in full to Agent on such date in immediately available funds and Agent may (but shall not be so required), in reliance upon such assumption, distribute to each Lender on such due date an amount equal to the amount then due such Lender. If and to the extent Borrowers (or any of them) do not make such payment in full to Agent on the date when due, each Lender severally shall repay to Agent on demand such amount distributed to such Lender, together with interest thereon at the Defaulting Lender Rate for each day from the date such amount is distributed to such Lender until the date repaid.

(b) Apportionment and Application.

- (i) So long as no Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, all principal and interest payments received by Agent shall be apportioned ratably among the Lenders (according to the unpaid principal balance of the portion of the Obligations to which such payments relate held by each Lender) and all payments of fees and expenses received by Agent (other than fees or expenses that are for Agent's separate account or for the separate account of Issuing Bank) shall be apportioned ratably among the Lenders having a Pro Rata Share of the type of Revolver Commitment or portion of the Obligation to which a particular fee or expense relates.
- (ii) Subject to Section 2.4(b)(vii) and Section 2.4(e), all payments to be made hereunder by U.S. Borrowers shall be remitted to Agent and all such payments, and all proceeds of Collateral securing the U.S. Obligations received by Agent, shall be applied, so long as no Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, subject to the Intercreditor Agreement, to reduce the balance of the U.S. Revolving Loans outstanding and, thereafter, to U.S. Borrowers (to be wired to the U.S. Designated Account) or such other Person entitled thereto under applicable law.
- (iii) Subject to Section 2.4(b)(vii) and Section 2.4(e), all payments to be made hereunder by Australian Borrowers shall be remitted to Agent and all such payments, and all proceeds of Collateral securing the Australian Obligations received by Agent, shall be applied, so long as no Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, subject to the Intercreditor Agreement, to reduce the balance of the Australian Revolving Loans outstanding and, thereafter, to Australian Borrowers (to be wired to the Australian Designated Account) or such other Person entitled thereto under applicable law.
- (iv) At any time that an Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, subject to the Intercreditor Agreement, all payments remitted to Agent by U.S. Borrowers and all proceeds of Collateral securing the U.S. Obligations received by Agent shall be applied as follows (it being understood that for purposes of this clause (iv) "U.S. Obligations" excludes the U.S. Loan Parties' obligations under their guarantee of the Australian Obligations):
- (A) first, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to Agent under the Loan Documents in respect of the U.S. Obligations, until paid in full,
 - (B) second, to pay any fees then due to Agent under the Loan Documents in respect of the U.S. Obligations until paid in full,
 - (C) third, ratably to pay interest due in respect of all U.S. Protective Advances until paid in full,
 - (D) fourth, ratably to pay the principal of all U.S. Protective Advances until paid in full,

- (E) <u>fifth</u>, ratably to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to any of the U.S. Revolving Lenders or U.S. Issuing Banks under the Loan Documents, until paid in full,
- (F) sixth, ratably to pay any fees then due to any of the U.S. Revolving Lenders or U.S. Issuing Banks under the Loan Documents until paid in full,
 - (G) seventh, ratably to pay interest accrued in respect of the U.S. Swing Loans until paid in full,
 - (H) eighth, ratably to pay the principal of all U.S. Swing Loans until paid in full,
- (I) <u>ninth</u>, to Agent, to be held by Agent, for the benefit of U.S. Issuing Banks (and for the ratable benefit of each of the U.S. Revolving Lenders that have an obligation to pay to Agent, for the account of U.S. Issuing Banks, a share of each Letter of Credit Disbursement in connection with each U.S. Letter of Credit), as cash collateral in an amount up to the sum of 103% of the U.S. Letter of Credit Usage (to the extent permitted by applicable law, such cash collateral shall be applied to the reimbursement of any Letter of Credit Disbursement as and when such disbursement occurs and, if a U.S. Letter of Credit expires undrawn, the cash collateral held by Agent in respect of such U.S. Letter of Credit shall, to the extent permitted by applicable law, be reapplied pursuant to this <u>Section 2.4(b)(iv)</u>, beginning with tier (A) hereof),
 - (J) <u>tenth</u>, ratably to pay interest accrued in respect of the U.S. Revolving Loans (other than U.S. Protective Advances) until

(K) eleventh,

paid in full,

- i. ratably to pay the principal of all U.S. Revolving Loans until paid in full,
- ii. ratably up to the amount (after taking into account any amounts previously paid pursuant to this clause (ii) during the continuation of the applicable Application Event) of the most recently established U.S. Hedge Reserves (after taking into account any amounts previously paid pursuant to this clause (ii) during the continuation of the applicable Application Event), to (I) the U.S. Hedge Providers based upon amounts then certified by the applicable U.S. Hedge Provider to Agent (in form and substance satisfactory to Agent) to be due and payable to such U.S. Hedge Providers on account of U.S. Hedge Obligations, and (II) with any balance to be paid to Agent, to be held by Agent, for the ratable benefit of the U.S. Hedge Providers, as cash collateral (which cash collateral may be released by Agent to the applicable U.S. Hedge Provider and applied by such U.S. Hedge Provider to the payment or reimbursement of any amounts due and payable with respect to U.S. Hedge Obligations owed to the applicable U.S. Hedge Provider as and when such amounts first become due and payable and, if and at such time as all such U.S. Hedge Obligations are paid or otherwise satisfied in full, the cash collateral held by Agent in respect of such U.S. Hedge Obligations shall be reapplied pursuant to this Section 2.4(b)(iv), beginning with tier (A) hereof),

THEN, unless otherwise agreed by the Supermajority Lenders, only after the earlier of (i) payment in full in cash of all Australian Obligations contemplated by clauses (A) through (K) of <u>Section 2.4(b)(v)</u> and (ii) the exercise of remedies against all Collateral securing the Australian Obligations and the application of the proceeds thereof in accordance with <u>Section 2.4(b)(v)</u>:

- (L) <u>twelfth</u>, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to Agent or Australian Security Trustee under the Loan Documents in respect of the Australian Obligations, until paid in full,
- (M) thirteenth, to pay any fees then due to Agent or Australian Security Trustee under the Loan Documents in respect of the Australian Obligations until paid in full,
 - (N) fourteenth, ratably to pay interest due in respect of all Australian Protective Advances until paid in full,

- (O) fifteenth, ratably to pay the principal of all Australian Protective Advances until paid in full,
- (P) sixteenth, ratably to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to any of the Australian Revolving Lenders or Australian Issuing Banks under the Loan Documents, until paid in full,
- (Q) seventeenth, ratably to pay any fees then due to any of the Australian Revolving Lenders or Australian Issuing Banks under the Loan Documents until paid in full,
 - (R) eighteenth, ratably to pay interest accrued in respect of the Australian Swing Loans until paid in full,
 - (S) <u>nineteenth</u>, ratably to pay the principal of all Australian Swing Loans until paid in full,
- (T) twentieth, to Agent, to be held by Agent, for the benefit of Australian Issuing Banks (and for the ratable benefit of each of the Australian Revolving Lenders that have an obligation to pay to Agent, for the account of Australian Issuing Banks, a share of each Letter of Credit Disbursement in connection with each Australian Letter of Credit), as cash collateral in an amount up to the sum of 103% of the Australian Letter of Credit Usage (to the extent permitted by applicable law, such cash collateral shall be applied to the reimbursement of any Letter of Credit Disbursement as and when such disbursement occurs and, if a Australian Letter of Credit expires undrawn, the cash collateral held by Agent in respect of such Australian Letter of Credit shall, to the extent permitted by applicable law, be reapplied pursuant to this Section 2.4(b)(iv), beginning with tier (A) hereof),
- (U) <u>twenty-first</u>, ratably to pay interest accrued in respect of the Australian Revolving Loans (other than Australian Protective Advances) until paid in full,
 - (V) twenty-second,
 - i. ratably to pay the principal of all Australian Revolving Loans until paid in full,
- ii. ratably up to the amount (after taking into account any amounts previously paid pursuant to this clause (ii) during the continuation of the applicable Application Event) of the most recently established Australian Hedge Reserves (after taking into account any amounts previously paid pursuant to this clause (ii) during the continuation of the applicable Application Event), to (I) the Australian Hedge Providers based upon amounts then certified by the applicable Australian Hedge Provider to Agent (in form and substance satisfactory to Agent) to be due and payable to such Australian Hedge Providers on account of Australian Hedge Obligations, and (II) with any balance to be paid to Agent, to be held by Agent, for the ratable benefit of the Australian Hedge Providers, as cash collateral (which cash collateral may be released by Agent to the applicable Australian Hedge Provider and applied by such Australian Hedge Provider to the payment or reimbursement of any amounts due and payable with respect to Australian Hedge Obligations owed to the applicable Australian Hedge Provider as and when such amounts first become due and payable and, if and at such time as all such Australian Hedge Obligations are paid or otherwise satisfied in full, the cash collateral held by Agent in respect of such Australian Hedge Obligations shall be reapplied pursuant to this Section 2.4(b)(iv), beginning with tier (A) hereof),
- (W) twenty-third, to pay any other U.S. Obligations other than U.S. Obligations owed to Defaulting Lenders (including being paid, ratably to the U.S. Bank Product Providers on account of all amounts then due and payable in respect of U.S. Bank Product Obligations, with any balance to be paid to Agent, to be held by Agent, for the ratable benefit of the U.S. Bank Product Providers, as cash collateral (which cash collateral may be released by Agent to the applicable U.S. Bank Product Provider and applied by such U.S. Bank Product Provider to the payment or reimbursement of any amounts due and payable with respect to U.S. Bank Product Obligations owed to the applicable U.S. Bank Product Provider as and when such amounts first become due and payable and, if and at such time as all such U.S. Bank Product Obligations are paid or otherwise satisfied in full, the cash collateral held by Agent in respect of such U.S. Bank Product Obligations shall be reapplied pursuant to this Section 2.4(b)(iv), beginning with tier (A) hereof),

- (X) twenty-fourth, to pay any other Australian Obligations other than Australian Obligations owed to Defaulting Lenders (including being paid, ratably to the Australian Bank Product Providers on account of all amounts then due and payable in respect of Australian Bank Product Obligations, with any balance to be paid to Agent, to be held by Agent, for the ratable benefit of the Australian Bank Product Providers, as cash collateral (which cash collateral may be released by Agent to the applicable Australian Bank Product Provider and applied by such Australian Bank Product Provider to the payment or reimbursement of any amounts due and payable with respect to Australian Bank Product Obligations owed to the applicable Australian Bank Product Provider as and when such amounts first become due and payable and, if and at such time as all such Australian Bank Product Obligations are paid or otherwise satisfied in full, the cash collateral held by Agent in respect of such Australian Bank Product Obligations shall be reapplied pursuant to this Section 2.4(b)(iv), beginning with tier (A) hereof),
 - (Y) twenty-fifth, ratably to pay any Obligations owed to Defaulting Lenders; and
- (Z) <u>twenty-sixth</u>, to U.S. Borrowers (to be wired to the U.S. Designated Account) or such other Person entitled thereto under applicable law.
- (v) At any time that an Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, subject to the Intercreditor Agreement, all payments remitted to Agent by Australian Borrowers and all proceeds of Collateral securing the Australian Obligations received by Agent or the Australian Security Trustee shall be applied as follows to the extent not applied or eligible to be applied pursuant to clause (iv) above:
- (A) <u>first</u>, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to Agent or the Australian Security Trustee under the Loan Documents in respect of the Australian Obligations, until paid in full,
- (B) <u>second</u>, to pay any fees then due to Agent (to the extent related to the Australian Obligations) or the Australian Security Trustee under the Loan Documents in respect of the Australian Obligations until paid in full,
 - (C) third, ratably to pay interest due in respect of all Australian Protective Advances until paid in full,
 - (D) fourth, ratably to pay the principal of all Australian Protective Advances until paid in full,
- (E) <u>fifth</u>, ratably to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to any of the Australian Revolving Lenders or Australian Issuing Banks under the Loan Documents, until paid in full,
- (F) sixth, ratably to pay any fees then due to any of the Australian Revolving Lenders or Australian Issuing Banks under the Loan Documents until paid in full,
 - (G) seventh, to pay interest accrued in respect of the Australian Swing Loans until paid in full,
 - (H) eighth, to pay the principal of all Australian Swing Loans until paid in full,
- (I) <u>ninth</u>, to Agent, to be held by Agent, for the benefit of Australian Issuing Banks (and for the ratable benefit of each of the Australian Revolving Lenders that have an obligation to pay to Agent, for the account of Australian Issuing Banks, a share of each Australian Letter of Credit Disbursement), as cash collateral in an amount up to 103% of the Australian Letter of Credit Usage (to the extent permitted by applicable law, such cash collateral shall be applied to the reimbursement of any Australian Letter of Credit Disbursement as and when such disbursement occurs and, if an Australian Letter of Credit expires undrawn, the cash collateral held by Agent in respect of such Australian Letter of Credit shall, to the extent permitted by applicable law, be reapplied pursuant to this <u>Section 2.4(b)(v)</u>, beginning with tier (A) hereof),
- (J) tenth, ratably, to pay interest accrued in respect of the Australian Revolving Loans (other than Australian Protective Advances) until paid in full,

(K) <u>eleventh</u>,

- i. ratably to pay the principal of all Australian Revolving Loans and until paid in full,
- ii. ratably up to the amount (after taking into account any amounts previously paid pursuant to this clause (ii) during the continuation of the applicable Application Event) of the most recently established Australian Hedge Reserves (after taking into account any amounts previously paid pursuant to this clause (ii) during the continuation of the applicable Application Event), to (I) the Australian Hedge Providers based upon amounts then certified by the applicable Australian Hedge Provider to Agent (in form and substance satisfactory to Agent) to be due and payable to such Australian Hedge Providers on account of Australian Hedge Obligations, and (II) with any balance to be paid to Agent, to be held by Agent, for the ratable benefit of the Australian Hedge Providers, as cash collateral (which cash collateral may be released by Agent to the applicable Australian Hedge Provider and applied by such Australian Hedge Provider to the payment or reimbursement of any amounts due and payable with respect to Australian Hedge Obligations owed to the applicable Australian Hedge Provider as and when such amounts first become due and payable and, if and at such time as all such Australian Hedge Obligations are paid or otherwise satisfied in full, the cash collateral held by Agent in respect of such Australian Hedge Obligations shall be reapplied pursuant to this Section 2.4(b)(v), beginning with tier (A) hereof),
- (L) twelfth, to pay any other Australian Obligations other than Australian Obligations owed to Defaulting Lenders (including being paid, ratably, to the Australian Bank Product Providers on account of all amounts then due and payable in respect of Australian Bank Product Obligations, with any balance to be paid to Agent, to be held by Agent, for the ratable benefit of the Australian Bank Product Providers, as cash collateral (which cash collateral may be released by Agent to the applicable Australian Bank Product Provider and applied by such Australian Bank Product Provider to the payment or reimbursement of any amounts due and payable with respect to Australian Bank Product Obligations owed to the applicable Australian Bank Product Provider as and when such amounts first become due and payable and, if and at such time as all such Australian Bank Product Obligations are paid or otherwise satisfied in full, the cash collateral held by Agent in respect of such Australian Bank Product Obligations shall be reapplied pursuant to this Section 2.4(b)(v), beginning with tier (A) hereof),
 - (M) thirteenth, ratably to pay any Australian Obligations owed to Defaulting Lenders; and
- (N) <u>fourteenth</u>, to Australian Borrowers (to be wired to the Australian Designated Account) or such other Person entitled thereto under applicable law.
- (vi) Agent promptly shall distribute to each Lender, pursuant to the applicable wire instructions received from each Lender in writing, such funds as it may be entitled to receive, subject to a Settlement delay as provided in <u>Section 2.3(g)</u>.
- (vii) In each instance, so long as no Application Event has occurred and is continuing, <u>Sections 2.4(b)(iv)</u> and <u>2.4(b)(v)</u> shall not apply to any payment made by Borrowers (or any of them) to Agent and specified by such Borrowers to be for the payment of specific Obligations then due and payable (or prepayable) under any provision of this Agreement or any other Loan Document.
- (viii) For purposes of Sections 2.4(b)(iv) and 2.4(b)(v), "paid in full" of a type of Obligation means payment in cash or immediately available funds of all amounts owing on account of such type of Obligation, including interest accrued after the commencement of any Insolvency Proceeding, default interest, interest on interest, and expense reimbursements, irrespective of whether any of the foregoing would be or is allowed or disallowed in whole or in part in any Insolvency Proceeding.
- (ix) In the event of a direct conflict between the priority provisions of this <u>Section 2.4</u> and any other provision contained in this Agreement or any other Loan Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, if the conflict relates to the provisions of <u>Section 2.3(i)</u> and this <u>Section 2.4</u>, then the provisions of <u>Section 2.3(i)</u> shall control and govern, and if otherwise, then the terms and provisions of this <u>Section 2.4</u> shall control and govern.
 - (c) Reduction of Commitments.

- (i) **U.S. Revolver Commitments.** The U.S. Revolver Commitments shall terminate on the Maturity Date. Borrowers may reduce the U.S. Revolver Commitments, without premium or penalty, to an amount not less than the sum of (A) the U.S. Revolver Usage as of such date, plus (B) the principal amount of all U.S. Revolving Loans not yet made as to which a request has been given by Borrowers under Section 2.3(a), plus (C) the amount of all U.S. Letters of Credit not yet issued as to which a request has been given by Borrowers pursuant to Section 2.11(a): provided that if after any such reduction of the U.S. Revolver Commitments, the Australian Revolver Commitments will exceed the U.S. Revolver Commitments, such reduction shall be accompanied by a reduction of the Australian Revolver Commitments pursuant to Section 2.4(c)(ii) in the amount of such excess. Each such reduction shall be in an amount which is not less than \$5,000,000 (unless the U.S. Revolver Commitments are being reduced to zero and the amount of the U.S. Revolver Commitments in effect immediately prior to such reduction are less than \$5,000,000), shall be made by providing not less than 5 Business Days prior written notice to Agent, and shall be irrevocable; provided that such notice of termination may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of one or more securities offerings or other transactions, in which case such notice may be revoked by Borrowers (by notice to Agent from Parent on or prior to the specified effective date) if such condition is not satisfied. Once reduced, the U.S. Revolver Commitments may not be increased. Each such reduction of the U.S. Revolver Commitments shall reduce the U.S. Revolver Commitments of each U.S. Revolver Commitments proportionately in accordance with its ratable share thereof.
- (ii) Australian Revolver Commitments. The Australian Revolver Commitments shall terminate on the Maturity Date. Borrowers may reduce the Australian Revolver Commitments, without premium or penalty, to an amount (which may be zero) not less than the sum of (A) the Australian Revolver Usage as of such date, plus (B) the principal amount of all Australian Revolving Loans not yet made as to which a request has been given by Borrowers under Section 2.3(a), plus (C) the amount of all Australian Letters of Credit not yet issued as to which a request has been given by Borrowers pursuant to Section 2.12(a). Each such reduction shall be in an amount which is not less than \$5,000,000 (unless the Australian Revolver Commitments are being reduced to zero and the amount of the Australian Revolver Commitments in effect immediately prior to such reduction are less than \$5,000,000), shall be made by providing not less than 5 Business Days prior written notice to Agent, and shall be irrevocable; provided that such notice of termination may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of one or more securities offerings or other transactions, in which case such notice may be revoked by Borrowers (by notice to Agent from Parent on or prior to the specified effective date) if such condition is not satisfied. Once reduced, the Australian Revolver Commitments may not be increased. Each such reduction of the Australian Revolver Commitments shall reduce the Australian Revolver Commitments of each Australian Revolving Lender proportionately in accordance with its ratable share thereof.

(d) Optional Prepayments.

- (i) U.S. Borrowers may prepay the principal of any U.S. Revolving Loan at any time in whole or in part, without premium or penalty, upon not less than (x) 3 Business Days prior written notice to Agent, in the case of LIBOR Rate U.S. Revolving Loan and (y) same day written notice to Agent, in the case of Base Rate U.S. Revolving Loans (each such notice to be irrevocable), provided that such notice may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of one or more securities offerings or other transactions, in which case such notice may be revoked by Borrowers (by notice to Agent from Parent on or prior to the specified effective date) if such condition is not satisfied; provided, further that no such notice shall be required during a Cash Dominion Trigger Period.
- (ii) Australian Borrowers may prepay the principal of any Australian Revolving Loan at any time in whole or in part, without premium or penalty.

(e) Mandatory Prepayments.

- (i) **U.S. Borrowing Base.** If, at any time and subject to <u>Section 1.7(d)</u>, (A) the U.S. Revolver Usage on such date exceeds (B) the U.S. Borrowing Base reflected in the Borrowing Base Certificate most recently delivered by Borrowers to Agent or the U.S. Maximum Revolver Amount, then U.S. Borrowers shall, within one Business Day, prepay the U.S. Obligations in accordance with <u>Section 2.4(f)(i)</u> in an aggregate amount equal to the amount of such excess
- (ii) Australian Borrowing Base. If, at any time and subject to Section 1.7(d), (A) the Australian Revolver Usage on such date exceeds (B) the Australian Borrowing Base reflected in the Borrowing Base Certificate most recently delivered by Borrowers to Agent or the Australian Maximum Revolver Amount, then Australian

Borrowers shall, within one Business Day, prepay the Australian Obligations in accordance with Section 2.4(f)(ii) in an aggregate amount equal to the amount of such excess.

(f) Application of Payments.

- (i) Each prepayment pursuant to Section 2.4(e)(i) shall, (A) so long as no Application Event shall have occurred and be continuing, be applied, first, to the outstanding principal amount of the U.S. Revolving Loans until paid in full, and second, to cash collateralize the U.S. Letters of Credit in an amount equal to the sum of (x) 103% of the U.S. Letter of Credit Usage that is denominated in Dollars, and (y) 103% of the U.S. Letter of Credit Usage that is denominated in an Agreed Currency other than Dollars, and (B) if an Application Event shall have occurred and be continuing, be applied in the manner set forth in Section 2.4(b)(iv).
- (ii) Each prepayment pursuant to <u>Section 2.4(e)(ii)</u> shall, (A) so long as no Application Event shall have occurred and be continuing, be applied, first, to the outstanding principal amount of the Australian Revolving Loans until paid in full, and second, to cash collateralize the Australian Letters of Credit in an amount equal to the sum of (x) 103% of the Australian Letter of Credit Usage that is denominated in Dollars, and (y) 103% of the Australian Letter of Credit Usage that is denominated in an Agreed Currency other than Dollars, and (B) if an Application Event shall have occurred and be continuing, be applied in the manner set forth in <u>Section 2.4(b)(v)</u>.
- (g) **Generally**. Any mandatory prepayments required under <u>Section 2.4(e)</u> shall not result in a permanent reduction of the Revolver Commitments. All prepayments made pursuant to <u>Section 2.4(d)</u> or <u>Section 2.4(e)</u> shall be accompanied by accrued and unpaid interest on the amount so prepaid.

2.5 Promise to Pay; Promissory Notes.

- (a) U.S. Borrowers agree to pay the Lender Group Expenses of Agent, the U.S. Issuing Banks, and the U.S. Revolving Lenders on the earlier of (i) the first day of the month following the date on which the applicable Lender Group Expenses were first incurred and invoiced or (ii) the date on which demand therefor is made by Agent (it being acknowledged and agreed that any charging of such costs, expenses or Lender Group Expenses to the applicable Loan Account pursuant to the provisions of Section 2.6(d) shall be deemed to constitute a demand for payment thereof for the purposes of this subclause (ii)). U.S. Borrowers promise to pay all of the U.S. Obligations (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses of Agent, the U.S. Issuing Banks, and the U.S. Revolving Lenders)) in full on the Maturity Date or, if earlier, on the date on which the U.S. Obligations (other than the U.S. Bank Product Obligations) become due and payable pursuant to the terms of this Agreement. U.S. Borrowers agree that their obligations contained in the first sentence of this Section 2.5(a) shall survive payment or satisfaction in full of all other U.S. Obligations.
- (b) Australian Borrowers agree to pay the Lender Group Expenses, in respect of the Australian Obligations, of Agent, the Australian Issuing Banks, and the Australian Revolving Lenders on the earlier of (i) the first day of the month following the date on which the applicable Lender Group Expenses were first incurred and invoiced or (ii) the date on which demand therefor is made by Agent (it being acknowledged and agreed that any charging of such costs, expenses or Lender Group Expenses to the applicable Loan Account pursuant to the provisions of Section 2.6(d) shall be deemed to constitute a demand for payment thereof for the purposes of this subclause (ii)). Australian Borrowers promise to pay all of the Australian Obligations (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses of Agent, the Australian Issuing Banks, and the Australian Revolving Lenders)) in full on the Maturity Date or, if earlier, on the date on which the Australian Obligations (other than the Australian Bank Product Obligations) become due and payable pursuant to the terms of this Agreement. Australian Borrowers agree that its obligations contained in the first sentence of this Section 2.5(b) shall survive payment or satisfaction in full of all other Australian Obligations.
- (c) Any Lender may request that any portion of its Revolver Commitments or the Revolving Loans made by it be evidenced by one or more promissory notes. In such event, the applicable Borrowers shall execute and deliver to such Lender or its registered assigns the requested promissory notes payable to such Lender in a form furnished by Agent and reasonably satisfactory to such Borrowers. Thereafter, the portion of the Revolver Commitments and Revolving Loans evidenced by such promissory notes and interest thereon shall at all times be represented by one or more promissory notes in such form payable to the payee named therein or its registered assigns.

2.6	Interest Rates	and Letter of	f Credit Fee: Rates.	, Payments	and Calculations.
-----	----------------	---------------	----------------------	------------	-------------------

- (a) Interest Rates. Except as provided in <u>Section 2.6(c)</u>, all Obligations (except for undrawn Letters of Credit) that have been charged to the applicable Loan Account pursuant to the terms hereof shall bear interest as follows:
- (i) if the relevant Obligation is a Revolving Loan that is a LIBOR Rate Loan, at a per annum rate equal to the LIBOR Rate plus the LIBOR Rate Margin.
- (ii) if the relevant Obligation is a Revolving Loan that is an Australian Base Rate Loan, at a per annum rate equal to the Australian Base Rate plus the Australian Base Rate Margin, and
 - (iii) otherwise, at a per annum rate equal to the Base Rate plus the Base Rate Margin.

Interest at the Default Rate shall be payable upon demand.

(b) Letter of Credit Fees.

- (i) U.S. Borrowers shall pay Agent (for the ratable benefit of the U.S. Revolving Lenders), a U.S. Letter of Credit fee (the "U.S. Letters") (which fee shall be in addition to the fronting fees and commissions, other fees, charges and expenses set forth in Section 2.11(I)) that shall accrue at a per annum rate equal to the LIBOR Rate Margin times the undrawn amount of all outstanding U.S. Letters of Credit.
- (ii) Australian Borrowers shall pay Agent (for the ratable benefit of the Australian Revolving Lenders), an Australian Letter of Credit fee (the "Australian Letter of Credit Fee"; and the U.S. Letter of Credit Fee and the Australian Letter of Credit Fee may also each be referred to herein as a "Letter of Credit Fee") (which fee shall be in addition to the fronting fees and commissions, other fees, charges and expenses set forth in Section 2.12(I)) that shall accrue at a per annum rate equal to the LIBOR Rate Margin times the undrawn amount of all outstanding Australian Letters of Credit.
 - (c) Default Rate. Upon the occurrence and during the continuation of an Event of Default,
- (i) the overdue Obligations shall bear interest at a per annum rate equal to two (2) percentage points above the per annum rate otherwise applicable thereunder, and
- (ii) the overdue Letter of Credit Fees shall be increased to two (2) percentage points above the per annum rate otherwise applicable hereunder.

(d) Payment.

- (i) Except to the extent provided to the contrary in <u>Section 2.4(g)</u>, <u>Section 2.10</u>, <u>Section 2.11(l)</u>, <u>Section 2.12(l)</u>, and <u>Section 2.13(a)</u>, (A) all U.S. Letter of Credit Fees, all U.S. Unused Line Fees and all other fees payable hereunder or under any of the other Loan Documents shall be due and payable, in arrears, on the first day of each quarter and on the Maturity Date, (B) all Australian Interest, Australian Letter of Credit Fees and Australian Unused Line Fees shall be due and payable, in arrears, on the first day of each month and on the Maturity Date, and (C) all costs and expenses payable hereunder or under any of the other Loan Documents, and all Lender Group Expenses shall be due and payable on the earlier of (x) the first day of the month following the date on which the applicable costs, expenses, or Lender Group Expenses were first incurred and invoiced or (y) the date on which demand therefor is made by Agent (it being acknowledged and agreed that any charging of such costs, expenses or Lender Group Expenses to the applicable Loan Account pursuant to the provisions of the following clauses (ii) and (iii) shall be deemed to constitute a demand for payment thereof for the purposes of this subclause (y)).
- (ii) U.S. Borrowers hereby authorize Agent, from time to time during any Cash Dominion Trigger Period, with prior notice to U.S. Borrowers that the Agent is exercising its rights under this <u>Section 2.6(d)(ii)</u> (provided that after delivery of such notice, no additional notice shall be required during such Cash Dominion Trigger Period), to charge to the U.S. Loan Account (A) on the first day of each quarter, all interest accrued during the prior quarter on the U.S. Revolving Loans hereunder, (B) on the first day of each quarter, all U.S. Letter of Credit Fees

accrued or chargeable hereunder during the prior quarter, (C) as and when due and payable, all fees and costs provided for in Section 2.10(a) or (c), (D) on the first day of each quarter, the Unused Line Fee accrued during the prior quarter pursuant to Section 2.10(b), (E) as and when due and payable, all other fees payable hereunder or under any of the other Loan Documents, (F) as and when due and payable, the fronting fees and all commissions, other fees, charges and expenses provided for in Section 2.11(l). (G) as and when due and payable, all other Lender Group Expenses of Agent, the U.S. Issuing Banks, and the U.S. Revolving Lenders, and (H) as and when due and payable all other payment obligations payable under any Loan Document or any U.S. Bank Product Agreement (including any amounts due and payable to the U.S. Bank Product Providers in respect of U.S. Bank Products). All amounts (including interest, fees, costs, expenses, Lender Group Expenses, or other amounts payable hereunder or under any other Loan Document or under any U.S. Bank Product Agreement) charged to the U.S. Loan Account shall thereupon constitute U.S. Revolving Loans hereunder, shall constitute U.S. Obligations hereunder, and shall initially accrue interest at the rate then applicable to U.S. Revolving Loans that are Base Rate Loans (unless and until converted into LIBOR Rate Loans in accordance with the terms of this Agreement).

- (iii) Australian Borrowers hereby authorize Agent, from time to time during the continuance of any Specified Event of Default, with prior notice to Australian Borrowers that the Agent is exercising its rights under this Section 2.6(d)(iii) (provided that after delivery of such notice, no additional notice shall be required during such Specified Event of Default), to charge to the Australian Loan Account (A) on the first day of each month, all interest accrued during the prior month on the Australian Revolving Loans hereunder, (B) on the first day of each month, all Australian Letter of Credit Fees and Australian Unused Line Fees accrued or chargeable hereunder during the prior month, (C) as and when due and payable, the fronting fees and all commissions, other fees, charges and expenses provided for in Section 2.11(l), (D) as and when due and payable, all other Lender Group Expenses of Agent, the Australian Issuing Banks, and the Australian Revolving Lenders, in each case, solely for amounts owed by the Australian Loan Parties, and (E) as and when due and payable all other payment obligations of Australian Borrowers payable under any Loan Document or any Australian Bank Product Agreement (including any amounts due and payable to the Australian Bank Product Providers in respect of Australian Bank Products). All amounts (including interest, fees, costs, expenses, Lender Group Expenses, or other amounts payable hereunder or under any other Loan Document or under any Australian Bank Product Agreement) charged to the Australian Loan Account shall thereupon constitute Australian Revolving Loans hereunder, shall constitute Australian Obligations hereunder, and shall initially accrue interest at the rate then applicable to Australian Revolving Loans that are Base Rate Loans (unless and until converted into LIBOR Rate Loans in accordance with the terms of this Agreement).
- (e) **Computation.** All interest and fees chargeable under the Loan Documents shall be computed on the basis of a 360-day year or, in the case of Base Rate Loans (or any fees or expenses based on the Base Rate), on the basis of a 365-day year or 366-day year, as applicable, in each case, for the actual number of days elapsed in the period during which the interest or fees accrue. In the event the Base Rate is changed from time to time hereafter, the rates of interest hereunder based upon the Base Rate automatically and immediately shall be increased or decreased by an amount equal to such change in the Base Rate.
- (f) Intent to Limit Charges to Maximum Lawful Rate. In no event shall the interest rate or rates payable under this Agreement, plus any other amounts paid in connection herewith, exceed the highest rate permissible under any law that a court of competent jurisdiction shall, in a final determination, deem applicable. Borrowers and the Lender Group, in executing and delivering this Agreement, intend legally to agree upon the rate or rates of interest and manner of payment stated within it; provided, that, anything contained herein to the contrary notwithstanding, if such rate or rates of interest or manner of payment exceeds the maximum allowable under applicable law, then, ipso facto, as of the date of this Agreement, the applicable Borrowers are and shall be liable only for the payment of such maximum amount as is allowed by law, and payment received from any Borrowers in excess of such legal maximum, whenever received, shall be applied to reduce the principal balance of the Obligations to the extent of such excess.
- 2.7 <u>Crediting Payments</u>. The receipt of any payment item by Agent shall not be required to be considered a payment on account unless such payment item is a wire transfer of immediately available federal funds made to the Agent's Account or unless and until such payment item is honored when presented for payment. Should any payment item not be honored when presented for payment, then Borrowers shall be deemed not to have made such payment and interest shall be calculated accordingly. Anything to the contrary contained herein notwithstanding, any payment item shall be deemed received by Agent only if it is received into the Agent's Account on a Business Day on or before 1:30 p.m. If any payment item is received into the Agent's Account on a non-Business Day or after 1:30 p.m. on a

Business Day (unless Agent, in its sole discretion, elects to credit it on the date received), it shall be deemed to have been received by Agent as of the opening of business on the immediately following Business Day.

- 2.8 <u>Designated Accounts</u>. Agent is authorized to make the Revolving Loans, and Issuing Bank is authorized to issue the Letters of Credit, under this Agreement based upon electronic or other instructions received from anyone purporting to be an Authorized Person or, without instructions, if pursuant to <u>Section 2.6(d)(ii)</u> and <u>(iii)</u>.
- (a) U.S. Borrowers agree to establish and maintain the U.S. Designated Account with the U.S. Designated Account Bank for the purpose of receiving the proceeds of the U.S. Revolving Loans requested by U.S. Borrowers and made by Agent or the U.S. Revolving Lenders hereunder. Unless otherwise agreed by Agent and U.S. Borrowers, any U.S. Revolving Loan or U.S. Swing Loan requested by U.S. Borrowers and made by Agent or the U.S. Revolving Lenders hereunder shall be made to the U.S. Designated Account.
- (b) Australian Borrowers agree to establish and maintain the Australian Designated Account with the Australian Designated Account Bank for the purpose of receiving the proceeds of the Australian Revolving Loans requested by Australian Borrowers and made by Agent or the Australian Revolving Lenders hereunder. Unless otherwise agreed by Agent and Australian Borrowers, any Australian Revolving Loan or Australian Swing Loan requested by Australian Borrowers and made by Agent or the Australian Revolving Lenders hereunder shall be made to the Australian Designated Account.

2.9 <u>Maintenance of Loan Accounts; Statements of Obligations.</u>

- (a) Agent shall maintain an account on its books in the name of U.S. Borrowers (the "U.S. Loan Account") on which U.S. Borrowers will be charged with all U.S. Revolving Loans (including U.S. Extraordinary Advances and U.S. Swing Loans) made by Agent, U.S. Swing Lender, or the U.S. Revolving Lenders to U.S. Borrowers or for U.S. Borrowers' account, the U.S. Letters of Credit issued or arranged by any U.S. Issuing Bank for U.S. Borrowers' account, and with all other payment U.S. Obligations hereunder or under the other Loan Documents, including, accrued interest, fees and expenses, and Lender Group Expenses. In accordance with Section 2.7, the U.S. Loan Account will be credited with all payments received by Agent from U.S. Borrowers or for U.S. Borrowers' account. Agent shall make available to U.S. Borrowers monthly statements regarding the U.S. Loan Account, including the principal amount of the U.S. Revolving Loans, interest accrued hereunder, fees accrued or charged hereunder or under the other Loan Documents, and a summary itemization of all charges and expenses constituting Lender Group Expenses accrued hereunder or under the other Loan Documents, and each such statement, absent manifest error, shall be conclusively presumed to be correct and accurate and constitute an account stated between U.S. Borrowers and the Lender Group unless, within 30 days after Agent first makes such a statement available to U.S. Borrowers, U.S. Borrowers shall deliver to Agent written objection thereto describing the error or errors contained in such statement.
- (b) Agent shall maintain an account on its books in the name of Australian Borrowers (the "Australian Loan Account") on which Australian Borrowers will be charged with all Australian Revolving Loans (including Australian Extraordinary Advances and Australian Swing Loans) made by Agent, Australian Swing Lender, or the Australian Revolving Lenders to Australian Borrowers or for Australian Borrowers' account, the Australian Letters of Credit issued or arranged by any Australian Issuing Bank for Australian Borrowers' account, and with all other payment Australian Obligations hereunder or under the other Loan Documents, including, accrued interest, fees and expenses, and Lender Group Expenses. In accordance with Section 2.7, the Australian Loan Account will be credited with all payments received by Agent from Australian Borrowers or for Australian Borrowers' account. Agent shall make available to Australian Borrowers monthly statements regarding the Australian Loan Account, including the principal amount of the Australian Revolving Loans, interest accrued hereunder, fees accrued or charged hereunder or under the other Loan Documents, and a summary itemization of all charges and expenses constituting Lender Group Expenses accrued hereunder or under the other Loan Documents, and each such statement, absent manifest error, shall be conclusively presumed to be correct and accurate and constitute an account stated between Australian Borrowers and the Lender Group unless, within 30 days after Agent first makes such a statement available to Australian Borrowers, Australian Borrowers shall deliver to Agent written objection thereto describing the error or errors contained in such statement.

2.10 **Fees.**

(a) **Agent Fees.** U.S. Borrowers shall pay to Agent, for the account of Agent, as and when due and payable under the terms of the Fee Letter, the fees set forth in Section 3 of the Fee Letter (it being understood that the "Closing Date" referred to therein shall mean March 30, 2015 and the reference therein to the "ABL Facility" shall be deemed a reference to Revolver Commitments established by this Agreement).

(b) Unused Line Fees.

- (i) U.S. Borrowers shall pay to Agent, for the ratable account of the U.S. Revolving Lenders, an unused line fee (the " <u>U.S. Unused Line Fee</u>") in an amount equal to the Applicable U.S. Unused Line Fee Percentage per annum times the result of (a) the aggregate amount of the U.S. Revolver Commitments, less (b) the average amount of the U.S. Revolver Usage during the immediately preceding quarter (or portion thereof), which U.S. Unused Line Fee shall be due and payable, in arrears, on the first day of each quarter from and after the Closing Date up to the first day of the quarter prior to the date on which the U.S. Obligations are paid in full and on the date on which the U.S. Obligations are paid in full. For purposes of calculating the U.S. Unused Line Fee, no portion of the U.S. Revolver Commitments shall be deemed used as a result of outstanding U.S. Swing Loans.
- (ii) Australian Borrowers shall pay to Agent, for the ratable account of the Australian Revolving Lenders, an unused line fee (the "Australian Unused Line Fee") in an amount equal to the Applicable Australian Unused Line Fee Percentage per annum times the result of (a) the aggregate amount of the Australian Revolver Commitments, less (b) the average amount of the Australian Revolver Usage during the immediately preceding quarter (or portion thereof), which Australian Unused Line Fee shall be due and payable, in arrears, on the first day of each month from and after the Closing Date up to the first day of the month prior to the date on which the Australian Obligations are paid in full and on the date on which the Australian Obligations are paid in full. For purposes of calculating the Australian Unused Line Fee, no portion of the Australian Revolver Commitments shall be deemed used as a result of outstanding Australian Swing Loans.
- (c) Field Examination and Other Fees. U.S. Borrowers shall pay to Agent, field examination, appraisal, and valuation fees and charges, as and when incurred or chargeable, the fees or charges paid or incurred by Agent (including allocated costs of employees of Agent) to perform field examinations of Parent or its Subsidiaries, to establish electronic collateral reporting systems, to appraise the Collateral, or any portion thereof, or to assess Parent's or its Subsidiaries' business valuation; provided, that so long as no Event of Default and no Field Examination/Appraisal Triggering Event shall have occurred and be continuing, U.S. Borrowers shall not be obligated to reimburse Agent for more than one (1) field examination, one (1) appraisal of the Collateral consisting of inventory, one (1) appraisal of the Collateral consisting of equipment, and one (1) business valuation during any calendar year; provided, further that (i) if a Field Examination/Appraisal Triggering Event occurs in any calendar year, the limits in the immediately preceding proviso shall each be increased to two (2) and (ii) during the existence and continuance of an Event of Default, the limits in the immediately preceding proviso shall each be increased to four (4).

2.11 U.S. Letters of Credit.

(a) Subject to the terms and conditions of this Agreement, upon the request of U.S. Borrowers made in accordance herewith, during the period from the Closing Date until the Letter of Credit Expiration Date, each U.S. Issuing Bank agrees to issue a requested U.S. Letter of Credit for the account of U.S. Borrowers or any of their respective Subsidiaries (it being understood that the U.S. Borrowers shall be primarily liable in respect of U.S. Letters of Credit Issued for the account of their Subsidiaries), it being agreed that Goldman Sachs Bank USA, Credit Suisse AG, Cayman Islands Branch and Deutsche Bank AG New York Branch shall only be required to issue standby U.S. Letters of Credit. By submitting a request to a U.S. Issuing Bank for the issuance of a U.S. Letter of Credit, U.S. Borrowers shall be deemed to have requested that such U.S. Issuing Bank issue the requested U.S. Letter of Credit. Each request for the issuance of a U.S. Letter of Credit, or the amendment, renewal, or extension of any outstanding U.S. Letter of Credit, shall be irrevocable and shall be made in writing by an Authorized Person and delivered to the applicable U.S. Issuing Bank via telefacsimile or other electronic method of transmission reasonably acceptable to such U.S. Issuing Bank and at least three (3) Business Days (or such shorter period as agreed by the applicable U.S. Issuing Bank) in advance of the requested date of issuance, amendment, renewal, or extension. Each such request shall be in form and substance reasonably satisfactory to such U.S. Issuing Bank and (i) shall specify (A) the amount and applicable Agreed Currency of such U.S. Letter of Credit, (B) the date of issuance, amendment, renewal, or

extension of such U.S. Letter of Credit, (C) the proposed expiration date of such U.S. Letter of Credit, (D) the name and address of the beneficiary of the U.S. Letter of Credit, and (E) such other information (including, the conditions to drawing, and, in the case of an amendment, renewal, or extension, identification of the U.S. Letter of Credit to be so amended, renewed, or extended) as shall be reasonably necessary to prepare, amend, renew, or extend such U.S. Letter of Credit, and (ii) shall be accompanied by such Issuer Documents as Agent or such U.S. Issuing Bank may reasonably request or require, to the extent that such requests or requirements are consistent with the Issuer Documents that such U.S. Issuing Bank generally requests for U.S. Letters of Credit in similar circumstances. The applicable U.S. Issuing Bank's records of the content of any such request will be conclusive, absent manifest error. Unless otherwise specified, all references herein to the amount of a U.S. Letter of Credit at any time shall be deemed to mean the maximum face amount of such U.S. Letter of Credit related thereto therefor, whether or not such maximum face amount is in effect at such time.

- (b) No U.S. Issuing Bank shall issue a U.S. Letter of Credit if any of the following would result after giving effect to the requested issuance:
- (i) the U.S. Letter of Credit Usage would exceed the U.S. Letter of Credit Sublimit, or the aggregate undrawn amount of all outstanding U.S. Letters of Credit issued by such U.S. Issuing Bank would exceed the U.S. Letter of Credit Sublimit with respect to such U.S. Issuing Bank,
- (ii) the U.S. Letter of Credit Usage would exceed the U.S. Maximum Revolver Amount less the sum of the outstanding principal amount of U.S. Revolving Loans (including U.S. Swing Loans) at such time,
- (iii) the U.S. Letter of Credit Usage would exceed the U.S. Borrowing Base at such time less the sum of the outstanding principal amount of U.S. Revolving Loans (including U.S. Swing Loans) at such time, or
- (iv) the expiration date of such requested U.S. Letter of Credit would occur after the Letter of Credit Expiration Date, unless such U.S. Issuing Bank and the Agent have approved such expiration date (it being understood that the obligation of U.S. Revolving Lenders to fund participations in U.S. Letters of Credit shall expire immediately following the Maturity Date (for this purpose, pursuant to clause (a) of the definition thereof)).

In addition, no U.S. Issuing Bank shall issue or be obligated to issue any U.S. Letter of Credit if it has actual knowledge that one or more of the applicable conditions precedent set forth in Section 3 is not satisfied on the requested Funding Date. On the Maturity Date, the US Borrowers shall provide Letter of Credit Collateralization to Agent to be held as security for U.S. Borrowers' reimbursement obligations in respect of drawings that may subsequently occur under issued and outstanding U.S. Letters of Credit.

- (c) In the event there is a Defaulting Lender as of the date of any request for the issuance of a U.S. Letter of Credit, no U.S. Issuing Bank shall be required to issue or arrange for such U.S. Letter of Credit to the extent (i) the Defaulting Lender's U.S. Letter of Credit Exposure with respect to such U.S. Letter of Credit may not be reallocated pursuant to Section 2.3(i)(ii), or (ii) such U.S. Issuing Bank has not otherwise entered into arrangements reasonably satisfactory to it and U.S. Borrowers to eliminate such U.S. Issuing Bank's risk with respect to the participation in such U.S. Letter of Credit of the Defaulting Lender, which arrangements may include U.S. Borrowers cash collateralizing such Defaulting Lender's U.S. Letter of Credit Exposure in accordance with Section 2.3(i)(iii). Additionally, no U.S. Issuing Bank shall have any obligation to issue a U.S. Letter of Credit if (A) any order, judgment, or decree of any Governmental Authority or arbitrator shall, by its terms, purport to enjoin or restrain such U.S. Issuing Bank from issuing such U.S. Letter of Credit, or any law applicable to such U.S. Issuing Bank or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such U.S. Issuing Bank shall prohibit or request that such U.S. Issuing Bank refrain from the issuance of letters of credit generally or such U.S. Letter of Credit would violate one or more policies of such U.S. Issuing Bank applicable to letters of credit generally, or (C) if amounts demanded to be paid under any U.S. Letter of Credit will or may not be in an Agreed Currency.
- (d) Any U.S. Issuing Bank (other than Bank of America or any of its Affiliates) shall notify Agent in writing no later than the Business Day immediately following the Business Day on which such U.S. Issuing Bank receives a request for issuance of any U.S. Letter of Credit. The applicable U.S. Issuing Bank will not issue any requested U.S. Letter of Credit if it receives written notice from the Agent (with a copy to Parent) that the proposed U.S. Letter of Credit would cause a U.S. Overadvance to occur. Each U.S. Letter of Credit shall be in form and substance reasonably acceptable to the applicable U.S. Issuing Bank, including the requirement that the amounts

payable thereunder must be payable in an Agreed Currency. If a U.S. Issuing Bank makes a payment under a U.S. Letter of Credit, U.S. Borrowers shall pay to Agent an amount equal to the applicable Letter of Credit Disbursement in the same currency as such U.S. Letter of Credit (i) within one Business Day after such Letter of Credit Disbursement in the case of U.S. Letters of Credit denominated in U.S. Dollars and (ii) within two Business Days after such Letter of Credit Disbursement in the case of U.S. Letters of Credit denominated in an Agreed Currency other than U.S. Dollars is made and, in the absence of such payment, the amount of the Letter of Credit Disbursement immediately and automatically shall be deemed to be a U.S. Revolving Loan hereunder (notwithstanding any failure to satisfy any condition precedent set forth in Section 3) and, initially, shall bear interest at the rate then applicable to U.S. Revolving Loans that are Base Rate Loans. If a Letter of Credit Disbursement is deemed to be a U.S. Revolving Loan hereunder, U.S. Borrowers' obligation to pay the amount of such Letter of Credit Disbursement to the applicable U.S. Issuing Bank shall be automatically converted into an obligation to pay the resulting U.S. Revolving Loan. Promptly following receipt by Agent of any payment from U.S. Borrowers pursuant to this paragraph, Agent shall distribute such payment to the applicable U.S. Issuing Bank, then to such U.S. Revolving Lenders have made payments pursuant to Section 2.11(e) to reimburse the applicable U.S. Issuing Bank, then to such U.S. Revolving Lenders and such U.S. Issuing Bank as their interests may appear.

- (e) Promptly following receipt of a notice of a Letter of Credit Disbursement pursuant to Section 2.11(d), each U.S. Revolving Lender agrees to fund its Pro Rata Share of any U.S. Revolving Loan deemed made pursuant to Section 2.11(d) on the same terms and conditions as if U.S. Borrowers had requested the amount thereof as a U.S. Revolving Loan and Agent shall promptly pay to the applicable U.S. Issuing Bank the amounts so received by it from the U.S. Revolving Lenders. By the issuance of a U.S. Letter of Credit (or an amendment, renewal, or extension of a U.S. Letter of Credit) and without any further action on the part of any U.S. Issuing Bank or the U.S. Revolving Lenders, the applicable U.S. Issuing Bank shall be deemed to have granted to each U.S. Revolving Lender, and each U.S. Revolving Lender shall be deemed to have purchased, a participation in each U.S. Letter of Credit issued by such U.S. Issuing Bank, in an amount equal to its Pro Rata Share of such U.S. Letter of Credit, and each such U.S. Revolving Lender agrees to pay to Agent, for the account of such U.S. Revolving Lender's Pro Rata Share of any Letter of Credit Disbursement made by U.S. Issuing Bank under the applicable U.S. Letter of Credit. In consideration and in furtherance of the foregoing, each U.S. Revolving Lender hereby absolutely and unconditionally agrees to pay to Agent, for the account of the applicable U.S. Issuing Bank, such Revolving Lender's Pro Rata Share of each Letter of Credit Disbursement made by such U.S. Issuing Bank and not reimbursed by U.S. Borrowers on the date due as provided in Section 2.11(d), or of any reimbursement payment that is required to be refunded (or that Agent or the applicable U.S. Issuing Bank elects, based upon the advice of counsel, to refund) to U.S. Borrowers for any reason. Each U.S. Revolving Lender acknowledges and agrees that its obligation to deliver to Agent, for the account of the applicable U.S. Issuing Bank, an amount equal to its respective Pro Rata Share of each Letter of Credit Disburs
- (f) Each U.S. Borrower agrees to indemnify, defend and hold harmless each member of the Lender Group (including each U.S. Issuing Bank and its branches, Affiliates, and correspondents) and each such Person's respective directors, officers, employees, attorneys and agents (each, including each U.S. Issuing Bank, a "U.S. Letter of Credit Related Person") (to the fullest extent permitted by law) from and against any and all claims, demands, suits, actions, investigations, proceedings, liabilities, fines, costs, penalties, and damages, and all reasonable fees and disbursements of one counsel in each relevant jurisdiction (and, solely in the case of an actual or perceived conflict of interest, one additional counsel to each group of affected persons similarly situated) and all other reasonable and documented costs and out-of-pocket expenses actually incurred in connection therewith or in connection with he enforcement of this indemnification (as and when they are incurred and irrespective of whether suit is brought), which may be incurred by or awarded against any U.S. Letter of Credit Related Person (other than any Taxes that are governed by Section 17) (the "U.S. Letter of Credit Indemnified Costs"), and which arise out of or in connection with, or as a result of this Agreement, any U.S. Letter of Credit, any Issuer Document, or any Drawing Document referred to in or related to any U.S. Letter of Credit, or any action or proceeding arising out of any of the foregoing (whether administrative, judicial or in connection with arbitration); in each case, including that resulting from the U.S. Letter of Credit Related Person's own negligence (other than resulting from such Person's bad faith or willful misconduct or constituting gross negligence); provided, however, that such indemnified Costs may be finally determined in a final, non-appealable judgment of a court of competent jurisdiction to have resulted from the bad faith, gross negligence or willful misconduct of the Letter of Credit Related Person claiming indemni

- (g) The liability of each U.S. Issuing Bank (or any other U.S. Letter of Credit Related Person) under, in connection with or arising out of any U.S. Letter of Credit (or pre-advice), regardless of the form or legal grounds of the action or proceeding, shall be limited to direct damages suffered by U.S. Borrowers that result from such U.S. Issuing Bank's bad faith, gross negligence or willful misconduct in (i) honoring a presentation under a U.S. Letter of Credit that on its face does not at least substantially comply with the terms and conditions of such U.S. Letter of Credit, (ii) failing to honor a presentation under a U.S. Letter of Credit that strictly complies with the terms and conditions of such U.S. Letter of Credit or (iii) retaining Drawing Documents presented under a U.S. Letter of Credit. Each U.S. Issuing Bank shall be deemed to have acted with due diligence and reasonable care if such U.S. Issuing Bank's conduct is in accordance with Standard Letter of Credit Practice or in accordance with this Agreement. U.S. Borrowers' aggregate remedies against each U.S. Issuing Bank and any U.S. Letter of Credit Related Person for wrongfully honoring a presentation under any U.S. Letter of Credit or wrongfully retaining honored Drawing Documents shall in no event exceed the aggregate amount paid by U.S. Borrowers to the applicable U.S. Issuing Bank in respect of the honored presentation in connection with such U.S. Letter of Credit under Section 2.11(d), plus interest at the rate then applicable to Base Rate Loans hereunder.
- (h) U.S. Borrowers are responsible for preparing or approving the final text of the U.S. Letter of Credit as issued by any U.S. Issuing Bank, irrespective of any assistance such U.S. Issuing Bank may provide such as drafting or recommending text or by such U.S. Issuing Bank's use or refusal to use text submitted by U.S. Borrowers. U.S. Borrowers are solely responsible for the suitability of the U.S. Letter of Credit for U.S. Borrowers' purposes. With respect to any U.S. Letter of Credit containing an "automatic amendment" to extend the expiration date of such U.S. Letter of Credit for additional consecutive periods of twelve (12) months (but in no event shall the expiration date extend beyond the Letter of Credit Expiration Date unless such U.S. Issuing Bank and the Agent have approved such expiration date (it being understood that the obligation of U.S. Revolving Lenders to fund participations in U.S. Letters of Credit shall expire immediately following the Maturity Date (for this purpose, pursuant to clause (a) of the definition thereof)), any U.S. Issuing Bank, in its sole and absolute discretion, may give notice of nonrenewal of such U.S. Letter of Credit and, if U.S. Borrowers do not at any time want such U.S. Letter of Credit to be renewed, U.S. Borrowers will so notify Agent and the applicable U.S. Issuing Bank at least 7 Business Days (or such later date as agreed to by the applicable U.S. Issuing Bank) before such U.S. Letter of Credit.
- (i) U.S. Borrowers' reimbursement and payment obligations under this Section are absolute, unconditional and irrevocable and shall be performed strictly in accordance with the terms of this Agreement under any and all circumstances whatsoever, provided, however, that subject to Section 2.11(g) above, the foregoing shall not release any U.S. Issuing Bank from such liability to U.S. Borrowers as may be finally determined in a final, non-appealable judgment of a court of competent jurisdiction against such U.S. Issuing Bank following reimbursement or payment of the obligations and liabilities, including reimbursement and other payment obligations, of U.S. Borrowers to such U.S. Issuing Bank arising under, or in connection with, this Section 2.11 or any U.S. Letter of Credit.
- (j) Without limiting any other provision of this Agreement, each U.S. Issuing Bank and each other U.S. Letter of Credit Related Person (if applicable) shall not be responsible to U.S. Borrowers for, and no U.S. Issuing Bank's rights and remedies against U.S. Borrowers and the obligation of U.S. Borrowers to reimburse each U.S. Issuing Bank for each drawing under each U.S. Letter of Credit shall be impaired by:
- (i) honor of a presentation under any U.S. Letter of Credit that on its face substantially complies with the terms and conditions of such U.S. Letter of Credit, even if the U.S. Letter of Credit requires strict compliance by the beneficiary;
- (ii) honor of a presentation of any Drawing Document that appears on its face to have been signed, presented or issued (A) by any purported successor or transferee of any beneficiary or other Person required to sign, present or issue such Drawing Document or (B) under a new name of the beneficiary;
- (iii) acceptance as a draft of any written or electronic demand or request for payment under a U.S. Letter of Credit, even if non-negotiable or not in the form of a draft or notwithstanding any requirement that such draft, demand or request bear any or adequate reference to a U.S. Letter of Credit;
- (iv) the identity or authority of any presenter or signer of any Drawing Document or the form, accuracy, genuineness or legal effect of any Drawing Document (other than the applicable U.S. Issuing Bank's determination that such Drawing Document appears on its face substantially to comply with the terms and conditions of a U.S. Letter of Credit);

- (v) acting upon any instruction or request relative to a U.S. Letter of Credit or requested a U.S. Letter of Credit that a U.S. Issuing Bank in good faith believes to have been given by a Person authorized to give such instruction or request;
- (vi) any errors, omissions, interruptions or delays in transmission or delivery of any message, advice or document (regardless of how sent or transmitted) or for errors in interpretation of technical terms or in translation or any delay in giving or failing to give notice to U.S. Borrowers;
- (vii) any acts, omissions or fraud by, or the insolvency of, any beneficiary, any nominated person or entity or any other Person or any breach of contract between any beneficiary and any U.S. Borrower or any of the parties to the underlying transaction to which the U.S. Letter of Credit relates;
- (viii) assertion or waiver of any provision of the ISP or UCP that primarily benefits an issuer of a letter of credit, including any requirement that any Drawing Document be presented to it at a particular hour or place;
- (ix) payment to any paying or negotiating bank (designated or permitted by the terms of the applicable Letter of Credit) claiming that it rightfully honored or is entitled to reimbursement or indemnity under Standard Letter of Credit Practice applicable to it;
- (x) acting or failing to act as required or permitted under Standard Letter of Credit Practice applicable to where Issuing Bank has issued, confirmed, advised or negotiated such Letter of Credit, as the case may be;
- (xi) honor of a presentation after the expiration date of any U.S. Letter of Credit notwithstanding that a presentation was made prior to such expiration date and dishonored by the applicable U.S. Issuing Bank if subsequently U.S. Issuing Bank or any court or other finder of fact determines such presentation should have been honored;
 - (xii) dishonor of any presentation that does not strictly comply or that is fraudulent, forged or otherwise not entitled to honor; or
- (xiii) honor of a presentation that is subsequently determined by the applicable U.S. Issuing Bank to have been made in violation of international, federal, state or local restrictions on the transaction of business with certain prohibited Persons.

(k) [Reserved.]

- (I) U.S. Borrowers shall pay immediately upon demand to Agent for the account of the applicable U.S. Issuing Bank as non-refundable fees, commissions, and charges (it being acknowledged and agreed that any charging of such fees, commissions, and charges to the U.S. Loan Account pursuant to the provisions of Section 2.6(d)(ii) shall be deemed to constitute a demand for payment thereof for the purposes of this Section 2.11(I)): (i) a fronting fee which shall be imposed by the applicable U.S. Issuing Bank upon the issuance of each Letter of Credit of 0.125% per annum of the face amount thereof, which fee shall be payable quarterly in arrears, on the first day of each quarter, and on maturity, plus (ii) any and all other customary commissions, fees and charges then in effect imposed by, and any and all reasonable and documented out-of-pocket expenses incurred by, the applicable U.S. Issuing Bank, or by any adviser, confirming institution or entity or other nominated person, relating to U.S. Letters of Credit, which charges shall be paid as and when incurred.
- (m) If by reason of (x) any Change in Law, or (y) compliance by any U.S. Issuing Bank or any other member of the Lender Group with any direction, request, or requirement (irrespective of whether having the force of law) of any Governmental Authority or monetary authority including, Regulation D of the Board of Governors as from time to time in effect (and any successor thereto):
- (i) any reserve, deposit, or similar requirement is or shall be imposed or modified in respect of any U.S. Letter of Credit issued or caused to be issued hereunder or hereby, or
- (ii) there shall be imposed on any U.S. Issuing Bank or any other member of the Lender Group any other condition regarding any U.S. Letter of Credit, or

(iii) there shall be imposed on any member of the Lender Group or Agent any Taxes (other than (A) Indemnified Taxes and (B) Excluded Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto.

and the result of the foregoing is to increase, directly or indirectly, the cost to any U.S. Issuing Bank or any other member of the Lender Group of issuing, making, participating in, or maintaining any U.S. Letter of Credit or to reduce the amount receivable in respect thereof, then, and in any such case, Agent may, at any time within 180 days after the additional cost is incurred or the amount received is reduced, notify U.S. Borrowers, and U.S. Borrowers shall pay within 30 days after demand therefor, such amounts as Agent may specify to be necessary to compensate the applicable U.S. Issuing Bank or any other member of the Lender Group for such additional cost or reduced receipt, together with interest on such amount from the date of such demand until payment in full thereof at the rate then applicable to Base Rate Loans hereunder; provided, that (A) U.S. Borrowers shall not be required to provide any compensation pursuant to this Section 2.11(m) for any such amounts incurred more than 180 days prior to the date on which the demand for payment of such amounts is first made to Borrowers, and (B) if an event or circumstance giving rise to such amounts is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof. The determination by Agent of any amount due pursuant to this Section 2.11(m), as set forth in a certificate setting forth the calculation thereof in reasonable detail, shall, in the absence of manifest or demonstrable error, be final and conclusive and binding on all of the parties hereto.

- (n) Unless otherwise expressly agreed by the applicable U.S. Issuing Bank and U.S. Borrowers when a U.S. Letter of Credit is issued, (i) the rules of the ISP and the UCP shall apply to each standby U.S. Letter of Credit, and (ii) the rules of the UCP shall apply to each commercial U.S. Letter of Credit.
- (o) In the event of a direct conflict between the provisions of this <u>Section 2.11</u> and any provision contained in any Issuer Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, the terms and provisions of this <u>Section 2.11</u> shall control and govern.
- (p) Notwithstanding anything to the contrary contained herein, any U.S. Issuing Bank may, upon thirty (30) days' notice to the Borrowers and the Lenders, resign as a U.S. Issuing Bank; provided that on or prior to the expiration of such 30-day period with respect to such resignation, the relevant U.S. Issuing Bank shall have identified a successor U.S. Issuing Bank reasonably acceptable to the Borrowers willing to accept its appointment as successor U.S. Issuing Bank. In the event of any such resignation of a U.S. Issuing Bank, the Borrowers shall be entitled to appoint from among the Lenders willing to accept such appointment a successor U.S. Issuing Bank hereunder; provided that no failure by the Borrowers to appoint any such successor shall affect the resignation of the relevant U.S. Issuing Bank, as the case may be, except as expressly provided above. If a U.S. Issuing Bank resigns as a U.S. Issuing Bank, it shall retain all the rights and obligations of a U.S. Issuing Bank hereunder with respect to all U.S. Letters of Credit outstanding as of the effective date of its resignation as a U.S. Issuing Bank and all obligations with respect thereto.

2.12 Australian Letters of Credit.

(a) Subject to the terms and conditions of this Agreement, upon the request of Australian Borrowers made in accordance herewith, during the period from the Closing Date until the Letter of Credit Expiration Date, each Australian Issuing Bank agrees to issue a requested Australian Letter of Credit for the account of Australian Borrowers and their respective Subsidiaries (it being understood that the Australian Borrowers shall be primarily liable in respect of Australian Letters of Credit Issued for the account of their Subsidiaries) it being agreed that Credit Suisse AG, Cayman Islands Branch and Deutsche Bank AG New York Branch shall only be required to issue standby Australian Letters of Credit. By submitting a request to an Australian Issuing Bank for the issuance of an Australian Letter of Credit, Australian Borrowers shall be deemed to have requested that such Australian Issuing Bank issue the requested Australian Letter of Credit. Each request for the issuance of an Australian Letter of Credit, or the amendment, renewal, or extension of any outstanding Australian Letter of Credit, shall be irrevocable and shall be made in writing by an Authorized Person and delivered to the applicable Australian Issuing Bank via telefacsimile or other electronic method of transmission reasonably acceptable to such Australian Issuing Bank and at least three (3) Business Days (or such shorter period as agreed to by the applicable Australian Issuing Bank) in advance of the requested date of issuance, amendment, renewal, or extension. Each such request shall be in form and substance reasonably satisfactory to such Australian Issuing Bank and (i) shall specify (A) the amount and applicable Agreed Currency of such Australian Letter of Credit, (B) the date of issuance, amendment, renewal, or extension of Such Australian Letter of Credit, (C) the

proposed expiration date of such Australian Letter of Credit, (D) the name and address of the beneficiary of the Australian Letter of Credit, and (E) such other information (including, the conditions to drawing, and, in the case of an amendment, renewal, or extension, identification of the Australian Letter of Credit to be so amended, renewed, or extended) as shall be reasonably necessary to prepare, amend, renew, or extend such Australian Letter of Credit, and (ii) shall be accompanied by such Issuer Documents as Agent or such Australian Issuing Bank may reasonably request or require, to the extent that such requests or requirements are consistent with the Issuer Documents that such Australian Issuing Bank generally requests for Australian Letters of Credit in similar circumstances. The applicable Australian Issuing Bank's records of the content of any such request will be conclusive, absent manifest error. Unless otherwise specified, all references herein to the amount of an Australian Letter of Credit at any time shall be deemed to mean the maximum face amount of such Australian Letter of Credit after giving effect to all increases thereof contemplated by such Australian Letter of Credit related thereto therefor, whether or not such maximum face amount is in effect at such time.

- (b) No Australian Issuing Bank shall issue an Australian Letter of Credit if any of the following would result after giving effect to the requested issuance:
- (i) the Australian Letter of Credit Usage would exceed the Australian Letter of Credit Sublimit, or the aggregate undrawn amount of all outstanding Australian Letters of Credit issued by such Australian Issuing Bank would exceed the Australian Letter of Credit Sublimit with respect to such Australian Issuing Bank,
- (ii) the Australian Letter of Credit Usage would exceed the Australian Maximum Revolver Amount less the outstanding principal amount of Australian Revolving Loans (including Australian Swing Loans) at such time,
- (iii) the Australian Letter of Credit Usage would exceed the Australian Borrowing Base at such time less the outstanding principal amount of Australian Revolving Loans (including Australian Swing Loans) at such time, or
- (iv) the expiration date of such requested Australian Letter of Credit would occur after the Letter of Credit Expiration Date, unless such Australian Issuing Bank and the Agent have approved such expiration date (it being understood that the obligation of Australian Revolving Lenders to fund participations in Australian Letters of Credit shall expire immediately following the Maturity Date (for this purpose, pursuant to clause (a) of the definition thereof)).

In addition, no Australian Issuing Bank shall issue or be obligated to issue any Australian Letter of Credit if it has actual knowledge that one or more of the applicable conditions precedent set forth in <u>Section 3</u> is not satisfied on the requested Funding Date. On the Maturity Date, the Australian Borrowers shall provide Letter of Credit Collateralization to Agent to be held as security for Australian Borrowers' reimbursement obligations in respect of drawings that may subsequently occur under issued and outstanding Australian Letters of Credit.

- (c) In the event there is a Defaulting Lender as of the date of any request for the issuance of an Australian Letter of Credit, no Australian Issuing Bank shall be required to issue or arrange for such Australian Letter of Credit to the extent (i) the Defaulting Lender's Australian Letter of Credit Exposure with respect to such Australian Letter of Credit may not be reallocated pursuant to Section 2.3(i)(iii), or (ii) such Australian Issuing Bank has not otherwise entered into arrangements reasonably satisfactory to it and Australian Borrowers to eliminate such Australian Issuing Bank's risk with respect to the participation in such Australian Letter of Credit of the Defaulting Lender, which arrangements may include Australian Borrowers cash collateralizing such Defaulting Lender's Australian Letter of Credit Exposure in accordance with Section 2.3(i)(iii). Additionally, no Australian Issuing Bank shall have any obligation to issue an Australian Issuing Bank from issuing such Australian Letter of Credit if (A) any order, judgment, or decree of any Governmental Authority or arbitrator shall, by its terms, purport to enjoin or restrain such Australian Issuing Bank from issuing such Australian Letter of Credit, or any law applicable to such Australian Issuing Bank or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such Australian Issuing Bank shall prohibit or request that such Australian Issuing Bank refrain from the issuance of letters of credit generally or such Australian Letter of Credit in particular, (B) the issuance of such Australian Letter of Credit would violate one or more policies of such Australian Issuing Bank applicable to letters of credit generally, or (C) if amounts demanded to be paid under any Australian Letter of Credit will or may not be in an Agreed Currency.
- (d) Any Australian Issuing Bank (other than Bank of America or any of its Affiliates) shall notify Agent in writing no later than the Business Day immediately following the Business Day on which such Australian

Issuing Bank receives a request for issuance of any Letter of Credit. The applicable Australian Issuing Bank will not issue any requested Australian Letter of Credit if it receives written notice from the Agent (with a copy to Parent) that the proposed Australian Letter of Credit would cause an Australian Overadvance to occur. Each Australian Letter of Credit shall be in form and substance reasonably acceptable to the applicable Australian Issuing Bank, including the requirement that the amounts payable thereunder must be payable in an Agreed Currency. If an Australian Issuing Bank makes a payment under an Australian Letter of Credit, Australian Borrowers shall pay to Agent an amount equal to the applicable Letter of Credit Disbursement in the same currency as such Australian Letter of Credit (i) within one Business Day after such Letter of Credit Disbursement in the case of Australian Letters of Credit denominated in U.S. Dollars and (ii) within two Business Days after such Letter of Credit Disbursement in the case of Australian Letters of Credit denominated in an Agreed Currency other than U.S. Dollars is made and, in the absence of such payment, the amount of the Letter of Credit Disbursement immediately and automatically shall be deemed to be an Australian Revolving Loan hereunder (notwithstanding any failure to satisfy any condition precedent set forth in Section 3) and, initially, shall bear interest at the rate then applicable to Australian Revolving Loans that are Base Rate Loans. If a Letter of Credit Disbursement is deemed to be an Australian Revolving Loan hereunder, Australian Borrowers' obligation to pay the amount of such Letter of Credit Disbursement to the applicable Australian Payment from Australian Borrowers pursuant to this paragraph, Agent shall distribute such payment to the applicable Australian Issuing Bank, then to such Australian Revolving Lenders have made payments pursuant to Section 2.12(e) to reimburse the applicable Australian Issuing Bank, then to such Australian Revolving Lender

- (e) Promptly following receipt of a notice of a Letter of Credit Disbursement pursuant to Section 2.12(d), each Australian Revolving Lender agrees to fund its Pro Rata Share of any Australian Revolving Loan deemed made pursuant to Section 2.12(d) on the same terms and conditions as if Australian Borrowers had requested the amount thereof as an Australian Revolving Loan and Agent shall promptly pay to the applicable Australian Issuing Bank the amounts so received by it from the Australian Revolving Lenders. By the issuance of an Australian Letter of Credit (or an amendment, renewal, or extension of an Australian Letter of Credit) and without any further action on the part of any Australian Issuing Bank or the Australian Revolving Lenders, the applicable Australian Issuing Bank shall be deemed to have granted to each Australian Revolving Lender, and each Australian Revolving Lender shall be deemed to have purchased, a participation in each Australian Letter of Credit issued by such Australian Issuing Bank, in an amount equal to its Pro Rata Share of such Australian Letter of Credit, and each such Australian Revolving Lender agrees to pay to Agent, for the account of such Australian Issuing Bank, such Australian Revolving Lender's Pro Rata Share of any Letter of Credit Disbursement made by Australian Issuing Bank under the applicable Australian Letter of Credit. In consideration and in furtherance of the foregoing, each Australian Revolving Lender hereby absolutely and unconditionally agrees to pay to Agent, for the account of the applicable Australian Issuing Bank, such Revolving Lender's Pro Rata Share of each Letter of Credit Disbursement made by such Australian Issuing Bank and not reimbursed by Australian Borrowers on the date due as provided in Section 2.12(d), or of any reimbursement payment that is required to be refunded (or that Agent or the applicable Australian Issuing Bank elects, based upon the advice of counsel, to refund) to Australian Borrowers for any reason. Each Australian Revolving Lender acknowledges and agrees that its obligation to deliver to Agent, for the account of the applicable Australian Issuing Bank, an amount equal to its respective Pro Rata Share of each Letter of Credit Disbursement pursuant to this Section 2.12(e) shall be absolute and unconditional and such remittance shall be made notwithstanding the occurrence or continuation of an Event of Default or Default or the failure to satisfy any condition set forth in Section 3.
- (f) Each Australian Borrower agrees to indemnify, defend and hold harmless each member of the Lender Group (including each Australian Issuing Bank and its branches, Affiliates, and correspondents) and each such Person's respective directors, officers, employees, attorneys and agents (each, including each Australian Issuing Bank, an "Australian Letter of Credit Related Person") (to the fullest extent permitted by law) from and against any and all claims, demands, suits, actions, investigations, proceedings, liabilities, fines, costs, penalties, and damages, and all reasonable fees and disbursements of counsel (limited to one primary counsel and one local counsel in each relevant jurisdiction (including Australia) for all Australian Letter of Credit Related Persons taken as a whole, and, solely in the case of an actual or perceived conflict of interest, one additional counsel to each group of Australian Letter of Credit Related Persons taken as a whole) and all other reasonable and documented costs and out-of-pocket expenses actually incurred in connection therewith or in connection with the enforcement of this indemnification (as and when they are incurred and irrespective of whether suit is brought), which may be incurred by or awarded against any Australian Letter of Credit Related Person (other than any Taxes that are governed by Section 17) (the "Australian Letter of Credit Indemnified Costs"), and which arise out of or in connection with, or as a result of this Agreement, any Australian Letter of Credit, any Issuer Document, or any Drawing Document referred to in or related to any Australian Letter of Credit, or any action or proceeding arising out of any of the foregoing (whether administrative, judicial or in

connection with arbitration); <u>provided</u>, <u>however</u>, that such indemnity shall not be available to any Australian Letter of Credit Related Person claiming indemnification to the extent that such Australian Letter of Credit Indemnified Costs may be finally determined in a final, non-appealable judgment of a court of competent jurisdiction to have resulted from the bad faith, gross negligence or willful misconduct of the Letter of Credit Related Person claiming indemnity. This indemnification provision shall survive termination of this Agreement and all Australian Letters of Credit.

- (g) The liability of each Australian Issuing Bank (or any other Australian Letter of Credit Related Person) under, in connection with or arising out of any Australian Letter of Credit (or pre-advice), regardless of the form or legal grounds of the action or proceeding, shall be limited to direct damages suffered by Australian Borrowers that result from such Australian Issuing Bank's bad faith, gross negligence or willful misconduct in (i) honoring a presentation under an Australian Letter of Credit that on its face does not at least substantially comply with the terms and conditions of such Australian Letter of Credit, (ii) failing to honor a presentation under an Australian Letter of Credit that strictly complies with the terms and conditions of such Australian Letter of Credit or (iii) retaining Drawing Documents presented under an Australian Letter of Credit. Each Australian Issuing Bank shall be deemed to have acted with due diligence and reasonable care if such Australian Issuing Bank's conduct is in accordance with Standard Letter of Credit Practice or in accordance with this Agreement. Australian Borrowers' aggregate remedies against each Australian Issuing Bank and any Australian Letter of Credit Related Person for wrongfully honoring a presentation under any Australian Letter of Credit or wrongfully retaining honored Drawing Documents shall in no event exceed the aggregate amount paid by Australian Borrowers to the applicable Australian Issuing Bank in respect of the honored presentation in connection with such Australian Letter of Credit under Section 2.12(d), plus interest at the rate then applicable to Base Rate Loans hereunder.
- (h) Australian Borrowers are responsible for preparing or approving the final text of the Australian Letter of Credit as issued by any Australian Issuing Bank, irrespective of any assistance such Australian Issuing Bank may provide such as drafting or recommending text or by such Australian Issuing Bank's use or refusal to use text submitted by the Australian Borrowers. Australian Borrowers are solely responsible for the suitability of the Australian Letter of Credit for Australian Borrowers' purposes. With respect to any Australian Letter of Credit containing an "automatic amendment" to extend the expiration date of such Australian Letter of Credit for additional consecutive periods of twelve (12) months (but in no event shall the expiration date extend beyond the Letter of Credit Expiration Date unless such Australian Issuing Bank and the Agent have approved such expiration date (it being understood that the obligation of Australian Revolving Lenders to fund participations in Australian Letters of Credit shall expire immediately following the Maturity Date (for this purpose, pursuant to clause (a) of the definition thereof)), any Australian Issuing Bank, in its sole and absolute discretion, may give notice of nonrenewal of such Australian Letter of Credit on the renewed, Australian Borrowers will so notify Agent and the applicable Australian Issuing Bank at least 7 Business Days (or such later date as agreed to by the applicable Australian Issuing Bank) before such Australian Issuing Bank is required to notify the beneficiary of such Australian Letter of Credit or any advising bank of such nonrenewal pursuant to the terms of such Australian Letter of Credit.
- (i) Australian Borrowers' reimbursement and payment obligations under this Section 2.12 are absolute, unconditional and irrevocable and shall be performed strictly in accordance with the terms of this Agreement under any and all circumstances whatsoever, provided, however, that subject to Section 2.12(g) above, the foregoing shall not release any Australian Issuing Bank from such liability to Australian Borrowers as may be finally determined in a final, non-appealable judgment of a court of competent jurisdiction against such Australian Issuing Bank following reimbursement or payment of the obligations and liabilities, including reimbursement and other payment obligations, of Australian Borrowers to such Australian Issuing Bank arising under, or in connection with, this Section or any Australian Letter of Credit.
- (j) Without limiting any other provision of this Agreement, each Australian Issuing Bank and each other Australian Letter of Credit Related Person (if applicable) shall not be responsible to Australian Borrowers for, and no Australian Issuing Bank's rights and remedies against Australian Borrowers and the obligation of Australian Borrowers to reimburse each Australian Issuing Bank for each drawing under each Australian Letter of Credit shall be impaired by:
- (i) honor of a presentation under any Australian Letter of Credit that on its face substantially complies with the terms and conditions of such Australian Letter of Credit, even if the Australian Letter of Credit requires strict compliance by the beneficiary;

- (ii) honor of a presentation of any Drawing Document that appears on its face to have been signed, presented or issued (A) by any purported successor or transferee of any beneficiary or other Person required to sign, present or issue such Drawing Document or (B) under a new name of the beneficiary;
- (iii) acceptance as a draft of any written or electronic demand or request for payment under an Australian Letter of Credit, even if non-negotiable or not in the form of a draft or notwithstanding any requirement that such draft, demand or request bear any or adequate reference to an Australian Letter of Credit;
- (iv) the identity or authority of any presenter or signer of any Drawing Document or the form, accuracy, genuineness or legal effect of any Drawing Document (other than the applicable Australian Issuing Bank's determination that such Drawing Document appears on its face substantially to comply with the terms and conditions of an Australian Letter of Credit);
- (v) acting upon any instruction or request relative to an Australian Letter of Credit or requested Australian Letter of Credit that an Australian Issuing Bank in good faith believes to have been given by a Person authorized to give such instruction or request;
- (vi) any errors, omissions, interruptions or delays in transmission or delivery of any message, advice or document (regardless of how sent or transmitted) or for errors in interpretation of technical terms or in translation or any delay in giving or failing to give notice to Australian Borrowers;
- (vii) any acts, omissions or fraud by, or the insolvency of, any beneficiary, any nominated person or entity or any other Person or any breach of contract between any beneficiary and any Australian Borrower or any of the parties to the underlying transaction to which the Australian Letter of Credit relates;
- (viii) assertion or waiver of any provision of the ISP or UCP that primarily benefits an issuer of a letter of credit, including any requirement that any Drawing Document be presented to it at a particular hour or place;
- (ix) payment to any paying or negotiating bank (designated or permitted by the terms of the applicable Letter of Credit) claiming that it rightfully honored or is entitled to reimbursement or indemnity under Standard Letter of Credit Practice applicable to it;
- (x) acting or failing to act as required or permitted under Standard Letter of Credit Practice applicable to where Issuing Bank has issued, confirmed, advised or negotiated such Letter of Credit, as the case may be;
- (xi) honor of a presentation after the expiration date of any Australian Letter of Credit notwithstanding that a presentation was made prior to such expiration date and dishonored by the applicable Australian Issuing Bank if subsequently Australian Issuing Bank or any court or other finder of fact determines such presentation should have been honored;
 - (xii) dishonor of any presentation that does not strictly comply or that is fraudulent, forged or otherwise not entitled to honor; or
- (xiii) honor of a presentation that is subsequently determined by the applicable Australian Issuing Bank to have been made in violation of international, federal, state or local restrictions on the transaction of business with certain prohibited Persons.

(k) [Reserved.]

(I) Australian Borrowers shall pay immediately upon demand to Agent for the account of the applicable Australian Issuing Bank as non-refundable fees, commissions, and charges (it being acknowledged and agreed that any charging of such fees, commissions, and charges to the Australian Loan Account pursuant to the provisions of Section 2.6(d)(iii) shall be deemed to constitute a demand for payment thereof for the purposes of this Section 2.12(i)): (i) a fronting fee which shall be imposed by the applicable Australian Issuing Bank upon the issuance of each Letter of Credit of 0.125% per annum of the face amount thereof, which fee shall be payable monthly in arrears, on the first day of each month, and on maturity, plus (ii) any and all other customary commissions, fees and charges then in effect imposed by, and any and all reasonable and documented out-of-pocket expenses incurred by, the

applicable Australian Issuing Bank, or by any adviser, confirming institution or entity or other nominated person, relating to Australian Letters of Credit, which charges shall be paid as and when incurred.

- (m) If by reason of (x) any Change in Law, or (y) compliance by any Australian Issuing Bank or any other member of the Lender Group with any direction, request, or requirement (irrespective of whether having the force of law) of any Governmental Authority or monetary authority including, Regulation D of the Board of Governors as from time to time in effect (and any successor thereto):
- (i) any reserve, deposit, or similar requirement is or shall be imposed or modified in respect of any Australian Letter of Credit issued or caused to be issued hereunder or hereby,
- (ii) there shall be imposed on any Australian Issuing Bank or any other member of the Lender Group any other condition regarding any Australian Letter of Credit, or
- (iii) there shall be imposed on any member of the Lender Group or Agent any Taxes (other than (A) Indemnified Taxes and (B) Excluded Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto.

and the result of the foregoing is to increase, directly or indirectly, the cost to any Australian Issuing Bank or any other member of the Lender Group of issuing, making, participating in, or maintaining any Australian Letter of Credit or to reduce the amount receivable in respect thereof, then, and in any such case, Agent may, at any time within a reasonable period after the additional cost is incurred or the amount received is reduced, notify Australian Borrowers, and Australian Borrowers shall pay within 30 days after demand therefor, such amounts as Agent may specify to be reasonably necessary to compensate the applicable Australian Issuing Bank or any other member of the Lender Group for such additional cost or reduced receipt, together with interest on such amount from the date of such demand until payment in full thereof at the rate then applicable to Base Rate Loans hereunder; provided, that (A) Australian Borrowers shall not be required to provide any compensation pursuant to this Section 2.12(m) for any such amounts incurred more than 180 days prior to the date on which the demand for payment of such amounts is first made to Borrowers, and (B) if an event or circumstance giving rise to such amounts is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof. The determination by Agent of any amount due pursuant to this Section 2.12(m), as set forth in a certificate setting forth the calculation thereof in reasonable detail, shall, in the absence of manifest or demonstrable error, be final and conclusive and binding on all of the parties hereto.

- (n) Unless otherwise expressly agreed by the applicable Australian Issuing Bank and Australian Borrowers when an Australian Letter of Credit is issued, (i) the rules of the ISP and the UCP shall apply to each standby Australian Letter of Credit, and (ii) the rules of the UCP shall apply to each commercial Australian Letter of Credit.
- (o) In the event of a direct conflict between the provisions of this <u>Section 2.12</u> and any provision contained in any Issuer Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, the terms and provisions of this <u>Section 2.12</u> shall control and govern.
- (p) Notwithstanding anything to the contrary contained herein, any Australian Issuing Bank may, upon thirty (30) days' notice to the Borrowers and the Lenders, resign as an Australian Issuing Bank; provided that on or prior to the expiration of such 30-day period with respect to such resignation, the relevant Australian Issuing Bank shall have identified a successor Australian Issuing Bank reasonably acceptable to the Borrowers willing to accept its appointment as successor Australian Issuing Bank. In the event of any such resignation of an Australian Issuing Bank, the Borrowers shall be entitled to appoint from among the Lenders willing to accept such appointment a successor Australian Issuing Bank hereunder; provided that no failure by the Borrowers to appoint any such successor shall affect the resignation of the relevant Australian Issuing Bank, as the case may be, except as expressly provided above. If an Australian Issuing Bank resigns as an Australian Issuing Bank, it shall retain all the rights and obligations of an Australian Issuing Bank hereunder with respect to all Australian Letters of Credit outstanding as of the effective date of its resignation as an Australian Issuing Bank and all obligations with respect thereto.

2.13 LIBOR Option.

(a) Interest and Interest Payment Dates. In lieu of having interest charged at the rate based upon the Base Rate, the applicable Borrowers shall have the option, subject to Section 2.13(b) below (the "LIBOR Option") to have interest on all or a portion of the Revolving Loans be charged (whether at the time when made (unless otherwise provided herein), upon conversion from a Base Rate Loan or an Australian Base Rate Loan to a LIBOR Rate Loan, or upon continuation of a LIBOR Rate Loan as a LIBOR Rate Loan) at a rate of interest based upon the LIBOR Rate. Interest on LIBOR Rate Loans shall be payable on the earliest of (i) the last day of the Interest Period applicable thereto; provided, that, subject to the following clauses (ii) and (iii), in the case of any Interest Period greater than 3 months in duration, interest shall be payable at 3 month intervals after the commencement of the applicable Interest Period and on the last day of such Interest Period), (ii) the date on which all or any portion of the Obligations are accelerated pursuant to the terms hereof, or (iii) the date on which this Agreement is terminated pursuant to the terms hereof. On the last day of each applicable Interest Period, unless Borrowers have properly exercised the LIBOR Option with respect thereto, the interest rate applicable to such LIBOR Rate Loan automatically shall convert to the rate of interest then applicable to Base Rate Loans of the same type hereunder. At any time that an Event of Default has occurred and is continuing, at the written election of the Required Lenders, no Borrower shall have the option to request that any Revolving Loans bear interest at a rate based upon the LIBOR Rate.

(b) LIBOR Election.

- (i) Borrowers may, at any time and from time to time, so long as no Event of Default has occurred and is continuing, and the Required Lenders have elected pursuant to Section 2.13(a) not to permit LIBOR Rate Loans, elect to exercise the LIBOR Option by notifying Agent prior to 11:00 a.m. at least three (3) Business Days prior to the commencement of the proposed Interest Period (the "LIBOR Deadline"). Notice of Borrowers' election of the LIBOR Option for a permitted portion of the Revolving Loans and an Interest Period pursuant to this Section shall be made by delivery to Agent of a LIBOR Notice received by Agent before the LIBOR Deadline, or by electronic notice received by Agent before the LIBOR Deadline (to be confirmed by delivery to Agent of a LIBOR Notice received by Agent prior to 5:00 p.m. on the same day). Promptly upon its receipt of each such LIBOR Notice, Agent shall provide a copy thereof to each of the affected Lenders. Each LIBOR Notice shall be irrevocable and binding on the applicable Borrowers.
- (ii) In connection with each LIBOR Rate Loan, each U.S. Borrower and each Australian Borrower, as applicable, of such U.S. Revolving Loan or Australian Revolving Loan, as applicable, shall indemnify, defend, and hold Agent and the Lenders harmless against any actual loss, cost, or expense incurred by Agent or any Lender as a result of (A) the payment of any principal of any applicable LIBOR Rate Loan other than on the last day of an Interest Period applicable thereto (including as a result of an Event of Default), (B) the conversion of any LIBOR Rate Loan other than on the last day of the Interest Period applicable thereto, or (C) the failure to borrow, convert, continue or prepay any LIBOR Rate Loan on the date specified in any LIBOR Notice delivered pursuant hereto (such losses, costs, or expenses, "Funding Losses"). A certificate of Agent or a Lender delivered to the applicable Borrowers setting forth in reasonable detail any amount or amounts that Agent or such Lender is entitled to receive pursuant to this Section 2.13 shall be conclusive absent manifest error. Such Borrowers shall pay such amount to Agent or the Lender, as applicable, within 30 days of the date of its receipt of such certificate. If a payment of any LIBOR Rate Loan on a day other than the last day of the applicable Interest Period would result in a Funding Loss, Agent may, in its sole discretion at the request of the applicable Borrowers, hold the amount of such payment as cash collateral in support of the Obligations until the last day of such Interest Period and apply such amounts to the payment of the applicable LIBOR Rate Loan on such last day, it being agreed that Agent has no obligation to so defer the application of payments to any LIBOR Rate Loan and that, in the event that Agent does not defer such application, the applicable Borrowers shall be obligated to pay any resulting Funding Losses.
- (iii) Unless Agent, in its sole discretion, agrees otherwise, Borrowers shall have not more than seven (7) LIBOR Rate Loans in effect at any given time. Borrowers may only exercise the LIBOR Option for proposed LIBOR Rate Loans of at <u>least</u> \$1,000,000 (and integral multiples of \$500,000 in excess thereof).
- (c) **Conversion.** The applicable Borrowers may convert LIBOR Rate Loans to Base Rate Loans at any time; provided, that in the event that LIBOR Rate Loans are converted or prepaid on any date that is not the last day of the Interest Period applicable thereto, including as a result of any prepayment through the required application by Agent of any payments or proceeds of Collateral in accordance with <u>Section 2.4(b)</u> or for any other reason, including early termination of the term of this Agreement or acceleration of all or any portion of the Obligations pursuant to the

terms hereof, each Borrower of such Revolving Loans shall indemnify, defend, and hold Agent and the Lenders and their Participants harmless against any and all Funding Losses in accordance with Section 2.13(b)(ii).

(d) Special Provisions Applicable to LIBOR Rate.

- (i) The LIBOR Rate may be adjusted by Agent with respect to any Lender on a prospective basis to take into account any additional or increased costs to such Lender of maintaining or obtaining any eurodollar deposits or increased costs, in each case, due to changes in applicable law occurring subsequent to the commencement of the then applicable Interest Period, including any Changes in Law (including any Changes in Laws relating to Taxes, other than Excluded Taxes and Indemnified Taxes, which shall be governed by Section 17) and changes in the reserve requirements imposed by the Board of Governors, which additional or increased costs would increase the cost of funding or maintaining loans bearing interest at the LIBOR Rate. In any such event, the affected Lender shall give Borrowers and Agent notice of such a determination and adjustment and Agent promptly shall transmit the notice to each other Lender and, upon its receipt of the notice from the affected Lender, Borrowers may, by notice to such affected Lender (A) require such Lender to furnish to Borrowers a statement setting forth in reasonable detail the basis for adjusting such LIBOR Rate and the method for determining the amount of such adjustment, or (B) repay the LIBOR Rate Loans of such Lender with respect to which such adjustment is made (together with any amounts due under Section 2.13(b)(ii)).
- (ii) If any Lender determines that any applicable law has made it unlawful for any Lender to make, maintain or fund LIBOR Rate Loans, or to determine or charge interest rates based upon the LIBOR Rate, or any Governmental Authority has imposed material restrictions on the authority of such Lender to purchase or sell, or to take deposits of, U.S. Dollars in the applicable interbank market, then, on notice thereof by such Lender to Parent through Agent, any obligation of such Lender to make or continue LIBOR Rate Loans or to convert Base Rate Loans to LIBOR Rate Loans, shall be suspended until such Lender notifies Agent and Parent that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Borrowers shall, upon demand from such Lender (with a copy to Agent), prepay or convert all such LIBOR Rate Loans of such Lender to Base Rate Loans, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such LIBOR Rate Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such LIBOR Rate Loans borrower shall also pay accrued interest on the amount of Loans so prepaid or converted.
- (iii) If the Required Lenders determine that for any reason in connection with any request for a LIBOR Rate Loan or a conversion to or continuation thereof that (a) deposits are not being offered to banks in the applicable offshore interbank market for the applicable amount and Interest Period of such LIBOR Rate Loan, (b) adequate and reasonable means do not exist for determining the LIBOR Rate for any requested Interest Period with respect to a proposed LIBOR Rate Loan, or (c) the LIBOR Rate for any requested Interest Period with respect to a proposed LIBOR Rate Loan does not adequately and fairly reflect the cost to such Lenders of funding such LIBOR Rate Loan, Agent will promptly so notify Parent and each Lender. Thereafter, the obligation of the Lenders to make or maintain LIBOR Rate Loans shall be suspended until Agent (upon the instruction of the Required Lenders) revokes such notice. Upon receipt of such notice, Parent (on behalf of the Borrowers) may revoke any pending request for a Borrowing of, conversion to or continuation of LIBOR Rate Loans or, failing that, will be deemed to have converted such request into a request for a Borrowing of Base Rate Loans in the amount specified therein.
- (iv) Notwithstanding anything to the contrary in this Agreement or any other Loan Documents, if Agent determines (which determination shall be conclusive absent manifest error), or Parent or the Required Lenders notify Agent (with, in the case of the Required Lenders, a copy to Parent) that Parent or the Required Lenders (as applicable) have determined, that:
- (A) adequate and reasonable means do not exist for ascertaining the LIBOR Rate for any requested Interest Period, including, without limitation, because the LIBOR Rate is not available or published on a current basis and such circumstances are unlikely to be temporary; or
- (B) the administrator of the LIBOR Rate or a Governmental Authority having jurisdiction over Agent has made a public statement identifying a specific date after which the LIBOR Rate shall no longer be made available, or used for determining the interest rate of loans (such specific date, the "Scheduled Unavailability Date"), or

(C) syndicated loans currently being executed, or that include language similar to that contained in this Section, are being executed or amended (as applicable) to incorporate or adopt a new benchmark interest rate to replace the LIBOR Rate,

then, reasonably promptly after such determination by Agent or receipt by Agent of such notice, as applicable, Agent and Parent may amend this Agreement to replace the LIBOR Rate with an alternate benchmark rate (including any mathematical or other adjustments to the benchmark (if any) incorporated therein), giving due consideration to any evolving or then existing convention for similar U.S. dollar denominated syndicated credit facilities for such alternative benchmarks (any such proposed rate, a "<u>LIBOR Successor Rate</u>"), together with any proposed LIBOR Successor Rate Conforming Changes and any such amendment shall become effective at 5:00 p.m. (New York time) on the fifth Business Day after Agent shall have posted such proposed amendment to all Lenders and Parent unless, prior to such time, Lenders comprising the Required Lenders have delivered to Agent written notice that such Required Lenders do not accept such amendment.

If no LIBOR Successor Rate has been determined and the circumstances under clause (A) above exist or the Scheduled Unavailability Date has occurred (as applicable), Agent will promptly so notify Parent and each Lender. Thereafter, (x) the obligation of the Lenders to make or maintain LIBOR Rate Loans shall be suspended (to the extent of the affected LIBOR Rate Loans or Interest Periods) and (y) the LIBOR Rate component shall no longer be utilized in determining the Base Rate. Upon receipt of such notice, any Borrower may revoke any pending request for a Borrowing of, conversion to or continuation of LIBOR Rate Loans (to the extent of the affected LIBOR Rate Loans or Interest Periods) or, failing that, will be deemed to have converted such request into a request for a Borrowing of Base Rate Loans (subject to the foregoing clause (y)) in the amount specified therein.

Notwithstanding anything else herein, any definition of LIBOR Successor Rate shall provide that in no event shall such LIBOR Successor Rate be less than zero for purposes of this Agreement.

(e) **No Requirement of Matched Funding.** Anything to the contrary contained herein notwithstanding, neither Agent, nor any Lender, nor any of their Participants, is required actually to acquire eurodollar deposits to fund or otherwise match fund any Obligation as to which interest accrues at the LIBOR Rate.

2.14 Capital Requirements.

(a) If, after the date hereof, any Issuing Bank or any Lender determines that (i) any Change in Law regarding capital or reserve requirements for banks or bank holding companies, or (ii) compliance by such Issuing Bank or such Lender, or their respective parent bank holding companies, with any guideline, request or directive of any Governmental Authority regarding capital adequacy or liquidity requirements (whether or not having the force of law), has the effect of reducing the return on such Issuing Bank's, such Lender's, or such holding companies' capital as a consequence of such Issuing Bank's or such Lender's commitments hereunder to a level below that which such Issuing Bank, such Lender, or such holding companies could have achieved but for such Change in Law or compliance (taking into consideration such Issuing Bank's, such Lender's, or such holding companies' then existing policies with respect to capital adequacy or liquidity requirements and assuming the full utilization of such entity's capital) by any amount deemed by such Issuing Bank or such Lender to be material, then such Issuing Bank or such Lender may notify Borrowers and Agent thereof. Following receipt of such notice, the applicable Borrowers agree to pay such Issuing Bank or such Lender on demand the amount of such reduction of return of capital as and when such reduction is determined, payable within 30 days after presentation by such Issuing Bank or such Lender of a statement in the amount and setting forth in reasonable detail such Issuing Bank's or such Lender's calculation thereof and the assumptions upon which such calculation was based (which statement shall be deemed true and correct absent manifest error). In determining such amount, such Issuing Bank or such Lender may use any reasonable averaging and attribution methods. Failure or delay on the part of such Issuing Bank or any Lender to demand compensation pursuant to this Section shall not constitute a waiver of such Issuing Bank's or such Lender's right to demand such compensation; provided that no Borrower shall be required to compensate an Issuing Bank or a Lender pursuant to this Section for any reductions in return incurred more than 180 days prior to the date that such Issuing Bank or such Lender notifies Borrowers of such Change in Law giving rise to such reductions and of such Lender's intention to claim compensation therefor, provided further that if such claim arises by reason of the Change in Law that is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof.

- (b) If Issuing Bank or any Lender requests additional or increased costs referred to in Section 2.11(m), Section 2.12(m), or Section 2.13(d)(i) or amounts under Section 2.14(a) or sends a notice under Section 2.13(d)(ii) relative to changed circumstances (such Issuing Bank or Lender, an " Affected Lender"), then such Affected Lender shall use reasonable efforts to promptly designate a different one of its lending offices or to assign its rights and obligations hereunder to another of its offices or branches, if (i) in the reasonable judgment of such Affected Lender, such designation or assignment would eliminate or reduce amounts payable pursuant to Section 2.11(m), Section 2.12(m), Section 2.13(d)(i) or Section 2.14(a), as applicable, or would eliminate the illegality or impracticality of funding or maintaining LIBOR Rate Loans and (ii) in the reasonable judgment of such Affected Lender, such designation or assignment would not subject it to any material unreimbursed cost or expense and would not otherwise be materially disadvantageous to it. Borrowers agree to pay all reasonable out-of-pocket costs and expenses incurred by such Affected Lender in connection with any such designation or assignment. If, after such reasonable efforts, such Affected Lender does not so designate a different one of its lending offices or assign its rights to another of its offices or branches so as to eliminate Borrowers' obligation to pay any future amounts to such Affected Lender pursuant to Section 2.11(m), Section 2.12(m), Section 2.13(d)(i) or Section 2.14(a), as applicable, or to enable the applicable Borrowers to obtain LIBOR Rate Loans, then Borrowers (without prejudice to any amounts then due to such Affected Lender under Section 2.11(m), Section 2.12(m), Section 2.13(d)(i) or Section 2.14(a), as applicable) may, unless prior to the effective date of any such assignment the Affected Lender withdraws its request for such additional amounts under Section 2.11(m), Section 2.12(m), Section 2.13(d)(i) or Section 2.14(a), as applicable, or indicates that it is no longer unlawful or impractical to fund or maintain LIBOR Rate Loans, may designate a different Issuing Bank or substitute a Lender, in each case, reasonably acceptable to Agent to purchase the Obligations owed to such Affected Lender and such Affected Lender's commitments hereunder (a "Replacement Lender"), and if such Replacement Lender agrees to such purchase, such Affected Lender shall assign to the Replacement Lender its Obligations and commitments, and upon such purchase by the Replacement Lender, which such Replacement Lender shall be deemed to be "Issuing Bank" or a "Lender" (as the case may be) for purposes of this Agreement and such Affected Lender shall cease to be "Issuing Bank" or a "Lender" (as the case may be) for purposes of this Agreement.
- (c) Notwithstanding anything herein to the contrary, the protection of Sections 2.11(m), 2.12(m), 2.13(d), and 2.14 shall be available to each Issuing Bank and each Lender (as applicable) regardless of any possible contention of the invalidity or inapplicability of the law, rule, regulation, judicial ruling, judgment, guideline, treaty or other change or condition which shall have occurred or been imposed, so long as it shall be customary for issuing banks or lenders affected thereby to comply therewith. Notwithstanding any other provision herein, neither any Issuing Bank nor any Lender shall demand compensation pursuant to this Section 2.14 if it shall not at the time be the general policy or practice of such Issuing Bank or such Lender (as the case may be) to demand such compensation in similar circumstances under comparable provisions of other credit agreements, if any.

2.15 **Joint and Several Liability.**

- (a) Each U.S. Borrower is accepting joint and several liability for the U.S. Obligations hereunder and under the other Loan Documents in consideration of the financial accommodations to be provided by the Lender Group under this Agreement, for the mutual benefit, directly and indirectly, of each U.S. Borrower and, with respect to Letters of Credit, their Subsidiaries, and in consideration of the undertakings of the other U.S. Borrowers to accept joint and several liability for the U.S. Obligations.
- (b) Each U.S. Borrower, jointly and severally, hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the other U.S. Borrowers, with respect to the payment and performance of all of the U.S. Obligations (including any U.S. Obligations arising under this Section 2.15), it being the intention of the parties hereto that all the U.S. Obligations shall be the joint and several obligations of each U.S. Borrower without preferences or distinction among them.
- (c) If and to the extent that any U.S. Borrower shall fail to make any payment with respect to any of the U.S. Obligations as and when due or to perform any of the U.S. Obligations in accordance with the terms thereof, then in each such event the other U.S. Borrowers will make such payment with respect to, or perform, such U.S. Obligation until such time as all of the U.S. Obligations are paid in full.
- (d) The U.S. Obligations of each U.S. Borrower under the provisions of this <u>Section 2.15</u> constitute the absolute and unconditional, full recourse U.S. Obligations of each U.S. Borrower enforceable against each U.S.

Borrower to the full extent of its properties and assets, irrespective of the validity, regularity or enforceability of the provisions of this Agreement (other than this Section 2.15(d)) or any other circumstances whatsoever.

- (e) Each Borrower is accepting joint and several liability for the Australian Obligations hereunder and under the other Loan Documents in consideration of the financial accommodations to be provided by the Lender Group under this Agreement, for the mutual benefit, directly and indirectly, of each Borrower and, with respect to Letters of Credit, their Subsidiaries, and in consideration of the undertakings of the other Borrowers to accept joint and several liability for the Australian Obligations.
- (f) Each Borrower, jointly and severally, hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the other Borrowers, with respect to the payment and performance of all of the Australian Obligations (including any Australian Obligations arising under this Section 2.15), it being the intention of the parties hereto that all the Australian Obligations shall be the joint and several obligations of each Borrower without preferences or distinction among them.
- (g) If and to the extent that any Borrower shall fail to make any payment with respect to any of the Australian Obligations as and when due or to perform any of the Australian Obligations in accordance with the terms thereof, then in each such event the other Borrowers will make such payment with respect to, or perform, such Australian Obligation until such time as all of the Australian Obligations are paid in full.
- (h) The Australian Obligations of each Borrower under the provisions of this <u>Section 2.15</u> constitute the absolute and unconditional, full recourse Australian Obligations of each Borrower enforceable against each Borrower to the full extent of its properties and assets, irrespective of the validity, regularity or enforceability of the provisions of this Agreement (other than this Section 2.15(h)) or any other circumstances whatsoever.
- Except as otherwise expressly provided in this Agreement, each Borrower hereby waives notice of acceptance of its joint and several liability with respect to a U.S. Borrower, the other Borrowers, and with respect to an Australian Borrower, the other Australian Borrowers, notice of any Revolving Loans or Letters of Credit issued under or pursuant to this Agreement, notice of the occurrence of any Default, Event of Default, or of any demand for any payment under this Agreement, notice of any action at any time taken or omitted by Agent or Lenders under or in respect of any of the Obligations, any requirement of diligence or to mitigate damages and, generally, to the extent permitted by applicable law, all demands, notices and other formalities of every kind in connection with this Agreement (except as otherwise provided in this Agreement). Each Borrower hereby assents to, and waives notice of, any extension or postponement of the time for the payment of any of the Obligations, the acceptance of any payment of any of the Obligations, the acceptance of any partial payment thereon, any waiver, consent or other action or acquiescence by Agent or Lenders at any time or times in respect of any default by any Borrower in the performance or satisfaction of any term, covenant, condition or provision of this Agreement, any and all other indulgences whatsoever by Agent or Lenders in respect of any of the Obligations, and the taking, addition, substitution or release, in whole or in part, at any time or times, of any security for any of the Obligations or the addition, substitution or release, in whole or in part, of any Borrower. Without limiting the generality of the foregoing, each Borrower assents to any other action or delay in acting or failure to act on the part of any Agent or Lender with respect to the failure by any Borrower to comply with any of its respective Obligations, including, without limitation, any failure strictly or diligently to assert any right or to pursue any remedy or to comply fully with applicable laws or regulations thereunder, which might, but for the provisions of this Section 2.15 afford grounds for terminating, discharging or relieving any Borrower, in whole or in part, from any of its Obligations under this Section 2.15, it being the intention of each Borrower that, so long as any of the Obligations hereunder remain unsatisfied, the Obligations of each Borrower under this Section 2.15 shall not be discharged except by performance and then only to the extent of such performance. The Obligations of each Borrower under this Section 2.15 shall not be diminished or rendered unenforceable by any winding up, reorganization, arrangement, liquidation, reconstruction or similar proceeding with respect to any other Borrower or any Agent or Lender.
- (j) Each Borrower represents and warrants to Agent and Lenders that such Borrower is currently informed of the financial condition of Borrowers and of all other circumstances which a diligent inquiry would reveal and which bear upon the risk of nonpayment of the Obligations. Each Borrower further represents and warrants to Agent and Lenders that such Borrower has read and understands the terms and conditions of the Loan Documents. Each Borrower hereby covenants that such Borrower will continue to keep informed of U.S. Borrowers' financial condition and of all other circumstances which bear upon the risk of nonpayment or nonperformance of the U.S. Obligations.

- (k) The provisions of this <u>Section 2.15</u> are made for the benefit of Agent, each member of the Lender Group, each Bank Product Provider, and their respective successors and assigns, and may be enforced by it or them from time to time against any or all Borrowers (to the extent provided in this <u>Section 2.15</u>) as often as occasion therefor may arise and without requirement on the part of Agent, any member of the Lender Group, any Bank Product Provider, or any of their successors or assigns first to marshal any of its or their claims or to exercise any of its or their rights against any Borrower or to exhaust any remedies available to it or them against any Borrower or to resort to any other source or means of obtaining payment of any of the Obligations hereunder or to elect any other remedy. The provisions of this <u>Section 2.15</u> shall remain in effect until all of the Obligations shall have been paid in full or otherwise fully agent or any Lender upon the insolvency, bankruptcy or reorganization of any Borrower, or otherwise, the provisions of this Section will forthwith be reinstated in effect, as though such payment had not been made.
- (I) Each Borrower hereby agrees that it will not enforce any of its rights of contribution or subrogation against any other Borrower with respect to any liability incurred by it hereunder or under any of the other Loan Documents, any payments made by it to Agent or Lenders with respect to any of the Obligations or any collateral security therefor until such time as all of the Obligations have been paid in full in cash. Any claim which any Borrower may have against any other Borrower with respect to any payments to any Agent or any member of the Lender Group hereunder or under any of the Bank Product Agreements are hereby expressly made subordinate and junior in right of payment, without limitation as to any increases in the Obligations arising hereunder or thereunder, to the prior payment in full in cash of the Obligations and, in the event of any insolvency, bankruptcy, receivership, liquidation, reorganization or other similar proceeding under the laws of any jurisdiction relating to any Borrower, its debts or its assets, whether voluntary or involuntary, all such Obligations shall be paid in full in cash before any payment or distribution of any character, whether in cash, securities or other property, shall be made to any other Borrower therefor.
- (m) Each U.S. Borrower hereby agrees that after the occurrence and during the continuance of any Event of Default, upon notice from the Agent, such U.S. Borrower will not demand, sue for or otherwise attempt to collect any indebtedness of any other Borrower owing to such U.S. Borrower until the Obligations shall have been paid in full in cash. If, notwithstanding the foregoing sentence, such U.S. Borrower shall collect, enforce or receive any amounts in respect of such indebtedness, such amounts shall be collected, enforced and received by such U.S. Borrower as trustee for Agent, and such U.S. Borrower shall deliver any such amounts to Agent for application to the Obligations in accordance with Section 2.4(b).
- (n) Each Australian Borrower hereby agrees that after the occurrence and during the continuance of any Event of Default, upon notice from the Agent such Australian Borrower will not demand, sue for or otherwise attempt to collect any indebtedness of any other Australian Borrower owing to such Australian Borrower until the Australian Obligations shall have been paid in full in cash. If, notwithstanding the foregoing sentence, such Australian Borrower shall collect, enforce or receive any amounts in respect of such indebtedness, such amounts shall be collected, enforced and received by such Australian Borrower as trustee for Agent, and such Australian Borrower shall deliver any such amounts to Agent for application to the Australian Obligations in accordance with Section 2.4(b)
- (o) **Separate and Distinct Australian Obligations**. For the avoidance of doubt, the parties hereto acknowledge and agree that, notwithstanding anything to the contrary in this Agreement or any of the other Loan Documents, and notwithstanding that the U.S. Loan Parties each are jointly and severally liable with the Australian Loan Parties for the Australian Obligations, the Obligations of the Australian Loan Parties under this Agreement or any of the other Loan Documents shall be separate and distinct from the Obligations of any U.S. Loan Party including, without limitation, the Parent, and shall be expressly limited to the Australian Obligations. In furtherance of the foregoing, each of the parties hereto acknowledges and agrees that the liability of any Australian Loan Party for the payment and performance of its covenants, representations and warranties set forth in this Agreement and the other Loan Documents shall be several from but not joint with the Obligations of the Parent and any other U.S. Loan Party; the Australian Loan Parties shall not guarantee the U.S. Obligations; and the Collateral of the Australian Loan Parties shall not secure or be applied in satisfaction, by way of payment, prepayment, or otherwise, of all or any portion of the U.S. Obligations of the Parent or any other U.S. Loan Party.

2.16 Incremental Borrowings.

- (a) The Borrowers may at any time or from time to time after the Closing Date, by notice from the Parent to the Agent (whereupon the Agent shall promptly deliver a copy to each of the Lenders), request one or more incremental Revolver Commitments (each an "Incremental Commitment" and all of them, collectively, the "Incremental Commitments" and any such loans thereunder, the "Incremental Loans"). Each tranche of Incremental Commitments shall be in an aggregate principal amount that is not less than \$10,000,000; provided that such amount may be less than \$10,000,000 if such amount represents all remaining availability under the limit set forth in the next sentence; provided, further that the allocation of any portion of such increase to the Australian Revolver Commitments shall be determined by the Agent in consultation with the Borrowers, with the approval of each Lender and Additional Lender (if any) agreeing to such increase; provided, further that no allocation of any Incremental Commitment to the Australian Revolver Commitments shall be permitted to the extent the Australian Revolver Commitments would exceed the U.S. Revolver Commitments after giving effect to such Incremental Commitment. Any such increase in Revolver Commitments may increase the U.S. Letter of Credit Sublimit or the Australian Letter of Credit Sublimit subject to the consent of the Agent and the applicable Issuing Bank; provided that any such increase in the U.S. Letter of Credit Sublimit or the Australian Letter of Credit Sublimit shall be provided by an Issuing Bank reasonably acceptable to the Agent and the Parent and the Issuing Bank at the time shall have no obligation to provide such increase. Notwithstanding anything to the contrary herein, the aggregate principal amount of the Incremental Commitments shall not exceed \$200,000,000. The Incremental Loans (i) shall, in the case of an Incremental Loan to the U.S. Borrowers, rank pari passu in right of payment and of security with the U.S. Loans and, in the case of an Incremental Loan to Australian Borrowers, rank pari passu in right of payment and of security with the Australian Loans and (ii) shall be implemented by way of increase of the Revolver Commitments and, except as to arrangement, underwriting or similar fees, shall be on terms identical, including the Applicable Margin and any other pricing matter related to the Revolver Commitments; provided that the OID or up-front fees (if any) applicable to any Incremental Loans will be determined by the Borrowers and the Lenders and/or Additional Lenders providing such Incremental Commitments and Incremental Loans. As a condition precedent to such an increase, (i) no Default or Event of Default shall exist on the date of the effectiveness of any Incremental Amendment, (ii) the representations and warranties contained in the Loan Documents shall be accurate in all material respects before and after the effectiveness of any Incremental Amendment referred to below; provided that, to the extent that such representations and warranties specifically refer to an earlier date, they shall be true and correct in all material respects as of such earlier date; provided, further, that, any representation and warranty that is qualified as to "materiality", "Material Adverse Effect" or similar language shall be true and correct (after giving effect to any qualification therein) in all respects on such respective dates, (iii) all fees and expenses owing in respect of any such Incremental Amendment to the Agent and the Lenders and/or Additional Lenders providing the Incremental Commitments thereunder shall have been paid and (iv) the Borrowers shall have delivered all customary agreements, certificates, opinions and other customary documents reasonably requested by the Agent.
- (b) Each notice from the Parent pursuant to this Section 2.16 shall set forth the requested amount and proposed terms of the relevant Incremental Commitments and Incremental Commitments and Incremental Loans may be made by any existing Lender (it being understood that no existing Lender will have an obligation to make a portion of any Incremental Commitment or Incremental Loan) or by any Additional Lender that is an Eligible Transferee reasonably acceptable to the Agent and each Issuing Bank (each such consent not to be unreasonably withheld, delayed or conditioned). Incremental Commitments shall become Revolver Commitments under this Agreement pursuant to an amendment (an "Incremental Amendment") to this Agreement and, as appropriate, the other Loan Documents, executed by the Borrowers, each Lender agreeing to provide such Incremental Commitment, if any, each Additional Lender, if any, the Agent and, if applicable, the Issuing Bank. The Incremental Amendment may, without the consent of any other Lenders, effect such amendments to this Agreement and the other Loan Documents as may be necessary or appropriate, in the reasonable opinion of the Agent and the Parent, to effect the provisions of this Section 2.16. The effectiveness of (and, in the case of any Incremental Amendment for an Incremental Loan, the Borrowers shall use the proceeds of the Incremental Loans for any purpose not prohibited by this Agreement.
- (c) Adjustment of Revolving Loans. Each Revolving Lender that is acquiring an Incremental Commitment on the effective date of any Incremental Amendment shall (i) make a Revolving Loan, the proceeds of which will be used to prepay the Revolving Loans of the other Revolving Lenders immediately prior to such effective date and (ii) automatically and without further act be deemed to have assumed a portion of the other Revolving Lenders' participations hereunder in outstanding Letters of Credit and Swing Loan reimbursement obligations, so that, after giving effect thereto, the Revolving Loans and Letter of Credit and Swing Loan reimbursement obligations

outstanding are held by the Revolving Lenders pro rata based on their Revolver Commitments after giving effect to such Incremental Amendment. If there is a new borrowing of Revolving Loans on the effective date of any Incremental Amendment, the Revolving Lenders after giving effect to such Incremental Amendment shall make such Revolving Loans in accordance with <u>Section 2.1</u>.

This <u>Section 2.16</u> shall supersede any provisions in <u>Section 14.1</u> to the contrary.

3	CONDITIONS:	TERM OF	AGREEMENT.

- 3.1 <u>Conditions Precedent to the Effectiveness of this Agreemen</u>t. The obligation of each Lender to make the initial extensions of credit provided for under this Agreement is subject to the fulfillment, to the satisfaction (or waiver) of Agent and each Lender, of each of the conditions precedent set forth on Schedule 3.1 (the delivery of a Lender's signature page to this Agreement being conclusively deemed to be its satisfaction or waiver of the conditions precedent).
- 3.2 Consequences of Closing Date. Upon the satisfaction of the conditions referenced in Section 3.1, (i) the Existing Syndicated Facility Agreement shall be automatically amended and restated to read in its entirety as set forth in this Agreement (with the Existing Letters of Credit being ratified and continued); provided that the rights and obligations of the parties hereto with respect to periods prior to the Closing Date shall be governed by the Existing Syndicated Facility Agreement and that except as amended and restated by the parties on or after the Closing Date, the Existing Syndicated Facility Agreement continues to operate and shall remain in full force and effect in accordance with its terms, (ii) all Liens securing obligations under the Existing Syndicated Facility Agreement shall be automatically continued and shall secure the Obligations under this Agreement (except that all Mortgages (as defined in the Existing Syndicated Facility Agreement) on Real Property, but not as-extracted collateral filings, securing the Obligations under (and as defined in) the Existing Syndicated Facility Agreement shall be released as security for the Obligations automatically upon the Closing Date) and (iii) each Lender agrees that the "Revolver Commitments" under (and as defined in) the Existing Syndicated Facility Agreement shall be reallocated among the Lenders (including as to risk participations in respect of Existing Letters of Credit and Swing Loans) such that the Revolver Commitments of the Lenders under this Agreement are as set forth on Schedule C-1 hereto and Agent may make such adjustments as it deems necessary in order to effectuate such reallocation among the Lenders. Borrowers hereby agree to compensate each Departing Lender for any and all losses, costs, and expenses incurred by such Lender in connection with any sale or assignment of LIBOR Rate Loans necessary to effect the reallocation heretofore described on terms and in the manner set forth in Section 2.13(b) of the Existing Syndicated Facility Agreement. Upon the Closing Date, automatically and without further action by any party hereto, (i) the Revolver Commitment of any Departing Lender shall be terminated, (ii) each Departing Lender will cease to be a Lender party to this Agreement and (iii) all outstanding Loans and accrued interest, fees and other amounts payable under the Existing Syndicated Facility Agreement for the account of such Departing Lender shall be due and payable on the Closing Date. Nothing contained in this Agreement or any other Loan Document shall constitute or be construed as a novation or displacement of any of the Obligations under the Existing Syndicated Facility Agreement. The Lenders that are parties to the Existing Syndicated Facility Agreement, comprising the "Required Lenders" under (and as defined in) the Existing Syndicated Facility Agreement hereby waive any requirement of prior notice of termination of the Revolver Commitments (as defined in the Existing Syndicated Facility Agreement) and of prepayment of loans thereunder, to the extent necessarv.
- 3.3 <u>Conditions Precedent to all Extensions of Credit</u>. The obligation of the Lender Group (or any member thereof) to make any Revolving Loans hereunder (or to extend any other credit hereunder) at any time shall be subject to the following conditions precedent:
- (a) the representations and warranties of Parent and its Subsidiaries contained in this Agreement or in the other Loan Documents shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) on and as of the date of such extension of credit, as though made on and as of such date (except to

the extent that such representations and warranties relate solely to an earlier date, in which case such representations and warranties shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of such earlier date);

- (b) no Default or Event of Default shall have occurred and be continuing on the date of such extension of credit, nor shall either immediately result from the making thereof; and
- (c) after giving effect to any extension of credit, (i) the U.S. Revolver Usage shall not exceed the U.S. Line Cap and (ii) the Australian Revolver Usage shall not exceed the Australian Line Cap.

Each request for an extension of credit submitted by a Borrower shall be deemed to be a representation and warranty that the conditions specified in this <u>Section 3.3</u> have been satisfied on and as of the date of the applicable credit extension.

3.4 Maturity. This Agreement shall continue in full force and effect for a term ending on the Maturity Date.

3.5 **Effect of Maturity.**

- (a) On the Maturity Date, all commitments of the Lender Group to provide additional credit hereunder shall automatically be terminated and all of the Obligations immediately shall become due and payable without notice or demand and (i) U.S. Borrowers shall be required to repay all of the U.S. Obligations in full, and (ii) Australian Borrowers shall be required to repay all of the Australian Obligations in full.
- (b) No termination of the obligations of any member of the Lender Group (other than payment in full of the U.S. Obligations and termination of the U.S. Revolver Commitments) shall relieve or discharge any U.S. Loan Party of its duties, obligations, or covenants hereunder or under any other Loan Document. Except as otherwise set forth herein or in the other Loan Documents, Agent's Liens in the Collateral securing the U.S. Obligations shall continue to secure the U.S. Obligations and shall remain in effect until all U.S. Obligations have been paid in full and the U.S. Revolver Commitments have been terminated, at which time Agent will, at Borrowers' sole expense, execute and deliver any termination statements, lien releases, discharges of security interests, and other similar discharge or release documents (and, if applicable, in recordable form) as are reasonably necessary to release, as of record, Agent's or such Lender's, as applicable, Liens in the Collateral securing the U.S. Obligations.
- (c) No termination of the obligations of any member of the Lender Group (other than payment in full of the Australian Obligations and termination of the Australian Revolver Commitments) shall relieve or discharge any Australian Loan Party of its duties, obligations, or covenants hereunder or under any other Loan Document. Except as otherwise set forth herein or in the other Loan Documents, Agent's or the Australian Security Trustee's Liens in the Collateral securing the Australian Obligations shall continue to secure the Australian Obligations and shall remain in effect until all Australian Obligations have been paid in full and the Australian Revolver Commitments have been terminated, at which time Agent will, at Borrowers' sole expense, execute and deliver any termination statements, lien releases, discharges of security interests, and other similar discharge or release documents (and, if applicable, in recordable form) as are reasonably necessary to release, as of record, Agent's or the Australian Security Trustee's Liens in the Collateral securing the Australian Obligations.
- 3.6 <u>Early Termination by Borrowers</u>. Borrowers have the option, at any time upon five (5) Business Days prior written notice to Agent, to terminate this Agreement and terminate the Revolver Commitments hereunder by repaying to Agent all of the Obligations in full. The foregoing notwithstanding, (a) Borrowers may rescind termination notices relative to proposed payments in full of the Obligations with the proceeds of third party Indebtedness or other transactions if the closing for such issuance or incurrence or other transaction does not happen on or before the date of the proposed termination (in which case, a new notice shall be required to be sent in connection with any subsequent termination), and (b) Borrowers may extend the date of termination at any time with the consent of Agent (which consent shall not be unreasonably withheld or delayed).

3.7 <u>Conditions Subsequent</u>. The obligation of the Lender Group (or any member thereof) to continue to make Revolving Loans (or otherwise extend credit hereunder) is subject to the fulfillment, on or before the date applicable thereto (unless such date is extended, in writing (including via electronic transmission), by Agent, which Agent may do without obtaining the consent of the other members of the Lender Group), of the conditions subsequent set forth on Schedule 3.7 (the failure by Borrowers to so perform or cause to be performed such conditions subsequent as and when required by the terms thereof, shall constitute an Event of Default).

4 REPRESENTATIONS AND WARRANTIES.

In order to induce the Lender Group to enter into this Agreement and to make credit extensions from time to time, each Borrower makes the following representations and warranties to the Lender Group and such representations and warranties shall survive the execution and delivery of this Agreement:

4.1 Due Organization and Qualification; Subsidiaries.

- (a) Each Loan Party (i) is duly organized or incorporated and existing and in good standing (where applicable) under the laws of the jurisdiction of its organization or incorporation, (ii) is qualified to do business in any state where the failure to be so qualified would reasonably be expected to result in a Material Adverse Effect, and (iii) has all requisite corporate or other organizational power and authority to own and operate its properties, to carry on its business as now conducted and as proposed to be conducted, to enter into the Loan Documents to which it is a party and to carry out the transactions contemplated thereby.
- (b) Set forth on Schedule 4.1(b) is a complete and accurate description, as of the Closing Date, of the authorized Equity Interests of each Borrower (other than Parent), by class, and, as of the Closing Date, a description of the number of shares of each such class that are issued and outstanding. No Borrower (other than Parent) is subject to any obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any shares of its Equity Interests or any security convertible into or exchangeable for any of its Equity Interests.
- (c) Set forth on Schedule 4.1(c), is, as of the Closing Date, a complete and accurate list of the Loan Parties' direct and indirect Subsidiaries, showing: (i) in the case of direct subsidiaries, the number of shares of each class of common and preferred Equity Interests authorized for each of such Subsidiaries, and (ii) the percentage of the outstanding shares of each such class owned directly or indirectly by Parent. All of the outstanding Equity Interests of each such Subsidiary has been validly issued and is fully paid and non-assessable (to the extent such concept is applicable).
- (d) Except as set forth on <u>Schedule 4.1(d)</u>, there are no subscriptions, options, warrants, or calls relating to any shares of any Subsidiaries' Equity Interests, including any right of conversion or exchange under any outstanding security or other instrument.

4.2 **Due Authorization; No Conflict.**

- (a) As to each Loan Party, the execution, delivery, and performance by such Loan Party of the Loan Documents to which it is a party have been duly authorized by all necessary corporate or organizational action on the part of such Loan Party.
- (b) As to each Loan Party, the execution, delivery, and performance by such Loan Party of the Loan Documents to which it is a party do not and will not (i) violate any material provision of federal, state, or local law or regulation applicable to any Loan Party or its Subsidiaries, the Governing Documents of any Loan Party or its Subsidiaries, or any order, judgment, or decree of any court or other Governmental Authority binding on any Loan Party or its Subsidiaries, (ii) conflict with, result in a breach of, or constitute (with due notice or lapse of time or both) a default under any Material Contract of any Loan Party or its Subsidiaries where any such conflict, breach or default

would individually or in the aggregate reasonably be expected to have a Material Adverse Effect, (iii) result in or require the creation or imposition of any Lien of any nature whatsoever upon any assets of any Loan Party, other than Permitted Liens, or (iv) require any approval of any holder of Equity Interests of a Loan Party or any approval or consent of any Person under any Material Contract of any Loan Party, other than (x) consents or approvals that have been obtained and that are still in force and effect or (y) except, in the case of a Material Contract, for consents or approvals, the failure to obtain would not individually or in the aggregate reasonably be expected to cause a Material Adverse Effect.

4.3 <u>Governmental Consents</u>. The execution, delivery, and performance by each Loan Party of the Loan Documents to which such Loan Party is a party and the consummation of the transactions contemplated by the Loan Documents do not and will not require any registration with, consent, or approval of, or notice to, or other action with or by, any Governmental Authority, except for (i) registrations, consents, approvals, notices, or other actions that have been obtained and that are still in force and effect, and (ii) filings and recordings with respect to the Collateral to be made, or otherwise delivered to Agent for filing or recordation

4.4 Binding Obligations; Perfected Liens.

- (a) Each Loan Document has been duly executed and delivered by each Loan Party that is a party thereto and is the legally valid and binding obligation of such Loan Party, enforceable against such Loan Party in accordance with its respective terms, except as enforcement may be limited by equitable principles or by bankruptcy, insolvency, receivership, administration, reorganization, moratorium, or similar laws relating to or limiting creditors' rights generally.
- (b) Subject to Section 3.7, Section 5.11, Section 5.12 and Section 5.16, Agent's and Australian Security Trustee's Liens in the Collateral (other than, for the avoidance of doubt, Excluded Property) are validly created and perfected, upon the filing of financing statements, the filing of PPSA Filings, the filing of as-extracted filings, the recordation of the intellectual property security agreements and the execution of Control Agreements, in each case, in the appropriate filing offices (and any necessary stamping of the Australian Security Documents for New South Wales stamp duty purposes), and, in the case of ABL Collateral, first priority Liens, subject only to Permitted Liens.
- 4.5 <u>Title to Assets; No Encumbrances</u>. Each of the Loan Parties and its Subsidiaries has (a) good, sufficient and legal title to (in the case of fee interests in Real Property), (b) valid leasehold interests in (in the case of leasehold interests in real or personal property), and (c) good and marketable title to (in the case of all other personal property), all of their respective assets reflected in their most recent financial statements delivered pursuant to <u>Section 5.1</u>, in each case except for assets disposed of since the date of such financial statements to the extent permitted hereby. All of such assets are free and clear of Liens except for Permitted Liens.
- 4.6 <u>Litigation</u>. Except for the Arbitration Award, there are no actions, suits, or proceedings pending or, to the knowledge of any Borrower, threatened in writing against a Loan Party or any of its Subsidiaries that either individually or in the aggregate would reasonably be expected to result in a Material Adverse Effect.
- 4.7 <u>Compliance with Laws</u>. No Loan Party nor any of its Subsidiaries (a) is in violation of any applicable laws, rules, regulations, executive orders, or codes (including Environmental Laws) that, individually or in the aggregate, would reasonably be expected to result in a Material Adverse Effect, or (b) is subject to or in default with respect to any final judgments, writs, injunctions, decrees, rules or regulations of any Governmental Authority, that, individually or in the aggregate, would reasonably be expected to result in a Material Adverse Effect.

4.8 No Material Adverse Effect. The audited financial statements relating to the Loan Parties and their Subsidiaries as of December 31, 2015, December 31, 2016 and December 31, 2017 that have been delivered by Borrowers to Agent have been prepared in all material respects in accordance with GAAP (except, in the case of unaudited financial statements, for the lack of footnotes and being subject to year-end audit adjustments) and present fairly in all material respects, the Loan Parties' and their Subsidiaries' consolidated financial condition as of the date thereof and results of operations for the period then ended. Since December 31, 2017, no event, circumstance, or change has occurred that has or would reasonably be expected to result in a Material Adverse Effect with respect to the Loan Parties and their Subsidiaries.

4.9 Solvency.

- (a) Each Borrower, individually, is Solvent and the Parent and its Subsidiaries, taken as a whole, are Solvent.
- (b) No transfer of property is being made by any Loan Party and no obligation is being incurred by any Loan Party in connection with the transactions contemplated by this Agreement or the other Loan Documents with the intent to hinder, delay, or defraud either present or future creditors of such Loan Party.

4.10 Employee Benefits.

- (a) Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect: (i) each Loan Party, each of its Subsidiaries and each of their ERISA Affiliates has complied with ERISA, the IRC and all applicable laws regarding each Employee Benefit Plan; (ii) each Employee Benefit Plan is, and has been, maintained in substantial compliance with ERISA, the IRC, all applicable laws and the terms of each such Employee Benefit Plan; (iii) no liability to the PBGC (other than for the payment of current premiums which are not past due) by any Loan Party or its Subsidiaries or the ERISA Affiliates has been incurred or is reasonably expected by any Loan Party or its Subsidiaries or the ERISA Affiliates to be incurred with respect to any Pension Plan; (iv) no Notification Event exists or has occurred in the past six (6) years; and (v) there exists no Unfunded Pension Liability with respect to any Pension Plans.
- (b) With respect to any scheme or arrangement mandated by a government other than the United States and with respect to each employee benefit plan maintained or contributed to by any Loan Party that is not subject to United States laws (such schemes, arrangements and employee benefit plans, collectively, "Foreign Plans"), none of the following events or conditions exists and is continuing that, either individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect: (i) non-compliance with its terms and with the requirements of any and all applicable laws, statutes, rules, regulations and orders, (ii) failure to be maintained, where required, in good standing with applicable regulatory authorities, (iii) non-compliance with any obligation of any Loan Party or its Subsidiaries in connection with the termination or partial termination of, or withdrawal from, any such Foreign Plan, (iv) any Lien on the property of any Loan Party or its Subsidiaries in favor of a Governmental Authority as a result of any action or inaction regarding such a Foreign Plan, (v) for each such Foreign Plan which is a funded or insured plan, failure to be funded or insured on an ongoing basis to the extent required by applicable non-U.S. law (using actuarial methods and assumptions which are consistent with the valuations last filed with the applicable Governmental Authorities) or (vi) any pending or threatened disputes that, to the knowledge of the Loan Party or any of its Subsidiaries, would reasonably be expected to result in liability to any Loan Party or any Subsidiaries.
- (c) Each Borrower represents and warrants as of the Closing Date that such Borrower is not and will not be using Plan Assets of one or more Plans in connection with the Loans, the Letters of Credit or the Revolver Commitments.
- 4 .11 <u>Environmental Condition</u>. Except as set forth on Schedule 4.10(a) or for any matters that, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect, (a) to each Borrower's knowledge, no Loan Party's nor any of its Subsidiaries' properties or assets has ever been used by a Loan Party, its Subsidiaries, or by previous owners or operators in the disposal of, or to produce, store, handle, treat, release, or

transport, any Hazardous Materials, where such disposal, production, storage, handling, treatment, release or transport was in violation of any applicable Environmental Law, (b) to each Borrower's knowledge, no Loan Party's nor any of its Subsidiaries' properties or assets have ever been listed on the National Priorities List, CERCLIS or any similar state or local list of Hazardous Materials disposal sites pursuant to any Environmental Law, and (c) no Loan Party nor any of its Subsidiaries nor any of their respective facilities or operations is subject to any Environmental Liability or to any outstanding written order, consent decree, negotiated agreements or settlement agreement with any Person relating to any violation of Environmental Law or Environmental Liability.

- 4.12 Complete Disclosure. All factual information taken as a whole (other than forward-looking information and projections and information of a general economic nature and general information about Borrowers' industry) furnished by or on behalf of a Loan Party or its Subsidiaries in writing to Agent or any Lender (including all information contained in the Schedules hereto or in the other Loan Documents) for purposes of or in connection with this Agreement or the other Loan Documents, and all other such factual information taken as a whole (other than forward-looking information and projections and information of a general economic nature and general information about Borrowers' industry) hereafter furnished by or on behalf of a Loan Party or its Subsidiaries in writing to Agent or any Lender will be, true and accurate, in all material respects, on the date as of which such information is dated or certified and not incomplete by omitting to state any material fact necessary to make such information (taken as a whole) not misleading in any material respect at such time in light of the circumstances under which such information was provided. The Projections delivered to Agent on February 13, 2018 represent, and as of the date on which any other Projections are delivered to Agent, such additional Projections, were prepared in good faith on the basis of the assumptions stated therein, which assumptions were believed to be reasonable by the Parent at the time made and at the time so furnished (it being understood that such Projections are subject to significant uncertainties and contingencies, many of which are beyond the control of the Loan Parties and their Subsidiaries, and no assurances can be given that such Projections will be realized, and although reflecting Borrowers' good faith estimate, projections or forecasts based on methods and assumptions which Borrowers believed to be reasonable at the time such Projections were prepared, are not to be viewed as facts, and that actual results during the period or pe
- 4.13 <u>Sanctions, PATRIOT Act, and FCPA</u>. To the extent applicable, each Loan Party and each Subsidiary of a Loan Party is in compliance, in all material respects, with the (a) Trading with the Enemy Act, as amended, the International Emergency Economic Powers Act, as amended, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) and any other enabling legislation or executive order relating thereto, (b) Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA PATRIOT Act of 2001) (the "Patriot Act"), and (c) anti-corruption laws applicable to such Loan Party or such Subsidiary, including the United States Foreign Corrupt Practices Act of 1977, as amended ("FCPA").

4.14 [Reserved.]

4.15 Payment of Taxes. Except as otherwise permitted under Section 5.5, all material tax returns and reports of each Loan Party and its Subsidiaries required to be filed by any of them have been timely filed, and all material taxes due and payable and all material assessments, fees and other governmental charges upon a Loan Party and its Subsidiaries and upon their respective assets, income, businesses and franchises that are due and payable have been paid when due and payable, except to the extent such taxes or assessments are being contested by such Loan Party or such Subsidiary diligently, in good faith, and by appropriate proceedings, and provided adequate provisions in accordance with GAAP has been made therefor. No Loan Party knows of any proposed material tax assessment against a Loan Party or any of its Subsidiaries.

- 4.16 <u>Margin Stock</u>. No Loan Party or any of its Subsidiaries is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying any Margin Stock. No part of the proceeds of the loans made to Borrowers will be used to purchase or carry any Margin Stock or to extend credit to others for the purpose of purchasing or carrying any Margin Stock or for any purpose that violates the provisions of Regulation T, U or X of the Board of Governors.
- 4.17 <u>Governmental Regulation</u>. No Loan Party or any of its Subsidiaries is subject to regulation under the Federal Power Act or the Investment Company Act of 1940 or under any other federal or state statute or regulation which may limit its ability to incur Indebtedness or which may otherwise render all or any portion of the Obligations unenforceable. No Loan Party or any of its Subsidiaries is a "registered investment company" or a company "controlled" by a "registered investment company" or a "principal underwriter" of a "registered investment company" as such terms are defined in the Investment Company Act of 1940.
- 4.18 OFAC. No Loan Party or any of its Subsidiaries is in violation in any material respect of any of the country or list based economic and trade sanctions administered and enforced by OFAC. No Loan Party nor any of its Subsidiaries, nor to the knowledge of any Loan Party, any director, officer, employee, agent, or affiliate of any Loan Party or their Subsidiaries, (a) is, or is owned or controlled by Persons that are, Sanctioned Persons or Sanctioned Entities, (b) has its assets located in Sanctioned Entities, or (c) derives revenues from investments in, or transactions with Sanctioned Persons or Sanctioned Entities.
- 4.19 <u>Employee and Labor Matters</u>. There is (a) no unfair labor practice complaint pending or, to the knowledge of any Loan Party or Subsidiary, threatened against any Loan Party or its Subsidiaries before any Governmental Authority and no grievance or arbitration proceeding pending or threatened against any Loan Party or any of its Subsidiaries which arises out of or under any collective bargaining agreement, (b) no strike, labor dispute, slowdown, stoppage or similar action or grievance pending or threatened in writing against any Loan Party or its Subsidiaries or (c) to the knowledge of any Loan Party or its Subsidiaries no union representation question existing with respect to the employees of any Loan Party or its Subsidiaries and no union organizing activity taking place with respect to any of the employees of Parent or its Subsidiaries, in each case in clause (a) through (c) above, which would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. None of any Loan Party or its Subsidiaries has incurred any liability or obligation under the Worker Adjustment and Retraining Notification Act or similar state law, which remains unpaid or unsatisfied and which would reasonably be expected to have a Material Adverse Effect. The hours worked and payments made to employees of any Loan Party and its Subsidiaries have not been in violation of the Fair Labor Standards Act or any other applicable legal requirements, except to the extent such violations would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. All material payments due from any Loan Party or its Subsidiaries on account of wages and employee health and welfare insurance and other benefits have been paid or accrued as a liability on the books of Parent, except where the failure to do so would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
 - 4.20 [Reserved.]
 - 4.21 [Reserved.]
- 4.22 <u>Eligible Accounts</u>. As to each Account that is identified by Borrowers as an Eligible Account in the most recent Borrowing Base Certificate submitted to Agent, such Account is not excluded as ineligible by virtue of one or more of the excluding criteria (other than any Agent-discretionary criteria) set forth in the definition of Eligible Accounts.

Borrowing Bas Borrowing Bas	<u>Eligible Inventory and Eligible Equipment</u> . As to each item of Inventory that is identified by Borrowers as Eligible Inventory in the most recent e Certificate submitted to Agent and as to each item of Equipment that is identified by Borrowers as Eligible Equipment in the most recent e Certificate submitted to Agent, such Inventory or Equipment (as the case may be) is not excluded as ineligible by virtue of one or more of the ria (other than any Agent-discretionary criteria) set forth in the definition of Eligible Inventory or Eligible Equipment (as the case may be).
4.24	[Reserved.]
	<u>Inventory and Equipment Records</u> . Each Loan Party keeps correct and accurate records itemizing and describing the type, quality and nventory and Equipment and the book value thereof, in each case, consistent with past practice, except with respect to Inventory of the U.S. Iron which shall be determined on a "first-in, first-out" basis.
4.26 Income Tax As	<u>Tax Consolidation</u> . Each Australian Loan Party is a member of a Tax Consolidated Group for which the Head Company (as defined in the sessment Act 1997 (Cth) of Australia) is Cliffs Natural Resources Holdings Pty Ltd.
4.27 Loan Party's co	Commercial Benefit. In relation to each Australian Loan Party, the entry into each Loan Document to which it is a party is for such Australian ommercial benefit.
	No Immunity . No Australian Loan Party has any right of immunity from set-off, legal action, suit or proceeding, attachment or execution or the iny court with respect to the Collateral owned by an Australian Loan Party or its obligations under the Loan Documents.
4.29	[Reserved.]
enforceable ag	Material Contracts. Except for matters which, either individually or in the aggregate, would not reasonably be expected to result in a Material, each Material Contract (other than those that have expired at the end of their normal terms) (a) is in full force and effect and is binding upon and ainst the applicable Loan Party or its Subsidiary and, to each Borrower's knowledge, each other Person that is a party thereto in accordance with b) is not in default due to the action or inaction of the applicable Loan Party or its Subsidiary.
4.31	EEA Financial Institutions. No Loan Party is an EEA Financial Institution.
5 AFFIRMAT	IVE COVENANTS.
	52

Each Borrower covenants and agrees that, until termination of all of the Revolver Commitments and payment in full of the Obligations:

- 5.1 Financial Statements, Reports, Certificates. Borrowers (i) will deliver to Agent (and if so requested by Agent, with copies for each Lender) each of the financial statements, reports, and other items set forth on Schedule 5.1 no later than the times specified therein, (ii) agree that any such financial statements covering periods on or after March 31, 2015 shall have been prepared as though the Canadian Entities are not Subsidiaries, (iii) agree to maintain a system of accounting that enables Borrowers to produce financial statements in accordance with GAAP in all material respects, and (iv) agree that they will, and will cause each other Loan Party to, (A) keep a reporting system that shows all additions, sales, claims, returns, and allowances with respect to their and their Subsidiaries' sales, and (B) maintain their billing systems and practices substantially as in effect as of the Closing Date and shall only make material modifications thereto with notice to, and with the consent of, Agent. The requirements of this Section 5.1 may be satisfied by notice to the Agent that such documents required to be delivered pursuant to this Section 5.1 (to the extent included on Form 10-K or Form 10-Q) have been filed with the SEC.
- 5.2 Reporting. Borrowers (a) will deliver to Agent (and if so requested by Agent, with copies for each Lender) each of the reports set forth on Schedule 5.2 at the times specified therein (including weekly reporting of the Borrowing Base during an Increased Borrowing Base Reporting Period, as more fully set forth in Schedule 5.2), and (b) agree to use commercially reasonable efforts in cooperation with Agent to facilitate and implement a system of electronic collateral reporting in order to provide electronic reporting of each of the items set forth on Schedule 5.2. All calculations of Availability in any Borrowing Base Certificate shall be made by the Parent and certified by a financial officer of the Parent; provided that the Agent may from time to time review and adjust any such calculation in consultation with the Parent to the extent the calculation is not made in accordance with this Agreement or does not accurately reflect the Reserves.
- 5.3 **Existence**. Except as otherwise permitted under <u>Section 6.3</u> or <u>Section 6.4</u>, each Borrower will, and will cause each of its Subsidiaries to, at all times preserve and keep in full force and effect such Person's (a) valid existence and good standing (where applicable) in its jurisdiction of organization or incorporation except with respect to any Subsidiary that is not a Loan Party, as would not reasonably be expected to result in a Material Adverse Effect and, (b) except as would not reasonably be expected to result in a Material Adverse Effect, good standing (where applicable) with respect to all other jurisdictions in which it is qualified to do business and any rights, franchises, permits, licenses, accreditations, authorizations, or other approvals material to their businesses.
- 5.4 <u>Maintenance of Properties</u>. Except as otherwise permitted under <u>Section 6.3</u> or <u>Section 6.4</u> or the shutdown, winding down, placing on care and maintenance or other similar transaction related to a mine (including in relation to the Permitted Reorganization) each Borrower will, and will cause each of its Subsidiaries to, maintain and preserve all of its assets that are necessary in the proper conduct of its business in good working order and condition, ordinary wear, tear, casualty, and condemnation and Permitted Dispositions excepted.
- 5.5 <u>Taxes</u>. Each Loan Party will, and will cause each of its Subsidiaries to, pay in full before delinquency or before the expiration of any extension period all material governmental assessments and taxes imposed, levied, or assessed against it, or any of its assets or in respect of any of its income, businesses, or franchises, except (i) to the extent that the validity of such governmental assessment or tax is the subject of a Permitted Protest and (ii) to the extent such failure to pay would not reasonably be expected to result in a Material Adverse Effect.
- 5.6 <u>Insurance</u>. Each Borrower will, and will cause each of its Subsidiaries to, at Borrowers' expense, (a) maintain insurance respecting each Borrower and its Subsidiaries' assets wherever located, covering liabilities, losses

or damages as are customarily insured against by other Persons engaged in same or similar businesses and similarly situated and located. All such policies of insurance shall be with financially sound and reputable insurance companies and in such amounts as is carried generally in accordance with sound business practice by companies in similar businesses similarly situated and located. All property insurance policies covering the Collateral are subject to the Intercreditor Agreement, to be made payable to Agent for the benefit of Agent and the Lenders, as their interests may appear, in case of loss, pursuant to a standard loss payable endorsement with a standard noncontributory "lender" or "secured party" clause and are to contain such other provisions as Agent may reasonably require to fully protect the Lenders' interest in the Collateral and to any payments to be made under such policies. All certificates of property and general liability insurance are to be delivered to Agent, with the loss payable (but only in respect of Collateral) or additional insured, as applicable, endorsements (it being understood that Agent shall not be named an additional insured with respect to liability insurance) in favor of Agent and shall provide for not less than 30 days (10 days in the case of non-payment) prior written notice to Agent of the exercise of any right of cancellation or such other terms reasonably acceptable to the Agent in its Permitted Discretion. If any Borrower or its Subsidiaries fails to maintain such insurance, Agent may arrange for such insurance, but at Borrowers' expense and without any responsibility on Agent's part for obtaining the insurance, the solvency of the insurance companies, the adequacy of the coverage, or the collection of claims. Upon the occurrence and during the continuance of an Event of Default, subject to the Intercreditor Agreement, Agent shall have the right to elect to file claims under any property and general liability insurance policies in respect of the Collateral, to receive, r

5.7 Inspection.

- (a) Each Borrower will, and will cause each of its Subsidiaries to, permit Agent, any Lender, and each of their respective duly authorized representatives or agents to visit any of its properties and inspect any of its assets or books and records, to examine and make copies of its books and records, and to discuss its affairs, finances, and accounts with its Responsible Officers, at such reasonable times and intervals as Agent or any Lender, as applicable, may designate and, so long as no Event of Default has occurred and is continuing, with reasonable prior notice to Borrowers and during regular business hours, provided that the Loan Parties shall only be obligated to reimburse Agent for one inspection and visit in any Fiscal Year so long as no Event of Default has occurred and is continuing.
- (b) Each Borrower will, and will cause each of its Subsidiaries to, permit Agent and each of its duly authorized representatives or agents to conduct field examinations, appraisals and valuations, with expenses for such field examinations, appraisals and valuations being subject to Section 2.10(c), at such reasonable times and intervals as Agent may designate in its Permitted Discretion. So long as no Event of Default has occurred and is continuing, Agent agrees to provide Borrowers with a copy of the report for any inventory or equipment appraisal upon request by Borrowers so long as (i) such report exists, (ii) the third person employed by Agent to perform such valuation consents to such disclosure, and (iii) Borrowers execute and deliver to Agent a non-reliance letter reasonably satisfactory to Agent.
- 5.8 <u>Compliance with Laws</u>. Each Borrower will, and will cause each of its Subsidiaries to, comply with the requirements of all applicable laws, rules, regulations, and orders of any Governmental Authority, other than laws, rules, regulations, and orders the non-compliance with which, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect.
- 5.9 **Environmental**. Except as would not reasonably be expected to result in a Material Adverse Effect, each Borrower will, and will cause each of its Subsidiaries to:
 - (a) comply with Environmental Laws; and
- (b) take any Remedial Action required to abate any release of which any Borrower has knowledge of a Hazardous Material in violation of any Environmental Law from or onto property owned or operated by any Borrower

or its Subsidiaries or resulting from the business of any Borrower or any of its Subsidiaries, to the extent required by applicable Environmental Law.

5.10 [Reserved.]

5.11 Formation of Subsidiaries. Each Borrower will, at the time that any Loan Party forms any direct or indirect Subsidiary (other than any such Subsidiary that is an Excluded Subsidiary) or acquires any direct or indirect Subsidiary after the Closing Date (other than any such Subsidiary that is an Excluded Subsidiary), within 30 days of such formation or acquisition (or such later date as permitted by Agent in its sole discretion) (a) cause such new Subsidiary to provide to Agent a joinder to the Guaranty and Security Agreement and an accession deed to the Australian Security Trust Deed (where applicable), together with such other security agreements and any applicable U.S. Additional Documents (as defined below)), as well as appropriate financing statements, all in form and substance reasonably satisfactory to Agent (including being sufficient to grant Agent or Australian Security Trustee (as the case may be) a Lien (subject to Permitted Liens) in and to the assets of such newly formed or acquired Subsidiary (excluding any Excluded Property and provided further that no Subsidiary which is incorporated in Australia shall be required to grant any real property mortgage or mining mortgage)), (b) provide, or cause the applicable Loan Party to provide, to Agent or the Australian Security Trustee a pledge agreement (or an addendum to the Guaranty and Security Agreement or an Australian Security Document) and appropriate certificates and powers or financing statements, pledging all of the direct or beneficial ownership interest in such new Subsidiary to the extent not constituting Excluded Property in form and substance reasonably satisfactory to Agent, provided, that, for the avoidance of doubt, not more than 65% of the total outstanding voting Equity Interest of any first tier Subsidiary of a Loan Party that is a CFC or a FSHCO (but none of the Equity Interest of any Subsidiary of such CFC or FSHCO) shall be required to be pledged, (c) if such new Subsidiary is to be a Borrower, cause such new Subsidiary to provide a joinder to this Agreement in form and substance reasonably satisfactory to Agent, and (d) provide to Agent all other documentation, including one or more opinions of counsel reasonably satisfactory to Agent, which, in its Permitted Discretion, is appropriate with respect to the execution and delivery of the applicable documentation referred to above. Any document, agreement, or instrument executed or issued pursuant to this Section 5.11 shall constitute a Loan Document. Notwithstanding the foregoing, Section 5.12 below or anything contained herein or in any other Loan Document to the contrary, it is understood and agreed that to the extent that the Fixed Asset Priority Collateral Agent is satisfied with or agrees to any deliveries in respect of any asset or property (other than ABL Collateral), Agent shall be deemed to be satisfied with such deliveries to the extent substantially the same as those delivered to the Fixed Asset Priority Collateral Agent and the Loan Parties shall not be required to deliver any U.S. Additional Documents with respect thereto. So long as the Intercreditor Agreement is in effect, a Loan Party may satisfy its obligations hereunder and under the other Loan Documents to deliver Collateral that constitutes Fixed Asset Priority Collateral to Agent by delivering such Collateral that constitutes Fixed Asset Priority Collateral to the Fixed Asset Priority Collateral to the Fixed Asset Priority Collateral Tourist Collateral Agent or its agent, designee or bailee.

5.12 Further Assurances.

(a) <u>U.S. Further Assurances</u>. Each U.S. Borrower will, and will cause each of the other U.S. Loan Parties to, at any time upon the reasonable request of Agent, subject to the terms of the Intercreditor Agreement and <u>Section 5.11</u> and <u>Section 18</u>, execute or deliver to Agent any and all financing statements, security agreements, as-extracted collateral filings, pledges, assignments, opinions of counsel and all other documents (the "<u>U.S. Additional Documents</u>") that Agent may reasonably request in form and substance reasonably satisfactory to Agent, to create, perfect, and continue to perfect Agent's Liens in all Collateral of each U.S. Loan Party (whether now owned or hereafter arising or acquired, tangible or intangible in each case, to the extent not constituting Excluded Property) to the extent not constituting Excluded Property or interest therein (other than as-extracted collateral interests) and to the extent any Liens, mortgages or other filings have been made with respect thereto, each Lender hereby authorizes the Agent to take such actions and make such filings as necessary or advisable to release or terminate any such Lien, mortgage or other filing (it being understood that all as-extracted collateral filings will remain in place).

(b) <u>Australian Further Assurances</u>. Each Australian Borrower will, and will cause each of the other Australian Loan Parties to, at the reasonable request of the Australian Security Trustee, do anything:

(i) to ensure any Loan Document (or any security interest (as defined in the Australian PPSA) or other Lien under any Loan Document) is fully effective, enforceable and perfected with the contemplated priority;
(ii) for more satisfactorily assuring or securing to the Lender Group the property the subject of any such security interest or other Lien in a manner consistent with the Loan Documents; or
(iii) for aiding the exercise of any power in any Loan Document, the Australian Loan Party shall do it promptly at its own cost.
This may include signing documents, getting documents completed and signed and supplying information, delivering documents and evidence of title and executed blank transfers, or otherwise giving possession or control with respect to any property the subject of any security interest or Security.
5.13 <u>Lender Meetings</u> . Parent will, within 90 days after the close of each fiscal year of Parent, at the request of Agent or of the Required Lenders and upon reasonable prior notice, hold a meeting (at a mutually agreeable location and time or, at the option of Agent, by conference call) with all Lenders who choose to attend such meeting at which meeting shall be reviewed the financial results of the previous fiscal year and the financial condition of Parent and its Subsidiaries and the projections presented for the current fiscal year of Parent.
5.14 [Reserved].
5.15 Compliance with ERISA and the IRC. Each Borrower will, and will cause each of its Subsidiaries to, comply with the provisions of ERISA and the IRC applicable to employee benefit plans as defined in Section 3(3) of ERISA and the laws applicable to any Foreign Plan, except to the extent any failure to comply, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect.
5.16 Cash Management.
(a) <u>U.S. Accounts</u> .
(i) Within 90 days of the acquisition or establishment of a lockbox or U.S. Blocked Account (in each case, other than with respect to Excluded Accounts), each U.S. Loan Party shall obtain a U.S. Control Agreement from such lockbox servicer or U.S. Blocked Account bank, establishing Agent's control over and Lien in the lockbox or U.S. Blocked Account, which may only be exercised by Agent during any Cash Dominion Trigger Period

(b) Australian Accounts.

Bank of America or to a U.S. Dominion Account.

(i) Each Australian Loan Party shall maintain its ADI Accounts (other than the Australian Exempt Accounts) pursuant to lockbox or other arrangements reasonably acceptable to the Agent, it being agreed that, subject to the other obligations following the Closing Date under this <u>Section 5.16(b)</u>, the arrangements as of the Closing Date are acceptable, including:

requiring immediate deposit of all remittances received in the lockbox or U.S. Blocked Account to a U.S. Dominion Account designated by Agent, and waiving offset rights of such servicer or bank, except for customary administrative charges. If a U.S. Blocked Account is not maintained with Bank of America, Agent may, during any Cash Dominion Trigger Period, require immediate and daily transfer of all funds in such account to a U.S. Blocked Account maintained with

(A) [reserved]; and

(B) maintenance of the China Accounts.

- (ii) On and from the Closing Date, each Australian Loan Party shall maintain an Australian Concentration Account.
- (iii) Funds standing to the credit of an Australian Concentration Account may be applied by the relevant Australian Loan Party towards the Australian Obligations then outstanding or otherwise towards funding any other ADI Account of an Australian Loan Party, in each case with the prior written consent of the Australian Security Trustee.
- (iv) On and from the Closing Date, each Australian Loan Party will maintain disbursement accounts (other than the China Accounts and the Australian Exempt Account) with Bank of America, N.A. (Australia Branch).
- (v) On and from the Closing Date (or such longer period that Agent may agree in its sole discretion), no Australian Loan Party may maintain any Australian Concentration Account, ADI Account, deposit account or disbursement account other than with Bank of America, N.A. (Australia Branch) without the prior written consent of the Agent (other than the China Accounts and the Australian Exempt Accounts).
- (vi) On and from the Closing Date, no Australian Loan Party may open any new Australian Concentration Account, ADI Account, deposit account or disbursement account other than with Bank of America, N.A. (Australia Branch) without the prior written consent of the Agent (other than the China Accounts and the Australian Exempt Accounts).

(c) Proceeds of Collateral.

- (i) Each U.S. Loan Party shall request in writing and otherwise take all necessary steps to ensure that all payments on Accounts or otherwise relating to ABL Priority Collateral are made directly to a U.S. Blocked Account (or a lockbox relating to a U.S. Blocked Account). If any U.S. Loan Party receives cash or payment items with respect to any ABL Priority Collateral, it shall hold same in trust for Agent and promptly (not later than three (3) Business Days thereafter) deposit same into a U.S. Blocked Account.
- (ii) Each Australian Loan Party shall request in writing and otherwise take all necessary steps to ensure that all payments on Accounts or otherwise relating to Australian ABL Collateral are made directly to an Australian Concentration Account. If any Australian Loan Party receives cash or payment items with respect to any Australian ABL Collateral, it shall hold same in trust for Australian Security Trustee and promptly (not later than three (3) Business Days thereafter) deposit same into an Australian Concentration Account (other than any amount controlled by the Australian Security Trustee under an Australian Control Agreement).
- (d) <u>Schedule 5.16(a)</u> sets forth all ADI Accounts maintained by the Australian Loan Parties and their Subsidiaries, as of the Closing Date. Each Australian Loan Party shall take all actions necessary to establish the Australian Security Trustee's control (within the meaning of s341 of the Australian PPSA) over each such ADI Account at all times (other than the China Accounts and the Australian Exempt Accounts). Each Australian Loan Party shall be the sole account holder of each ADI Account and shall not allow any other Person (other than the Australian Security Trustee and the applicable depositary bank) to have control over such ADI Account or any deposits therein. Each Australian Loan Party shall promptly notify the Australian Security Trustee of any opening or closing of an ADI Account, and shall not open any ADI Account without the prior written consent of the Australian Security Trustee.
- (e) <u>Schedule 5.16(b)</u> sets forth all Deposit Accounts, including all U.S. Dominion Accounts maintained by the U.S. Loan Parties and their Subsidiaries, as of the Closing Date. Each U.S. Loan Party shall take all actions necessary to establish the Agent's control (within the meaning of the Code) over each such Deposit Account (other than Excluded Accounts) at all times. Each U.S. Loan Party shall be the sole account holder of each Deposit Account and shall not allow any other Person (other than the Agent and the applicable depositary bank) to have control over any such Deposit Account (other than Excluded Accounts) or any deposits therein. Each U.S. Loan Party shall promptly notify the Agent of any opening or closing of a Deposit Account (other than an Excluded Account), and shall not open any Deposit Account (other than an Excluded Account) unless such Deposit Account is a U.S. Blocked Account.

6	NFG/	TIVE	COV	FΝΔ	PLIN

Each Borrower covenants and agrees that, until termination of all of the Revolver Commitments and pay	ment in full of the Obligations:
---	----------------------------------

- 6.1 <u>Indebtedness</u>. Each Borrower will not, and will not permit any of its Subsidiaries to create, incur, assume, suffer to exist, guarantee, or otherwise become or remain, directly or indirectly, liable with respect to any Indebtedness, except for Permitted Indebtedness.
- 6.2 <u>Liens</u>. Each Borrower will not, and will not permit any of its Subsidiaries to create, incur, assume, or suffer to exist, directly or indirectly, any Lien on or with respect to any of its assets, of any kind, whether now owned or hereafter acquired, or any income or profits therefrom, except for Permitted Liens.
 - 6.3 Restrictions on Fundamental Changes. Each Borrower will not, and will not permit any of its Subsidiaries to,
- (a) other than in order to consummate a Permitted Acquisition, Permitted Investment or Permitted Disposition or a Permitted Reorganization, enter into any merger, consolidation, or amalgamation, except for (i) any merger, consolidation or amalgamation between Loan Parties, provided, that (w) no U.S. Loan Party shall merge, consolidate an amalgamate with an Australian Loan Party (and vice-versa) unless the Parent shall have delivered an updated Borrowing Base Certificate reflecting such transaction, (x) if such transaction involves a Borrower, a Borrower must be the surviving entity of any such transaction, (y) Parent must be the surviving entity of any such transaction to which it is a party and (z) in the case of any transaction involving a U.S. Loan Party must be the surviving entity of such transaction, (ii) any merger, consolidation or amalgamation among a Loan Party and a Subsidiary that is not a Loan Party so long a Loan Party is the surviving entity of any such transaction, and (iii) any merger, consolidation or amalgamation among Subsidiaries of any Borrower that are not Loan Parties; or
- (b) liquidate, wind up, or dissolve itself (or suffer any liquidation or dissolution), except for (i) the liquidation, winding up or dissolution of any Subsidiary (other than a Borrower) so long as such dissolution, winding up or liquidation, as applicable, would not reasonably be expected to have a Material Adverse Effect, (ii) the liquidation or dissolution of a Borrower (other than Parent) so long as all of the assets (including any interest in any Equity Interests) of such liquidating or dissolving Borrower are transferred to a Borrower that is not liquidating or dissolving, or (iii) a Permitted Reorganization.
- 6.4 <u>Disposal of Assets</u>. Other than Permitted Dispositions, each Borrower will not, and will not permit any of its Subsidiaries to convey, sell, lease, license, assign, transfer, or otherwise dispose of any of its or their assets.
- 6.5 <u>Nature of Business</u>. Each Borrower will not, and will not permit any of its Subsidiaries to change in any material respect the general nature of their business, taken as a whole, from the general nature of the business as of the Closing Date; provided, that the foregoing shall not prevent any Borrower and its Subsidiaries from (i) engaging in any business that is reasonably related, complementary or ancillary thereto, (ii) disposing of any business pursuant to a Permitted Disposition, or (iii) a Permitted Reorganization.
 - 6.6 Prepayments and Amendments. Each Borrower will not, and will not permit any of its Subsidiaries to:

- (a) except in connection with the Transaction or any Refinancing Indebtedness permitted by Section 6.1,
- optionally prepay, redeem, defease, purchase, or otherwise acquire any Indebtedness of any Borrower or its Subsidiaries consisting of Indebtedness permitted under clauses (f), (p), (q), (t), (u), (v) or (z) of the definition of Permitted Indebtedness prior to the maturity date applicable to such Indebtedness, except (A) any prepayment, redemption, defeasance, purchase or other acquisition with Qualified Equity Interests so long as at the time of such prepayment, redemption, defeasance, purchase or other acquisition no Default or Event of Default has occurred and is continuing or would result therefrom, (B) any prepayment, redemption, defeasance, purchase or other acquisition with the net cash proceeds of an issuance of Qualified Equity Interests within 60 days of such issuance (or such later date as agreed to by the Agent in its sole discretion)) so long as (1) at the time of such prepayment, redemption, defeasance, purchase or other acquisition no Default or Event of Default has occurred and is continuing or would result therefrom and (2) the net cash proceeds of such issuance of Qualified Equity Interests are maintained in a segregated Deposit Account subject to the "control" of the Agent until the earlier of (a) application toward such prepayment, redemption, defeasance, purchase or other acquisition and (b) the date that is 60 days after such issuance, (C) any prepayment, redemption, defeasance, purchase or other acquisition so long as (1) for each of the 30 consecutive days immediately preceding such prepayment. redemption, defeasance, purchase or other acquisition, and both before and after giving effect to such prepayment, redemption, defeasance, purchase or other acquisition, no Loans are outstanding and (2) at the time of such prepayment, redemption, defeasance, purchase or other acquisition, no Default or Event of Default has occurred and is continuing or would result therefrom; provided, further that the foregoing conditions under this clause (C) shall not be required to be satisfied with respect to prepayments, redemptions, defeasances, purchases or other acquisitions of any such Indebtedness in an aggregate principal amount (for all such prepayments, redemptions, defeasances, purchases or other acquisitions) of up to \$25,000,000 during the term of this Agreement and (D) any prepayment, redemption, defeasance, purchase or other acquisition of the Convertible Notes with Qualified Equity Interests; provided that this Section 6.6(a)(i) shall not apply to any prepayment, redemption, defeasance, purchase, or other acquisition of the Convertible Notes to the extent such event or condition occurs as a result of (x) the satisfaction of a conversion contingency pursuant to the Convertible Notes (as in effect on the date hereof) or the exercise by a holder of the Convertible Notes of a conversion right resulting from the satisfaction of a conversion contingency pursuant to the Convertible Notes (as in effect on the date hereof) (it being understood that any such prepayment, redemption, defeasance, purchase, or other acquisition of the Convertible Notes made in cash in reliance on this clause (x) shall be subject to satisfaction of the Payment Conditions at the time thereof, other than prepayments, redemptions, defeasances, purchases or other acquisitions (i) of less than \$15,000,000 in the aggregate during the term of this Agreement, or (ii) paid in lieu of fractional shares)) or (y) a required repurchase under the Convertible Notes; provided further that nothing in this Section 6.6 shall prohibit the payment of Indebtedness permitted under this Agreement at the time of the final maturity of the obligations under such Indebtedness, or
- (ii) make any payment on account of Indebtedness that has been contractually subordinated in right of payment to the Obligations if such payment is not permitted at such time under the subordination terms and conditions, or
- (b) except in connection with the Transaction or any Refinancing Indebtedness permitted by Section 6.1, directly or indirectly, amend, modify, or change any of the terms or provisions of
- (i) any agreement, instrument, document, indenture, or other writing evidencing or concerning Permitted Indebtedness permitted under clauses (f), (p), (q), (t), (u), (v) or (z) of the definition of Permitted Indebtedness if such Indebtedness could not have been incurred on such terms (without limiting clause (iii) below), or
- (ii) the Governing Documents of any Loan Party or any of its Subsidiaries, the Existing Senior Notes, the Convertible Notes or the Senior Secured Notes, in each case if the effect thereof, either individually or in the aggregate, would reasonably be expected to be materially adverse to the interests of the Lenders.
- 6.7 Restricted Payments. Each Borrower will not, and will not permit any of its Subsidiaries to make any Restricted Payment; provided, that, so long as it is permitted by law and the Governing Documents of such Borrower or its Subsidiaries,

- (a) the Borrowers and their respective Subsidiaries may make Restricted Payments to purchase, redeem or otherwise acquire or retire pursuant to a management or employee benefit plan in an aggregate amount not to exceed \$25,000,000 per fiscal year,
- (b) Parent and each Subsidiary may declare and make dividend payments or other distributions payable solely in Equity Interests (other than Disqualified Equity Interests),
- (c) (i) any Borrower may make Restricted Payments to another Borrower, (ii) any Subsidiary that is not a Borrower may make Restricted Payments to any Borrower or any Guarantor, (iii) any Subsidiary that is not a Loan Party may make Restricted Payments to any other Subsidiary that is not a Loan Party and (iv) any Borrower (other than Parent) or any Subsidiary may make any Restricted Payments to its parent entity,

(d) [Reserved],

- (e) in addition to the foregoing, Parent may make any other Restricted Payments so long as (i) the Payment Conditions are satisfied at the time declared and (ii) until such time as such Restricted Payment is made, a Reserve has been established by Agent in an amount equal to the Restricted Payment so declared, and
- (f) Parent may make Restricted Payments of the type described in clauses (b) and (c) of the definition thereof so long as (i) no Default or Event of Default has occurred and is continuing or would result therefrom and (ii) for each of the 30 consecutive days immediately preceding such Restricted Payment, and both before and after giving effect to such Restricted Payment, (A) no Loans are outstanding and (B) Liquidity is not less than \$250,000,000.
- Accounting Methods. Each Borrower will not, and will not permit any of its Subsidiaries to modify or change its fiscal year or its method of accounting (other than as may be required to conform to GAAP or, except to the extent that such modification or change would impact the calculation of the Fixed Charge Coverage Ratio or the Borrowing Base (or any component definition of any of the foregoing), such modification or change of its method of accounting is permitted by GAAP), or in the case of any Subsidiary or any Borrower (other than Parent), in order to conform the fiscal year of such Subsidiary or Borrower to the fiscal year of the Parent.
- 6.9 <u>Investments</u>. Each Borrower will not, and will not permit any of its Subsidiaries to, directly or indirectly, make or acquire any Investment except for Permitted Investments.
- 6.10 <u>Transactions with Affiliates</u>. Each Borrower will not, and will not permit any of its Subsidiaries to, directly or indirectly, enter into or permit to exist any transaction with any Affiliate of any Borrower or any of its Subsidiaries except for:
- (a) transactions between such Borrower or its Subsidiaries, on the one hand, and any Affiliate of such Borrower or its Subsidiaries, on the other hand, so long as such transactions (i) are no less favorable, taken as a whole, to such Borrower or its Subsidiaries, as applicable, than would be obtained in an arm's-length transaction with a non-Affiliate or (ii) have been approved by a majority of the disinterested members of the Parent's board of directors,
- (b) so long as it has been approved by such Borrower's or its applicable Subsidiary's board of directors (or comparable governing body) in accordance with applicable law, any indemnity provided for the benefit of directors (or comparable managers) of such Borrower or its applicable Subsidiary,
- (c) the payment of reasonable compensation, severance, or employee benefit arrangements to employees, officers, and outside directors of such Borrower and its Subsidiaries in the ordinary course of business,
 - (d) transactions permitted by <u>Section 6.1</u>, <u>Section 6.3</u>, <u>Section 6.7</u> or <u>Section 6.9</u>,

- (e) the Joint Venture Agreements,
- (f) transactions between and among U.S. Loan Parties and between and among Australian Loan Parties, and
- (g) transactions between Parent and its Subsidiaries with the Canadian Entities existing on the Closing Date or, to the extent the Payment Conditions are satisfied at the time of, and after giving pro forma effect to, such transactions, otherwise on terms determined in the business judgment of Parent.
- 6.11 <u>Use of Proceeds</u>. (a) Each Borrower will not, and will not permit any of its Subsidiaries to use the proceeds of any loan made hereunder for any purpose other than (a) on the Closing Date, to pay the fees, costs, and expenses incurred in connection with this Agreement and the other Loan Documents and (b) thereafter, to provide for working capital and general corporate purposes to the extent not prohibited by the terms hereof (other than the refinancing of any commercial paper), for their lawful and permitted purposes (including that no part of the proceeds of the loans made to Borrowers will be used to purchase or carry any such Margin Stock or to extend credit to others for the purpose of purchasing or carrying any such Margin Stock or for any purpose that violates the provisions of Regulation T, U or X of the Board of Governors).
- (b) No part of the proceeds of the loans made hereunder will be used by any Loan Party or any of their Affiliates, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the FCPA or any other applicable anti-corruption laws.
- (c) No proceeds of any loan made hereunder will be used, directly or indirectly, by any Loan Party or Subsidiary thereof to fund any operations in or with, finance any investments or activities in or with, or make any payments to, a Sanctioned Person or a Sanctioned Entity.

7 FINANCIAL COVENANT.

Each Borrower covenants and agrees that, until termination of all of the Revolver Commitments and payment in full of the Obligations, commencing on the date on which a Financial Covenant Period begins and measured as of the end of the fiscal quarter immediately preceding the date on which a Financial Covenant Period first begins and as of each fiscal quarter end thereafter during such Financial Covenant Period, the Parent and its Subsidiaries on a consolidated basis will have a Fixed Charge Coverage Ratio, measured on a quarter-end basis, of at least 1.00:1.00 for the 12-month period ending as of the end of each fiscal quarter.

8 EVENTS OF DEFAULT.

Any one or more of the following events shall constitute an event of default (each, an " Event of Default") under this Agreement:

8.1 Payments. If Borrowers (or any of them) fail to pay when due and payable, or when declared due and payable, (a) all or any portion of the Obligations consisting of interest, fees, or charges due the Lender Group, reimbursement of Lender Group Expenses, or other amounts (other than any portion thereof constituting principal) constituting Obligations (including any portion thereof that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), and such failure continues for a period of five (5) calendar days, (b) all or any portion of the principal of the Loans, or (c) any amount payable to any Issuing Bank in reimbursement of any drawing under a Letter of Credit;

8.2 Covenants. If any Loan Party or any of its Subsidiaries

- (a) fails to perform or observe any covenant or other agreement contained in any of (i) Sections 3.7, 5.1, 5.2, 5.3 (solely if any Borrower is not validly existing or in good standing (to the extent such concept is applicable) in its jurisdiction of organization or incorporation), Section 5.6, Section 5.16, Section 5.12, Section 5.13, or Section 5.16 of this Agreement, (ii) Section 6 of this Agreement or (iii) Section 7 of this Agreement;
- (b) fails to perform or observe any covenant or other agreement contained in this Agreement, or in any of the other Loan Documents, in each case, other than any such covenant or agreement that is the subject of another provision of this <u>Section 8</u> (in which event such other provision of this <u>Section 8</u> shall govern), and such failure continues for a period of 30 days after the earlier of (i) the date on which such failure shall first become known to any Responsible Officer of any Borrower or (ii) the date on which written notice thereof is given to Borrowers by Agent;
- 8.3 <u>Judgments</u>. If one or more judgments, orders, or awards for the payment of money (other than in respect of the Arbitration Award) involving an aggregate amount of \$75,000,000 or more (except to the extent covered by insurance pursuant to which the insurer has not denied coverage) is entered or filed against a Loan Party or any of its Subsidiaries, or with respect to any of their respective assets, and either (a) there is a period of 60 consecutive days at any time after the entry of any such judgment, order, or award during which (1) the same is not discharged, satisfied, vacated, or bonded pending appeal, or (2) a stay of enforcement thereof is not in effect, or (b) enforcement proceedings are commenced upon such judgment, order, or award; provided that at any time that Liquidity is less than \$250,000,000, the threshold included in this <u>Section 8.3</u> shall be \$25,000,000 (which shall apply to all judgments, orders or awards then outstanding, whether entered before or after the date Liquidity was less than \$250,000,000).
- 8.4 <u>Voluntary Bankruptcy, etc.</u> If an Insolvency Proceeding (other than a Permitted Reorganization) is commenced by a Loan Party or any of its Significant Subsidiaries or an Australian Loan Party is or is presumed or deemed (under the Corporations Act) to be unable or admits inability to pay its debts as they fall due or suspends making payments on any of its debts;
- 8.5 <u>Involuntary Bankruptcy, etc.</u> If an Insolvency Proceeding (other than a Permitted Reorganization) is commenced against a Loan Party or any of its Significant Subsidiaries and any of the following events occur: (a) such Loan Party or such Significant Subsidiary consents to the institution of such Insolvency Proceeding against it, (b) the petition commencing the Insolvency Proceeding is not dismissed within 60 calendar days of the date of the filing thereof, (d) with respect to an Australian Loan Party, an application is not made to the court disputing the petition commencing the Insolvency Proceeding within 15 Business Days of the date of the filing thereof, (e) an interim trustee is appointed to take possession of all or any substantial portion of the properties or assets of, or to operate all or any substantial portion of the business of, such Loan Party or its Significant Subsidiary, or (f) an order for relief shall have been issued or entered therein;
- 8.6 <u>Default Under Other Agreements</u>. If there is a default in one or more agreements to which a Loan Party or any of its Subsidiaries is a party with one or more third Persons relative to a Loan Party's or any of its Subsidiaries' Indebtedness (other than any letter of credit fully secured by cash or Cash Equivalents) involving an aggregate amount of \$75,000,000 or more, and such default (a) occurs at the final maturity of the obligations thereunder, or (b) results in a right by such third Person, irrespective of whether exercised, to accelerate the maturity of such Loan Party's or its Subsidiary's obligations thereunder; provided that at any time that Liquidity is less than \$250,000,000, the threshold included in this <u>Section 8.6</u> shall be \$25,000,000 (which shall apply to any and all such defaults under all such agreements, whether entered before or after the date Liquidity was less than \$250,000,000);
- 8.7 Representations, etc. If any warranty, representation, certificate, statement, or Record made herein or in any other Loan Document or delivered in writing to Agent or any Lender in connection with this Agreement or any

other Loan Document proves to be untrue in any material respect (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of the date of issuance or making or deemed making thereof;
8.8 Security Documents. If the Guaranty and Security Agreement, any Australian Security Document, or any other Loan Document that purports to create a Lien, shall, for any reason, fail or cease to create a valid and perfected Lien on a material portion of the Collateral covered thereby (or a Loan Party shall so assert), except for a failure or cessation (a) pursuant to the terms hereof or thereof or (b) as the result of an action or failure to act on the part of Agent or the Australian Security Trustee (as the case may be);
8.9 Loan Documents. The validity or enforceability of any Loan Document shall at any time for any reason (other than solely as the result of an action or failure to act on the part of Agent or the Australian Security Trustee) be declared to be null and void, or a proceeding shall be commenced by a Loan Party or its Subsidiaries, or by any Governmental Authority having jurisdiction over a Loan Party, seeking to establish the invalidity or unenforceability thereof, or a Loan Party shall deny that such Loan Party has any liability or obligation purported to be created under any Loan Document;
8.10 Change in Control. A Change in Control shall occur, whether directly or indirectly;
8.11 ERISA and Australian Pension Events. The occurrence of any of the following events: (a) any Loan Party or any of its Subsidiaries or the ERISA Affiliates fails to make full payment within 30 days when due of all amounts which any Loan Party or any of its Subsidiaries or the ERISA Affiliates is required to pay as contributions, installments, or otherwise to or with respect to a Pension Plan or Multiemployer Plan, and such failure, individually or in the aggregate, would reasonably be expected to result in a Material Adverse Effect, (b) an accumulated funding deficiency or funding shortfall occurs or exists, whether or not waived, with respect to any Pension Plan, that, individually or in the aggregate, would reasonably be expected to result in a Material Adverse Effect, (c) a Notification Event, which, individually or in the aggregate, would reasonably be expected to result in a Material Adverse Effect, (d) any Loan Party or any of its Subsidiaries or their ERISA Affiliates completely or partially withdraws from one or more Multiemployer Plans and (i) incurs Withdrawal Liability, (ii) requires payment in any one calendar year, or (iii) fails to make any Withdrawal Liability payment when due, which in each case of (i), (ii) and (iii), individually or in the aggregate would reasonably be expected to result in a Material Adverse Effect, or (e) the existence of any facts or circumstances with respect to the Employee Benefit Plans in the aggregate that results in or is likely to result in a Material Adverse Effect; or
8.12 Expropriation, etc. Solely prior to the Australian End Date, if any expropriation, attachment, sequestration, distress or execution affects any assets or assets of an Australian Loan Party included in the Borrowing Base where the value of those assets exceeds \$10,000,000.
9 RIGHTS AND REMEDIES.
9.1 Rights and Remedies. Upon the occurrence and during the continuation of an Event of Default, Agent (or the Australian Security Trustee, as the case may be) may, and, at the instruction of the Required Lenders, shall (in each case under clauses (a) or (b) by written notice to Borrowers), in addition to any other rights or remedies provided for hereunder or under any other Loan Document or by applicable law, do any one or more of the following:

- (b) declare the Revolver Commitments terminated, whereupon the Revolver Commitments shall immediately be terminated together with (i) any obligation of any Revolving Lender to make Revolving Loans, (ii) the obligation of any Swing Lender to make Swing Loans, and (iii) the obligation of any Issuing Bank to issue Letters of Credit; and
- (c) exercise (and the Australian Security Trustee may exercise, as the case may be) all other rights and remedies available to Agent, the Australian Security Trustee or the Lenders under the Loan Documents, under applicable law, or in equity.

The foregoing to the contrary notwithstanding, upon the occurrence of any Event of Default described in Section 8.4 or Section 8.5, in addition to the remedies set forth above, without any notice to Borrowers or any other Person or any act by the Lender Group, the Revolver Commitments shall automatically terminate and the Obligations (other than the Bank Product Obligations), inclusive of the principal of, and any and all accrued and unpaid interest and fees in respect of, the Loans and all other Obligations (other than the Bank Product Obligations), whether evidenced by this Agreement or by any of the other Loan Documents, shall automatically become and be immediately due and payable and (x) U.S. Borrowers shall be obligated to repay all of such Australian Obligations in full (including Borrowers being obligated to provide (and Borrowers agree that they will provide) (1) Letter of Credit Collateralization to Agent to be held as security for Borrowers' reimbursement obligations in respect of drawings that may subsequently occur under issued and outstanding Letters of Credit and (2) Bank Product Collateralization to be held as security for Borrowers' or their Subsidiaries' obligations in respect of outstanding Bank Products), without presentment, demand, protest, or notice or other requirements of any kind, all of which are expressly waived by each Borrower.

9.2	Remedies Cumu	<u>ative</u> . The righ	ts and remedies	of the Lend	er Group unde	r this Agreement	, the other Loan	Documents,	and all other
agreements	shall be cumulative.	The Lender Gr	oup shall have a	II other right	s and remedies	not inconsistent	herewith as pro	ovided under t	he Code, the
Corporation	s Act by law, or in equi-	y. No exercise	by the Lender Gro	oup of one rig	ht or remedy sh	all be deemed an	election, and no	waiver by the l	Lender Group
of any Even	t of Default shall be de	emed a continui	ng waiver. No del	ay by the Ler	der Group shall	constitute a waive	er, election, or ac	quiescence by	it.

10 WAIVERS; INDEMNIFICATION.

- 10.1 <u>Demand; Protest; etc.</u> Each Borrower waives demand, protest, notice of protest, notice of default or dishonor, notice of payment and nonpayment, nonpayment at maturity, release, compromise, settlement, extension, or renewal of documents, instruments, chattel paper, and guarantees at any time held by the Lender Group on which any Borrower may in any way be liable.
- 10.2 <u>The Lender Group's Liability for Collateral</u>. Each Borrower hereby agrees that: (a) so long as Agent complies with its obligations, if any, under the Code, the Lender Group shall not in any way or manner be liable

or responsible for: (i) the safekeeping of the Collateral, (ii) any loss or damage thereto occurring or arising in any manner or fashion from any cause, (iii) any diminution in the value thereof, or (iv) any act or default of any carrier, warehouseman, bailee, forwarding agency, or other Person, and (b) all risk of loss, damage, or destruction of the Collateral shall be borne by Borrowers.

10.3 Indemnification. Subject to Section 2.15 (including Section 2.15(o)), each Borrower shall pay, indemnify, defend, and hold the Agent-Related Persons and the Lender-Related Persons (each, an "Indemnified Person") harmless (to the fullest extent permitted by law) from and against any and all claims, demands, suits, actions, investigations, proceedings, liabilities, fines, costs, penalties, and damages, and all reasonable and documented fees and disbursements of counsel (limited to one primary counsel and one local counsel in each relevant jurisdiction (including Australia) for all Indemnified Persons, taken as a whole, and, solely in the case of an actual or perceived conflict of interest, one additional counsel to each group of Indemnified Persons similarly situated) and one environmental consultant and all other reasonable and documented out-of-pocket costs and expenses incurred in connection therewith or in connection with the enforcement of this indemnification (as and when they are incurred and irrespective of whether suit is brought), at any time asserted against, imposed upon, or incurred by any of them (a) in connection with or as a result of or related to the execution and delivery (provided that Borrowers shall not be liable for costs and expenses (including attorneys' fees) of any Lender (other than Bank of America) incurred in advising, structuring, drafting, reviewing, administering or syndicating the Loan Documents), enforcement, performance, or administration (including any restructuring or workout with respect hereto) of this Agreement, any of the other Loan Documents, or the transactions contemplated hereby or thereby or the monitoring of Parent's and its Subsidiaries' compliance with the terms of the Loan Documents (provided, that the indemnification in this clause (a) shall not extend to (i) disputes solely between or among the Lenders that do not involve any acts or omissions of any Loan Party, or (ii) disputes solely between or among the Lenders and their respective Affiliates that do not involve any acts or omissions of any Loan Party; it being understood and agreed that the indemnification in this clause (a) shall extend to Agent (but not the Lenders) relative to disputes between or among Agent on the one hand, and one or more Lenders, or one or more of their Affiliates, on the other hand, or (iii) any Taxes or any costs attributable to Taxes that are governed by Section 17), (b) with respect to any actual or prospective investigation, litigation, or proceeding related to this Agreement, any other Loan Document, the making of any Loans or issuance of any Letters of Credit hereunder, or the use of the proceeds of the Loans or the Letters of Credit provided hereunder (irrespective of whether any Indemnified Person is a party thereto), or any act, omission, event, or circumstance in any manner related thereto, and (c) in connection with or arising out of any presence or release of Hazardous Materials giving rise to liability or for which any Lender is required to incur costs at, on, under, to or from the business or any assets or properties owned, leased or operated by any Borrower or any of its Subsidiaries or any Environmental Actions, Environmental Liabilities or Remedial Actions related in any way to the Borrowers or any of their Subsidiaries or the business or any such assets or properties of any Borrower or any of its Subsidiaries (each and all of the foregoing, the "Indemnified Liabilities"). The foregoing to the contrary notwithstanding, no Borrower shall have any obligation to any Indemnified Person under this Section 10.3 with respect to any Indemnified Liability that a court of competent jurisdiction finally determines to have resulted from the bad faith, gross negligence or willful misconduct of. by such Indemnified Person or its officers, directors, employees, attorneys, or agents. This provision shall survive the termination of this Agreement and the repayment in full of the Obligations. If any Indemnified Person makes any payment to any other Indemnified Person with respect to an Indemnified Liability as to which Borrowers were required to indemnify the Indemnified Person receiving such payment, the Indemnified Person making such payment is entitled to be indemnified and reimbursed by Borrowers with respect thereto. WITHOUT LIMITATION, THE FOREGOING INDEMNITY SHALL APPLY TO EACH INDEMNIFIED PERSON WITH RESPECT TO INDEMNIFIED LIABILITIES WHICH IN WHOLE OR IN PART ARE CAUSED BY OR ARISE OUT OF ANY NEGLIGENT ACT OR OMISSION (OTHER THAN A GROSSLY NEGLIGENT ACT OR OMISSION OR TO THE EXTENT CONSTITUTING BAD FAITH OR WILLFUL MISCONDUCT) OF SUCH INDEMNIFIED PERSON OR OF ANY OTHER PERSON. The indemnity in this clause from the Australian Borrowers applies only to the extent such claims, demands, suits, actions, investigations, proceedings, liabilities, fines, costs, penalties, damages, fees and disbursements of counsel and other out-of-pocket expenses relate to the Australian Obligations.

11 NOTICES.

Unless otherwise provided in this Agreement, all notices or demands relating to this Agreement or any other Loan Document shall be in writing and (except for financial statements and other informational documents which may be sent by first-class mail, postage prepaid) shall be personally delivered or sent by registered or certified mail (postage

prepaid, return receipt requested), overnight courier, electronic mail (at such email addresses as a party may designate in accordance herewith), or telefacsimile. In the case of notices or demands to any Borrower or Agent, as the case may be, they shall be sent to the respective address set forth below:

If to any Borrower: c/o CLEVELAND-CLIFFS INC.

200 Public Square #3300 Cleveland, Ohio 44114 Attn: James Graham Fax No. 216-694-6509

Email: james.graham@clevelandcliffs.com

With a copy to: Cleveland-Cliffs Inc. 200 Public Square #3300 Cleveland, Ohio 44114 Attn: Timothy K. Flanagan Fax No. 216-694-6509

Email: timothy.flanagan@clevelandcliffs.com

If to Agent: BANK OF AMERICA, N.A.

135 S. LaSalle St., Suite 925

Chicago, IL 60604 Attn: Thomas Herron Tel no.: 312-992-6107 Fax No. 312-904-7190

Email: thomas.h.herron@baml.com

If to Issuing Bank: To Agent

Any party hereto may change the address at which they are to receive notices hereunder, by notice in writing in the foregoing manner given to the other party. Each notice shall be effective only (a) if given by facsimile transmission, when transmitted to the applicable facsimile number, if confirmation of receipt is received; (b) if given by mail, three Business Days after deposit in the mail, with first-class postage pre-paid, addressed to the applicable address; or (c) if given by personal delivery, when duly delivered to the notice address with receipt acknowledged. Any written communication that is not sent in conformity with the foregoing provisions shall nevertheless be effective on the date actually received by the noticed party. Any notice received by Parent shall be deemed received by all Borrowers. Electronic communications (including e-mail, messaging and websites) may be used only in a manner reasonably acceptable to Agent and, unless otherwise agreed by Agent, only for routine communications, such as delivery of administrative matters and distribution of Loan Documents. Agent makes no assurance as to the privacy or security of electronic communications. Voice mail shall not be effective notices under the Loan Documents.

12 CHOICE OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION.

- (a) THE VALIDITY OF THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS (UNLESS EXPRESSLY PROVIDED TO THE CONTRARY IN ANOTHER LOAN DOCUMENT IN RESPECT OF SUCH OTHER LOAN DOCUMENT), THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT HEREOF AND THEREOF, THE RIGHTS OF THE PARTIES HERETO AND THERETO WITH RESPECT TO ALL MATTERS ARISING HEREUNDER OR THEREUNDER OR RELATED HERETO OR THERETO, AND ANY CLAIMS, CONTROVERSIES OR DISPUTES ARISING HEREUNDER OR THEREUNDER OR RELATED HERETO OR THERETO SHALL BE DETERMINED UNDER, GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.
- (b) THE PARTIES AGREE THAT ALL ACTIONS OR PROCEEDINGS ARISING IN CONNECTION WITH THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS SHALL BE TRIED AND LITIGATED ONLY IN THE STATE AND, TO THE EXTENT PERMITTED BY APPLICABLE LAW, FEDERAL COURTS LOCATED IN THE COUNTY

OF NEW YORK, STATE OF NEW YORK; PROVIDED, THAT ANY SUIT SEEKING ENFORCEMENT AGAINST ANY COLLATERAL OR OTHER PROPERTY MAY BE BROUGHT, AT AGENT'S OPTION, IN THE COURTS OF ANY JURISDICTION WHERE AGENT ELECTS TO BRING SUCH ACTION OR WHERE SUCH COLLATERAL OR OTHER PROPERTY MAY BE FOUND. EACH BORROWER AND EACH MEMBER OF THE LENDER GROUP WAIVE, TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, ANY RIGHT EACH MAY HAVE TO ASSERT THE DOCTRINE OF FORUM NON CONVENIENS OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS SECTION 12(b).

- (c) TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, EACH BORROWER AND EACH MEMBER OF THE LENDER GROUP HEREBY WAIVE THEIR RESPECTIVE RIGHTS, IF ANY, TO A JURY TRIAL OF ANY CLAIM, CONTROVERSY, DISPUTE OR CAUSE OF ACTION DIRECTLY OR INDIRECTLY BASED UPON OR ARISING OUT OF ANY OF THE LOAN DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED THEREIN, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW OR STATUTORY CLAIMS (EACH A "CLAIM"). EACH BORROWER AND EACH MEMBER OF THE LENDER GROUP REPRESENT THAT EACH HAS REVIEWED THIS WAIVER AND EACH KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. IN THE EVENT OF LITIGATION, A COPY OF THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.
- (d) EACH BORROWER HEREBY IRREVOCABLY AND UNCONDITIONALLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS LOCATED IN THE COUNTY OF NEW YORK AND THE STATE OF NEW YORK, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO ANY LOAN DOCUMENTS, OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT SHALL AFFECT ANY RIGHT THAT AGENT MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AGAINST ANY LOAN PARTY OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION.
- (e) NO CLAIM MAY BE MADE BY ANY PARTY HERETO AGAINST ANY LOAN PARTY, THE AGENT, THE SWING LENDER, ANY OTHER LENDER, ISSUING BANK, OR ANY AFFILIATE, DIRECTOR, OFFICER, EMPLOYEE, COUNSEL, REPRESENTATIVE, AGENT, OR ATTORNEY-IN-FACT OF ANY OF THEM FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES OR LOSSES IN RESPECT OF ANY CLAIM FOR BREACH OF CONTRACT OR ANY OTHER THEORY OF LIABILITY ARISING OUT OF OR RELATED TO THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, OR ANY ACT, OMISSION, OR EVENT OCCURRING IN CONNECTION THEREWITH, AND EACH LOAN PARTY HEREBY WAIVES, RELEASES, AND AGREES NOT TO SUE UPON ANY CLAIM FOR SUCH DAMAGES, WHETHER OR NOT ACCRUED AND WHETHER OR NOT KNOWN OR SUSPECTED TO EXIST IN ITS FAVOR; PROVIDED THAT NOTHING HEREIN SHALL LIMIT THE BORROWERS' OBLIGATIONS TO INDEMNIFY THE AGENT-RELATED PERSONS AND THE LENDER-RELATED PERSONS AS REQUIRED UNDER, AND SUBJECT TO, SECTION 10.3.

13 ASSIGNMENTS AND PARTICIPATIONS; SUCCESSORS.

13.1 Assignments and Participations.

- (a) (i) Subject to the conditions set forth in clause (a)(ii) below, any Lender may assign and delegate all or any portion of its rights and duties under the Loan Documents (including the Obligations owed to it and its Revolver Commitments) to one or more assignees so long as such prospective assignee is an Eligible Transferee (each, an "Assignee"), with the prior written consent (such consent not be unreasonably withheld or delayed) of:
- (A) Borrowers; provided, that no consent of Borrowers shall be required (1) if an Event of Default has occurred and is continuing, or (2) in connection with an assignment to a Person that is a Lender or an Affiliate (other than natural persons) of a Lender; provided <u>further</u>, that Borrowers shall be deemed to have

consented to a proposed assignment unless they object thereto by written notice to Agent within 5 Business Days after having received notice thereof; and

- (B) Agent and each Issuing Bank; provided, that no consent of Agent or any Issuing Bank shall be required in connection with an assignment to a Person that is a Lender or an Affiliate (other than natural persons) of a Lender.
 - (ii) Assignments shall be subject to the following additional conditions:
 - (A) [Reserved,]
 - (B) no assignment may be made, (i) to a Competitor, or (ii) to a natural person,
 - (C) no assignment may be made to a Loan Party or an Affiliate of a Loan Party or an Offshore Associate of an Australian

Loan Party,

- (D) the amount of the Revolver Commitments and the other rights and obligations of the assigning Lender hereunder and under the other Loan Documents subject to each such assignment (determined as of the date the Assignment and Acceptance with respect to such assignment is delivered to Agent) shall be in a minimum amount (unless waived by Agent) of \$5,000,000 (except such minimum amount shall not apply to (1) an assignment or delegation by any Lender to any other Lender, an Affiliate of any Lender, or a Related Fund of such Lender or (2) a group of new Lenders, each of which is an Affiliate of each other or a Related Fund of such new Lender to the extent that the aggregate amount to be assigned to all such new Lenders is at least \$5,000,000) or (3) in the case of an assignment of the entire remaining amount of the assigning Lender's Revolver Commitment and/or Obligations at the time owing it,
- (E) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement,
- (F) the parties to each assignment shall execute and deliver to Agent an Assignment and Acceptance; provided, that Borrowers and Agent may continue to deal solely and directly with the assigning Lender in connection with the interest so assigned to an Assignee until written notice of such assignment, together with payment instructions, addresses, and related information with respect to the Assignee, have been given to Borrowers and Agent by such Lender and the Assignee,
- (G) unless waived by Agent, the assigning Lender or Assignee has paid to Agent, for Agent's separate account, a processing fee in the amount of \$3,500, and
- (H) the assignee, if it is not a Lender, shall deliver to Agent an administrative questionnaire in a form approved by Agent (the "Administrative Questionnaire").
- (b) From and after the date that Agent receives the executed Assignment and Acceptance, records it in the Register, and if applicable, receives payment of the required processing fee, (i) the Assignee thereunder shall be a party hereto and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment and Acceptance, subject to Agent's recording of the Assignment and Acceptance in the Register as required by Section 13.1(h), shall be a "Lender" and shall have the rights and obligations of a Lender under the Loan Documents, and (ii) the assigning Lender shall, to the extent that rights and obligations hereunder and under the other Loan Documents have been assigned by it pursuant to such Assignment and Acceptance, relinquish its rights (except with respect to Section 10.3) and be released from any future obligations under this Agreement (and in the case of an Assignment and Acceptance covering all or the remaining portion of an assigning Lender's rights and obligations under this Agreement and the other Loan Documents, such Lender shall cease to be a party hereto and thereto); provided, that nothing contained herein shall release any assigning Lender from obligations that survive the termination of this Agreement, including such assigning Lender's obligations under Section 18.9(a).
- (c) By executing and delivering an Assignment and Acceptance, the assigning Lender thereunder and the Assignee thereunder confirm to and agree with each other and the other parties hereto as follows: (i) other than as provided in such Assignment and Acceptance, such assigning Lender makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Agreement or the execution, legality, validity, enforceability, genuineness, sufficiency or value of this Agreement

or any other Loan Document furnished pursuant hereto, (ii) such assigning Lender makes no representation or warranty and assumes no responsibility with respect to the financial condition of any Borrower or the performance or observance by any Borrower of any of its obligations under this Agreement or any other Loan Document furnished pursuant hereto, (iii) such Assignee confirms that it has received a copy of this Agreement, together with such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Acceptance, (iv) such Assignee will, independently and without reliance upon Agent, such assigning Lender or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement, (v) such Assignee appoints and authorizes Agent to take such actions and to exercise such powers under this Agreement and the other Loan Documents as are delegated to Agent, by the terms hereof and thereof, together with such powers as are reasonably incidental thereto, and (vi) such Assignee agrees that it will perform all of the obligations which by the terms of this Agreement are required to be performed by it as a Lender.

- (d) Immediately upon Agent's receipt of the required processing fee, if applicable, and delivery of notice to the assigning Lender pursuant to Section 13.1(b), this Agreement shall be deemed to be amended to the extent, but only to the extent, necessary to reflect the addition of the Assignee and the resulting adjustment of the Revolver Commitments arising therefrom. The Revolver Commitment allocated to each Assignee shall reduce such Revolver Commitments of the assigning Lender pro tanto.
- (e) Any Lender may at any time sell to one or more commercial banks, financial institutions, or other Persons that is an Eligible Transferee and is not a Competitor (a "Participant") participating interests in all or any portion of its Obligations, its Revolver Commitment, and the other rights and interests of that Lender (the "Originating Lender") hereunder and under the other Loan Documents; provided, that (i) the Originating Lender shall remain a "Lender" for all purposes of this Agreement and the other Loan Documents and the Participant receiving the participating interest in the Obligations, the Revolver Commitments, and the other rights and interests of the Originating Lender hereunder shall not constitute a "Lender" hereunder or under the other Loan Documents and the Originating Lender's obligations under this Agreement shall remain unchanged, (ii) the Originating Lender shall remain solely responsible for the performance of such obligations, (iii) Borrowers, Agent, and the Lenders shall continue to deal solely and directly with the Originating Lender in connection with the Originating Lender's rights and obligations under this Agreement and the other Loan Documents (it being understood that the documentation required under Section 17.2 shall be delivered to the participating Lender), (iv) no Lender shall transfer or grant any participating interest under which the Participant has the right to approve any amendment to, or any consent or waiver with respect to, this Agreement or any other Loan Document, except to the extent such amendment to, or consent or waiver with respect to this Agreement or of any other Loan Document would (A) extend the final maturity date of the Obligations hereunder in which such Participant is participating, (B) reduce the interest rate applicable to the Obligations hereunder in which such Participant is participating (excluding the imposition of the Default Rate), (C) release all or substantially all of the Collateral or guaranties (except to the extent expressly provided herein or in any of the Loan Documents) supporting the Obligations hereunder in which such Participant is participating, (D) postpone the payment of, or reduce the amount of, the interest or fees payable to such Participant through such Lender (other than a waiver of default interest), or (E) decreases the amount or postpones the due dates of scheduled principal repayments or prepayments payable to such Participant through such Lender. (v) no participation shall be sold to a natural person, (vi) no participation shall be sold to a Loan Party or an Affiliate of a Loan Party, and (vii) all amounts payable by Borrowers hereunder shall be determined as if such Lender had not sold such participation. The rights of any Participant only shall be derivative through the Originating Lender with whom such Participant participates and no Participant shall have any rights under this Agreement or the other Loan Documents or any direct rights as to the other Lenders, Agent, Borrowers, the Collateral, or otherwise in respect of the Obligations. Notwithstanding the preceding sentence, the Borrower agrees that each Participant shall be entitled to the benefits of Section 17 (Withholding Taxes) (subject to the requirements and limitations therein, including the requirements under Section 17.2 (Exemptions) (it being understood that the documentation required under Section 17.2 shall be delivered to the Lender granting the participation only) to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to paragraph (a) of this Section 13.1; provided that such Participant (A) agrees to be subject to the provisions of Sections 14.2 (Replacement of Certain Lenders) as if it were an assignee under paragraph (a) of this Section 13.1; and (B) shall not be entitled to receive any greater payment under Section 17 (Withholding Taxes), with respect to any participation, than its Originating Lender would have been entitled to receive, except to the extent such entitlement to receive a greater payment results from a Change in Law that occurs after the Participant acquired the applicable participation. No Participant shall have the right to participate directly in the making of decisions by the Lenders among themselves.

(f) In connection with any such assignment or participation or proposed assignment or participation or any grant of a security interest in, or pledge of, its rights under and interest in this Agreement, a Lender

may, subject to the provisions of Section	<u>n 18.9</u> , disclose all documents	s and information which it no	ow or hereafter may ha	ve relating to any B	orrower and its
Subsidiaries and their respective busines	ses.				

- (g) Any other provision in this Agreement notwithstanding, any Lender may at any time create a security interest in, or pledge, all or any portion of its rights under and interest in this Agreement in favor of any Federal Reserve Bank in accordance with Regulation A of the Federal Reserve Bank or U.S. Treasury Regulation 31 CFR §203.24, and such Federal Reserve Bank may enforce such pledge or security interest in any manner permitted under applicable law.
- (h) Agent (as a non-fiduciary agent on behalf of Borrowers) shall maintain, or cause to be maintained, a register (the "Register") on which it enters the name and address of each Lender as the registered owner of the Loans (and the principal amount thereof and stated interest thereon) held by such Lender (each, a "Registered Loan"). A Registered Loan (and the registered note, if any, evidencing the same) may be assigned or sold in whole or in part only by registration of such assignment or sale on the Register (and each registered note shall expressly so provide) and any assignment or sale of all or part of such Registered Loan (and the registered note, if any, evidencing the same) may be effected only by registration of such assignment or sale on the Register, together with the surrender of the registered note, if any, evidencing the same duly endorsed by (or accompanied by a written instrument of assignment or sale duly executed by) the holder of such registered note, whereupon, at the request of the designated assignee(s) or transferee(s), one or more new registered notes in the same aggregate principal amount shall be issued to the designated assignee(s) or transferee(s). Prior to the registration of assignment or sale of any Registered Loan (and the registered note, if any, evidencing the same) is registered as the owner thereof for the purpose of receiving all payments thereon and for all other purposes, notwithstanding notice to the contrary.
- (i) In the event that a Lender sells participations in the Registered Loan, such Lender, as a non-fiduciary agent on behalf of Borrowers, shall maintain (or cause to be maintained) a register on which it enters the name and address of all participants in the Registered Loans held by it (and the principal amount (and stated interest thereon) of the portion of such Registered Loans that is subject to such participants) (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any Commitments, Loans or participations in Letters of Credit and Swing Loans or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such Commitment, Loan or participation in a Letter of Credit or Swing Loan is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. A Registered Loan (and the registered note, if any, evidencing the same) may be participated in whole or in part only by registration of such participation on the Participant Register (and each registered note hall expressly so provide). Any participation of such Registered Loan (and the registered note, if any, evidencing the same) may be effected only by the registration of such participation on the Participant Register. The conclusiveness of the Participant Register shall be subject to the qualification "absent manifest error".
- (j) Agent shall make a copy of the Register (and each Lender shall make a copy of its Participant Register in the extent it has one) available for review by Borrowers from time to time as Borrowers may reasonably request.
- 13.2 <u>Successors</u>. This Agreement shall bind and inure to the benefit of the respective successors and assigns of each of the parties; provided, that no Borrower may assign this Agreement or any rights or duties hereunder without the Lenders' prior written consent and any prohibited assignment shall be absolutely void ab initio. No consent to assignment by the Lenders shall release any Borrower from its Obligations. A Lender may assign this Agreement and the other Loan Documents and its rights and duties hereunder and thereunder pursuant to <u>Section 13.1</u> and, except as expressly required pursuant to Section 13.1, no consent or approval by any Borrower is required in connection with any such assignment.
- 14 AMENDMENTS; WAIVERS.

14.1 Amendments and Waivers.

- (a) No amendment, waiver or other modification of any provision of this Agreement or any other Loan Document (other than Bank Product Agreements or the Fee Letter), and no consent with respect to any departure by any Borrower therefrom, shall be effective unless the same shall be in writing and signed by the Required Lenders (or by Agent at the written request of the Required Lenders) and the Loan Parties that are party thereto and then any such waiver or consent shall be effective, but only in the specific instance and for the specific purpose for which given; provided that no such waiver, amendment, or consent shall, unless in writing and signed by all of the Lenders directly affected thereby and all of the Loan Parties that are party thereto, do any of the following:
- (i) increase the amount of or extend the expiration date of any Revolver Commitment of any Lender or amend, modify, or eliminate the last sentence of Section 2.4(c)(i) or Section 2.4(c)(ii),
- (ii) postpone or delay any date fixed by this Agreement or any other Loan Document for any payment of principal, interest, fees, or other amounts due hereunder or under any other Loan Document,
- (iii) reduce the principal of, or the rate of interest on, any loan or other extension of credit hereunder, or reduce any fees or other amounts payable hereunder or under any other Loan Document (except in connection with the waiver of applicability of <u>Section 2.6(c)</u> (which waiver shall be effective with the written consent of the Required Lenders)),
 - (iv) amend, modify, or eliminate this Section or any provision of this Agreement providing for consent or other action by all Lenders,
 - (v) amend, modify, or eliminate Section 3.1 or 3.3,
 - (vi) amend, modify, or eliminate Section 15.9 or 15.10,
 - (vii) other than as permitted by Section 15.9, release Agent's Lien in and to all or substantially all of the Collateral,
- (viii) amend, modify, or eliminate the definitions of "<u>Required Lenders</u>", "<u>Supermajority Lenders</u>" or "<u>Pro Rata Share</u>", or amend or modify the percentage set forth in any other provision of any Loan Document (including this <u>Section 14.1</u>) specifying the number or percentage of Lenders required to waive, amend or modify any rights thereunder or make any determination or grant any consent thereunder.
 - (ix) contractually subordinate any of Agent's Liens on the ABL Collateral,
- (x) other than in connection with a merger, liquidation, dissolution or sale of such Person expressly permitted by the terms hereof or the other Loan Documents, or if such Person constitutes an Excluded Subsidiary, release any Borrower or any Guarantor from any obligation for the payment of money or consent to the assignment or transfer by any Borrower or any Guarantor of any of its rights or duties under this Agreement or the other Loan Documents if such release would release all or substantially all of the guarantees provided thereunder; provided that, notwithstanding the foregoing, Agent and Australian Security Trustee shall be entitled to, and agrees to, release the Australian Loan Parties from the Loan Documents, and to release their Liens on the assets of the Australian Loan Parties, without the consent of any Lender so long as all Australian Obligations have been paid in full, all Australian Letters of Credit have been terminated (or Cash Collateralized in accordance with the terms of this Agreement) and all Australian Revolver Commitments have been permanently terminated (such time, the "Australian End Date"), and upon the occurrence of the Australian End Date, (1) the Australian Loan Parties shall not be required to grant Liens on their assets, (2) each Australian Loan Party shall no longer constitute a "Australian Loan Party" or a "Loan Party" under this Agreement or any of the Loan Documents, (3) the Agent and the Australian Security Trustee shall be authorized to enter into such amendments to this Agreement and the other Loan Documents to evidence the termination of the Australian Revolver Commitments and the release of the Australian Loan Parties, (4) the Agent and the Australian Security Trustee will enter into such documents as reasonably requested by the Borrower Representative to effectuate such release and evidence the foregoing, including amendments to the Loan Documents, and (5) the release of the Australian Loan Parties from the Loan Documents and the termination of such Loan Documents and

- (xi) amend, modify, or eliminate any of the provisions of Section 2.4(b)(i), (iii), (iii), (iii), (iiv) or (v) or Section 2.4(e) or (f), or
- (xii) amend, modify, or eliminate any of the provisions of <u>Section 13.1</u> (1) with respect to assignments to, or participations with, Persons who are Loan Parties or their Affiliates or (2) in a manner which further restricts assignments by Lenders thereunder.
 - (b) No amendment, waiver, modification, or consent shall amend, modify, waive, or eliminate,
- (i) the definition of, or any of the terms or provisions of, the Fee Letter, without the written consent of Agent and Borrowers (and shall not require the written consent of any of the Lenders),
- (ii) any provision of <u>Section 15</u> pertaining to Agent, or any other rights or duties of Agent under this Agreement or the other Loan Documents, without the written consent of Agent, Borrowers, and the Required Lenders;
- (c) No amendment, waiver, modification, elimination, or consent shall amend, modify, or eliminate (i) the definition of U.S. Borrowing Base (or any of the defined terms (including the definitions of Eligible Accounts, Eligible Inventory, and Eligible Equipment) that are used in such definition) to the extent that any such change results in more credit being made available to U.S. Borrowers based upon the U.S. Borrowing Base, but not otherwise, the last paragraph of Section 2.1(a), (ii) the definition of U.S. Maximum Revolver Amount, (iii) the definition of "ABL Priority Collateral" contained in the Intercreditor Agreement or (iv) Section 4.01 of the Intercreditor Agreement, in each case, without the written consent of Agent, Borrowers, and the Supermajority Lenders;
- (d) No amendment, waiver, modification, elimination, or consent shall amend, modify, or eliminate the definition of Australian Borrowing Base (or any of the defined terms (including the definitions of Eligible Accounts, Eligible Inventory, In-Transit Inventory, and Eligible Equipment) that are used in such definition) to the extent that any such change results in more credit being made available to Australian Borrowers based upon the Australian Borrowing Base, but not otherwise, the last paragraph of Section 2.1(b), or the definition of Australian Maximum Revolver Amount, in each case, without the written consent of Agent, Borrowers, and the Supermajority Lenders;
- (e) No amendment, waiver, modification, elimination, or consent shall amend, modify, or waive any provision of this Agreement or the other Loan Documents pertaining to any Issuing Bank, or any other rights or duties of any Issuing Bank under this Agreement or the other Loan Documents, without the written consent of such Issuing Bank, Agent, Borrowers, and the Required Lenders;
- (f) No amendment, waiver, modification, elimination, or consent shall amend, modify, or waive any provision of this Agreement or the other Loan Documents pertaining to any Swing Lender, or any other rights or duties of any Swing Lender under this Agreement or the other Loan Documents, without the written consent of such Swing Lender, Agent, Borrowers, and the Required Lenders; and
- (g) Anything in this <u>Section 14.1</u> to the contrary notwithstanding, (i) any amendment, modification, elimination, waiver, consent, termination, or release of, or with respect to, any provision of this Agreement or any other Loan Document that relates only to the relationship of the Lender Group among themselves, and that does not affect the rights or obligations of any Borrower, shall not require consent by or the agreement of any Loan Party, (ii) any amendment, waiver, modification, elimination, or consent of or with respect to any provision of this Agreement or any other Loan Document may be entered into without the consent of, or over the objection of, any Defaulting Lender other than any of the matters governed by <u>Section 14.1(a)(i)</u> through (iv) that affect such Lender and (iii) Agent and Parent shall be permitted to amend any provision of this Agreement or any other Loan Document (and such amendment shall become effective without any further action or consent of any other party to any Loan Document) if Agent and Parent shall have jointly identified an obvious error or omission of a technical or immaterial nature in any such provision.
- (h) Notwithstanding the foregoing, the Agent and the Borrowers may enter into an Incremental Amendment in accordance with Section 2.16 and such document shall be effective to amend the terms of this Agreement and the other applicable Loan Documents, in each case without any further action or consent of any other party to any Loan Documents except as otherwise required by such Section.

14.2 Replacement of Certain Lenders.

- (a) If (i) any action to be taken by the Lender Group or Agent hereunder requires the consent, authorization, or agreement of all Lenders or all Lenders affected thereby and if such action has received the consent, authorization, or agreement of the Required Lenders but not of all Lenders or all Lenders affected thereby, or (ii) any Lender makes a claim for compensation under <u>Section 17</u>, then Borrowers, at their sole cost and expense (including any assignment fees), upon at least five Business Days prior irrevocable notice, may permanently replace any Lender that failed to give its consent, authorization, or agreement (a "<u>Non-Consenting Lender</u>") or any Lender that made a claim for compensation (a "<u>Tax Lender</u>") with one or more Replacement Lenders, and the Non-Consenting Lender or Tax Lender, as applicable, shall have no right to refuse to be replaced hereunder. Such notice to replace the Non-Consenting Lender or Tax Lender, as applicable, shall specify an effective date for such replacement, which date shall not be later than 15 Business Days after the date such notice is given.
- (b) Prior to the effective date of such replacement, the Non-Consenting Lender or Tax Lender, as applicable, and each Replacement Lender shall execute and deliver an Assignment and Acceptance, subject only to the Non-Consenting Lender or Tax Lender, as applicable, being repaid in full its share of the outstanding Obligations (without any premium or penalty of any kind whatsoever, but including (i) all interest, fees and other amounts that may be due in payable in respect thereof, and (ii) an assumption of its Pro Rata Share of participations in the Letters of Credit). If the Non-Consenting Lender or Tax Lender, as applicable, shall refuse or fail to execute and deliver any such Assignment and Acceptance prior to the effective date of such replacement, Agent may, but shall not be required to, execute and deliver such Assignment and Acceptance in the name or and on behalf of the Non-Consenting Lender or Tax Lender, as applicable, and irrespective of whether Agent executes and delivers such Assignment and Acceptance, the Non-Consenting Lender or Tax Lender, as applicable, shall be deemed to have executed and delivered such Assignment and Acceptance. The replacement of any Non-Consenting Lender or Tax Lender, as applicable, shall be made in accordance with the terms of Section 13.1. Until such time as one or more Replacement Lenders shall have acquired all of the Obligations, the Revolver Commitments, and the other rights and obligations of the Non-Consenting Lender or Tax Lender, as applicable, hereunder and under the other Loan Documents, the Non-Consenting Lender or Tax Lender, as applicable, Pro Rata Share of Revolving Loans and to purchase a participation in each Letter of Credit, in an amount equal to its Pro Rata Share of participations in such Letters of Credit.
- 14.3 <u>No Waivers; Cumulative Remedies</u>. No failure by Agent or any Lender to exercise any right, remedy, or option under this Agreement or any other Loan Document, or delay by Agent or any Lender in exercising the same, will operate as a waiver thereof. No waiver by Agent or any Lender will be effective unless it is in writing, and then only to the extent specifically stated. No waiver by Agent or any Lender on any occasion shall affect or diminish Agent's and each Lender's rights thereafter to require strict performance by Parent and Borrowers of any provision of this Agreement. Agent's and each Lender's rights under this Agreement and the other Loan Documents will be cumulative and not exclusive of any other right or remedy that Agent or any Lender may have.

15 AGENT AND AUSTRALIAN SECURITY TRUSTEE.

15.1 Appointment, Authority and Duties of Agent.

(a) Appointment and Authority. Each Lender appoints and designates Bank of America as Agent and Australian Security Trustee under all Loan Documents. Agent and Australian Security Trustee may, and each Lender authorizes Agent and Australian Security Trustee to, enter into all Loan Documents to which Agent or Australian Security Trustee is intended to be a party and accept the Guaranty and Security Agreement and all Australian Security Documents. Any action taken by Agent and Australian Security Trustee in accordance with the provisions of the Loan Documents, and the exercise by Agent and Australian Security Trustee of any rights or remedies set forth therein, together with all other powers reasonably incidental thereto, shall be authorized by and binding upon all Lenders (an Bank Product Providers). Without limiting the generality of the foregoing, Agent and Australian Security Trustee, as applicable, shall have the sole and exclusive authority to (a) act as the disbursing and collecting agent for Lenders with respect to all payments and collections arising in connection with the Loan Documents; (b) execute and deliver

as Agent and Australian Security Trustee, as applicable, each Loan Document, including any intercreditor or subordination agreement, and accept delivery of each Loan Document; (c) act as collateral agent and security trustee for the Lenders for purposes of perfecting and administering Liens under the Loan Documents, and for all other purposes stated therein; (d) manage, supervise or otherwise deal with Collateral; and (e) take any Enforcement Action or otherwise exercise any rights or remedies with respect to any Collateral or under any Loan Documents, applicable law or otherwise. Agent alone shall be authorized to determine eligibility and applicable advance rates under the U.S. Borrowing Base and the Australian Borrowing Base, whether to impose or release any reserve, or whether any conditions to funding or issuance of a Letter of Credit have been satisfied, which determinations and judgments, if exercised in good faith, shall exonerate Agent from liability to any Lender, Bank Product Provider or other Person for any error in judgment.

- (b) <u>Duties</u>. The titles of "Agent" and "Security Trustee" are used solely as a matter of market custom and the duties of Agent and Australian Security Trustee are administrative in nature only. Neither Agent nor, to the extent permitted by law, Australian Security Trustee have any duties except those expressly set forth in the Loan Documents, and in no event does Agent or Australian Security Trustee have any agency, fiduciary or implied duty to or relationship with any Lender, Bank Product Provider or other Person by reason of any Loan Document or related transaction. The conferral upon Agent and Australian Security Trustee of any right shall not imply a duty to exercise such right, unless instructed to do so by Lenders in accordance with this Agreement.
- (c) <u>Agent Professionals</u>. Each of Agent and Australian Security Trustee may perform its duties through agents and employees. Each of Agent and Australian Security Trustee may consult with and employ Agent Professionals, and shall be entitled to act upon, and shall be fully protected in any action taken in good faith reliance upon, any advice given by an Agent Professional. Neither Agent nor Australian Security Trustee shall be responsible for the negligence or misconduct of any agents, employees or Agent Professionals selected by it with reasonable care.
- (d) Instructions of Required Lenders. The rights and remedies conferred upon Agent and Australian Security Trustee under the Loan Documents may be exercised without the necessity of joining any other party, unless required by applicable law. In determining compliance with a condition for any action hereunder, including satisfaction of any condition in Section 3, Agent may presume that the condition is satisfactory to a Lender and Bank Product Provider unless Agent has received written notice to the contrary from such Lender or Bank Product Provider before Agent takes the action. Agent and Australian Security Trustee may request instructions from Required Lenders or other Lenders or Bank Product Providers with respect to any act (including the failure to act) in connection with any Loan Documents or Collateral, and may seek assurances to its reasonable satisfaction from Lenders or Bank Product Providers of their indemnification obligations against Applicable Claims that could be incurred by Agent or Australian Security Trustee. Agent and Australian Security Trustee may refrain from any act until it has received such instructions or assurances, and shall not incur liability to any Person by reason of so refraining. Instructions of Required Lenders shall be binding upon all Lenders (and Bank Product Providers), and no Lender (or Bank Product Provider) shall have any right of action whatsoever against Agent or Australian Security Trustee as a result of Agent or Australian Security Trustee acting or refraining from acting pursuant to instructions of Required Lenders. Notwithstanding the foregoing, instructions by and consent of specific parties shall be required to the extent provided in Section 14.1. In no event shall Agent or Australian Security Trustee be required to take any action that it determines in its discretion is contrary to applicable law or any Loan Documents or could subject any Agent-Related Person to liability.
- 15.2 <u>Liability of Agent and Australian Security Trustee</u>. Neither Agent nor Australian Security Trustee shall be liable to any Lender or Bank Product Provider for any action taken or omitted to be taken under the Loan Documents, except for losses directly and solely caused by its bad faith, gross negligence or willful misconduct. Neither Agent nor Australian Security Trustee assume any responsibility for any failure or delay in performance or any breach by any Loan Party, Lender or Bank Product Provider of any obligations under the Loan Documents. Neither Agent nor Australian Security Trustee make any express or implied representation, warranty or guarantee to Lenders or Bank Product Providers with respect to any Obligations, Collateral, Liens, Loan Documents or Loan Party. No Agent-Related Person shall be responsible to Lender (and Bank Product Providers) for any recitals, statements, information, representations or warranties contained in any Loan Documents or Borrower Materials; the execution, validity, genuineness, effectiveness or enforceability of any Loan Documents; the genuineness, enforceability, collectability, value, sufficiency, location or existence of any Collateral, or the validity, extent, perfection or priority of any Lien therein; the validity, enforceability or collectability of any Obligations; or the assets, liabilities, financial condition, results of operations, business, creditworthiness or legal status of any Loan Party or Account Debtor. No Agent-Related Person shall have any obligation to any Lender or Bank Product Provider to ascertain or inquire into the existence of any

Default or Event of Default, the observance by any Loan Party of any terms of the Loan Documents, or the satisfaction of any conditions precedent contained in any Loan Documents.

- 15.3 Reliance by Agent. Each of the Agent and the Australian Security Trustee shall be entitled to rely, and shall be fully protected in relying, upon any certification, notice or other communication (including those by telephone, telex, telegram, telecopy or e-mail) believed by it to be genuine and correct and to have been signed, sent or made by the proper Person. Each of the Agent and the Australian Security Trustee shall have a reasonable and practicable amount of time to act upon any instruction, notice or other communication under any Loan Document, and shall not be liable for any delay in acting.
- 15.4 <u>Notice of Default or Event of Default.</u> Agent shall not be deemed to have knowledge of any Default or Event of Default, or of any failure to satisfy any conditions in <u>Section 3</u>, unless it has received written notice from a Borrower or Required Lenders specifying the occurrence and nature thereof. If any Lender acquires knowledge of a Default, Event of Default or failure of such conditions, it shall promptly notify Agent and the other Lenders thereof in writing. Each Lender (and Bank Product Provider) agrees that, except as otherwise provided in any Loan Documents or with the written consent of Agent and Required Lenders, it will not take any Enforcement Action, accelerate Obligations (other than Bank Product Obligations) or assert any rights relating to any Collateral.
- Due Diligence and Non-Reliance. Each Lender acknowledges and agrees that it has, independently and without reliance upon Agent, Australian Security Trustee or any other Lenders, and based upon such documents, information and analyses as it has deemed appropriate, made its own credit analysis of each Loan Party and its own decision to enter into this Agreement and to fund Loans and participate in Letters of Credit hereunder. Each Lender (and Bank Product Provider) has made such inquiries as it feels necessary concerning the Loan Documents, Collateral and Loan Parties. Each Lender (and Bank Product Provider) acknowledges and agrees that the other Lenders (and Bank Product Providers) have made no representations or warranties concerning any Loan Party, any Collateral or the legality, validity, sufficiency or enforceability of any Loan Documents or Obligations. Each Lender (and Bank Product Provider) will, independently and without reliance upon any other Lender (or Bank Product Provider), and based upon such financial statements, documents and information as it deems appropriate at the time, continue to make and rely upon its own credit decisions in making Loans and participating in Letters of Credit, and in taking or refraining from any action under any Loan Documents. Except for notices, reports and other information expressly requested by a Lender, Agent shall have no duty or responsibility to provide any Lender (or Bank Product Provider) with any notices, reports or certificates furnished to Agent by any Loan Party or any credit or other information concerning the affairs, financial condition, business or properties of any Loan Party (or any of its Affiliates) which may come into possession of any Agent-Related Person.
- 15.6 Indemnification. EACH LENDER (AND BANK PRODUCT PROVIDER) SHALL INDEMNIFY AND HOLD HARMLESS AGENT-RELATED PERSONS AND ISSUING BANK INDEMNITEES, TO THE EXTENT NOT REIMBURSED BY LOAN PARTIES (AND WITHOUT LIMITING THEIR OBLIGATION TO DO SO), ON A PRO RATA BASIS, AGAINST ALL APPLICABLE CLAIMS THAT MAY BE INCURRED BY OR ASSERTED AGAINST ANY SUCH INDEMNIFIED PERSON, PROVIDED THAT ANY APPLICABLE CLAIM AGAINST AN AGENT-RELATED PERSON RELATES TO OR ARISES FROM ITS ACTING AS OR FOR AGENT (IN THE CAPACITY OF AGENT); PROVIDED THAT NO LENDER SHALL BE REQUIRED TO INDEMNIFY ANY AGENT-RELATED PERSON OR ISSUING BANK INDEMNITEE FOR APPLICABLE CLAIMS THAT A COURT OF COMPETENT JURISDICTION HAS FINALLY DETERMINED TO HAVE RESULTED FROM THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF SUCH AGENT-RELATED PERSON OR ISSUING BANK INDEMNITEE. In Agent's discretion, it may reserve for any Applicable Claims made against an Agent-Related Person or Issuing Bank Indemnitee, and may satisfy any judgment, order or settlement relating thereto, from proceeds of Collateral prior to making any distribution of Collateral proceeds to Lenders (and Bank Product Providers). If Agent or Australian Security Trustee is sued by any receiver, trustee or other Person for any alleged preference or fraudulent transfer, then any monies paid by Agent or Australian Security Trustee in settlement or satisfaction of such proceeding, together with all interest, costs and expenses (including attorneys' fees)

incurred in the defense of same, shall be promptly reimbursed to Agent or Australian Security Trustee, as applicable, by each Lender (and Bank Product Provider) to the extent of its Pro Rata Share.

15.7 Individual Capacities. As a Lender, Bank of America shall have the same rights and remedies under the Loan Documents as any other Lender, and the terms "Lenders," "Required Lenders" or any similar term shall include Bank of America in its capacity as a Lender. Agent, Lenders, Australian Security Trustee and their Affiliates may accept deposits from, lend money to, provide Bank Products to, act as financial or other advisor to, and generally engage in any kind of business with, Loan Parties and their Affiliates, as if they were not Agent, Australian Security Trustee or Lenders hereunder, without any duty to account therefor to any Lender (or Bank Product Provider). In their individual capacities, Agent, Australian Security Trustee, Lenders and their Affiliates may receive information regarding Loan Parties, their Affiliates and their Account Debtors (including information subject to confidentiality obligations), and shall have no obligation to provide such information to any Lender (or Bank Product Provider).

15.8 Successor Agent.

- (a) Resignation; Successor Agent. Agent may resign at any time by giving at least 30 days written notice thereof to Lenders and Borrowers. Required Lenders may appoint a successor to replace the resigning Agent, which successor shall be (a) a Lender or an Affiliate of a Lender; or (b) a financial institution reasonably acceptable to Required Lenders and, in either case, provided no Event of Default exists, reasonably acceptable to Borrowers. If no successor agent is appointed prior to the effective date of Agent's resignation, then Agent may appoint a successor agent that is a financial institution acceptable to it (which shall be a Lender unless no Lender accepts the role) or in the absence of such appointment, Required Lenders shall on such date assume all rights and duties of Agent hereunder, provided that Agent shall consult with Parent prior to such appointment. Upon acceptance by any successor Agent of its appointment hereunder, such successor Agent shall thereupon succeed to and become vested with all the powers and duties of the retiring Agent without further act. On the effective date of its resignation, the retiring Agent shall be discharged from its duties and obligations hereunder in its capacity as Agent but shall continue to have all rights and protections under the Loan Documents with respect to actions taken or omitted to be taken by it while Agent, including the indemnification set forth in Sections 15.6 and 10.3, and all rights and protections under this Section 15. Any successor to Bank of America by merger or acquisition of stock or this loan shall continue to be Agent hereunder without further act on the part of any Lender (or Bank Product Provider) or Loan Party.
- (b) <u>Co-Collateral Agent</u>. If appropriate under applicable law, Agent may appoint a Person to serve as a co-collateral agent or separate collateral agent under any Loan Document. Each right, remedy and protection intended to be available to Agent under the Loan Documents shall also be vested in such agent. Lenders (and Bank Product Providers) shall execute and deliver any instrument or agreement that Agent may request to effect such appointment. If any such agent shall die, dissolve, become incapable of acting, resign or be removed, then all the rights and remedies of the agent, to the extent permitted by applicable law, shall vest in and be exercised by Agent until appointment of a new agent.

15.9 Agreements Regarding Collateral and Borrower Materials.

(a) <u>Lien Releases; Care of Collateral</u>. Lenders (and Bank Product Providers) authorize Agent and Australian Security Trustee, as applicable, to release any Lien with respect to any Collateral (a) upon payment in full of the Obligations (other than unasserted contingent obligations); (b) that is the subject of a disposition or Lien that Borrowers certify in writing is a Permitted Disposition or any other disposition permitted by the Loan Documents or a Permitted Lien entitled to priority over Agent's or Australian Security Trustee's Liens (and Agent and Australian Security Trustee may rely conclusively on any such certificate without further inquiry); (c) that does not constitute a material part of the Collateral; (d) that is owned by a Loan Party if such Loan Party becomes an Excluded Subsidiary; (e) if required or permitted under the terms of any other Loan Documents, including any intercreditor agreement, or (f) subject to Section 14.1, with the consent of Required Lenders. Lenders (and Bank Product Providers) authorize Agent and Australian Security Trustee, as applicable, to subordinate its Liens to any Permitted Purchase Money Indebtedness or other Lien entitled to priority hereunder. Neither Agent nor Australian Security Trustee have any obligation to assure that any Collateral exists or is owned by a Loan Party, or is cared for, protected or insured, nor to assure that Agent's

or Australian Security Trustee's Liens have been properly created, perfected or enforced, or are entitled to any particular priority, nor to exercise any duty of care with respect to any Collateral.

- (b) <u>Possession of Collateral</u>. Agent, the Australian Security Trustee, Lenders and Bank Product Providers appoint each Lender as agent (for the benefit of Lender (and Bank Product Providers)) for the purpose of perfecting Liens in any Collateral held or controlled by such Lender, to the extent such Liens are perfected by possession or control. If any Lender obtains possession or control of any Collateral, it shall notify Agent thereof and, promptly upon Agent's request, deliver such Collateral to Agent or otherwise deal with it in accordance with Agent's instructions.
- (c) Agent shall promptly provide to Lenders, when complete, any field examination, audit, appraisal report or valuation prepared for Agent with respect to any Loan Party or Collateral ("Report") and other Borrower Materials received pursuant to Schedule 5.1 or 5.2. Reports and other Borrower Materials may be made available to Lenders by providing access to them on the Platform, but Agent shall not be responsible for system failures or access issues that may occur from time to time. Each Lender agrees (a) that Reports are not intended to be comprehensive audits or examinations, and that Agent or any other Person performing an audit or examination will inspect only limited information and will rely significantly upon Borrowers' books, records and representations; (b) that Agent makes no representation or warranty as to the accuracy or completeness of any Borrower Materials and shall not be liable for any information contained in or omitted from any Borrower Materials, including any Report; and (c) to keep all Borrower Materials confidential and strictly for such Lender's internal use, not to distribute any Report or other Borrower Materials (or the contents thereof) to any Person (except to such Lender's Participants, attorneys and accountants on a confidential basis), and to use all Borrower Materials solely for administration of the Obligations. Each Lender shall indemnify and hold harmless Agent and any other Person preparing a Report from any action such Lender may take as a result of or any conclusion it may draw from any Borrower Materials, as well as from any Applicable Claims arising as a direct or indirect result of Agent furnishing same to such Lender, via the Platform or otherwise.

15.10 Ratable Sharing.

(a) If any Lender obtains any payment or reduction of any Obligation, whether through set-off or otherwise, in excess of its ratable share of such Obligation, such Lender shall forthwith purchase from Lenders (and Bank Product Providers) participations in the affected Obligation as are necessary to share the excess payment or reduction on a pro rata basis or in accordance with Sections 2.4(b)(iv) or 2.4(b)(v), as applicable. If any of such payment or reduction is thereafter recovered from the purchasing Lender, the purchase shall be rescinded and the purchase price restored to the extent of such recovery, but without interest. Notwithstanding the foregoing, if a Defaulting Lender obtains a payment or reduction of any Obligation, it shall immediately turn over the full amount thereof to Agent for application under Section 2.3(i) and it shall provide a written statement to Agent describing the Obligation affected by such payment or reduction. No Lender shall set off against an Agent Account without Agent's prior consent.

15.11 Remittance of Payments and Collections.

- (a) Remittances Generally. All payments by any Lender to Agent shall be made by the time and on the day set forth in this Agreement, in immediately available funds. If no time for payment is specified or if payment is due on demand by Agent and request for payment is made by Agent by 1:00 p.m. on a Business Day, payment shall be made by Lender not later than 3:00 p.m. on such day, and if request is made after 1:00 p.m., then payment shall be made by 11:00 a.m. on the next Business Day. Payment by Agent to any Lender (or Bank Product Provider) shall be made by wire transfer, in the type of funds received by Agent. Any such payment shall be subject to Agent's right of offset for any amounts due from such payee under the Loan Documents.
- (b) <u>Failure to Pay</u>. If any Lender (or Bank Product Provider) fails to pay any amount when due by it to Agent pursuant to the terms hereof, such amount shall bear interest, from the due date until paid in full, at the greater of the Federal Funds Rate or the rate determined by Agent as customary for interbank compensation for two Business Days and thereafter at the Default Rate for Base Rate Loans. In no event shall Borrowers be entitled to credit for any interest paid by a Lender (or Bank Product Provider) to Agent, nor shall a Defaulting Lender be entitled to interest on amounts held by Agent pursuant to <u>Section 2.3(i)</u>.

- (c) <u>Recovery of Payments</u>. If Agent pays an amount to a Lender (or Bank Product Provider) in the expectation that a related payment will be received by Agent from a Loan Party and such related payment is not received, then Agent may recover such amount from the Lender (or Bank Product Provider). If Agent determines that an amount received by it must be returned or paid to a Loan Party or other Person pursuant to applicable law or otherwise, then Agent shall not be required to distribute such amount to any Lender (or Bank Product Provider). If any amounts received and applied by Agent to Obligations held by a Lender (or Bank Product Provider) are later required to be returned by Agent pursuant to applicable law, such Lender (or Bank Product Provider) shall pay to Agent, on demand, its share of the amounts required to be returned.
- 15.12 <u>Titles</u>. Each Lender, other than Bank of America, that is designated in connection with this credit facility as a "Joint Lead Arranger" or "Joint Bookrunner" of any kind shall have no right or duty under any Loan Documents other than those applicable to all Lenders, and shall in no event have any fiduciary duty to any Lender (or any Bank Product Provider).
- 15.13 <u>Bank of Product Providers</u>. Each U.S. Bank Product Provider and Australian Bank Product Provider, by delivery of a notice to Agent of a Bank Product, agrees to be bound by the Loan Documents, including <u>Sections 2.4(b)(iv)</u> or <u>2.4(b)(v)</u>, as applicable, <u>Section 18.9(c)</u> and <u>Section 15</u>. Each Bank Product Provider shall indemnify and hold harmless Agent-Related Persons, to the extent not reimbursed by Loan Parties, against all Applicable Claims that may be incurred by or asserted against any Agent-Related Person in connection with such provider's Bank Product Obligations.

15.14 Australian Security Trustee.

- (a) In this Agreement, any rights and remedies exercisable by, any documents to be delivered to, or any other indemnities or obligations in favor of Agent shall be, as the case may be, exercisable by, delivered to, or be indemnities or other obligations in favor of Agent (or any other Person acting in such capacity) in its capacity as Australian Security Trustee to the extent that the rights, remedies, deliveries, indemnities or other obligations relate to the Australian Security Documents or the security thereby created. Any obligations of Agent (or any other Person acting in such capacity) in this Agreement shall be obligations of Agent in its capacity as Australian Security Trustee to the extent that the obligations relate to any Australian Security Document or the security thereby created. Additionally, in its capacity as Australian Security Trustee, Agent (or any other Person acting in such capacity) shall have:
 - (i) all the rights, remedies and benefits in favor of Agent contained in the provisions of the whole of this Section 15.14;
 - (ii) all the powers of an absolute owner of the security constituted by the Australian Security Documents; and
- (iii) all the rights, remedies and powers granted to it and be subject to all the obligations and duties owed by it under the Australian Security Documents.
- (b) Each Australian Bank Product Provider, Australian Issuing Bank, Australian Revolving Lender, Australian Swing Lender, Australian Hedge Provider and the Agent (the "<u>Australian Secured Creditors</u>") appoint Australian Security Trustee under and pursuant to the terms of the Australian Security Trust Deed to act as its trustee under and in relation to the Australian Security Documents and any other Loan Document which the Australian Security Trustee may enter into from time to time and each Australian Secured Creditor (and each Australian Bank Product Provider shall be deemed to) authorize Australian Security Trustee under the terms of the Australian Security Trust Deed to exercise such rights, remedies, powers and discretions as are specifically delegated to Australian Security Trustee by the terms of the Australian Security Trust Deed and the Loan Documents together with all such rights, remedies, powers and discretions as are reasonably incidental thereto and Australian Security Trustee accepts that appointment.

- (c) Each Australian Secured Creditor hereby (and each Australian Bank Product Provider by entering into a Bank Product Provider Agreement is deemed to):
 - (i) acknowledges that they are aware of, and consent to, the terms of the Australian Security Trust Deed; and
- (ii) agrees to comply with and be bound by the Australian Security Trust Deed as a Beneficiary (as that term is defined in the Australian Security Trust Deed);
- (iii) acknowledges that it has received a copy of the Australian Security Trust Deed together with the other information which it has required in connection with the Australian Security Trust Deed, the Australian Security Documents and the other Loan Documents to which the Australian Security Trustee is a party;
- (iv) without limiting the general application of paragraph (i) above, acknowledges and agrees as specified in <u>Section 7.12</u> of the Australian Security Trust Deed and provides the indemnities as specified in <u>Section 10.3</u> of the Australian Security Trust Deed; and
- (v) without limiting the general application of paragraph (i) above, for consideration received, irrevocably appoints as its attorney each person who under the terms of the Australian Security Trust Deed is appointed an attorney of a Beneficiary (as defined in the Australian Security Trust Deed) on the same terms and for the same purposes as contained in the Australian Security Trust Deed.

This clause is executed as a deed poll in favor of Australian Security Trustee and each Beneficiary (as defined in the Australian Security Trust Deed) from time to time.

- (d) The Australian Secured Creditors (and each Australian Bank Product Provider by entering into a Bank Product Provider Agreement is deemed to) agree that at any time that the Australian Security Trustee shall be a Person other than Agent, such other Person shall have the rights, remedies, benefits and powers granted to the Agent in its capacity as Australian Security Trustee in this Agreement.
- 15.15 <u>No Third Party Beneficiaries.</u> This <u>Section 15</u> is an agreement solely among Lender (and Bank Product Providers), Agent and Australian Security Trustee, and shall survive payment in full of the Obligations. This <u>Section 15</u> does not confer any rights or benefits upon Borrowers or any other Person. As between Borrowers, Agent and Australian Security Trustee, any action that Agent or Australian Security Trustee may take under any Loan Documents or with respect to any Obligations shall be conclusively presumed to have been authorized and directed by Lenders (and Bank Product Providers).

16 COLLECTION ALLOCATION MECHANISM.

(a) On the CAM Exchange Date, (i) the Revolver Commitments shall automatically and without further act be terminated as provided in Section 8, (ii) each Lender shall become obligated to fund, within one Business Day, all participations in outstanding Swing Loans held by it (it being agreed that the CAM Exchange shall not result in a reallocation of such funding obligations, but only of the funded participations resulting therefrom) and (iii) the Lenders shall automatically and without further act be deemed to have made reciprocal purchases of interests in the Designated Obligations such that, in lieu of the interests of each Lender in the particular Designated Obligations that it shall own as of such date and immediately prior to the CAM Exchange, such Lender shall own an interest equal to such Lender's CAM Percentage in such Designated Obligations. Each Lender, each person acquiring a participation from any Lender as contemplated by Section 13.1 and each Borrower hereby consents and agrees to the CAM Exchange. Each Borrower and each Lender agrees from time to time to execute and deliver to the Agent all such promissory notes and other instruments and documents as the Agent shall reasonably request to evidence and confirm the respective interests and obligations of the Lenders after giving effect to the CAM Exchange, and each Lender agrees to surrender any promissory notes originally received by it hereunder to the Agent against delivery of any promissory notes so executed and delivered; provided that the failure of any Borrower to execute or deliver or of any Lender to accept any such promissory note, instrument or document shall not affect the validity or effectiveness of the CAM Exchange.

- (b) As a result of the CAM Exchange, on and after the CAM Exchange Date, each payment received by the Agent pursuant to any Loan Document in respect of the Designated Obligations shall be distributed to the Lenders pro rata in accordance with their respective CAM Percentages (to be redetermined as of each such date of payment or distribution to the extent required by clause (c) below), but giving effect to assignments after the CAM Exchange Date, it being understood that nothing herein shall be construed to prohibit the assignment of a proportionate part of all an assigning Lender's rights and obligations in respect of a single class of Revolver Commitments or Loans.
- (c) In the event that, after the CAM Exchange, the aggregate amount of the Designated Obligations shall change as a result of the making of a Letter of Credit Disbursement by an Issuing Bank that is not reimbursed by the applicable Borrower, then (a) each U.S. Revolving Lender or Australian Revolving Lender, as applicable, shall, in accordance with Section 2.11(e) or Section 2.12(e), as applicable, promptly purchase from the applicable Issuing Bank a participation in such Letter of Credit Disbursement in the amount equal to its Pro Rata Share of such Letter of Credit Disbursement (without giving effect to the CAM Exchange), (b) the Agent shall redetermine the CAM Percentages after giving effect to such Letter of Credit Disbursement and the purchase of participations therein by the applicable Lenders, and the Lenders shall automatically and without further act be deemed to have made reciprocal purchases of interests in the Designated Obligations such that each Lender shall own an interest equal to such Lender's CAM Percentage in each of the Designated Obligations and (c) in the event distributions shall have been made in accordance with the preceding paragraph, the Lenders shall make such payments to one another as shall be necessary in order that the amounts received by them shall be equal to the amounts they would have received had each Letter of Credit Disbursement been outstanding immediately prior to the CAM Exchange. Each such redetermination shall be binding on each of the Lenders and their successors and assigns and shall be conclusive absent manifest error.

17 WITHHOLDING TAXES.

1 7 . 1 Payments. All payments made by any Loan Party hereunder or under any note or other Loan Document will be made without setoff, counterclaim, or other defense. In addition, all such payments will be made free and clear of, and without deduction or withholding for, any present or future Taxes, except as required by applicable law. If any applicable law requires any deduction or withholding of an Indemnified Tax, then the applicable Loan Party agrees (subject to Section 2.15 of this Agreement, including Section 2.15(o)) to pay the full amount of such Indemnified Taxes and such additional amounts as may be necessary so that every payment of all amounts due under this Agreement, any note, or Loan Document, including any amount paid pursuant to this Section 17.1 after withholding or deduction for or on account of any Indemnified Taxes, will not be less than the amount provided for herein. The applicable Loan Party will furnish to Agent as promptly as possible after the date the payment of any Indemnified Tax is due pursuant to applicable law, certified copies of tax receipts evidencing such payment by the applicable Loan Party. The applicable Loan Party agrees to timely pay any present or future stamp, value added or documentary taxes or any other excise or property taxes, charges, or similar levies that arise from any payment made hereunder or from the execution, delivery, performance, recordation, or filing of, or otherwise with respect to this Agreement or any other Loan Document in accordance with applicable law.

17.2 Exemptions.

- (a) If a Lender or Participant is entitled to claim an exemption or reduction from United States withholding tax, such Lender or Participant agrees with and in favor of Agent and Parent, to deliver to Agent and Parent (or, in the case of a Participant, to the Lender granting the participation only) the following at the time or times prescribed by applicable law and at the time or times reasonably requested by Agent or Parent:
- (i) if such Lender or Participant is entitled to claim an exemption from United States withholding tax pursuant to the portfolio interest exception, (A) a statement of the Lender or Participant, signed under penalty of perjury, that it is not (1) a "bank" as described in Section 881(c)(3)(A) of the IRC, (2) a 10 percent shareholder of Parent (within the meaning of Section 881(c)(3)(B) of the IRC), or (3) a controlled foreign corporation related to

Borrowers within the meaning of Section 881(c)(3)(C) of the IRC, and (B) a properly completed and executed IRS Form W-8BEN or W-8BEN-E, as applicable, or Form W-8IMY (with proper attachments);

- (ii) if such Lender or Participant is entitled to claim an exemption from, or a reduction of, withholding tax under a United States tax treaty, a properly completed and executed copy of IRS Form W-8BEN or W-8BEN-E, as applicable;
- (iii) if such Lender or Participant is entitled to claim that interest paid under this Agreement is exempt from United States withholding Tax because it is effectively connected with a United States trade or business of such Lender, a properly completed and executed copy of IRS Form W-8ECI;
- (iv) if such Lender or Participant is entitled to claim that interest paid under this Agreement is exempt from United States withholding Tax because such Lender or Participant serves as an intermediary, a properly completed and executed copy of IRS Form W-8IMY (with proper attachments); or
- (v) a properly completed and executed copy of any other form or forms, including IRS Form W-9, as may be required under the IRC or other laws of the United States as a condition to exemption from, or reduction of, United States withholding or backup withholding Tax.
- (b) Each Lender or Participant shall provide new forms (or successor forms) upon the expiration or obsolescence of any previously delivered forms and to notify promptly Agent and Parent (or, in the case of a Participant, to the Lender granting the participation only) of any change in circumstances which would modify or render invalid any claimed exemption or reduction.
- (c) If a Lender or Participant claims an exemption from withholding Tax imposed by a jurisdiction other than the United States, such Lender or such Participant agrees with and in favor of Agent and Parent, to deliver to Agent and Parent (or, in the case of a Participant, to the Lender granting the participation only) any form or forms that may be required under the laws of such jurisdiction as a condition to exemption from, or reduction of, foreign withholding or backup withholding Tax at the time or times prescribed by applicable law and at the time or times reasonably requested by Agent or Parent. Notwithstanding anything to the contrary in the preceding sentence, the completion, execution and submission of the documentation described in the preceding sentence shall not be required if in the Lender's or the Participant's reasonable judgment such completion, execution or submission would subject such Lender or Participant to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender or Participant. Each Lender and each Participant shall provide new forms (or successor forms) upon the expiration or obsolescence of any previously delivered forms and to promptly notify Agent and Parent (or, in the case of a Participant, to the Lender granting the participation only) of any change in circumstances which would modify or render invalid any claimed exemption or reduction.
- (d) If a payment made to a Lender or Participant under any Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender or Participant were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the IRC, as applicable), such Lender shall deliver to Parent and the Agent, or, in the case of a Participant, to the Lender granting the participation, at the time or times prescribed by law and at such time or times reasonably requested by Parent or the Agent such documentation prescribed by applicable law (including as prescribed by Section 1471(b)(3)(C)(i) of the IRC) and such additional documentation reasonably requested by Parent or the Agent as may be necessary for Parent and the Agent to comply with their obligations under FATCA and to determine that such Lender or Participant has complied with its obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (d), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

17.3 Reductions.

(a) If a Lender or a Participant is subject to an applicable withholding tax, Borrowers or Agent (or, in the case of a Participant, the Lender granting the participation) may withhold from any payment to such Lender or such Participant an amount equivalent to the applicable withholding tax. If the forms or other documentation required by Section 17.2(a) or 17.2(c) are not delivered to Agent and Parent (or, in the case of a Participant, to the Lender granting the participation), then Agent or Borrowers (or, in the case of a Participant, to the Lender granting the

participation) may withhold from any payment to such Lender or such Participant not providing such forms or other documentation an amount equivalent to the applicable withholding tax.

- (b) Each Loan Party shall (subject to Section 2.15 of this Agreement, including Section 2.15(o)) indemnify Lender or Agent, within 10 days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section 17) payable or paid by such Lender or Agent or required to be withheld or deducted from a payment to such Lender or Agent and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the applicable Loan Party by such Lender (with a copy to Agent), or by the Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.
- (c) Each Lender shall severally indemnify Agent, within 10 days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that any Loan Party has not already indemnified Agent for such Indemnified Taxes and without limiting the obligation of the Loan Parties to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of Section 13.1(i) relating to the maintenance of a Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by Agent in connection with any Loan Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by Agent shall be conclusive absent manifest error. Each Lender hereby authorizes Agent to set off and apply any and all amounts at any time owing to such Lender under any Loan Document or otherwise payable by Agent to the Lender from any other source against any amount due to the Agent under this paragraph (c).
- 17.4 Refunds. If Agent or a Lender determines, in its sole discretion, that it has received a refund of any Indemnified Taxes for which it has received additional amounts pursuant to this Section 17, so long as no Default or Event of Default has occurred and is continuing, it shall pay over such refund to the applicable Loan Party (but only to the extent of payments made, or additional amounts paid, by the applicable Loan Party under this Section 17 with respect to Indemnified Taxes giving rise to such a refund), net of all out-of-pocket expenses of Agent or such Lender and without interest (other than any interest paid by the applicable Governmental Authority with respect to such a refund); provided, that the applicable Loan Party, upon the request of Agent or such Lender, agrees to repay the amount paid over to the applicable Loan Party (plus any penalties, interest or other charges imposed by the applicable Governmental Authority, other than such penalties, interest or other charges imposed as a result of the bad faith, willful misconduct or gross negligence of Agent hereunder) to Agent or such Lender in the event Agent or such Lender is required to repay such refund to such Governmental Authority. Notwithstanding anything to the contrary in this Section 17.4, in no event will the Agent or Lender be required to pay any amount to a Loan Party pursuant to this Section 17.4 the payment of which would place Agent or Lender in a less favorable net after-Tax position than the Lender or Agent would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. Notwithstanding anything in this Agreement to the contrary, this Section 17.4 shall not be construed to require Agent or any Lender to make available its tax returns (or any other information which it deems confidential) to the applicable Loan Party or any other Person.

18 **GENERAL PROVISIONS**.

- 18.1 <u>Effectiveness.</u> This Agreement shall be binding and deemed effective when executed by each Borrower, Agent, the Australian Security Trustee and each Lender whose signature is provided for on the signature pages hereof.
- 18.2 <u>Section Headings</u>. Headings and numbers have been set forth herein for convenience only. Unless the contrary is compelled by the context, everything contained in each Section applies equally to this entire Agreement.

18.3	Interpretation. Neither this Agreement nor any uncertainty or ambiguity herein shall be construed against the Lender Group or any Borrow	ver,
whether under	r any rule of construction or otherwise. On the contrary, this Agreement has been reviewed by all parties and shall be construed and interpre	eted
according to th	ne ordinary meaning of the words used so as to accomplish fairly the purposes and intentions of all parties hereto.	

- 18.4 <u>Severability of Provisions</u>. Each provision of this Agreement shall be severable from every other provision of this Agreement for the purpose of determining the legal enforceability of any specific provision.
- Bank Product Providers. Each Bank Product Provider in its capacity as such shall be deemed a third party beneficiary hereof and of the provisions of the other Loan Documents for purposes of any reference in a Loan Document to the parties for whom Agent is acting. Agent hereby agrees to act as agent for such Bank Product Providers and, by virtue of entering into a Bank Product Agreement, the applicable Bank Product Provider shall be automatically deemed to have appointed Agent as its agent and to have accepted the benefits of the Loan Documents. It is understood and agreed that the rights and benefits of each Bank Product Provider under the Loan Documents consist exclusively of such Bank Product Provider's being a beneficiary of the Liens and security interests (and, if applicable, guarantees) granted to Agent and the Australian Security Trustee and the right to share in payments and collections out of the Collateral as more fully set forth herein. In addition, each Bank Product Provider, by virtue of entering into a Bank Product Agreement, shall be automatically deemed to have agreed that Agent shall have the right, but shall have no obligation, to establish, maintain, relax, or release reserves in respect of the Bank Product Obligations and that if reserves are established there is no obligation on the part of Agent to determine or insure whether the amount of any such reserve is appropriate or not. In connection with any such distribution of payments or proceeds of Collateral, Agent shall be entitled to assume no amounts are due or owing to any Bank Product Provider unless such Bank Product Provider has provided a written certification (setting forth a reasonably detailed calculation) to Agent as to the amounts that are due and owing to it and such written certification is received by Agent a reasonable period of time prior to the making of such distribution. Agent shall have no obligation to calculate the amount due and payable with respect to any Bank Products, but may rely upon the written certification of the amount due and payable from the applicable Bank Product Provider. In the absence of an updated certification, Agent shall be entitled to assume that the amount due and payable to the applicable Bank Product Provider is the amount last certified to Agent by such Bank Product Provider as being due and payable (less any distributions made to such Bank Product Provider on account thereof). Borrowers may obtain Bank Products from any Bank Product Provider, although Borrowers are not required to do so. Each Borrower acknowledges and agrees that no Bank Product Provider has committed to provide any Bank Products and that the providing of Bank Products by any Bank Product Provider is in the sole and absolute discretion of such Bank Product Provider. Notwithstanding anything to the contrary in this Agreement or any other Loan Document, no provider or holder of any Bank Product shall have any voting or approval rights hereunder (or be deemed a Lender) solely by virtue of its status as the provider or holder of such agreements or products or the Obligations owing thereunder, nor shall the consent of any such provider or holder be required (other than in their capacities as Lenders, to the extent applicable) for any matter hereunder or under any of the other Loan Documents, including as to any matter relating to the Collateral or the release of Collateral or Guarantors.
- 18.6 <u>Debtor-Creditor Relationship</u>. The relationship between the Lenders and Agent, on the one hand, and the Loan Parties, on the other hand, is solely that of creditor and debtor. No member of the Lender Group has (or shall be deemed to have) any fiduciary relationship or duty to any Loan Party arising out of or in connection with the Loan Documents or the transactions contemplated thereby, and there is no agency or joint venture relationship between the members of the Lender Group, on the one hand, and the Loan Parties, on the other hand, by virtue of any Loan Document or any transaction contemplated therein. Each Lender and their affiliates may have economic interests that conflict with those of the Borrower.
- 18.7 <u>Counterparts; Electronic Execution</u>. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, shall be

deemed to be an original, and all of which, when taken together, shall constitute but one and the same Agreement. Delivery of an executed counterpart of this Agreement by telefacsimile or other electronic method of transmission shall be equally as effective as delivery of an original executed counterpart of this Agreement. Any party delivering an executed counterpart of this Agreement by telefacsimile or other electronic method of transmission also shall deliver an original executed counterpart of this Agreement but the failure to deliver an original executed counterpart shall not affect the validity, enforceability, and binding effect of this Agreement. The foregoing shall apply to each other Loan Document mutatis mutandis.

Revival and Reinstatement of Obligations; Certain Waivers. If any member of the Lender Group or any Bank Product Provider repays, refunds, restores, or returns in whole or in part, any payment or property (including any proceeds of Collateral) previously paid or transferred to such member of the Lender Group or such Bank Product Provider in full or partial satisfaction of any Obligation or on account of any other obligation of any Loan Party under any Loan Document or any Bank Product Agreement, because the payment, transfer, or the incurrence of the obligation so satisfied is asserted or declared to be void, voidable, or otherwise recoverable under any applicable law relating to creditors' rights, including provisions of the Bankruptcy Code relating to fraudulent transfers, preferences, or other voidable or recoverable obligations or transfers (each, a "Voidable Transfer"), or because such member of the Lender Group or Bank Product Provider elects to do so on the reasonable advice of its counsel in connection with a claim that the payment, transfer, or incurrence is or may be a Voidable Transfer, then, as to any such Voidable Transfer, or the amount thereof that such member of the Lender Group or Bank Product Provider elects to repay, restore, or return (including pursuant to a settlement of any claim in respect thereof), and as to all reasonable costs, expenses, and attorneys' fees of such member of the Lender Group or Bank Product Provider related thereto, (i) the liability of the Loan Parties with respect to the amount or property paid, refunded, restored, or returned will automatically and immediately be revived, reinstated, and restored and will exist and (ii) Agent's and Australian Security Trustee's Liens securing such liability shall be effective, revived, and remain in full force and effect, in each case, as fully as if such Voidable Transfer had never been made. If, prior to any of the foregoing, (A) Agent's and Australian Security Trustee's Liens shall have been released or terminated or (B) any pro

18.9 Confidentiality.

(a) Agent and Lenders each individually (and not jointly or jointly and severally) agree that material, non-public information regarding Parent and its Subsidiaries, their operations, assets, and existing and contemplated business plans ("Confidential Information") shall be treated by Agent and the Lenders in a confidential manner, and shall not be disclosed by Agent and the Lenders to Persons who are not parties to this Agreement, except: (i) to attorneys for and other advisors, accountants, auditors, and consultants to any member of the Lender Group and to employees, directors and officers of any member of the Lender Group (the Persons in this clause (i), "Lender Group Representatives") on a "need to know" basis in connection with this Agreement and the transactions contemplated hereby and on a confidential basis, (ii) to Subsidiaries and Affiliates of any member of the Lender Group (including the Bank Product Providers), provided that any such Subsidiary or Affiliate shall have agreed to receive such information hereunder subject to the terms of this Section 18.9, (iii) as may be required by regulatory authorities so long as such authorities are informed of the confidential nature of such information, (iv) as may be required by statute, decision, or judicial or administrative order, rule, or regulation; provided that (x) prior to any disclosure under this clause (iv), the disclosing party agrees to provide Borrowers with prior notice thereof, to the extent that it is practicable to do so and to the extent that the disclosing party is permitted to provide such prior notice to Borrowers pursuant to the terms of the applicable statute, decision, or judicial or administrative order, rule, or regulation and (y) any disclosure under this clause (iv) shall be limited to the portion of the Confidential Information as may be required by such statute, decision, or judicial or administrative order, rule, or regulation, (v) as may be agreed to in advance in writing by Borrowers, (vi) as requested or required by any Governmental Authority pursuant to any subpoena or other legal process, provided, that, (x) prior to any disclosure under this clause (vi) the disclosing party agrees to provide Borrowers with prior written notice thereof, to the extent that it is practicable to do so and to the extent that the disclosing party is permitted to provide such prior written notice to Borrowers pursuant to the terms of the subpoena or other legal process and (y) any disclosure under this clause (vi) shall be limited to the portion of the Confidential Information as may be required

by such Governmental Authority pursuant to such subpoena or other legal process, (vii) as to any such information that is or becomes generally available to the public (other than as a result of prohibited disclosure by Agent or the Lenders or the Lender Group Representatives), (viii) in connection with any assignment, participation or pledge of any Lender's interest under this Agreement, provided that prior to receipt of Confidential Information any such assignee, participant, or pledgee shall have agreed in writing to receive such Confidential Information either subject to the terms of this Section 18.9 or pursuant to confidentiality requirements substantially similar to those contained in this Section 18.9 (and such Person may disclose such Confidential Information to Persons employed or engaged by them as described in clause (i) above), (ix) in connection with any litigation or other adversary proceeding involving parties hereto which such litigation or adversary proceeding involves claims related to the rights or duties of such parties under this Agreement or the other Loan Documents; provided, that, prior to any disclosure to any Person (other than any Loan Party, Agent, any Lender, any of their respective Affiliates, or their respective counsel) under this clause (ix) with respect to litigation involving any Person (other than any Borrower, Agent, any Lender, any of their respective Affiliates, or their respective counsel), the disclosing party agrees to provide Borrowers with prior written notice thereof, and (x) in connection with, and to the extent reasonably necessary for, the exercise of any secured creditor remedy under this Agreement or under any other Loan Document. In addition, the Agent and the Lenders may disclose the existence of this Agreement and publicly available information about this Agreement to (i) any actual or prospective direct or indirect counterparty (or its advisors) to any swap or derivative transaction relating to the Borrowers or (ii) market data collectors, similar service pr

- (b) Anything in this Agreement to the contrary notwithstanding, Agent may disclose customary information concerning the terms and conditions of this Agreement and the other Loan Documents to loan syndication and pricing reporting services or in its marketing or promotional materials, with such information to consist of deal terms and other information customarily found in such publications or marketing or promotional materials and may otherwise use the name, logos, and other insignia of any Borrower or the other Loan Parties and the Revolver Commitments provided hereunder in any "tombstone" or other advertisements, on its website or in other marketing materials of the Agent.
- (c) The Loan Parties hereby acknowledge that Agent or its Affiliates may make available to the Lenders materials or information provided by or on behalf of Borrowers hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks, SyndTrak or another similar electronic system (the "Platform") and certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Loan Parties or their securities) (each, a "Public Lender"). The Loan Parties shall be deemed to have authorized Agent and its Affiliates and the Lenders to treat Borrower Materials marked "PUBLIC" or otherwise at any time filed with the SEC as not containing any material non-public information with respect to the Loan Parties or their securities for purposes of United States federal and state securities laws. All Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated as "Public Investor" (or another similar term). Agent and its Affiliates and the Lenders shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" or that are not at any time filed with the SEC as being suitable only for posting on a portion of the Platform not marked as "Public Investor" (or such other similar term).
- 18.10 <u>Survival</u>. All representations and warranties made by the Loan Parties in the Loan Documents and in the certificates or other instruments delivered in connection with or pursuant to this Agreement or any other Loan Document shall be considered to have been relied upon by the other parties hereto and shall survive the execution and delivery of the Loan Documents and the making of any Loans and issuance of any Letters of Credit, regardless of any investigation made by any such other party or on its behalf and notwithstanding that Agent, Issuing Bank, or any Lender may have had notice or knowledge of any Default or Event of Default or incorrect representation or warranty at the time any credit is extended hereunder, and shall continue in full force and effect as long as the principal of, or any accrued interest on, any Loan or any fee or any other amount payable under this Agreement is outstanding or unpaid or any Letter of Credit is outstanding and so long as the Revolver Commitments have not expired or been terminated.
- 18.11 Patriot Act. Each Lender that is subject to the requirements of the Patriot Act hereby notifies Borrowers that pursuant to the requirements of the Act, it is required to obtain, verify and record information that identifies each Borrower, which information includes the name and address of each Borrower and other information that will allow

such Lender to identify each Borrower in accordance with the Patriot Act. In addition, if Agent is required by law or regulation or internal policies to do so, it shall have the right to periodically conduct (a) Patriot Act searches, OFAC/PEP searches, and customary individual background checks for the Loan Parties and (b) OFAC/PEP searches and customary individual background checks for the Loan Parties' senior management and key principals, and each Borrower agrees to cooperate in respect of the conduct of such searches and further agrees that the reasonable costs and charges for such searches shall constitute Lender Group Expenses hereunder and be for the account of Borrowers.

- 18.12 <u>Integration</u>. This Agreement, together with the other Loan Documents, reflects the entire understanding of the parties with respect to the transactions contemplated hereby and shall not be contradicted or qualified by any other agreement, oral or written, before the date hereof. The foregoing to the contrary notwithstanding, all Bank Product Agreements, if any, are independent agreements governed by the written provisions of such Bank Product Agreements, which will remain in full force and effect, unaffected by any repayment, prepayments, acceleration, reduction, increase, or change in the terms of any credit extended hereunder, except as otherwise expressly provided in such Bank Product Agreement.
- 18.13 Parent as Agent for Borrowers. Each Borrower hereby irrevocably appoints Parent as the borrowing agent and attorney-in-fact for all Borrowers (the "Administrative Borrower") which appointment shall remain in full force and effect unless and until Agent shall have received prior written notice signed by each Borrower that such appointment has been revoked and that another Borrower has been appointed Administrative Borrower. Each Borrower hereby irrevocably appoints and authorizes the Administrative Borrower (a) to provide Agent with all notices with respect to Revolving Loans and Letters of Credit obtained for the benefit of any Borrower and all other notices and instructions under this Agreement and the other Loan Documents (and any notice or instruction provided by Administrative Borrower shall be deemed to be given by Borrowers hereunder and shall bind each Borrower), (b) to receive notices and instructions from members of the Lender Group (and any notice or instruction provided by any member of the Lender Group to the Administrative Borrower in accordance with the terms hereof shall be deemed to have been given to each Borrower), (c) to take such action as the Administrative Borrower deems appropriate on its behalf to obtain Revolving Loans and Letters of Credit and to exercise such other powers as are reasonably incidental thereto to carry out the purposes of this Agreement and (d) execute and deliver any amendments, consents, waivers or other instruments related to this Agreement and the other Loan Documents on behalf of such Borrower and any such amendment, consent, waiver or other instrument shall be binding upon and enforceable against such other Borrower to the same extent as if made directly by such Borrower (but Agent shall be entitled to require execution thereof by all Borrowers). It is understood that the handling of the U.S. Loan Accounts and the Australian Loan Accounts and Collateral in a combined fashion, as more fully set forth herein (and subject to the terms herein), is done solely as an accommodation to the applicable Borrowers in order to utilize the collective borrowing powers of the applicable Borrowers in the most efficient and economical manner and at their request, and that Lender Group shall not incur liability to any Borrower as a result hereof. Each Borrower expects to derive benefit, directly or indirectly, from the handling of the U.S. Loan Accounts and the Australian Loan Accounts and the Collateral in a combined fashion, as more fully set forth herein (and subject to the terms herein), since the successful operation of each Borrower is dependent on the continued successful performance of the integrated group. To induce the Lender Group to do so, and in consideration thereof, (x) each U.S. Borrower hereby jointly and severally agrees to indemnify each member of the Lender Group and holds each member of the Lender Group harmless against any and all liability, expense, loss or claim of damage or injury, made against the Lender Group by any U.S. Borrower or by any third party whosoever, arising from or incurred by reason of (i) the handling of the U.S. Loan Account and Collateral securing the U.S. Obligations as herein provided, or (ii) the Lender Group's relying on any instructions of the Administrative Borrower, except that U.S. Borrowers will have no liability to the relevant Agent-Related Person or Lender-Related Person under this Section 18.13 with respect to any liability that has been finally determined by a court of competent jurisdiction to have resulted solely from the bad faith, gross negligence or willful misconduct of such Agent-Related Person or Lender-Related Person, as the case may be, and (y) Australian Borrowers hereby agree to indemnify each member of the Lender Group and holds each member of the Lender Group harmless against any and all liability, expense, loss or claim of damage or injury, made against the Lender Group by Australian Borrowers or by any third party whosoever, arising from or incurred by reason of (i) the handling of the Australian Loan Account and Collateral securing the Australian Obligations as herein provided, or (ii) the Lender Group's relying on any instructions of the Administrative Borrower, except that Australian Borrowers will have no liability to the relevant Agent-Related Person or Lender-Related Person under this Section with respect to any liability that has been finally determined by a court of competent jurisdiction to have resulted solely from the

bad faith, gross negligence or willful misconduct of such Agent-Related Person or Lender-Related Person, as the case may be.

18.14 **Public Offer Representations.**

- (a) Merrill Lynch, Pierce, Fenner & Smith Incorporated in its capacity as Joint Lead Arranger (" <u>Lead Left Arranger"</u>) makes the following representations, warranties and agreements:
- (i) on behalf of the Borrowers, such Lead Left Arranger has made invitations to become a Lender under the Existing Syndicated Facility Agreement and this Agreement to at least ten parties (each of whom has been disclosed to the Parent);
- (ii) at least ten of the parties to whom such Lead Left Arranger has made invitations pursuant to clause (i) above are not, as at the date the invitations were made, to the knowledge of the relevant officers of such Lead Left Arranger and in reliance on the representations made by the Borrowers to it, Associates of any of the others of those ten invitees;
- (iii) such Lead Left Arranger has not made and will not make offers of invitations referred to in clause (i) above to parties whom any Borrower have notified it, are Offshore Associates of a Borrower; and
- (iv) such Lead Left Arranger will provide to the Borrowers, when reasonably requested, any factual information in its possession or which it is reasonably able to provide to assist the Borrowers to demonstrate (based upon tax advice received by the Borrowers) that s128F of the Australian Tax Act has been satisfied, solely to the extent where doing so will not, in such Lead Left Arranger's reasonable opinion, breach any law or regulation or any duty of confidence applicable to it.
- (b) Each of the Lenders as of the date of the making of the Revolving Loans (or other extension of credit) under the Existing Syndicated Facility Agreement and the Closing Date make the following representations, warranties and agreements:
- (i) at the time such Lender received the invitation (if any) as contemplated by clause (a)(i) above, it was carrying on the business of providing finance, or investing or dealing in securities, in the course of operating in financial markets; and
- (ii) such Lender will provide to the Borrowers, when reasonably requested, any factual information in its possession or which it is reasonably able to provide to assist the Borrowers to demonstrate (based upon tax advice received by the Borrowers) that s128F of the Australian Tax Act has been satisfied, solely to the extent where doing so will not, in such Lender's reasonable opinion, breach any law or regulation or any duty of confidence applicable to it; and
- (iii) such Lender will not assign or transfer rights hereunder that relate to an Australian Borrower to a person whom the officers of the relevant existing Lender involved on a day to day basis in the administration of the Lender's participation hereunder know to be an Offshore Associate of the relevant Australian Borrower.
- (c) The Borrowers confirm that none of the potential invitees whose names were disclosed to it by the Lead Left Arranger before the date of the Existing Syndicated Facility Agreement and this Agreement were known or suspected by it to be an Offshore Associate of any Borrower and that at least ten of these potential invitees disclosed to it by the Lead Left Arranger before the date of the Existing Syndicated Facility Agreement and this Agreement were not known or suspected by it to be an Associate of any of the others of those ten invitees.
- (d) If for any reason, the requirements of section 128F of the Australian Tax Act have not been satisfied in relation to interest payable on Loans (except to an Offshore Associate of any Borrower), then on request by an Agent, a Joint Lead Arranger or any Borrower, such Borrower and the Agent shall co-operate and take steps reasonably requested with a view to satisfying those requirements:
- (i) where the Lead Left Arranger breached <u>Section 18.14(a)</u> or where a Lender breached <u>Section 18.14(b)</u>, at the costs of the Lead Left Arranger or Lender (as relevant); or

- (ii) in all other cases, at the cost of such Borrower.
- 18.15 <u>Acknowledgement and Consent to Bail-In of EEA Financial Institutions</u>. Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any Lender that is an EEA Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the write-down and conversion powers of an EEA Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:
- (a) the application of any Write-Down and Conversion Powers by an EEA Resolution Authority to any such liabilities arising hereunder which may be payable to it by any Lender that is an EEA Financial Institution; and
 - (b) the effects of any Bail-in Action on any such liability, including, if applicable:
 - (i) a reduction in full or in part or cancellation of any such liability;
- (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such EEA Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or
- (iii) the variation of the terms of such liability in connection with the exercise of the write-down and conversion powers of any EEA Resolution Authority.

As used herein, the following terms have the following meanings:

"Bail-In Action" means the exercise of any Write-Down and Conversion Powers by the applicable EEA Resolution Authority in respect of any liability of an EEA Financial Institution.

"Bail-In Legislation" means, with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule.

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" means any public administrative authority or any person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"EU Bail-In Legislation Schedule" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"<u>Write-Down and Conversion Powers</u>" means, with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule.

18.16 <u>Lender Representation</u>. Each Lender as of the Closing Date represents and warrants as of the Closing Date that such Lender is not and will not be (a) an employee benefit plan subject to Title I of ERISA, (b) a plan or account subject to Section 4975 of the IRC, (c) an entity deemed to hold Plan Assets of any such plans or accounts or (d) a "governmental plan" within the meaning of Section 3(32) of ERISA. No portion of any Loan shall be funded or held with Plan Assets.

19 INTERCREDITOR AGREEMENT

The terms of this Agreement and the other Loan Documents (other than the Intercreditor Agreement), any Lien granted to the Agent pursuant to any Loan Document and the exercise of any right or remedy by the Agent hereunder are subject to the provisions of the Intercreditor Agreement. In the event of any inconsistency between the provisions of this Agreement and the Loan Documents (other than the Intercreditor Agreement), on the one hand, and the Intercreditor Agreement, on the other, the provisions of the Intercreditor Agreement shall supersede the provisions of this Agreement and the Loan Documents (other than the Intercreditor Agreement). Without limiting the generality of the foregoing, and notwithstanding anything herein to the contrary, all rights and remedies of the Agent (and the Lender Group) shall be subject to the terms of the Intercreditor Agreement, and until the Discharge of Fixed Asset Obligations (as defined in the Intercreditor Agreement), (i) except for express requirements of this Agreement, no Loan Party shall be required hereunder or under any other Loan Document to take any action in respect of the Fixed Asset Priority Collateral that is inconsistent with such Loan Party's obligations under the Senior Secured Notes Documents except if otherwise provided in the Intercreditor Agreement and (ii) any obligation of any Loan Party hereunder or under any other Loan Document with respect to the delivery or control of any Fixed Asset Priority Collateral, the novation of any lien on any certificate of title, bill of lading or other document, the giving of any notice to any bailee or other Person, the provision of voting rights or the obtaining of any consent of any Person, in each case in respect of any Fixed Asset Priority Collateral shall be deemed to be satisfied if such Loan Party complies with the requirements of the similar provision of the applicable Senior Secured Notes Document.

[Signature pages to follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed and delivered as of the date first above written.

U.S. BORROWERS

CLEVELAND-CLIFFS INC.
NORTHSHORE MINING COMPANY
CLIFFS MINING COMPANY
THE CLEVELAND-CLIFFS IRON COMPANY

/s/ James D. Graham

Name: James D. Graham

Title: Executive Vice President, Chief Legal Officer and

Secretary

UNITED TACONITE LLC LAKE SUPERIOR & ISHPEMING RAILROAD COMPANY

/s/ James D. Graham

Name: James D. Graham

Title: Secretary

AUSTRALIAN BORROWERS

Executed by Cliffs Natural Resources Pty Ltd ACN 112 437 180 in accordance with section 127 of the Corporations Act 2001:						
/s/ Stuart Taylor	/s/ Jason Grace					
Director/ Company Secretary	Director					
STUART TAYLOR	JASON GRACE					
Name of Director/ Company Secretary	Name of Director					
(BLOCK LETTERS)	(BLOCK LETTERS)					
Executed by Cliffs Asia Pacific Iron Ore Pty Ltd ACN 001 892 995 in accordance with section 127 of the <i>Corporations Act 2001:</i>						
/s/ Stuart Taylor	/s/ Jason Grace					
Director/ Company Secretary	Director					
STUART TAYLOR	JASON GRACE					
Name of Director/ Company Secretary	Name of Director					
(BLOCK LETTERS)	(BLOCK LETTERS)					

Bank Of America, N.A., as Agent, as Australian Security Trustee, as Joint Lead Arranger, as Joint Book Runner, and as a Lender

By: /s/ Thomas H. Herron

Name: Thomas H. Herron

Title: Senior Vice President

PNC Bank, National Association, as a Lender

By: /s/ Robert T. Brown
Name: Robert T. Brown
Title: Vice President

Citizens Bank, N.A.

By: /s/ Debra L. McAllonis
Name: Debra L. McAllonis
Title: Senior Vice President

DEUTSCHE BANK AG NEW YORK BRANCH

By: /s/ Alicia Schug

Name: Alicia Schug
Title: Vice President
If second signatures is needed:
By: /s/ Marguerite Sutton

Name: Marguerite Sutton
Title: Vice President

Regions Bank

By: /s/ Stephen J. McGreevy

Name: Stephen J. McGreevy
Title: Managing Director

CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH

By: /s/ Mikhail Faybusovich
Name: Mikhail Faybusovich

Title: Authorized Signatory

By: /s/ Christopher Zybrick

Name: Christopher Zybrick
Title: Authorized Signatory

GOLDMAN SACHS BANK USA

By: /s/ Josh Rosenthal
Name: Josh Rosenthal

Title: Authorized Signatory

GOLDMAN SACHS LENDING PARTNERS LLC

By: /s/ Josh Rosenthal

Name: Josh Rosenthal
Title: Authorized Signatory

The Huntington National Bank By: /s/ Paul Weybrecht

Name: Paul Weybrecht
Title: Senior Vice President

JEFFERIES FINANCE LLC

By: /s/ John Koehler

Name: John Koehler

Title: Senior Vice President

JFIN BUSINESS CREDIT FUND I LLC

By: /s/ John Koehler

Name: John Koehler

Title: Senior Vice President

EXHIBIT A-1

FORM OF ASSIGNMENT AND ACCEPTANCE AGREEMENT

This ASSIGNMENT AND ACCEPTANCE AGREEMENT (" Assignment Agreement") is entered into as of [I] between [I] ("Assignor") and [I] ("Assignee").

Reference is made to the Amended and Restated Syndicated Facility Agreement dated as of February 28, 2018 (as amended, restated, amended and restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), by and among CLEVELAND-CLIFFS INC., an Ohio corporation, as parent ("Parent"), the Subsidiaries of Parent identified on the signature pages thereto (such Subsidiaries, together with Parent, the "Borrowers"), the lenders party thereto and BANK OF AMERICA, N.A., a national banking association, as agent for each member of the Lender Group and the Bank Product Providers (the "Agent"). Capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in the Credit Agreement.

- 1. Assignor hereby assigns to Assignee and Assignee hereby purchases and assumes from Assignor (a) a principal amount of \$_____ of Assignor's outstanding Revolving Loans and \$_____ of Assignor's participations in Letters of Credit and (b) the amount of \$____ of Assignor's Revolver Commitment (which represents _____% of the total Revolver Commitments) (the foregoing items being, collectively, "Assigned Interest"), together with an interest in the Loan Documents corresponding to the Assigned Interest. This Assignment Agreement shall be effective as of the date ("Effective Date") indicated in the corresponding Assignment Notice attached hereto as Annex I delivered to Agent, provided that such Assignment Notice is executed by Assignor, Assignee and, if applicable, Agent, each Issuing Bank and the Borrowers. From and after the Effective Date, Assignee hereby expressly assumes, and undertakes to perform, all of Assignor's obligations in respect of the Assigned Interest, and all principal, interest, fees and other amounts which would otherwise be payable to or for Assignor's account in respect of the Assigned Interest shall be payable to or for Assignee's account, to the extent such amounts accrue on or after the Effective Date.
- 2. Assignor (a) represents that as of the date hereof, prior to giving effect to this assignment, its Revolver Commitment is \$_____ and the outstanding balance of its Revolving Loans and participations in Letters of Credit is \$_____; (b) makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with the Credit Agreement or the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Credit Agreement or any other instrument or document furnished pursuant thereto, other than that Assignor is the legal and beneficial owner of the interest being assigned by it hereunder and that such interest is free and clear of any adverse claim; and (c) makes no representation or warranty and assumes no responsibility with respect to the financial condition of Borrowers or the performance by Borrowers of their obligations under the Loan Documents. [Assignor is attaching the promissory note[s] held by it and requests that Agent exchange such note[s] for new promissory notes payable to Assignee [and Assignor].]
- 3. Assignee (a) represents and warrants that it is legally authorized to enter into this Assignment Agreement; (b) confirms that it has received copies of the Credit Agreement and such other Loan Documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment Agreement; (c) agrees that it shall, independently and without reliance upon Assignor and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents; (d) confirms that it is an Eligible Transferee; (e) appoints and authorizes Agent to take such action as agent on its behalf and to exercise such powers under the Credit Agreement as are delegated to Agent by the terms thereof, together with such powers as are incidental thereto; (f) agrees that it will observe and perform all obligations that are required to be performed by it as a "Lender" under the Loan Documents; (g) represents and warrants that the assignment evidenced hereby will not result in a non-exempt "prohibited transaction" under Section 406 of ERISA; and (h) attaches the forms prescribed by the Internal Revenue Service of the United States certifying as to the Assignee's status for purposes of determining exemption from United States withholding taxes with respect to all payments to be made to the Assignee under the Credit Agreement or such other documents as are necessary to indicate that all such payments are subject to such rates at a rate reduced by an applicable tax treaty.

law, it shall be		re only to the extent of such invalidity and the remaining provisions of this Assignment Agreement shall remain in full force and effect.
5. first-class mail.		n notice or other communication hereunder shall be in writing, shall be sent by messenger, by telecopy or facsimile transmission, or by deemed given when sent and shall be sent as follows:
	(a)	If to Assignee, to the following address (or to such other address as Assignee may designate from time to time):
	(b)	If to Assignor, to the following address (or to such other address as Assignor may designate from time to time):
Payme	ents here	eunder shall be made by wire transfer of immediately available Dollars as follows:
If to As	ssignee,	to the following account (or to such other account as Assignee may designate from time to time):
		ABA No
		Account NoReference:
If to As	ssignor, i	to the following account (or to such other account as Assignor may designate from time to time):
		ABA No
		Account NoReference:
		IDEMAINDED OF THIS DAGE INTENTIONALLY LEFT BLANKI
		[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]
		A-1-2

IN WITNESS WHEREOF, the parties hereto have caused this Assignment Agreement to be executed by their respective officers, as of the first date written above.						
		[NAME OF ASSIGNOR] as Assignor				
		By:				
			Name:			
			Title:			
		[NAME OF ASSIGNEE] as Assignee				
		By:				
			Name:			
			Title:			
	A-1-3					

ANNEX I TO ASSIGNMENT AND ACCEPTANCE

ASSIGNMENT NOTICE

Reference is made to the Amended and Restated Syndicated Facility Agreement dated as of February 28, 2018 (as amended, restated, amended and restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), by and among CLEVELAND-CLIFFS INC., an Ohio corporation, as parent ("Parent"), the Subsidiaries of Parent identified on the signature pages thereto (such Subsidiaries, together with Parent, the "Borrowers"), the lenders party thereto and BANK OF AMERICA, N.A., a national banking association, as agent for each member of the Lender Group and the Bank Product Providers (the "Agent"). Capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in the Credit Agreement. Assignor hereby notifies Borrowers and Agent of Assignor's intent to assign to Assignee pursuant to the Assignment Agreement (a) a principal amount of \$_____ of Assignor's outstanding Revolving Loans and \$_____ of Assignor's participations in Letters of Credit and (b) the amount of \$_____ of Assignor's Revolver Commitment (which represents _____% of the total Revolver Commitments) (the foregoing items being, collectively, the "Assigned Interest"), together with an interest in the Loan Documents corresponding to the Assigned Interest. This Assignment Agreement shall be effective as of the date ("Effective Date") indicated below, provided this Assignment Notice is executed by Assignor, Assignee and, if applicable, Agent, each Issuing Bank and Borrowers. Pursuant to the Assignment Agreement, Assignee has expressly assumed all of Assignor's obligations under the Credit Agreement to the extent of the Assigned Interest, as of the Effective Date. For purposes of the Credit Agreement, Agent shall deem Assignor's Revolver Commitment to be reduced by \$_____, and Assignee's Revolver Commitment to be increased by \$ The address of Assignee to which notices and information are to be sent under the terms of the Credit Agreement is: The address of Assignee to which payments are to be sent under the terms of the Credit Agreement is shown in the Assignment Agreement. This Assignment Notice is being delivered to Borrowers and Agent pursuant to Section 13.1 of the Credit Agreement. Please acknowledge your acceptance of this Assignment Notice by executing and returning to Assignee and Assignor a copy of this Assignment Notice.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

bove.	
	[NAME OF ASSIGNOR]
	as Assignor
	By: Name: Title:
	Title.
	[NAME OF ASSIGNEE]
	as Assignee By:
	Name: Title:
ACCEPTED AS OF THE DATE FIRST WRITTEN ABOVE	
BANK OF AMERICA, N.A., as Agent] ¹	
Name: Title:	
[ISSUING BANK], as Issuing Bank] ²	
By: Name:	
Title:	
INSERT BORROWERS' SIGNATURE BLOCKS]] ³	
¹ Include to the extent required by Section 13(a)(i) of the Credit Agreement.	
² Include to the extent required by Section 13(a)(i) of the Credit Agreement.	
³ Include to the extent required by Section 13(a)(i) of the Credit Agreement.	
	A-1-5

IN WITNESS WHEREOF, the parties hereto have caused this Assignment Notice to be executed by their respective officers, as of the first date written

EXHIBIT B-1

FORM OF BORROWING BASE CERTIFICATE

[•] [•], 20[•]

Reference is made to that certain Amended and Restated Syndicated Facility Agreement, dated as of February 28, 2018 (as amended, restated, amended and restated, supplemented or otherwise modified and in effect on the date hereof, the "Credit Agreement"), by and among CLEVELAND-CLIFFS INC., an Ohio corporation, as parent ("Parent"), the Subsidiaries of Parent identified on the signature pages thereto (such Subsidiaries, together with Parent, the "Borrowers"), the lenders party thereto and BANK OF AMERICA, N.A., a national banking association, as agent for each member of the Lender Group and the Bank Product Providers (the "Agent"). Capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in the Credit Agreement.

Pursuant to <u>Section 5.2</u> of the Credit Agreement, the undersigned Responsible Officer of the Parent hereby certifies, solely in his/her capacity as a Responsible Officer and not in any individual capacity, that the items set forth on <u>Schedule I</u> hereto, calculated in accordance with the terms and definitions set forth in the Credit Agreement for such items, are true and correct in all material respects.

For the avoidance of doubt, nothing contained in this Borrowing Base Certificate shall constitute a waiver, modification or limitation in the terms or conditions set forth in the Credit Agreement.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

In witness whereof, this Borrowing Base Certificate is executed by the undersigned as of the date first written above.

	CLEVELAND-CLIFFS INC., an Ohio corporation, as Parent By:	
	Name: Title:	
P.4.2		

0-	L	1	_ 1
Sc	nec	ıuı	еı

[Attached]

B-1-3

EXHIBIT B-2

FORM OF Australian commitment Reallocation Notice

[•] [•], 20[•]

Cleveland-Cliffs Inc., an Ohio corporation ("Parent"), pursuant to Section 2.2 of that certain Amended and Restated Syndicated Facility Agreement, dated as of February 28, 2018 (as amended, restated, amended and restated, supplemented or otherwise modified and in effect on the date hereof, the "Credit Agreement"), by and among Parent, the Subsidiaries of Parent identified on the signature pages thereto (such Subsidiaries, together with Parent, the "Borrowers"), the lenders party thereto and Bank of America, N.A., a national banking association, as agent for each member of the Lender Group and the Bank Product Providers (the "Agent") provides this notice (the "Notice") to the Agent to certify the following:

		This Nation is an Australian Commitment Declination Nation delivered in accordance with and action in a requirement
of, <u>S</u>	ection 2.2(a).	 This Notice is an Australian Commitment Reallocation Notice delivered in accordance with, and satisfying the requirement
Commitment	(b) s shall be \$[_	Pursuant to <u>Section 2.2(a)</u> , the amount of the Australian Revolver Commitments to be permanently reallocated as U.S. Revolving].
Maximum Re	(c) evolver Amou	After giving effect to such reallocation, the U.S. Maximum Revolver Amount shall be \$[] and the Australian at shall be \$[].
	(d)	The reallocation requested by this Notice is requested to occur on [DATE].
effect to such	(e) n reallocation.	No Default or Event of Default has occurred and is continuing on the date hereof, nor will any thereof occur immediately after giving
U.S. Revolve	(f) er Commitmer	Immediately after giving effect to such reallocation, the aggregate Australian Revolver Commitments shall not exceed the aggregate its.
Unle Agreement.	ess otherwise	defined herein, capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in the Credit
The Parent.	undersigned	certifies that he/she is a Responsible Officer of Parent, and that as such he/she is authorized to execute this certificate on behalf of
		[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]
		B-2-1

In witness whereof	this Notice is executed b	v the undersigned as of	f the date first written above.

	CLEVEI By:	LAND-CLIFFS INC., an Ohio corporation, as Parent
	·	Name:
		Title:
B-2	1-2	

EXHIBIT B-3

FORM OF BANK PRODUCT PROVIDER AGREEMENT

[•] [•], 20[•]

To: Bank of America, N.A. 135 S. LaSalle Street; Suite 925 Chicago, IL 60603

Attn: Peter Walther Tel No.: 312-904-8225 Fax No.: 312-453-5555

Email: peter.walther@baml.com

Re: Bank Product Obligations

Reference is made to that certain Amended and Restated Syndicated Facility Agreement, dated as of February 28, 2018 (as amended, restated, amended and restated, supplemented or otherwise modified and in effect on the date hereof, the "Credit Agreement"), by and among CLEVELAND-CLIFFS INC., an Ohio corporation, as parent ("Parent"), the Subsidiaries of Parent identified on the signature pages thereto (such Subsidiaries, together with Parent, the "Borrowers"), the lenders party thereto and BANK OF AMERICA, N.A., a national banking association, as agent for each member of the Lender Group and the Bank Product Providers (the "Agent"). Each of the undersigned hereby acknowledge the following as of the date hereof:

The undersigned is [a Lender][an Affiliate of [insert name of Lender], which is a Lender], and is providing to [insert name(s)], which [is a Borrower][are Borrowers][a Subsidiary of a Borrower], the following Bank Product(s) that are permitted under the terms of the Credit Agreement:

[insert description of Bank Product(s)].

The maximum amount to be secured by the Collateral with respect to such Bank Product is \$[insert amount]. The methodology to be used in calculating such amount is: [describe].

The undersigned hereby agrees to be bound by the provisions of <u>Sections 15.13</u> and <u>18.5</u> of the Credit Agreement, and the provisions of the Credit Agreement referred to in such Sections, as provided therein.

Upon execution of this letter agreement by the undersigned and acceptance by Agent, the undersigned shall be a Bank Product Provider with regard to the Bank Product Obligations described above (in each case, subject to the terms of the Credit Agreement and the other Loan Documents).

This letter agreement may not be amended or any provision hereof waived or modified except by an instrument in writing signed by each of the parties hereto. This letter agreement may be executed in any number of counterparts, each of which shall be an original and all of which, when taken together, shall constitute one agreement. Delivery of an executed counterpart of a signature page of this letter agreement by telecopier, facsimile or other electronic transmission (i.e. a "pdf" or "tiff") shall be effective as delivery of a manually executed counterpart hereof.

Unless otherwise defined herein, capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in the Credit Agreement.

This letter agreement shall be governed by, and construed in accordance with, the laws of the State of New York.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

		Very truly yours,	
		[insert name of Bank Product Provider]	
		Ву:	
		Name:	
		Title:	
	WLEDGED AND AGREED TODAY OF, 20:		
BANK Cas Ager	DF AMERICA, N.A., nt		
[insert n	name of Bank Product Provider]		
By:			
	Name:		
	Title:		

EXHIBIT C-1

FORM OF COMPLIANCE CERTIFICATE

[on Parent's letterhead]

To: Bank of America, N.A. 135 S. LaSalle Street; Suite 925

Chicago, IL 60603 Attn: Peter Walther Tel No.: 312-904-8225 Fax No.: 312-453-5555

Email: peter.walther@baml.com

Re: Compliance Certificate dated [I], 20[I]

Ladies and Gentlemen:

Reference is made to that certain Amended and Restated Syndicated Facility Agreement dated as of February 28, 2018 (as amended, restated, supplemented, or otherwise modified from time to time, the "Credit Agreement") by and among CLEVELAND-CLIFFS INC., an Ohio corporation, as parent ("Parent"), the Subsidiaries of Parent identified on the signature pages thereto (such Subsidiaries, together with Parent, the "Borrowers"), the lenders party thereto (the "Lenders") and BANK OF AMERICA, N.A., a national banking association ("Bank of America"), as agent for each member of the Lender Group and the Bank Product Providers. All initially capitalized terms used herein and in the schedules hereto and not otherwise defined herein shall have the meanings ascribed to them in the Credit Agreement.

Pursuant to Section 5.1 and Section 5.2 of the Credit Agreement, the undersigned officer of Parent hereby certifies as of the date hereof that:

- 1. The financial information of Parent and its Subsidiaries furnished in Schedule 1 attached hereto, has been prepared in accordance with GAAP (except, in the case of unaudited financial statements, for year-end audit adjustments and the lack of footnotes), and fairly presents in all material respects the consolidated financial condition and the results of operations of the Loan Parties and their Subsidiaries as of the date set forth therein.
- 2. Such officer has reviewed the terms of the Credit Agreement and has made, or caused to be made under his/her supervision, a review in reasonable detail of the transactions and financial condition of Parent and its Subsidiaries during the accounting period covered by the financial statements delivered pursuant to Section 5.1 of the Credit Agreement.
- 3. Such review has not disclosed the existence on and as of the date hereof, and the undersigned does not have knowledge of the existence as of the date hereof, of any event or condition that constitutes a Default or Event of Default, except for such conditions or events listed on Schedule 2 attached hereto, in each case specifying the nature and period of existence thereof and what action Parent and/or its Subsidiaries have taken, are taking, or propose to take with respect thereto.
- 4. The financial calculations, analyses and information set forth on <u>Schedule 3</u> attached hereto are delivered in compliance with the applicable provisions of the Credit Agreement requiring delivery thereof.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, this Compliance Certificate is executed by the undersigned as of the date first written above.

	CLEVELAND-CLIFFS INC., an Ohio corporation, as Parent By:
	Name:
	Title:
SCHEDULE 1	
Financial Informat	tion

C-1-2

SCHEDULE 2

Default or Event of Default

SCHEDULE 3

Financial Calculations⁴

Fixed Charge (Ratio	Coverage	
(a)	EBITDA ⁷ for the Relevant Period ended on the Computation Date	\$
(b)	Capital Expenditures made (to the extent not already incurred in a prior period) or incurred during the Relevant Period (excluding any capitalized interest with respect thereto) to the extent required to be paid in cash (provided, however, that any Capital Expenditures made or incurred during such period in connection with the Toledo, Ohio hot-briquetted iron production plant shall not be included in the calculation of this clause (b) solely to the extent that such Capital Expenditures are funded with balance sheet cash that constitutes net cash proceeds of (x) the Senior Secured Notes or (y) the Convertibles Notes).	
(c)	Federal, state and local income taxes paid in cash during the Relevant Period (net of federal, state and local tax refunds received in cash during such period)	\$
(d)	Result of line (a) minus line (b) minus line (c)	\$
(e)	Fixed Charges ⁸	\$
(f)	Ratio of line (d) to line (e)	Actual::1.00
		Required: ≥ 1.00:1.00 ⁹
4 The fellowing and	forth in this Schedule are summaries and are qualified in their entirety by reference to the full text	of the control of the

 $^{^{7}}$ Insert amount from line 2(d) below.

⁸ Insert amount from line 3(d) below.

⁹ Commencing on the date on which a Financial Covenant Period begins and measured as of the end of the fiscal quarter immediately preceding the date on which a Financial Covenant Period first begins and as of each fiscal quarter end thereafter during such Financial Covenant Period, the Parent and its Subsidiaries on a consolidated basis will have a Fixed Charge Coverage Ratio, measured on a quarter-end basis, of at least 1.00:1.00 for the 12-month period ending as of the end of each fiscal quarter.

2. EBITDA

(a) (b)

(c)

	Parent's consolidated net earnings (or loss),	\$
	<i>minus</i> (without duplication) the sum of items (i) through (iii) below of Parent for the Relevant Period to the extent included in determining consolidated net earnings (or loss) for such period:	\$
i.	any extraordinary, unusual or non-recurring gains,	\$
ii.	interest income,	\$
iii.	exchange, translation or performance gains relating to any hedging transactions or foreign currency fluctuations, and	\$
	plus (without duplication), the sum of the following items (i) through (xiii) of Parent for the Relevant Period to the extent included in determining consolidated net earnings (or loss) for such period	\$
i.	any non-cash extraordinary, unusual, or non-recurring losses,	\$
ii.	the aggregate of the interest expense of Parent for the Relevant Period, determined on a consolidated basis in accordance with GAAP,	\$
iii.	tax expense based on income, profits or capital, including federal, foreign, state, franchise and similar taxes (and for the avoidance of doubt, specifically excluding any sales taxes or any other taxes held in trust for a Governmental Authority),	\$
iv.	depreciation and amortization for such period,	\$
V.	with respect to any Permitted Acquisition after the Closing Date, costs, fees, charges, or expenses consisting of out-of-pocket expenses owed by Parent or any of its Subsidiaries to any Person for services performed by such Person in connection with such Permitted Acquisition incurred within 180 days of the consummation of such Permitted Acquisition, up to an aggregate amount (for all such items in this clause (v)) for such Permitted Acquisition not to exceed the greater of (A) \$2,500,000 and (B) 2.50% of the Purchase Price of such Permitted Acquisition,	\$
vi.	(A) purchase accounting adjustments, including, without limitation, a dollar for dollar adjustment for that portion of revenue that would have been recorded in the relevant period had the balance of deferred revenue (unearned income) recorded on the closing balance sheet and before application of purchase accounting not been adjusted downward to fair value to be recorded on the opening balance sheet in accordance with GAAP purchase accounting rules; and (B) non-cash adjustments in accordance with GAAP purchase accounting rules under Statement of Financial Accounting Standards No. 805, in the event that such an adjustment is required by Parent's independent auditors, in each case, as determined in accordance with GAAP,	\$
vii.	costs and expenses incurred during such period in connection with the Canadian Restructuring or a restructuring of APIO in an aggregate amount for all such costs and expenses incurred during such period not to exceed \$50,000,000 for such period,	\$

		VIII	expense), or other non-cash expenses or charges, arising from the sale or issuance of Equity Interests, the granting of stock options, and the granting of stock appreciation rights and similar arrangements (including any repricing, amendment, modification, substitution, or change of any such Equity Interests, stock option, stock appreciation rights, or similar arrangements) minus the amount of any such expenses or charges when paid in cash to the extent not deducted in the computation of net earnings (or loss),	\$
		ix	one-time non-cash restructuring charges,	\$
		x	. non-cash exchange, translation, or performance losses relating to any hedging transactions or foreign currency fluctuations,	\$
		xi	. non-cash losses on sales of fixed assets or write-downs of fixed or intangible assets (excluding ABL Collateral),	\$
		xii	any non-cash loss, charge or expense (but only to the extent not relating to the ABL Collateral), and	\$
		xiii	financing fees, costs, accruals, payments and expenses (including rationalization, legal, tax, structuring and other costs and expenses and non-operating or non-recurring professional fees, costs and expenses related thereto), related to, to the extent permitted under this Agreement, any Permitted Investments, Permitted Dispositions (other than in the ordinary course of business), issuances of Equity Interests and issuances, amendments, modifications, refinancings or repayments of Permitted Indebtedness (in each case, regardless of whether or not consummated),	\$
		(d)	EBITDA: (a) - (b) + (c) =	\$
2.	Fixed Charges			
		(a)	the aggregate of the interest expense of Parent for the Relevant Period, determined on a consolidated basis in accordance with GAAP but subject to Section 1.2 of the Credit Agreement ("Interest Expense") accrued (other than (x) interest paid-in-kind, amortization of financing fees, and other non-cash Interest Expense and (y) make-whole payments, premiums or similar payments related to the prepayment or extinguishment of Indebtedness) during such period to the extent required to be paid in cash, <i>plus</i>	\$
		(b)	Scheduled principal payments in respect of all Indebtedness for borrowed money or letters of credit of Parent, determined on a consolidated basis in accordance with GAAP ("Funded Indebtedness") that are required to be paid in cash during such period (excluding, for the avoidance of doubt, any mandatory or voluntary prepayments or any payments made in connection with the Transaction or any Refinancing of Funded Indebtedness), <i>plus</i>	\$
		(c)	All Restricted Payments paid in cash pursuant to $\underline{\text{Section 6.7(a)}}$ or $\underline{\text{(e)}}$ of the Credit Agreement during the Relevant Period	\$
		(d)	Fixed Charges: without duplication, the sum of: (a) + (b) + (c)	\$

EXHIBIT L-1

FORM OF LIBOR NOTICE

Bank of America, N.A., as Agent under the below referenced Credit Agreement Waukesha Operations 20975 Swenson Dr. Suite 200 Waukesha, WI 53186

Ladies and Gentlemen:

Reference hereby is made to that certain Amended and Restated Syndicated Facility Agreement dated as of February 28, 2018 (as amended, restated, supplemented, or otherwise modified from time to time, the "Credit Agreement") by and among CLEVELAND-CLIFFS INC., an Ohio corporation, as parent ("Parent"), the Subsidiaries of Parent identified on the signature pages thereto (such Subsidiaries, together with Parent, the "Borrowers"), the lenders party thereto (the "Lenders") and BANK OF AMERICA, N.A., as agent for each member of the Lender Group and the Bank Product Providers (the "Agent"). All initially capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed to them in the Credit Agreement.

This LIBOR Notice represents Borrowers' request to elect the LIBOR Option with respect to outstanding [U.S. Revolving Loans] [Australian Revolving Loans] in the amount of \$[II]^10 (the "LIBOR Rate Advance")[, and is a written confirmation of the electronic notice of such election given to Agent].

The LIBOR Rate Advance will have an Interest Period of [1, 2, 3, 6, [or 12]] month(s) commencing on [0]11.

This LIBOR Notice further confirms Borrowers' acceptance, for purposes of determining the rate of interest based on the LIBOR Rate under the Credit Agreement, of the LIBOR Rate as determined pursuant to the Credit Agreement.

Each Borrower represents and warrants that no Default or Event of Default has occurred and is continuing on the date hereof, nor will any thereof occur immediately after giving effect to the request above.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

10 Borrowers may only exercise the LIBOR Option for proposed LIBOR Rate Loans of at least \$1,000,000 (and integral multiples of \$500,000 in excess thereof).

9 Unless Agent, in its sole discretion, agrees otherwise, Borrowers shall have not more than seven (7) LIBOR Rate Loans in effect at any given time.

	Бy.	
		Name:
Dated:		Title:
Acknowledged by:		
DANK OF AMERICA N.A Asset		
BANK OF AMERICA, N.A., as Agent		
Ву:		_
Name:		
Title:		
		I -1-2

CLEVELAND-CLIFFS INC., an Ohio corporation, as Parent

EXHIBIT P-1

FORM OF U.S. PERFECTION CERTIFICATE

				1,	20	ſ

Reference is hereby made to (i) that certain Amended and Restated U.S. Guaranty and Security Agreement (the "ABL Guaranty and Security Agreement"), among Cleveland-Cliffs Inc., an Ohio corporation (the "Parent"), the other grantors party thereto (collectively, the "Grantors") and Bank of America, N.A., in its capacity as agent for each member of the Lender Group and the Bank Product Providers (in such capacity, together with its successors and assigns, if any, in such capacity, "ABL Agent") and (ii) that certain Amended and Restated Syndicated Facility Agreement, among Parent, certain Subsidiaries of Parent identified on the signature pages thereto, the lenders party thereto and ABL Agent (as the same may be amended, amended and restated, supplemented or otherwise modified from time to time, the "Credit Agreement").

Capitalized terms used but not defined herein have the meanings assigned in the Agreements, as applicable. Any terms (whether capitalized or lower case) used in this Perfection Certificate that are defined in the Code shall be construed and defined as set forth in the Code unless otherwise defined herein or in the Agreements; provided that to the extent that the Code is used to define any term used herein and if such term is defined differently in different Articles of the Code, the definition of such term contained in Article 9 of the Code shall govern. As used herein, the term "Code" shall mean the "Code" as that term is defined in the ABL Guaranty and Security Agreement.

As used herein, the term "Company" shall mean each of Cleveland-Cliffs Inc. and the other Grantors, individually, and the term "Companies" shall mean Cleveland-Cliffs Inc. and the other Grantors, collectively.

The undersigned hereby certify to the Agents as follows:

- 1. Names. (a) The exact legal name of each Company, as such name appears in its respective certificate of incorporation or any other organizational document, is set forth in <u>Schedule 1(a)</u>. Each Company is (i) the type of entity disclosed next to its name in <u>Schedule 1(a)</u> and (ii) a registered organization except to the extent disclosed in <u>Schedule 1(a)</u>. Also set forth in <u>Schedule 1(a)</u> is the organizational identification number and tax identification number, if any, of each Company that is a registered organization and the jurisdiction of formation or incorporation of each Company.
 - (b) Set forth in <u>Schedule 1(b)</u> hereto is any other corporate or organizational names each Company has had in the past five years, together with the date of the relevant change.
 - (c) Set forth in <u>Schedule 1(c)</u> is a list of all other names (including trade names or similar appellations) used by each Company, or any other business or organization to which each Company became the successor by merger, consolidation, acquisition, change in form, nature or jurisdiction of formation or incorporation or otherwise, at any time in the past five years. Also set forth in <u>Schedule 1(c)</u> is the information required by <u>Section 1(a)</u> of this certificate for any other business or organization to which each Company became the successor by merger, consolidation, acquisition, change in form, nature or jurisdiction of organization or otherwise, at any time in the past five years. Except as set forth in <u>Schedule 1(c)</u>, no Company has changed its jurisdiction of organization at any time during the past four months.
 - 2. Current Locations. (a) The chief executive office of each Company is located at the address set forth in Schedule 2(a) hereto.
 - (b) Set forth in Schedule 2(b) are all locations where each Company maintains any material books or records relating to any Collateral.
 - (c) Set forth in <u>Schedule 2(c)</u> hereto are all the other places of material business of each Company.
 - (d) Set forth in <u>Schedule 2(d)</u> hereto are all other locations where each Company maintains any of the Collateral consisting of (i) inventory in excess of \$1,500,000 per location or (ii) equipment that is included in the U.S. Borrowing Base not identified above.

- (e) Set forth in <u>Schedule 2(e)</u> hereto are the names and locations of all persons or entities other than each Company, such as lessors, consignees, bailees or warehouseman which have possession of any of the Collateral consisting of (i) inventory in excess of \$1,500,000 per location or (ii) equipment that is included in the U.S. Borrowing Base, or instruments and chattel paper executed in connection therewith.
- 3. <u>Prior Locations</u>. (a) Set forth in <u>Schedule 3</u> is the information required by <u>Schedule 2(a)</u>, <u>Schedule 2(b)</u> or <u>Schedule 2(c)</u> with respect to each location or place of material business previously maintained by each Company at any time during the past four months.
 - (b) Set forth in Schedule 3(b) is the information required by Schedule 2(d) or Schedule 2(e) with respect to each other location at which, or other person or entity with which, any of the Collateral consisting of (i) inventory in excess of \$1,500,000 per location or (ii) equipment that is included in the U.S. Borrowing Base has been previously held at any time during the past twelve months.
- 4. Real Estate. Attached hereto as Schedule 4 is a chart of (a) all real property (owned and leased) of each Company, including mineral interests, (b) legal (or other) descriptions, approximate acreages and (where applicable) lessors thereof, and (c) other information relating thereto contemplated by such Schedule.
- 5. <u>Extraordinary Transactions</u>. Except for those purchases, mergers, acquisitions, consolidations, and other transactions described in <u>Schedule 5</u> attached hereto, during the prior five years, all of the Collateral constituting ABL Priority Collateral with a value in excess of \$10,000,000 in the aggregate has been originated by each Company in the ordinary course of business or consists of goods which have been acquired by such Company in the ordinary course of business from a person in the business of selling goods of that kind.
- 6. <u>Stock Ownership and Other Equity Interests</u>. Attached hereto as <u>Schedule 6</u> is a true and correct list of all of the authorized, and the issued and outstanding, Equity Interests owned by each Grantor and the record and beneficial owners of such Equity Interests. Also set forth in <u>Schedule 6</u> is a true and correct organizational chart of Parent and its subsidiaries. Set forth on <u>Schedule 6</u> is each equity investment of each Company that represents 50% or less of the equity of the entity in which such investment was made.
- 7. <u>Instruments and Chattel Paper</u>. Attached hereto as <u>Schedule 7</u> is a true and correct list of all promissory notes, instruments (other than checks to be deposited in the ordinary course of business), tangible chattel paper, electronic chattel paper, and other evidence of indebtedness held by each Company as of the date hereof including all intercompany notes between or among any two or more Companies, in each case, having an individual value of \$5,000,000.00 or more.
- 8. <u>Intellectual Property. Schedule 8</u> provides a complete and correct list of all United States registered Copyrights (as defined in the ABL Guaranty and Security Agreement) owned by any Company and all United States applications for registration of Copyrights owned by any Company, in each case, material to the business of the Companies, taken as a whole. <u>Schedule 8</u> provides a complete and correct list of all United States Patents (as defined in the ABL Guaranty and Security Agreement) owned by any Company and all applications for United States Patents owned by any Company, in each case, material to the business of the Companies, taken as a whole. <u>Schedule 8</u> provides a complete and correct list of all United States registered Trademarks (as defined in the ABL Guaranty and Security Agreement) owned by any Company and all United States applications for registration of Trademarks owned by any Company, in each case, material to the business of the Companies, taken as a whole.
- 9. <u>Commercial Tort Claims</u>. Attached hereto as <u>Schedule 9</u> is a true and correct list of all commercial tort claims that, in the reasonable judgment of the Parent, are reasonably expected to result in proceeds in excess of \$5,000,000 held by each Company, including a brief description thereof.
- 10. <u>Deposit Accounts and Securities Accounts</u>. Attached hereto as <u>Schedule 10</u> is a true and complete list of all Deposit Accounts, Securities Accounts and Commodities Accounts (each as defined in the ABL Guaranty and Security Agreement) maintained by each Company, including the name of each institution where each such account is held, the name of each such account and the name of each entity that holds each account.
- 11. <u>Letter-of-Credit Rights</u>. Attached hereto as <u>Schedule 11</u> is a true and correct list of all Letters of Credit issued in favor of each Company, as beneficiary thereunder, with a face amount in excess of \$5,000,000.

Aircraft:		Yes No	
Vessels, boats or ships:		Yes No	
Railroad rolling stock:		Yes No	
Assets subject to a certificate of title or other r	registration statute of the United States:	Yes No	
If the answer is yes to any of these other types of assets, other	ner than immaterial assets, please describe	n <u>Schedule 12</u> .	
[The remai	inder of this page has been intentionally left	ank]	
IN WITNESS WHEREOF, we have hereunto signed	I this Perfection Certificate as of the date firs	above written.	
	[Signature Blocks to Come]	S TO PERFECTION	SCHEDULE 1(a)

12. Other Assets. A Company owns the following kinds of assets:

Legal Names, etc.

Legal Name	Type of Entity	Registered Organization (Yes/No)	Charter Number	Jurisdiction of Formation or Incorporation	Tax ID Number

Prior Organizational Names

Company/Subsidiary	Prior Name	Date of Change

SCHEDULE 1(c) TO PERFECTION CERTIFICATE

Changes in Corporate Identity: Other Names

Company/Subsidiary	Corporate Name of Entity	Action	Date of Prior Name Usage	Formation or	List of Names used During the Past Five Years

Chief Executive Offices

Entity Name	Address	County	State

Location of Books

Company/Subsidiary	Address	County	State
	<u>-</u>		

Other Places of Business

Company/Subsidiary	Address	County	State

Additional Locations of Equipment and Inventory

Company/Subsidiary	Address	County	State

SCHEDULE 2(e) TO PERFECTION CERTIFICATE

Locations of Collateral in Possession of Persons Other Than Any Company

Company/Subsidiary	Name of Entity in Possession of Collateral	Location of Collateral	County	State/Province

Prior Locations

Chief Executive Offices

Entity Name	Address	County	State

Location of Books

Company/Subsidiary	Address	County	State

Other Places of Business

Company/Subsidiary	Address	County	State

Prior Locations/Other Entities

Company/Subsidiary	Prior Locations of Collateral: Address Including County	Other Entities in Possession of Collateral/Capacity	Address of Such Other Entity Including County

SCHEDULE 4
TO PERFECTION CERTIFICATE

Real Property

<u>Transactions Other Than in the Ordinary Course of Business</u>

Company/Subsidiary	Description of Transaction Including Parties Thereto	Date of Transaction

Equity Interests of Companies and Subsidiaries

Issuer	Jurisdiction of Organization	Owner of Equity Interest	Percentage Owned	Percentage Pledged	Number of Shares or Units / Percentage of Membership Interest	Cert No.

Organizational Chart

Other Equity Interests

Issuer	Jurisdiction of Organization	Owner of Equity Interest	Percentage Owned	Percentage Pledged	Number of Shares or Units	Cert No.

Instruments and Chattel Paper

Promissory Notes:	
Chattel Paper:	
	P-1-15

Copyrights, Patents and Trademarks

UNITED STATES COPYRIGHTS:

Title	Regn No.	Regn Date	Company

UNITED STATES PATENTS:

Title	Appln. No	Filed	Patent No.	Grant Date	Current Owner

UNITED STATES TRADEMARKS:

Mark	Appln. No.	Filed	Regn. No.	Regn. Date	Company	Status

SCHEDULE 9 TO PERFECTION CERTIFICATE

Commercial Tort Claims

SCHEDULE 10 TO PERFECTION CERTIFICATE

Deposit Accounts and Securities Accounts

SCHEDULE 11 TO PERFECTION CERTIFICATE

Letter of Credit Rights

SCHEDULE 12 TO PERFECTION CERTIFICATE

Other Assets

P-1-17

FORM OF SUPPLEMENT TO PERFECTION CERTIFICATE

Supplement (this " <u>Supplement</u> "), dated as of, 20, to the Perfection Certificate, dated as of February 28, 2018 (as amended, restated, supplemented or otherwise modified from time to time, the " <u>Perfection Certificate</u> ") by each of the parties listed on the signature pages hereto.
Reference is hereby made to (i) that certain Amended and Restated U.S. Guaranty and Security Agreement (the " <u>ABL Guaranty and Security Agreement"</u>), among Cleveland-Cliffs Inc., an Ohio corporation (the " <u>Parent"</u>), the other grantors party thereto (collectively, the " <u>Grantors"</u>) and Bank of America, N.A., in its capacity as agent for each member of the Lender Group and the Bank Product Providers (in such capacity, together with its successors and assigns, if any, in such capacity, " <u>ABL Agent"</u>), and (ii) that certain Amended and Restated Syndicated Facility Agreement, among Parent, certain Subsidiaries of Parent identified on the signature pages thereto, the lenders party thereto and ABL Agent (as the same may be amended, amended and restated, supplemented or otherwise modified from time to time, the " <u>Credit Agreement"</u>).
All initially capitalized terms used herein without definition shall have the meanings ascribed thereto in the Agreements, as applicable. Any terms (whether capitalized or lower case) used in this Perfection Certificate that are defined in the Code shall be construed and defined as set forth in the Code unless otherwise defined herein or in the Agreements; provided that to the extent that the Code is used to define any term used herein and if such term is defined differently in different Articles of the Code, the definition of such term contained in Article 9 of the Code shall govern. As used herein, the term "Code" shall mean the "Code" as that term is defined in the ABL Guaranty and Security Agreement.
WHEREAS, the signatories hereto must execute and deliver a Perfection Certificate and the execution and delivery of the Perfection Certificate may be accomplished by the execution of this Supplement in favor of Agents, for the benefit of the applicable secured parties;
The undersigned, the of, hereby certify (in my capacity as and not in my individual capacity) to Agents and each of the secured parties as follows as of, 20: [the information in the Perfection Certificate delivered on or prior to the Closing Date is true, correct, and complete on and as of the date hereof.] [Schedule 1(a), "Legal Names, Etc.", Schedule 1(b), "Prior Organization Names", Schedule 1(c), "Changes in Corporate Identity; Other Names", Schedule 2(a), "Chief Executive Offices", Schedule 2(b), "Location of Books", Schedule 2(c), "Other Prior Business", Schedule 2(d), "Additional Locations of Equipment and Inventory", Schedule 2(e), "Locations of Collateral in Possession of Persons Other Than Any Company", Schedule 3(a), "Prior Locations", Schedule 3(b), "Prior Locations/Other Entities", Schedule 4, "Real Property", Schedule 5, "Transactions Other Than in the Ordinary Course of Business", Schedule 6, "Equity Interests", Schedule 7, "Instruments and Chattel Paper", Schedule 8, "Copyrights, Patents and Trademarks", Schedule 9, "Commercial Tort Claims", Schedule 10, "Deposit Accounts and Securities Accounts", Schedule 11, "Letter-of-Credit Rights", and Schedule 12, "Other Assets" attached hereto supplement Schedule 1(a), Schedule 1(b), Schedule 1(c), Schedule 2(a), Schedule 2(b), Schedule 2(c), Schedule 2(c), Schedule 2(d), Schedule 2(d), Schedule 2(d), Schedule 3(a), Schedule 3(b), Schedule 4, Schedule 5, Schedule 6, Schedule 7, Schedule 8, Schedule 9, Schedule 10, Schedule 11, and Schedule 12 respectively, to the Perfection Certificate and shall be deemed a part thereof for all purposes of the Perfection Certificate.] The undersigned officers of each of the entities party hereto hereby certify as of the date hereof on behalf of such entities in their capacity as officers of such entities and not in their individual capacities that no additional fillings or actions are required to create, preserve or perfect the security interests in the Collateral granted, assigned or pledged to
Except as expressly supplemented hereby, the Perfection Certificate shall remain in full force and effect.
IN WITNESS WHEREOF, we have hereunto signed this Supplement to Perfection Certificate as of this day of, 20
[Signature Blocks to Come]
P-1-18

SCHEDULE 1(a) TO SUPPLEMENTAL PERFECTION CERTIFICATE

Legal Names, etc.

Legal Name	Type of Entity	Registered Organization (Yes/No)	Charter Number	Jurisdiction of Formation or Incorporation	Tax ID Number
			+		

SCHEDULE 1(b) TO SUPPLEMENTAL PERFECTION CERTIFICATE

Prior Organizational Names

Company/Subsidiary	Prior Name	Date of Change

SCHEDULE 1(c) TO SUPPLEMENTAL PERFECTION CERTIFICATE

Changes in Corporate Identity: Other Names

Company/Subsidiary	Corporate Name of Entity	Action	Date of Prior Name Usage	Jurisdiction of Formation or Incorporation	List of Names used During the Past Five Years

Chief Executive Offices

Entity Name	Address	County	State

SCHEDULE 2(b) TO SUPPLEMENTAL PERFECTION CERTIFICATE

Location of Books

Company/Subsidiary	Address	County	State

SCHEDULE 2(c) TO SUPPLEMENTAL PERFECTION CERTIFICATE

Other Places of Business

Company/Subsidiary	Address	County	State

Additional Locations of Equipment and Inventory

Company/Subsidiary	Address	County	State

Locations of Collateral in Possession of Persons Other Than Any Company

Company/Subsidiary	Name of Entity in Possession of Collateral/Capacity of such Entity	Address/Location of Collateral	County	State

Prior Locations

Chief Executive Offices

Entity Name	Address	County	State

Location of Books

Company/Subsidiary	Address	County	State

Other Places of Business

Company/Subsidiary	Address	County	State

SCHEDULE 3(b) TO SUPPLEMENTAL PERFECTION CERTIFICATE

Prior Locations/Other Entities

Company/Subsidiary	Prior Locations of Collateral:	Other Entities in Possession of	Address of Such Other Entity
	Address Including County	Collateral/Capacity	Including County

Real Property

SCHEDULE 5 TO SUPPLEMENTAL PERFECTION CERTIFICATE

<u>Transactions Other Than in the Ordinary Course of Business</u>

Company/Subsidiary	Description of Transaction Including Parties Thereto	Date of Transaction

SCHEDULE 6 TO SUPPLEMENTAL PERFECTION CERTIFICATE

Equity Interests of Companies and Subsidiaries

Current Legal Entities Owned	Record Owner	Certificate No.	No. Shares/Interest	Percent Pledged

Organizational Chart

SCHEDULE 7 TO SUPPLEMENTAL PERFECTION CERTIFICATE

Instruments and Chattel Paper

Promissory Notes:	
Chattel Paper:	
P-1-3	32

Copyrights, Patents and Trademarks

UNITED STATES COPYI Registrations:	RIGHTS:		
OWNER	TITLE	REGIS	STRATION NU
Annlications:			
Applications:			
OWNER	APPLICATION	NUMBER	
UNITED STATES PATEN Registrations:	NTS:		
	REGISTRATION		
OWNER	NUMBER	DESCRIPTION	<u></u>
Applications:			
OWNER	APPLICATION NUMBER	DESCRIPTION	
2			<u> </u>
		P-1-33	

UNITED STATES TRADEMARKS:

Registrations:

REGISTRATION
OWNER NUMBER TRADEMARK

Applications:

OWNER APPLICATION NUMBER TRADEMARK

SCHEDULE 9 TO SUPPLEMENTAL PERFECTION CERTIFICATE

Commercial Tort Claims

SCHEDULE 10 TO SUPPLEMENTAL PERFECTION CERTIFICATE

Deposit Accounts and Securities Accounts

P-1-35

Letter of Credit Rights

SCHEDULE 12 TO SUPPLEMENTAL PERFECTION CERTIFICATE

Other Assets

EXHIBIT P-2

FORM OF PERFECTION CERTIFICATE (AUSTRALIA)

Г	1 201	
], 20	

Reference is hereby made to:

- (i) the general security deed and guarantee dated March 30, 2015 (the "<u>General Security Deed</u>"), among the Australian Companies (as defined below) and Bank of America, N.A., a national banking association ("<u>Bank of America</u>"), in its capacity as Australian security trustee for the Beneficiaries (as defined therein) ("<u>Australian Security Trustee</u>"), and
- (ii) the Amended and Restated Credit Agreement dated February 28, 2018, among Cleveland-Cliffs Inc. (the "Parent"), certain Subsidiaries of the Parent identified on the signature pages thereto, the lenders party thereto as "Lenders" and Bank of America in its capacity as agent for each member of the Lender Group and the Bank Product Providers (as the same may be amended, amended and restated, supplemented or otherwise modified from time to time. the "Credit Agreement").

Capitalized terms used but not defined herein have the meanings assigned in the General Security Deed or the Credit Agreement. Any terms (whether capitalized or lower case) used in this Perfection Certificate that are defined in the PPSA shall be construed and defined as set forth in the PPSA unless otherwise defined herein or in the Credit Agreement or General Security Deed. As used herein, the term "PPSA" shall mean the "PPSA" as that term is defined in the General Security Deed.

As used herein, the term "Australian Company" shall mean each of Cliffs Asia Pacific Iron Ore Holdings Pty Ltd, Cliffs Asia Pacific Iron Ore Pty Ltd, Cliffs Natural Resources Holdings Pty Ltd and Cliffs Natural Resources Pty Ltd, individually, and the term "Australian Companies" shall mean each Australian Company, collectively.

Each of the undersigned hereby certifies to the Agent as follows:

- 1. Names. (a) The exact legal name of each Australian Company, as such name appears in its respective certificate of incorporation, its constitution and at the Australian Securities Investment Commission, is set forth in Schedule 1(a). Also set forth in Schedule 1(a) is the ACN and ABN, if any, of each Australian Company, and the state of registration of each Australian Company.
 - (b) Set forth in Schedule 1(b) hereto is each other company name that an Australian Company has had.
 - 2. <u>Current Locations</u>. (a) The registered office of each Australian Company is located at the address set forth in <u>Schedule 2(a)</u> hereto.
 - (b) Set forth in <u>Schedule 2(b)</u> are all locations where each Australian Company maintains any material books or records relating to any Collateral.
 - (c) Set forth in Schedule 2(c) hereto are all the other places of material business of each Australian Company.
- 3. <u>Equity Interests.</u> Attached hereto as <u>Schedule 3</u> is a true and correct list of all of the authorized, issued and outstanding Equity Interests of each Australian Company and its [Material] Subsidiaries and the registered and beneficial owners of such Equity Interests.
- 4. <u>Instruments and Chattel Paper</u>. Attached hereto as <u>Schedule 4</u> is a true and correct list of all promissory notes, instruments (other than checks to be deposited in the ordinary course of business), tangible chattel paper, electronic chattel paper, and other evidence of indebtedness held by each Australian Company as of the date hereof

including	all intercompan	v notoc hotwoon o	r amana ani	two or more Lean	Dartice in each case	having	loubivibai ao a	value of \$5,000,000 or more.
IIICIUUIII	ali ilitercompan	y notes between or	i ailiony any	wo or more Loan	railles, ili eacii case,	, mavini	ari iriuiviuuai	value of \$5,000,000 of filore.

- 5. <u>Litigation</u>. Attached hereto as <u>Schedule 5</u> is a true and correct list of all current litigation, arbitration or administrative proceedings of or before any court, arbitral body or agency to which any Australian Company with a value in excess of \$5,000,000 in the aggregate is a party to, including a brief description thereof.
- 6. <u>Deposit Accounts and Marketable Securities</u>. Attached hereto as <u>Schedule 6</u> is a true and complete list of all Deposit Accounts maintained by each Australian Company, including the name of each institution where each such account is held, the name of each such account and the name of each entity that holds each account and all Marketable Securities of each Australian Company.
- 7. <u>Letter-of-Credit Rights.</u> Attached hereto as <u>Schedule 7</u> is a true and correct list of all Letters of Credit issued in favor of each Australian Company, as beneficiary thereunder, with a face amount in excess of \$5,000,000.
- 8. <u>Serial Numbered Collateral.</u> Attached hereto as <u>Schedule 8</u> is a true and correct list of all (a) Serial Numbered Collateral subject to the General Security Deed.

[The remainder of this page has been intentionally left blank]

IN WITNESS WHEREOF, we have hereunto signed this Perfection Certificate as of the date first above written.

[Signature Blocks to Come]

Legal Names, etc.

Legal Name of Australian Company	ACN	ABN	State of registration

Prior Corporate Names

Australian Company	Prior Name	

SCHEDULE 1(c) TO PERFECTION CERTIFICATE (AUSTRALIA)

Changes in Corporate Identity: Other Names

Australian Company	Corporate Name of Entity	Action	Date of Prior Name Usage	Formation or	List of Names used During the Past Five Years

Registered Offices

Australian Company	Address	State

Location of Books

Australian Company	Address	State

Other Places of Business

Australian Company	Address	State

Additional Locations of Equipment and Inventory

Australian Company	Address	County	State

SCHEDULE 2(e) TO PERFECTION CERTIFICATE (AUSTRALIA)

Locations of Collateral in Possession of Persons Other Than Any Australian Company

Australian Company	Name of Entity in Possession of Collateral/Capacity of such Entity	Address/Location of Collateral	County	State

Equity Interests of Companies and Subsidiaries

Current Legal Entities Owned	Registered Owner / Beneficial Owner	Description of Equity Interests	No. Shares/Interest	Percent Pledged

SCHEDULE 4 TO PERFECTION CERTIFICATE (AUSTRALIA)

Instruments and Chattel Paper

Promissory Notes:	
Chattel Paper:	
	P-2-11

SCHEDULE 5 TO PERFECTION CERTIFICATE (AUSTRALIA)

Litigation

SCHEDULE 6
TO PERFECTION CERTIFICATE (AUSTRALIA)

Deposit Accounts and Marketable Securities

Letter of Credit Rights

Serial Numbered Collateral

FORM OF SUPPLEMENT TO PERFECTION CERTIFICATE (AUSTRALIA)

	,
otherwi	Supplement (this "Supplement"), dated, 20, to the Perfection Certificate, dated March 30, 2015 (as amended, restated, supplemented or se modified from time to time, the "Perfection Certificate") by each of the parties listed on the signature pages thereto.
	Reference is hereby made to:
(i)	the general security deed and guarantee dated March 30, 2015 (the "General Security Deed"), among the Australian Companies (as defined below) and Bank of America, N.A., a national banking association ("Bank of America"), in its capacity as Australian security trustee for the Beneficiaries (as defined therein) ("Australian Security Trustee"), and
(ii)	the Amended and Restated Syndicated Facility Agreement dated February 28, 2018, among Cleveland-Cliffs Inc. (the "Parent"), certain Subsidiaries of the Parent identified on the signature pages thereto, the lenders party thereto as "Lenders", Bank of America in its capacity as agent for each member of the Lender Group and the Bank Product Providers and the Australian Security Trustee (as the same may be amended, amended and restated, supplemented or otherwise modified from time to time, the "Credit Agreement").
herein	Capitalized terms used but not defined herein have the meanings assigned in the General Security Deed or the Credit Agreement. Any terms (whether zed or lower case) used in this Supplement are defined in the PPSA shall be construed and defined as set forth in the PPSA unless otherwise defined or in the Credit Agreement or General Security Deed. As used herein, the term "PPSA" shall mean the "PPSA" as that term is defined in the General y Deed.
	As used herein, the term "Australian Company" shall mean each of Cliffs Asia Pacific Iron Ore Holdings Pty Ltd, Cliffs Asia Pacific Iron Ore Pty Ltd, Cliffs Resources Holdings Pty Ltd and Cliffs Natural Resources Pty Ltd, individually, and the term "Australian Companies" shall mean each Australian ny, collectively.
	WHEREAS, pursuant to Section 3.1 of the Credit Agreement, the Australian Companies must execute and deliver a Perfection Certificate and the on and delivery of the Perfection Certificate may be accomplished by the execution of this Supplement in favor of Agent, for the benefit of each member ender Group and the Bank Product Providers;
2(a), "Fand In Schedu Rights"	In accordance with Section 3.1 of the Credit Agreement, the undersigned, the of, hereby certify (in my capacity as and not in my individual capacity) to Agent and each of the other members of the Lender Group and the Bank Product Providers as follows as of, 20: [the information in the Perfection Certificate delivered on or prior to the Closing Date is true, correct, and complete on and as of the date Schedule 1(a), "Legal Names, Etc.", Schedule 1(b), "Prior Corporate Names", Schedule 1(c), "Changes in Corporate Identity: Other Names", Schedule Registered Offices", Schedule 2(b), "Location of Books", Schedule 2(c), "Other Places of Business", Schedule 2(d), "Additional Locations of Equipment ventory", Schedule 2(e), "Locations of Collateral in Possession of Persons Other Than Any Australian Company", Schedule 3, "Equity Interests", alle 4, "Instruments and Chattel Paper", Schedule 5 "Litigation", Schedule 6, "Deposit Accounts and Marketable Securities", Schedule 7, "Letter of Credit, and Schedule 8, "Serial Numbered Collateral" attached hereto supplement Schedule 1(a), Schedule 1(b), Schedule 1(c), Schedule 2(a), Schedule 2(b), alle 2(c), Schedule 2(d), Schedule 2(e), Schedule 3, Schedule 4, Schedule 5, Schedule 6, Schedule 7 and Schedule 8 respectively, to the Perfection ate and shall be deemed a part thereof for all purposes of the Perfection Certificate.]
	Except as expressly supplemented hereby, the Perfection Certificate shall remain in full force and effect.
	IN WITNESS WHEREOF, we have hereunto signed this Supplement to Perfection Certificate as of this day of, 20
	[Signature Blocks to Come]
	P-2-15
-	

SCHEDULE 1(a) TO SUPPLEMENTAL PERFECTION CERTIFICATE (AUSTRALIA)

Legal Names, etc.

Legal Name of Australian Company	ACN	ABN	State of registration

SCHEDULE 1(b) TO SUPPLEMENTAL PERFECTION CERTIFICATE (AUSTRALIA)

Prior Corporate Names

Australian Company	Prior Name	

Registered Offices

Address	State
	Address

Location of Books

Australian Company	Address	State

Other Places of Business

Australian Company	Address	State

SCHEDULE 3 TO SUPPLEMENTAL PERFECTION CERTIFICATE (AUSTRALIA)

Equity Interests of Companies and Subsidiaries

Current Legal Entities Owned	Description of Equity Interests	No. Shares/Interest	Percent Pledged

SCHEDULE 4 TO SUPPLEMENTAL PERFECTION CERTIFICATE (AUSTRALIA)

Instruments and Chattel Paper

Promissory Notes:	
Chattel Paper:	
	P-2-21
	P-2-21

SCHEDULE 5 TO SUPPLEMENTAL PERFECTION CERTIFICATE (AUSTRALIA)

Litigation

SCHEDULE 6
TO SUPPLEMENTAL PERFECTION CERTIFICATE (AUSTRALIA)

Deposit Accounts and Marketable Securities

P-2-22

SCHEDULE 7 TO SUPPLEMENTAL PERFECTION CERTIFICATE (AUSTRALIA)

Letter of Credit Rights

SCHEDULE 8 TO SUPPLEMENTAL PERFECTION CERTIFICATE (AUSTRALIA)

Serial Numbered Collateral

SCHEDULE 1.1

As used in the Agreement, the following terms shall have the following definitions:

"ABL Collateral" means, collectively, ABL Priority Collateral and Australian ABL Collateral.

"ABL Priority Collateral" has the meaning specified therefor in the Intercreditor Agreement.

"Account" means an account (as that term is defined in the Code or the Ascustralian PPSA).

"Account Debtor" means any Person who is obligated on an Account, chattel paper, or a general intangible.

"<u>Accounting Change</u>" means any change in accounting principles required or permitted by the promulgation of any rule, regulation, pronouncement or opinion by the Financial Accounting Standards Board of the American Institute of Certified Public Accountants (or successor thereto or any agency with similar functions).

"Acquired Indebtedness" means Indebtedness of a Person whose assets or Equity Interests are acquired by Parent or any of its Subsidiaries in a Permitted Acquisition; provided, that such Indebtedness (a) is either purchase money Indebtedness or a Capital Lease with respect to Equipment or mortgage financing with respect to Real Property, (b) was in existence prior to the date of such Permitted Acquisition, and (c) was not incurred in connection with, or in contemplation of, such Permitted Acquisition.

"Acquisition" means (a) the purchase or other acquisition by a Person or its Subsidiaries of all or substantially all of the assets of (or any division or business line of) any other Person (other than of a Subsidiary or a Canadian Entity), or (b) the purchase or other acquisition (whether by means of a merger, consolidation, or otherwise) by a Person or its Subsidiaries of all or substantially all of the Equity Interests of any other Person (other than of a Subsidiary or a Canadian Entity).

"Additional Lender" means, at any time, any bank, other financial institution or institutional investor that, in any case, is not an existing Lender, that is an Eligible Transferee and that agrees to provide any portion of any Incremental Commitments in accordance with Section 2.16; provided that each Additional Lender shall be subject to the approval of the Agent and each Issuing Bank (such approval not to be unreasonably withheld, delayed or conditioned), in each case to the extent any such consent would be required from the Agent and such Issuing Bank, as applicable, under Section 13.1 for an assignment of Revolver Commitments to such Additional Lender.

"ADI Account" shall have the meaning provided in Section 10 of the Australian PPSA.

"Administrative Borrower" has the meaning specified therefor in Section 18.13 of the Agreement.

"Administrative Questionnaire" has the meaning specified therefor in Section 13.1(a)(ii)(H) of this Agreement.

"Affected Lender" has the meaning specified therefor in Section 2.14(b) of the Agreement.

"Affiliate" means, as applied to any Person, any other Person who controls, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" means the possession, directly or indirectly through one or more intermediaries, of the power to direct the management and policies of a Person, whether through the ownership of Equity Interests, by contract, or otherwise; provided, that, for purposes of the definition of Eligible Accounts and Section 6.10 of the Agreement: (a) any Person which owns directly or indirectly 10% or more of the Equity Interests having ordinary voting power to the election of directors or other members of the governing body of a Person or 10% or more of the partnership or other ownership interests of a Person (other than as a limited partner of such Person) shall be deemed an Affiliate of such Person, (b) each director (or comparable manager) of a Person shall be deemed to be an Affiliate of such Person, and (c) each partnership in which a Person is a general partner shall be deemed an Affiliate of such Person.

"Agent" has the meaning specified therefor in the preamble to the Agreement.

"Agent Professionals" attorneys, accountants, appraisers, auditors, business valuation experts, environmental engineers or consultants, turnaround consultants, and other professionals and experts retained by Agent and Australian Security Trustee.

- "Agent-Related Persons" means Agent, Australian Security Trustee together with their respective Affiliates, officers, directors, employees, attorneys, and agents.
- "Agent's Account" means the Deposit Account of Agent identified on Schedule A-1 to this Agreement (or such other Deposit Account of Agent that has been designated as such, in writing, by Agent to Borrowers and the Lenders).
 - "Agent's Liens" means the Liens granted by any Loan Party to Agent under the Loan Documents and securing the Obligations (or any portion thereof).
- "Agreed Currency" means (a) Dollars, (b) Australian Dollars, (c) Canadian Dollars, (d) Pounds Sterling, (e) Euros, (f) Japanese Yen, (g) New Zealand Dollars, (h) Swiss Francs and (i) any other currency approved in writing by the Agent and the applicable Issuing Bank that is freely traded and exchangeable into U.S. Dollars.
- "Agreed Currency Equivalent" means, at any time, with respect to any amount denominated in an Agreed Currency other than Dollars, the equivalent amount thereof in the applicable Agreed Currency, as determined by Agent on the basis of the Spot Rate for the purchase of such Agreed Currency with Dollars.
- "Agreement" means the Amended and Restated Syndicated Facility Agreement to which this Schedule 1.1 is attached, as may be further amended, restated or otherwise modified from time to time.
 - "Allocated U.S. Availability" means any portion of the U.S. Excess Availability (which shall be calculated without reference to the U.S. Revolver Commitments in effect at such time) designated by the Parent to the Agent in writing to be allocated to the Australian Borrowing Base in accordance with the definition thereof
 - "APIO" means Cliffs Asia Pacific Iron Ore Pty Ltd ACN 001 892 995, a company incorporated under the laws of Australia.
- "Applicable Australian Unused Line Fee Percentage" means, as of any date of determination, the applicable percentage set forth in the following table that corresponds to the Average Australian Revolver Usage of Australian Borrowers for the most recently completed quarter; provided, that for the period from the Closing Date through the end of the first full fiscal quarter of the Borrowers ending after the Closing Date, the Applicable Australian Unused Line Fee Percentage shall be set at the rate in the row styled "Level II":

<u>Level</u>	Average Australian Revolver Usage	Applicable Australian Unused Line Fee Percentage
I	≥ 50% of the Australian Maximum Revolver Amount	0.250 percentage points
II	< 50% of the Australian Maximum Revolver Amount	0.300 percentage points

The Applicable Australian Unused Line Fee Percentage shall be re-determined on the first day of each quarter by Agent.

"Applicable Claims" means all claims, liabilities, obligations, losses, damages, penalties, judgments, proceedings, interest, costs and expenses of any kind (including remedial response costs, reasonable and documented attorneys' fees and Lender Group Expenses) at any time (including after payment in full of the Obligations or replacement of Agent or any Lender) incurred by any Indemnified Person or asserted against any Indemnified Person by any Loan Party or other Person, in any way relating to (a) any Loans, Letters of Credit, Loan Documents, Borrower Materials, or the use thereof or transactions relating thereto, (b) any action taken or omitted in connection with any Loan Documents, (c) the existence or perfection of any Liens, or realization upon any Collateral, (d) exercise of any rights or remedies under any Loan Documents or applicable law, or (e) failure by any Loan Party to perform or observe any terms of any Loan Document, in each case including all costs and expenses relating to any investigation, litigation, arbitration or other proceeding (including an Insolvency Proceeding or appellate proceedings), whether or not the applicable Indemnified Person is a party thereto.

"Applicable Margin" means, as of any date of determination and with respect to Base Rate Loans, LIBOR Rate Loans and Australian Base Rate Loans, as applicable, the applicable margin set forth in the following table that corresponds to the average daily Excess Availability for the most recently ended fiscal quarter; provided, that for the

period from the Closing Date through the end of the first full fiscal quarter of the Borrowers ending after the Closing Date, the Applicable Margin shall be no less than the margin in the row styled "Level II":

Level	Average Daily Aggregate Excess Availability	Applicable Margin Relative to Base Rate Loans (the "Base Rate Margin")	Applicable Margin Relative to LIBOR Rate Loans (the "LIBOR Rate Margin") and Australian Base Rate Loans (the "Australian Base Rate Margin")
I	≥ 66 2/3% of the Line Cap	0.25%	1.25%
П	< 66 2/3% of Line Cap	0.50%	1.50%

Except as set forth in the foregoing proviso, the Applicable Margin shall be re-determined quarterly on the first day of the month following the date of delivery to Agent of the certified calculation of Excess Availability pursuant to Section 5.1 of the Agreement. In the event that the information contained in any Borrowing Base Certificate is shown to be inaccurate, and such inaccuracy, if corrected, would have led to the application of a higher Applicable Margin for any period (an "Applicable Period") than the Applicable Margin actually applied for such Applicable Period, then (i) Borrowers shall immediately deliver to Agent a correct Borrowing Base Certificate for such Applicable Period, (ii) the Applicable Margin shall be determined as if the correct Applicable Margin (as set forth in the table above) were applicable for such Applicable Period, and (iii) the applicable Borrowers shall immediately deliver to Agent full payment in respect of the accrued additional interest as a result of such increased Applicable Margin for such Applicable Period, which payment shall be promptly applied by Agent to the affected Obligations. If Borrowers fail to provide such certification when such certification is due, the Applicable Margin shall be set at the margin in the row styled "Level II" as of the first day of the month following the date on which the certification was required to be delivered until the date on which such certification is delivered (on which date (but not retroactively), without constituting a waiver of any Default or Event of Default occasioned by the failure to timely deliver such certification, the Applicable Margin shall be set at the margin based upon the calculations disclosed by such certification.

"Applicable U.S. Unused Line Fee Percentage" means, as of any date of determination, the applicable percentage set forth in the following table that corresponds to the Average U.S. Revolver Usage of U.S. Borrowers for the most recently completed quarter; provided, that for the period from the Closing Date through the end of the first full fiscal quarter of the Borrowers ending after the Closing Date, the Applicable U.S. Unused Line Fee Percentage shall be set at the rate in the row styled "Level II":

<u>Level</u>	Average U.S. Revolver Usage	Applicable U.S. Unused Line Fee Percentage
Ι	≥ 50% of the U.S. Maximum Revolver Amount	0.250 percentage points
II	< 50% of the U.S. Maximum Revolver Amount	0.300 percentage points

The Applicable U.S. Unused Line Fee Percentage shall be re-determined on the first day of each quarter by Agent.

"Application Event" means the occurrence of (a) a failure by Borrowers (or any of them) to repay all of the Obligations in full on the Maturity Date, (b) an exercise by Agent or the Required Lenders of the remedies set forth in Section 9.1(a) or 9.1(b) or (c) an Event of Default and the election by the Required Lenders to require that payments and proceeds of Collateral be applied pursuant to Section 2.4(b)(iv) or Section 2.4(b)(v) of the Agreement.

"Arbitration Award" means that certain arbitration award granted pursuant to Case No. 18209/VRO/AGF/ZF by the ICC International Court of Arbitration in favor of Worldlink Resources Limited against Cliffs Quebec Iron Mining ULC, The Bloom Lake Iron Ore Mine Limited Partnership and Bloom Lake General Partner Limited.

"Assignee" has the meaning specified therefor in Section 13.1(a) of the Agreement.

"Assignment and Acceptance" means an Assignment and Acceptance Agreement substantially in the form of Exhibit A-1 to the Agreement.

"Associate" shall have the meaning given to it in section 128F(9) of the Australian Tax Act.

"Australia" means the Commonwealth of Australia.

"Australian ABL Collateral" has the meaning given to ABL Priority Collateral in the Australian Security Document described in paragraph (b) of that definition.

"Australian Bank Product" means any one or more of the following financial products or accommodations extended to Australian Borrowers or their respective Subsidiaries by an Australian Bank Product Provider: (a) credit cards (including commercial cards (including so-called "purchase cards", "procurement cards" or "p-cards")), (b) credit card processing services, (c) debit cards, (d) stored value cards, (e) supply chain financing, (f) Cash Management Services, or (g) transactions under Hedge Agreements.

"Australian Bank Product Agreements" means those agreements entered into from time to time by Australian Borrowers or their respective Subsidiaries with an Australian Bank Product Provider in connection with the obtaining of any of the Australian Bank Products.

"Australian Bank Product Obligations." means (a) all obligations, liabilities, reimbursement obligations, fees, or expenses owing by Australian Borrowers and their respective Subsidiaries to any Australian Bank Product Provider pursuant to or evidenced by an Australian Bank Product Agreement and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, (b) all Australian Hedge Obligations, and (c) all amounts that Agent or any Australian Revolving Lender is obligated to pay to an Australian Bank Product Provider as a result of Agent or such Australian Revolving Lender purchasing participations from, or executing guarantees or indemnities or reimbursement obligations to, an Australian Bank Product Provider with respect to the Australian Bank Products provided by such Australian Bank Product Provider to Australian Borrowers or their respective Subsidiaries; provided that in order for any item described in clauses (a) (b), or (c) above, as applicable, to constitute "Australian Bank Product Obligations", if the applicable Australian Bank Product Provider is any Person other than Bank of America or its Affiliates, then the applicable Australian Bank Product must have been provided on or after the Closing Date and Agent shall have received a Bank Product Provider Agreement within 10 days (or such later date as Agent and Parent may agree) after the date of the provision of the applicable Australian Bank Product to Australian Borrowers or their respective Subsidiaries; provided further that the Australian Bank Product Obligations shall not include any Excluded Swap Obligations.

"Australian Bank Product Provider" means any Australian Revolving Lender or any of its Affiliates, including each of the foregoing in its capacity, if applicable, as an Australian Hedge Provider, provided, that no such Person (other than Bank of America or its Affiliates) shall constitute an Australian Bank Product Provider with respect to an Australian Bank Product unless and until Agent receives a Bank Product Provider Agreement from such Person and with respect to the applicable Australian Bank Product within 10 days (or such later date as Agent and Parent may agree) after the provision of such Australian Bank Product to Australian Borrowers or their respective Subsidiaries; provided further, that if, at any time, an Australian Revolving Lender ceases to be a Lender under the Agreement, then, from and after the date on which it ceases to be a Lender thereunder, neither it nor any of its Affiliates shall constitute Australian Bank Product Providers and the obligations with respect to Australian Bank Products provided by such former Australian Revolving Lender or any of its Affiliates shall no longer constitute Australian Bank Product Obligations.

"Australian Bank Product Reserves" means, as of any date of determination, those reserves that Agent deems necessary or appropriate to establish (based upon the Australian Bank Product Providers' determination of the liabilities and obligations of Australian Borrowers and their respective Subsidiaries in respect of Australian Bank Product Obligations) in respect of Australian Bank Products then provided or outstanding.

"Australian Base Rate" shall mean, with respect to U.S. Dollars funded outside the U.S., LIBOR as of 11.00 a.m. on the first Business Day in each month for a one month period, <u>provided</u> that to the extent a comparable or successor rate is approved by the Agent in connection herewith, the approved rate shall be applied in a manner consistent with market practice; <u>provided</u>, <u>further</u> that to the extent such market practice is not administratively feasible for the Agent, such approved rate shall be applied in a manner as otherwise reasonably determined by the Agent in consultation with Parent. Any change in such rate shall take effect at the opening of business on the day of such change.

"Australian Base Rate Loan" shall mean each Revolving Loan which is designated or deemed designated as an Australian Base Rate Loan by the Parent at the time of the incurrence thereof or conversion thereto.

"Australian Base Rate Margin" has the meaning set forth in the definition of Applicable Margin.

"Australian Borrowers" means APIO, Cliffs Natural Resources Pty Ltd, ACN 112 437 180, a company incorporated under the laws of Australia, all Subsidiaries of Parent incorporated or organized under the laws of Australia whose assets comprise a portion of the Australian Borrowing Base, and shall include, as applicable, any Subsidiary of Parent incorporated under the laws of Australia that becomes a Borrower after the Closing Date pursuant to Section 5.11 of the Agreement.

"Australian Borrowing Base" means, as of any date of determination, the sum of:

- (a) 85% of the amount of Eligible Accounts of Australian Borrowers, plus
- (b) the lesser of (i) the product of 80% multiplied by the value (calculated at the lower of cost or market on a basis consistent with Borrowers' historical accounting practices) of Eligible Inventory of the Australian Borrowers at such time, and (ii) the product of 85% multiplied by the Net Orderly Liquidation Value of Eligible Inventory of Australian Borrowers at such time, *plus*
- (c) the lesser of (i) 100% of the Net Book Value of Eligible Equipment of Australian Borrowers at such time, and (ii) 85% multiplied by the Net Orderly Liquidation Value of Eligible Equipment of Australian Borrowers at such time; provided that this clause (c) of the Australian Borrowing Base shall not account for more than 35% of the availability created by the Australian Borrowing Base, plus
 - (d) at the Parent's election, the Allocated U.S. Availability (if any) at the time of determination, minus
 - (e) the aggregate amount of reserves, if any, established by Agent under Section 2.1(b) of the Agreement.

Notwithstanding the foregoing, the Australian Borrowing Base shall not exceed 25% of the sum of the U.S. Borrowing Base and the Australian Borrowing Base.

"<u>Australian Borrowing Base Deficiency</u>" occurs if at any time the Australian Revolving Loan Exposures exceeds the Australian Borrowing Base then in effect at such time.

"Australian Commitment Reallocation" has the meaning assigned to such term in Section 2.2.

"Australian Commitment Reallocation Notice" has the meaning assigned to such term in Section 2.2.

"Australian Concentration Account" shall mean, subject to the proviso in Section 5.16(b)(ii), a special concentration account established and maintained by an Australian Loan Party with Bank of America, N.A. (Australia Branch) (or if Bank of America is not Agent, such other financial institution reasonably acceptable to Agent and Parent) over which the Australian Security Trustee has sole dominion and exclusive control for withdrawal purposes and which, if required by Agent, is subject to an Australian Control Agreement.

"Australian Control Agreement" means, with respect to any applicable bank account established and maintained by an Australian Loan Party, an agreement, in form and substance reasonably satisfactory to the Agent, establishing "control" (as defined in section 341 of the Australian PPSA) of such an account by the Australian Security Trustee and whereby the Person maintaining such account agrees to comply only with the instructions originated by the Australian Security Trustee without the further consent of any Australian Loan Party.

"Australian Designated Account" means the Deposit Account of Australian Borrowers identified on Schedule D-2 to the Agreement (or such other Deposit Account of Australian Borrowers located at Australian Designated Account Bank that has been designated as such, in writing, by Borrowers to Agent).

"Australian Designated Account Bank" has the meaning specified therefor in <u>Schedule D-2</u> to the Agreement (or such other bank that is located within the United States or Australia that has been designated as such, in writing, by Borrowers to Agent).

"Australian Dilution Percent" means the percent, determined as of the end of the Australian Loan Parties' most recent field examination, equal to (a) bad debt write-downs or write-offs, discounts, returns, promotions, credits, credit

memos and other dilutive items with respect to active Accounts of the Australian Loan Parties, divided by (b) active gross sales.

"Australian Dilution Percentage" means at any time, one percentage point (or fraction thereof) for each percentage point (or fraction thereof) by which the Australian Dilution Percent for the Australian Loan Parties exceeds five percent (5.0%).

"Australian Dilution Reserve" means a reserve equal to the product of (x) the Australian Dilution Percentage times (y) the value of all Eligible Accounts of the Australian Loan Parties at such time.

"Australian Dollars" or "A\$" means the lawful currency of Australia as in effect from time to time.

"Australian End Date" has the meaning set forth in Section 14.1(a)(x).

"Australian Equipment" means any Equipment of an Australian Loan Party located on any Australian Mine Site.

"Australian Excess Availability" means, as of any date of determination, (i) the Australian Line Cap, minus (ii) the outstanding Australian Revolver Usage.

"Australian Exempt Accounts" means the bank accounts of Cliffs Natural Resources Pty Limited at Commonwealth Bank of Australia with account numbers 600011280350 and 374028211.

"Australian Extraordinary Advances" has the meaning specified therefor in Section 2.3(f)(vi) of the Agreement.

"Australian Featherweight Collateral" has the meaning given to Featherweight Collateral in the Australian Security Document described in paragraph (b) of that definition.

"Australian General Security Deed and Guarantee" means the general security deed and guarantee, dated as of March 30, 2015, among the Australian Security Trustee and each Australian Loan Party.

"Australian Hedge Obligations" means any and all obligations or liabilities, whether absolute or contingent, due or to become due, now existing or hereafter arising, of Australian Borrowers and their respective Subsidiaries arising under, owing pursuant to, or existing in respect of Hedge Agreements entered into with one or more of the Australian Hedge Providers; provided, however that the Australian Hedge Obligations shall not include any Excluded Swap Obligations.

"Australian Hedge Provider" means any Australian Revolving Lender or any of its Affiliates; provided, that no such Person (other than Bank of America or its Affiliates) shall constitute an Australian Hedge Provider unless and until Agent receives a Bank Product Provider Agreement from such Person and with respect to the applicable Hedge Agreement within 10 days (or such later date as Agent and Parent may agree) after the execution and delivery of such Hedge Agreement with Australian Borrowers or their respective Subsidiaries; provided further, that if, at any time, an Australian Revolving Lender ceases to be an Australian Revolving Lender the Agreement, then, from and after the date on which it ceases to be an Australian Revolving Lender thereunder, neither it nor any of its Affiliates shall constitute Australian Hedge Providers and the obligations with respect to Hedge Agreements entered into with such former Australian Revolving Lender or any of its Affiliates shall no longer constitute Australian Hedge Obligations.

"Australian Hedge Reserves" means, as of any date of determination, those reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to the last paragraph of Section 2.1(b), to establish and maintain with respect to the Australian Hedge Obligations of Australian Borrowers

"Australian In-Transit Inventory" means Inventory of the Australian Borrowers which is in-transit from the Australian Mine Site to a Designated Port located in Australia.

"Australian Inventory/Equipment Reserves" means, as of any date of determination, (a) Landlord Reserves for locations of Australian Borrowers, (b) those reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to the last paragraph of Section 2.1(b), to establish and maintain (including reserves for slow-moving Inventory and Inventory shrinkage) with respect to Eligible Inventory of Australian Borrowers or the Australian Maximum Revolver Amount, and (c) with respect to Australian In-Transit Inventory, those reserves that Agent deems

necessary or appropriate, in its Permitted Discretion and subject to the last paragraph of Section 2.1(b), to establish and maintain with respect to Australian In-Transit Inventory of Australian Borrowers that is Eligible Inventory or the Australian Maximum Revolver Amount (i) for the estimated costs relating to unpaid freight charges, warehousing or storage charges, taxes, duties, and other similar unpaid costs associated with the acquisition of such Australian In-Transit Inventory, plus (ii) for the estimated reclamation claims of unpaid sellers of such Australian In-Transit Inventory.

"Australian Issuing Bank" means Bank of America (or any of its Affiliates), PNC Bank, National Association, Citizens Bank, N.A., Deutsche Bank AG New York Branch, Regions Bank, Credit Suisse AG, Cayman Islands Branch, The Huntington National Bank and each other Australian Revolving Lender appointed by Parent and agreed by the Agent and such Australian Revolving Lender, in their sole discretion, to become an Australian Issuing Bank for the purpose of issuing Australian Letters of Credit pursuant to Section 2.12 of the Agreement.

"Australian Letter of Credit" means a letter of credit (as that term is defined in the Code) issued or any standby, documentary, bankers acceptance, foreign guarantee, indemnity or bank guarantee by an Australian Issuing Bank for the account of an Australian Borrower pursuant to the Agreement.

"Australian Letter of Credit Exposure" means, as of any date of determination with respect to any Australian Lender, such Australian Lender's Pro Rata Share of the Australian Letter of Credit Usage on such date.

"Australian Letter of Credit Fee" has the meaning specified therefor in Section 2.6(b)(ii) of the Agreement.

"Australian Letter of Credit Indemnified Costs" has the meaning specified therefor in Section 2.12(f) of the Agreement.

"Australian Letter of Credit Related Person" has the meaning specified therefor in Section 2.12(f) of the Agreement.

"Australian Letter of Credit Sublimit" means an amount equal to \$24,444,444.44, which amount (i) shall be allocated in the following amounts: \$7,222,222.22 to Bank of America (or any of its Affiliates), \$4,444,444.44 to PNC Bank, National Association, \$2,777,777.78 to Citizens Bank, N.A., \$2,777,777.78 to Deutsche Bank AG New York Branch, \$2,777,777.78 to Regions Bank, \$2,777,777.78 to Credit Suisse AG, Cayman Islands Branch, \$1,666,666.66 to The Huntington National Bank and (ii) may be increased to an amount to not exceed \$25,000,000 upon the agreement of one or more Australian Issuing Banks to increase their allocation of the Australian Letter of Credit Sublimit.

"<u>Australian Letter of Credit Usage</u>" means, as of any date of determination, the aggregate undrawn amount of all outstanding Australian Letters of Credit.

"Australian Line Cap" means, as of any date of determination, the lesser of (a) the Australian Maximum Revolver Amount and (b) the Australian Borrowing Base as of such date.

"Australian Loan Account" has the meaning specified therefor in Section 2.9(b) of the Agreement.

"Australian Loan Party" means any Australian Borrower or any other guarantor of the Australian Obligations (other than a U.S. Loan Party).

"Australian Maximum Revolver Amount" means \$50,000,000, as the same may be decreased by the amount of reductions in the Australian Revolver Commitments made in accordance with Section 2.4(c)(ii) of the Agreement, and as the same may be increased pursuant to Section 2.16 or decreased pursuant to Section 2.2.

"Australian Mine Site" means any land in Australia which is the subject of a statutory mining lease or similar mineral tenement granted by a Governmental Authority in favor of an Australian Loan Party.

"Australian Obligations" means (a) all loans (including the Australian Revolving Loans (inclusive of Australian Extraordinary Advances and Australian Swing Loans)), debts, principal, interest (including any interest that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), reimbursement or indemnification obligations with respect to Australian Letters of Credit (irrespective of whether contingent), premiums, liabilities (including all amounts charged to the Australian Loan Account pursuant to the Agreement), obligations (including indemnification obligations), fees, Lender Group Expenses (including any fees or expenses that accrue after the commencement of an Insolvency Proceeding,

regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), guaranties, and all covenants and duties of any other kind and description owing by any Australian Loan Party arising out of, under, pursuant to, in connection with, or evidenced by the Agreement or any of the other Loan Documents and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, and including all interest not paid when due and all other expenses or other amounts that the Australian Loan Parties are required to pay or reimburse by the Loan Documents or by law or otherwise in connection with the Loan Documents, and (b) all Australian Bank Product Obligations. Without limiting the generality of the foregoing, the Australian Obligations of the Australian Loan Parties under the Loan Documents include the obligation to pay (i) the principal of the Australian Revolving Loans, (ii) interest accrued on the Australian Revolving Loans, (iii) the amount necessary to reimburse any Australian Issuing Bank for amounts paid or payable pursuant to Australian Letters of Credit, (iv) Australian Letter of Credit commissions, fees (including fronting fees) and charges, (v) Lender Group Expenses, (vi) fees payable under the Agreement or any of the other Loan Documents, and (vii) indemnities and other amounts payable by any Australian Loan Party under any Loan Document. Any reference in the Agreement or in the Loan Documents to the Australian Obligations shall include all or any portion thereof and any extensions, modifications, renewals, or alterations thereof, both prior and subsequent to any Insolvency Proceeding.

"Australian Overadvance" means, as of any date of determination, that the Australian Revolver Usage is greater than any of the limitations set forth in Section 2.1(b) or Section 2.12, subject to Section 1.7(d).

"Australian Overadvance Loan" shall mean an Australian Revolving Loan that is a Base Rate Loan or an Australian Base Rate Loan made when an Australian Overadvance exists or is caused by the funding thereof.

"Australian Pension Plan" means a superannuation, retirement benefit or pension fund (whether established by deed or under any statute of Australia or any state or territory of Australia) contributed to by, or to which there is or may be an obligation to contribute by, any Loan Party in respect of its Australian employees and officers or former employees and officers.

"Australian PPS Law" means: (a) the Australian PPSA; (b) regulations made under the Australian PPSA; or (c) any amendment made at any time to any other legislation as a consequence of an Australian PPSA Law referred to in clauses (a) and (b) of this definition, including amendments to the Corporations Act.

"Australian PPSA" means the Personal Property Securities Act 2009 (Cth) of Australia.

"Australian Priority Payables Reserve" means those reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to the last paragraph of Section 2.1(b) with respect to amounts secured by any Liens, choate or inchoate, or any rights, whether imposed by applicable Law in Australia or elsewhere (and including rights to the payment or reimbursement of any costs, charges or other amounts in connection with any Insolvency Proceeding), which rank or are capable of ranking in priority to the Liens on the Australian ABL Collateral in favor of Agent or the Australian Security Trustee of and/or for amounts which may represent costs relating to the enforcement of Agent's and/or the Australian Security Trustee's Liens including, without limitation, to the extent applicable by operation of law, any such amounts due or which may become due and not paid for wages, long service leave, retrenchment, payment in lieu of notice, or vacation pay (including in all respects amounts protected by or payable pursuant to the Fair Work Act 2009 (Cth)), any preferential claims as set out in the Corporations Act, amounts due or which may become due and not paid under any legislation relating to workers' compensation or to employment insurance, all amounts deducted or withheld and not paid and remitted when due under the Taxation Administration Act 1953 (Cth) (but excluding Pay as You Go income withholding tax) and amounts in the future, currently or past due and not contributed, remitted or paid in respect of any Australian Pension Plan, together with any charges which may be levied by a Governmental Authority as a result of any default in payment obligations in respect of any Australian Pension Plan.

"Australian Protective Advances" has the meaning specified therefor in Section 2.3(f)(iv) of the Agreement.

"Australian Receivable Reserves" means, as of any date of determination, those reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to the last paragraph of Section 2.1(b), to establish and maintain (including reserves for rebates, discounts, warranty claims, and returns) with respect to the Eligible Accounts of Australian Borrowers or the Australian Maximum Revolver Amount.

"Australian Revolver Commitment" means, with respect to each Australian Revolving Lender, its Australian Revolver Commitment, and, with respect to all Australian Revolving Lenders, their Australian Revolver Commitments,

in each case as such Dollar amounts are set forth beside such Australian Revolving Lender's name under the applicable heading on <u>Schedule C-1</u> to the Agreement or in the Assignment and Acceptance pursuant to which such Australian Revolving Lender became an Australian Revolving Lender under the Agreement, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of <u>Section 13.1</u> of the Agreement.

"<u>Australian Revolver Usage</u>" means, as of any date of determination, the sum of (a) the amount of outstanding Australian Revolving Loans (inclusive of Australian Swing Loans and Australian Protective Advances), *plus* (b) the amount of the Australian Letter of Credit Usage.

"<u>Australian Revolving Lender</u>" means a Lender that has an Australian Revolver Commitment or that has an outstanding Australian Revolving Loan and includes each Australian Issuing Bank if the concept so requires.

"<u>Australian Revolving Loan Exposure</u>" means, with respect to any Australian Revolving Lender, as of any date of determination (a) prior to the termination of the Australian Revolver Commitments, the amount of such Australian Revolving Lender's Australian Revolver Commitments, and (b) after the termination of the Australian Revolver Commitments, the aggregate outstanding principal amount of the Australian Revolving Loans of such Australian Revolving Lender.

"Australian Revolving Loans" has the meaning specified therefor in Section 2.1(b) of the Agreement and includes Australian Letter of Credit Disbursements pursuant to Section 2.12(d).

"Australian Secured Creditors" has the meaning specified therefor in Section 15.14(b) of the Agreement.

"Australian Security Documents" means (a) the Australian Security Trust Deed, (b) the Australian General Security Deed and Guarantee, (c) any other documents which are incidental, collateral or supplementary to the security documents referred to in clauses (a) and (b) reasonably required by Agent, including for purposes of payment of stamp duty and registration and any accession deed to the Australian Security Trust Deed, (d) any deed of priority or similar document entered into by the Australian Security Trustee relating to any other Australian Security Document, (e) each other security agreement or other instrument or document executed and delivered by any Australian Loan Party to the Australian Security Trustee pursuant to the Agreement or any other Loan Document granting a Lien on assets of any Australian Loan Party as security for the Australian Obligations, and (f) any other document designated as such by the Australian Security Trustee and Parent.

"Australian Security Trust Deed" means the security trust deed, dated as of March 30, 2015, between, amongst others, the Australian Security Trustee and each Australian Loan Party.

"Australian Security Trustee" has the meaning specified therefor in the preamble to the Agreement.

"Australian Swing Lender" means Bank of America or any other Australian Revolving Lender that, at the request of Borrowers and with the consent of Agent agrees, in such Australian Revolving Lender's sole discretion, to become the Australian Swing Lender under Section 2.3(d) of the Agreement.

"Australian Swing Loan" has the meaning specified therefor in Section 2.3(d) of the Agreement.

"Australian Subsidiary" means each of APIO, Cliffs Natural Resources Pty Ltd, ACN 112 437 180, a company incorporated under the laws of Australia, any Subsidiary incorporated or organized under the laws of Australia or any subsidiary of the foregoing.

"Australian Tax Act" shall mean the Income Tax Assessment Act 1936 of Australia.

"Australian Unused Line Fee" has the meaning specified therefor in Section 2.10(b)(ii) of the Agreement.

"Authorized Person" means any one of the individuals identified on Schedule A-2 to the Agreement, as such schedule is updated from time to time by written notice from Borrowers to Agent.

"Availability" means, as of any date of determination, the amount that Borrowers are entitled to borrow as Revolving Loans under Section 2.1 of the Agreement (after giving effect to the then outstanding Revolver Usage).

"Bank of America" means Bank of America, N.A., a national banking association.

"Bank Product" means any U.S. Bank Product or any Australian Bank Product.

"Bank Product Agreements" means the U.S. Bank Product Agreements and the Australian Bank Product Agreements.

"Bank Product Collateralization" means providing cash collateral (pursuant to documentation reasonably satisfactory to Agent) to be held by Agent for the benefit of the applicable Bank Product Providers (other than the Hedge Providers) in an amount reasonably determined by Agent in its Permitted Discretion as sufficient to satisfy the reasonably estimated credit exposure with respect to the then existing Bank Product Obligations owed to such Bank Product Providers (other than Hedge Obligations).

"Bank Product Obligations" means the U.S. Bank Product Obligations and the Australian Bank Product Obligations; provided that Bank Product Obligations shall not include any Excluded Swap Obligations.

"Bank Product Provider" means any U.S. Bank Product Provider or any Australian Bank Product Provider.

"Bank Product Provider Agreement" means an agreement in substantially the form attached hereto as Exhibit B-3 to the Agreement or otherwise in form and substance reasonably satisfactory to Agent, duly executed by the applicable Bank Product Provider, the applicable Borrowers, and Agent.

"Bankruptcy Code" means title 11 of the United States Code, as in effect from time to time.

"Base Rate" means the greatest of (a) the Federal Funds Rate plus ½%, (b) the LIBOR Rate (which rate shall be calculated based upon an Interest Period of 1 month and shall be determined on a daily basis), plus 1.00%, (c) the Prime Rate and (d) 1.00%. Any change in the Base Rate due to a change in the Prime Rate, the Federal Funds Rate or the LIBOR Rate, as the case may be, shall be effective from and including the effective date of such change in the Prime Rate, the Federal Funds Rate or the LIBOR Rate, as the case may be.

"Base Rate Loan" means each portion of the Loans that bears interest at a rate determined by reference to the Base Rate.

"Base Rate Margin" has the meaning set forth in the definition of Applicable Margin.

"Benefit Plan" means an employee benefit plan as defined in Section 3(3) of ERISA (other than a Pension Plan or Multiemployer Plan) to which any Loan Party incurs or otherwise has any obligation or liability, contingent or otherwise.

"Board of Directors" means, as to any Person, the board of directors (or comparable managers) of such Person, or any committee thereof duly authorized to act on behalf of the board of directors (or comparable managers).

"Board of Governors" means the Board of Governors of the Federal Reserve System of the United States (or any successor).

"Borrower" and "Borrowers" have the respective meanings specified therefor in the preamble to the Agreement.

"Borrower Materials" has the meaning specified therefor in Section 18.9(c) of the Agreement.

"Borrowing" means a borrowing consisting of Revolving Loans made on the same day by Lenders (or Agent on behalf thereof), or by a Swing Lender in the case of a Swing Loan, or by Agent in the case of an Extraordinary Advance.

"Borrowing Base Certificate" means a certificate substantially in the form of Exhibit B-1.

"Business Day" means any day except Saturday, Sunday, or other day on which banks are authorized or required to close in the states of California or New York, except that (a) if a determination of a Business Day shall relate to a LIBOR Rate Loan, the term "Business Day" also shall exclude any day on which banks are closed for dealings in Dollar deposits in the London interbank market, and (b) if a determination of a Business Day shall relate

to any Loan to or Letter of Credit issued for the account of the Australian Borrowers, the term " <u>Business Day</u>" also shall exclude any day on which banks are authorized or required to be closed in any of Sydney or Melbourne, Australia or Hong Kong.

"CAM" means the mechanism for the allocation and exchange of interests in the U.S. Revolver Commitments and the Australian Revolver Commitments and the collections thereunder established under Section 16.

"CAM Exchange" means the exchange of the Lenders' interests provided for in Section 16.

"CAM Exchange Date" means the date on which (a) any event referred to in Section 8.4 or Section 8.5 shall occur with respect to any Borrower or (b) any remedy set forth in Section 9.1(a) or 9.1(b) is exercised.

"CAM Percentage" means, as to each Lender, a fraction, expressed as a decimal, of which (a) the numerator shall be the sum of the Dollar Equivalents (determined on the basis of Spot Rates prevailing on the CAM Exchange Date) of the Designated Obligations owed to such Lender (whether or not at the time due and payable) immediately prior to the CAM Exchange and (b) the denominator shall be the sum of the Dollar Equivalents (as so determined) of the Designated Obligations owed to all the Lenders (whether or not at the time due and payable) immediately prior to the CAM Exchange.

"Canadian Dollars" means the lawful currency of Canada as in effect from time to time.

"Canadian Entities" means the CCAA Entities and the Other Canadian Entities.

"Canadian Restructuring" means all or any part of the transaction or event or series of transactions or events described to Agent and the Lenders in that certain letter agreement, dated March 30, 2015, addressed to Agent and the Lenders and acknowledged by Agent, delivered by the Parent on March 30, 2015, together with such changes, modifications and supplements thereto that are reasonably acceptable to Agent.

"Canadian Restructuring Commencement Date" means, (i) with respect to the CCAA Entities, January 27, 2015, and (ii) with respect to the Other Canadian Entities, the earliest date on which any of the transactions or events in the definition of Canadian Restructuring is initiated (it being understood that, in the case of any Other Canadian Entity, its Canadian Restructuring Commencement Date shall be the earliest date on which any of the transactions or events in the definition of Canadian Restructuring is initiated with respect to such Canadian Entity).

"Capital Expenditures" means, with respect to any Person for any period, the amount of all expenditures by such Person and its Subsidiaries during such period that are capital expenditures as determined in accordance with GAAP, unless such expenditures are financed with Indebtedness incurred after the Closing Date that is not Loans.

"Capitalized Lease Obligation" means that portion of the obligations under a Capital Lease that is required to be capitalized in accordance with GAAP.

"Capital Lease" means a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP.

"Cash Collateral Account" means a blocked, non-interest bearing deposit account at Bank of America or a financial institution selected by the Agent, in the name of the Parent and under the sole dominion and control of the Agent, and otherwise established in a manner reasonably satisfactory to the Agent.

"Cash Collateralize" means to pledge and deposit with or deliver to the Agent, for the benefit of the Agent, each applicable Issuing Bank, the applicable Swing Lender or the Revolving Lenders, as collateral or other credit support for Letter of Credit Obligations, Obligations in respect of Swing Loans or obligations of Revolving Lenders to fund participations in respect of either thereof (as the context may require), (a) cash or deposit account balances or (b) if the applicable Swing Lender or the applicable Issuing Bank benefiting from such collateral shall agree in its sole discretion, other credit support, in each case pursuant to documentation in form and substance satisfactory to the Agent and the Swing Lender or the applicable Issuing Bank, as applicable. "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"Cash Dominion Trigger Event" means (a) the occurrence and continuance of any Specified Event of Default or (b) the failure of the U.S. Borrowers to maintain U.S. Excess Availability of at least the greater of (x) \$35,000,000 and (y) 10% of the U.S. Line Cap.

"Cash Dominion Release Event" means U.S. Excess Availability is at least the greater of (a) \$35,000,000 and (b) 10% of the U.S. Line Cap for thirty (30) consecutive days and no Specified Event of Default is outstanding during such thirty (30) consecutive day period.

"Cash Dominion Trigger Period" means the period commencing with a Cash Dominion Trigger Event and ending with a Cash Dominion Release Event.

"Cash Equivalents" means (a) Domestic Cash Equivalents; and (b) Foreign Cash Equivalents.

"Cash Management Services" means any cash management or related services including treasury, depository, return items, overdraft, controlled disbursement, merchant store value cards, e-payables services, electronic funds transfer, interstate depository network, automatic clearing house transfer (including the Automated Clearing House processing of electronic funds transfers through the direct Federal Reserve Fedline system) and other customary cash management arrangements.

"CCAA Entities" means Bloom Lake General Partner Limited Partnership, an Ontario corporation, Quinto Mining Corporation, 8568391 Canada Limited, Cliffs Quebec Iron Mining ULC (f/k/a Cliffs Quebec Iron Mining Limited), an unlimited liability company organized under the laws of British Columbia, Canada, The Bloom Lake Iron Ore Mine Limited Partnership, a limited partnership formed under the laws of Ontario, and Bloom Lake Railway Company Limited.

"CFC" means a controlled foreign corporation (as that term is defined in the IRC).

"Change in Control" means that:

- (a) any "person" or "group" (as those terms are used in Section 13(d)(3) of the Exchange Act, it being agreed that an employee of the Parent or any of its Subsidiaries for whom shares are held under an employee stock ownership, employee retirement, employee savings or similar plan and whose shares are voted in accordance with the instructions of such employee shall not be a member of a "group" (as that term is used in Section 13(d)(3) of the Exchange Act) solely because such employee's shares are held by a trustee under said plan) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, of Equity Interests of Parent (or other securities convertible into such Equity Interests) representing 50% or more of the combined voting power of all Equity Interests of Parent entitled (without regard to the occurrence of any contingency) to vote for the election of members of the Board of Directors of Parent;
- (b) except in connection with a transaction permitted by <u>Section 6.3</u> or 6.4, Borrowers fail to own and control, directly or indirectly, 100% of the Equity Interests of each other Loan Party (or if such Subsidiary becomes a Loan Party after the Closing Date, the amount owned and controlled, directly or indirectly, as of the date such Subsidiary becomes a Loan Party);
 - (c) [reserved]; or
- (d) the occurrence of any "Change in Control" (or any similar or like term) as defined in the Senior Secured Notes Indenture, the Convertible Notes Indenture, any Existing Senior Notes Indenture or any indenture, agreement, note or similar document governing or evidencing Indebtedness that is outstanding in an aggregate principal amount of \$25,000,000 or more.

"Change in Law" means the occurrence after the date of the Agreement of: (a) the adoption or effectiveness of any law, rule, regulation, judicial ruling, judgment or treaty, (b) any change in any law, rule, regulation, judicial ruling, judgment or treaty or in the administration, interpretation, implementation or application by any Governmental Authority of any law, rule, regulation, guideline or treaty, or (c) the making or issuance by any Governmental Authority of any request, rule, guideline or directive, whether or not having the force of law; provided that notwithstanding anything in the Agreement to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (ii) all requests, rules, guidelines or directives concerning capital adequacy promulgated by the Bank for International Settlements, the Basel Committee

on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities shall, in each case, be deemed to be a "Change in Law," regardless of the date enacted, adopted or issued.

"China Accounts" means bank accounts of APIO in the People's Republic of China for the purposes of paying operating expenses associated with its representative office located in the People's Republic of China, provided the aggregate credit balance in such accounts does not at any time exceed \$600,000 (or its Dollar Equivalent in any other currency or currencies).

"Claims" has the meaning assigned to such term in Section 12(c).

"Closing Date" means the date on which Agent sends Borrowers a written notice that each of the conditions precedent set forth on Schedule 3.1 either have been satisfied or have been waived, which date is February, 2018.

"Code" means the New York Uniform Commercial Code, as in effect from time to time; provided, however, that in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection, priority, or remedies with respect to Agent's Lien on any Collateral is governed by the Uniform Commercial Code as enacted and in effect in a jurisdiction other than the State of New York, the term "Code" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions thereof relating to such attachment, perfection, priority, or remedies.

"Collateral" means all assets and interests in assets and proceeds thereof now owned or hereafter acquired by any Loan Party in or upon which a Lien is granted by such Person in favor of Agent, the Australian Security Trustee, or the Lenders under any of the Loan Documents.

"Collateral Access Agreement" means a landlord waiver, bailee letter, carrier agreement or acknowledgement agreement of any lessor, warehouseman, processor, carrier, consignee, or other Person (including any Joint Venture) in possession of, having a Lien upon, or having rights or interests in any Loan Parties' books and records, Equipment, or Inventory, in each case, in form and substance reasonably satisfactory to Agent.

"Commencement Date" has the meaning assigned to such term on Schedule 5.2.

"Commodity Exchange Act" means the Commodity Exchange Act (7 U.S.C. § 1 et seq.), as amended from time to time, and any successor statute.

"Competitor" means any Person which is a direct competitor of Borrowers or their Subsidiaries and is named on Schedule A-3 and such Person's parent entities, affiliates and subsidiaries, in each case, that are readily identifiable as such by virtue of their names; provided, that in connection with any assignment or participation, the Assignee or Participant with respect to such proposed assignment or participation that is an investment bank, a commercial bank, a finance company, a fund, or other Person which merely has an economic interest in any such direct competitor, and is not itself such a direct competitor of Borrowers or their Subsidiaries, shall not be deemed to be a direct competitor for the purposes of this definition.

"Compliance Certificate" means a certificate substantially in the form of Exhibit C-1 to the Agreement delivered by the chief financial officer or other senior financial officer of Parent to Agent.

"Confidential Information" has the meaning specified therefor in Section 18.9(a) of the Agreement.

"Consolidated Net Tangible Assets" means the aggregate amount of assets (less applicable reserves and other properly deductible items) after deducting therefrom (a) all current liabilities (excluding any Indebtedness for money borrowed having a maturity of less than 12 months from the date of the most recent consolidated balance sheet of Parent but which by its terms is renewable or extendable beyond 12 months from such date at the option of Parent) and (b) all goodwill, trade names, patents, unamortized debt discount and expense and any other like intangibles, in each case as set forth on the most recent consolidated balance sheet of Parent and computed in accordance with GAAP.

"Consolidated Total Assets" means, as to any Person as of any date of determination, all amounts that would, in conformity with GAAP, be set forth opposite the caption "total assets" (or any like caption) on a consolidated balance sheet of such Person as of such date of determination.

"Control Agreement" means an Australian Control Agreement or a U.S. Control Agreement.

"Convertible Notes" means those certain 1.50% Convertible Senior Notes due 2025 issued by Parent on December 19, 2017 in the initial aggregate principal amount of \$316,250,000.

"Convertible Notes Documents" means the Convertible Notes, each Convertible Notes Indenture, and all other agreements, documents and instruments entered into now or in the future in connection with the Convertible Notes or any Convertible Notes Indenture.

"Convertible Notes Indenture" means the Indenture dated as of March 17, 2010 (as supplemented by the Seventh Supplemental Indenture dated as of May 7, 2013 and by the Eighth Supplemental Indenture dated as of December 19, 2017) governing the Convertible Notes, by and between Parent, as issuer, and U.S. Bank National Association, as trustee.

"Copyright Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"Corporations Act" means the Corporations Act 2001 (Cth) of Australia.

"Default" means an event, condition, or default that, with the giving of notice, the passage of time, or both, would be an Event of Default.

"Defaulting Lender" means any Lender that (a) has failed to fund any amounts required to be funded by it under the Agreement within 2 Business Days of the date that it is required to do so under the Agreement (including the failure to make available to Agent amounts required pursuant to a Settlement or to make a required payment in connection with a Letter of Credit Disbursement), (b) notified Borrowers, Agent, or any Lender in writing that it does not intend to comply with all or any portion of its funding obligations under the Agreement, (c) has made a public statement to the effect that it does not intend to comply with its funding obligations under the Agreement or under other agreements generally (as reasonably determined by Agent) under which it has committed to extend credit, (d) failed, within 3 Business Days after written request by Agent, to confirm that it will comply with the terms of the Agreement relating to its obligations to fund any amounts required to be funded by it under the Agreement, (e) otherwise failed to pay over to Agent or any other Lender any other amount required to be paid by it under the Agreement within 2 Business Days of the date that it is required to do so under the Agreement, or (f) (i) becomes or is insolvent or has a parent company that has become or is insolvent, (ii) becomes the subject of a bankruptcy or insolvency proceeding, or has had a receiver, conservator, trustee, or custodian or appointed for it, or has taken any action in furtherance of, or indicating its consent to, approval of or acquiescence in any such proceeding or appointment or has a parent company that has become the subject of a bankruptcy or insolvency proceeding, or has had a receiver, conservator, trustee, or custodian appointed for it, or has taken any action in furtherance of, or indicating its consent to, approval of or acquiescence in any such proceeding or appointment or (iii) has, or has a direct or indirect parent company that has, become the subject of a Bail-In Action (as defined in Section 18.15); provided, however that a Lender shall not be a Defaulting Lender solely by virtue of a Governmental Authority's ownership of an equity interest in such Lender or parent company unless the ownership provides immunity for such Lender from jurisdiction of courts within the United States or from enforcement of judgments or writs of attachment on its assets, or permits such Lender or Governmental Authority to repudiate or otherwise to reject such Lender's agreements; provided, further, that a Lender shall not be deemed to be a Defaulting Lender under clauses (a), (b) or (c) if it has notified Agent and Parent in writing that it will not make a funding because a condition to funding (specifically identified in the notice) is not or cannot be satisfied.

"<u>Defaulting Lender Rate</u>" means (a) for the first 3 days from and after the date the relevant payment is due, the Base Rate, and (b) thereafter, the interest rate then applicable to Revolving Loans that are Base Rate Loans (inclusive of the Base Rate Margin applicable thereto).

"Departing Lender" means any lender party to the Existing Syndicated Facility Agreement but not listed in Schedule C-1 of this Agreement on the Closing Date.

"Deposit Account" means any deposit account (as that term is defined in the Code) or any ADI Account.

"<u>Designated Obligations</u>" means all obligations of the Borrowers with respect to (a) principal of and interest on the Revolving Loans, (b) participations in Swing Loans funded (or required to be funded as provided in <u>Section 16</u>) by the U.S. Revolver Lenders or the Australian Revolver Lenders, as applicable, (c) participations in undrawn Letters

of Credit, (d) unreimbursed Letter of Credit Disbursements and interest thereon and (e) all facility fees and Letter of Credit participation fees.

"<u>Designated Non-Cash Consideration</u>" means the fair market value of non-cash consideration received by any Loan Party in connection with a Permitted Disposition that is so designated as Designated Non-Cash Consideration pursuant to an officer's certificate delivered by Parent to Agent at least 5 Business Days prior to the consummation of such Permitted Disposition (which certificate will set forth the basis for such valuation), less the amount of cash or Cash Equivalents, received in connection with a subsequent sale of or collection on such Designated Non-Cash Consideration.

"Designated Port" means each port, dock, marine terminal or similar location that is an Identified Location.

"Disqualified Equity Interests" means any Equity Interest that, by its terms (or by the terms of any security or other Equity Interests into which it is convertible or for which it is exchangeable), or upon the happening of any event or condition (a) matures or is mandatorily redeemable (other than solely for Qualified Equity Interests), pursuant to a sinking fund obligation or otherwise (except as a result of a change of control or asset sale so long as any rights of the holders thereof upon the occurrence of a change of control or asset sale event shall be subject to the prior repayment in full of the Loans and all other Obligations that are accrued and payable and the termination of the Revolver Commitments), (b) is redeemable at the option of the holder thereof (other than solely for Qualified Equity Interests), in whole or in part, (c) provides for the scheduled payments of dividends in cash, or (d) is or becomes convertible into or exchangeable for Indebtedness or any other Equity Interests that would constitute Disqualified Equity Interests, in each case, prior to the date that is 181 days after the Maturity Date.

"Dollar Equivalent" means, at any time, (a) with respect to any amount denominated in Dollars, such amount, and (b) with respect to any amount denominated in any currency other than Dollars, the equivalent amount thereof in Dollars, as determined by Agent on the basis of the Spot Rate for the purchase of Dollars with such currency.

"Dollars" or "\$" means United States dollars.

"Domestic Cash Equivalents" means (a) marketable direct obligations issued by, or unconditionally guaranteed by, the United States or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing within 1 year from the date of acquisition thereof, (b) marketable direct obligations issued or fully guaranteed by any state of the United States or any political subdivision of any such state or any public instrumentality thereof maturing within 1 year from the date of acquisition thereof and, at the time of acquisition, having one of the two highest ratings obtainable from either Standard & Poor's Rating Group ("S&P") or Moody's Investors Service, Inc. ("Moody's"), (c) commercial paper maturing no more than 270 days from the date of creation thereof and, at the time of acquisition, having a rating of at least A-1 from S&P or at least P-1 from Moody's, (d) certificates of deposit, time deposits, overnight bank deposits or bankers' acceptances maturing within 1 year from the date of acquisition thereof issued by any Lender or any other bank organized under the laws of the United States or any state thereof or the District of Columbia or any United States branch of a foreign bank having at the date of acquisition thereof combined capital and surplus of not less than \$1,000,000,000, (e) Deposit Accounts maintained with (i) any bank that satisfies the criteria described in clause (d) above, or (ii) any other bank organized under the laws of the United States or any state thereof so long as the full amount maintained with any such other bank is insured by the Federal Deposit Insurance Corporation, (f) repurchase obligations of any commercial bank satisfying the requirements of clause (d) of this definition or recognized securities dealer having combined capital and surplus of not less than \$1,000,000,000, having a term of not more than seven days, with respect to securities satisfying the criteria in clauses (a) or (d) above, (g) debt securities with maturities of six months or less from the date of acquisition backed by standby letters of credit issued by any commercial bank satisfying the criteria described in clause (d) above, (h) Investments in money market funds substantially all of whose assets are invested in the types of assets described in clauses (a) through (g) above and (i) Investments of the type described in the "Cash Investment Policy" of Parent, dated as of June 30, 2017, together with any modifications thereto reasonably acceptable to the Agent.

"Drawing Document" means any Letter of Credit or other document presented for purposes of drawing under any Letter of Credit.

"EBITDA" means, with respect to any fiscal period:

(a) Parent's consolidated net earnings (or loss),

minus

- (b) without duplication, the sum of the following amounts of Parent for such period to the extent included in determining consolidated net earnings (or loss) for such period:
 - (i) any extraordinary, unusual, or non-recurring gains,
 - (ii) interest income, and
 - (iii) exchange, translation or performance gains relating to any hedging transactions or foreign currency fluctuations,

plus

- (c) without duplication, the sum of the following amounts of Parent for such period to the extent included in determining consolidated net earnings (or loss) for such period:
 - (i) any non-cash extraordinary, unusual, or non-recurring losses,
 - (ii) Interest Expense,
- (iii) tax expense based on income, profits or capital, including federal, foreign, state, franchise and similar taxes (and for the avoidance of doubt, specifically excluding any sales taxes or any other taxes held in trust for a Governmental Authority),
 - (iv) depreciation and amortization for such period,
- (v) with respect to any Permitted Acquisition after the Closing Date, costs, fees, charges, or expenses consisting of out-of-pocket expenses owed by Parent or any of its Subsidiaries to any Person for services performed by such Person in connection with such Permitted Acquisition incurred within 180 days of the consummation of such Permitted Acquisition, up to an aggregate amount (for all such items in this clause (v)) for such Permitted Acquisition not to exceed the greater of (A) \$2,500,000 and (B) 2.50% of the Purchase Price of such Permitted Acquisition,
- (vi) (A) purchase accounting adjustments, including, without limitation, a dollar for dollar adjustment for that portion of revenue that would have been recorded in the relevant period had the balance of deferred revenue (unearned income) recorded on the closing balance sheet and before application of purchase accounting not been adjusted downward to fair value to be recorded on the opening balance sheet in accordance with GAAP purchase accounting rules; and (B) non-cash adjustments in accordance with GAAP purchase accounting rules under Statement of Financial Accounting Standards No. 805, in the event that such an adjustment is required by Parent's independent auditors, in each case, as determined in accordance with GAAP,
- (vii) costs and expenses incurred during such period in connection with the Canadian Restructuring or a restructuring of APIO, in an aggregate amount for all such costs and expenses incurred during such period not to exceed \$50,000,000 for such period,
- (viii) non-cash compensation expense (including deferred non-cash compensation expense), or other non-cash expenses or charges, arising from the sale or issuance of Equity Interests, the granting of stock options, and the granting of stock appreciation rights and similar arrangements (including any repricing, amendment, modification, substitution, or change of any such Equity Interests, stock option, stock appreciation rights, or similar arrangements) *minus* the amount of any such expenses or charges when paid in cash to the extent not deducted in the computation of net earnings (or loss),
 - (ix) one-time non-cash restructuring charges,
 - (x) non-cash exchange, translation, or performance losses relating to any hedging transactions or foreign currency fluctuations,
 - (xi) non-cash losses on sales of fixed assets or write-downs of fixed or intangible assets (excluding ABL Collateral),

- (xii) any non-cash loss, charge or expense (but only to the extent not relating to the ABL Collateral), and
- (xiii) financing fees, costs, accruals, payments and expenses (including rationalization, legal, tax, structuring and other costs and expenses and non-operating or non-recurring professional fees, costs and expenses related thereto), related to, to the extent permitted under this Agreement, any Permitted Investments, Permitted Dispositions (other than in the ordinary course of business), issuances of Equity Interests and issuances, amendments, modifications, refinancings or repayments of Permitted Indebtedness (in each case, regardless of whether or not consummated),

in each case, determined on a consolidated basis in accordance with GAAP.

"Eligible Accounts" means those Accounts created by a Borrower in the ordinary course of its business, that arise out of such Borrower's sale of goods or rendition of services, that comply with each of the representations and warranties respecting Eligible Accounts made in the Loan Documents, and that are not excluded as ineligible by virtue of one or more of the excluding criteria set forth below; provided, that such criteria may be revised from time to time by Agent in Agent's Permitted Discretion to address the results of any field examination performed by (or on behalf of) Agent from time to time after the Closing Date. In determining the amount to be included, Eligible Accounts shall be calculated net of customer deposits, unapplied cash, taxes, discounts, credits, allowances, and rebates. Eligible Accounts shall not include the following:

- (a) Accounts that (i) the Account Debtor has failed to pay within 120 days of original invoice date or (ii) are more than 60 days past due,
- (b) Accounts owed by an Account Debtor (or its Affiliates) where 50% or more of all Accounts owed by that Account Debtor (or its Affiliates) are deemed ineligible under clause (a) above,
- (c) Accounts with respect to which the Account Debtor is an Affiliate of any Borrower or an employee or agent of any Borrower or any Affiliate of any Borrower; <u>provided</u> that the foregoing shall not apply to Accounts owed by partners in Joint Ventures, Eligible Customers or their respective Affiliates that constitute Affiliates to the extent that such Accounts were generated on terms that are no less favorable, taken as a whole, to such Borrower than would be obtained in an arm's-length transaction with a non-Affiliate,
- (d) Accounts arising in a transaction wherein goods are placed on consignment or are sold pursuant to a guaranteed sale, a sale or return, a sale on approval, a bill and hold, or any other terms by reason of which the payment by the Account Debtor may be conditional,
 - (e) Accounts that are not payable in Dollars (or, with respect to Account Debtors of the Australian Borrowers, Dollars or Australian Dollars),
- (f) Accounts with respect to which the Account Debtor either (i) does not maintain its chief executive office in an Eligible Country, (ii) is not organized under the laws of an Eligible Country or any political subdivision thereof, or (iii) is the government of any foreign country or sovereign state, or of any state, province, municipality, or other political subdivision thereof, or of any department, agency, public corporation, or other instrumentality thereof, unless, in each case, either (x) the Account is supported by an irrevocable letter of credit reasonably satisfactory to Agent in its Permitted Discretion (as to form, substance, and issuer or domestic confirming bank) that has been delivered to Agent and is directly drawable by Agent, or (y) the Account is covered by credit insurance in form, substance, and amount, and by an insurer, reasonably satisfactory to Agent in its Permitted Discretion; provided that (i) Accounts whereby Hyundai Motor Company or Posco, or one of their respective subsidiaries organized in South Korea, is the Account Debtor, (ii) Accounts whereby China Steel or one of its subsidiaries organized in Taiwan is the Account Debtor, (iii) up to \$10,000,000 of Accounts whereby a Subsidiary of ThyssenKrupp AG organized in Brazil is the Account Debtor and (iii) up to \$10,000,000 of Accounts of U.S. Borrowers and Australian Borrowers taken as a whole, in each case may be deemed to be Eligible Accounts if they do not satisfy this clause (f) so long as they satisfy the other eligibility criteria set forth in this definition.
- (g) Accounts with respect to which the Account Debtor is either (i) the United States or any department, agency, or instrumentality of the United States (exclusive, however, of Accounts with respect to which Borrowers have complied, to the reasonable satisfaction of Agent, with the Assignment of Claims Act, 31 U.S.C. §3727), or (ii) any state or municipality of the United States,

- (h) Accounts with respect to which the Account Debtor is a creditor of a Borrower, has or has asserted a right of recoupment or setoff, or has disputed its obligation to pay all or any portion of the Account, to the extent of such claim, right of recoupment or setoff, or dispute,
- (i) with respect to Account Debtors of U.S. Borrowers, Accounts with respect to an Account Debtor whose total obligations owing to U.S. Borrowers exceed \$125,000,000 (such amount, as applied to a particular Account Debtor, being subject to reduction by Agent in its Permitted Discretion if the creditworthiness of such Account Debtor deteriorates), to the extent of the obligations owing by such Account Debtor in excess of such amount; provided, that, in each case, the amount of Eligible Accounts that are excluded because they exceed the foregoing amount shall be determined by Agent based on all of the otherwise Eligible Accounts prior to giving effect to any eliminations based upon the foregoing concentration limit,
- (j) with respect to Account Debtors of Australian Borrowers, Accounts with respect to an Account Debtor whose total obligations owing to Australian Borrowers exceed \$25,000,000 (such amount, as applied to a particular Account Debtor, being subject to reduction by Agent in its Permitted Discretion if the creditworthiness of such Account Debtor deteriorates), to the extent of the obligations owing by such Account Debtor in excess of such amount; provided, that, in each case, the amount of Eligible Accounts that are excluded because they exceed the foregoing amount shall be determined by Agent based on all of the otherwise Eligible Accounts prior to giving effect to any eliminations based upon the foregoing concentration limit,
- (k) Accounts with respect to which the Account Debtor is subject to an Insolvency Proceeding, is not, in the reasonable determination of Parent, Solvent, has gone out of business, or as to which any Borrower has received notice of an imminent Insolvency Proceeding or a material impairment of the financial condition of such Account Debtor.
- (I) Accounts, the collection of which, Agent, in its Permitted Discretion, believes to be doubtful, including by reason of the Account Debtor's financial condition,
- (m) Accounts that are not subject to a valid and perfected first priority Agent's Lien (or, with respect to Accounts owned by Australian Borrowers, a valid and perfected first priority Lien granted in favor of the Australian Security Trustee), including accounts that arose from the sale of coal, iron ore or other asextracted collateral (as defined in the Code) in the United States, if an as-extracted collateral filing in the applicable jurisdiction has not been filed for the benefit of the Agent with respect to the location of the mining operation and/or mineheads from which such coal, iron ore or other as-extracted collateral was extracted; provided that, in the event that any as-extracted collateral filing is not made within 30 days after the date of acquisition of an interest in the applicable location or minehead, any such Account covered by such filing shall not constitute an Eligible Account until the day that is 91 days after such as-extracted collateral filing is made,
- (n) Accounts with respect to which (i) the goods giving rise to such Account have not been shipped and billed to the Account Debtor, or (ii) the services giving rise to such Account have not been performed and billed to the Account Debtor; *provided* that up to \$50,000,000 of unbilled Accounts may be deemed to be Eligible Accounts if (x) any such Account has not been unbilled for more than 30 days, (y) the applicable goods have been shipped or the applicable services have been performed, as applicable, and (z) such Accounts satisfy the other eligibility criteria set forth in this definition,
 - (o) Accounts with respect to which the Account Debtor is a Sanctioned Person or Sanctioned Entity,
- (p) Accounts that represent the right to receive progress payments or other advance billings that are due prior to the completion of performance by the applicable Borrower of the subject contract for goods or services,
- (q) Accounts owned by a target acquired in connection with a Permitted Acquisition, until the completion of an appraisal and field examination with respect to such target, in each case, reasonably satisfactory to Agent (which appraisal and field examination may be conducted prior to the closing of such Permitted Acquisition), or
- (r) Accounts sold pursuant to the Purchase and Sale Agreement dated as of April 22, 2014 among certain Loan Parties party thereto, as sellers, Parent, as servicer, and CNR Receivables LLC.

"Eligible Contract" means a contract among a U.S. Borrower, as seller, and one or more Eligible Customers, as buyer, relating to the sale of iron ore in the United States (i) whereby the applicable U.S. Borrower retains title to the applicable iron ore Inventory until payment is made by the Eligible Customer in respect of such iron ore Inventory

and (ii) that provides for minimum purchases annually or a requirements contract of a quality of iron ore that is customized to the requirements of such Eligible Customer.

"Eligible Contract Inventory Location" means a location owned, leased or operated by an Eligible Customer where iron ore Inventory of a U.S. Borrower is held prior to the passage of title of such Inventory from such U.S. Borrower to the Eligible Customer pursuant to an Eligible Contract.

"Eligible Country" means each of the United States, Australia, Canada, the United Kingdom, Germany, Italy, Japan, the Netherlands, Singapore, Switzerland, Austria, any other member state of the European Union prior to May 1, 2014 and any other country approved by the Required Lenders in their sole discretion; provided, however, that upon fifteen (15) days' notice to Parent, Agent may, at its sole discretion, remove as an Eligible Country any country that does not have foreign currency ratings of "A" or better by S&P and "A2" or better by Moody's.

"Eligible Customer" means Ispat Inland Inc., International Steel Group Inc., ISG Cleveland Inc., ISG Indiana Harbor Inc., Severstal North America, Inc., AK Steel Corporation, ArcelorMittal Dofasco Inc., Algoma Steel Inc. and their respective Subsidiaries and any other Person (including successors-in-interest of the foregoing) approved by the Agent in its Permitted Discretion.

"<u>Eligible Equipment</u>" means Equipment of a Borrower that is Mobile Equipment, that complies with each of the representations and warranties respecting Eligible Equipment made in the Loan Documents, and that is not excluded as ineligible by virtue of one or more of the excluding criteria set forth below; <u>provided</u>, that such criteria may be revised from time to time by Agent in Agent's Permitted Discretion to address the results of any field examination or appraisal performed by Agent from time to time after the Closing Date. An item of Equipment shall not be included in Eligible Equipment if:

- (a) a Borrower does not have good, valid, and marketable title thereto,
- (b) other than with respect to Australian Equipment, a Borrower does not have actual and exclusive possession thereof (either directly or through a bailee or agent of a Borrower).
- (c) it is not located at one of the Identified Locations located in the continental United States (or in-transit from one such location in the continental United States to another such location in the continental United States) (or with respect to Equipment of Australian Borrowers, one of the Identified Locations located in Australia (or in-transit from one such location in Australia to another such location in Australia)),
- (d) it is in-transit to or from a location of a Borrower (other than (i) with respect to Equipment of U.S. Borrowers in-transit from one of the Identified Location located in the continental United States to another Identified Location located in the continental United States, or (ii) with respect to Equipment of Australian Borrowers in-transit from one of the Identified Locations located in Australia to another Identified Location located in Australia),
- (e) it is located on real property leased by a Loan Party from a third party (other than any Australian Mine Site) or with a bailee, in a contract warehouse or at the location of a Joint Venture, customer or other third party, in each case, unless (A) it is subject to a Collateral Access Agreement executed by the lessor, bailee, warehouseman, Joint Venture, customer or other third party, as the case may be, (B) with respect to Equipment located on leased real property, with a bailee or in a contract warehouse, it is the subject of Landlord Reserves, or (C) in respect of a bailee of Australian Equipment, the relevant Australian Borrower who is the owner of such Australian Equipment has perfected a purchase money security interest as bailor against the applicable bailee in accordance with the Australian PPSA covering the relevant Australian Equipment in the possession of such bailee,
- (f) it is not subject to a valid and perfected first priority Agent's Lien (or with respect to Equipment owned by Australian Borrowers, a valid and perfected first priority Lien in favor of the Australian Security Trustee),
- (g) it (i) is not in good repair and normal operating condition, ordinary wear and tear excepted, in accordance with its intended use in the business of the Loan Parties, (ii) is out for repair, (iii) does not meet in all material respects all standards imposed by any Governmental Authority having regulatory authority over such Equipment, (iv) is substantially worn, damaged, defective or obsolete, or (v) constitutes furnishings, real property or fixtures,
 - (h) it constitutes spare parts inventory or "surplus" equipment,

- (i) it is "subject to" (within the meaning of Section 9-311 of the Code) any certificate of title or comparable statute,
- (j) it was acquired in connection with a Permitted Acquisition, until the completion of an appraisal and field examination of such Inventory, in each case, reasonably satisfactory to Agent (which appraisal and field examination may be conducted prior to the closing of such Permitted Acquisition),
- (k) in relation to Equipment owned by the Australian Borrowers, the Australian Borrowers have not provided the Australian Security Trustee with such Equipment's serial number and such other information necessary to make an effective serial numbered collateral registration with respect to such Equipment under the Australian PPSA, or
- (I) in relation to Australian Equipment, such Australian Equipment is located on an Australian Mine Site over which a registered mining mortgage has been granted unless a Collateral Access Agreement has been executed by the mortgagee.

"Eligible Inventory" means Inventory of a Borrower that consists of goods held for sale in the ordinary course of such Borrower's business and that complies with each of the representations and warranties respecting Eligible Inventory made in the Loan Documents, and that is not excluded as ineligible by virtue of one or more of the excluding criteria set forth below; provided, that such criteria may be revised from time to time by Agent in Agent's Permitted Discretion to address the results of any field examination or appraisal performed by Agent from time to time after the Closing Date. In determining the amount to be so included, Inventory shall be valued at the lower of cost or market on a monthly basis in connection with delivery of each monthly Borrowing Base Certificate and a basis consistent with Borrowers' historical accounting practices. An item of Inventory shall not be included in Eligible Inventory if:

- (a) a Borrower does not have good, valid, and marketable title thereto,
- (b) a Borrower does not have ownership thereof,
- (c) except with respect to up to \$10,000,000 of Inventory owned by a U.S. Borrower that is located in Canada, it is not located at an Identified Location located in the continental United States (or in-transit from one such location in the continental United States to another such location in the continental United States) (or with respect to Inventory of Australian Borrowers, one of the Identified Locations located in Australia (or in-transit from one such location in Australia to another such location in Australia)),
 - (d) [reserved],
- (e) it is located on real property leased by a Loan Party from a third party (other than any Australian Mine Site) or with a bailee or in a contract warehouse or at the location of a Joint Venture, customer or other third party, in each case, unless (i) either (A) it is subject to a Collateral Access Agreement executed by the lessor, bailee, warehouseman, Joint Venture, customer or other third party, as the case may be, (B) with respect to Inventory located on leased real property, with a bailee or in a contract warehouse, it is the subject of Landlord Reserves, or (C) it is located at an Eligible Contract Inventory Location, and (ii) other than with respect to Inventory at a location of a Joint Venture or customer or at an Eligible Contract Inventory Location or a Designated Port located in the United States, it is segregated or otherwise separately identifiable from goods of others, if any, on the premises,
- (f) it is in-transit to a Designated Port on a carrier not owned by one of the Loan Parties unless Agent has received a Collateral Access Agreement with the applicable carrier with respect thereto.
- (g) it is the subject of a bill of lading or other document of title other than those delivered to Agent as to goods in-transit as set forth in clauses (c), (e) or (f) above,
- (h) it is not subject to a valid and perfected first priority Agent's Lien (or with respect to Inventory owned by Australian Borrowers, a valid and perfected first priority Lien in favor of the Australian Security Trustee).
 - (i) it consists of goods returned or rejected by a Borrower's customers,
- (j) it consists of goods that are obsolete or slow moving, restrictive or custom items, packaging and shipping materials, bill and hold goods, defective goods, "seconds," or Inventory acquired on consignment,

- (k) it is subject to third party trademark, licensing or other proprietary rights, unless Agent is satisfied in its Permitted Discretion that such Inventory can be freely sold by Agent on and after the occurrence of an Event of a Default despite such third party rights,
- (I) it was acquired in connection with a Permitted Acquisition, until the completion of an appraisal and field examination of such Inventory, in each case, reasonably satisfactory to Agent (which appraisal and field examination may be conducted prior to the closing of such Permitted Acquisition).
- (m) it was acquired from a Sanctioned Person or Sanctioned Entity, or it does not meet all standards imposed by any Governmental Authority or constitutes Hazardous Materials,
- (n) it constitutes work-in process, raw material or supplies, other than (i) in the case of iron ore located in the United States, work-in-process that has been converted into concentrate, (ii) in the case of coal located in the United States, work-in-process and raw material, and (iii) in the case of iron ore located in Australia, work-in-process and raw material,
- (o) it constitutes coal, iron ore or other as-extracted collateral (as defined in the UCC) in the United States, unless an as-extracted collateral filing in the applicable jurisdiction has been filed for the benefit of the Agent with respect to the location of the mining operation and/or mineheads from which such coal, iron ore or other as-extracted collateral was extracted, or
- (p) in relation to Inventory owned by Australian Loan Parties, such Inventory is located on an Australian Mine Site over which a registered mining mortgage has been granted unless a Collateral Access Agreement has been executed by the mortgagee.

"Eligible Transferee" means (a) any Lender (other than a Defaulting Lender), any Affiliate of any Lender and any Related Fund of any Lender; (b) (i) a commercial bank organized under the laws of the United States or any state thereof, and having total assets in excess of \$1,000,000,000; (ii) a savings and loan association or savings bank organized under the laws of the United States or any state thereof, and having total assets in excess of \$1,000,000,000; (iii) a commercial bank organized under the laws of any other country or a political subdivision thereof; provided that (A) (x) such bank is acting through a branch or agency located in the United States or (y) such bank is organized under the laws of a country that is a member of the Organization for Economic Cooperation and Development or a political subdivision of such country, and (B) such bank has total assets in excess of \$1,000,000,000; (c) any other entity (other than a natural person, Loan Party or an Affiliate of a Loan Party) that is an "accredited investor" (as defined in Regulation D under the Securities Act) that extends credit or buys loans as one of its businesses including insurance companies, investment or mutual funds and lease financing companies, and having total assets in excess of \$1,000,000,000; and (d) any other Person (other than a natural person, Loan Party or Affiliate of a Loan Party) approved by Agent, each Issuing Bank and, so long as no Event of Default is continuing, Parent.

"Employee Benefit Plan" means any Benefit Plan, Multiemployer Plan or Pension Plan.

"Enforcement Action" any action to enforce any Obligations (other than Bank Product Obligations) or Loan Documents or to exercise any rights or remedies relating to any Collateral (whether by judicial action, self-help, notification of Account Debtors, setoff or recoupment, credit bid, action in a Loan Party's Insolvency Proceeding or otherwise) while an Event of Default exists.

"Environmental Action" means any written complaint, summons, citation, notice, directive, order, claim, litigation, investigation, judicial or administrative proceeding, judgment, letter, or other written communication from any Governmental Authority, or any third party involving violations of Environmental Laws or releases of Hazardous Materials (a) from any assets, properties, or businesses of any Borrower, any Subsidiary of any Borrower, or, to the extent potentially giving rise to liability to any Borrower or any of their respective Subsidiaries, any of their predecessors in interest, (b) from adjoining properties or businesses, or (c) from or onto any facilities which received Hazardous Materials generated by any Borrower, any Subsidiary of any Borrower, or, to the extent potentially giving rise to liability to any Borrower or any of their respective Subsidiaries, any of their predecessors in interest.

"Environmental Law" means any applicable federal, state, provincial, foreign or local statute, law, rule, regulation, ordinance, code, binding and enforceable guideline, binding and enforceable written policy, or rule of common law now or hereafter in effect and in each case as amended, or any judicial or administrative order, consent decree or judgment, in each case, to the extent binding on any Borrower or its Subsidiaries, relating to protection of

the environment, the generation, handling, storage, treatment, release or disposal of, or exposure to, hazardous or toxic materials, or the effect of the environment on human health, in each case as amended from time to time.

"Environmental Liabilities" means all liabilities, monetary obligations, losses, damages, costs and expenses (including all reasonable fees, disbursements and expenses of counsel, experts, or consultants, and costs of investigation and feasibility studies), fines, penalties, sanctions, and interest incurred or arising under any Environmental Law or as a result of any claim or demand relating to any Environmental Law, or Remedial Action required, by any Governmental Authority or any third party.

"Equipment" means equipment (as that term is defined in the Code).

"Equity Interest" means, with respect to a Person, all of the shares, options, warrants, interests, participations, or other equivalents (regardless of how designated) of or in such Person, whether voting or nonvoting, including capital stock (or other ownership or profit interests or units), preferred stock, or any other "equity security" (as such term is defined in Rule 3a11-1 of the General Rules and Regulations promulgated by the SEC under the Exchange Act), but excluding any debt securities convertible into, exchangeable for or referencing any of the foregoing, including without limitation the Convertible Notes (prior to any conversion thereof into Equity Interests).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and any successor statutes, and all regulations and guidance promulgated thereunder. Any reference to a specific section of ERISA shall be deemed to be a reference to such section of ERISA and any successor statutes, and all regulations and guidance promulgated thereunder.

"ERISA Affiliate" means (a) any Person subject to ERISA whose employees are treated as employed by the same employer as the employees of any Loan Party under IRC Section 414(b), (b) any trade or business subject to ERISA whose employees are treated as employed by the same employer as the employees of any Loan Party under IRC Section 414(c), (c) solely for purposes of Section 302 of ERISA and Section 412 of the IRC, any organization subject to ERISA that is a member of an affiliated service group of which any Loan Party is a member under IRC Section 414(m), or (d) solely for purposes of Section 302 of ERISA and Section 412 of the IRC, any Person subject to ERISA that is a party to an arrangement with any Loan Party and whose employees are aggregated with the employees of any Loan Party under IRC Section 414(o).

"Euro" means the single currency of the Participating Member States.

"Event of Default" has the meaning specified therefor in Section 8 of the Agreement.

"Excess Availability" means, as of any date of determination, the amount equal to the sum of the U.S. Excess Availability and the Australian Excess Availability.

"Exchange Act" means the Securities Exchange Act of 1934, as in effect from time to time.

"Excluded Accounts" has the meaning specified in the Guaranty and Security Agreement.

"Excluded Property" has the meaning specified therefor in the Guaranty and Security Agreement.

"Excluded Subsidiary" means (i) any direct or indirect Foreign Subsidiary of Parent (prior to the Australian End Date, other than a Subsidiary incorporated under the laws of Australia formed or acquired after the Closing Date but solely with respect to the Australian Obligations), (ii) any non-Foreign Subsidiary if substantially all of its assets consist of the Voting Stock of one or more direct or indirect Foreign Subsidiaries of Parent, (iii) any non-Foreign Subsidiary of a Foreign Subsidiary, (iv) any Subsidiary that is an Immaterial Subsidiary, (v) any non-Wholly Owned Subsidiary, to the extent, and for so long as, a guarantee by such Subsidiary of the obligations of the U.S. Borrowers under the Loan Documents would be prohibited by the terms of any organizational document, joint venture agreement or shareholder's agreement applicable to such Subsidiary, provided that such prohibition existed on the Closing Date or, with respect to any Subsidiary formed or acquired after the Closing Date or which became a Permitted Joint Venture after the Closing Date (and, in the case of any Subsidiary acquired after the Closing Date, for so long as such prohibition was not incurred in contemplation of such acquisition), on the date such Subsidiary is so formed or acquired or became a Permitted Joint Venture, (vi) any parent entity of any non-Wholly Owned Subsidiary, to the extent, and for so long as, a guarantee by such Subsidiary, of the obligations of the U.S. Borrowers under the Loan Documents would be prohibited by the terms of any organizational document, joint venture agreement or shareholder's agreement applicable to the

non-Wholly Owned Subsidiary to which such Subsidiary is a parent, provided that (A) such prohibition existed on the Closing Date or, with respect to any Subsidiary formed or acquired after the Closing Date (and, in the case of any Subsidiary acquired after the Closing Date, for so long as such prohibition was not incurred in contemplation of such acquisition), on the date such Subsidiary is so formed or acquired and (B) a direct or indirect parent company of such parent entity (1) shall be a Guarantor and (2) shall be a holding company not engaged in any business activities or having any assets or liabilities other than (x) its ownership and acquisition of the Equity Interest of the applicable joint venture (or any other entity holding an ownership interest in such joint venture), together with activities directly related thereto, (y) actions required by law to maintain its existence and (z) activities incidental to its maintenance and continuance and to the foregoing activities, (vii) Cleveland-Cliffs International Holding Company, so long as substantially all of its assets consist of equity interests in, or indebtedness of, one or more Foreign Subsidiaries, (viii) Wabush Iron Co. Limited, and (ix) any Subsidiary of a Person described in the foregoing clauses (i), (ii), (iii), (iv), (v), (vi), (vii) or (viii), provided in each case that such Subsidiary has not guaranteed any Obligations of U.S. Borrowers or guarantors under the Existing Senior Notes Documents.

"Excluded Swap Obligations" means with respect to any Loan Party, any Swap Obligation if, and to the extent that, all or a portion of the Guaranty Obligations of such Person of, or the grant by such Person of a security interest to secure, such Swap Obligation (or any Guaranty Obligation thereof) is or becomes illegal under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof). If a Swap Obligation arises under a master agreement governing more than one swap, such exclusion shall apply only to the portion of such Swap Obligation that is attributable to swaps for which such Guaranty Obligation or security interest is or becomes illegal.

"Excluded Taxes" means (i) any tax imposed on the net income or net profits of any Lender or any Participant, franchise Taxes and branch profits Taxes, in each case, (A) imposed as a result of such Lender or such Participant being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision or taxing authority thereof) or (B) imposed as a result of a present or former connection between such Lender or such Participant and the jurisdiction or taxing authority imposing the Tax (other than any such connection arising solely from such Lender or such Participant having executed, delivered or performed its obligations or received payment under, or enforced its rights or remedies under this Agreement or any other Loan Document); (ii) Taxes resulting from a Lender's or a Participant's failure to comply with the requirements of Section 17.2 of this Agreement, (iii) any United States federal or Australian withholding Taxes that would be imposed on amounts payable to a Lender pursuant to a law in effect at the time such Lender becomes a party to this Agreement (or designates a new lending office), except for (A) any amount that such Lender (or its assignor, if any) was previously entitled to receive pursuant to Section 17.1 of this Agreement, if any, with respect to such withholding Tax at the time such Lender became a party to this Agreement (or designated a new lending office), as a result of a change in law, rule, regulation, order or other decision with respect to any of the foregoing by any Governmental Authority, and (iv) any withholding Taxes imposed under FATCA.

"Existing Debt" means (a) the Existing Senior Notes, (b) the Senior Secured Notes, (c) the Convertible Notes and (d) any other agreement, indenture or other instrument with respect to indebtedness for borrowed money (excluding capital leases) of the Loan Parties of more than \$15,000,000 with a scheduled maturity prior to the date that is five years from the Closing Date.

"Existing Hedge Obligations" means the obligations or liabilities arising under, owing pursuant to, or existing in respect of the Hedge Agreements set forth on Schedule E-1, but only for so long as the counterpart constitutes a Hedge Provider hereunder.

"Existing Letters of Credit" means the letters of credit issued prior to the Closing Date pursuant to the Existing Syndicated Facility Agreement, as set forth on Schedule E-2.

"Existing Senior Notes" means collectively (a) those certain 5.900% Senior Notes due 2020 issued by Parent in the initial aggregate principal amount of \$400,000,000; (b) those certain 4.80% Senior Notes due 2020 issued by Parent in the initial aggregate principal amount of \$500,000,000; (c) those certain 6.25% Senior Notes due 2040 issued by Parent in the initial aggregate principal amount of \$500,000,000 and (d) those certain 4.875% Senior Notes due 2021 issued by Parent in the initial aggregate principal amount of \$700,000,000.

"Existing Senior Notes Documents" means the Existing Senior Notes, each Existing Senior Notes Indenture, and all other agreements, documents and instruments entered into now or in the future in connection with the Existing Senior Notes or any Existing Senior Notes Indenture.

"Existing Senior Notes Indenture" means any indenture governing any of the Existing Senior Notes.

"Existing Syndicated Facility Agreement" has the meaning specified therefor in the preamble to the Agreement.

"Extraordinary Advances" has the meaning specified therefor in Section 2.3(f)(vi) of the Agreement.

"FATCA" means Sections 1471 through 1474 of the IRC, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof, any agreement entered into pursuant to Section 1471(b)(1) of the IRC and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Authorities and implementing such Sections of the IRC.

"FCPA" has the meaning assigned to such term in Section 4.13

"Fee Letter" means that certain administrative agent fee letter, dated as of March 13, 2015, between Parent and Agent.

"Federal Funds Rate" means, for any period, a fluctuating interest rate per annum equal to, for each day during such period, the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by Agent from three Federal funds brokers of recognized standing selected by it.

"Field Examination/Appraisal Triggering Event" means any date on which Excess Availability is less than the greater of (a) 17.5% of the Line Cap, and (b) \$70,000,000.

"Financial Covenant Period" means a period which shall commence on any date of determination on which Excess Availability is less than the greater of (i) 10% of the Line Cap and (ii) \$35,000,000, and shall continue until Excess Availability is not less than the greater of (a) 10% of the Line Cap and (b) \$35,000,000 for a period of 60 consecutive days.

"Fixed Asset Priority Collateral" has the meaning specified therefor in the Intercreditor Agreement.

"Fixed Asset Priority Collateral Agent" has the meaning specified therefor in the Intercreditor Agreement.

"Fixed Charges" means, with respect to any fiscal period and with respect to Parent determined on a consolidated basis in accordance with GAAP but subject to Section 1.2, the sum, without duplication, of (a) Interest Expense accrued (other than (x) interest paid-in-kind, amortization of financing fees, and other non-cash Interest Expense and (y) make-whole payments, premiums or similar payments related to the prepayment or extinguishment of Indebtedness) during such period to the extent required to be paid in cash, (b) scheduled principal payments in respect of Funded Indebtedness that are required to be paid in cash during such period (excluding, for the avoidance of doubt, any mandatory or voluntary prepayments or any payments made in connection with the Transaction or any Refinancing of such Funded Indebtedness) and (c) all Restricted Payments paid in cash pursuant to Section 6.7(a) or (e) during such period.

"Fixed Charge Coverage Ratio" means, with respect to any fiscal period and with respect to Parent determined on a consolidated basis in accordance with GAAP, the ratio of (a) EBITDA for such period *minus* (i) Capital Expenditures made (to the extent not already incurred in a prior period) or incurred during such period (excluding any capitalized interest with respect thereto) to the extent required to be paid in cash (provided, however, any Capital Expenditures made or incurred during such period in connection with the Toledo, Ohio hot-briquetted iron ("HBI") production plant shall not be included in the calculation of this clause (i) solely to the extent that such Capital Expenditures are funded with balance sheet cash that constitutes net cash proceeds of (x) the Senior Secured Notes or (y) the Convertibles Notes) *minus* (ii) federal, state and local income taxes paid in cash during such period (net of federal, state and local tax refunds received in cash during such period) (it being understood that the amount subtracted pursuant to this clause (ii) shall not be less than \$0), to (b) Fixed Charges for such period.

"Foreign Benefit Event" shall mean, with respect to any Foreign Plan, (a) the existence of unfunded liabilities in excess of the amount permitted under any applicable law, (b) the failure to make the required contributions or payments, under any applicable law, on or before the due date for such contributions or payments, (c) the receipt of a notice by a Governmental Authority relating to the intention to terminate any such Foreign Plan or to appoint a trustee or similar official to administer any such Foreign Plan, or alleging the insolvency of any such Foreign Plan, (d) the incurrence of any liability by any Loan Party or any Subsidiary under applicable law on account of the complete or partial termination of such Foreign Plan or the complete or partial withdrawal of any participating employer therein, or (e) the occurrence of any transaction that is prohibited under any applicable law and that could reasonably be expected to result in the incurrence of any liability by any Loan Party or any of its Subsidiaries, or the imposition on any Loan Party or any of its Subsidiaries of any fine, excise tax or penalty resulting from any noncompliance with any applicable law.

"Foreign Cash Equivalents" means (a) certificates of deposit, bankers' acceptances, or time deposits maturing within 1 year from the date of acquisition thereof, in each case payable in an Agreed Currency and issued by any bank organized under the laws of any Specified State and having at the date of acquisition thereof combined capital and surplus of not less than \$1,000,000,000 (calculated at the then applicable Spot Rate), (b) Deposit Accounts maintained with any bank that satisfies the criteria described in clause (a) above, (c) Investments in money market funds substantially all of whose assets are invested in the types of assets described in clauses (a) and (b) above and (d) Investments of the type described in the "Cash Investment Policy" of Parent, dated as of June 30, 2017, together with any modifications thereto reasonably acceptable to the Agent.

"Foreign Plan" has the meaning specified therefor in Section 4.10(b) of the Agreement and shall include an Australian Pension Plan.

"Foreign Subsidiary" means any Subsidiary of Parent that was not formed under the laws of the United States or any state of the United States or the District of Columbia.

"FSHCO" means a Subsidiary that is organized under the laws of the United States, any state thereof or the District of Columbia substantially all of the assets of which consist of Equity Interests of one or more CFCs or FSHCOs. For the avoidance of doubt, Cleveland-Cliffs International Holding Company shall be considered a FSHCO.

"Funded Indebtedness" means, as of any date of determination, all Indebtedness for borrowed money or letters of credit of Parent, determined on a consolidated basis in accordance with GAAP.

"Funding Date" means the date on which a Borrowing occurs.

"Funding Losses" has the meaning specified therefor in Section 2.13(b)(ii) of the Agreement.

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States, consistently applied.

"Governing Documents" means, with respect to any Person, the certificate or articles of incorporation, by-laws, articles of association, or other organizational documents of such Person.

"Governmental Authority" means the government of any nation or any political subdivision thereof, whether at the national, state, territorial, provincial, municipal or any other level, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of, or pertaining to, government (including any supra-national bodies such as the European Union or the European Central Bank).

"Guarantor" means (a) each Subsidiary of each Borrower (other than any Excluded Subsidiary), and (b) each other Person that becomes a guarantor after the Closing Date pursuant to Section 5.11 of the Agreement.

"Guaranty and Security Agreement" means an amended and restated guaranty and security agreement, dated as of the Closing Date, executed and delivered by each of the U.S. Borrowers and each of the U.S. Guarantors to Agent.

"Guaranty Obligations" means as to any Person (without duplication) any obligation of such Person guaranteeing any Indebtedness ("primary Indebtedness") of any other Person (the "primary obligor") in any manner,

whether directly or indirectly, including, without limitation, any obligation of such Person, whether or not contingent: (i) to purchase any such primary Indebtedness or any property constituting direct or indirect security therefor; (ii) to advance or supply funds for the purchase or payment of any such primary Indebtedness or to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor; (iii) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary Indebtedness of the ability of the primary obligor to make payment of such primary Indebtedness; or (iv) otherwise to assure or hold harmless the owner of such primary Indebtedness against loss in respect thereof, provided, however, that the definition of Guaranty Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guaranty Obligation shall be deemed to be an amount equal to the stated or determinable amount of the primary Indebtedness in respect of which such Guaranty Obligation is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof (assuming such Person is required to perform thereunder).

"<u>Hazardous Materials</u>" means (a) substances that are defined or listed in, or otherwise classified pursuant to, any applicable Environmental Laws as "hazardous substances," "hazardous materials," "hazardous wastes," "toxic substances," or any other formulation intended to define, list, or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, carcinogenicity, reproductive toxicity, or "EP toxicity", (b) oil, petroleum, or petroleum derived substances, natural gas, natural gas liquids or synthetic gas, (c) any explosives or any radioactive materials, and (d) asbestos in any form or electrical equipment that contains any oil or dielectric fluid containing levels of polychlorinated biphenyls in excess of 50 parts per million.

"<u>Hedge Agreement</u>" means a "<u>swap agreement</u>" as that term is defined in Section 101(53B)(A) of the Bankruptcy Code.

"Hedge Obligations" means the U.S. Hedge Obligations and the Australian Hedge Obligations; provided that the Hedge Obligations shall not include any Excluded Swap Obligations.

"Hedge Provider" means any U.S. Hedge Provider or any Australian Hedge Provider.

"<u>Identified Locations</u>" means the locations identified on the most recently delivered location report provided pursuant to clause (m) of Schedule 5.2 of the Agreement or such other locations identified by the Borrowers to the Agent from time to time.

"Immaterial Subsidiary" means (a) the Persons identified on Schedule I-1 to this Agreement and (b) any other Subsidiary that, together with its Subsidiaries, does not have (i) Consolidated Total Assets in excess of 3.0% of the Consolidated Total Assets of Parent and its Subsidiaries on a consolidated basis as of the date of the most recent consolidated balance sheet of Parent or (ii) consolidated total revenues in excess 3.0% of the consolidated total revenues of Parent and its Subsidiaries on a consolidated basis for the most recently ended four fiscal quarters for which internal financial statements of Parent are available immediately preceding such calculation date; provided that any such Subsidiary, when taken together with all other Immaterial Subsidiaries does not, in each case together with their respective Subsidiaries, have (i) Consolidated Total Assets with a value in excess of 7.5% of the Consolidated Total Assets of Parent and its Subsidiaries on a consolidated basis or (ii) consolidated total revenues in excess of 7.5% of the consolidated total revenues of Parent and its Subsidiaries on a consolidated basis. For the avoidance of doubt, no Borrower shall be an Immaterial Subsidiary.

"Increased Borrowing Base Reporting Period" has the meaning specified in Schedule 5.2.

"Incremental Amendment" has the meaning specified in Section 2.16(b).

"Incremental Commitment" has the meaning specified in Section 2.16(a).

"Incremental Loans" has the meaning specified in Section 2.16(a).

"Indebtedness" as to any Person means (a) all obligations of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes, or other similar instruments and all reimbursement or other obligations in respect of letters of credit, bankers acceptances, or other financial products, (c) all obligations of such Person as a lessee under Capital Leases, (d) all obligations or liabilities of others secured by a Lien on any asset of such Person, irrespective of whether such obligation or liability is assumed, (e) all obligations of such Person to pay the deferred purchase price of assets (other than trade payables incurred in the ordinary course of business), (f) all

monetary obligations of such Person owing under Hedge Agreements (which amount shall be calculated based on the amount that would be payable by such Person if the Hedge Agreement were terminated on the date of determination), (g) any Disqualified Equity Interests of such Person, and (h) any obligation of such Person guaranteeing or intended to guarantee (whether directly or indirectly guaranteed, endorsed, co-made, discounted, or sold with recourse) any obligation of any other Person that constitutes Indebtedness under any of clauses (a) through (g) above. For purposes of this definition, (i) the amount of any Indebtedness represented by a guaranty or other similar instrument shall be the lesser of the principal amount of the obligations guaranteed and still outstanding and the maximum amount for which the guaranteeing Person may be liable pursuant to the terms of the instrument embodying such Indebtedness, and (ii) the amount of any Indebtedness which is limited or is non-recourse to a Person or for which recourse is limited to an identified asset shall be valued at the lesser of (A) if applicable, the limited amount of such obligations, and (B) if applicable, the fair market value of such assets securing such obligation.

"Indemnified Liabilities" has the meaning specified therefor in Section 10.3 of the Agreement.

"Indemnified Person" has the meaning specified therefor in Section 10.3 of the Agreement.

"Indemnified Taxes" means, any Taxes other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of any Loan Party under any Loan Document.

"Insolvency Proceeding" means with respect to any Person, (a) any proceeding, corporate action, procedure or step commenced or taken by or against that Person under any provision of the Bankruptcy Code, the Corporations Act or under any other state or federal bankruptcy or insolvency law, assignments for the benefit of creditors, formal or informal moratoria, compositions, extensions generally with creditors, or proceedings seeking reorganization, arrangement, or other similar relief, or (b) the appointment of a custodian, trustee, receiver, interim receiver, national receiver, receiver-manager, monitor, liquidator, administrator, judicial manager, administrative receiver, supervisor, compulsory manager, Controller or similar custodian for that Person or for substantially all of its assets.

"Intercompany Subordination Agreement" means an intercompany subordination agreement, dated as of March 30, 2015, executed and delivered by each of the other Loan Parties and Agent.

"Intercreditor Agreement" means that certain ABL Intercreditor Agreement, dated as of December 19, 2017, between Agent and U.S. Bank National Association, as collateral agent in respect of the Senior Secured Notes and acknowledged and agreed to by the U.S. Loan Parties.

"Interest Expense" means, for any period, the aggregate of the interest expense of Parent for such period, determined on a consolidated basis in accordance with GAAP.

"Interest Period" means, with respect to each LIBOR Rate Loan, a period commencing on the date of the making of such LIBOR Rate Loan (or the continuation of a LIBOR Rate Loan or the conversion of a Base Rate Loan to a LIBOR Rate Loan) and ending 1, 2, 3, or 6 months thereafter or, if agreed to by all Lenders, 12 months thereafter; provided, that (a) interest shall accrue at the applicable rate based upon the LIBOR Rate from and including the first day of each Interest Period to, but excluding, the day on which any Interest Period expires, (b) any Interest Period that would end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day, (c) with respect to an Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period), the Interest Period shall end on the last Business Day of the calendar month that is 1, 2, 3, 6 or 12 months after the date on which the Interest Period began, as applicable, and (d) Borrowers may not elect an Interest Period which will end after the Maturity Date.

"Inventory" means inventory (as that term is defined in the Code or the Australian PPSA).

"Investment" means, with respect to any Person, any investment by such Person in any other Person (including Affiliates) in the form of loans, guarantees, advances, capital contributions (excluding (a) commission, travel, and similar advances to directors, officers and employees of such Person made in the ordinary course of business, and (b) bona fide accounts receivable arising in the ordinary course of business), or acquisitions of Equity Interests, or all or substantially all of the assets of such other Person (or of any division or business line of such other Person), and any other items that are or would be classified as investments on a balance sheet prepared in accordance with GAAP. The amount of any Investment shall be the original cost of such Investment plus the cost of all additions thereto, without

any adjustment for increases or decreases in value, or write-ups, write-downs, or write-offs with respect to such Investment.

"IRC" means the Internal Revenue Code of 1986, as amended.

"ISP" means, with respect to any Letter of Credit, the International Standby Practices 1998 (International Chamber of Commerce Publication No. 590) and any subsequent revision thereof adopted by the International Chamber of Commerce on the date such Letter of Credit is issued.

"Issuer Document" means, with respect to any Letter of Credit, a letter of credit application, a letter of credit agreement, or any other document, agreement or instrument entered into (or to be entered into) by a Borrower in favor of Issuing Bank and relating to such Letter of Credit.

"Issuing Bank" means any U.S. Issuing Bank or any Australian Issuing Bank.

"Issuing Bank Indemnitees" Issuing Bank and its officers, directors, employees, Affiliates, agents and attorneys.

"Joint Book Runners" means Merrill Lynch, Pierce, Fenner & Smith Incorporated, PNC Capital Markets LLC, Deutsche Bank Securities Inc., Citizens Bank N.A. and Regions Business Capital, a Division of Regions Bank.

"Joint Lead Arrangers" means Merrill Lynch, Pierce, Fenner & Smith Incorporated PNC Capital Markets LLC, Deutsche Bank Securities Inc., Citizens Bank N.A. and Regions Business Capital, a Division of Regions Bank.

"Joint Ventures" means the joint ventures established pursuant to the Joint Venture Agreements.

"Joint Venture Agreements" means, collectively (and in each case, as amended, restated, supplemented or modified from time to time) (a) that certain Hibbing Taconite Joint Venture Agreement dated as of January 1, 1974, as amended by that certain Amended Joint Venture Agreement dated as of January 1, 1975, that certain Agreement of Amendment dated as of July 31, 1978, that certain Agreement of Amendment dated as of February 1, 1981, that certain Agreement of Amendment dated as of January 1, 1999, that certain Agreement of Amendment dated as of November 3, 2008, and that certain Amendment dated as of February 9, 2011, among ArcelorMittal Hibbing Inc., a Delaware corporation ("ArcelorMittal Hibbing") (as successor in interest to Bethlehem Steel Corporation), Cliffs Mining Holding Sub Company, a Delaware corporation (as successor in interest to Pickands Mather & Co.), Ontario Hibbing Company, a Minnesota corporation ("Ontario Hibbing"), and Hibbing Development Company, a general partnership organized under the Uniform Partnership Act as adopted in the State of Minnesota; (b) that certain Partnership Agreement of Hibbing Development Corporation dated as of July 31, 1978, as amended by that certain Agreement of Amendment dated as of February 1, 1981, that certain Agreement of Amendment dated December 31, 1985, and that certain Agreement of Amendment dated July 18, 2002, among ArcelorMittal Hibbing (as successor in interest to Bethlehem Steel Corporation), Ontario Hibbing (as successor in interest to Republic Hibbing Corporation) and Pickands Hibbing Corporation, a Minnesota corporation (as successor in interest to Pickands Mather & Co.); (c) that certain Restated Empire Iron Ore Mining Partnership Agreement dated as of December 1, 1978, as amended by that certain Amendment to Empire Partnership Agreement dated as of July 1, 1983, that certain Empire Partnership Omnibus Agreement dated as of December 13, 1991, that certain Second Empire Partnership Omnibus Agreement dated as of December 31, 2002, and that certain 2014 Extension Agreement effective as of January 1, 2014, between Cliffs Empire, Inc., a Michigan corporation, and Cliffs Empire II Inc., a Delaware corporation (as successor in interest to ArcelorMittal Empire Inc (as successors in interest to The Cleveland-Cliffs Iron Company, McLouth Steel Corporation, WSC Empire, Inc. and Inland Steel Company)); (d) that certain Operating Agreement of Tilden Mining Company L.C. dated as of June 30, 1995, as amended, restated, supplemented or otherwise modified from time to time, including as amended by that certain Amendment that became effective initially on January 1, 1999 (with certain additional provisions subsequently becoming effective on January 1, 2000), between Cliffs TIOP, Inc., a Michigan corporation, and Cliffs TIOP II, LLC, a Delaware limited liability company (as successor in interest to Ontario Tilden Company (as successors in interest to J&L Cliffs Ore Partnership, Tilden Iron Ore Partnership, Tilden Magnetite Partnership, Cliffs Tilden. Inc.. Stelco Coal Company and Cannelton Iron Ore Company)), (e) that certain Wabush Mines Joint Venture Agreement and Wabush Mines Supplemental Joint Venture Agreement, each dated January 1, 1967, and as amended by that certain Amendment dated as of January 1, 1999, that certain Amendment to Wabush Mines Project Agreements and Management Agreement dated as of November 5, 2008, and that Wabush Mines Joint Venture Interim Agreement dated as of February 1, 2010 each by and among entered into by and between Wabush Iron Co. Limited and Wabush Resources Inc. or its predecessors in interest, (f) Amended and Restated Limited Partnership Agreement dated July 20, 2009 by and between Bloom Lake General Partner Limited, Cliffs Quebec Iron Mining Limited, Wugang Canada

Resources Investment Limited and Minerals Corporation Limited of Wuhan Iron and Steel (Group), as amended, and Shareholders Agreement of Bloom Lake General Partner Limited dated July 20, 2009 by and between Cliffs Quebec Iron Mining Limited and Wugang Canada Resources Investment Limited, as amended, and (g) any other agreement which establishes any corporation, partnership, limited liability company or other entity or organization that has voting Equity Interests directly or indirectly owned by Parent; provided, however, that, notwithstanding clause (g) of this definition, none of the following shall be a Joint Venture hereunder: (i) any wholly-owned Subsidiary and (ii) any trade creditor or customer in which Parent or any of its Subsidiaries has made an Investment pursuant to clause (t) of the definition of Permitted Investments.

"Landlord Reserve" means, as to each location not owned by a Loan Party at which a Borrower has Inventory of a type included in the Borrowing Base, Equipment of a type included in the Borrowing Base or books and records related to Accounts of a type included in the Borrowing Base or any such Inventory or Equipment located and as to which a Collateral Access Agreement has not been received by Agent, a reserve in an amount equal to the greater of (a) the number of months' rent for which the landlord, bailee or warehousemen will have, under applicable law, a Lien in the Inventory or Equipment of such Borrower to secure the payment of rent or other amounts under the lease or other agreement relative to such location, or (b) with respect to a location where Inventory is located, 6 months' rent and charges under the lease or other agreement relative to such location and with respect to a location where Equipment is located, 12 months' rent and charges under the lease or other agreement relative to such location.

"Lead Left Arranger" has the meaning specified therefor in Section 18.14(a) of the Agreement.

"Lender" has the meaning set forth in the preamble to the Agreement, shall include each Issuing Bank and each Swing Lender, and shall also include any other Person made a party to the Agreement pursuant to the provisions of <u>Section 13.1</u> of the Agreement and "<u>Lenders</u>" means each of the Lenders or any one or more of them. For avoidance of doubt, each Additional Lender is a Lender to the extent any such Person has executed and delivered an Incremental Amendment and to the extent such Incremental Amendment shall have become effective in accordance with the terms hereof and thereof.

"Lender Group" means each of the Lenders (including each Issuing Bank and each Swing Lender), Agent, and the Australian Security Trustee, or any one or more of them.

"Lender Group Expenses" means all (a) reasonable and documented costs and out-of-pocket expenses (including taxes and insurance premiums) required to be paid by any Borrower or its Subsidiaries under any of the Loan Documents that are paid, advanced, or incurred by Agent, (b) reasonable and documented out-of-pocket fees or charges paid or incurred by Agent and Australian Security Trustee in connection with the Lender Group's transactions with each Borrower and its Subsidiaries under any of the Loan Documents, including, photocopying, notarization, couriers and messengers, telecommunication, public record searches, filing fees, recording fees, publication, real estate surveys and environmental audits, (c) Agent's customary fees and charges imposed or incurred in connection with any background checks or OFAC/PEP searches related to any Borrower or its Subsidiaries, (d) Agent's customary fees and charges (as adjusted from time to time) with respect to the disbursement of funds (or the receipt of funds) to or for the account of any Borrower (whether by wire transfer or otherwise), together with any out-of-pocket costs and expenses incurred in connection therewith, (e) customary charges imposed or incurred by Agent resulting from the dishonor of checks payable by or to any Loan Party, (f) reasonable and documented out-of-pocket costs and expenses paid or incurred by Agent or Australian Security Trustee to correct any default or enforce any provision of the Loan Documents, or during the continuance of an Event of Default, in gaining possession of, maintaining, handling, preserving, storing, shipping, selling, preparing for sale, or advertising to sell the Collateral, or any portion thereof, irrespective of whether a sale is consummated, (g) field examination, appraisal, and valuation fees and expenses of Agent related to any field examinations, appraisals, or valuation to the extent of the fees and charges (and up to the amount of any limitation) provided in Section 2.10 of the Agreement, (h) Agent's and Australian Security Trustee's reasonable and documented costs and out-of-pocket expenses (including reasonable documented attorneys' fees and expenses) relative to third party claims or any other lawsuit or adverse proceeding paid or incurred, whether in enforcing or defending the Loan Documents or otherwise in connection with the transactions contemplated by the Loan Documents, Agent's or Australian Security Trustee's Liens in and to the Collateral, or the Lender Group's relationship with any Borrower or any of its Subsidiaries, (i) Agent's and Australian Security Trustee's reasonable and documented costs and out-of-pocket expenses (including reasonable and documented attorneys' fees and due diligence expenses) incurred in advising, structuring, drafting, reviewing, administering (including travel, meals, and lodging), syndicating (including reasonable and documented costs and out-of-pocket expenses relative to CUSIP, DXSyndicate™, SyndTrak or other communication costs incurred in connection with a syndication of the loan facilities), or amending, waiving, or modifying the Loan Documents, and (i) Agent's, Australian Security Trustee's and each Lender's reasonable and

documented costs and out-of-pocket expenses (including reasonable and documented attorneys', accountants', consultants', and other advisors' fees and expenses) incurred in terminating, enforcing (including attorneys', accountants', consultants', and other advisors' fees and expenses incurred in connection with a "workout," a "restructuring," or an Insolvency Proceeding concerning any Borrower or any of its Subsidiaries or in exercising rights or remedies under the Loan Documents), or defending the Loan Documents, irrespective of whether a lawsuit or other adverse proceeding is brought, or in taking any Enforcement Action or any Remedial Action with respect to the Collateral.

"Lender Group Representatives" has the meaning specified therefor in Section 18.9(a) of the Agreement.

"Lender-Related Person" means, with respect to any Lender, such Lender, together with such Lender's Affiliates, officers, directors, employees, attorneys, and agents.

"Letter of Credit" means a letter of credit (as that term is defined in the Code) issued by an Issuing Bank for the account of a Borrower pursuant to the Agreement, including, without limitation, the Existing Letters of Credit.

"Letter of Credit Collateralization" means either (a) providing cash collateral (pursuant to documentation reasonably satisfactory to Agent and the Issuing Bank, including provisions that specify that the Letter of Credit Fees and all commissions, fees, charges and expenses provided for in Section 2.11(I) and Section 2.12(I) of the Agreement (including any fronting fees) will continue to accrue while the Letters of Credit are outstanding) to be held by Agent for the benefit of the applicable Revolving Lenders in an amount equal to the sum of (i) 103% of the then existing Letter of Credit Usage that is denominated in 2013 of the then existing Letter of Credit Usage that is denominated in an Agreed Currency other than Dollars, (b) delivering to Agent documentation executed by all beneficiaries under the Letters of Credit, in form and substance reasonably satisfactory to Agent and Issuing Bank, terminating all of such beneficiaries' rights under the Letters of Credit, or (c) providing Agent with a standby letter of credit, in form and substance reasonably satisfactory to Agent, from a commercial bank acceptable to Agent (in its sole discretion) in an amount equal to the sum of (i) 103% of the then existing Letter of Credit Usage that is denominated in Dollars, and (ii) 103% of the then existing Letter of Credit Usage that is denominated in 2013 of the then existing Letter of Credit Usage that is denominated in 2014 of the Letters of Credit Fee and all fronting fees set forth in the Agreement will continue to accrue while the Letters of Credit are outstanding and that any such fees that accrue must be an amount that can be drawn under any such standby letter of credit).

"Letter of Credit Disbursement" means a payment made by any Issuing Bank pursuant to a Letter of Credit.

"Letter of Credit Expiration Date" shall mean the date which is five (5) Business Days prior to the Maturity Date (as determined pursuant to clause (a) of the definition thereof).

"Letter of Credit Fee" has the meaning specified therefor in Section 2.6(b)(ii) of the Agreement.

"Letter of Credit Usage" means, as of any date of determination, the aggregate undrawn amount of all outstanding Letters of Credit.

"LIBOR Deadline" has the meaning specified therefor in Section 2.13(b)(i) of the Agreement.

"LIBOR Notice" means a written notice substantially in the form of Exhibit L-1 to the Agreement.

"LIBOR Option" has the meaning specified therefor in Section 2.13(a) of the Agreement.

"LIBOR Rate" means the per annum rate of interest (rounded up, if necessary, to the nearest 1/8th of 1%) determined by Agent at or about 11:00 a.m. (London time) two Business Days prior to an interest period (but in the case of a LIBOR Rate Loan denominated in Pounds Sterling, Agent may determine LIBOR on the first day of the Interest Period), for a term equivalent to such period, equal to the London Interbank Offered Rate, or comparable or successor rate approved by Agent, as published on the applicable Reuters screen page (or other commercially available source designated by Agent from time to time); provided, that any such comparable or successor rate shall be applied by Agent, if administratively feasible, in a manner consistent with market practice (and, if any such rate is below zero, the LIBOR Rate shall be deemed to be zero).

"LIBOR Rate Loan" means each portion of a Loan that bears interest at a rate determined by reference to the LIBOR Rate.

"LIBOR Rate Margin" has the meaning set forth in the definition of Applicable Margin.

"LIBOR Successor Rate" has the meaning specified in Section 2.13(d)(iv) of this Agreement.

"LIBOR Successor Rate Conforming Changes" means, with respect to any proposed LIBOR Successor Rate, any conforming changes to the definition of Base Rate, Interest Period, timing and frequency of determining rates and making payments of interest and other administrative matters as may be appropriate, in the discretion of Agent, to reflect the adoption of such LIBOR Successor Rate and to permit the administration thereof by Agent in a manner substantially consistent with market practice (or, if Agent determines that adoption of any portion of such market practice is not administratively feasible or that no market practice for the administration of such LIBOR Successor Rate exists, in such other manner of administration as Agent determines in consultation with Parent).

"<u>Lien</u>" means any mortgage, deed of trust, pledge, hypothecation, assignment, charge, deposit arrangement, encumbrance, easement, lien (statutory or other), security interest, or other security arrangement and any other preference, priority, or preferential arrangement of any kind or nature whatsoever, including any conditional sale contract or other title retention agreement, the interest of a lessor under a Capital Lease and any synthetic or other financing lease having substantially the same economic effect as any of the foregoing.

"Line Cap" means, as of any date of determination, the sum of the U.S. Line Cap and the Australian Line Cap.

"Liquidity" means, as of any date of determination, the sum of Excess Availability plus Qualified Cash held in an account agreed between Parent and Agent (and any successor account agreed by Parent and Agent) as of such date of determination, which account is subject to the Cash Collateral Account letter agreement between Bank of America and the Parent dated as of March 30, 2015.

"Loan" means any Revolving Loan, Swing Loan, Incremental Loan or Extraordinary Advance made (or to be made) hereunder.

"Loan Account" means the U.S. Loan Account or the Australian Loan Account, as applicable.

"Loan Documents" means the Agreement, the Control Agreements, the Copyright Security Agreement, any Borrowing Base Certificate, the Fee Letter, the Guaranty and Security Agreement, the Australian Security Documents, the Intercompany Subordination Agreement, the Intercreditor Agreement, any Issuer Documents, the Letters of Credit, the Patent Security Agreement, the Trademark Security Agreement, any Incremental Amendment, any note or notes executed by Borrowers in connection with the Agreement and payable to any member of the Lender Group, and any other instrument or agreement entered into, now or in the future, by any Borrower or any of its Subsidiaries and any member of the Lender Group in connection with the Agreement.

"Loan Party" means any Borrower or any Guarantor. "Loan Parties" means, collectively, the Borrowers and the Guarantors.

"Margin Stock" as defined in Regulation U of the Board of Governors as in effect from time to time.

"Material Adverse Effect" means (a) a material adverse effect in the business, operations, results of operations, assets, liabilities or financial condition of Borrowers and their Subsidiaries, taken as a whole, (b) a material impairment of Borrowers' and their Subsidiaries' ability, when taken as a whole, to perform their obligations under the Loan Documents to which they are parties or of the Lender Group's ability to enforce the Obligations or realize upon a material portion of the Collateral (other than as a result of as a result of an action taken or not taken that is solely in the control of Agent), or (c) a material impairment of the enforceability or priority of Agent's or Australian Security Trustee's Liens with respect to all or a material portion of the Collateral; provided, however, that in no event shall (x) the Canadian Restructuring or its effect on the Canadian Entities constitute a Material Adverse Effect unless it constitutes a Material Adverse Effect on the Parent and its Subsidiaries, taken as a whole, (y) the failure of any Canadian Entity to repay any intercompany Indebtedness constitute a Material Adverse Effect.

"Material Contract" means, with respect to any Person, (a) the Joint Venture Agreements, (b) the Senior Secured Notes Documents, (c) the Existing Senior Notes Documents, (d) the Convertible Notes Documents and (e) any contract or agreement, the loss of which would reasonably be expected to result in a Material Adverse Effect.

"Material Subsidiary" means any Subsidiary that, together with its Subsidiaries, has (i) Consolidated Total Assets in excess of 3.0% of the Consolidated Total Assets of Parent and its Subsidiaries on a consolidated basis as of the date of the most recent consolidated balance sheet of Parent or (ii) consolidated total revenues in excess 3.0% of the consolidated total revenues of Parent and its Subsidiaries on a consolidated basis for the most recently ended four fiscal quarters for which internal financial statements of Parent are available immediately preceding such calculation date; provided that at no time shall any Subsidiary that is not a Material Subsidiary, when taken together with all other Subsidiaries that are not Material Subsidiaries, in each case together with their respective Subsidiaries, have (i) Consolidated Total Assets with a value in excess of 7.5% of the Consolidated Total Assets of Parent and its Subsidiaries on a consolidated basis or (ii) consolidated total revenues in excess of 7.5% of the consolidated total revenues of Parent and its Subsidiaries on a consolidated basis. For the avoidance of doubt, each Loan Party is a Material Subsidiary.

"Maturity Date" means the earlier of (a) February 28, 2023 and (b) the date that is 60 days before the stated maturity date of any portion of the Existing Debt, if (in the case of this clause (b)) on such date the aggregate principal amount of all Existing Debt outstanding that would mature on or before such 60th day exceeds \$50,000,000. References in the Agreement to "181 days after the Maturity Date" shall mean the Maturity Date as determined pursuant to clause (a) of this definition.

"Maximum Revolver Amount" means \$450,000,000, decreased by the amount of reductions in the Revolver Commitments made in accordance with Section 2.4(c)(ii) and Section 2.4(c)(iii) of the Agreement and as may be increased pursuant to Section 2.16.

"Mobile Equipment" means any forklifts, trailers, graders, dump trucks, water trucks, grapple trucks, lift trucks, flatbed trucks, fuel trucks, other trucks, dozers, cranes, loaders, skid steers, excavators, back hoes, shovels, drill crawlers, other drills, scrappers, graders, gondolas, flat cars, ore cars, shuttle cars, jenny cars, conveyors, locomotives, miners, other rail cars, and any other vehicles, mobile equipment and other equipment similar to any of the foregoing.

"Moody's" has the meaning specified therefor in the definition of Domestic Cash Equivalents.

"Multiemployer Plan" means any multiemployer plan within the meaning of Section 3(37) or 4001(a)(3) of ERISA with respect to which any Loan Party or any of its Subsidiaries or their respective ERISA Affiliates has an obligation to contribute or has any liability, contingent or otherwise or could be assessed Withdrawal Liability assuming a complete or partial withdrawal from any such multiemployer plan.

"Net Book Value" means, with respect to Equipment, the net book value under GAAP (but net of delivery charges, sales tax, federal excise tax and other costs incidental to the purchase thereof) as reported by Borrowers to Agent; <u>provided</u> that for purposes of the calculation of the U.S. Borrowing Base or the Australian Borrowing Base, the Net Book Value of the Equipment shall not include (a) the portion of the value of the Equipment equal to the profit earned by any Affiliate of Parent on the sale thereof to any Loan Party, or (b) write-ups or write-downs in value with respect to currency exchange rates.

"Net Orderly Liquidation Value" means, with respect to Equipment or Inventory, the orderly liquidation value of such Equipment or Inventory determined for each category of such Equipment or Inventory (but net of all associated costs and expenses of such liquidation) as specified in the most recent appraisal received by Agent from an appraisal company reasonably acceptable to Agent in its Permitted Discretion. With respect to Eligible Equipment, the Net Orderly Liquidation Value shall be determined as of the date of each Borrowing Base Certificate giving effect to depreciation and assuming that since the last appraisal the ratio of Net Book Value to Net Orderly Liquidation Value has remained constant.

"Non-Consenting Lender" has the meaning specified therefor in Section 14.2(a) of the Agreement.

"Non-Defaulting Lender" means each Lender other than a Defaulting Lender.

"Notification Event" means (a) the occurrence of a "reportable event" described in Section 4043 of ERISA for which the 30-day notice requirement has not been waived by applicable regulations issued by the PBGC with respect to a Pension Plan, (b) the withdrawal of any Loan Party or ERISA Affiliate from a Pension Plan during a plan year in which it was a "substantial employer" as defined in Section 4001(a)(2) of ERISA, (c) the termination of a Pension Plan, the filling of a notice of intent to terminate a Pension Plan or the treatment of a Pension Plan amendment as a termination, under Section 4041 of ERISA, if the plan assets are not sufficient to pay all plan liabilities, (d) the institution of proceedings

to terminate, or the appointment of a trustee with respect to, any Pension Plan or Multiemployer Plan by the PBGC, (e) any other event or condition that would constitute grounds under Sections 4042(a)(1)-(3) of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan, (f) the imposition of a Lien pursuant to the IRC or ERISA in connection with any Employee Benefit Plan or the existence of any facts or circumstances that would reasonably be expected to result in the imposition of a Lien, (g) the partial or complete withdrawal of any Loan Party or ERISA Affiliate from a Multiemployer Plan or the receipt of notification that a partial or complete withdrawal has occurred, (h) the reorganization or insolvency of a Multiemployer Plan under ERISA, (i) the termination of a Multiemployer Plan under Section 4041A of ERISA or the institution by the PBGC of proceedings to terminate or to appoint a trustee to administer a Multiemployer Plan under ERISA including the filing of a notice of reorganization, insolvency or termination (or treatment of a plan amendment as termination) under Section 4041A of ERISA, (j) any Pension Plan being in "at risk status" within the meaning of IRC Section 430(i), (k) receipt of notification by any Loan Party or ERISA Affiliate that any Multiemployer Plan being in "endangered status" or "critical status" within the meaning of IRC Section 432(b) or the determination that any Multiemployer Plan is insolvent or in reorganization within the meaning of Title IV of ERISA, (l) with respect to any Pension Plan to meet the minimum funding standards within the meaning of the IRC or Section 302 of ERISA), in each case, whether or not waived, (n) the filing of an application for a waiver of the minimum funding standards within the meaning of the IRC or Section 302 of ERISA) with respect to any Pension Plan, or (o) any event that results in or would reasonably be expected to result in a liability by a Loan Party pursuant to the excise tax provisions of the IRC relating

"Obligations" means the U.S. Obligations and the Australian Obligations; provided that the Obligations shall not include any Excluded Swap Obligations.

"OFAC" means The Office of Foreign Assets Control of the U.S. Department of the Treasury.

"Offshore Associate" shall mean an Associate (a)(x) which is a non-resident of Australia and does not become a Lender or receive payment in carrying on a business in Australia at or through a permanent establishment of the Associate in Australia or (y) which is a resident of Australia and which becomes a Lender or receives a payment in carrying on a business in a country outside Australia at or through a permanent establishment of the Associate in that country; and (b) which does not become a Lender and receive payment in the capacity of a clearing house, custodian, funds manager or responsible entity of a registered scheme.

"OID" means original issue discount.

"Originating Lender" has the meaning specified therefor in Section 13.1(e) of the Agreement.

"Other Canadian Entities" means each subsidiary of Parent (other than the CCAA entities) organized under the laws of Canada or any province thereof and Wabush Iron Co. Limited, an Ohio corporation, including, for the avoidance of doubt, Wabush Mines, an unincorporated joint venture and Knoll Lake Minerals Limited, a company organized under the laws of Canada, other than any subsidiary of Cliffs Natural Resources Exploration Inc.

"Overadvance" means, as of any date of determination, that the Revolver Usage is greater than any of the limitations set forth in Section 2.1, Section 2.11, or Section 2.12.

"Parent" has the meaning specified therefor in the preamble to the Agreement.

"Participant" has the meaning specified therefor in Section 13.1(e) of the Agreement.

"Participant Register" has the meaning set forth in Section 13.1(i) of the Agreement.

"Participating Member State" means any member state of the European Union that adopts or has adopted the Euro as its lawful currency in accordance with legislation of the European Union relating to economic and monetary union.

"Patent Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"Patriot Act" has the meaning specified therefor in Section 4.13 of the Agreement.

"Payment Conditions" means:

- (a) with respect to any proposed transaction pursuant to clause (g) or clause (m) of the definition of Permitted Investments or clause (d) of the definition of Permitted Acquisitions, that (A) no Default or Event of Default has occurred and is continuing or would result therefrom and (B) (i) Borrowers' Excess Availability for each of the 30 consecutive days immediately preceding the date of such transaction, and both immediately before and immediately after giving effect to such transaction, is in excess of the greater of (I) \$70,000,000 and (II) 17.5% of the Line Cap or (ii) (1) Borrowers' Excess Availability for each of the 30 consecutive days immediately preceding the date of such transaction, and both immediately before and immediately after giving effect to such transaction, is in excess of the greater of (I) \$50,000,000 and (II) 12.5% of the Line Cap and (2) immediately after giving pro forma effect to such transaction, the Fixed Charge Coverage Ratio of Parent and its Subsidiaries is at least 1.00:1.00 as of the 12-month period ending as of the most recently ended fiscal quarter (calculated assuming that such transaction was consummated as of the end of such fiscal quarter); and
- (b) with respect to any other proposed transaction, that (A) no Default or Event of Default has occurred and is continuing or would result therefrom and (B) (i) Borrowers' Excess Availability for each of the 30 consecutive days immediately preceding the date of such transaction, and both immediately before and immediately after giving effect to such transaction, is in excess of the greater of (I) \$85,000,000 and (II) 20% of the Line Cap or (ii) (1) Borrowers' Excess Availability for each of the 30 consecutive days immediately preceding the date of such transaction, and both immediately before and immediately after giving effect to such transaction, is in excess of the greater of (I) \$60,000,000 and (II) 15% of the Line Cap and (2) immediately after giving pro forma effect to such transaction, the Fixed Charge Coverage Ratio of Parent and its Subsidiaries is at least 1.00:1.00 as of the 12-month period ending as of the most recently ended fiscal quarter (calculated assuming that such transaction was consummated as of the end of such fiscal quarter).

"PBGC" means the Pension Benefit Guaranty Corporation or any successor agency.

"Pension Plan" means any employee benefit plan, other than a Multiemployer Plan, which is subject to the provisions of Title IV or Section 302 of ERISA or Sections 412 or 430 of the IRC sponsored, maintained, or contributed to by any Loan Party, Subsidiary or their respective ERISA Affiliates or to which any Loan Party, Subsidiary or their respective ERISA Affiliate has any liability, contingent or otherwise.

"Perfection Certificates" means the certificates substantially in the forms of Exhibit P-1 and Exhibit P-2 to the Agreement.

"Permitted Acquisition" means any Acquisition so long as:

- (a) no Default or Event of Default shall have occurred and be continuing or would result from the consummation of the proposed Acquisition and the proposed Acquisition is consensual,
- (b) for any Permitted Acquisition pursuant to which the aggregate purchase price is greater than \$100,000,000, Borrowers have provided Agent with five (5) Business Days' (or such shorter period as Agent may agree) prior notice, and such information, including historical financial statements and projections, as Agent shall have reasonably requested (to the extent available).
- (c) for any Permitted Acquisition pursuant to which the aggregate purchase price is greater than \$100,000,000, Borrowers have provided Agent with written confirmation, supported by reasonably detailed calculations, that on a pro forma basis (including pro forma adjustments arising out of events which are directly attributable to such proposed Acquisition, are factually supportable, and are expected to have a continuing impact, in each case, determined as if the combination had been accomplished at the beginning of the relevant period; such eliminations and inclusions determined on a basis consistent with Article 11 of Regulation S-X promulgated under the Securities Act and as interpreted by the staff of the SEC) created by adding the historical combined financial statements of Parent (including the combined financial statements of any other Person or assets that were the subject of a prior Permitted Acquisition during the relevant period) to the historical consolidated financial statements of the Person to be acquired (or the historical financial statements related to the assets to be acquired) pursuant to the proposed Acquisition, Parent and its Subsidiaries (i) would have been in compliance with the financial covenant in Section 7 of the Agreement (regardless of whether a Financial Covenant Period is then in effect) for the fiscal quarter ended immediately prior to the proposed date of consummation of such proposed Acquisition and (ii) are projected to be in compliance with the financial covenant

in <u>Section 7</u> of the Agreement (regardless of whether a Financial Covenant Period is then in effect) for each of the four (4) fiscal quarters in the period ended one year after the proposed date of consummation of such proposed Acquisition,

- (d) the Payment Conditions will be satisfied, and
- (e) the subject assets or Equity Interests, as applicable, are being acquired directly by a Borrower or one of its Subsidiaries that is a Loan Party, and, in connection therewith, the applicable Loan Party shall have complied with Section 5.11 or 5.12 of the Agreement, as applicable.

"Permitted Discretion" means a determination made in good faith based upon the exercise of reasonable (from the perspective of a secured asset-based lender) business judgment.

"Permitted Dispositions" means:

- (a) sales, abandonment, or other dispositions of any personal property or Real Property that, in the reasonable judgment of any Borrower or any of its Subsidiaries, has become obsolete, worn out or no longer used or useful or, except with respect to personal property included in the Borrowing Base, has become uneconomic, in each case in the ordinary course of business and (ii) licenses, leases or subleases of Real Property or personal property in the ordinary course of business so long as such licenses, leases or subleases do not individually or in the aggregate interfere in any material respect with the ordinary conduct of the business of Borrowers and their Subsidiaries,
 - (b) sales of Inventory to buyers in the ordinary course of business,
 - (c) the use or transfer of money or Cash Equivalents in a manner that is not prohibited by the terms of the Agreement or the other Loan Documents,
 - (d) the licensing or sublicensing of patents, trademarks, copyrights, and other intellectual property rights in the ordinary course of business,
 - (e) the granting of Permitted Liens,
- (f) the sale or discount, in each case without recourse, of accounts receivable arising in the ordinary course of business, but only in connection with the compromise or collection thereof,
 - (g) any involuntary loss, damage or destruction of property,
- (h) any involuntary condemnation, seizure or taking, by exercise of the power of eminent domain or otherwise, or confiscation or requisition of use of property,
 - (i) the licensing, leasing or subleasing of assets (other than Real Property) of any Borrower or its Subsidiaries in the ordinary course of business,
 - (j) the sale or issuance of Equity Interests (other than Disqualified Equity Interests) of Parent,
- (k) the lapse or abandonment of registered patents, trademarks, copyrights and other intellectual property of any Borrower or any of its Subsidiaries to the extent that, in the reasonable judgment of such Borrower or Subsidiary, such intellectual property is not economically desirable in the conduct of its business,
 - (I) the making of Restricted Payments (in cash or stock) that are expressly permitted to be made pursuant to the Agreement,
 - (m) the making of Permitted Investments (in cash) or the consummation of a transaction permitted by Section 6.3,
- (n) the sale, transfer, lease or other disposition of assets (i) from any Loan Party to another Loan Party, (ii) from any Subsidiary of any Borrower that is not a Loan Party to any Loan Party or any other Subsidiary of any Borrower, or (iii) from any Loan Party to a Subsidiary that is not a Loan Party, in the case of this clause (iii) only, in an aggregate amount not exceed \$10,000,000 during the term of this Agreement,

- (o) dispositions of assets acquired by Borrowers and the Subsidiaries pursuant to a Permitted Acquisition consummated within 12 months of the date of the proposed disposition so long as (i) the consideration received for the assets to be so disposed is at least equal to the fair market value of such assets and for at least 75% cash, (ii) the assets to be so disposed are not necessary or economically desirable in connection with the business of Borrowers and their Subsidiaries, (iii) the assets to be so disposed are readily identifiable as assets acquired pursuant to the subject Permitted Acquisition and (iv) if any such assets consist of ABL Collateral that are of a type included in the Borrowing Base and have a fair market value in excess of \$5,000,000, the Parent shall have delivered an updated Borrowing Base Certificate reflecting the disposition of those assets,
 - (p) sales or dispositions set forth on Schedule P-3 to the Agreement,
- (q) sales or dispositions of assets not otherwise permitted in clauses (a) through (p) above or clauses (r) through (v) below so long as (i) made at fair market value for at least 75% cash, (ii) if any such assets consist of ABL Collateral that are of a type included in the Borrowing Base and have a fair market value in excess of \$5,000,000, the Parent shall have delivered an updated Borrowing Base Certificate reflecting the disposition of those assets and (iii) the aggregate fair market value of all assets disposed of during the term of the Agreement (including the proposed disposition) would not exceed \$100,000,000,
- (r) sales or dispositions of assets for fair market value and at least 75% cash so long as (i) the Payment Conditions are satisfied and (ii) if any such assets consist of ABL Collateral that are of a type included in the Borrowing Base and have a fair market value in excess of \$5,000,000, the Parent shall have delivered an updated Borrowing Base Certificate reflecting the disposition of those assets,
- (s) parting with possession by an Australian Loan Party of goods to any mining contractor in the ordinary course of business where the Australian Loan Party retains ownership of those goods and (if the parting with possession constitutes a security interest under the Australian PPSA) it has perfected an appropriate security interest against the relevant mining contractor in accordance with the Australian PPSA which covers those goods,
- (t) the direct or indirect sale or other disposition of Collateral by a Loan Party in connection with any Permitted Joint Venture; provided that such Loan Party receives consideration at least equal to the fair market value of the Collateral subject to the sale or other disposition (and, in the case of ABL Collateral, for 100% cash),
- (u) the sale or disposition of any asset of an Australian Subsidiary, including the sale by an Australian Subsidiary or any part of its entire business and mining operations in any manner (including by way of asset sales, a share sale, or liquidation) on arm's length terms (provided that, if such sale or disposition occurs prior to the Australian End Date and such Australian Subsidiary's assets then contribute to the Borrowing Base, the Parent shall have delivered an updated Borrowing Base Certificate), and
 - (v) any Permitted Reorganization.

provided that (i) for purposes of clauses (o), (q) and (r) above the following shall be deemed cash with respect to the disposition of assets (other than ABL Collateral) (A) the repayment or assumption by the transferee of Indebtedness secured by Liens with a priority to the Liens securing the Obligations (other than Indebtedness incurred in contemplation of such disposition), (B) the repayment or assumption by the transferee of liabilities (as shown on the Parent's most recent balance sheet or in the notes thereto), other than liabilities that are subordinated in right of payment to the Obligations, (C) any securities, notes or other obligations received by any Loan Party that are, within 180 days of the disposition of such assets, converted by such Loan Party into cash or Cash Equivalents and (D) any Designated Non-Cash Consideration received by such Loan Party in such disposition having an aggregate fair market value, taken together with all other Designated Non-Cash Consideration received pursuant to this proviso that has at that time not been converted into cash or Cash Equivalents not to exceed the greater of (x) \$75.0 million and (y) 2.0% of Consolidated Net Tangible Assets at the time of the receipt of such Designated Non-Cash Consideration (with the fair market value of each item of Designated Non-Cash Consideration being measured at the time received and without giving effect to subsequent changes in value), (ii) in no event shall any disposition of any material portion of the U.S. Iron Ore Business, constitute a Permitted Disposition, and (iii) in connection with a disposition of assets of any Canadian Entity or the disposition of any Equity Interests of a Canadian Entity made pursuant to clause (r) above, the Payment Conditions need not be satisfied.

"Permitted Indebtedness" means:

- (a) Indebtedness evidenced by the Agreement or the other Loan Documents,
- (b) Indebtedness set forth on Schedule P-4 to the Agreement and any Refinancing Indebtedness in respect of such Indebtedness,
- (c) Permitted Purchase Money Indebtedness in an aggregate amount outstanding at any time not to exceed \$300,000,000 and any Refinancing Indebtedness in respect of such Indebtedness,
 - (d) endorsement of instruments or other payment items for deposit,
- (e) Indebtedness (i) incurred in the ordinary course of business in respect of surety and appeal bonds, performance bonds, bid bonds, appeal bonds, reclamation bonds, completion guarantee and similar obligations, or any similar financial assurance obligations under Environmental Laws or worker's compensation laws or with respect to self-insurance obligations and (ii) consisting of guarantees arising with respect to customary indemnification obligations to purchasers in connection with Permitted Dispositions,
- (f) (i) Indebtedness of any Borrower that is incurred on the date of the consummation of a Permitted Acquisition solely for the purpose of consummating such Permitted Acquisition so long as (A) no Event of Default has occurred and is continuing or would result therefrom, (B) such Indebtedness is not incurred for working capital purposes, (C) such Indebtedness does not mature prior to the date that is 181 days after the Maturity Date, (D) such Indebtedness does not amortize until 181 days after the Maturity Date, (E) such Indebtedness does not provide for the payment of interest thereon in cash or Cash Equivalents other than current market interest, as determined by the Parent in its reasonable business judgment, (F) if such Indebtedness is secured (1) the Liens securing such Indebtedness shall be pari passu with or junior to the Liens securing the Senior Secured Notes (or if the Senior Secured Notes are no longer outstanding at such time, such Liens would have been pari passu with or junior to the Liens securing the Senior Secured Notes had the Senior Secured Notes been outstanding), (2) the Liens securing such Indebtedness are subject to an intercreditor agreement in form and substance reasonably satisfactory to Agent (it being agreed the form and substance of the Intercreditor Agreement is acceptable), (3) such Indebtedness shall not be secured by ABL Collateral and (4) on a pro forma basis after giving effect to such Indebtedness, Parent is in compliance with the financial covenant set forth in Section 7 of this Agreement (regardless of whether a Financial Covenant Period is then in effect), and (ii) any Refinancing Indebtedness in respect of such Indebtedness.
- (g) Acquired Indebtedness in an aggregate principal amount not to exceed \$25,000,000 outstanding at any one time and any Refinancing Indebtedness in respect thereof,
- (h) Indebtedness (i) incurred in the ordinary course of business under performance, surety, statutory, or appeal bonds or guarantees, and completion guarantees (or obligations in respect of letters of credit related thereto in an amount not to exceed \$15,000,000 outstanding at any one time) and (ii) in respect of letters of credit which are backstopped or supported by a Letter of Credit issued hereunder,
- (i) Indebtedness owed to any Person providing property, casualty, liability, or other insurance to any Borrower or any of its Subsidiaries, so long as the amount of such Indebtedness is not in excess of the amount of the unpaid cost of, and shall be incurred only to defer the cost of, such insurance for the year in which such Indebtedness is incurred and such Indebtedness is outstanding only during such year,
- (j) the incurrence by Parent or any of its Subsidiaries of Indebtedness under Hedge Agreements that are incurred for the bona fide purpose of hedging the interest rate, commodity, or foreign currency risks associated with Parent's and its Subsidiaries' operations and not for speculative purposes,
- (k) Indebtedness incurred in the ordinary course of business in respect of credit cards, credit card processing services, debit cards, stored value cards, commercial cards (including so-called "purchase cards", "procurement cards" or "p-cards"), Cash Management Services, or other Bank Products,
- (I) Guaranty Obligations in respect of Indebtedness otherwise permitted under this definition; <u>provided</u> that any guaranty by a Loan Party of Indebtedness of a Subsidiary that is not a Loan Party must be permitted by <u>clause (q)</u> of the definition of "Permitted Investment",

- (m) contingent liabilities in respect of any indemnification obligation, adjustment of purchase price, non-compete, or similar obligation of any Loan Party incurred in connection with the consummation of one or more Permitted Acquisitions,
 - (n) Indebtedness composing Permitted Investments (other than clause (u)(i) of the definition thereof),
- (o) Indebtedness incurred in respect of netting services, overdraft protection, and other like services, in each case, incurred in the ordinary course of business.
- (p) Indebtedness of any Borrower or its Subsidiaries in respect of earn-outs owing to sellers of assets or Equity Interests to such Borrower or its Subsidiaries that is incurred in connection with the consummation of one or more Permitted Acquisitions and any Refinancing Indebtedness in respect of such Indebtedness,
- (q) Indebtedness incurred in connection with any sale/leaseback transaction and any Refinancing Indebtedness in respect of such Indebtedness; provided, that such Indebtedness incurred from and after the Closing Date shall be in an aggregate principal amount not to exceed \$100,000,000 at any time outstanding,
 - (r) customer advances for prepayment of ore sales,
- (s) accrual of interest, accretion or amortization of original issue discount, or the payment of interest in kind, in each case, on Indebtedness that otherwise constitutes Permitted Indebtedness,
- (t) Indebtedness evidenced by the Existing Senior Notes in an aggregate principal amount not to exceed the aggregate principal amount thereof outstanding on the Closing Date (after giving effect to any exchange offers in respect thereof consummated on or before the Closing Date), and any Refinancing Indebtedness in respect thereof,
- (u) (i) Indebtedness evidenced by the Senior Secured Notes in an aggregate principal amount not to exceed \$400,000,000 at any one time outstanding and any Refinancing Indebtedness in respect thereof, (ii) [reserved] and (iii) Indebtedness in an aggregate principal amount not to exceed the greater of (x) an amount equal to (1) \$1,250,000,000 minus (2) the principal amount of Indebtedness incurred pursuant to clause (u)(i) above and (y) an amount that, on a pro forma basis upon giving effect the incurrence thereof, would not cause Parent's "Consolidated Secured Leverage Ratio" (as defined in the Senior Secured Notes Indenture as of the date hereof) to exceed 3.00:1.00 (and any Refinancing Indebtedness in respect of any such Indebtedness), so long as, in the case of this clause (iii), (A) if such Indebtedness is secured, (x) (1) the Liens securing such Indebtedness shall be *pari passu* with or junior to the Liens securing the Senior Secured Notes of the Senior Secured Notes are no longer outstanding at such time, such Liens would have been *pari passu* with or junior to the Liens securing the Senior Secured Notes had the Senior Secured Notes been outstanding), and (y) the Liens securing such Indebtedness are subject to an intercreditor agreement in form and substance satisfactory to Agent (it being agreed the form and substance of the Intercreditor Agreement is acceptable), (B) such Indebtedness does not mature prior to the date that is 181 days after the Maturity Date, (C) such Indebtedness does not amortize until 181 days after the Maturity Date, and (D) after giving effect to the incurrence of any such Indebtedness after the Closing Date, Parent is in compliance with the financial covenant set forth in Section 7 of this Agreement (regardless of whether a Financial Covenant Period is then in effect),
- (v) any other unsecured Indebtedness incurred by any Borrower or any of its Subsidiaries so long as (i) no Default or Event of Default has occurred and is continuing or would result from the incurrence of such Indebtedness, (ii) such Indebtedness does not mature prior to the date that is 181 days after the Maturity Date, (iii) such Indebtedness does not amortize until 181 days after the Maturity Date and (iv) after giving effect to the incurrence of any such Indebtedness, Parent is in compliance with the financial covenant set forth in <u>Section 7</u> of this Agreement (regardless of whether a Financial Covenant Period is then in effect); <u>provided</u> that conditions (ii) and (iii) above shall not be required to be satisfied with respect to Indebtedness in an aggregate principal amount not to exceed \$25,000,000 during the term of the Agreement.
- (w) Indebtedness of Excluded Subsidiaries (i) listed on Schedule E-3 and any Refinancing Indebtedness in respect of such Indebtedness, and (ii) other Indebtedness in an aggregate amount at any time outstanding not to exceed \$75,000,000,
 - (x) Guaranty Obligations of Indebtedness of the Canadian Entities existing as of the Closing Date,
 - (y) Permitted Intercompany Advances and Contributions, and

(z) Indebtedness evidenced by the Convertible Notes in an aggregate principal amount at any time outstanding not to exceed \$316,250,000 and any Refinancing Indebtedness in respect thereof.

"Permitted Intercompany Advances and Contribution" means loans, advances or capital contributions made by (a) a U.S. Loan Party to another U.S. Loan Party, (b) an Australian Loan Party to another Australian Loan Party or a U.S. Loan Party, (c) a Subsidiary of a Borrower that is not a Loan Party to a Canadian Entity or another Subsidiary of a Borrower that is not a Loan Party, (d) a Subsidiary of a Borrower that is not a Loan Party to a Loan Party to a Loan Party to a Loan Party to the Intercompany Subordination Agreement, (e) a Loan Party to a Joint Venture so long as such loans, advances or capital contributions are required by and made in accordance with the Joint Venture Agreements as in effect on the date hereof, (f) a Loan Party to a Joint Venture, Subsidiary or Canadian Entity, the sum of which shall not exceed an aggregate amount equal to \$200,000,000 at any time outstanding (of which the sum, without duplication, of (1) the aggregate amount of Investments made in, to or on behalf of the Canadian Entities from and after the Closing Date under any "e-payables" program of Parent or any of its Subsidiaries in respect of amounts due to any supplier or vendor of the Canadian Entities, shall not exceed an aggregate amount equal to \$100,000,000), provided that with respect to a Canadian Entity, from and after the Canadian Entities, shall not exceed an aggregate amount equal to \$100,000,000), provided that with respect to a Canadian Entity, from and after the Canadian Restructuring Commencement Date for such Canadian Entity, any Investments made under clause (f) hereof in or to the Canadian Entities shall be made in the form of intercompany loans that are secured by the assets of the Canadian Entities, and (g) Investments (other than transfers of assets that constitute Accounts, Inventory or Mobile Equipment) in IronUnits LLC which shall not exceed an aggregate amount equal to \$250,000,000; provided that, in the case of Investments made in reliance on clause (g) above that are made at a time when any Loans are o

"Permitted Investments" means:

- (a) Investments in cash and Cash Equivalents,
- (b) Investments in negotiable instruments deposited or to be deposited for collection in the ordinary course of business,
- (c) advances made in connection with purchases of goods or services in the ordinary course of business,
- (d) Investments received in settlement of amounts due to any Loan Party or any of its Subsidiaries effected in the ordinary course of business or owing to any Loan Party or any of its Subsidiaries as a result of Insolvency Proceedings involving an account debtor or upon the foreclosure or enforcement of any Lien in favor of a Loan Party or its Subsidiaries,
- (e) Investments owned by any Loan Party or any of its Subsidiaries on the Closing Date in their respective Subsidiaries and the Canadian Entities and other Investments set forth on Schedule P-1 to the Agreement,
 - (f) guarantees and other contingent liabilities permitted under the definition of Permitted Indebtedness (other than clause (n) thereof),
 - (g) Permitted Intercompany Advances and Contributions,
- (h) Equity Interests or other securities acquired in connection with the satisfaction or enforcement of Indebtedness or claims due or owing to a Loan Party or its Subsidiaries (in bankruptcy of customers or suppliers or otherwise outside the ordinary course of business) or as security for any such Indebtedness or claims,
 - (i) deposits of cash made in the ordinary course of business to secure performance of operating leases,
- (j) loans and advances to employees, officers, and directors of a Borrower or any of its Subsidiaries for bona fide business purposes in the ordinary course of business,
 - (k) Permitted Acquisitions,
 - (I) acquisitions of property, plant and equipment to be used in the ordinary course of business,

- (m) Investments so long as the Payment Conditions are satisfied at the time thereof,
- (n) Investments resulting from entering into (i) Bank Product Agreements, or (ii) agreements relative to Indebtedness that is permitted under clause (j) of the definition of Permitted Indebtedness,
- (o) equity Investments by any Loan Party in any Subsidiary of such Loan Party which is required by law to maintain a minimum net capital requirement or as may be otherwise required by applicable law,
- (p) Investments held by a Person acquired in a Permitted Acquisition to the extent that such Investments were not made in contemplation of or in connection with such Permitted Acquisition and were in existence on the date of such Permitted Acquisition.
- (q) so long as no Event of Default has occurred and is continuing or would result therefrom, any other Investments in an aggregate amount not to exceed \$25,000,000 during the term of the Agreement,
 - (r) Hedge Obligations incurred in the ordinary course of business and not for speculative purposes,
 - (s) mergers, consolidations or amalgamations permitted by Section 6.3,
- (t) Investments in securities of trade creditors or customers in the ordinary course of business that are received (i) in settlement of bona fide disputes or pursuant to any plan of reorganization or liquidation or similar arrangement upon the bankruptcy or insolvency of such trade creditors or customers or (ii) in the settlement of debts created in the ordinary course of business,
 - (u) Permitted Dispositions (other than in reliance on clause (m) thereof), and
- (v) Investments in Joint Ventures in the United States existing as of the Closing Date for the purpose of financing such entities' (i) operating expenses incurred in the ordinary course of business, (ii) reasonable Capital Expenditures and (iii) other reasonable obligations that are accounted for by the Parent and its Subsidiaries as increases in equity in such Joint Ventures.

"Permitted Joint Venture" means any Person at least 50% of the capital stock of which is owned by a Loan Party if (a) such Person is engaged in a business related to that of a Loan Party and (b) the Loan Party has the right to appoint at least half of the members of the Board of Directors (or equivalent governing body, including, without limitation, of the general partner of a limited partnership) of such Person.

"Permitted Liens" means

- (a) Liens granted to, or for the benefit of, Agent or the Australian Security Trustee to secure the Obligations or any portion thereof,
- (b) Liens for unpaid taxes, assessments, or other governmental charges or levies that either (i) are not yet delinquent by more than 30 days, or (ii) do not have priority over Agent's or the Australian Security Trustee's Liens with respect to the ABL Collateral and the underlying taxes, assessments, or charges or levies are the subject of Permitted Protests,
- (c) judgment Liens arising solely as a result of the existence of judgments, orders, or awards that do not constitute an Event of Default under <u>Section 8.3</u> of the Agreement,
- (d) Liens set forth on <u>Schedule P-2</u> to the Agreement, any continuation or extension thereof; <u>provided</u>, that to qualify as a Permitted Lien, any such Lien described on <u>Schedule P-2</u> to the Agreement shall only secure the Indebtedness that it secures on the Closing Date and any Refinancing Indebtedness in respect thereof,
 - (e) the interests of lessors under operating leases and non-exclusive licensors under license agreements,
- (f) purchase money Liens or the interests of lessors under Capital Leases to the extent that such Liens or interests secure Permitted Purchase Money Indebtedness and so long as (i) in the case of Permitted Purchase Money Indebtedness described in clauses (a) and (b) of the definition thereof, (A) such Lien attaches only to the asset purchased or acquired and the proceeds thereof, and (B) such Lien only secures the Indebtedness that was incurred

to acquire the asset purchased or acquired or any Refinancing Indebtedness in respect thereof, and (ii) in the case of Permitted Purchase Money Indebtedness described in clauses (a) through (c) of the definition thereof, such Lien does not attach to ABL Collateral (other than with respect to Permitted Purchase Money Indebtedness (and Refinancing Indebtedness in respect thereof) in an aggregate principal amount not to exceed \$50,000,000 at any time outstanding secured by Mobile Equipment (and not any other ABL Collateral) but only to the extent that Parent shall have delivered a notice to Agent clearly identifying the Mobile Equipment subject to such Lien prior to the granting of such Lien).

- (g) Liens arising by operation of law in favor of warehousemen, landlords, carriers, vendors', mechanics, materialmen, laborers, suppliers and other like Liens, incurred in the ordinary course of business, and which Liens either (i) are for sums not yet delinquent by more than 30 days, or (ii) are the subject of Permitted Protests.
- (h) Liens on amounts pledged or deposited to secure any Borrower's and its Subsidiaries obligations in connection with worker's compensation, other unemployment insurance and similar legislation.
- (i) Liens on amounts deposited to secure any Borrower's and its Subsidiaries obligations in connection with the making or entering into of bids, tenders, or leases in the ordinary course of business and not in connection with the borrowing of money,
- (j) Liens on amounts deposited to secure any Borrower's and its Subsidiaries reimbursement obligations with respect to surety or appeal bonds obtained in the ordinary course of business,
- (k) with respect to any Real Property, easements, rights of way, zoning and similar restrictions, reservations (including severances, leases or reservations of oil, gas, coal, minerals or water rights), restrictions or encumbrances in respect of real property or title defects that were not incurred in connection with indebtedness and do not in the aggregate materially impair their use in the operation of the business of the Parent and its Subsidiaries,
 - (I) non-exclusive licenses of patents, trademarks, copyrights, and other intellectual property rights in the ordinary course of business,
- (m) Liens that are replacements of Permitted Liens to the extent that the original Indebtedness is the subject of permitted Refinancing Indebtedness and so long as the replacement Liens only encumber those assets that secured the original Indebtedness,
- (n) rights of setoff, bankers' liens and other similar Liens upon deposits of funds in favor of banks or other depository institutions, solely to the extent incurred in connection with the maintenance of such Deposit Accounts in the ordinary course of business,
- (o) Liens granted in the ordinary course of business on the unearned portion of insurance premiums securing the financing of insurance premiums to the extent the financing is permitted under the definition of Permitted Indebtedness,
- (p) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods,
- (q) Liens solely on any cash earnest money deposits made by a Borrower or any of its Subsidiaries in connection with any letter of intent or purchase agreement with respect to a Permitted Acquisition,
 - (r) Liens assumed by any Borrower or its Subsidiaries in connection with a Permitted Acquisition that secure Acquired Indebtedness,
- (s) Liens on the Collateral securing the Senior Secured Notes and any Indebtedness incurred pursuant to clause (f) or (u) of the definition of Permitted Indebtedness and Refinancing Indebtedness in respect thereof, so long as such Liens are subject to the Intercreditor Agreement or such intercreditor agreement in form and substance satisfactory to the Agent,
- (t) Liens in the nature of royalties, dedications of reserves under supply agreements, mining leases or similar rights or interests granted, taken subject to or otherwise imposed on properties consistent with normal practices in the mining industry and any precautionary UCC financing statement filings in respect of leases or consignment arrangements (and not any Indebtedness) entered into in the ordinary course of business,

- (u) leases or subleases of properties, in each case entered into in the ordinary course of business so long as such leases or subleases do not, individually or in the aggregate, (i) interfere in any material respect with the ordinary conduct of the business of Borrowers or their respective Subsidiaries or (ii) materially impair the use or the value of the property subject thereto,
- (v) Liens arising out of conditional sale, title retention, consignment or similar arrangements for the sale of goods entered into in the ordinary course of business in accordance with the past business practices of such Person, and any products or proceeds thereof to the extent covered by such Liens,
 - (w) the filing of UCC financing statements in connection with operating leases, consignment of goods or bailment agreements,
 - (x) Liens on assets of a Subsidiary that is not a Loan Party in favor of a Borrower or a Subsidiary of a Borrower,
- (y) Liens created solely for the purpose of securing Indebtedness permitted by clause (q) of the definition of "Permitted Indebtedness"; <u>provided</u> that any such Liens attach only to the property being leased or acquired pursuant to such Indebtedness and do not encumber any other property (other than any products or proceeds thereof to the extent covered by such Liens),
- (z) Liens on (i) letters of credit securing another standby letter of credit and (ii) cash or Cash Equivalents securing reimbursement or counterindemnity obligations with respect to any standby letter of credit as to which the aggregate amount of the obligations secured thereby does not exceed \$15,000,000 at any time outstanding.
- (aa) Liens on the assets of Excluded Subsidiaries securing Indebtedness of Excluded Subsidiaries permitted under clause (w) of the definition of "Permitted Indebtedness".
- (bb) Liens in the nature of royalties, dedications of reserves or similar rights or interests granted, taken subject to or otherwise imposed on properties consistent with normal practices in the iron ore or coal mining industries and any precautionary UCC financing statement filings in respect of leases or consignment arrangements (and not any Indebtedness) entered into in the ordinary course of business,
 - (cc) an interest that is a security interest by virtue only of the operation of section 12(3) of the Australian PPSA,
- (dd) Liens over Australian Featherweight Collateral (other than any registered mining mortgage over any Australian Mine Site upon which any Australian ABL Collateral is located unless a Collateral Access Agreement has been executed by the mortgagee), and
- (ee) other Liens as to which the aggregate amount of the obligations secured thereby does not exceed the greater of (x) \$100,000,000 and (y) 6% of the Parent's "Consolidated Net Tangible Assets" (as defined in the Senior Secured Notes Indenture) at any time outstanding; provided, however, that no more than \$5,000,000 of such Liens permitted under this clause (ee) shall be secured by the ABL Collateral.

"Permitted Protest" means the right of any Borrower or any of its Subsidiaries to protest any Lien (other than any Lien that secures the Obligations), taxes (other than payroll taxes or taxes that are the subject of a United States federal tax lien), or rental payment, provided that (a) a reserve with respect to such obligation is established on such Borrower's or its Subsidiaries' books and records in such amount as is required under GAAP, (b) any such protest is instituted by such Borrower or its Subsidiary, as applicable, in good faith, and (c) with respect to any Lien on ABL Collateral, Agent or Australian Security Trustee, as applicable, is satisfied in its Permitted Discretion that, while any such protest is pending, there will be no impairment of the enforceability, validity, or priority of any of Agent's or Australian Security Trustee's Liens on such ABL Collateral.

"Permitted Purchase Money Indebtedness." means, as of any date of determination, (a) purchase money Indebtedness, including any such Indebtedness assumed in connection with a Permitted Acquisition, (b) Capitalized Lease Obligations, including any such obligations assumed in connection with a Permitted Acquisition, and (c) Indebtedness incurred to finance the acquisition, construction or improvement of any fixed or capital assets, including any indebtedness assumed in connection with the acquisition of any such assets or secured by a Lien on such assets before the acquisition thereof, and any Refinancing Indebtedness thereof, provided that, with respect to the

Indebtedness described in clause (c) of this definition ("Project Indebtedness"), (w) such Project Indebtedness is incurred before or within 180 days after such acquisition or the completion of such construction or improvement, (x) such Project Indebtedness shall be secured only by the property acquired, constructed or improved in connection with the incurrence of such Project Indebtedness, (y) with respect to such Project Indebtedness assumed in connection with a Permitted Acquisition, the amount of such Project Indebtedness shall not exceed 100% of the total consideration paid in connection with such Permitted Acquisition and (z) with respect to Project Indebtedness incurred to finance the acquisition of any fixed or capital assets, such Project Indebtedness shall constitute not more than 100% of the aggregate consideration paid with respect to such fixed or capital assets; provided, further that Parent shall be in pro forma compliance with the financial covenant set forth in Section 7 of this Agreement (regardless of whether a Financial Covenant Period is then in effect) after giving effect to any such Indebtedness described in clauses (a) through (c) of this definition.

"Permitted Reorganization" means (a) the solvent liquidation, winding up, sale, transfer, disposal or reorganization of any Australian Subsidiary (or its assets), or shares in any Australian Subsidiary or (b) the sale, shutdown, winding down or similar transaction by an Australian Subsidiary of any or all of its business and mining operations in any manner (including by way of asset sales, a share sale or liquidation), so long as, in the case of both (a) and (b), all Australian Obligations have been paid in full, all Australian Letters of Credit have been terminated (or Cash Collateralized in accordance with the terms of this Agreement) and all Australian Revolver Commitments have been permanently terminated.

"Person" means natural persons, corporations, limited liability companies, limited partnerships, general partnerships, limited liability partnerships, joint ventures, trusts, land trusts, business trusts, or other organizations, irrespective of whether they are legal entities, and governments and agencies and political subdivisions thereof.

"Plan" means any of (a) an "employee benefit plan" (as defined in ERISA) that is subject to Title I of ERISA, (b) a "plan" as defined in Section 4975 of the Code or (c) any Person whose assets include Plan Assets of any such "employee benefit plan" or "plan".

"Plan Assets" means "plan assets" within the meaning of U.S. Department of Labor Regulation 29 C.F.R. Section 2510.3-101, as modified by Section 3(42) of ERISA.

"Platform" has the meaning specified therefor in Section 18.9(c) of the Agreement.

"Pounds Sterling" shall mean British Pounds Sterling or any successor currency in the United Kingdom.

"Prime Rate" the rate of interest announced by Bank of America from time to time as its prime rate. Such rate is set by Bank of America on the basis of various factors, including its costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above or below such rate. Any change in such rate publicly announced by Bank of America shall take effect at the opening of business on the day specified in the announcement.

"Pro Rata Share" means, as of any date of determination:

- (a) with respect to a Lender's obligation to make all or a portion of the U.S. Revolving Loans, with respect to such Lender's right to receive payments of interest, fees, and principal with respect to the U.S. Revolving Loans, and with respect to all other computations and other matters related to the U.S. Revolver Commitments or the U.S. Revolving Loans, the percentage obtained by dividing (i) the U.S. Revolving Loan Exposure of such Lender by (ii) the aggregate U.S. Revolving Loan Exposure of all Lenders,
- (b) with respect to a Lender's obligation to participate in the U.S. Letters of Credit, with respect to such Lender's obligation to reimburse any U.S. Issuing Bank, and with respect to such Lender's right to receive payments of U.S. Letter of Credit Fees, and with respect to all other computations and other matters related to the U.S. Letters of Credit, the percentage obtained by dividing (i) the U.S. Revolving Loan Exposure of such Lender by (ii) the aggregate U.S. Revolving Loan Exposure of all Lenders; provided, that if all of the U.S. Revolving Loans have been repaid in full and all U.S. Revolver Commitments have been terminated, but U.S. Letters of Credit remain outstanding, Pro Rata Share under this clause shall be determined as if the U.S. Revolver Commitments had not been terminated and based upon the U.S. Revolver Commitments as they existed immediately prior to their termination,

- (c) with respect to a Lender's obligation to make all or a portion of the Australian Revolving Loans, with respect to such Lender's right to receive payments of interest, fees, and principal with respect to the Australian Revolving Loans, and with respect to all other computations and other matters related to the Australian Revolver Commitments or the Australian Revolving Loans, the percentage obtained by dividing (i) the Australian Revolving Loan Exposure of such Lender by (ii) the aggregate Australian Revolving Loan Exposure of all Lenders,
- (d) with respect to a Lender's obligation to participate in the Australian Letters of Credit, with respect to such Lender's obligation to reimburse any Australian Issuing Bank, and with respect to such Lender's right to receive payments of Australian Letter of Credit Fees, and with respect to all other computations and other matters related to the Australian Letters of Credit, the percentage obtained by dividing (i) the Australian Revolving Loan Exposure of such Lender by (ii) the aggregate Australian Revolving Loan Exposure of all Lenders; provided, that if all of the Australian Revolving Loans have been repaid in full and all Australian Revolver Commitments have been terminated, but Australian Letters of Credit remain outstanding, Pro Rata Share under this clause shall be determined as if the Australian Revolver Commitments had not been terminated and based upon the Australian Revolver Commitments as they existed immediately prior to their termination, and
- (e) with respect to all other matters and for all other matters as to a particular Lender (including the indemnification obligations arising under Section 15.6 of the Agreement), the percentage obtained by dividing (i) the Revolving Loan Exposure of such Lender by (ii) the aggregate Revolving Loan Exposure of all Lenders, in any such case as the applicable percentage may be adjusted by assignments permitted pursuant to Section 13.1; provided, that if all of the Loans have been repaid in full, all Letters of Credit have been made the subject of Letter of Credit Collateralization, and all Revolver Commitments have been terminated, Pro Rata Share under this clause shall be determined as if the Revolving Loan Exposures had not been repaid, collateralized, or terminated and shall be based upon the Revolving Loan Exposures as they existed immediately prior to their repayment, collateralization, or termination.

"Projections" means Parent's forecasted (a) balance sheets, (b) profit and loss statements, and (c) cash flow statements, all prepared on a basis consistent with Parent's historical financial statements, together with appropriate supporting details and a statement of underlying assumptions.

"Protective Advances" has the meaning specified therefor in Section 2.3(f)(iv) of the Agreement.

"Public Lender" has the meaning specified therefor in Section 18.9(c) of the Agreement.

"<u>Purchase Price</u>" means, with respect to any Acquisition, an amount equal to the aggregate consideration, whether cash, property or securities (including the fair market value of any Equity Interests of Parent issued in connection with such Acquisition), paid or delivered by a Borrower or one of its Subsidiaries in connection with such Acquisition (whether paid at the closing thereof or payable thereafter and whether fixed or contingent), but excluding therefrom (a) any cash of the seller and its Affiliates used to fund any portion of such consideration and (b) any cash or Cash Equivalents acquired in connection with such Acquisition.

"Qualified Cash" means, as of any date of determination, the amount of unrestricted cash and Cash Equivalents of the Loan Parties that is not identifiable proceeds of Collateral and that is held by Bank of America in the account referred to in the definition of Liquidity.

"Qualified Equity Interest" means and refers to any Equity Interests issued by Parent (and not by one or more of its Subsidiaries) that is not a Disqualified Equity Interest.

"Real Property" means, collectively, all right, title and interest in and to any and all the parcels of or interests in real property owned or leased by a person or as to which a person otherwise has a possessory interest (including by license or easement), together with, in each case, all easements, hereditaments and appurtenances relating thereto, all improvements and appurtenant fixtures.

"Record" means information that is inscribed on a tangible medium or that is stored in an electronic or other medium and is retrievable in perceivable form.

"Refinancing Indebtedness" means refinancings, renewals, or extensions of Indebtedness so long as:

- (a) such refinancings, renewals, or extensions do not result in an increase in the principal amount of the Indebtedness so refinanced, renewed, or extended, other than by the amount of premiums paid thereon and the fees and expenses incurred in connection therewith and by the amount of unfunded commitments with respect thereto.
- (b) such refinancings, renewals, or extensions do not result in a shortening of the average weighted maturity (measured as of the refinancing, renewal, or extension) of the Indebtedness so refinanced, renewed, or extended, nor are they on terms or conditions that, taken as a whole, (i) are more burdensome in any material respect to the Parent or any of its Subsidiaries than the Indebtedness being refinanced as determined by Parent in its reasonable business judgment or (ii) are or could be expected to be materially adverse to the interests of the Lenders; provided that the Existing Senior Notes may be refinanced on substantially the same terms as the terms of the Senior Secured Notes,
- (c) if the Indebtedness that is refinanced, renewed, or extended was subordinated in right of payment to the Obligations, then the terms and conditions of the refinancing, renewal, or extension must include subordination terms and conditions that, when taken as a whole, as determined by Parent in its reasonable business judgment, are at least as favorable to the Lender Group as those that were applicable to the refinanced, renewed, or extended Indebtedness,
- (d) the Indebtedness that is refinanced, renewed, or extended is not recourse to any Person that is liable on account of the Obligations other than those Persons which were obligated with respect to the Indebtedness that was refinanced, renewed, or extended; provided that the Existing Senior Notes may be refinanced on substantially the same terms as the terms of the Senior Secured Notes,
 - (e) [reserved], and
- (f) if such Refinancing Indebtedness in respect of any Existing Senior Notes is secured, such Refinancing Indebtedness shall be subject to an intercreditor agreement in form and substance reasonably acceptable to the Agent, it being agreed the form and substance of the Intercreditor Agreement is acceptable; provided that other than the Existing Senior Notes no unsecured Indebtedness may be refinanced with secured Indebtedness.

"Register" has the meaning set forth in Section 13.1(h) of the Agreement.

"Registered Loan" has the meaning set forth in Section 13.1(h) of the Agreement.

"Related Fund" means any Person (other than a natural person) that is engaged in making, purchasing, holding or investing in bank loans and similar extensions of credit in the ordinary course of and that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"Remedial Action" means all actions required by Environmental Laws or a Governmental Authority taken to (a) clean up, remove, remediate, contain, treat, monitor, assess, evaluate, or in any way address Hazardous Materials in the indoor or outdoor environment, (b) prevent or minimize a release or threatened release of Hazardous Materials so they do not migrate or endanger or threaten to endanger public health or welfare or the indoor or outdoor environment, (c) restore or reclaim natural resources or the environment, (d) perform any pre-remedial studies, investigations, or post-remedial operation and maintenance activities, or (e) conduct any other similar response actions with respect to Hazardous Materials required by Environmental Laws or a Governmental Authority.

"Replacement Lender" has the meaning specified therefor in Section 2.14(b) of the Agreement.

"Report" has the meaning specified therefor in Section 15.9(c) of this Agreement.

"Required Lenders" means, at any time, Lenders having or holding more than 50% of the aggregate Revolving Loan Exposure of all Lenders; provided, that the Revolving Loan Exposure of any Defaulting Lender shall be disregarded in the determination of the Required Lenders.

"Reserves" means, as of any date of determination, those reserves without duplication to U.S. Receivable Reserves, Australian Receivable Reserves, U.S. Bank Product Reserves, Australian Bank Product Reserves, Australian Priority Payables Reserves, U.S. Dilution Reserves, Australian Dilution Reserves, U.S. Inventory/Equipment Reserves, Australian Inventory/Equipment Reserves, U.S. Hedge Reserves and Australian Hedge Reserves that Agent

deems necessary or appropriate, in its Permitted Discretion and subject to the last paragraphs of Section 2.1(a) and Section 2.1(b), to establish and maintain (including reserves with respect to (a) sums that any Borrower or its Subsidiaries are required to pay under any Section of the Agreement or any other Loan Document (such as taxes, assessments, insurance premiums, or, in the case of leased assets, rents or other amounts payable under such leases) and has failed to pay, and (b) amounts owing by any Borrower or its Subsidiaries to any Person to the extent secured by a Lien on, or trust over, any of the ABL Collateral (other than a Permitted Lien), which Lien or trust, in the Permitted Discretion of Agent likely would have a priority superior to the Agent's or Australian Security Trustee's Liens in and to such item of the ABL Collateral) with respect to the U.S. Borrowing Base, the Australian Borrowing Base, the U.S. Maximum Revolver Amount, or the Australian Maximum Revolver Amount.

"Responsible Officer" shall mean any of the President, Chairman, Chief Executive Officer, Chief Operating Officer, Vice Chairman, any Executive Vice President, Chief Financial Officer, General Counsel, Chief Legal Officer, Treasurer or Assistant Treasurer, of Parent.

"Restricted Payment" means to (a) declare or pay any dividend or make any other payment or distribution, directly or indirectly, on account of Equity Interests issued by any Borrower or any of its Subsidiaries or to the direct or indirect holders of Equity Interests issued by such Borrower or such Subsidiary in their capacity as such (other than dividends or distributions payable in Qualified Equity Interests issued by Parent, (b) purchase, redeem, make any sinking fund or similar payment, or otherwise acquire or retire for value (including in connection with any merger or consolidation involving any Borrower or any of its Subsidiaries) any Equity Interests issued by any Borrower or any of its Subsidiaries, and (c) make any payment to retire, or to obtain the surrender of, any outstanding warrants, options, or other rights to acquire Equity Interests of any Borrower or any of its Subsidiaries now or hereafter outstanding.

"Revolver Commitment" means, with respect to each Lender, its U.S. Revolver Commitment or its Australian Revolver Commitment, as the context requires, and, with respect to all Lenders, their U.S. Revolver Commitments or their Australian Revolver Commitments, as the context requires, in each case as such Dollar amounts are set forth beside such Lender's name under the applicable heading on Schedule C-1 to the Agreement or in the Assignment and Acceptance or Incremental Amendment pursuant to which such Lender became a Lender under the Agreement, as such amounts may be reduced or increased from time to time pursuant to Australian Commitment Reallocations pursuant to Section 2.2 or assignments made in accordance with the provisions of Section 13.1 of the Agreement.

"Revolver Usage" means, as of any date of determination, the sum of (a) the amount of outstanding Revolving Loans (inclusive of Swing Loans and Protective Advances), plus (b) the amount of the Letter of Credit Usage.

"Revolving Lender" means a Lender that has a Revolver Commitment or that has an outstanding Revolving Loan.

"Revolving Loan Exposure" means, with respect to any Revolving Lender, as of any date of determination (a) prior to the termination of the Revolver Commitments, the amount of such Lender's Revolver Commitments, and (b) after the termination of the Revolver Commitments, the aggregate outstanding principal amount of the Revolving Loans of such Lender.

"Revolving Loans" has the meaning specified therefor in Section 2.1(b) of the Agreement and includes any Letter of Credit Disbursement, as provided in Section 2.11(d) and Section 2.12(d).

"Sanctioned Entity" means (a) a country or a government of a country, (b) an agency of the government of a country, (c) an organization directly or indirectly owned or controlled by a country or its government, (d) a Person located, organized, or resident in or determined to be resident in a country, in each case, that is subject to a country-based sanctions program administered and enforced by OFAC, Her Majesty's Treasury, the European Union, or the United Nations Security Council.

"Sanctioned Person" means a Person named on the Specially Designated Nationals and Blocked Persons List, the Sectoral Sanctions Identification List, and/or the Foreign Sanctions Evaders List maintained by OFAC, Her Majesty's Treasury, the European Union, or the United Nations Security Council, or any Person owned, directly or indirectly, 50% or more by one or more Persons named on any of the foregoing lists.

"S&P" has the meaning specified therefor in the definition of Domestic Cash Equivalents.

"Scheduled Unavailability Date" has the meaning specified in Section 2.13(d)(iv) of this Agreement.

"SEC" means the United States Securities and Exchange Commission and any successor thereto.

"Securities Account" means a securities account (as that term is defined in the Code).

"Securities Act" means the Securities Act of 1933, as amended from time to time, and any successor statute.

"Senior Secured Notes" means those certain 4.875% Senior Secured Notes due 2024 issued by Parent on December 19, 2017 in the initial aggregate principal amount of \$400,000,000.

"Senior Secured Notes Documents" means the Senior Secured Notes, each Senior Secured Notes Indenture, and all other agreements, documents and instruments entered into now or in the future in connection with the Senior Secured Notes or any Senior Secured Notes Indenture.

"Senior Secured Notes Indenture" means the Indenture, dated as of December 19, 2017, governing the Senior Secured Notes, by and among Parent, as issuer, the guarantors from time to time party thereto, and U.S. Bank National Association, as trustee and collateral agent.

"Settlement" has the meaning specified therefor in Section 2.3(g)(i)(A) of the Agreement.

"Settlement Date" has the meaning specified therefor in Section 2.3(g)(i)(A) of the Agreement.

"Significant Subsidiary" means any Subsidiary that would be a "significant subsidiary" as defined in Article 1, Rule 1-02(w)(1) or (2) of Regulation S-X promulgated under the Securities Act, as such regulation is in effect on the Issue Date.

"Solvent" means, with respect to any Person as of any date of determination, that (a) at fair valuations, the sum of such Person's debts (including contingent liabilities) is less than all of such Person's assets, (b) such Person is not engaged or about to engage in a business or transaction for which the remaining assets of such Person are unreasonably small in relation to the business or transaction or for which the property remaining with such Person is an unreasonably small capital, and (c) such Person has not incurred and does not intend to incur, or reasonably believe that it will incur, debts beyond its ability to pay such debts as they become due (whether at maturity or otherwise), and (d) such Person is "solvent" or not "insolvent", as applicable within the meaning given those terms and similar terms under applicable laws relating to fraudulent transfers and conveyances. For purposes of this definition, the amount of any contingent liability at any time shall be computed as the amount that, in light of all of the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability (irrespective of whether such contingent liabilities meet the criteria for accrual under Statement of Financial Accounting Standard No. 5).

"Specified Event of Default" shall mean any Event of Default arising under Section 8.1, 8.2 (relating to a failure to deliver a Borrowing Base Certificate when required (after the applicable grace period) or a failure to comply with, Section 5.16 or Section 7), 8.4, 8.5, or 8.7 (relating to misrepresentations related to the U.S. Borrowing Base or the Australian Borrowing Base).

"Specified State" means any one of (a) Australia, (b) Canada, (c) The Netherlands, (d) Luxembourg and (e) the United States of America.

"Spot Rate" means the exchange rate, as determined by Agent, that is applicable to conversion of one currency into another currency, which is (a) the exchange rate reported by Bloomberg (or other commercially available source designated by Agent) as of the end of the preceding business day in the financial market for the first currency; or (b) if such report is unavailable for any reason, the spot rate for the purchase of the first currency with the second currency as in effect during the preceding business day in Agent's principal foreign exchange trading office for the first currency.

"Standard Letter of Credit Practice" means, for any Issuing Bank, any domestic or foreign law or letter of credit practices applicable in the city in which such Issuing Bank issued the applicable Letter of Credit or, for its branch or correspondent, such laws and practices applicable in the city in which it has advised, confirmed or negotiated such Letter of Credit, as the case may be, in each case, (a) which letter of credit practices are of banks that regularly issue letters of credit in the particular city, and (b) which laws or letter of credit practices are required or permitted under ISP or UCP, as chosen in the applicable Letter of Credit.

"Subsidiary" of a Person means a corporation, partnership, limited liability company, or other entity in which that Person directly or indirectly owns or controls the Equity Interests having ordinary voting power to elect a majority of the Board of Directors of such corporation, partnership, limited liability company, or other entity. Notwithstanding the foregoing, as used herein and in the Loan Documents, Subsidiary shall not include the Canadian Entities and none of the Canadian Entities shall be deemed to be a Subsidiary of Parent; provided that, notwithstanding the foregoing, (i) the Canadian Entities shall constitute Subsidiaries for purposes of Section 4.9, Section 4.13, Section 5.8 (as it relates to compliance with laws referred to in Section 5.8 (as it relates to compliance with laws referred to in Section 4.11).

"Supermajority Lenders" means, at any time, Lenders having or holding more than 66 2/3% of the aggregate Revolving Loan Exposure of all Lenders; provided, that (a) the Revolving Loan Exposure of any Defaulting Lender shall be disregarded in the determination of the Supermajority Lenders, and (b) at any time there are two (2) or more Lenders with Revolving Loan Exposure, "Supermajority Lenders" must include at least two (2) Lenders (who are not Affiliates of one another).

"Swap Obligation" means, with respect to any Loan Party, any obligation to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act.

"Swing Lender" means any U.S. Swing Lender or any Australian Swing Lender.

"Swing Loan" has the meaning specified therefor in Section 2.3(d) of the Agreement.

"Swing Loan Exposure" means, as of any date of determination with respect to any Lender, such Lender's Pro Rata Share of the Swing Loans on such date.

"<u>Taxes</u>" means any taxes, levies, imposts, duties, fees, assessments or other charges of whatever nature now or hereafter imposed by any jurisdiction or by any political subdivision or taxing authority thereof or therein, and all interest, penalties or similar liabilities with respect thereto.

"Tax Lender" has the meaning specified therefor in Section 14.2(a) of the Agreement.

"Trademark Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"<u>Transactions</u>" means (a) the execution and delivery of the Agreement, (b) the occurrence of the Closing Date and (c) the payment of fees, costs and expenses incurred in connection with the foregoing transactions.

"<u>UCP</u>" means, with respect to any Letter of Credit, the Uniform Customs and Practice for Documentary Credits 2007 Revision, International Chamber of Commerce Publication No. 600 and any subsequent revision thereof adopted by the International Chamber of Commerce on the date such Letter of Credit is issued.

"<u>Unfunded Pension Liability</u>" means, with respect to any Pension Plan at any time, the amount of any of its unfunded benefit liabilities as defined in Section 4001(a)(18) of ERISA.

"United States" and "U.S." mean the United States of America.

"Unused Line Fee" means an Australian Unused Line Fee or a U.S. Unused Line Fee.

"U.S. Additional Documents" has the meaning specified therefor in Section 5.12(a) of the Agreement.

"U.S. Bank Product" means any one or more of the following financial products or accommodations extended to a U.S. Borrower or its Subsidiaries by a U.S. Bank Product Provider: (a) credit cards (including commercial cards (including so-called "purchase cards", "procurement cards" or "p-cards")), (b) credit card processing services, (c) debit cards, (d) stored value cards, (e) Cash Management Services, (f) supply chain financing, or (g) transactions under Hedge Agreements.

"<u>U.S. Bank Product Agreements</u>" means those agreements entered into from time to time by a U.S. Borrower or its Subsidiaries with a U.S. Bank Product Provider in connection with the obtaining of any of the U.S. Bank Products.

- "U.S. Bank Product Obligations" means (a) all obligations, liabilities, reimbursement obligations, fees, or expenses owing by U.S. Borrowers and their Subsidiaries to any U.S. Bank Product Provider pursuant to or evidenced by an U.S. Bank Product Agreement and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, (b) all U.S. Hedge Obligations, and (c) all amounts that Agent or any U.S. Revolving Lender is obligated to pay to a U.S. Bank Product Provider as a result of Agent or such U.S. Revolving Lender purchasing participations from, or executing guarantees or indemnities or reimbursement obligations to, a U.S. Bank Product Provider with respect to the U.S. Bank Products provided by such Bank Product Provider to U.S. Borrowers or their Subsidiaries; provided that in order for any item described in clauses (a), (b), or (c) above, as applicable, to constitute "U.S. Bank Product Obligations", if the applicable U.S. Bank Product Provider is any Person other than Bank of America or its Affiliates or any Person holding Existing Hedge Obligations, then the applicable U.S. Bank Product must have been provided on or after the Closing Date and Agent shall have received a Bank Product Provider Agreement within ten (10) days after the date of the provision of the applicable U.S. Bank Product to U.S. Borrowers or their Subsidiaries; provided further that the U.S. Bank Product Obligations shall not include any Excluded Swap Obligations.
- "U.S. Bank Product Provider" means any U.S. Revolving Lender or any of its Affiliates, including each of the foregoing in its capacity, if applicable, as a U.S. Hedge Provider; provided, that no such Person (other than Bank of America or its Affiliates) shall constitute a U.S. Bank Product Provider with respect to a U.S. Bank Product untiless and until Agent receives a Bank Product Provider Agreement from such Person and with respect to the applicable U.S. Bank Product within ten (10) days after the provision of such U.S. Bank Product to U.S. Borrowers or their Subsidiaries; provided further, that if, at any time, a U.S. Revolving Lender ceases to be a Lender under the Agreement, then, from and after the date on which it ceases to be a Lender thereunder, neither it nor any of its Affiliates shall constitute U.S. Bank Product Providers and the obligations with respect to U.S. Bank Products provided by such former U.S. Revolving Lender or any of its Affiliates shall no longer constitute U.S. Bank Product Obligations.
- "<u>U.S. Bank Product Reserves</u>" means, as of any date of determination, those reserves that Agent deems necessary or appropriate to establish (based upon the U.S. Bank Product Providers' determination of the liabilities and obligations of U.S. Borrowers and their Subsidiaries in respect of U.S. Bank Product Obligations) in respect of U.S. Bank Products then provided or outstanding.
 - "U.S. Blocked Account" means a deposit account located in the U.S. that is subject to a U.S. Control Agreement.
- "U.S. Borrowers" means Cleveland-Cliffs Inc., an Ohio corporation, Lake Superior & Ishpeming Railroad Company, a Michigan corporation, The Cleveland-Cliffs Iron Company, an Ohio corporation, Cliffs Mining Company, a Delaware corporation, Northshore Mining Company, a Delaware corporation, United Taconite LLC, a Delaware limited liability company, Cliffs North American Coal LLC, a Delaware limited liability company, Oak Grove Resources, LLC, a Delaware limited liability company, Pinnacle Mining Company, LLC, a Delaware limited liability company, all domestic Subsidiaries of Parent whose assets comprise a portion of the U.S. Borrowing Base and shall include, as applicable, any Subsidiary of Parent organized under the laws of the United States, any State thereof, or the District of Columbia that becomes a Borrower after the Closing Date pursuant to Section 5.11 of the Agreement.
 - "U.S. Borrowing Base" means, as of any date of determination, the sum of:
 - (a) 85% of the amount of Eligible Accounts of U.S. Borrowers, plus
- (b) the lesser of (i) the product of 80% multiplied by the value (calculated at the lower of cost or market on a basis consistent with Borrowers' historical accounting practices) of Eligible Inventory of the U.S. Borrowers at such time, and (ii) the product of 85% multiplied by the Net Orderly Liquidation Value of Eligible Inventory of U.S. Borrowers at such time, *plus*
- (c) the lesser of (i) 100% of the Net Book Value of Eligible Equipment of the U.S. Borrowers at such time, and (ii) 85% multiplied by the Net Orderly Liquidation Value of Eligible Equipment of U.S. Borrowers at such time; provided that this clause (c) of the U.S. Borrowing Base shall not account for more than 35% of the availability created by the U.S. Borrowing Base, minus
 - (d) the Allocated U.S. Availability (if any) at the time of determination; minus
 - (e) the aggregate amount of reserves, if any, established by Agent under Section 2.1(a) of the Agreement.

- "U.S. Control Agreement" means, with respect to any applicable deposit or securities account established by a U.S. Loan Party, an agreement, in form and substance reasonably satisfactory to the Agent, establishing Control (as defined in the Uniform Commercial Code) of such an account by the Agent and whereby the Person maintaining such account agrees to comply only with the instructions originated by the Agent without the further consent of any U.S. Loan Party.
- "<u>U.S. Designated Account</u>" means the Deposit Account of Administrative Borrower identified on <u>Schedule D-1</u> to the Agreement (or such other Deposit Account of Administrative Borrower located at Designated Account Bank that has been designated as such, in writing, by Borrowers to Agent).
- "<u>U.S. Designated Account Bank</u>" has the meaning specified therefor in <u>Schedule D-1</u> to the Agreement (or such other bank that is located within the United States that has been designated as such, in writing, by Borrowers to Agent).
- "U.S. Dilution Percent" means the percent, determined as of the end of the U.S. Loan Parties' most recent field examination, equal to (a) bad debt write-downs or write-offs, discounts, returns, promotions, credits, credit memos and other dilutive items with respect to active Accounts of the U.S. Loan Parties, divided by (b) active gross sales.
- "<u>U.S. Dilution Percentage</u>" means at any time, one percentage point (or fraction thereof) for each percentage point (or fraction thereof) by which the U.S. Dilution Percent for the U.S. Loan Parties exceeds five percent (5.0%).
- "<u>U.S. Dilution Reserve</u>" means a reserve equal to the product of (x) the U.S. Dilution Percentage times (y) the value of all Eligible Accounts of the U.S. Loan Parties at such time.
 - "U.S. Dominion Account" a collection account at Bank of America in the United States over which Agent has exclusive control.
 - "<u>U.S. Excess Availability</u>" means, as of any date of determination, (i) the U.S. Line Cap, minus (ii) the outstanding U.S. Revolver Usage.
 - "U.S. Extraordinary Advances" has the meaning specified therefor in Section 2.3(f)(iii) of the Agreement.
- "<u>U.S. Hedge Obligations</u>" means any and all obligations or liabilities, whether absolute or contingent, due or to become due, now existing or hereafter arising, of U.S. Borrowers and their Subsidiaries arising under, owing pursuant to, or existing in respect of Hedge Agreements entered into with one or more of the U.S. Hedge Providers, including without limitation, the Existing Hedge Obligations; <u>provided</u>, however, that the U.S. Hedge Obligations shall not include any Excluded Swap Obligations.
- "U.S. Hedge Provider" means any U.S. Revolving Lender or any of its Affiliates; provided, that no such Person (other than Bank of America or its Affiliates) shall constitute a U.S. Hedge Provider unless and until Agent receives a Bank Product Provider Agreement from such Person and with respect to the applicable Hedge Agreement within 10 days after the execution and delivery of such Hedge Agreement with a U.S. Borrower or its Subsidiaries; provided further, that if, at any time, a U.S. Revolving Lender ceases to be a U.S. Revolving Lender under the Agreement, then, from and after the date on which it ceases to be a U.S. Revolving Lender thereunder, neither it nor any of its Affiliates shall constitute U.S. Hedge Providers and the obligations with respect to Hedge Agreements entered into with such former U.S. Revolving Lender or any of its Affiliates shall no longer constitute U.S. Hedge Obligations.
- "<u>U.S. Hedge Reserves</u>" means, as of any date of determination, those reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to the last paragraph of <u>Section 2.1(a)</u>, to establish and maintain with respect to the U.S. Hedge Obligations of U.S. Borrowers.
- "<u>U.S. Inventory/Equipment Reserves</u>" means, as of any date of determination, (a) Landlord Reserves for locations of any U.S. Borrower, and (b) those reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to the last paragraph of <u>Section 2.1(a)</u>, to establish and maintain (including reserves for slow moving Inventory and Inventory shrinkage) with respect to Eligible Inventory or Eligible Equipment of U.S. Borrowers or the U.S. Maximum Revolver Amount.

- "<u>U.S. Iron Ore Business</u>" means the iron ore mining, processing, distribution and sales operations, and related assets, of the Parent and its Subsidiaries conducted, or located in, the United States, excluding Cliffs Erie L.L.C., a Delaware corporation and its assets.
- "U.S. Issuing Bank" means Bank of America (or any of its Affiliates), PNC Bank, National Association, Citizens Bank, N.A., Deutsche Bank AG New York Branch, Regions Bank, Credit Suisse AG, Cayman Islands Branch, Goldman Sachs Bank USA, The Huntington National Bank and each other U.S. Revolving Lender appointed by Parent and agreed by the Agent and such U.S. Revolving Lender, in their sole discretion, to become a U.S. Issuing Bank for the purpose of issuing U.S. Letters of Credit pursuant to Section 2.11 of the Agreement, and U.S. Issuing Bank shall be deemed to be a Lender and a U.S. Revolving Lender.
- "<u>U.S. Letter of Credit</u>" means a letter of credit (as that term is defined in the Code) or any standby, documentary, bankers acceptance, foreign guarantee, or indemnity issued by a U.S. Issuing Bank for the account of a U.S. Borrower pursuant to the Agreement.
- "<u>U.S. Letter of Credit Exposure</u>" means, as of any date of determination with respect to any U.S. Revolving Lender, such U.S. Revolving Lender's Pro Rata Share of the U.S. Letter of Credit Usage on such date.
 - "U.S. Letter of Credit Fee" has the meaning specified therefor in Section 2.6(b)(i) of the Agreement.
 - "U.S. Letter of Credit Indemnified Costs" has the meaning specified therefor in Section 2.11(f) of the Agreement.
 - "U.S. Letter of Credit Related Person" has the meaning specified therefor in Section 2.11(f) of the Agreement.
- "U.S. Letter of Credit Sublimit" means an amount equal to \$248,777,777.78, which amount (i) shall be allocated in the following amounts: \$46,000,000.00 to Bank of America (or any of its Affiliates), \$44,444,444.44 to PNC Bank, National Association, \$27,777,777.78 to Citizens Bank, N.A, \$27,777,777.78 to Deutsche Bank AG New York Branch, \$27,777,777.78 to Regions Bank, \$27,777,777.78 to Credit Suisse AG, Cayman Islands Branch, \$30,555,555.55 to Goldman Sachs Bank USA and \$16,666,666.67 to The Huntington National Bank and (ii) may be increased to an amount not to exceed \$250,000,000.00 upon the agreement of one or more U.S. Issuing Banks to increase their allocation of the U.S. Letter of Credit Sublimit.
 - "U.S. Letter of Credit Usage" means, as of any date of determination, the aggregate undrawn amount of all outstanding U.S. Letters of Credit.
- "<u>U.S. Line Cap</u>" means, as of any date of determination, the lesser of (a) the U.S. Maximum Revolver Amount and (b) the U.S. Borrowing Base as of such date.
 - "U.S. Loan Account" has the meaning specified therefor in Section 2.9(a) of the Agreement.
 - "U.S. Loan Party" means any U.S. Borrower or any guarantor of the U.S. Obligations.
- "<u>U.S. Maximum Revolver Amount</u>" means \$400,000,000, decreased by the amount of reductions in the U.S. Revolver Commitments made in accordance with <u>Section 2.4(c)(i)</u> of the Agreement, and as the same may be increased pursuant to <u>Section 2.16</u> or increased pursuant to <u>Section 2.2</u>.
- "U.S. Obligations" means (a) all loans (including the U.S. Revolving Loans (inclusive of U.S. Extraordinary Advances and U.S. Swing Loans)), debts, principal, interest (including any interest that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), reimbursement or indemnification obligations with respect to U.S. Letters of Credit (irrespective of whether contingent), premiums, liabilities (including all amounts charged to the U.S. Loan Account pursuant to the Agreement), obligations (including indemnification obligations), fees, Lender Group Expenses (including any fees or expenses that accrue after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), guaranties, and all covenants and duties of any other kind and description owing by any U.S. Loan Party arising out of, under, pursuant to, in connection with, or evidenced by the Agreement or any of the other Loan Documents and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, and including all interest not paid when due and all other expenses or other amounts that the U.S. Loan Parties are required to pay or reimburse by the Loan Documents or by law or otherwise in connection with the Loan Documents,

- and (b) all U.S. Bank Product Obligations; provided, however, that the U.S. Obligations shall not include any Excluded Swap Obligations. Without limiting the generality of the foregoing, the U.S. Obligations of the U.S. Loan Parties under the Loan Documents include the obligation to pay (i) the principal of the U.S. Revolving Loans, (ii) interest accrued on the U.S. Revolving Loans, (iii) the amount necessary to reimburse any U.S. Issuing Bank for amounts paid or payable pursuant to U.S. Letters of Credit, (iv) U.S. Letter of Credit commissions, fees (including fronting fees) and charges, (v) Lender Group Expenses, (vi) fees payable under the Agreement or any of the other Loan Documents, and (vii) indemnities and other amounts payable by any U.S. Loan Party under any Loan Document. Any reference in the Agreement or in the Loan Documents to the U.S. Obligations shall include all or any portion thereof and any extensions, modifications, renewals, or alterations thereof, both prior and subsequent to any Insolvency Proceeding.
- "<u>U.S. Overadvance</u>" means, as of any date of determination, that the U.S. Revolver Usage is greater than any of the limitations set forth in <u>Section 2.1(a)</u> or <u>Section 2.11</u>, in each case subject to <u>Section 1.7(d)</u>.
- "<u>U.S. Overadvance Loan</u>" shall mean a U.S. Revolving Loan that is a Base Rate Loan or an Australian Base Rate Loan made when a U.S. Overadvance exists or is caused by the funding thereof.
 - "U.S. Protective Advances" has the meaning specified therefor in Section 2.3(f)(i) of the Agreement.
- "<u>U.S. Receivable Reserves</u>" means, as of any date of determination, those reserves that Agent deems necessary or appropriate, in its Permitted Discretion and subject to the last paragraph of <u>Section 2.1(a)</u>, to establish and maintain (including reserves for rebates, discounts, warranty claims, and returns) with respect to the Eligible Accounts of U.S. Borrowers or the U.S. Maximum Revolver Amount.
- "U.S. Revolver Commitment" means, with respect to each U.S. Revolving Lender, its U.S. Revolver Commitment, and, with respect to all U.S. Revolving Lenders, their U.S. Revolver Commitments, in each case as such Dollar amounts are set forth beside such U.S. Revolving Lender's name under the applicable heading on Schedule C-1 to the Agreement or in the Assignment and Acceptance pursuant to which such U.S. Revolving Lender became a U.S. Revolving Lender under the Agreement, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of Section 13.1 of the Agreement.
- "<u>U.S. Revolver Usage</u>" means, as of any date of determination, the sum of (a) the amount of outstanding U.S. Revolving Loans (inclusive of U.S. Swing Loans and U.S. Protective Advances), plus (b) the amount of the U.S. Letter of Credit Usage.
 - "U.S. Revolving Lender" means a Lender that has a U.S. Revolver Commitment or that has an outstanding U.S. Revolving Loan.
- "<u>U.S. Revolving Loan Exposure</u>" means, with respect to any U.S. Revolving Lender, as of any date of determination (a) prior to the termination of the U.S. Revolver Commitments, the amount of such U.S. Revolving Lender's U.S. Revolver Commitments, and (b) after the termination of the U.S. Revolver Commitments, the aggregate outstanding principal amount of the U.S. Revolving Loans of such U.S. Revolving Lender.
- "U.S. Revolving Loans" has the meaning specified therefor in <u>Section 2.1(a)</u> of the Agreement and includes U.S. Letter of Credit Disbursements pursuant to <u>Section 2.11(d)</u>.
- "<u>U.S. Swing Lender</u>" means Bank of America or any other U.S. Revolving Lender that, at the request of Borrowers and with the consent of Agent agrees, in such U.S. Revolving Lender's sole discretion, to become the U.S. Swing Lender under <u>Section 2.3(c)</u> of the Agreement.
 - "U.S. Swing Loan" has the meaning specified therefor in Section 2.3(c) of the Agreement.
- "<u>U.S. Swing Loan Exposure</u>" means, as of any date of determination with respect to any U.S. Revolving Lender, such U.S. Revolving Lender's Pro Rata Share of the U.S. Swing Loans on such date.
 - "U.S. Unused Line Fee" has the meaning specified therefor in Section 2.10(b)(i) of the Agreement.
 - "Voidable Transfer" has the meaning specified therefor in Section 18.8 of the Agreement.

"Withdrawal Liabili	ty" means liability with respe	ct to a Multiemployer	Plan as a result of	of a complete or par	tial withdrawal from such	Multiemployer Plan,
as such terms are defined	in Part I of Subtitle E of Title	IV of ERISA.				

AGENT'S ACCOUNT

BANK OF AMERICA, N.A. NEW YORK, NY ABA #0260-0959-3 ACCOUNT NAME: BofA Waukesha Collections ACCOUNT #936 933 7536 REF: [Insert Client or Company Name]

For International Incoming Wires: Swift Code: BOFAUS3NXXX CHIPS Routing Number: 959 Bank of America, N.A. United States

AUTHORIZED PERSONS

James Graham Celso L. Goncalves Denise Caruso Timothy Flanagan

COMPETITORS

USIO ArcelorMittal Rio Tinto- IOC US Steel

APIO Fortescue Metals Group BHP Rio Tinto Vale Anglo American

Revolver Commitments

Name of Lender	Total Commitment	U.S. Revolver Commitment	Australian Revolver Commitment
Bank of America, N.A.	\$80,000,000.00	\$71,111,111.11	\$8,888,888.89
PNC Bank, National Association	\$80,000,000.00	\$71,111,111.11	\$8,888,888.89
Citizens Bank, N.A.	\$50,000,000.00	\$44,444,444.44	\$5,555,555.56
Deutsche Bank AG New York Branch	\$50,000,000.00	\$44,444,444.44	\$5,555,555.56
Regions Bank	\$50,000,000.00	\$44,444,444.44	\$5,555,555.56
Credit Suisse AG	\$50,000,000.00	\$44,444,444.44	\$5,555,555.56
Goldman Sachs Bank USA	\$44,444,444.44	\$44,444,444.44	\$0.00
Goldman Sachs Lending Partners LLC	\$5,555,555.56	\$0.00	\$5,555,555.56
The Huntington National Bank	\$30,000,000.00	\$26,666,666.68	\$3,333,333.32
Jefferies Finance LLC	\$1,111,111.10	\$0.00	\$1,111,111.10
JFIN Business Credit Fund I LLC	\$8,888,888.90	\$8,888,888.90	\$0.00
TOTAL	\$450,000,000.00	\$400,000,000.00	\$50,000,000.00

U.S. DESIGNATED ACCOUNT

PNC Bank, N.A.
Firstside Center, Mail Stop P7-PFSC-03-W
500 First Avenue, Pittsburgh, PA 15219
ABA: 041000124
SWIFT:
Bnf A/C:
Bnf: Cleveland-Cliffs Inc.
200 Public Square, Suite 3300
Cleveland, Ohio 44114

AUSTRALIAN DESIGNATED ACCOUNT

Bank of America Merrill Lynch Sydney, Australia SWIFT: Bnf: Cliffs Natural Resources Pty Ltd. Level 12, The Quadrant Building 1 William Street, Perth, WA 6000 Acct No.:

SCHEDULE E-1

EXISTING HEDGE OBLIGATIONS

Trade <u>Date</u>	SWAP Month	Index	Ticker	Quantity	Units	Settlement Date
12/4/2017	Mar-18	Nat Gas - NYMEX	NG1	615,000	MMBtu	03/05/18
12/4/2017	Apr-18	Nat Gas - NYMEX	NG1	363,000	MMBtu	04/04/18
12/4/2017	May-18	Nat Gas - NYMEX	NG1	350,000	MMBtu	05/03/18
12/4/2017	Jun-18	Nat Gas - NYMEX	NG1	331,000	MMBtu	06/05/18
12/4/2017	Jul-18	Nat Gas - NYMEX	NG1	245,000	MMBtu	07/05/18
12/4/2017	Aug-18	Nat Gas - NYMEX	NG1	242,000	MMBtu	08/03/18
12/4/2017	Sep-18	Nat Gas - NYMEX	NG1	186,000	MMBtu	09/06/18
12/4/2017	Oct-18	Nat Gas - NYMEX	NG1	158,000	MMBtu	10/03/18
12/4/2017	Nov-18	Nat Gas - NYMEX	NG1	147,000	MMBtu	11/05/18
12/4/2017	Dec-18	Nat Gas - NYMEX	NG1	155,000	MMBtu	12/05/18

EXISTING LETTERS OF CREDIT

Account Party	L/C Issuer	Expiry Date	Letter of Credit Amount (USD Value)	<u>Beneficiary</u>
CNR Pty Ltd.	Bank of America, N.A.	30-Mar-20	\$1,323,481.92	COMMONWEALTH BANK OFFICERS SUPERANNUATIO
CNR Pty Ltd.	Bank of America, N.A.	30-Mar-20	\$60,427.50	MINISTER FOR TRANSPORT
Cleveland-Cliffs Inc.	Bank of America, N.A.	15-May-18	\$6,326,260.00	ACE AMERICAN INSURANCE COMPANY
Empire Iron Mining Partnership	Bank of America, N.A.	30-Jun-18	\$750,000.00	BUREAU OF WORKERS DISABILITY
Cleveland-Cliffs Iron Company	Bank of America, N.A.	30-Jun-18	\$400,000.00	DEPT OF ENERGY LABOR AND ECONOMIC
Northshore Mining Company	Bank of America, N.A.	24-Oct-18	\$4,000,000.00	MINNESOTA DEPARTMENT OF NATURAL
Cleveland-Cliffs Inc.	Bank of America, N.A.	24-Oct-18	\$8,563,521.00	NATIONAL UNION FIRE INSURANCE CO.
Cleveland-Cliffs Inc.	Bank of America, N.A.	5-Nov-18	\$2,716,227.00	NORTHERN NATURAL GAS COMPANY
Tilden Mining Company L.C.	Bank of America, N.A.	9-Nov-18	\$750,000.00	DEPARTMENT OF LICENSING
Cleveland-Cliffs Inc.	Bank of America, N.A.	18-Nov-18	\$1,343,818.00	ROCKWOOD CASUALTY INSURANCE CO.
Cleveland-Cliffs Inc.	Bank of America, N.A.	19-Nov-18	\$6,658,929.00	NATIONAL UNION FIRE INSURANCE CO.
Cleveland-Cliffs Inc.	Bank of America, N.A.	24-Nov-18	\$13,662,495.00	ARGONAUT INSURANCE CO.

EXCLUDED SUBSIDIARY INDEBTEDNESS

None.

IMMATERIAL SUBSIDIARIES

Cliffs Renewable Energies LLC
Cliffs Reduced Iron Corporation
IronUnits LLC
Cliffs Technical Products LLC
Syracuse Mining Company
Seignelay Resources, Inc.
Cliffs Erie L.L.C.
Cleveland-Cliffs Ore Corporation
Cliffs Marquette, Inc.
The Cleveland-Cliffs Steamship Company
Marquette Iron Mining Partnership
Cliffs Oil Shale Corp.
Northern Conservation, LLC
Republic Wetlands Preserve LLC
Cliffs West Virginia Coal Inc.
Cleveland-Cliffs Minnesota Land Development LLC

PERMITTED INVESTMENTS

- Intercompany Note (Note 78), between Cliffs Natural Resources Luxembourg S.à r.l., as lender, and Cliffs Natural Resources Holdings Pty Ltd., as borrower, in the principal amount of AUD \$283,073,400.00
- Intercompany Note (Note 61), between Cliffs Natural Resources Pty Ltd, as lender, and Cliffs Quebec Iron Mining Ltd., as borrower, in a principal amount of USD \$248,924,803.57
- Intercompany Note (Note 76), between Luxembourg S.à r.l., as lender, and Cleveland-Cliffs Inc. (f/k/a/ Cliffs Natural Resources Inc.), as borrower, in a principal amount of USD \$242,000,000.00

PERMITTED LIENS

None.

PERMITTED DISPOSITIONS

Sale of all of the membership interests of Cliffs Erie L.L.C., a wholly owned subsidiary of Cliffs Mining Company, or all or a portion of the assets of Cliffs Erie L.L.C.

PERMITTED INDEBTEDNESS

- Intercompany Note (Note 78), between Cliffs Natural Resources Luxembourg S.à r.l., as lender, and Cliffs Natural Resources Holdings Pty Ltd., as borrower, in the principal amount of AUD \$283,073,400.00
- Intercompany Note (Note 61), between Cliffs Natural Resources Pty Ltd, as lender, and Cliffs Quebec Iron Mining Ltd., as borrower, in a principal amount of USD \$248,924,803.57
- Intercompany Note (Note 76), between Luxembourg S.à r.l., as lender, and Cleveland-Cliffs Inc. (f/k/a Cliffs Natural Resources Inc.), as borrower, in a principal amount of USD \$242,000,000.00
- The Existing Letters of Credit set forth on Schedule E-2 hereto

SCHEDULE 3.1

The effectiveness of the Agreement and the obligation of each Lender to make its initial extension of credit provided for in the Agreement are subject to the fulfillment, to the reasonable satisfaction of each Lender (the delivery of a signature page to the Agreement by a Lender being conclusively deemed to be its satisfaction or waiver of the following), of each of the following conditions precedent:

- (a) Agent shall have received executed counterparts of the Agreement signed by each Borrower and each Lender:
- (b) Subject to Section 3.7, Agent shall have received evidence that appropriate Uniform Commercial Code financing statements, Australian PPSA financing statements and as-extracted collateral filings have been (or, on the Closing Date, will be) duly filed in such office or offices as may be necessary or, in the opinion of Agent, desirable to perfect the Agent's or the Australian Security Trustee's Liens in and to the Collateral, and, in the case of Australian PPSA financing statements, Agent shall have received searches reflecting the filing of all such financing statements;
- (c) Agent shall have received each of the following documents (including any restatements or reaffirmations thereof), in form and substance reasonably satisfactory to Agent, duly executed and delivered, and each such document shall be in full force and effect:
 - (i) the Fee Letter,
 - (ii) the Guaranty and Security Agreement, and
 - (iii) a completed perfection certificate for each of the Loan Parties.
- (d) Agent shall have received a certificate from the Secretary or a director of each Loan Party (i) attesting to the resolutions of such Loan Party's board of directors authorizing its execution, delivery, and performance of the Loan Documents to which it is a party, (ii) authorizing specific officers of such Loan Party to execute the same, and (iii) attesting to the incumbency and signatures of such specific officers of such Loan Party:
- (e) Agent shall have received copies of each Loan Party's Governing Documents, as amended, modified, or supplemented to the Closing Date, which Governing Documents shall be (i) certified by the Secretary or a director of such Loan Party, and (ii) with respect to Governing Documents that are charter documents, certified as of a recent date by the appropriate governmental official;
- (f) Agent shall have received a certificate of status with respect to each Loan Party (where applicable, or such other customary functionally equivalent certificates, to the extent available in the applicable jurisdiction), such certificate to be issued by the appropriate officer of the jurisdiction of organization of such Loan Party, which certificate shall indicate that such Loan Party is in good standing in such jurisdiction;
- (g) Agent shall have received a certificate of insurance, together with the endorsements thereto, as are required by Section 5.6 of the Agreement, which Agent acknowledges it has received;
- (h) Agent shall have received (i) from Jones Day, special New York, Delaware, Ohio, Michigan and Minnesota counsel to the Loan Parties, an opinion addressed to the Agent and each of the Lenders and dated the Closing Date in form and substance reasonably satisfactory to the Agent, and (ii) from Norton Rose Fulbright Australia, special Australian counsel to the Agent, an opinion addressed to the Agent and each of the Lenders and dated the Closing Date in form and substance reasonably satisfactory to the Agent;
 - (i) [reserved];
 - (j) [reserved];
- (I) Agent shall have received a set of Projections of Borrower for the five year period following the Closing Date on a year by year basis in form and substance (including as to scope and underlying assumptions) reasonably satisfactory to Agent;

- (m) Borrowers shall have paid all Lender Group Expenses incurred in connection with the transactions evidenced by the Agreement and the other Loan Documents for which invoices have been presented at least two Business Days prior to the Closing Date;
 - (n) [reserved];
- (o) each Borrower and each of its Subsidiaries shall have received all governmental and third party approvals (including shareholder approvals) necessary or, in the reasonable opinion of Agent, advisable in connection with the Agreement or the Transactions, which shall all be in full force and effect;
- (p) Agent (on behalf of the Lenders) shall have received not later than three Business Days prior to the Closing Date (or such later date as shall be acceptable to it), all documentation and other information about the Loan Parties as had been reasonably requested at least five Business Days prior to the Closing Date by Agent (on behalf of the Lenders) that it reasonably determines is required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act.
- (q) the Borrowers shall have paid (i) all principal, interest, fees and other amounts due and payable to the Departing Lenders (if any) and (ii) all accrued interest and fees under the Existing Syndicated Facility Agreement to any lender party to the Existing Syndicated Facility Agreement that is not a Departing Lender; and
- (r) all other documents and legal matters in connection with the transactions contemplated by the Agreement shall have been delivered, executed, or recorded and shall be in form and substance satisfactory to Agent.

SCHEDULE 3.7

CONDITIONS SUBSEQUENT

- 1. By June 17, 2018 (or such later date as the Agent shall determine in its sole discretion), the Company shall cause as-extracted collateral filings to be made against Empire Iron Mining Partnership in Marquette County, Michigan, and shall provide, simultaneously with the making of such filing, an opinion of counsel addressed to Agent and each of the Lenders in form and substance reasonably satisfactory to Agent (it being agreed that the form of opinion previously delivered is acceptable) which, in its Permitted Discretion, is appropriate with respect to the making of such filing.
- 2. By March 14, 2018 (or such later date as the Agent shall determine in its sole discretion), the Agent shall have received evidence that appropriate Uniform Commercial Code as-extracted collateral filings have been duly filed in such office or offices as may be necessary or, in the opinion of Agent, desirable to perfect the Agent's or the Australian Security Trustee's Liens in and to the Collateral, and the Agent shall have received searches reflecting that the as-extracted collateral filings that were made in connection with the execution of the existing Syndicated Facility Agreement have not been released or terminated and that no intervening liens have been filed, other than Permitted Liens.

SCHEDULE 4.1(b)

CAPITALIZATION OF BORROWERS

Issuer	Jurisdiction of Organization	Owner of Equity Interest	Percentage Owned	Number of Shares or Units / Percentage of Membership Interest
Cliffs Mining Company	Delaware	Cleveland-Cliffs Inc.	100%	27,000
Lake Superior & Ishpeming Railroad Company	Michigan	Cleveland-Cliffs Inc.	100%	107,196
Northshore Mining Company	Delaware	Cleveland-Cliffs Inc.	100%	100
The Cleveland-Cliffs Iron Company	Ohio	Cleveland-Cliffs Inc.	100%	1,000
United Taconite LLC	Delaware	Cliffs Minnesota Mining Company	70%	N/A
United Taconite LLC	Delaware	Cliffs UTAC Holding LLC	30%	N/A
Cliffs Natural Resources Pty Ltd	Australia	Cliffs Natural Resources Holdings Pty Ltd	100%	104,975,948
Cliffs Asia Pacific Iron Ore Pty Ltd	Australia	Cliffs Asia Pacific Iron Ore Holdings Pty Ltd	100%	N/A

SCHEDULE 4.1(c)

CAPITALIZATION OF BORROWERS' SUBSIDIARIES

Issuer	Jurisdiction of Organization	Owner of Equity Interest	Percentage Owned	Number of Shares or Units / Percentage of Membership Interest
CLF PinnOak LLC	Delaware	Cliffs Mining Company	100%	N/A
Cliffs Pickands Holding, LLC	Delaware	Cliffs Mining Company	100%	100
Cliffs Mining Holding, LLC	Delaware	Cliffs Mining Company	100%	100
Cliffs Technical Products LLC	Delaware	Cliffs Mining Company	100%	N/A
Syracuse Mining Company	Minnesota	Cliffs Mining Company	100%	500
Seignelay Resources, Inc.	Delaware	Cliffs Mining Company	100%	200
Wabush Iron Co. Limited	Ohio	Cliffs Mining Company	100%	1,540
Cliffs Erie L.L.C.	Delaware	Cliffs Mining Company	100%	100% of Membership Interest
Cliffs UTAC Holding LLC	Delaware	Cliffs Minnesota Mining Company	100%	N/A
United Taconite LLC	Delaware	Cliffs Minnesota Mining Company	70%	N/A
Cleveland-Cliffs International Holding Company	Delaware	Cleveland-Cliffs Inc.	100%	650 350
Cliffs Finance US LLC	Ohio	Cleveland-Cliffs International Holding Company	100%	1,050
Cliffs Finance Lux SCS	Luxembourg	Cleveland-Cliffs International Holding Company	99%	3,376,598,301
Cliffs Finance Lux SCS	Luxembourg	Cliffs Finance U.S. LLC	1%	34,107,053
Cliffs (Gibraltar) Holdings Limited	Gibraltar	Cliffs Finance Lux SCS	100%	167,407,000
Cliffs (Gibraltar) Limited	Gibraltar	Cliffs (Gibraltar) Holdings Limited	100%	167,406,851
Cliffs (Gibraltar) Holdings Limited Luxembourg S.C.S.	Luxembourg	Cliffs (Gibraltar) Holdings Limited	0.0019450.002%	75
Cliffs (Gibraltar) Holdings Limited Luxembourg S.C.S.	Luxembourg	Cliffs (Gibraltar) Limited	99.998055%	3,8555,000N/A
Cliffs Natural Resources Luxembourg S.ar.l.	Luxembourg	Cliffs (Gibraltar) Holdings Limited Luxembourg S.C.S.	100%	1,976,150
Cliffs International Lux I S.ar.I.	Luxembourg	Cliffs Natural Resources Luxembourg S.ar.l.	99.991963100%	1,878,554
Cliffs International Lux II S.ar.l.	Luxembourg	Cliffs International Lux I S.ar.l.	100%	1,871,852

Cliffs International Mineracao Brasil Ltda.	Brazil	Cliffs International Lux I S.ar.l.	53.19%	669,379
Cliffs International Mineracao Brasil Ltda.	Brazil	Cliffs International Lux II S.ar.l.	46.81%	589,077
Cliffs Internatioanl Luxembourg S.ar.l.	Luxembourg	Cliffs International Lux II S.ar.I.	100%	30,375
Cliffs International Management Company LLC	Delaware	Cleveland-Cliffs Inc.	100%	100% of Membership Interest
Cliffs Mining Company	Delaware	Cleveland-Cliffs Inc.	100%	27,000
Cliffs Mining Services Company	Delaware	Cleveland-Cliffs Inc.	100%	1,000
Cliffs Minnesota Mining Company	Delaware	Cleveland-Cliffs Inc.	100%	1,000
Lake Superior & Ishpeming Railroad Company	Michigan	Cleveland-Cliffs Inc.	100%	107,196
Northshore Mining Company	Delaware	Cleveland-Cliffs Inc.	100%	100
The Cleveland-Cliffs Iron Company	Ohio	Cleveland-Cliffs Inc.	100%	1,000
Cliffs Reduced Iron Corporation	Delaware	Cleveland-Cliffs Inc.	100%	1,000
Cliffs Reduced Iron Management Company	Delaware	Cliffs Reduced Iron Corporation	100%	1,000
Cliffs and Associates Limited	Trinidad	Cliffs Reduced Iron Corporation	82.4%	8,559,891
CALipso Sales Company	Delaware	Cliffs and Associates Limited	100%	1,000
Cliffs West Virginia Coal Inc.	Delaware	Cleveland-Cliffs Inc.	100%	100
Cliffs Renewable Energies LLC	Delaware	Cleveland-Cliffs Inc.	100%	N/A
Cleveland-Cliffs Minnesota Land Development LLC	Delaware	Cleveland-Cliffs Inc.	100%	100% of Membership Interest
IronUnits LLC	Delaware	Cleveland-Cliffs Inc.	100%	100% of Membership Interest
United Taconite LLC	Delaware	Cliffs UTAC Holding LLC	30%	N/A
Silver Bay Power Company	Delaware	Northshore Mining Company	100%	100
Cliffs TIOP Holding, LLC	Delaware	The Cleveland-Cliffs Iron Company	100%	100
Cliffs Empire Holding, LLC	Delaware	The Cleveland-Cliffs Iron Company	100%	100
Cleveland-Cliffs Ore Corporation	Ohio	The Cleveland-Cliffs Iron Company	100%	5
Cliffs Marquette, Inc.	Michigan	The Cleveland-Cliffs Iron Company	100%	500
Cliffs Biwabik Ore Corporation	Minnesota	Cleveland-Cliffs Ore Corporation	100%	100
Marquette Iron Mining Partnership	Michigan	Cleveland-Cliffs Ore Corporation	62%	N/A

Marquette Iron Mining Partnership	Michigan	Cliffs Marquette, Inc.	38%	N/A
Cliffs Oil Shale Corp.	Colorado	Marquette Iron Mining Partnership	100%	100
Northern Conservation, LLC	Minnesota	Marquette Iron Mining Partnership	100%	1,000
Republic Wetlands Preserve LLC	Michigan	Marquette Iron Mining Partnership	100%	N/A
The Cleveland-Cliffs Steamship Company	Delaware	The Cleveland-Cliffs Iron Company	100%	12,000
Pickands Hibbing Corporation	Minnesota	Cliffs Pickands Holding, LLC	100%	1,000
Cliffs Mining Holding Sub Company	Delaware	Cliffs Mining Holding, LLC	100%	100
Hibbing Development Company	Minnesota	Pickands Hibbing Corporation	39%	N/A
Hibbing Taconite Company	Minnesota	Cliffs Mining Holding Sub Company	10%	N/A
Hibbing Taconite Company	Minnesota	Hibbing Development Company	33.3%	N/A
Cliffs Empire, Inc.	Michigan	Cliffs Empire Holding, LLC	100%	500
Empire Iron Mining Partnership	Michigan	Cliffs Empire, Inc.	79%	N/A
Empire Iron Mining Partnership	Michigan	Cliffs Empire II, Inc.	21%	N/
Cliffs TIOP, Inc.	Michigan	Cliffs TIOP Holding, LLC	100%	500
Tilden Mining Company L.C.	Michigan	Cliffs TIOP, Inc.	85%	N/A
Tilden Mining Company L.C.	Michigan	Cliffs TIOP II, LLC	15%	N/A
Marquette Range Coal Service Company	Michigan	Empire Iron Mining Partnership	48.57%	392 93.70
Marquette Range Coal Service Company	Michigan	Tilden Mining Company L.C.	51.43%	514.3
Cliffs Empire II Inc.	Delaware	Cliffs Empire Holding, LLC	100%	100
Cliffs TIOP II, LLC	Delaware	Cliffs TIOP Holding, LLC	100%	500
Cliffs Australia Coal Pty Ltd	Australia	Cliffs Natural Resources Pty Ltd	100%	65,132,311
Cliffs Asia Pacific Iron Ore Holdings Pty Ltd	Australia	Cliffs Natural Resources Pty Ltd	100%	165,916,353
Cliffs Natural Resources Holdings Pty Ltd	Australia	Cliffs Natural Resources Luxembourg S.ar.l.	100%	N/A
Cliffs Natural Resources Pty Ltd	Australia	Cliffs Natural Resources Holdings Pty Ltd	100%	104,975,948
Cliffs Asia Pacific Iron Ore Pty Ltd	Australia	Cliffs Asia Pacific Iron Ore Holdings Pty Ltd	100%	N/A
Cliffs Asia Pacific Iron Ore Management Pty Ltd	Australia	Cliffs Asia Pacific Iron Ore Holdings Pty Ltd	100%	1,505,000

Cliffs Magnetite Holdings Pty Ltd	Australia	Cliffs Asia Pacific Iron Ore Holdings Pty Ltd	100%	2
Cliffs Asia Pacific Iron Ore Investment Pty Ltd	Australia	Cliffs Asia Pacific Iron Ore Holdings Pty Ltd	100%	100
Cliffs Australia Washplant Operations Pty Ltd	Australia	Cliffs Australia Coal Pty Ltd	100%	54,189,415

SCHEDULE 4.1(d)

SUBSCRIPTIONS, OPTIONS, WARRANTS, CALLS

None.

SCHEDULE 4.11

ENVIRONMENTAL MATTERS

None.

SCHEDULE 5.1

Deliver to Agent (and if so requested by Agent, with copies for each Lender) each of the financial statements, reports, or other items set forth below at the following times in a form reasonably satisfactory to Agent:

within 45 days after the end of each of Parent's fiscal quarters,	(a) an unaudited consolidated balance sheet, income statement, statement of cash flow, and statement of shareholder's equity covering Parent's and its Subsidiaries' operations during such period and compared to the prior period and plan, together with a corresponding discussion and analysis of results from management (or notice to the Administrative Agent that such discussion and analysis is in the Parent's Form 10-Q), and
	(b) a Compliance Certificate along with the underlying calculations, including the calculations to arrive at EBITDA and Fixed Charge Coverage Ratio.
within 90 days after the end of each of Parent's fiscal years,	(c) consolidated financial statements of Parent and its Subsidiaries for each such fiscal year, audited by independent certified public accountants of national standing or otherwise reasonably acceptable to Agent and accompanied by an unqualified opinion (as defined in <u>Section 1.2</u> of the Agreement), of such accountants to the effect that such audited financial statements have been prepared in accordance with GAAP (such audited financial statements to include a balance sheet, income statement, statement of cash flow, and statement of shareholder's equity, and, if prepared, such accountants' letter to management),
	(d) unaudited consolidating financial statements of Parent and the other Loan Parties for each such fiscal year, and
	(e) a Compliance Certificate along with the underlying calculations, including the calculations to arrive at EBITDA and Fixed Charge Coverage Ratio.
within 60 days after the end of each of Parent's fiscal years,	(f) copies of Parent's Projections, in a form reasonably satisfactory to Agent, in its Permitted Discretion, for the forthcoming 3 years, year by year, and for the forthcoming fiscal year, fiscal quarter by fiscal quarter, certified by the chief financial officer or other senior financial officer of Parent as being prepared in good faith on the basis of the assumptions stated therein, which assumptions were believed to be reasonable by Parent at the time made and at the time so furnished.
if and when filed by Parent,	(g) Form 10-Q quarterly reports, Form 10-K annual reports, and Form 8-K current reports,
	(h) any other filings made by Parent with the SEC, and
	(i)any other information that is provided by Parent to its shareholders generally.
within 5 Business Days after any Borrower has knowledge of any event or condition that constitutes a Default or an Event of Default,	(j) notice of such event or condition and a statement of the curative action that Borrowers propose to take with respect thereto.
within 5 Business Days after any Borrower has knowledge thereof,	(k) notice of all actions, suits, or proceedings brought by or against any Borrower or any of its Restricted Subsidiaries before any Governmental Authority which would reasonably be expected to result in a Material Adverse Effect.
promptly and in any event within 30 days after the filing thereof with the United States Department of Labor or other Governmental Authority,	(I) copies of each Schedule SB (Actuarial Information) to the Annual Report (Form 5500 Series) (or any equivalent under Australian law) including, to the extent required, the related financial and actuarial statements and opinions and other supporting statements, certifications, schedules and information with respect to each Pension Plan.

Agent,	(m) (i) commencement of any Environmental Action or written notice that an Environmental Action will be filed against a Borrower or its Subsidiaries, in each case, to the extent such Environmental Action would reasonably be expected to result in a Material Adverse Effect, and (ii) written notice of a violation, citation, or other administrative order from a Governmental Authority, to the extent such violation, citation, or other administrative order would reasonably be expected to result in a Material Adverse Effect.
	(n) any other information reasonably requested relating to the financial condition of any Borrower or its Restricted Subsidiaries.
promptly after any Borrower has knowledge thereof,	(o)any loss of ABL Priority Collateral exceeding \$10,000,000 covered by a Borrower's or a Borrower's Subsidiary's casualty or business interruption insurance; <u>provided</u> that during the continuance of an Event of Default or a Cash Dominion Trigger Period, the threshold included in this sentence shall be \$2,500,000.

For the avoidance of doubt, each document described above may be delivered electronically pursuant to the terms of Section 5.1 or Section 11 of the Agreement.

SCHEDULE 5.16(a)

ADI ACCOUNTS

Account Name	Account number	Bank	CNCY	Туре
CLIFFS ASIA PACIFIC IRON ORE PTY LTD				
CLIFFS ASIA PACIFIC IRON ORE PTY LTD		Bank of China	CNY	Operating Account
CLIFFS ASIA PACIFIC IRON ORE PTY LTD		Bank of America Merrill Lynch	AUD	Operating Account
CLIFFS ASIA PACIFIC IRON ORE PTY LTD		Bank of America Merrill Lynch	USD	
CLIFFS NATURAL RESOURCES PTY LTD				
CLIFFS NATURAL RESOURCES PTY LTD		Commonwealth Bank of Australia	AUD	Operating Account
CLIFFS NATURAL RESOURCES PTY LTD		Commonwealth Bank of Australia	AUD	Term Deposit
CLIFFS NATURAL RESOURCES PTY LTD		Bank of America Merrill Lynch	USD	Master ZBA
CLIFFS NATURAL RESOURCES PTY LTD		Bank of America Merrill Lynch	USD	Operating Account
CLIFFS NATURAL RESOURCES PTY LTD		Bank of America Merrill Lynch	AUD	Master ZBA
CLIFFS NATURAL RESOURCES PTY LTD		Bank of America Merrill Lynch	USD	Operating Account

SCHEDULE 5.16(b)

DEPOSIT ACCOUNTS

CLEVELAND-CLIFFS INC.	Account number	Institution	Туре
CLEVELAND-CLIFFS INC.		Bank of America Merrill Lynch	Master
CLEVELAND-CLIFFS INC.		Bank of America Merrill Lynch	MMDA
CLEVELAND-CLIFFS INC.		Bank of America Merrill Lynch	INVESTMENT
CLEVELAND-CLIFFS INC.		Huntington National Bank	MMDA
CLEVELAND-CLIFFS INC CDA		PNC Bank N.A.	CDA
CLEVELAND-CLIFFS INC DDA		PNC Bank N.A.	Master ZBA
CLEVELAND-CLIFFS INC.		PNC Bank N.A.	MEDICAL BENEFITS
CLEVELAND-CLIFFS INC.		PNC Bank N.A.	MMDA
CLEVELAND-CLIFFS INC.		CITIZENS BANK	MMDA
CLEVELAND-CLIFFS INC.		TREASURY CURVE	TreasuryCurve MMF Portal
CLEVELAND-CLIFFS INC.		Regions Bank	Enhanced Cash Portfolio
CLEVELAND-CLIFFS INC.		BANK OF NEW YORK MELLON	DB/CCI Salary
CLEVELAND-CLIFFS INC.		BANK OF AMERICA MERRILL LYNCH	Deposit Account (Notes Collateral Account)
CLEVELAND-CLIFFS INC.		PNC Bank, N.A.	Accelerated Maturity Option Certificate of Deposit
CLEVELAND-CLIFFS INC.		PNC Bank, N.A.	Accelerated Maturity Option Certificate of Deposit
CLEVELAND-CLIFFS INC.		PNC Bank, N.A.	Accelerated Maturity Option Certificate of Deposit
CLEVELAND-CLIFFS INC.		PNC Bank, N.A.	Accelerated Maturity Option Certificate of Deposit
CLEVELAND-CLIFFS INC.		PNC Bank, N.A.	Enhance Cash Deposit
THE CLEVELAND-CLIFFS IRON COMPANY			
CCICo HOURLY		BANK OF NEW YORK MELLON	DB/CCI Salary
CCICo - EMPIRE		BANK OF NEW YORK MELLON	VEBA
CCICo - TILDEN		BANK OF NEW YORK MELLON	VEBA
CCICo		BANK OF NEW YORK MELLON	SUPP UNEMP BENE CCICO
CLIFFS MINING COMPANY			
CLIFFS MINING COMPANY DBA MITTAL USA - ONTARIO IRON		PNC Bank N.A.	DDA
CLIFFS MINING COMPANY		Bank of Montreal	DDA
CLIFFS MINING COMPANY		Bank of America Merrill Lynch	DDA
EMPIRE IRON MINING PARTNERSHIP			
EMPIRE IRON MINING PARTNERSHIP		RANGE BANK	Commercial Checking

LAKE SUPERIOR & ISHPEMING RAILROAD COMPANY		
LAKE SUPERIOR & ISHPEMING RAILROAD COMPANY	WELLS FARGO N.A.	Checking
LAKE SUPERIOR & ISHPEMING RAILROAD COMPANY	PNC Bank, N.A.	DDA
NORTHSHORE MINING COMPANY		
NORTHSHORE MINING COMPANY	WELLS FARGO N.A.	DDA
NORTHSHORE MINING COMPANY	PNC Bank, N.A.	DDA
UNITED TACONITE LLC		
UNITED TACONITE LLC	WELLS FARGO N.A.	Checking
UNITED TACONITE LLC	WELLS FARGO N.A.	Savings
ORE MINING COMPANIES	BANK OF NEW YORK MELLON	Ore Mining (Utac)
UNITED TACONITE LLC	PNC Bank N.A.	DDA
UNITED TACONITE LLC	BANK OF NEW YORK MELLON	SUPP UNEMP BENE UTAC

SCHEDULE 5.2

Provide Agent (and if so requested by Agent, with copies for each Lender) with each of the documents set forth below at the following times in a form reasonably satisfactory to Agent:

monthly, within 20 days after the end of each	(a) an executed Borrowing Base Certificate,
fiscal month during each of Parent's fiscal years, or during any Increased	(b) a detailed aging, by total, of Borrowers' Accounts, together with a reconciliation and supporting documentation for any reconciling items noted,
Borrowing Base Period (as defined below), weekly (no later than	(c) a monthly Account roll-forward, in a format reasonably acceptable to Agent in its Permitted Discretion, tied to the beginning and ending account receivable balances of Borrowers' general ledger,
Wednesday of each week as of and for the immediately preceding	(d) notice of all material claims, offsets, or disputes asserted by Account Debtors with respect to Borrowers' Accounts,
week),	(e) Inventory system/perpetual reports specifying the cost and the wholesale market value and Net Orderly Liquidation Value of each Borrower's Inventory, by category, with additional detail showing additions to and deletions therefrom,
	(f) Equipment reports specifying the Net Book Value and Net Orderly Liquidation Value of each Borrower's Mobile Equipment, with reasonable additional detail showing additions to and deletions therefrom,
	(g) a detailed calculation of Inventory and Equipment categories that are not eligible for the Borrowing Base,
	(h) a summary aging, by vendor, of Borrowers' and their Restricted Subsidiaries' accounts payable and any book overdraft and an aging, by vendor, of any held checks, and
	(i) a detailed report regarding Borrowers' and their Restricted Subsidiaries' cash and Cash Equivalents, including an indication of which amounts constitute Qualified Cash (items (a) - (i) are referred to as the "Borrowing Base Materials").
promptly upon the request of Agent,	(j) other reasonably requested supporting documentation related to the Borrowing Base Materials.
monthly, within 30 days after the end of each fiscal month during each of Parent's fiscal years,	(k) a reconciliation of Accounts, trade accounts payable, Inventory and Mobile Equipment of Borrowers' general ledger accounts to their monthly financial statements including any book reserves related to each category.
quarterly, within 45 days after the end of each	(I) a report regarding each Borrower's and its Restricted Subsidiaries' accrued, but unpaid, ad valorem taxes.
fiscal quarter during each of Parent's fiscal years,	(m) a report setting forth the current location of Borrowers' Inventory and Mobile Equipment included in the most recent U.S. Borrowing Base Certificate.
promptly, upon the reasonable request by Agent (but no more frequently than semi-	(n) copies of material purchase orders and invoices for Inventory and Equipment of any Borrower or its Restricted Subsidiaries and/or corresponding shipping and delivery documents and credit memos, in each case, together with corresponding supporting documentation.
annually),	(o) such other reports as to the Collateral or the financial condition of any Borrower and its Restricted Subsidiaries, as Agent may request in its Permitted Discretion, including intra-quarter updates as to the locations of Inventory and Mobile Equipment included in the most recent U.S. Borrowing Base Certificate.

As used herein, "Increased Borrowing Base Reporting Period" means a period which shall commence on any date of determination (the "Commencement Date") on which Excess Availability is less than the greater of (a) 12.5% of the Line Cap, and (b) \$50,000,000, and shall continue until the last day of the first fiscal month ending after the

Commencement Date when Liquidity has been greater than or equal to (x) 12.5% of the Line Cap, and (y) \$50,000,000 for a period of 60 consecutive days.

For the avoidance of doubt, each document described above may be delivered electronically, to the extent such an electronic delivery system has been implemented, pursuant to the terms of Section 5.2 or Section 11 of the Agreement.

CLEVELAND-CLIFFS INC. AMENDED AND RESTATED 2015 EQUITY AND INCENTIVE COMPENSATION PLAN

RESTRICTED STOCK UNIT AWARD MEMORANDUM

Employee:	PARTICIPANT NAME
Date of Grant:	2/21/2018
Grant Price:	\$ <u></u>
Number of Restricted Stock Units (Common Shares) Subject to Award:	SHARES GRANTED
Vesting Date:	December 31, 2020

Additional terms and conditions of your award are included in the Restricted Stock Unit Award Agreement. As a condition to your receipt of this award, you must log on to Fidelity's website at www.netbenefits.fidelity.com and accept the terms and conditions of this award within 90 calendar days of your Date of Grant. If you do not accept the terms and conditions of this award within such time at www.netbenefits.fidelity.com, this award may be forfeited and immediately terminate.

<u>Note</u>: Section 2.1 of the Restricted Stock Unit Award Agreement contains provisions that restrict your activities. These provisions apply to you and, by accepting this award, you agree to be bound by these restrictions.

CLEVELAND-CLIFFS INC. AMENDED AND RESTATED 2015 EQUITY AND INCENTIVE COMPENSATION PLAN

Restricted Stock Unit Award Agreement

This Restricted Stock Unit Award Agreement (this "Agreement") is between Cleveland-Cliffs Inc., an Ohio corporation (the "Company"), and you, the person named in the Restricted Stock Unit Award Memorandum (the "Award Memorandum") who is an employee of the Company or a Subsidiary of the Company (the "Participant"). For purposes of this Agreement, "Employer" means the entity (the Company or Subsidiary) that employs the Participant on the applicable date. This Agreement is effective as of the Date of Grant set forth in the Award Memorandum.

The Company wishes to award to the Participant Restricted Stock Units representing the opportunity to earn a number of Common Shares, subject to the terms and conditions set forth in this Agreement, in order to carry out the purpose of the Cliffs Natural Resources Inc. Amended and Restated 2015 Equity and Incentive Compensation Plan (the "Plan"). All capitalized terms not defined in this Agreement shall have the same meaning as set forth in the Plan. See Section 2 of the Plan for a list of certain defined terms.

In the event of a conflict between the terms of this Agreement, the Award Memorandum and the terms of the Plan, the terms of the Plan shall govern. In the event of a conflict between the terms of this Agreement and the Award Memorandum, the terms of this Agreement shall govern.

ARTICLE 1 Grant and Terms of Restricted Stock Units

- 1.1 <u>Grant of Restricted Stock Units</u>. Pursuant to the Plan, the Company has granted to the Participant the number of Restricted Stock Units as specified in the Award Memorandum, with dividend equivalents ("<u>Restricted Stock Units</u>"), effective as of the Date of Grant.
- 1.2 <u>Vesting As Condition of Payment</u>. The Restricted Stock Units evidenced by this Agreement and these terms and conditions shall only result in the issuance of Common Shares equal in number to the Restricted Stock Units to the extent the Participant is "<u>Vested</u>" in the Restricted Stock Units on the date the Restricted Stock Units are to be paid as specified in Section 1.4. The Restricted Stock Units will become Vested as follows:
- (a) <u>Employment Through Vesting Period.</u> The Participant will become 100% Vested in all the Restricted Stock Units subject to this award if the Participant remains in the continuous employ of the Company or a Subsidiary throughout the period beginning on the Date of Grant and ending on the Vesting Date, as set forth in the Award Memorandum ("Vesting Period").
- (b) <u>Death, Disability, Retirement or a Termination Without Cause</u>. If the Participant experiences a termination of employment with the Company because of the Participant's death, Disability (as defined herein) or Retirement (as defined herein) or a termination of employment by the Company without Cause (as defined herein) during the Vesting Period, the Participant shall become Vested in a prorated number of Restricted Stock Units equal to the product of the number of Restricted Stock Units subject to this award, multiplied by a fraction, the numerator of which is the number of full months the Participant was employed with the Company or a Subsidiary between January 1, 2018 and the date of the Participant's termination of employment, and the denominator of which is 36, rounded down to the nearest whole Restricted Stock Unit.

For purposes of this Agreement, "<u>Disability</u>" shall mean a medically determinable physical or mental impairment that can be expected to result in death or can be expected to last for a continuous period of not less than 12 months and that results in the Participant: (i) being unable to engage in any substantial gainful activity; or (ii) receiving income replacement benefits for a period of not less than three months under an accident or health plan covering employees of the Company. For purposes of this Agreement, "<u>Retirement</u>" shall mean the Participant's retirement from active employment with the Company or Subsidiary upon or after the attainment of at least age 55 and at least a 5-year period of service with the Company and/or Subsidiary.

(c) <u>Change in Control</u>. In the event a Change in Control occurs during the Vesting Period, the Participant will become Vested in the Restricted Stock Units only to the extent provided in Section 1.3.

In the event the Participant otherwise terminates employment prior to becoming Vested in the Restricted Stock Units or the Participant's employment is terminated by the Company for Cause, the Participant shall forfeit all rights to any Restricted Stock Units evidenced by this Agreement.

1.3 Change in Control Vesting.

- (a) If the Participant remains in the continuous employ of the Company or a Subsidiary throughout the period beginning on the Date of Grant and ending on the date of a Change in Control, the Participant will become 100% Vested in all the Restricted Stock Units evidenced by this Agreement upon the Change in Control, except to the extent that an award meeting the requirements of Section 1.3(d) (a "Replacement Award") is provided to the Participant in accordance with Section 1.3(d) to replace, adjust or continue the award of Restricted Stock Units covered by this Agreement (the "Replaced Award"). If a Replacement Award is provided, references to Restricted Stock Units in this Agreement shall be deemed to refer to the Replacement Award after the Change in Control.
- (b) If, upon or after receiving a Replacement Award, the Participant experiences a termination of employment with the Company or a Subsidiary of the Company (or any of their successors) (as applicable, the "Successor") by reason of the Participant terminating employment for Good Reason or the Successor terminating the Participant's employment other than for Cause, in each case within a period of two years after the Change in Control and during the Vesting Period, the Participant shall become 100% Vested in the Replacement Award upon such termination.
- (c) If a Replacement Award is provided, notwithstanding anything in this Agreement to the contrary, any outstanding Restricted Stock Units that at the time of the Change in Control are not subject to a "substantial risk of forfeiture" (within the meaning of Section 409A of the Code) will be deemed to be Vested at the time of such Change in Control and will be paid as provided for in Section 1.4(c).
- (d) For purposes of this Agreement, a "Replacement Award" means an award: (i) of the same type (e.g., time-based restricted stock units) as the Replaced Award; (ii) that has a value at least equal to the value of the Replaced Award; (iii) that relates to publicly traded equity securities of the Company or its successor in the Change in Control or another entity that is affiliated with the Company or its successor following the Change in Control; (iv) if the Participant holding the Replaced Award is subject to U.S. federal income tax under the Code, the tax consequences of which to such Participant under the Code are not less favorable to such Participant than the tax consequences of the Replaced Award; and (v) the other terms and conditions of which are not less favorable to the Participant holding the Replaced Award than the terms and conditions of the Replaced Award (including the provisions that would apply in the event of a subsequent Change in Control). A Replacement Award may be granted only to the extent it does not result in the Replaced Award or Replacement Award failing to comply with or be exempt from Section 409A of the Code. Without limiting the generality of the foregoing, the Replacement Award may take the form of a continuation of the Replaced Award if the requirements of the two preceding sentences are satisfied. The determination of whether the conditions of this Section 1.3(d) are satisfied will be made by the Committee, as constituted immediately before the Change in Control, in its sole discretion.
- (e) For purposes of this Agreement, a termination for "Cause" shall mean that, prior to termination of employment, the Participant shall have committed: (i) and been convicted of a criminal violation involving fraud, embezzlement or theft in connection with his or her duties or in the course of his or her employment with the Company or any Affiliate (or the Successor, if applicable); (iii) intentional wrongful damage to property of the Company or any Affiliate (or the Successor, if applicable); (iii) intentional wrongful disclosure of secret processes or confidential information of the Company or any Affiliate (or the Successor, if applicable); or (iv) intentional wrongful engagement in any competitive activity; and any such act shall have been demonstrably and materially harmful to the Company or any Affiliate (or the Successor, if applicable). For purposes of this Agreement, no act or failure to act on the part of the Participant shall be deemed "intentional" if it was due primarily to an error in judgment or negligence, but shall be deemed "intentional" only if done or omitted to be done by the Participant not in good faith and without reasonable belief that the Participant's action or omission was in the best interest of the Company or an Affiliate (or the Successor, if applicable).
- (f) A termination "for Good Reason" shall mean the Participant's termination of employment with the Successor as a result of the initial occurrence, without the Participant's consent, of one or more of the following events:

- i. a material diminution in the Participant's annual base salary rate as in effect from time to time (" Base Pay");
- ii. a material diminution in the Participant's authority, duties or responsibilities;
- iii. a material change in the geographic location at which the Participant must perform services;
- iv. a reduction in the Participant's opportunity regarding annual bonus, incentive or other payment of compensation, in addition to Base Pay, made or to be made in regard to services rendered in any year or other period pursuant to any bonus, incentive, profit-sharing, performance, discretionary pay or similar agreement, policy, plan, program or arrangement (whether or not funded) of the Successor; and
- v. any other action or inaction that constitutes a material breach by the Participant's employer of the employment agreement, if any, under which the Participant provides services.

Notwithstanding the foregoing, "Good Reason" shall not be deemed to exist unless: (A) the Participant has provided notice to his or her employer of the existence of one or more of the conditions listed in (i) through (v) above within 90 days after the initial occurrence of such condition or conditions; and (B) such condition or conditions have not been cured by the Participant's employer within 30 days after receipt of such notice.

1.4 Payment of Restricted Stock Units.

- (a) Payment After the Vesting Period. Subject to Sections 1.4(b) and (c), the Restricted Stock Units that are Vested as of the Vesting Date shall be paid after the end of the Vesting Period, but in any event no later than 2-1/2 months after the end of the Vesting Period to the extent they have not been previously paid to the Participant.
- (b) <u>Payment After Death, Disability, Retirement or a Termination Without Cause</u>. Notwithstanding Section 1.4(a), if the Participant experiences a termination of employment with the Company because of the Participant's death, Disability or Retirement or termination of employment by the Company without Cause or by the Participant for Good Reason during the Vesting Period, the Vested Restricted Stock Units will be paid within 30 days following the date of such termination. Any payment of Restricted Stock Units to a deceased Participant shall be paid to the estate of the Participant, unless the Participant files a completed Designation of Death Beneficiary with the Company in accordance with its procedures.
- (c) <u>Change in Control</u>. Notwithstanding Section 1.4(a) and Section 1.4(b), to the extent any Restricted Stock Units are Vested as of a Change in Control, such Vested Restricted Stock Units will be paid within 10 days of the Change in Control; <u>provided</u>, <u>however</u>, that if such Change in Control would not qualify as a permissible date of distribution under Section 409A(a)(2)(A) of the Code, and the regulations thereunder, and where Section 409A of the Code applies to such distribution, payment will be made on the date that would have otherwise applied pursuant to this Section 1.4.
- (d) <u>Payment Following a Change in Control</u>. Notwithstanding Section 1.2 and Section 1.4(a), if, during the two-year period following a Change in Control, the Participant experiences a qualifying termination of employment (as described in Section 1.3(b)), the Restricted Stock Units that are Vested as of the date of such termination of employment shall be paid within 30 days of such termination of employment to the extent they have not been previously paid to the Participant; <u>provided</u>, <u>however</u>, that if such Change in Control would not qualify as a permissible date of distribution under Section 409A(a)(2)(A) of the Code, and the regulations thereunder, and where Section 409A of the Code applies to such distribution, payment will be made on the date that would have otherwise applied pursuant to this Section 1.4.
- (e) <u>General</u>. The Restricted Stock Units are to be settled in Common Shares. The Committee shall withhold Common Shares to the extent necessary to satisfy income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related item withholding requirements, as described in Section 4.3. In addition, the Committee may restrict 50% of the Common Shares to be issued in satisfaction of the total Restricted Stock Units, before income tax withholding, so that they cannot be sold by the Participant unless immediately after such sale the Participant is in compliance with the Company's share ownership guidelines that are applicable to the Participant at the time of sale.

(f)	Payment Obliga	tion. Prior to payment,	the Company sh	all only have an u	ınfunded and u	nsecured obliga	tion to make p	aymen
of Restricted Stock Uni	ts to the Participant.	The Restricted Stock	Units evidenced	by this Agreement	t that have not	yet been earned	d, and any inte	rests o
the Participant with res	pect thereto, are not	transferable other than	n pursuant to the	laws of descent a	nd distribution,	or in accordanc	e with Section	1.4(b).

(g) No Shareholder Rights. The Participant shall have no rights of ownership in the Common Shares underlying the Restricted Stock Units and no right to vote the Common Shares underlying the Restricted Stock Units until the date on which the Common Shares underlying the Restricted Stock Units are issued or transferred to the Participant pursuant to this Section 1.4.

ARTICLE 2 Other Terms and Conditions

2.1 Non-Compete and Confidentiality.

- (a) The Participant shall not render services for any organization or engage directly or indirectly in any business that is a competitor of the Company or any Affiliate of the Company, or which organization or business is or plans to become prejudicial to or in conflict with the business interests of the Company or any Affiliate of the Company or distribute any secret or confidential information belonging to the Company or any Affiliate of the Company.
- (b) Failure to comply with Section 2.1(a) above will cause the Participant to forfeit the right to Restricted Stock Units and require the Participant to reimburse the Company for the taxable income received on Restricted Stock Units that have been paid out in Common Shares within the 90-day period preceding the Participant's termination of employment.

ARTICLE 3 Acknowledgments

- 3. Acknowledgments. In accepting the award, the Participant acknowledges, understands and agrees to the following:
 - (a) The Plan is established voluntarily by the Company, it is discretionary in nature and it may be modified, amended, suspended or terminated by the Company at any time, to the extent permitted by the Plan;
 - (b) The grant of the Restricted Stock Units is voluntary and occasional and does not create any contractual or other right to receive future grants of Restricted Stock Units, or benefits in lieu of Restricted Stock Units, even if Restricted Stock Units have been granted in the past;
 - (c) All decisions with respect to future Restricted Stock Units or other grants, if any, will be at the sole discretion of the Company;
 - (d) The Participant's participation in the Plan is voluntary;
 - (e) The Restricted Stock Unit Award and the Participant's participation in the Plan shall not create a right to employment or be interpreted as forming an employment or services contract with the Company or any Subsidiary and shall not interfere with the ability of the Company, or any Subsidiary, as applicable, to terminate the Participant's employment or service relationship (if any);
 - (f) The future value of the underlying Common Shares is unknown, indeterminable and cannot be predicted with certainty:
 - (g) No claim or entitlement to compensation or damages shall arise from forfeiture of any Restricted Stock Units resulting from the Participant ceasing to provide employment or other services to the Company or a Subsidiary (for any reason whatsoever whether or not later found to be invalid or in breach of employment laws in the jurisdiction where the Participant

is employed or the terms of the Participant's employment agreement, if any), and in consideration of the grant of the Restricted Stock Units to which the Participant is otherwise not entitled, the Participant irrevocably agrees never to institute any claim against the Company or any of its Subsidiaries, and the Participant waives his or her ability, if any, to bring any such claim, and releases the Company and its Subsidiaries from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, the Participant shall be deemed irrevocably to have agreed not to pursue such claim and agrees to execute any and all documents necessary to request dismissal or withdrawal of such claim;

- (h) Neither the Plan nor the Restricted Stock Units shall be construed to create an employment relationship where any employment relationship did not otherwise already exist;
- (i) The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding the Participant's participation in the Plan, or the Participant's acquisition or sale of the underlying Common Shares. The Participant is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the Restricted Stock Units;
- (j) The Restricted Stock Units and the Common Shares subject to the Restricted Stock Units, and the income and value of same, are not part of normal or expected compensation for purposes of calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments; and
- (k) The Company reserves the right to impose other requirements on participation in the Restricted Stock Units and on any Common Shares acquired under the Plan, to the extent the Company determines it is necessary or advisable in order to comply with local law or other applicable rules or facilitate the administration of the Plan, and to require the Participant to sign any additional agreements or undertakings that may be necessary to accomplish the foregoing.

ARTICLE 4 General Provisions

- 4.1 Compliance with Law. The Company shall make reasonable efforts to comply with all applicable federal and state securities laws; provided, however, notwithstanding any other provision of the Agreement and these terms and conditions, the Company shall not be obligated to issue any Common Shares pursuant to the Agreement and these terms and conditions if the issuance or payment thereof would result in a violation of any such law; provided further, however, that the Common Shares will be issued at the earliest date at which the Company reasonably anticipates that the issuance of the Common Shares will not cause such violation. Notwithstanding anything in this Agreement to the contrary, nothing in this Agreement prevents the Participant from providing, without prior notice to the Company, information to governmental authorities regarding possible legal violations or otherwise testifying or participating in any investigation or proceeding by any governmental authorities regarding possible legal violations, and for purpose of clarity the Participant is not prohibited from providing information voluntarily to the Securities and Exchange Commission pursuant to Section 21F of the Exchange Act.
- **4.2** <u>Dividend Equivalents</u>. During the period beginning on the Date of Grant and ending on the date that the Restricted Stock Units are paid in accordance with Section 1.4, the Participant will be entitled to dividend equivalents on Restricted Stock Units equal to the cash dividend or distribution that would have been paid on the Restricted Stock Units had the Restricted Stock Units been issued and outstanding Common Shares on the record date for the dividend or distribution. Such accrued dividend equivalents (a) will vest and become payable upon the same terms and at the same time of settlement as the Restricted Stock Units to which they relate, and (b) will be denominated and payable solely in cash.
- 4.3 <u>Withholding Taxes</u>. To the extent that the Company is required to withhold federal, state, local or foreign taxes in connection with any payment made or benefit realized by the Participant under this Agreement, the Company shall withhold Common Shares having a value equal to the amount required to be withheld. The shares used for tax withholding will be valued at an amount equal to the fair market value on the date the benefit is to be included in the Participant's income. In no event will the market value of the Common Shares to be withheld and

delivered pursuant to this Section to satisfy applicable withholding taxes in connection with the benefit exceed the maximum amount of taxes that could be required to be withheld (subject to the Participant's estimated tax obligations attributable to the applicable transaction).

- **4.4** Continuous Employment. For purposes of this Agreement, the continuous employment of the Participant with the Company shall not be deemed to have been interrupted, and the Participant shall not be deemed to have separated from service with the Company, by reason of the transfer of his employment among the Company or Subsidiaries or an approved leave of absence, unless otherwise indicated in the Plan or if required to comply with Section 409A of the Code.
- 4.5 <u>Relation to Other Benefits</u>. Any economic or other benefit to the Participant under the Agreement and these terms and conditions or the Plan shall not be taken into account in determining any benefits to which the Participant may be entitled under any profit-sharing, retirement or other benefit or compensation plan maintained by the Company or a Subsidiary and shall not affect the amount of any life insurance coverage available to any beneficiary under any life insurance plan covering employees of the Company or Subsidiary.
 - 4.6 Adjustments. Restricted Stock Units evidenced by this Agreement are subject to mandatory adjustment as provided in Section 11 of the Plan.
- **4.7** These Terms and Conditions Subject to Plan. The Restricted Stock Units covered under the Agreement and all of the terms and conditions hereof are subject to all of the terms and conditions of the Plan, a copy of which is available upon request.
- **4.8** <u>Transferability.</u> Except as otherwise provided in the Plan, the Restricted Stock Units are non-transferable and any attempts to assign, pledge, hypothecate or otherwise alienate or encumber (whether by law or otherwise) any Restricted Stock Units shall be null and void.
- **4.9** <u>Data Privacy</u>. The Participant hereby explicitly and unambiguously consents to the collection, use and transfer, in electronic or other form, of the Participant's personal data as described in this Agreement and any other Restricted Stock Unit award materials by and among, as applicable, the Company or Subsidiaries for the exclusive purpose of implementing, administering and managing the Participant's participation in the Plan.

The Participant understands that the Company or Subsidiary may hold certain personal information about the Participant, including, but not limited to, the Participant's name, home address and telephone number, date of birth, social security number or other identification number, salary, nationality, job title, any Common Shares of or directorships in the Company that are held, details of all Restricted Stock Units or any other entitlement to Common Shares awarded, canceled, exercised, vested, unvested or outstanding in the Participant's favor, for the exclusive purpose of implementing, administering and managing the Plan ("Data").

The Participant understands that Data will be transferred to the Company's broker, or such other stock plan service provider as may be selected by the Company in the future, which is assisting the Company with the implementation, administration and management of the Plan. The Participant understands that the recipients' use of the Data may be located in the United States or elsewhere, and that the recipients' country (e.g., the United States) may have different data privacy laws and protections than the Participant's country. The Participant understands that if he or she resides outside the United States, he or she may request a list with the names and addresses of any potential recipients of the Data by contacting his or her local human resources representative. The Participant authorizes the Company, the Company's broker and any other possible recipients which may assist the Company (presently or in the future) with implementing, administering and managing the Plan to receive, possess, use, retain and transfer the Data, in electronic or other form, for the sole purpose of implementing, administering and managing the Participants's participation in the Plan. The Participant understands that Data will be held only as long as is necessary to implement, administer and manage the Participant's participation in the Plan. The Participant understands if he or she resides outside the United States, he or she may, at any time, view their respective Data, request additional information about the storage and processing of their Data, require any necessary amendments to their Data or refuse or withdraw the consents herein, in any case without cost, by contacting in writing his or her local human resources representative. Further, the Participant understands that he or she is providing the consents herein on a purely voluntary basis. If the Participant does not consent, or if the Participant later seeks to revoke his or her consent, his or her employment status or service and career with the Employer will not be adversely

withdrawing his or her consent may affect the Participant's ability to participate in the Plan. For more information on the consequences of the Participant's refusal to consent or withdrawal of consent, the Participant understands that he or she may contact his or her local human resources representative.

- **4.10** <u>Amendments</u>. This Agreement can be amended at any time by the Committee. Any amendment to the Plan shall be deemed to be an amendment to this Agreement to the extent that the amendment is applicable hereto. Except for amendments necessary to bring this Agreement into compliance with current law including Section 409A of the Code, no amendment to this Agreement shall materially and adversely affect the rights of the Participant without the Participant's written consent.
- **4.11** <u>Severability</u>. The provisions of this Agreement are severable and if any one or more provisions are determined to be illegal or otherwise unenforceable, in whole or in part, the remaining provisions shall nevertheless be binding and enforceable.
- **4.12** <u>Electronic Delivery</u>. The Company may, in its sole discretion, decide to deliver any documents related to the Restricted Stock Units by electronic means. By accepting this award of Restricted Stock Units, the Participant hereby consents to receive such documents by electronic delivery and agrees to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.
- **4.13** Headings. Headings are given to the articles or sections of this Agreement solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Agreement or any provision hereof.
 - 4.14 Governing Law. This Agreement is governed by and construed in accordance with the internal substantive laws of the State of Ohio.
- 4.15 Section 409A of the Code. To the extent applicable, it is intended that this Agreement and the Plan comply with the provisions of Section 409A of the Code. This Agreement and the Plan shall be administered in a manner consistent with this intent, and any provision that would cause the Agreement or the Plan to fail to satisfy Section 409A of the Code shall have no force and effect until amended to comply with Section 409A of the Code (which amendment may be retroactive to the extent permitted by Section 409A of the Code and may be made by the Company without the consent of the Participant). The terms "termination of employment," "terminates employment," and similar words and phrases used in this Agreement mean a "separation from service" within the meaning of Treasury Regulation section 1.409A-1(h). If, at the time of the Participant's separation from service (within the meaning of Section 409A of the Code), (a) the Participant will be a specified employee (within the meaning of Section 409A of the Code and using the identification methodology selected by the Company from time to time) and (b) the Company makes a good faith determination that an amount payable hereunder constitutes deferred compensation (within the meaning of Section 409A of the Code) the payment of which is required to be delayed pursuant to the six-month delay rule set forth in Section 409A of the Code in order to avoid taxes or penalties under Section 409A of the Code, then the Company will not pay such amount on the otherwise scheduled payment date but will instead pay it, without interest, on the fifth business day of the seventh month after such separation from service.

[Acceptance Page Contained in Exhibit A]

Exhibit A

ELECTRONIC ACCEPTANCE

Acceptance by the Participant

By selecting the "Accept Grant" box on the website of the Company's administrative agent, the Participant acknowledges acceptance of, and consents to be bound by, the Plan and this Agreement and any other rules, agreements or other terms and conditions incorporated herein by reference.

IF I FAIL TO ACKNOWLEDGE ACCEPTANCE OF THE AWARD WITHIN NINETY (90) DAYS OF THE DATE OF GRANT SET FORTH IN THE AGREEMENT, THE COMPANY MAY DETERMINE THAT THIS AWARD HAS BEEN FORFEITED.

PARTICIPANT NAME	ACCEPTANCE DATE
Participant Name	Date
ELECTRONIC SIGNATURE	
Participant Signature	

PRODUCT ID LTIPSU18

CLEVELAND-CLIFFS INC. AMENDED AND RESTATED 2015 EQUITY AND INCENTIVE COMPENSATION PLAN

PERFORMANCE SHARE AWARD MEMORANDUM (TSR)

Employee:

Date of Grant:

Number of Shares Subject to Award:

Performance Metric: Incentive Period:

----PARTICIPANT NAME

2/21/2018

SHARES GRANTED

Relative Total Shareholder Return January 1, 2018 - December 31, 2020

Additional terms and conditions of your award are included in the Performance Share Award Agreement. As a condition to your receipt of Shares, you must log on to Fidelity's website at www.netbenefits.fidelity.com and accept the terms and conditions of this award within 90 calendar days of your Date of Grant. If you do not accept the terms and conditions of this award within such time at www.netbenefits.fidelity.com, this award may be forfeited and immediately terminate.

Note: Section 3.1 of the Performance Share Award Agreement contains provisions that restrict your activities. These provisions apply to you and, by accepting this award, you agree to be bound by these restrictions.

CLEVELAND-CLIFFS INC. AMENDED AND RESTATED 2015 EQUITY AND INCENTIVE COMPENSATION PLAN

Performance Share Award Agreement

This Performance Share Award Agreement (this "<u>Agreement</u>") is between Cleveland-Cliffs Inc., an Ohio corporation (the "<u>Company</u>"), and you, the person named in the Performance Share Award Memorandum (the "<u>Award Memorandum</u>") who is an employee of the Company or a Subsidiary of the Company (the "<u>Participant</u>"). For purposes of this Agreement, "<u>Employer</u>" means the entity (the Company or Subsidiary) that employs the Participant on the applicable date. This Agreement is effective as of the Date of Grant set forth in the Award Memorandum.

The Company wishes to award to the Participant Performance Shares representing the opportunity to earn a number of Common Shares (the "<u>Shares</u>"), subject to the terms and conditions set forth in this Agreement, in order to carry out the purpose of the Cliffs Natural Resources Inc. Amended and Restated 2015 Equity and Incentive Compensation Plan (the "<u>Plan</u>"). All capitalized terms not defined in this Agreement shall have the same meaning as set forth in the Plan. See Section 2 of the Plan for a list of certain defined terms.

In the event of a conflict between the terms of this Agreement, the Award Memorandum and the terms of the Plan, the terms of the Plan shall govern. In the event of a conflict between the terms of this Agreement and the Award Memorandum, the terms of this Agreement shall govern.

ARTICLE1 Definitions

All terms used herein with initial capital letters shall have the meanings assigned to them in the Plan and the following additional terms, when used herein with initial capital letters, shall have the following meanings:

- 1.1 "Incentive Period" shall be the time period as set forth in the Award Memorandum.
- 1.2 "Market Value Price" shall mean the latest available closing price of a Share of the Company or the latest available closing price per share of a common share of each of the entities in the Peer Group, as the case may be, on the New York Stock Exchange or other recognized market if the shares do not trade on the New York Stock Exchange at the relevant time.
- 1.3 "Peer Group" shall mean the group of companies, as more particularly set forth on attached Exhibit A, against which the Relative Total Shareholder Return of the Company is measured over the Incentive Period.
- **1.4** "Performance Objective(s)" shall mean, for the Incentive Period, the predetermined objectives of the Company with respect to the Relative Total Shareholder Return goal established by the Committee and reported to the Board for this award, as more particularly set forth on attached Exhibit B.
- 1.5 "Performance Shares Earned" shall mean the number of Shares of the Company (or cash equivalent) earned by a Participant, as determined under Section 2.3.
- **1.6** "Relative Total Shareholder Return" shall mean for the Incentive Period the Total Shareholder Return of the Company compared to the Total Shareholder Return of the Peer Group, as more particularly set forth on attached Exhibit C.
- 1.7 "Share Ownership Guidelines" shall mean the Company's share ownership guidelines, as amended from time to time, which encourage such Directors and Officers to hold a meaningful stake in the Company.
- 1.8 "Total Shareholder Return" or "TSR" shall mean, for the Incentive Period, the cumulative return to shareholders of the relevant entity during the Incentive Period, measured by the change in Market Value Price per share of a common share of the entity, plus dividends (or other distributions, excluding franking credits) reinvested over the Incentive Period, determined on the last business day of the Incentive Period compared to a base measured by the average Market Value Price per share of a common share of the entity on the last business day of the year immediately preceding the Incentive Period. Dividends (or other distributions, excluding franking credits) per share are assumed to be reinvested in the applicable stock on the last business day of the quarter during which they are

paid at the then Market Value Price per share, resulting in a fractionally higher number of shares owned at the market price.

ARTICLE 2 Grant and Terms of Performance Shares

- 2.1 <u>Grant of Performance Shares</u>. Pursuant to the Plan, the Company has granted to the Participant an award covering the number of Performance Shares as specified in the Award Memorandum, with dividend equivalents ("<u>Performance Shares</u>"), effective as of the Date of Grant.
- 2.2 <u>Issuance of Performance Shares</u>. The Performance Shares covered by this Agreement and these terms and conditions shall only result in the issuance of Shares (or cash or a combination of Shares and cash, as decided by the Committee in its sole discretion), to the extent such Performance Shares have become Performance Shares Earned, as provided in Section 2.3, on the date the Performance Shares Earned are to be paid as specified in Section 2.5.

2.3 Performance Shares Earned.

- (a) Achievement of Company Performance Objective(s). Subject to Sections 2.3(b), and 2.3(c), the number of Performance Shares Earned, if any, shall be based upon the degree of achievement of the Company Performance Objective(s), all as more particularly set forth in Exhibit B, with actual Performance Shares Earned interpolated between the performance levels shown on Exhibit B, as determined and certified by the Committee as of the end of the Incentive Period. The percentage level of achievement determined for the Company Performance Objective(s) shall be multiplied by the number of Performance Shares to determine the actual number of Performance Shares Earned, rounded down to the nearest whole Performance Share. The calculation as to whether the Company has met or exceeded the Company Performance Objective(s) shall be determined and certified by the Committee in accordance with the award and these terms and conditions. Subject to the terms of the Plan, except as provided in Sections 2.3(b) and 2.3(c), no Performance Shares will become Performance Shares Earned unless the Participant remains in the continuous employment of the Company or a Subsidiary during the entire Incentive Period
- (b) <u>Death, Disability, Retirement or a Termination Without Cause</u>. If the Participant experiences a termination of employment with the Company because of the Participant's death, Disability (as defined herein) or Retirement (as defined herein) or a termination of employment by the Company without Cause (as defined herein) during the Incentive Period, the number of the Participant's Performance Shares that become Performance Shares Earned will be determined after the end of the Incentive Period under Section 2.3(a) (without regard to the requirement that employment continue until the end of the Incentive Period), prorated based upon the number of full months the Participant was employed with the Company or a Subsidiary between January 1, 2018 and the date of the Participant's termination of employment compared to the number of full months from January 1, 2018 to December 31, 2020, rounded down to the nearest whole Performance Share.

For purposes of this Agreement, "Disability" shall mean a medically determinable physical or mental impairment that can be expected to result in death or can be expected to last for a continuous period of not less than 12 months and that results in the Participant: (i) being unable to engage in any substantial gainful activity; or (ii) receiving income replacement benefits for a period of not less than three months under an accident or health plan covering employees of the Company. For purposes of this Agreement, "Retirement" shall mean the Participant's retirement from active employment with the Company or Subsidiary upon or after the attainment of age 55 and at least a 5-year period of service with the Company and/or Subsidiary.

(c) <u>Change in Control</u>. In the event a Change in Control occurs during the Incentive Period, the Participant's Performance Shares will become Performance Shares Earned only to the extent provided in Section 2.4.

In the event the Participant otherwise terminates employment prior to becoming entitled to Performance Shares Earned or the Participant's employment is terminated by the Company for Cause, the Participant shall forfeit all rights to any Performance Shares that were granted under this Agreement.

2.4 Change in Control Vesting.

- (a) If the Participant remains in the continuous employ of the Company or Subsidiary throughout the period beginning on the Date of Grant and ending on the date of a Change in Control, upon the Change in Control, 100% of the Performance Shares shall become Performance Shares Earned, except to the extent that an award meeting the requirements of Section 2.5(e) (a "Replacement Award") is provided to the Participant in accordance with Section 2.5(e) to replace, adjust, or continue the award of Performance Shares covered by this Agreement (the "Replaced Award"). If a Replacement Award is provided, references to Performance Shares in this Agreement shall be deemed to refer to the Replacement Award after the Change in Control.
- (b) If, upon or after receiving a Replacement Award, the Participant experiences a termination of employment with the Company or a Subsidiary of the Company (or any of their successors) (as applicable, the "<u>Successor</u>") by reason of the Participant terminating employment for Good Reason or the Successor terminating the Participant's employment other than for Cause, in each case within a period of two years after the Change in Control and during the Incentive Period, 100% of the Replacement Award will become earned and nonforfeitable upon such termination.
- (c) If a Replacement Award is provided, notwithstanding anything in this Agreement to the contrary, any outstanding Performance Shares that at the time of the Change in Control are not subject to a "substantial risk of forfeiture" (within the meaning of Section 409A of the Code) will be deemed to be Performance Shares Earned at the time of such Change in Control and will be paid as provided for in Section 2.5(b).
 - (d) For purposes of this Agreement, the following terms have the following meanings:
- i. A "Replacement Award" means an award (A) of the same type (e.g., performance shares) as the Replaced Award, (B) that has a value at least equal to the value of the Replaced Award, (C) that relates to publicly traded equity securities of the Company or its successor in the Change in Control or another entity that is affiliated with the Company or its successor following the Change in Control, (D) if the Participant holding the Replaced Award is subject to U.S. federal income tax under the Code, the tax consequences of which to such Participant under the Code are not less favorable to such Participant than the tax consequences of the Replaced Award, and (E) the other terms and conditions of which are not less favorable to the Participant holding the Replaced Award than the terms and conditions of the Replaced Award (including the provisions that would apply in the event of a subsequent Change in Control). A Replacement Award may be granted only to the extent it does not result in the Replaced Award or Replacement Award failing to comply with or be exempt from Section 409A of the Code. Without limiting the generality of the foregoing, the Replacement Award may take the form of a continuation of the Replaced Award if the requirements of the two preceding sentences are satisfied. The determination of whether the conditions of this Section 2.4(d) are satisfied will be made by the Committee, as constituted immediately before the Change in Control, in its sole discretion.
- ii. A termination for "Cause" shall mean that, prior to termination of employment, the Participant shall have committed: (A) and been convicted of a criminal violation involving fraud, embezzlement or theft in connection with his or her duties or in the course of his or her employment with the Company or any Affiliate (or the Successor, if applicable); (B) intentional wrongful damage to property of the Company or any Affiliate (or the Successor, if applicable); or (D) intentional wrongful disclosure of secret processes or confidential information of the Company or any Affiliate (or the Successor, if applicable); or (D) intentional wrongful engagement in any competitive activity; and any such act shall have been demonstrably and materially harmful to the Company or any Affiliate (or the Successor, if applicable). For purposes of this Agreement, no act or failure to act on the part of the Participant shall be deemed "intentional" only if done or omitted to be done by the Participant not in good faith and without reasonable belief that the Participant's action or omission was in the best interest of the Company or an Affiliate (or the Successor, if applicable).
- iii. A termination "<u>for Good Reason</u>" shall mean the Participant's termination of employment with the Successor as a result of the initial occurrence, without the Participant's consent, of one or more of the following events:
 - (A) a material diminution in the Participant's annual base salary rate as in effect from time to time (" Base Pay");
 - (B) a material diminution in the Participant's authority, duties or responsibilities;

- (C) a material change in the geographic location at which the Participant must perform services;
- (D) a reduction in the Participant's opportunity regarding annual bonus, incentive or other payment of compensation, in addition to Base Pay, made or to be made in regard to services rendered in any year or other period pursuant to any bonus, incentive, profit-sharing, performance, discretionary pay or similar agreement, policy, plan, program or arrangement (whether or not funded) of the Successor; and
- (E) any other action or inaction that constitutes a material breach by the Participant's employer of the employment agreement, if any, under which the Participant provides services.

Notwithstanding the foregoing, "Good Reason" shall not be deemed to exist unless: (I) the Participant has provided notice to his or her employer of the existence of one or more of the conditions listed in (A) through (E) above within 90 days after the initial occurrence of such condition or conditions; and (II) such condition or conditions have not been cured by the Participant's employer within 30 days after receipt of such notice.

2.5 Payment of Performance Shares Earned.

- (a) <u>Payment After the Incentive Period</u>. Subject to Sections 2.5(b) and (c), the Performance Shares Earned shall be paid after the end of the Incentive Period and after the determination and certification by the Committee of the level of attainment of the Company Performance Objective(s), but in any event no later than 2-½ months after the end of the Incentive Period, to the extent not previously paid to the Participant.
- (b) <u>Change in Control</u>. Notwithstanding Section 2.5(a), to the extent there are any Performance Shares Earned as of a Change in Control, such Performance Shares Earned will be paid within 10 days of the Change in Control; <u>provided</u>, <u>however</u>, that if such Change in Control would not qualify as a permissible date of distribution under Section 409A(a)(2)(A) of the Code, and the regulations thereunder, and where Section 409A of the Code applies to such distribution, payment will be made on the date that would have otherwise applied pursuant to this Section 2.5.
- (c) <u>Payment Following a Change in Control</u>. Notwithstanding Sections 2.2 and 2.5(a), if, during the two-year period following a Change in Control, the Participant experiences a qualifying termination of employment (as described in Section 2.4(b)), the Performance Shares Earned as of the date of such termination of employment shall be paid **[in cash (pursuant to Section 2.5(d))]** within 10 days of the termination of employment to the extent they have not been previously paid to the Participant; <u>provided</u>, <u>however</u>, that if such Change in Control would not qualify as a permissible date of distribution under Section 409A(a)(2)(A) of the Code, and the regulations thereunder, and where Section 409A of the Code applies to such distribution, payment will be made on the date that would have otherwise applied pursuant to Section 2.5(a).
- (d) General. The Committee, in its sole discretion, may settle the Performance Shares Earned in cash or a combination of Shares and cash, in lieu of issuing only Shares. In the event that all or any portion of the Performance Shares Earned are paid in cash, the cash equivalent of one Performance Share Earned shall be equal to the Fair Market Value of one Share on the last trading day of the Incentive Period or, if earlier, the trading day immediately prior to the payment date. Notwithstanding the foregoing, no Performance Shares granted hereunder may be paid in cash in lieu of Shares to any Participant who is subject to the Share Ownership Guidelines unless and until such Participant is either in compliance with, or no longer subject to, such Share Ownership Guidelines; provided, however, that the Committee shall withhold Shares to the extent necessary to satisfy income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related item withholding requirements, as described in Section 5.3. In addition, the Committee may restrict 50% of the Shares to be issued in satisfaction of the total Performance Shares Earned, before income tax withholding, so that they cannot be sold by the Participant unless immediately after such sale the Participant is in compliance with the Share Ownership Guidelines that are applicable to the Participant at the time of sale.
- (e) <u>Payments After Death</u>. Any payment of Performance Shares Earned to a deceased Participant shall be paid to the estate of the Participant, unless the Participant files a completed Designation of Death Beneficiary with the Company in accordance with its procedures.
- (f) <u>Payment Obligation</u>. Prior to payment, the Company shall only have an unfunded and unsecured obligation to make payment of Performance Shares Earned to the Participant. The Performance Shares

covered by this Agreement that have not yet been earned as Performance Shares Earned, and any interests of the Participant with respect thereto, are not transferable other than pursuant to the laws of descent and distribution, or in accordance with Section 2.5(e).

ARTICLE 3 Other Terms and Conditions

3.1 Non-Compete and Confidentiality.

- (a) The Participant shall not render services for any organization or engage directly or indirectly in any business that is a competitor of the Company or any Affiliate of the Company, or which organization or business is or plans to become prejudicial to or in conflict with the business interests of the Company or any Affiliate of the Company or distribute any secret or confidential information belonging to the Company or any Affiliate of the Company.
- (b) Failure to comply with Section 3.1(a) above will cause the Participant to forfeit the right to Performance Shares and require the Participant to reimburse the Company for the taxable income received on Performance Shares that become payable to the Participant.

ARTICLE 4 Acknowledgments

- 4.1 Acknowledgments. In accepting the award, the Participant acknowledges, understands and agrees to the following:
 - (a) The Plan is established voluntarily by the Company, it is discretionary in nature and it may be modified, amended, suspended or terminated by the Company at any time, to the extent permitted by the Plan;
 - (b) The grant of the Performance Shares is voluntary and occasional and does not create any contractual or other right to receive future grants of Performance Shares, or benefits in lieu of Performance Shares, even if Performance Shares have been granted in the past;
 - (c) All decisions with respect to future Performance Shares or other grants, if any, will be at the sole discretion of the Company;
 - (d) The Participant's participation in the Plan is voluntary;
 - (e) The Performance Share award and the Participant's participation in the Plan shall not create a right to employment or be interpreted as forming an employment or services contract with the Company or any Subsidiary and shall not interfere with the ability of the Company, or any Subsidiary, as applicable, to terminate the Participant's employment or service relationship (if any);
 - (f) The future value of the underlying Shares is unknown, indeterminable and cannot be predicted with certainty;
 - (g) No claim or entitlement to compensation or damages shall arise from forfeiture of any Performance Shares resulting from the Participant ceasing to provide employment or other services to the Company or a Subsidiary (for any reason whatsoever whether or not later found to be invalid or in breach of employment laws in the jurisdiction where the Participant is employed or the terms of the Participant's employment agreement, if any), and in consideration of the grant of the Performance Shares to which the Participant is otherwise not entitled, the Participant irrevocably agrees never to institute any claim against the Company or any of its Subsidiaries, and the Participant waives his or her ability, if any, to bring any such claim, and releases the Company and its Subsidiaries from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, the Participant shall be deemed irrevocably to have agreed not to pursue such claim and agrees to execute any and all documents necessary to request dismissal or withdrawal of such claim;

- (h) Neither the Plan nor the Performance Shares shall be construed to create an employment relationship where any employment relationship did not otherwise already exist;
- (i) The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding the Participant's participation in the Plan, or the Participant's acquisition or sale of the underlying Shares. The Participant is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the Performance Shares;
- (j) The Performance Shares and the Shares subject to the Performance Shares, and the income and value of same, are not part of normal or expected compensation for purposes of calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments;
- (k) The Company reserves the right to impose other requirements on participation in the Performance Shares and on any Shares acquired under the Plan, to the extent the Company determines it is necessary or advisable in order to comply with local law or other applicable rules or facilitate the administration of the Plan, and to require the Participant to sign any additional agreements or undertakings that may be necessary to accomplish the foregoing; and
- (I) Notwithstanding anything in this Agreement to the contrary, the Participant acknowledges and agrees that this Performance Share Award, this Agreement and any related benefits or compensation under this Agreement are subject to the terms and conditions of the Company's clawback policy (if any) as may be in effect from time to time specifically to implement Section 10D of the Exchange Act and any applicable rules or regulations promulgated thereunder (including applicable rules and regulations of any national securities exchange on which the Common Shares may be traded) (the "Compensation Recovery Policy"), and that applicable provisions of this Agreement shall be deemed superseded by and subject to the terms and conditions of the Compensation Recovery Policy from and after the effective date thereof.

ARTICLE 5 General Provisions

- 5.1 <u>Compliance with Law.</u> The Company shall make reasonable efforts to comply with all applicable federal and state securities laws; provided, however, notwithstanding any other provision of the Agreement and these terms and conditions, the Company shall not be obligated to issue any Shares pursuant to the Agreement and these terms and conditions if the issuance or payment thereof would result in a violation of any such law; provided further, however, that the Shares will be issued at the earliest date at which the Company reasonably anticipates that the issuance of the Shares will not cause such violation. Notwithstanding anything in this Agreement to the contrary, nothing in this Agreement prevents the Participant from providing, without prior notice to the Company, information to governmental authorities regarding possible legal violations or otherwise testifying or participating in any investigation or proceeding by any governmental authorities regarding possible legal violations, and for purpose of clarity the Participant is not prohibited from providing information voluntarily to the Securities and Exchange Commission pursuant to Section 21F of the Exchange Act.
- **5 . 2** <u>Dividend Equivalents</u>. During the period beginning on the Date of Grant and ending on the date that Performance Shares are paid in accordance with Section 2.5, the Participant will be entitled to dividend equivalents on Performance Shares Earned equal to the cash dividend or distribution that would have been paid on the Performance Shares Earned had the Performance Shares Earned been issued and outstanding Shares on the record date for the dividend or distribution. Such accrued dividend equivalents (a) will vest and become payable upon the same terms and at the same time of settlement as the Performance Shares to which they relate, and (b) will be denominated and payable solely in cash.
- 5.3 <u>Withholding Taxes</u>. To the extent that the Company is required to withhold federal, state, local or foreign taxes in connection with any payment made or benefit realized by the Participant under this Agreement, the Company shall withhold Shares having a value equal to the amount required to be withheld. The Shares used for tax withholding will be valued at an amount equal to the fair market value on the date the benefit is to be included in the Participant's income. In no event will the market value of the Shares to be withheld and delivered pursuant to this

Section to satisfy applicable withholding taxes in connection with the benefit exceed the maximum amount of taxes that could be required to be withheld (subject to the Participant's estimated tax obligations attributable to the applicable transaction).

- 5.4 <u>Continuous Employment</u>. For purposes of this Agreement, the continuous employment of the Participant with the Company shall not be deemed to have been interrupted, and the Participant shall not be deemed to have separated from service with the Company, by reason of the transfer of his employment among the Company or Subsidiaries or an approved leave of absence, unless otherwise indicated in the Plan or if required to comply with Section 409A of the Code.
- 5.5 <u>Relation to Other Benefits</u>. Any economic or other benefit to the Participant under the Agreement and these terms and conditions or the Plan shall not be taken into account in determining any benefits to which the Participant may be entitled under any profit-sharing, retirement or other benefit or compensation plan maintained by the Company or a Subsidiary and shall not affect the amount of any life insurance coverage available to any beneficiary under any life insurance plan covering employees of the Company or Subsidiary.
 - 5.6 Adjustments. Restricted Stock Units evidenced by this Agreement are subject to mandatory adjustment as provided in Section 11 of the Plan.
- **5.7** These Terms and Conditions Subject to Plan. The Performance Shares covered under the Agreement and all of the terms and conditions hereof are subject to all of the terms and conditions of the Plan, a copy of which is available upon request.
- **5.8** <u>Transferability.</u> Except as otherwise provided in the Plan, the Performance Shares are non-transferable and any attempts to assign, pledge, hypothecate or otherwise alienate or encumber (whether by law or otherwise) any Performance Shares shall be null and void.
- 5.9 <u>Data Privacy</u>. The Participant hereby explicitly and unambiguously consents to the collection, use and transfer, in electronic or other form, of the Participant's personal data as described in this Agreement and any other Performance Share award materials by and among, as applicable, the Company or Subsidiaries for the exclusive purpose of implementing, administering and managing the Participant's participation in the Plan.

The Participant understands that the Company or Subsidiary may hold certain personal information about the Participant, including, but not limited to, the Participant's name, home address and telephone number, date of birth, social security number or other identification number, salary, nationality, job title, any Shares of or directorships in the Company that are held, details of all Performance Shares or any other entitlement to Shares awarded, canceled, exercised, vested, unvested or outstanding in the Participant's favor, for the exclusive purpose of implementing, administering and managing the Plan ("<u>Data</u>").

The Participant understands that Data will be transferred to the Company's broker, or such other stock plan service provider as may be selected by the Company in the future, which is assisting the Company with the implementation, administration and management of the Plan. The Participant understands that the recipients' use of the Data may be located in the United States or elsewhere, and that the recipients' country (e.g., the United States) may have different data privacy laws and protections than the Participant's country. The Participant understands that if he or she resides outside the United States, he or she may request a list with the names and addresses of any potential recipients of the Data by contacting his or her local human resources representative. The Participant authorizes the Company, the Company's broker and any other possible recipients which may assist the Company (presently or in the future) with implementing, administering and managing the Plan to receive, possess, use, retain and transfer the Data, in electronic or other form, for the sole purpose of implementing, administering and managing the Participants' participation in the Plan. The Participant understands that Data will be held only as long as is necessary to implement, administer and manage the Participants' participation in the Plan. The Participant understands if he or she resides outside the United States, he or she may, at any time, view their respective Data, request additional information about the storage and processing of their Data, require any necessary amendments to their Data or refuse or withdraw the consents herein, in any case without cost, by contacting in writing his or her local human resources representative. Further, the Participant understands that he or she is providing the consents herein on a purely voluntary basis. If the Participant later seeks to revoke his or her consent, his or her employment status or service and career with the Employer will not be adversely affected; the only adverse consequence of

withdrawing his or her consent may affect the Participant's ability to participate in the Plan. For more information on the consequences of the Participant's refusal to consent or withdrawal of consent, the Participant understands that he or she may contact his or her local human resources representative.

- **5.10** Amendments. This Agreement can be amended at any time by the Committee. Any amendment to the Plan shall be deemed to be an amendment to this Agreement to the extent that the amendment is applicable hereto. Except for amendments necessary to bring this Agreement into compliance with current law including Section 409A of the Code, no amendment to this Agreement shall materially and adversely affect the rights of the Participant without the Participant's written consent.
- **5.11** Severability. The provisions of this Agreement are severable and if any one or more provisions are determined to be illegal or otherwise unenforceable, in whole or in part, the remaining provisions shall nevertheless be binding and enforceable.
- **5.12** Electronic Delivery. The Company may, in its sole discretion, decide to deliver any documents related to the Performance Shares by electronic means. By accepting this Award of Performance Shares, the Participant hereby consents to receive such documents by electronic delivery and agrees to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.
- **5.13** Headings. Headings are given to the articles or sections of this Agreement solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Agreement or any provision hereof.
 - 5.14 Governing Law. This Agreement is governed by, and construed in accordance with the internal substantive laws of the State of Ohio.
- 5.15 Section 409A of the Code. To the extent applicable, it is intended that this Agreement and the Plan comply with the provisions of Section 409A of the Code. This Agreement and the Plan shall be administered in a manner consistent with this intent, and any provision that would cause the Agreement or the Plan to fail to satisfy Section 409A of the Code shall have no force and effect until amended to comply with Section 409A of the Code (which amendment may be retroactive to the extent permitted by Section 409A of the Code and may be made by the Company without the consent of the Participant). The terms "termination of employment," "terminates employment," and similar words and phrases used in this Agreement mean a "separation from service" within the meaning of Treasury Regulation section 1.409A-1(h). If, at the time of the Participant's separation from service (within the meaning of Section 409A of the Code), (a) the Participant will be a specified employee (within the meaning of Section 409A of the Code and using the identification methodology selected by the Company from time to time) and (b) the Company makes a good faith determination that an amount payable hereunder constitutes deferred compensation (within the meaning of Section 409A of the Code) the payment of which is required to be delayed pursuant to the six-month delay rule set forth in Section 409A of the Code in order to avoid taxes or penalties under Section 409A of the Code, then the Company will not pay such amount on the otherwise scheduled payment date but will instead pay it, without interest, on the fifth business day of the seventh month after such separation from service.

[Acceptance Page Contained in Exhibit D]

EXHIBITS

Exhibit A Exhibit B Exhibit C Exhibit D

Peer Group Performance Objectives Relative Total Shareholder Return Electronic Acceptance

Exhibit A

PEER GROUP

(2018 - 2020)

The Peer Group will be the constituents as defined by the SPDR S&P Metals and Mining ETF Index on the first day of trading of the Incentive Period.

The value of the stock of a Peer Group company will be determined in accordance with the following:

- 1. If the stock is listed on an exchange in the U.S. or Canada, then the value on such exchange will be used:
- Otherwise, if the stock is traded in the U.S. as an American Depositary Receipt ("ADR"), then the value of the ADR will be used;
- Otherwise, the value on the exchange in the country where the company is headquartered will be used.

Exhibit B

PERFORMANCE OBJECTIVES (TSR)

(2018 - 2020)

The Performance Objective of the Company is based on Relative Total Shareholder Return (share price plus reinvested dividends) over the three-year Incentive Period from January 1, 2018 to December 31, 2020. Achievement of the Relative Total Shareholder Return objective shall be determined by the Total Shareholder Return of the Company relative to that of the Peer Group, interpolating where necessary. Achievement shall be determined against the scale set forth in the table below:

	Performance Level							
Performance Factor	Below Threshold	Threshold	Target	Maximum				
Relative TSR	less than 25 th percentile	25 th percentile	50 th percentile	75 th or greater percentile				
Payout For Relative TSR	0%	50%	100%	200%				
		40						
		12						

Exhibit C

RELATIVE TOTAL SHAREHOLDER RETURN

(2018 - 2020)

Relative Total Shareholder Return for the Incentive Period is calculated as follows:

- 1. The Total Shareholder Return as defined in Section 1.8 of these terms and conditions for the Incentive Period for the Company shall be compared to the Total Shareholder Return for each of the entities within the Peer Group for the Incentive Period. The results shall be ranked to determine the Company's Relative Total Shareholder Return percentile ranking compared to the Peer Group.
- 2. The Company's Relative Total Shareholder Return for the Incentive Period shall be compared to the Relative Total Shareholder Return performance target range established for the Incentive Period.
- 3. The Relative Total Shareholder Return performance target range has been established for the 2018 2020 Incentive Period as follows:

2018 - 2020 Relative Total Shareholder Return

Performance Level Percentile Ranking

Maximum 75th Percentile
Target 50th Percentile
Threshold 25th Percentile

Exhibit D

ELECTRONIC ACCEPTANCE

Acceptance by the Participant

By selecting the "Accept Grant" box on the website of the Company's administrative agent, the Participant acknowledges acceptance of, and consents to be bound by, the Plan and this Agreement and any other rules, agreements or other terms and conditions incorporated herein by reference.

IF I FAIL TO ACKNOWLEDGE ACCEPTANCE OF THE AWARD WITHIN NINETY (90) DAYS OF THE DATE OF GRANT SET FORTH IN THE AGREEMENT, THE COMPANY MAY DETERMINE THAT THIS AWARD HAS BEEN FORFEITED.

PARTICIPANT NAME	ACCEPTANCE DATE
Participant Name	Date
ELECTRONIC SIGNATURE	
Participant Signature	

Product ID LTICATSR18

CLEVELAND-CLIFFS INC. AMENDED AND RESTATED 2015 EQUITY AND INCENTIVE COMPENSATION PLAN

CASH INCENTIVE AWARD MEMORANDUM (TSR)

Employee: PARTICIPANT NAME

Date of Grant: 2/21/2018

Target Amount of Cash Subject to Award: \$ Cash Granted

Performance Metric: Relative Total Shareholder Return Incentive Period: January 1, 2018 - December 31, 2020

Additional terms and conditions of your award are included in the Cash Incentive Award Agreement. As a condition to your receipt of this award, you must log on to Fidelity's website at www.netbenefits.fidelity.com and accept the terms and conditions of this award within 90 calendar days of your Date of Grant. If you do not accept the terms and conditions of this award within such time at www.netbenefits.fidelity.com, this award may be forfeited and immediately terminate.

<u>Note</u>: Section 3.1 of the Cash Incentive Award Agreement contains provisions that restrict your activities. These provisions apply to you and, by accepting this award, you agree to be bound by these restrictions.

CLEVELAND-CLIFFS INC. AMENDED AND RESTATED 2015 EQUITY AND INCENTIVE COMPENSATION PLAN

Cash Incentive Award Agreement (TSR)

This Cash Incentive Award Agreement (this " <u>Agreement</u>") is between Cleveland-Cliffs Inc., an Ohio corporation (the "<u>Company</u>"), and you, the person named in the Cash Incentive Award Memorandum (the "<u>Award Memorandum</u>") who is an employee of the Company or a Subsidiary of the Company (the "<u>Participant</u>"). For purposes of this Agreement, "<u>Employer</u>" means the entity (the Company or Subsidiary) that employs the Participant on the applicable date. This Agreement is effective as of the Date of Grant set forth in the Award Memorandum.

The Company wishes to award to the Participant the opportunity to earn an amount of cash, subject to the terms and conditions set forth in this Agreement, in order to carry out the purpose of the Cliffs Natural Resources Inc. Amended and Restated 2015 Equity and Incentive Compensation Plan (the "Plan"). All capitalized terms not defined in this Agreement shall have the same meaning as set forth in the Plan. See Section 2 of the Plan for a list of certain defined terms.

In the event of a conflict between the terms of this Agreement, the Award Memorandum and the terms of the Plan, the terms of the Plan shall govern. In the event of a conflict between the terms of this Agreement and the Award Memorandum, the terms of this Agreement shall govern.

ARTICLE 1 Definitions

All terms used herein with initial capital letters shall have the meanings assigned to them in the Plan and the following additional terms, when used herein with initial capital letters, shall have the following meanings:

- 1.1 "Earned Cash Incentive" shall mean the amount of cash earned by the Participant, as determined under Section 2.3.
- 1.2 "Incentive Period" shall be the time period as set forth in the Award Memorandum.
- 1.3 "Market Value Price" shall mean the latest available closing price of a Common Share of the Company or the latest available closing price per share of a common share of each of the entities in the Peer Group, as the case may be, on the New York Stock Exchange or other recognized market if the shares do not trade on the New York Stock Exchange at the relevant time.
- 1.4 "Peer Group" shall mean the group of companies, as more particularly set forth on attached Exhibit A, against which the Relative Total Shareholder Return of the Company is measured over the Incentive Period.
- 1.5 "Performance Objective(s)" shall mean for the Incentive Period the predetermined objectives of the Company with respect to the Management Objectives and any applicable goals established by the Committee and reported to the Board for this award, as more particularly set forth on attached Exhibit B.
- **1.6** "Relative Total Shareholder Return" shall mean for the Incentive Period the Total Shareholder Return of the Company compared to the Total Shareholder Return of the Peer Group, as more particularly set forth on attached Exhibit C.
- 1.7 "Total Shareholder Return" or "TSR" shall mean, for the Incentive Period, the cumulative return to shareholders of the relevant entity during the Incentive Period, measured by the change in Market Value Price per share of a common share of the entity plus dividends (or other distributions, excluding franking credits) reinvested over the Incentive Period, determined on the last business day of the Incentive Period compared to a base measured by the average Market Value Price per share of a common share of the entity on the last business day of the year immediately preceding the Incentive Period. Dividends (or other distributions, excluding franking credits) per share are assumed to be reinvested in the applicable stock on the last business day of the quarter during which they are paid at the then Market Value Price per share, resulting in a fractionally higher number of shares owned at the market price.

ARTICLE 2 Grant and Terms of Cash Incentive Award

- 2.1 Grant of Cash Incentive Award Opportunity. Pursuant to the Plan, the Company has granted to the Participant the opportunity to earn a percentage (from 0% to 200%) of the target amount of cash as specified in the Award Memorandum ("Cash Incentive Award"), effective as of the Date of Grant.
- 2.2 <u>Performance as Condition of Payment</u>. The Cash Incentive Award evidenced by this Agreement and these terms and conditions shall only result in the payment of cash to the extent such Cash Incentive Award has become an Earned Cash Incentive, as provided in Section 2.3, on the date the Earned Cash Incentive is to be paid as specified in Section 2.5.

2.3 Earned Cash Incentive.

- (a) Achievement of Performance Objective(s). Subject to Sections 2.3(b) and 2.3(c), the amount of Earned Cash Incentive, if any, shall be based upon the degree of achievement of the Performance Objective(s), all as more particularly set forth in Exhibit B, with the actual amount of the Earned Cash Incentive interpolated between the performance levels shown on Exhibit B, as determined and certified by the Committee as of the end of the Incentive Period. The percentage level of achievement determined for the Performance Objective(s) shall be multiplied by the target amount of cash subject to the Cash Incentive Award, as specified in the Award Memorandum, to determine the actual amount of Earned Cash Incentive, rounded down to the nearest whole cent. The calculation as to whether the Company has met or exceeded the Performance Objective(s) shall be determined and certified by the Committee in accordance with the award and these terms and conditions. Subject to the terms of the Plan, except as provided in Sections 2.3(b) and 2.3(c), no Cash Incentive Award will become an Earned Cash Incentive unless the Participant remains in the continuous employment of the Company or a Subsidiary during the entire Incentive Period.
- (b) <u>Death, Disability, Retirement or a Termination Without Cause</u>. If the Participant experiences a termination of employment with the Company because of the Participant's death, Disability (as defined herein) or Retirement (as defined herein) or a termination of employment by the Company without Cause (as defined herein) during the Incentive Period, the amount of the Participant's Cash Incentive Award that becomes an Earned Cash Incentive will be a prorated amount equal to the product of the amount determined after the end of the Incentive Period under Section 2.3(a) (without regard to the requirement that employment continue until the end of the Incentive Period), multiplied by a fraction, the numerator of which is the number of full months the Participant was employed with the Company or a Subsidiary between the start of the Incentive Period and the date of the Participant's termination of employment, and the denominator of which is 36, rounded down to the nearest whole cent.

For purposes of this Agreement, "Disability" shall mean a medically determinable physical or mental impairment that can be expected to result in death or can be expected to last for a continuous period of not less than 12 months and that results in the Participant: (i) being unable to engage in any substantial gainful activity; or (ii) receiving income replacement benefits for a period of not less than three months under an accident or health plan covering employees of the Company. For purposes of this Agreement, "Retirement" shall mean the Participant's retirement from active employment with the Company or Subsidiary upon or after the attainment of age 55 and at least a 5-year period of service with the Company and/or Subsidiary.

(c) <u>Change in Control</u>. In the event a Change in Control occurs during the Incentive Period, the Participant's Cash Incentive Award will become an Earned Cash Incentive only to the extent provided in Section 2.4.

In the event the Participant otherwise terminates employment prior to becoming entitled to an Earned Cash Incentive or the Participant's employment is terminated by the Company for Cause, the Participant shall forfeit all rights to any Cash Incentive Award evidenced by this Agreement.

2.4 Change in Control Vesting.

(a) If the Participant remains in the continuous employ of the Company or a Subsidiary throughout the period beginning on the Date of Grant and ending on the date of a Change in Control, upon the Change in Control, 100% of the Cash Incentive Award shall become an Earned Cash Incentive, except to the extent that an award meeting the requirements of Section 2.4(d) (a "Replacement Award") is provided to the Participant in accordance with Section

2.4(d) to replace, ad	just, or continue the	Cash Incentive Awa	ard evidenced by thi	s Agreement (the	" Replaced Award").	If a Replacement	Award is provided,
references to Cash Ir	ncentive Award in thi	s Agreement shall be	e deemed to refer to	the Replacement A	Award after the Chang	e in Control.	

- (b) If, upon or after receiving a Replacement Award, the Participant experiences a termination of employment with the Company or a Subsidiary of the Company (or any of their successors) (as applicable, the "Successor") by reason of the Participant terminating employment for Good Reason or the Successor terminating the Participant's employment other than for Cause, in each case within a period of two years after the Change in Control and during the Incentive Period, 100% of the Replacement Award will become earned and nonforfeitable upon such termination.
- (c) If a Replacement Award is provided, notwithstanding anything in this Agreement to the contrary, any portion of the outstanding Cash Incentive Award that at the time of the Change in Control is not subject to a "substantial risk of forfeiture" (within the meaning of Section 409A of the Code) will be deemed to be an Earned Cash Incentive at the time of such Change in Control and will be paid as provided for in Section 2.5(b).
- (d) For purposes of this Agreement, a "Replacement Award" means an award: (i) of the same type (e.g., performance-based cash award opportunity) as the Replaced Award; (ii) that has a value at least equal to the value of the Replaced Award; (iii) that is payable in cash; (iv) if the Participant holding the Replaced Award is subject to U.S. federal income tax under the Code, the tax consequences of which to such Participant under the Code are not less favorable to such Participant than the tax consequences of the Replaced Award; and (v) the other terms and conditions of which are not less favorable to the Participant holding the Replaced Award than the terms and conditions of the Replaced Award (including the provisions that would apply in the event of a subsequent Change in Control). A Replacement Award may be granted only to the extent it does not result in the Replaced Award or Replacement Award failing to comply with or be exempt from Section 409A of the Code. Without limiting the generality of the foregoing, the Replacement Award may take the form of a continuation of the Replaced Award if the requirements of the two preceding sentences are satisfied. The determination of whether the conditions of this Section 2.4(d) are satisfied will be made by the Committee, as constituted immediately before the Change in Control, in its sole discretion.
- (e) For purposes of this Agreement, a termination for "Cause" shall mean that, prior to termination of employment, the Participant shall have committed: (i) and been convicted of a criminal violation involving fraud, embezzlement or theft in connection with his or her duties or in the course of his or her employment with the Company or any Affiliate (or the Successor, if applicable); (iii) intentional wrongful damage to property of the Company or any Affiliate (or the Successor, if applicable); (iii) intentional wrongful disclosure of secret processes or confidential information of the Company or any Affiliate (or the Successor, if applicable); or (iv) intentional wrongful engagement in any competitive activity; and any such act shall have been demonstrably and materially harmful to the Company or any Affiliate (or the Successor, if applicable). For purposes of this Agreement, no act or failure to act on the part of the Participant shall be deemed "intentional" if it was due primarily to an error in judgment or negligence, but shall be deemed "intentional" only if done or omitted to be done by the Participant not in good faith and without reasonable belief that the Participant's action or omission was in the best interest of the Company or an Affiliate (or the Successor, if applicable).
- (f) A termination "for Good Reason" shall mean the Participant's termination of employment with the Successor as a result of the initial occurrence, without the Participant's consent, of one or more of the following events:
 - i. a material diminution in the Participant's annual base salary rate as in effect from time to time (" Base Pay");
 - ii. a material diminution in the Participant's authority, duties or responsibilities;
 - iii. a material change in the geographic location at which the Participant must perform services;
 - iv. a reduction in the Participant's opportunity regarding annual bonus, incentive or other payment of compensation, in addition to Base Pay, made or to be made in regard to services rendered in any year or other period pursuant to any bonus, incentive, profit-sharing, performance, discretionary pay or similar agreement, policy, plan, program or arrangement (whether or not funded) of the Successor; and

v. any other action or inaction that constitutes a material breach by the Participant's employer of the employment agreement, if any, under which the Participant provides services.

Notwithstanding the foregoing, "Good Reason" shall not be deemed to exist unless: (A) the Participant has provided notice to his or her employer of the existence of one or more of the conditions listed in (i) through (v) above within 90 days after the initial occurrence of such condition or conditions; and (B) such condition or conditions have not been cured by the Participant's employer within 30 days after receipt of such notice.

2.5 Payment of Earned Cash Incentive.

- (a) <u>Payment After the Incentive Period</u>. Subject to Sections 2.5(b) and (c), the Earned Cash Incentive shall be paid after the end of the Incentive Period and after the determination and certification by the Committee of the level of attainment of the Performance Objective(s), but in any event no later than 2-½ months after the end of the Incentive Period, to the extent it has not been previously paid to the Participant.
- (b) <u>Change in Control</u>. Notwithstanding Section 2.5(a), to the extent there is any Earned Cash Incentive as of a Change in Control, such Earned Cash Incentive will be paid within 10 days of the Change in Control; <u>provided</u>, <u>however</u>, that if such Change in Control would not qualify as a permissible date of distribution under Section 409A(a)(2)(A) of the Code, and the regulations thereunder, and where Section 409A of the Code applies to such distribution, payment will be made on the date that would have otherwise applied pursuant to this Section 2.5.
- (c) <u>Payment Following a Change in Control</u>. Notwithstanding Section 2.2 and 2.5(a), if, during the two-year period following a Change in Control, the Participant experiences a qualifying termination of employment (as described in Section 2.4(b)), the Earned Cash Incentive as of the date of such termination of employment shall be paid within 10 days of such termination of employment to the extent it has not been previously paid to the Participant; <u>provided</u>, <u>however</u>, that if such Change in Control would not qualify as a permissible date of distribution under Section 409A(a)(2)(A) of the Code, and the regulations thereunder, and where Section 409A of the Code applies to such distribution, payment will be made on the date that would have otherwise applied pursuant to this Section 2.5.
- (d) <u>General</u>. The Cash Incentive Award is to be settled in cash. The Committee may withhold cash to the extent necessary to satisfy income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related item withholding requirements, as described in Section 5.3.
- (e) <u>Payments After Death</u>. Any payment of Earned Cash Incentive to a deceased Participant shall be paid to the estate of the Participant, unless the Participant files a completed Designation of Death Beneficiary with the Company in accordance with its procedures.
- (f) <u>Payment Obligation</u>. Prior to payment, the Company shall only have an unfunded and unsecured obligation to make payment of the Earned Cash Incentive to the Participant. The Cash Incentive Award evidenced by this Agreement that has not yet been earned as an Earned Cash Incentive, and any interests of the Participant with respect thereto, are not transferable other than pursuant to the laws of descent and distribution, or in accordance with Section 2.5(e).

ARTICLE 3 Other Terms and Conditions

3.1 Non-Compete and Confidentiality.

- (a) The Participant shall not render services for any organization or engage directly or indirectly in any business that is a competitor of the Company or any Affiliate of the Company, or which organization or business is or plans to become prejudicial to or in conflict with the business interests of the Company or any Affiliate of the Company or distribute any secret or confidential information belonging to the Company or any Affiliate of the Company.
- (b) Failure to comply with Section 3.1(a) above will cause the Participant to forfeit the right to the Cash Incentive Award and require the Participant to reimburse the Company for the taxable income received as a result of the Cash Incentive Award within the 90-day period preceding the Participant's termination of employment.

ARTICLE 4 Acknowledgments

- 4.1 Acknowledgments. In accepting the award, the Participant acknowledges, understands and agrees to the following:
 - (a) The Plan is established voluntarily by the Company, it is discretionary in nature and it may be modified, amended, suspended or terminated by the Company at any time, to the extent permitted by the Plan;
 - (b) The grant of the Cash Incentive Award is voluntary and occasional and does not create any contractual or other right to receive future grants of Cash Incentive Awards, or benefits in lieu of Cash Incentive Awards, even if Cash Incentive Awards have been granted in the past;
 - (c) All decisions with respect to future Cash Incentive Awards or other grants, if any, will be at the sole discretion of the Company;
 - (d) The Participant's participation in the Plan is voluntary;
 - (e) The Cash Incentive Award and the Participant's participation in the Plan shall not create a right to employment or be interpreted as forming an employment or services contract with the Company or any Subsidiary and shall not interfere with the ability of the Company, or any Subsidiary, as applicable, to terminate the Participant's employment or service relationship (if any);
 - (f) No claim or entitlement to compensation or damages shall arise from forfeiture of any Cash Incentive Award resulting from the Participant ceasing to provide employment or other services to the Company or a Subsidiary (for any reason whatsoever whether or not later found to be invalid or in breach of employment laws in the jurisdiction where the Participant is employed or the terms of the Participant's employment agreement, if any), and in consideration of the grant of the Cash Incentive Award to which the Participant is otherwise not entitled, the Participant irrevocably agrees never to institute any claim against the Company or any of its Subsidiaries, and the Participant waives his or her ability, if any, to bring any such claim, and releases the Company and its Subsidiaries from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, the Participant shall be deemed irrevocably to have agreed not to pursue such claim and agrees to execute any and all documents necessary to request dismissal or withdrawal of such claim;
 - (g) Neither the Plan nor the Cash Incentive Award shall be construed to create an employment relationship where any employment relationship did not otherwise already exist;
 - (h) The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding the Participant's participation in the Plan, or the Participant's acquisition of cash thereunder. The Participant is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the Cash Incentive Award;
 - (i) The Cash Incentive Award, and the income and value of same, are not part of normal or expected compensation for purposes of calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments;
 - (j) The Company reserves the right to impose other requirements on participation in the Cash Incentive Award, to the extent the Company determines it is necessary or advisable in order to comply with local law or other applicable rules or facilitate the administration of the Plan, and to require the Participant to sign any additional agreements or undertakings that may be necessary to accomplish the foregoing; and

(k) Notwithstanding anything in this Agreement to the contrary, the Participant acknowledges and agrees that this Cash Incentive Award, this Agreement and any related benefits or compensation under this Agreement are subject to the terms and conditions of the Company's clawback policy (if any) as may be in effect from time to time specifically to implement Section 10D of the Exchange Act and any applicable rules or regulations promulgated thereunder (including applicable rules and regulations of any national securities exchange on which the Common Shares may be traded) (the "Compensation Recovery Policy"), and that applicable provisions of this Agreement shall be deemed superseded by and subject to the terms and conditions of the Compensation Recovery Policy from and after the effective date thereof.

ARTICLE 5 General Provisions

5 . 1 <u>Compliance with Law</u>. The Company shall make reasonable efforts to comply with all applicable federal and state securities laws. Notwithstanding anything in this Agreement to the contrary, nothing in this Agreement prevents the Participant from providing, without prior notice to the Company, information to governmental authorities regarding possible legal violations or otherwise testifying or participating in any investigation or proceeding by any governmental authorities regarding possible legal violations, and for purpose of clarity the Participant is not prohibited from providing information voluntarily to the Securities and Exchange Commission pursuant to Section 21F of the Exchange Act.

5.2 Reserved.

- 5.3 Withholding Taxes. To the extent that the Company is required to withhold federal, state, local or foreign taxes in connection with any payment made or benefit realized by the Participant under this Agreement, and the amounts available to the Company for such withholding are insufficient, it will be a condition to the receipt of such payment or the realization of such benefit that the Participant make arrangements satisfactory to the Company for payment of the balance of such taxes required to be withheld, which arrangements (in the discretion of the Committee) may include relinquishment of a portion of such benefit.
- 5.4 <u>Continuous Employment</u>. For purposes of this Agreement, the continuous employment of the Participant with the Company shall not be deemed to have been interrupted, and the Participant shall not be deemed to have separated from service with the Company, by reason of the transfer of his employment among the Company or Subsidiaries or an approved leave of absence, unless otherwise indicated in the Plan or if required to comply with Section 409A of the Code.
- 5.5 Relation to Other Benefits. Any economic or other benefit to the Participant under the Agreement and these terms and conditions or the Plan shall not be taken into account in determining any benefits to which the Participant may be entitled under any profit-sharing, retirement or other benefit or compensation plan maintained by the Company or a Subsidiary and shall not affect the amount of any life insurance coverage available to any beneficiary under any life insurance plan covering employees of the Company or Subsidiary.
 - 5.6 Adjustments. The Cash Incentive Award evidenced by this Agreement is subject to mandatory adjustment as provided in Section 11 of the Plan.
- 5.7 <u>These Terms and Conditions Subject to Plan</u>. The Cash Incentive Award covered under the Agreement and all of the terms and conditions hereof are subject to all of the terms and conditions of the Plan, a copy of which is available upon request.
- **5.8** <u>Transferability.</u> Except as otherwise provided in the Plan, the Cash Incentive Award is non-transferable and any attempts to assign, pledge, hypothecate or otherwise alienate or encumber (whether by law or otherwise) any portion of the Cash Incentive Award shall be null and void.
- **5.9** <u>Data Privacy</u>. The Participant hereby explicitly and unambiguously consents to the collection, use and transfer, in electronic or other form, of the Participant's personal data as described in this Agreement and any other Cash Incentive Award materials by and among, as applicable, the Company or Subsidiaries for the exclusive purpose of implementing, administering and managing the Participant's participation in the Plan.

The Participant understands that the Company or Subsidiary may hold certain personal information about the Participant, including, but not limited to, the Participant's name, home address and telephone number, date of birth, social security number or other identification number, salary, nationality, job title, any Common Shares of or directorships in the Company that are held, details of all Cash Incentive Awards awarded, canceled, vested, unvested or outstanding in the Participant's favor, for the exclusive purpose of implementing, administering and managing the Plan ("Data").

The Participant understands that Data will be transferred to the Company's broker, or such other stock plan service provider as may be selected by the Company in the future, which is assisting the Company with the implementation, administration and management of the Plan. The Participant understands that the recipients' use of the Data may be located in the United States or elsewhere, and that the recipients' country (e.g., the United States) may have different data privacy laws and protections than the Participant's country. The Participant understands that if he or she resides outside the United States, he or she may request a list with the names and addresses of any potential recipients of the Data by contacting his or her local human resources representative. The Participant authorizes the Company, the Company's broker and any other possible recipients which may assist the Company (presently or in the future) with implementing, administering and managing the Plan to receive, possess, use, retain and transfer the Data, in electronic or other form, for the sole purpose of implementing, administering and managing the Participants' participation in the Plan. The Participant understands that Data will be held only as long as is necessary to implement, administer and manage the Participant's participation in the Plan. The Participant understands if he or she resides outside the United States, he or she may, at any time, view their respective Data, request additional information about the storage and processing of their Data, require any necessary amendments to their Data or refuse or withdraw the consents herein, in any case without cost, by contacting in writing his or her local human resources' representative. Further, the Participant understands that he or she is providing the consents herein on a purely voluntary basis. If the Participant does not consent, or if the Participant later seeks to revoke his or her consent, his or her employment status or service and career with the Employer will not be adversely affected; the only adverse consequence of refusing or withdrawing the Participant's consent is that the Company would not be able to grant Cash Incentive Awards or equity awards or administer or maintain such awards. Therefore, the Participant understands that refusing or withdrawing his or her consent may affect the Participant's ability to participate in the Plan. For more information on the consequences of the Participant's refusal to consent or withdrawal of consent, the Participant understands that he or she may contact his or her local human resources representative.

- **5.10** <u>Amendments</u>. This Agreement can be amended at any time by the Committee. Any amendment to the Plan shall be deemed to be an amendment to this Agreement to the extent that the amendment is applicable hereto. Except for amendments necessary to bring this Agreement into compliance with current law including Section 409A of the Code, no amendment to this Agreement shall materially and adversely affect the rights of the Participant without the Participant's written consent.
- **5.11** Severability. The provisions of this Agreement are severable and if any one or more provisions are determined to be illegal or otherwise unenforceable, in whole or in part, the remaining provisions shall nevertheless be binding and enforceable.
- **5.12** Electronic Delivery. The Company may, in its sole discretion, decide to deliver any documents related to the Cash Incentive Award by electronic means. By accepting this Cash Incentive Award, the Participant hereby consents to receive such documents by electronic delivery and agrees to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.
- **5.13** Headings. Headings are given to the articles or sections of this Agreement solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Agreement or any provision hereof.
 - 5.14 Governing Law. This Agreement is governed by and construed in accordance with the internal substantive laws of the State of Ohio.
- 5.15 Section 409A of the Code. To the extent applicable, it is intended that this Agreement and the Plan comply with the provisions of Section 409A of the Code. This Agreement and the Plan shall be administered in a manner consistent with this intent, and any provision that would cause the Agreement or the Plan to fail to satisfy Section 409A of the Code shall have no force and effect until amended to comply with Section 409A of the Code (which amendment may be retroactive to the extent permitted by Section 409A of the Code and may be made by the Company without the consent of the Participant). The terms "termination of employment," "terminates employment," and similar

words and phrases used in this Agreement mean a "separation from service" within the meaning of Treasury Regulation section 1.409A-1(h). If, at the time of the Participant's separation from service (within the meaning of Section 409A of the Code), (a) the Participant will be a specified employee (within the meaning of Section 409A of the Code and using the identification methodology selected by the Company from time to time) and (b) the Company makes a good faith determination that an amount payable hereunder constitutes deferred compensation (within the meaning of Section 409A of the Code) the payment of which is required to be delayed pursuant to the six-month delay rule set forth in Section 409A of the Code in order to avoid taxes or penalties under Section 409A of the Code, then the Company will not pay such amount on the otherwise scheduled payment date but will instead pay it, without interest, on the fifth business day of the seventh month after such separation from service.

[Acceptance Page Contained in Exhibit D]

EXHIBITS

Exhibit A Exhibit B Exhibit C Exhibit D

Peer Group Performance Objectives Relative Total Shareholder Return Electronic Acceptance

Exhibit A

PEER GROUP (2018 - 2020)

The Peer Group will be the constituents as defined by the SPDR S&P Metals and Mining ETF Index on the first day of trading of the Incentive Period.

The value of the stock of a Peer Group company will be determined in accordance with the following:

- If the stock is listed on an exchange in the U.S. or Canada, then the value on such exchange will be used:
- Otherwise, if the stock is traded in the U.S. as an American Depositary Receipt ("ADR"), then the value of the ADR will be used; or
- Otherwise, the value on the exchange in the country where the company is headquartered will be used.

Exhibit B

PERFORMANCE OBJECTIVES (TSR)

(2018 - 2020)

The Performance Objective of the Company is based on Relative Total Shareholder Return (share price plus reinvested dividends) over the three-year Incentive Period from January 1, 2018 to December 31, 2020. Achievement of the Relative Total Shareholder Return objective shall be determined by the Total Shareholder Return of the Company relative to that of the Peer Group, interpolating where necessary. Achievement shall be determined against the scale set forth in the table below:

	Performance Level						
Performance Factor	Below Threshold	Threshold	Target	Maximum			
Relative TSR	less than 25th percentile	25 th percentile	50 th percentile	75 th or greater percentile			
Payout For Relative TSR	0%	50%	100%	200%			

Exhibit C

RELATIVE TOTAL SHAREHOLDER RETURN (2018-2020)

Relative Total Shareholder Return for the Incentive Period is calculated as follows:

- 1. The Total Shareholder Return as defined in Section 1.7 of these terms and conditions for the Incentive Period for the Company shall be compared to the Total Shareholder Return for each of the entities within the Peer Group for the Incentive Period. The results shall be ranked to determine the Company's Relative Total Shareholder Return percentile ranking compared to the Peer Group.
- 2. The Company's Relative Total Shareholder Return for the Incentive Period shall be compared to the Relative Total Shareholder Return performance target range established for the Incentive Period.
- 3. The Relative Total Shareholder Return performance target range has been established for the 2018 2020 Incentive Period as follows:

2018 - 2020 Relative Total Shareholder Return Percentile Ranking

75th Percentile

Performance Level

Maximum 50th Percentile

Target 25th Percentile

Threshold

13

Exhibit D

ELECTRONIC ACCEPTANCE

Acceptance by the Participant

By selecting the "Accept Grant" box on the website of the Company's administrative agent, the Participant acknowledges acceptance of, and consents to be bound by, the Plan and this Agreement and any other rules, agreements or other terms and conditions incorporated herein by reference.

IF I FAIL TO ACKNOWLEDGE ACCEPTANCE OF THE AWARD WITHIN NINETY (90) DAYS OF THE DATE OF GRANT SET FORTH IN THE AGREEMENT, THE COMPANY MAY DETERMINE THAT THIS AWARD HAS BEEN FORFEITED.

PARTICIPANT NAME	ACCEPTANCE DATE
Participant Name	Date
ELECTRONIC SIGNATURE	
Participant Signature	_

CERTIFICATION

I, Lourenco Goncalves, certify that:

- I have reviewed this quarterly report on Form 10-Q of Cleveland-Cliffs Inc.
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report:
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(f)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)), for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected or is reasonably likely to materially affect the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date:	4/24/2018	Ву:	/s/ Lourenco Goncalves
			Lourenco Goncalves
			Chairman, President and Chief Executive Officer

CERTIFICATION

I, Timothy K. Flanagan, certify that:

- I have reviewed this quarterly report on Form 10-Q of Cleveland-Cliffs Inc:
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report.
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(f)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)), for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected or is reasonably likely to materially affect the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date:	4/24/2018	Ву:	/s/ Timothy K. Flanagan
			Timothy K. Flanagan
			Executive Vice President, Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Cleveland-Cliffs Inc. (the "Company") on Form 10-Q for the period ended March 31, 2018, as filed with the Securities and Exchange Commission on the date hereof (the "Form 10-Q"), I, Lourenco Goncalves, Chairman, President and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to such officer's knowledge:

- (1) The Form 10-Q fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m or 78o(d)); and
- (2) The information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company as of the dates and for the periods expressed in the Form 10-Q.

Date: 4/24/2018

By: /s/ Lourenco Goncalves

Lourenco Goncalves

Chairman, President and Chief Executive Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Cleveland-Cliffs Inc. (the "Company") on Form 10-Q for the period ended March 31, 2018, as filed with the Securities and Exchange Commission on the date hereof (the "Form 10-Q"), I, Timothy K. Flanagan, Executive Vice President, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to such officer's knowledge:

- (1) The Form 10-Q fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m or 78o(d)); and
- (2) The information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company as of the dates and for the periods expressed in the Form 10-Q.

Date: 4/24/2018

By: /s/ Timothy K. Flanagan

Timothy K. Flanagan Executive Vice President, Chief Financial Officer

Mine Safety Disclosures

The operation of our mines located in the United States is subject to regulation by MSHA under the FMSH Act. MSHA inspects these mines on a regular basis and issues various citations and orders when it believes a violation has occurred under the FMSH Act. We present information below regarding certain mining safety and health citations that MSHA has issued with respect to our mining operations. In evaluating this information, consideration should be given to factors such as: (i) the number of citations and orders will vary depending on the size of the mine; (ii) the number of citations issued will vary from inspector to inspector and mine to mine, and (iii) citations and orders can be contested and appealed and, in that process, are often reduced in severity and amount, and are sometimes dismissed.

Under the Dodd-Frank Act, each operator of a coal or other mine is required to include certain mine safety results within its periodic reports filed with the SEC. As required by the reporting requirements included in §1503(a) of the Dodd-Frank Act, we present the following items regarding certain mining safety and health matters, for the period presented, for each of our mine locations that are covered under the scope of the Dodd-Frank Act:

- (A) The total number of violations of mandatory health or safety standards that could significantly and substantially contribute to the cause and effect of a coal or other mine safety or health hazard under section 104 of the FMSH Act (30 U.S.C. 814) for which the operator received a citation from MSHA;
- (B) The total number of orders issued under section 104(b) of the FMSH Act (30 U.S.C. 814(b));
- (C) The total number of citations and orders for unwarrantable failure of the mine operator to comply with mandatory health or safety standards under section 104(d) of the FMSH Act (30 U.S.C. 814(d));
- (D) The total number of imminent danger orders issued under section 107(a) of the FMSH Act (30 U.S.C. 817(a));
- (E) The total dollar value of proposed assessments from MSHA under the FMSH Act (30 U.S.C. 801 et seq.);
- (F) Legal actions pending before the Federal Mine Safety and Health Review Commission involving such coal or other mine as of the last day of the period;
- (G) Legal actions initiated before the Federal Mine Safety and Health Review Commission involving such coal or other mine during the period; and
- (H) Legal actions resolved before the Federal Mine Safety and Health Review Commission involving such coal or other mine during the period.

During the three months ended March 31, 2018, our U.S. mine locations did not receive any flagrant violations under section 110(b)(2) of the FMSH Act or any written notices of a pattern of violations, or the potential to have such a pattern of violations, under section 104(e) of the FMSH Act. In addition, there were no mining-related fatalities at any of our U.S. mine locations during this same period.

Following is a summary of the information listed above for the three months ended March 31, 2018:

			Three Months Ended March 31, 2018							
		(A)	(B)	(C)	(D)	(E)	(F)		(G)	(H)
Mine Name/ MSHA ID No.	Operation	Section 104 S&S Citations	Section 104(b) Orders	Section 104(d) Citations & Orders	Section 107(a) Orders	Total Dollar Value of MSHA Proposed Assessments (1)	Legal Actions Pending as of Last Day of Period		Legal Actions Initiated During Period	Legal Actions Resolved During Period
Tilden / 2000422	Iron Ore	14				\$ 250,467	9	(2)	3	1
Empire / 2001012	Iron Ore	_	_	_	_	_	_		_	_
Northshore Plant / 2100831	Iron Ore	42	_	_	_	38,920	6	(3)	4	_
Northshore Mine / 2100209	Iron Ore	1	_	_	_	818	_		_	_
Hibbing / 2101600	Iron Ore	7	_	2	_	18,241	2	(4)	2	_
United Taconite Plant / 2103404	Iron Ore	5	_	_	_	5,215	_		_	_
United Taconite Mine / 2103403	Iron Ore	1	_	_	_	1,844	_		_	_

- (1) Amounts included under the heading "Total Dollar Value of MSHA Proposed Assessments" are the total dollar amounts for proposed assessments received from MSHA for the three months ended March 31, 2018.
- (2) This number consists of 8 pending legal actions related to contests of proposed penalties referenced in Subpart C of FMSH Act's procedural rules, and 1 pending legal action related to complaints of discharge, discrimination, or interference referenced in Subpart E of FMSH Act's procedural rules.
- (3) This number consists of 6 pending legal actions related to contests of proposed penalties referenced in Subpart C of FMSH Act's procedural rules.
- (4) This number consists of 1 pending legal action related to contests of proposed penalties referenced in Subpart C of FMSH Act's procedural rules, and 1 pending legal action related to complaints of discharge, discrimination, or interference referenced in Subpart E of FMSH Act's procedural rules.